#### REPUBLIC OF SOUTH AFRICA

# DIVISION OF REVENUE BILL

(As introduced in the National Assembly (proposed section 76); explanatory summary of Bill published in Government Gazette No. 33959 of 24 January 2011)
(The English text is the official text of the Bill)

(MINISTER OF FINANCE)

[**B 4—2011**] ISBN 978-1-77037-797-4

## **BILL**

To provide for the equitable division of revenue raised nationally among the national, provincial and local spheres of government for the 2011/12 financial year and the responsibilities of all three spheres pursuant to such division; and to provide for matters connected therewith.

#### **PREAMBLE**

**WHEREAS** section 214(1) of the Constitution of the Republic of South Africa, 1996, requires an Act of Parliament to provide for—

- (a) the equitable division of revenue raised nationally among the national, provincial and local spheres of government;
- (b) the determination of each province's equitable share of the provincial share of that revenue; and
- (c) any other allocations to provinces, local government or municipalities from the national government's share of that revenue, and any conditions on which those allocations may be made,

**B**<sup>E</sup> IT THEREFORE ENACTED by the Parliament of the Republic of South Africa, as follows:—

#### ARRANGEMENT OF PROVISIONS OF ACT

Sections

#### CHAPTER 1

#### INTERPRETATION AND OBJECTS OF ACT

- 1. Interpretation
- 2. Objects of Act

### **CHAPTER 2**

## EQUITABLE SHARE ALLOCATIONS

10

- 3. Equitable division of revenue raised nationally among spheres of government
- 4. Equitable division of provincial share among provinces
- 5. Equitable division of local government share among municipalities
- 6. Shortfalls, excess revenue and additional allocations

## **CHAPTER 3**

## CONDITIONAL ALLOCATIONS TO PROVINCES AND MUNICIPALITIES

P	ar	t	1

	1	1 1	11	. •
Co	nditio	nai ai	toca	tions

7. 8.	Conditional allocations to provinces Conditional allocations to municipalities	5
	Part 2	
j	Duties of accounting officers in respect of Schedule 4 to 9 allocations	
9. 10. 11. 12. 13.	Duties of transferring national officer in respect of Schedule 4 allocations Duties of transferring national officer in respect of Schedule 5 to 8 allocations Duties of receiving officer in respect of Schedule 4 allocations Duties of receiving officer in respect of Schedule 5, 6, 8 or 9 allocations Duties in respect of annual financial statements and annual reports for 2011/12	10
	Part 3	
	Matters relating to Schedule 4 to 9 allocations	15
14. 15. 16. 17. 18. 19. 20.	Publication of allocations and frameworks Spending in terms of purpose and subject to conditions Withholding of allocation Stopping of allocation Re-allocation of funds Conversion of allocations Unspent conditional allocations	20
	Part 4	
	Matters relating to Schedule 1 to 9 allocations	
21. 22.	Payment schedule Amendment of payment schedule	25
	CHAPTER 4	
	MATTERS RELATING TO ALL ALLOCATIONS	
23. 24. 25. 26. 27.	Transfers made in error or fraudulently Allocations not listed in Schedules Transitional arrangement relating to re-demarcations of municipal boundaries Preparations for next financial year and 2013/14 financial year Expenditure prior to commencement of Division of Revenue Act, 2012	30
	CHAPTER 5	
D	UTIES OF MUNICIPALITIES, PROVINCIAL TREASURIES AND NATIONAL TREASURY	35
28. 29. 30.	Duties of municipalities Duties of provincial treasuries Duties of National Treasury	

## CHAPTER 6

GENERAL	
<ul> <li>31. Allocations by public entities to provinces or municipalities</li> <li>32. Liability for costs incurred in violation of principles of cooperative governance and intergovernmental relations</li> <li>33. Irregular expenditure</li> <li>34. Financial misconduct</li> <li>35. Delegations and assignments</li> <li>36. Exemptions</li> <li>37. Regulations</li> <li>38. Repeal of laws</li> </ul>	5
39. Short title and commencement	
SCHEDULE 1	
Equitable division of revenue raised nationally among the three spheres of government	15
SCHEDULE 2	
Determination of each province's equitable share of the provincial sphere's share of revenue raised nationally (as a direct charge against the National Revenue Fund)	
SCHEDULE 3	
Determination of each municipality's equitable share of the local government sphere's share of revenue raised nationally	20
SCHEDULE 4	
Allocations to provinces and municipalities to supplement the funding of programmes or functions funded from provincial or municipal budgets	
SCHEDULE 5	25
Specific purpose allocations to provinces	
SCHEDULE 6	
Specific purpose allocations to municipalities	
SCHEDULE 7	
Allocations-in-kind to provinces and municipalities for designated special programmes	30
SCHEDULE 8	
Incentives to provinces and municipalities to meet targets with regard to priority government programmes	

SCHEDULE 9

35

Provision to specifically cater for immediate release of funds to provinces and municipalities for disaster response

## **CHAPTER 1**

#### INTERPRETATION AND OBJECTS OF ACT

60

Interpretation	
1. (1) In this Act, unless the context indicates otherwise, any word or expression to which a meaning has been assigned in the Public Finance Management Act or the Municipal Finance Management Act has the meaning assigned to it in the Act in question, and—	5
"accreditation" means accreditation of a municipality, in terms of section 10(2) of the Housing Act, 1997 (Act No. 107 of 1997), to administer national housing programmes, read with Part 3 of the National Housing Code, 1999 (Financial Interventions: Accreditation of Municipalities); "category A, B or C municipality" has the meaning assigned to each category in	10
terms of the Municipal Structures Act; "conditional allocation" means a conditional allocation to a province, local government or municipality from the national government's share of revenue raised nationally, contemplated in section 214(1)(c) of the Constitution of the Republic of South Africa, 1996;	15
"corporation for public deposits account" means a bank account of the Provincial Revenue Fund held with the Corporation for Public Deposits, established by the Corporation for Public Deposits Act, 1984 (Act No. 46 of 1984); "Disaster Management Act" means the Disaster Management Act, 2002 (Act No.	20
57 of 2002); "financial year" means the financial year commencing on 1 April 2011 and ending on 31 March 2012; "framework" means the conditions and other information in respect of a conditional allocation published by the National Treasury in terms of section 14; "level one accreditation" means accreditation to render beneficiary management, subsidy budget planning and allocation, and priority programme management and	25
administration; "level three accreditation" means accreditation to render financial administration in addition to the responsibilities under a level one accreditation and a level two	30
accreditation;  "level two accreditation" means accreditation to render full programme management and administration of all housing instruments and housing programmes in addition to the responsibilities under a level one accreditation;  "Municipal Finance Management Act" means the Local Government: Municipal Finance Management Act, 2003 (Act No. 56 of 2003);  "Municipal Structures Act" means the Local Government: Municipal Structures	35
Act, 1998 (Act No. 117 of 1998); "municipal financial year" means the financial year of a municipality commencing on 1 July 2011 and ending on 30 June 2012; "next financial year" means the financial year commencing on 1 April 2012 and	40
ending on 31 March 2013; "next municipal financial year" means the financial year of a municipality commencing on 1 July 2012 and ending on 30 June 2013; "organ of state" means an organ of state as defined in section 239 of the	45
Constitution of the Republic of South Africa, 1996;  "payment schedule" means a schedule which sets out—  (a) the amount of each transfer of an equitable share or any conditional allocation in terms of this Act to be transferred to a province or municipality in the financial year;	50
(b) the date on which each transfer must be paid; and (c) to whom, and to which bank account, each transfer must be paid; "prescribe" means prescribe by regulation in terms of section 37; "primary bank account"—	55

(a) in relation to a province, means a bank account of the Provincial Revenue Fund held with a commercial bank which the head of the department in the provincial treasury has certified to the National Treasury as the bank account into which all conditional allocations, other than the Gautrain Rapid Rail Link Grant, in terms of this Act must be deposited; and

(b) in relation to a municipality, means the bank account of the municipality as determined in terms of section 8 of the Municipal Finance Management Act; "Public Finance Management Act," means the Public Finance Management Act,	
1999 (Act No. 1 of 1999); "quarter" means the period from—  (a) 1 April to 30 June;  (b) 1 July to 30 September;	5
<ul> <li>(b) 1 July to 30 September;</li> <li>(c) 1 October to 31 December; or</li> <li>(d) 1 January to 31 March;</li> <li>"receiving officer"—</li> </ul>	10
(a) in relation to a Schedule 4, 5, 8 or 9 allocation transferred to a province, means the accounting officer of the provincial department which receives that allocation or a portion thereof for spending via an appropriation from its Provincial Revenue Fund; or	10
(b) in relation to a Schedule 4, 6, 7, 8 or 9 allocation transferred to or provided in kind to a municipality, means the accounting officer of the municipality; "this Act" includes any framework or allocation published, or any regulation	15
made under this Act;  "transferring national officer" means the accounting officer of a national department that transfers a Schedule 4, 5, 6, 8 or 9 allocation to a province or municipality or spends a Schedule 7 allocation on behalf of a municipality.  (2) Any determination, instruction or request in terms of this Act must be in writing.	20
Objects of Act	
<ul><li>2. The objects of this Act are to—</li><li>(a) provide for the equitable division of revenue raised nationally among the three spheres of government;</li></ul>	25
<ul> <li>(b) promote predictability and certainty in respect of all allocations to provinces and municipalities, in order that provinces and municipalities may plan their budgets over a multi-year period and thereby promote better coordination between policy, planning and budgeting;</li> <li>(c) promote transparency and accountability in the resource allocation process, by ensuring that all allocations are reflected on the budgets of receiving provinces and municipalities and by ensuring that the expenditure of</li> </ul>	30
conditional allocations is reported on by the receiving provinces and municipalities.	35
CHAPTER 2	
EQUITABLE SHARE ALLOCATIONS	
Equitable division of revenue raised nationally among spheres of government	
<b>3.</b> (1) Revenue raised nationally in respect of the financial year must be divided among the national, provincial and local spheres of government for their equitable share allocations as set out in Column A of Schedule 1.	40
(2) An envisaged division of revenue anticipated to be raised in respect of the next financial year and the 2013/14 financial year, and which is subject to the provisions of the annual Division of Revenue Acts in respect of those financial years, is set out in Column B of Schedule 1.	45
Equitable division of provincial share among provinces	
<b>4.</b> (1) Each province's equitable share of the provincial share of revenue raised nationally in respect of the financial year is set out in Column A of Schedule 2. (2) An envisaged division for each province of revenue anticipated to be raised nationally in respect of the next financial year and the 2013/14 financial year, and which is subject to the provisions of the annual Division of Revenue Acts for those financial	50

(3) Each province's equitable share allocation contemplated in subsection (1) must be

years, is set out in Column B of Schedule 2.

transferred to the corporation for public deposits account of the province, in accordance with a payment schedule determined by the National Treasury in terms of section 21.

#### Equitable division of local government share among municipalities

- **5.** (1) Each municipality's share of local government's equitable share of revenue raised nationally in respect of the financial year, is set out in Column A of Schedule 3.
- (2) An envisaged division between municipalities of revenue anticipated to be raised nationally in respect of the next financial year and the 2013/14 financial year, and which is subject to the provisions of the annual Division of Revenue Acts for those financial years, is set out in Column B of Schedule 3.
- (3) Each municipality's equitable share contemplated in subsection (1) must be transferred to the primary bank account of the municipality in three transfers on 7 July 2011, 30 November 2011 and 23 March 2012, in accordance with a payment schedule 10 determined by the National Treasury in terms of section 21.

#### Shortfalls, excess revenue and additional allocations

**6.** (1) If actual revenue raised nationally in respect of the financial year falls short of the anticipated revenue set out in Schedule 1, the national government bears the shortfall.

15

20

25

30

- (2) If actual revenue raised nationally in respect of the financial year exceeds the anticipated revenue set out in Schedule 1, the excess accrues to the national government, subject to subsection (3), to be used to reduce borrowing or pay debt as part of its share of revenue raised nationally, in addition to its share in Column A of Schedule 1.
  - (3) The national government may—
    - (a) appropriate a portion of its equitable share or excess revenue contemplated in subsection (2) to make further allocations in an adjustments budget to—
      - (i) national departments; or
      - (ii) provinces or municipalities, as a conditional or an unconditional allocation; and
    - (b) increase a conditional allocation to a province or municipality through a virement under section 43 of the Public Finance Management Act or section 28(2)(d) of the Municipal Finance Management Act, as the case may be.

#### **CHAPTER 3**

#### CONDITIONAL ALLOCATIONS TO PROVINCES AND MUNICIPALITIES

#### Part 1

#### Conditional allocations

#### Conditional allocations to provinces

- **7.** (1) Conditional allocations to provinces in respect of the financial year from the national government's share of revenue raised nationally are set out in Column A of the 35 following Schedules:
  - (a) Schedule 4, specifying allocations to provinces to supplement the funding of programmes or functions funded from provincial budgets;
  - (b) Schedule 5, specifying specific-purpose allocations to provinces;
  - (c) Schedule 7, specifying allocations-in-kind to provinces for designated special 40 programmes;
  - (d) Schedule 8, specifying incentives to provinces to meet targets with regard to priority government programmes; and
  - (e) Schedule 9, specifying funds that are currently not allocated to specific provinces, that may be released to provinces to fund disaster response within 45 a period from three days up to three months following a declared disaster in terms of the conditions of the Disaster Management Act.
- (2) An envisaged division of conditional allocations to provinces from the national government's share of revenue anticipated to be raised nationally for the next financial year and the 2013/14 financial year, which is subject to the annual Division of Revenue 50 Acts for those years, is set out in Column B of the Schedules referred to in subsection (1).

#### Conditional allocations to municipalities

- **8.** (1) Conditional allocations to local government in respect of the financial year from the national government's share of revenue raised nationally are set out in Column A of the following Schedules:
  - (a) Schedule 4, specifying allocations to municipalities to supplement the funding of functions funded from municipal budgets;
  - (b) Schedule 6, specifying specific-purpose allocations to municipalities;
  - (c) Schedule 7, specifying allocations-in-kind to municipalities for designated special programmes;
  - (d) Schedule 8, specifying incentives to municipalities to meet targets with regard 10 to priority government programmes; and
  - (e) Schedule 9, specifying funds that are currently not allocated to specific municipalities, that may be released to local government or municipalities to fund disaster response within a period from three days up to three months following a declared disaster in terms of the conditions of the Disaster 15 Management Act.
- (2) An envisaged division of conditional allocations to local government from the national government's share of revenue anticipated to be raised nationally for the next financial year and the 2013/14 financial year, which, with the exception of what is provided in subsection (4) in relation to the Public Transport Infrastructure and Systems 20 Grant, is subject to the annual Division of Revenue Acts for those years, as set out in Column B of the Schedules referred to in subsection (1).
- (3) The National Treasury must, in terms of section 14, publish the share or indicative allocation of each municipality in respect of the local government allocations contemplated in subsections (1)(a) to (d) and (2) in the *Gazette*.

25

35

40

45

- (4) Despite subsection (2), in respect of the Public Transport Infrastructure and Systems Grant, funding which is specifically approved by the National Treasury in relation to transport contracts for capital projects must be regarded as being firm allocations for the next financial year and the 2013/14 financial year that will not be altered downwards in the Division of Revenue Acts in respect of those financial years.
- (5) A municipality may, only after obtaining the approval of the National Treasury, pledge, offer as security or commit to a person or institution future conditional allocation transfers due to the municipality for the next financial year and the 2013/14 financial year, for the purpose of securing a loan or any other form of financial or other support from that person or institution.

#### Part 2

#### Duties of accounting officers in respect of Schedule 4 to 9 allocations

#### Duties of transferring national officer in respect of Schedule 4 allocations

- 9. (1) The transferring national officer of a Schedule 4 allocation is responsible for—
  - (a) ensuring that transfers to all provinces and municipalities are—

    (i) deposited only into the primary bank account of a province
    - (i) deposited only into the primary bank account of a province or municipality; and
    - (ii) made in accordance with the payment schedule approved in terms of section 21, unless allocations are withheld or stopped in terms of section 16 or 17:
  - (b) monitoring expenditure and non-financial performance information on programmes funded by an allocation, in accordance with subsections (2) and (3) and the requirements of the applicable framework;
  - (c) complying with the requirements of the applicable framework; and
  - (d) evaluating the performance of programmes funded or partially funded by the sallocation and the submission of such evaluations to the National Treasury, within four months in respect of a province, and six months in respect of a municipality, after the end of the financial year.
- (2) (a) Subsection (1)(b) does not apply to the Urban Settlements Development Grant.
- (b) Municipalities that receive the Urban Settlements Development Grant must report 55 expenditure and non-financial performance information against their capital budgets, in accordance with section 11 of this Act and with the requirements of section 71 of the Municipal Finance Management Act.

- (c) Provinces that receive the Education Infrastructure Grant, the Health Infrastructure Grant or the Provincial Roads Maintenance Grant must report expenditure and non-financial performance information against their capital budgets, in accordance with section 11 of this Act and with the requirements of section 32 of the Public Finance Management Act.
- (d) Provinces that receive the Education Infrastructure Grant, the Health Infrastructure Grant and the Provincial Roads Maintenance Grant must provide the transferring national officer, not later than 14 days after this Act takes effect, with a list of infrastructure projects for education, health and roads which the provinces will implement during the 2011/12 to 2013/14 financial years, in a format that will be 10 determined by the National Treasury.
- (e) The transferring national officer must publish the information in terms of paragraph (d) in the *Gazette*, in a format that will be determined by the National Treasury, not later than 28 days after this Act comes into effect.
- (f) Failure to comply with paragraphs (d) and (e) may result in withholding of the 15 allocations in terms of section 16.
- (3) With respect to Schedule 4 Grants, any monitoring programme or system that is utilised to monitor expenditure and non-financial performance information funded by an allocation must.—
  - (i) be approved by the National Treasury;
  - (ii) not impose any excessive administrative responsibility on receiving provinces or receiving municipalities beyond the provision of standard management information:

25

30

40

- (iii) be compatible and integrated with and not unnecessarily duplicate other relevant and related national and provincial systems; and
- (iv) be consistent with section 11(2).

#### Duties of transferring national officer in respect of Schedule 5 to 8 allocations

- 10. (1) A transferring national officer must—
  - (a) not later than 14 days after this Act takes effect, certify to the National Treasury that—
    - (i) allocation frameworks, including conditions and monitoring provisions, are reasonable and do not impose excessive administrative responsibility on receiving provincial departments and municipalities beyond the provision of standard management information;
    - (ii) monitoring provisions are compatible and integrated with and do not 35 duplicate other relevant and related national, provincial and local systems;
    - (iii) in respect of a Schedule 5 allocation, any business plans requested in respect of how allocations will be utilised by a province have been approved prior to the start of the financial year;
    - (iv) in respect of a Schedule 6 allocation transferred to a municipality, any business plans requested in respect of how allocations will be utilised by a municipality have been approved prior to the start of the financial year;
  - (b) transfer funds only after information required in terms of this Act has been secured and all relevant information has been provided to the National 45 Treasury;
  - (c) transfer funds only in accordance with a payment schedule determined in accordance with section 21;
  - (d) deposit funds only into the primary bank account of a province or municipality, or, where appropriate, into the corporation for public deposits 50 account of a province or the bank account designed for the transfer of the Gautrain Rapid Rail Link Grant; and
  - (e) ensure that all other arrangements or requirements as stipulated in this Act or in the relevant framework for the particular allocation necessary for the transfer of an allocation have been complied with prior to the start of the 55 financial year, and are complied with throughout the financial year.
- (2) The transferring national officer must submit all relevant information and documentation referred to in subsection (1)(a) to the National Treasury within 14 days after this Act takes effect.

- (3) A transferring national officer who has not complied with subsection (1) must transfer the allocation in the manner instructed by the National Treasury, including transferring the allocation as an unconditional allocation.
- (4) Before making the first transfer of any allocation, the transferring national officer must take note of any notice in terms of section 30(1) from the National Treasury outlining the details of the account for each province or municipality.
- (5) Despite anything to the contrary contained in any law, a transferring national officer must in respect of any allocation, as part of the report contemplated in section 40(4)(c) of the Public Finance Management Act, not later than 20 days after the end of each month, and in the format determined by the National Treasury, submit to the National Treasury information for the month reported on and for the financial year up to the end of that month on—
  - (a) the amount of funds transferred to a province or municipality;
  - (b) the amount of funds withheld or stopped from any province or municipality, the reasons for the withholding or stopping and the steps taken by the transferring national officer and the receiving officer to deal with the matters or causes that necessitated the withholding or stopping of the payment;
  - (c) the actual expenditure incurred by the province or municipality in respect of a Schedule 5 or 6 allocation;
  - (d) the actual expenditure incurred by the transferring national officer in respect 20 of a Schedule 7 allocation;
  - (e) any matter or information that may be prescribed in the relevant framework for the particular allocation; and
  - (f) such other issues as the National Treasury may determine.
- (6) A transferring national officer must submit a quarterly performance report within 25 45 days after the end of each quarter to the National Treasury, in accordance with the requirements of the relevant framework.
- (7) The transferring national officer must evaluate the performance of programmes funded or partially funded by the allocation and submit such evaluations to the National Treasury, within four months in respect of a province, and six months in respect of a 30 municipality, after the end of the financial year.
- (8) The transferring national officer for the Human Settlements Development Grant, after consultation with the receiving officer and the National Treasury, must determine the allocations from the Human Settlements Development Grant for each municipality that receives the Urban Settlements Development Grant and submit those allocations to 35 the National Treasury not later than 14 days after this Act takes effect, and the National Treasury must publish those allocations in the *Gazette* within 14 days after receipt of the allocations.

#### Duties of receiving officer in respect of Schedule 4 allocations

- **11.** (1) A receiving officer of a Schedule 4 allocation is responsible for—
  - (a) complying with the framework for a Schedule 4 allocation as published in terms of section 14; and
  - (b) the manner in which a Schedule 4 allocation is allocated and spent.
- (2) The receiving officer of a municipality must—
  - (a) ensure and certify to the National Treasury that the municipality— 45
    - (i) indicates or, if required, exclusively appropriates each programme funded or partially funded by this allocation in its annual budget; and
    - (ii) makes public, in accordance with the requirements of section 21A of the Municipal Systems Act, the conditions and other information in respect of the allocation, to facilitate performance measurement and the use of 50 required inputs and outputs; and
  - (b) report to the transferring national officer, the relevant provincial treasury and the National Treasury, on—
    - (i) spending and financial performance against programmes funded by a Schedule 4 allocation, other than the Urban Settlements Development 55 Grant; and
    - (ii) in the case of the Urban Settlements Development Grant, spending and financial performance against the municipality's entire capital budget and performance against the targets stipulated in the infrastructure performance framework that is required in terms of the grant framework, 60

- when the municipality prepares its monthly budget statements as required in terms of section 71 of the Municipal Finance Management Act for the months of September 2011, December 2011, March 2012 and June 2012; and within 30 days after the end of each quarter, report to the transferring national officer and the National Treasury on—

  (i) non-financial performance against programmes, in respect of Schedule 4
- Grants other than the Urban Settlements Development Grant; and

  (ii) performance for that quarter against the targets stipulated in the infrastructure performance framework that is required in terms of the grant framework, in respect of the Urban Settlements Development 10 Grant.

15

20

35

- (3) The National Treasury must make the report submitted to it in terms of subsection (2)(b) or (c) available to any other national departments that have responsibilities relating to the grant.
  - (4) The receiving officer in a province must—
    - (a) submit, as part of the report required in section 40(4)(c) of the Public Finance Management Act, reports to the relevant provincial treasury on spending and performance against programmes; and
    - (b) submit a quarterly performance report within 30 days after the end of each quarter to the relevant provincial treasury, and the National Treasury.
- (5) The receiving officer must report against programmes funded or partially funded by a Schedule 4 allocation against the relevant framework in its annual financial statements and annual report.
- (6) (a) The receiving officer must, within two months after the end of the financial year, and where relevant, the municipal financial year, evaluate the performance of the province or municipality, as the case may be, in respect of programmes funded or partially funded by an allocation and submit such evaluation to the transferring national officer.
- (b) With respect to the Education Infrastructure Grant, the Health Infrastructure Grant, the Provincial Roads Maintenance Grant and the Urban Settlements Development Grant, reporting on performance will be based on the overall capital budget of the provincial department or the municipality concerned.

#### Duties of receiving officer in respect of Schedule 5, 6, 8 or 9 allocation

- **12.** (1) The receiving officer of a Schedule 5, 6, 8 or 9 allocation must ensure compliance with the requirements of the relevant framework.
- (2) The relevant receiving officer must, in respect of a Schedule 5, 6, 8 or 9 allocation transferred to—
  - (a) a province, as part of the report required in section 40(4)(c) of the Public Finance Management Act, report on the matters referred to in subsection (3) and submit a copy of the section 40(4)(c) report to the relevant provincial 40 treasury and the transferring national officer;
  - (b) a municipality, as part of the report required in terms of section 71 of the Municipal Finance Management Act, report on the matters referred to in subsection (4) and submit a copy of the section 71 report to the relevant provincial treasury, the National Treasury and the relevant transferring 45 national officer; and
  - (c) a province or a municipality, submit a quarterly performance report within 30 days after the end of each quarter to the transferring national officer, the relevant provincial treasury and the National Treasury.
- (3) A report by a province in terms of subsection (2)(a) must set out for that month and 50 for the financial year up to the end of that month—
  - (a) the amount received by the province;
  - (b) the amount of funds stopped or withheld from the province;
  - (c) the actual expenditure by the province in respect of a Schedule 5 allocation;
  - (d) the amount transferred to any national or provincial public entity to implement 55 a programme funded by a Schedule 5 allocation on behalf of a province or to assist the province in implementing such a programme;
  - (e) the most up to date available figures regarding the expenditure by a public entity referred to in paragraph (d);
  - (f) the extent of compliance with this Act and with the conditions of an allocation 60 provided for in a framework, based on the most up to date available information at the time of reporting;

- 12 (g) an explanation of any material problems experienced by the province regarding an allocation which has been received and a summary of the steps taken to deal with such problems; (h) any matter or information that may be prescribed in the relevant framework for the particular allocation; and 5 such other issues and information as the National Treasury may determine. (4) A report by a municipality in terms of subsection (2)(b) must set out for that month and for the financial year up to the end of that month-(a) the amount received by the municipality; (b) the amount of funds stopped or withheld from the municipality; 10 (c) the extent of compliance with this Act and with the conditions of an allocation or part of an allocation provided for in a framework; an explanation of any material problems experienced by the municipality regarding an allocation which has been received and a summary of the steps taken to deal with such problems; 15 (e) any matter or information that may be prescribed in the relevant framework for the particular allocation; and such other issues and information as the National Treasury may determine. (5) (a) Subsections (2) and (3) do not apply to the receiving officer of the Gautrain Rapid Rail Link Grant. 20 (b) The receiving officer of the Gautrain Rapid Rail Link Grant must, at the end of each quarter, submit a report to the transferring national officer, detailing the payment made in that quarter to meet its payment obligation in terms of the public-private partnership agreement entered into by the province in accordance with regulations issued under the Public Finance Management Act. 25 (c) Copies of payment certificates issued in terms of the public-private partnership agreement must be submitted together with the report referred to in paragraph (b). (6) A receiving officer must, within two months after the end of the financial year, and where relevant, the municipal financial year, evaluate its performance in respect of programmes or functions funded or partially funded by an allocation and submit such 30 evaluation to the transferring national officer. Duties in respect of annual financial statements and annual reports for 2011/12 13. (1) The 2011/12 financial statements of a national department transferring any funds in respect of an allocation set out in Schedule 4, 5, 6, 8 or 9 must, in addition to 35 any requirements in terms of any other applicable law— (a) indicate the total amount of that allocation transferred to a province or municipality: indicate the transfers, if any, that were withheld in respect of each province or municipality; indicate any re-allocations by the National Treasury in terms of section 18 or 40 by the transferring national officer in respect of the Expanded Public Works Programme Incentive Grant; (d) certify that all transfers to a province or municipality were deposited into the primary bank account of a province or municipality or, where appropriate, into the corporation for public deposits account of a province; and 45 (e) indicate the funds, if any, utilised for the administration of the allocation by the receiving officer. (2) The 2011/12 annual report of a national department transferring any funds in respect of an allocation set out in Schedule 4, 5, 6, 8 or 9 must, in addition to any requirements in terms of any other applicable law-50 (a) indicate the reasons for the withholding of any transfers to a province or municipality; (b) indicate to what extent provinces or municipalities were monitored for
- steps taken to deal with such non-compliance.
  (3) The 2011/12 financial statements of a provincial department receiving an allocation in terms of Schedule 4, 5, 8 or 9 must, in addition to any requirements in terms 60 of any other applicable law—

the relevant framework;

compliance with this Act and the conditions of an allocation provided for in

(c) indicate to what extent the allocation achieved its objectives and outputs; and(d) indicate any non-compliance with this Act or the relevant framework, and the

- 13 (a) indicate the total amount of all allocations received; (b) indicate the total amount of actual expenditure on all allocations except Schedule 4 allocations; and (c) certify that all transfers in terms of this Act to the province were deposited into the primary bank account of the province or, where appropriate, into the corporation for public deposits account of a province. (4) The 2011/12 annual report of a provincial department receiving an allocation in terms of Schedule 4, 5, 8 or 9 must, in addition to any requirements in terms of any other
- applicable law-(a) indicate to what extent the provincial department complied with the 10 provisions of this Act and met the conditions provided for in the relevant framework of such an allocation;
  - (b) indicate the steps taken to deal with non-compliance with any of the provisions of this Act or the conditions provided for in the relevant framework of such an allocation;

40

45

- (c) indicate the extent to which the objectives and outputs of the allocation were achieved;
- (d) contain any other information that may be specified in the relevant framework for the allocation; and
- (e) contain such other information as the National Treasury may determine. 20
- (5) The 2011/12 financial statements and annual report of a municipality must be prepared in accordance with the Municipal Finance Management Act.
- (6) The National Treasury may determine how transferring departments and receiving municipalities report on local government allocations on a quarterly basis, to facilitate the audit of allocations for both the national and municipal financial years. 25

#### Part 3

#### Matters relating to Schedule 4 to 9 allocations

#### Publication of allocations and frameworks

- 14. (1) The National Treasury must, within 14 days of this Act taking effect, publish in the Gazette— 30
  - (a) the allocations per municipality, in respect of Schedule 4 and 6 allocations;
  - (b) the indicative allocations per province and municipality, in respect of Schedule 7 allocations;
  - (c) the indicative allocations for provinces and municipalities, along with their initial threshold and performance targets, in respect of Schedule 8 allocations; 35
  - (d) the framework for each allocation in Schedules 4 to 9.
  - (2) The National Treasury must publish in the Gazette—
    - (a) any revisions or amendments of the allocations and frameworks published in terms of subsection (1)(a) and (d);
    - (b) any virement which is made in accordance with the requirements of section 6(3) and the Public Finance Management Act;
    - (c) any revisions of or amendments to the allocations and frameworks published in terms of subsection (1) necessary to give effect to the conversion of an allocation in terms of section 19;
    - any re-allocations by the National Treasury in accordance with section 18 or the transferring national officer in respect of the Expanded Public Works Programme Incentive Grant;
    - (e) any revised allocations in respect of Schedule 7 Grants; and
    - (f) any revised indicative allocations, performance and threshold targets for the 50 Expanded Public Works Programme Incentive Grant.
- (3) The National Treasury may at any time, after consultation with or at the written request of a transferring national officer, revise or amend a framework published in terms of subsection (1) or (2), to correct any error or omission.
- (4) An amendment, revision, virement or re-allocation takes effect on publication 55 thereof in the Gazette, with the exception of a Schedule 9 Grant.

#### Spending in terms of purpose and subject to conditions

- **15.** (1) Despite anything to the contrary contained in any law, an allocation referred to in Schedules 4 to 9 may only be utilised for the purpose stipulated in the Schedule concerned and in accordance with the framework published in terms of section 14.
- (2) A receiving officer may not transfer any Schedule 5 or 6 allocation or a portion of such an allocation to any other entity or other sphere of government for the performance of a function envisaged in terms of the allocation, unless the receiving officer has entered into a payment schedule with the entity or other sphere of government that will be performing the function, that has been approved by the National Treasury, and—
  - (a) it is a transfer that is approved in the budget of the receiving province or 10 municipality or a framework published in terms of section 14;
  - (b) it is a payment for services rendered or goods received, which services or goods were procured in accordance with the supply chain management policy or procurement policy of the relevant province or municipality and for which adequate documentation for payment has been received; or
  - (c) in the case of an advance payment or a transfer which is not consistent with the budget of the receiving province or municipality—
    - (i) the receiving officer has certified to the National Treasury that the transfer is not an attempt to artificially inflate its spending estimates and that there are good reasons for the advance payment or transfer; and

15

20

45

55

60

- (ii) the National Treasury has approved the advance payment or transfer.
- (3) No public entity, other than Eskom Holdings Limited in respect of funds received from the Department of Energy and water boards in respect of funds received from the Department of Water Affairs and the Department of Human Settlements for the implementation of Schedule 7 allocations, may receive funds for the provision of a 25 municipal service or municipal function on behalf of a municipality from a national or provincial organ of state, except via the municipality responsible for that service or function, unless the National Treasury approves otherwise in respect of municipalities it deems to have low capacity.
- (4) In respect of the Human Settlements Development Grant, a receiving officer and 30 a municipality with level three accreditation must, by 31 May 2011, comply with the requirements of subsection (2), by—
  - (a) entering into a payment schedule; and
  - (b) submitting the payment schedule to the National Treasury for approval.
- (5) A receiving officer must request the National Treasury to amend the payment 35 schedule referred to in subsection (4), in accordance with the procedure set out in section 22, in instances where a transfer to a municipality with level three accreditation has been withheld or stopped in terms of section 16 or 17.

#### Withholding of allocation

- **16.** (1) Subject to subsections (2) and (3), a transferring national officer may withhold 40 the transfer of a Schedule 4, 5 or 6 allocation, or any portion of such allocation, for a period not exceeding 30 days, if—
  - (a) the province or municipality does not comply with the provisions of this Act or conditions to which the allocation, as provided for in the relevant framework, is subject;
  - (b) roll-overs of conditional allocations approved by the National Treasury in accordance with section 20 have not been spent; or
  - (c) expenditure on previous transfers during the financial year reflects significant under-spending, for which no satisfactory explanation is given.
- (2) A maximum of five per cent of the Health Professions Training and Development 50 Grant and the National Tertiary Services Grant may be withheld in terms of this section.
- (3) A transferring national officer must, at least seven working days prior to withholding an allocation in terms of subsection (1)—
  - (a) give the relevant receiving officer—
    - (i) written notice of the intention to withhold the allocation; and
    - (ii) an opportunity to submit written representations as to why the allocation should not be withheld; and
  - (b) inform the relevant provincial treasury and the National Treasury, and in respect of any conditional grant to local government, also the provincial department responsible for local government.

- (4) A notice contemplated in subsection (3) must include the reasons for withholding the allocation and the intended duration of the withholding.
- (5) (a) The National Treasury may, when a transferring national officer is withholding an allocation in terms of subsection (1), instruct or approve a request from that transferring national officer to withhold an allocation for a period longer than 30 days, but not exceeding 120 days, if the withholding will—
  - (i) facilitate compliance with this Act or the conditions to which the allocation is subject; or
  - (ii) minimise the risk of under-spending.
- (b) A transferring national officer must, when requesting the withholding of an 10 allocation in terms of this subsection, submit proof of its compliance with subsection (3) and any representations received from the receiving officer, to the National Treasury.

20

35

(c) The transferring national officer must again comply with subsection (3) when the National Treasury instructs or approves a request by the transferring national officer in terms of paragraph (a).

#### Stopping of allocation

- 17. (1) Despite section 16, the National Treasury may, in its discretion or at the request of a transferring national officer, or a receiving officer, stop the transfer of a Schedule 4, 5 or 6 allocation referred to in section 16(1) to a province or municipality—
  - (a) on the grounds of persistent and material non-compliance with—
    - (i) this Act; or
    - (ii) a condition to which the allocation, as provided for in the relevant framework, is subject;
  - (b) if the National Treasury anticipates that a province or municipality will substantially under-spend on that programme or allocation in the financial 25 year;
  - (c) if a function is reassigned from a province to a municipality; or
  - (d) if a province implementing an infrastructure project does not comply with best practice standards and guidelines made in terms of section 5 of the Construction Industry Development Board Act, 2000 (Act No. 38 of 2000).
- (2) Except where a function is reassigned from a province to a municipality, the National Treasury must, when stopping an allocation in terms of this section—
  - (a) comply with the requirements set out in section 16(3)(a), and in respect of a municipality, also with the requirements of section 38 of the Municipal Finance Management Act; and
  - (b) inform the relevant provincial treasury of its intention to stop the allocation.
- (3) Any stopping of an allocation contemplated in subsection (1) must, together with an explanatory memorandum, be published by the National Treasury in the *Gazette*.
- (4) (a) The Minister may, by notice in the *Gazette*, approve that an allocation or any portion of such allocation stopped in terms of subsection (1), be utilised to meet that 40 province's or municipality's outstanding statutory and contractual financial commitments.
- (b) The utilisation of funds contemplated in this subsection is a direct charge against the National Revenue Fund.

#### Re-allocation of funds 45

- **18.** (1) (*a*) The National Treasury may, when it stops a Schedule 4, 5 or 6 allocation in terms of section 17, after consultation with the transferring national officer and the relevant provincial treasury, determine that a portion of the allocation that will not be spent be reallocated, as the same type of grant allocation as it was allocated originally, to one or more provinces or municipalities, on condition that the allocation will be spent in the financial year or the next financial year.
- (b) The reallocation of a portion of an allocation or the full allocation on condition that the allocation will be spent in the next financial year, in terms of paragraph (a), must be deemed to be a roll-over approved by the National Treasury in terms of section 20(2)(a), and the roll-over process set out in Treasury Regulation 6.4.2 would not need 55 to be followed.
- (2) (a) Despite subsection (1), the National Treasury may, when an intervention in terms of the Constitution or section 137, 139 or 150 of the Municipal Finance Management Act is taking place, on such conditions as it may determine, authorise—

- (i) in relation to section 100 of the Constitution, the transferring national officer to spend an allocation stopped in terms of section 17 on behalf of the relevant province;
- (ii) in relation to section 139 of the Constitution and sections 137 and 139 of the Municipal Finance Management Act, the intervening province to spend an allocation stopped in terms of section 17 on behalf of the relevant municipality; or
- (iii) in relation to section 150 of the Municipal Finance Management Act, the relevant transferring national officer to spend an allocation stopped in terms of section 17 on behalf of the relevant municipality.

50

- (b) An allocation that is spent by the transferring national officer or intervening province referred to in paragraph (a) must, for the purposes of this Act, be regarded as a Schedule 7 allocation from the date on which the authorisation is given.
- (3) The National Treasury may, after consultation with the transferring national officer, determine a portion of the allocation that will be reallocated in the event of the 15 reassignment of a function from a province to a municipality.

#### Conversion of allocations

- **19.** (1) The National Treasury may, in its discretion or at the request of the transferring national officer, convert an allocation listed in Schedule 6 to become an allocation listed in Schedule 7, or convert an allocation listed in Schedule 7 to become an allocation 20 listed in Schedule 6, if the National Treasury is satisfied that the conversion will prevent under-spending on the allocation.
- (2) In respect of the School Infrastructure Backlogs Grant, the National Treasury may, at the request of the transferring national officer and a receiving province, convert the allocation to the province to become an allocation to the Education Infrastructure Grant, 25 if the National Treasury and the transferring national officer are satisfied the province has demonstrated the capacity to implement projects and that the conversion will not be likely to result in—
  - (a) underspending on the allocation; or
  - (b) a lesser level of service delivery compared to if the allocation is provided to 30 the province as a Schedule 7 grant-in-kind.
- (3) An allocation that is converted in terms of this section must be paid to or expended on behalf of the same province or municipality to which the allocation was originally made.
- (4) A conversion referred to in subsections (1) and (2) takes effect on the date of 35 publication referred to in section 14(2)(c).
- (5) The National Treasury must inform the transferring national officer and each affected receiving province or municipality of a conversion.

#### **Unspent conditional allocations**

- **20.** (1) Despite the provisions of the Public Finance Management Act or the Municipal Finance Management Act relating to roll-overs, any conditional allocation, excluding the Gautrain Rapid Rail Link Grant and the Expanded Public Works Programme Incentive Grant, that is, in the case of a province, not spent at the end of a financial year or, in the case of a municipality, at the end of a municipal financial year, reverts to the National Revenue Fund, unless the relevant receiving officer can prove to the satisfaction of the National Treasury that the unspent allocation is committed to identifiable projects.
- (2) The National Treasury may, at the request of a transferring national officer, provincial treasury or municipality, approve—
  - (a) a roll-over from a conditional allocation to the next financial year; and
  - (b) spending of a portion of a conditional allocation on activities related to the purpose of that allocation, where the province or municipality projects significant unforeseeable and unavoidable over-spending on its budget.
- (3) Any funds which must revert to the National Revenue Fund in terms of subsection (1), and which have not been approved by the National Treasury to be retained in terms 55 of subsection (2), must be repaid to the National Revenue Fund.
- (4) The National Treasury, in accordance with subsection (5), may offset any funds which must be repaid to the National Revenue Fund in terms of subsections (1) and (3), but which have not been repaid—

- (a) in the case of a province, against future advances for conditional grant allocations to that province; and
- (b) in the case of a municipality, against future advances for the equitable share or conditional grant allocations to that municipality.
- (5) Prior to the National Treasury setting-off any amounts against allocations to provinces or municipalities in terms of subsection (4), the National Treasury must give the relevant transferring national officer, province or municipality—
  - (a) written notice of the intention to offset amounts against upcoming advances for allocations; and
  - (b) an opportunity, within 14 days of receipt of the notice referred to in paragraph 10 (a), to—
    - (i) submit written representations that prove to the satisfaction of the National Treasury that the unspent allocation was either spent in accordance with the relevant framework, or is committed to identifiable projects:
    - (ii) propose alternative means acceptable to the National Treasury by which the unspent allocations can be repaid to the National Revenue Fund; and

25

55

- (iii) propose an alternative payment schedule in terms of which the unspent allocations will be repaid to the National Revenue Fund.
- (6) A notice contemplated in subsection (5) must include the intended amount to be 20 offset against allocations, and the reasons for offsetting the amounts.
- (7) Despite this section, the retention of funds which should revert to the National Revenue Fund in terms of subsections (1) and (3), and which have not been approved by the National Treasury to be retained in terms of subsection (2), constitutes financial misconduct in terms of section 34.

#### Part 4

#### Matters relating to Schedule 1 to 9 allocations

## Payment schedule

- **21.** (1) (a) The National Treasury determines the payment schedule for the transfer of a province's equitable share allocation, after consultation with the head of the 30 department in the provincial treasury.
- (b) In determining the payment schedule, the National Treasury must take account of the monthly spending commitments of provinces and seek to minimise risk and debt servicing costs for national and provincial government.
- (c) Despite paragraph (a), the National Treasury may, for cash management purposes 35 relating to the corporation for public deposits account or when an intervention in terms of section 100 of the Constitution is taking place, on such conditions as it may determine, advance funds to a province in respect of its equitable share or a portion of it which has not yet fallen due for transfer in accordance with the payment schedule.
- (d) Any advances in terms of paragraph (c) must be offset against transfers to the 40 province which would otherwise become due in terms of that payment schedule.
- (2) (a) The National Treasury determines the payment schedule for the transfer of a municipality's equitable share allocation, after consultation with the accounting officer of the national department responsible for local government.
- (b) Despite paragraph (a), the National Treasury, after consultation with the 45 accounting officer of the national department responsible for local government, may, for cash management purposes in the municipality or when an intervention in terms of section 139 of the Constitution or section 137, 139 or 150 of the Municipal Finance Management Act is taking place, on such conditions as it may determine, approve a request or direct that the equitable share or a portion of the equitable share which has not yet fallen due for transfer in accordance with the payment schedule, be advanced to a municipality.
- (c) Any advances in terms of paragraph (b) must be offset against transfers to the municipality which would otherwise become due in terms of the applicable payment schedule.
- (3) (a) The National Treasury must approve the payment schedule for the transfer of an allocation listed in Schedules 4 to 9 to a province or municipality.

10	
(b) The transferring national officer of a Schedule 4, 5, 6 or 8 allocation must submit a payment schedule to the National Treasury for approval within 14 days after this Act	
takes effect.  (c) Prior to the submission of a payment schedule in terms of paragraph (b), the transferring national officer must—  (i) in relation to a Schedule 4 allocation, consult the relevant receiving officer;  (ii) in relation to the Gautrain Rapid Rail Link Grant, ensure that the payment schedule—	5
(aa) is consistent with the projected dates for payments to the private party in terms of the public-private partnership agreement entered into by the relevant province in accordance with regulations issued under the Public Finance Management Act; and	10
<ul><li>(bb) reflects the portion of any payments due under the agreement referred to in subparagraph (aa) payable from the allocation; and</li><li>(iii) in relation to a Schedule 5 or 6 allocation, consult the relevant province or municipality.</li></ul>	15
<ul><li>(4) The transferring national officer of a Schedule 4, 5 or 6 allocation must provide the receiving officer with a copy of the approved payment schedule prior to making the first transfer in accordance therewith.</li><li>(5) Schedule 4 and 6 transfers must be done in line with the municipal financial year.</li></ul>	20
	20
Amendment of payment schedule	
to the National Treasury prior to any further transfers being made.	25
(2) The National Treasury may, in the interest of better debt and cash-flow management or to deal with financial mismanagement or financial misconduct, slow spending or accelerated spending, amend any payment schedule for an allocation listed in Schedules 2 to 6, on notification to—  (a) the accounting officer of a provincial treasury, in the case of a provincial	30
allocation; and (b) the accounting officer of the national department responsible for local government, in the case of a local government allocation.  (3) A payment schedule amended in terms of subsection (1) or (2) must take account of—	35
<ul> <li>(a) the monthly spending commitments of provinces or municipalities;</li> <li>(b) the revenue at the disposal of provinces or municipalities; and</li> <li>(c) the minimisation of risk and debt servicing costs for all three spheres of government.</li> <li>(4) An amendment of a payment schedule in terms of subsection (2) prevails over any</li> </ul>	40
amendment made in terms of subsection (1).  (5) The transferring national officer must immediately inform the receiving officer of any amendment to a payment schedule in accordance with subsection (1) or (2).	
CHAPTER 4	45
MATTERS RELATING TO ALL ALLOCATIONS	
Transfers made in error or fraudulently	
23. (1) Despite anything to the contrary contained in any law, the transfer of an allocation to a province, municipality or public entity in error or fraudulently is regarded as not legally due to that province, municipality or public entity, as the case may be.  (2) A transfer contemplated in subsection (1) must be recovered, without delay, by the responsible transferring national officer.	50
(3) Despite subsection (2), the National Treasury may instruct that the recovery contemplated in subsection (2) be effected by set-off against future transfers to the province, municipality or public entity, which would otherwise become due in accordance with a payment schedule.	55

#### Allocations not listed in Schedules

- **24.** (1) An allocation, other than—
  - (a) a re-allocation referred to in section 18 or in respect of the Expanded Public Works Programme Incentive Grant; or
  - (b) a revised indicative allocation in respect of a Schedule 7 or 8 Grant, which is not listed in the Schedules referred to in sections 7 and 8,

10

15

20

45

may only be made in terms of section 6(3).

- (2) The National Treasury must publish the allocations referred to in subsection (1) and frameworks for such allocations in the *Gazette*, prior to the transfer of any funds to a province or municipality.
- (3) The national transferring officer must notify the relevant provincial treasury and the National Treasury of the excess incentive payment earned to be appropriated either in the provincial adjustments appropriation bill, if made before the end of November each year, or in the appropriation bill for the next financial year.
  - (4) Subsection (2) does not apply to Schedule 9 allocations.
- (5) A transfer to a province or municipality as a Schedule 9 allocation may only be made after the National Treasury has approved the transfer.
- (6) (a) The transferring national officer must notify the relevant provincial treasury and the National Treasury in writing within 14 days of a transfer of a Schedule 9 allocation to a province or municipality.
- (b) The National Treasury must publish a Schedule 9 allocation in the *Gazette* within 16 days after being notified in terms of paragraph (a).
- (7) Schedule 9 allocations must be appropriated or authorised either in the provincial adjustments appropriation legislation, municipal adjustments budgets or other appropriations legislation.

#### Transitional arrangement relating to re-demarcation of municipal boundaries

- **25.** (1) (a) A municipality (the releasing municipality) from which a particular area is transferred at the commencement of or through the implementation of legislation, which re-demarcates municipal boundaries and addresses associated matters, must continue to spend its allocations for the 2011/12 financial year made in terms of this Act, in that particular area or municipality, as if that area was not transferred to another municipality (the receiving municipality), unless the affected municipalities have entered into an agreement that ensures that the relocated area or municipality is not negatively affected.
- (b) The transferring national officer of an allocation made in terms of this Act and the receiving municipality must monitor that the releasing municipality complies with 35 paragraph (a).
- (c) The releasing municipality must, at the request of the transferring national officer, the receiving municipality or the National Treasury, demonstrate compliance with paragraph (a).
- (2) (a) The provisions of sections 16 and 17 apply with the necessary changes where 40 a releasing municipality fails to comply with subsection (1) (a) or (c) in respect of a Schedule 4, 6 or 7 allocation.
- (b) The National Treasury may, where it withholds or stops an allocation in terms of paragraph (a), after consultation with the transferring national officer, determine that a portion of the allocation be reallocated to the receiving municipality.
- (3) The National Treasury may, where a releasing municipality fails to comply with subsection (1) (a) or (c), reallocate a portion of the releasing municipality's equitable share allocation referred to in section 5 to the receiving municipality.
- (4) (a) The allocations referred to in sections 5 (2) and 8 (2) are subject to adjustments necessitated by the implementation of the legislation which re-demarcates the municipal 50 boundaries
- (b) The transferring national officer of a Schedule 4, 6 or 7 allocation must, by 15 September 2011, inform the National Treasury of any adjustments to the allocations referred to in section 8 (2) that must be reflected in the Division of Revenue Act for the next financial year.

#### Preparations for next financial year and 2013/14 financial year

**26.** (1) (a) A category C municipality that receives a conditional allocation in terms of this Act must, using the envisaged conditional allocations to that municipality for the

next financial year and the 2013/14 financial year as set out in Column B of the Schedules, by 3 October 2011—

- (i) agree on the provisional allocations and the projects to be funded from those allocations in the next financial year and the 2013/14 financial year with each category B municipality within the category C municipality's area of jurisdiction; and
- (ii) submit to the transferring national officer—
  - (aa) the provisional allocations referred to in subparagraph (i); and
  - (bb) the projects referred to in subparagraph (i), listed per municipality to be funded from the allocations for the next financial year and the 2013/14 10 financial year.
- (b) Where a category C municipality and a category B municipality cannot agree on the allocations and projects referred to in paragraph (a), the category C municipality must request the relevant transferring national officer to facilitate agreement.
- (c) The transferring national officer must take all necessary steps to facilitate 15 agreement as soon as possible, but no later than 60 days after receiving a request referred to in paragraph (b).
- (d) Any proposed amendment or adjustment of the allocations that is intended to be published in terms of section 29(3)(b) must be agreed with the relevant category B municipality, the transferring national officer and the National Treasury, prior to 20 publication, and prior to the submission of the allocations referred to in paragraph (a)(ii).
- (e) Should agreement not be reached between the category C municipality and the category B municipality on the provisional allocations and projects referred to in paragraph (a) prior to 3 October 2011, the National Treasury may proceed to determine 25 the provisional allocations and provide those provisional allocations to the municipalities concerned and the transferring national officer.
- (f) (i) The final allocations based on the provisional allocations referred to in paragraphs (a)(i) and (ii) and (e) must be submitted to the National Treasury by 7 December 2011.

30

40

- (ii) If the transferring national officer fails to submit the allocations referred to in subparagraph (i) by 7 December 2011, the National Treasury may determine the appropriate allocations, taking into consideration the envisaged allocations for the next financial year.
- (2) (a) The transferring national officer of a conditional allocation, using the 35 envisaged conditional allocations for the next financial year and the 2013/14 financial year as set out in Column B of the Schedules to this Act, must, by 3 October 2011, submit to the National Treasury for approval—
  - (i) the provisional allocations to each province or municipality in respect of new conditional allocations to be made in the next financial year;
  - (ii) any amendments to the envisaged allocations for each province or municipality set out in Column B of the Schedules in respect of existing conditional allocations; and
  - (iii) the draft frameworks for the allocations referred to in subparagraphs (i) and (ii) in the format to be determined by the National Treasury.
- (b) Any proposed amendment or adjustment for the next financial year of the allocation criteria of an existing conditional allocation must be agreed with the National Treasury prior to the submission of the provisional allocations and draft frameworks referred to in paragraph (a)(ii) and (iii).
- (c) The National Treasury may, if the transferring national officer fails to comply with 50 paragraph (a) by 3 October 2011, determine—
  - (i) the provisional allocations in paragraph (a)(i);
  - (ii) any amendments to the envisaged allocations contemplated in paragraph (a)(ii); and
  - (iii) the draft frameworks for the allocations referred to in paragraph (a)(iii), and 55 submit that information to the relevant provinces or municipalities.
- (d) (i) The final allocations based on the provisional allocations referred to in paragraph (a)(i) and (ii) must be submitted to the National Treasury by 7 December 2011.
- (ii) If the transferring national officer fails to submit the allocations referred to in 60 subparagraph (i) by 7 December 2011, the National Treasury may determine the appropriate allocations, taking into consideration the envisaged allocations for the next financial year.

(3) The National Treasury may, in preparation for the next financial year, instruct transferring national officers, receiving officers and municipalities to submit to it such plans and information for any conditional allocation as it may determine at specified times prior to the start of the next financial year.

#### Expenditure prior to commencement of Division of Revenue Act, 2012

27. Despite sections 3(2), 7(2) and 8(2), if the annual Division of Revenue Act for the next financial year has not commenced before or on 1 April 2012, the National Treasury may determine that an amount not exceeding 45 per cent of the total amount of each allocation made in terms of sections 3(1), 7(1) and 8(1) be transferred to the relevant province or municipality as a direct charge against the National Revenue Fund.

#### **CHAPTER 5**

#### DUTIES OF MUNICIPALITIES, PROVINCIAL TREASURIES AND NATIONAL TREASURY

#### **Duties of municipalities**

- **28.** (1) (a) In addition to the requirements of the Municipal Finance Management Act, 15 the accounting officer of a category C municipality must, no later than 14 days after this Act takes effect, submit to the National Treasury and all category B municipalities within that municipality's area of jurisdiction, the budget, as tabled in accordance with section 16 of the Municipal Finance Management Act, for the 2011/12 municipal financial year, and the two following municipal financial years.
- (b) The budget must indicate all allocations from its equitable share and conditional allocations to be transferred to each category B municipality within the category C municipality's area of jurisdiction and disclose the criteria for allocating funds between the category B municipalities.
- (2) A category C municipality that is providing a municipal service must, before implementing any capital project for water, electricity, roads or any other municipal service, consult the category B municipalities within whose area of jurisdiction the project will be implemented, and agree in writing which municipality is responsible for the operational costs and the collection of user fees.
- (3) A category C municipality must ensure that it does not duplicate a function 30 currently performed by a category B municipality and must transfer funds for the provision of services, including basic services, to the relevant category B municipality that is providing municipal services, irrespective of the fact that-
  - (a) the category C municipality retains the power or function in terms of the Municipal Structures Act; and
  - (b) a service delivery agreement for the provision of services by the category B municipality on behalf of the category C municipality has not been concluded.
- (4) A category B municipality which is not authorised to perform a function in terms of the Municipal Structures Act may not extend the scope or type of services that it currently provides, without—
  - (a) entering into a service delivery agreement with the category C municipality which is authorised to perform the function in terms of the Municipal Structures Act: or
  - (b) obtaining the legal authorisation to perform the function in terms of the Municipal Structures Act.
- (5) (a) A category C municipality and a category B municipality must, before the commencement of a municipal financial year, agree to a payment schedule in respect of the allocations referred to in subsection (1)(b) to be transferred to the category B municipality in that financial year, and the category C municipality must submit that payment schedule to the National Treasury before the commencement of a municipal 50 financial year.
- (b) A category C municipality must make transfers in accordance with the payment schedule submitted in terms of paragraph (a).
- (6) (a) The National Treasury may withhold or stop any allocation to the category C municipality and reallocate the allocation to the relevant category B municipalities if a 55 category C municipality fails to—
  - (i) make allocations referred to in subsection (1)(b);

10

5

20

35

40

- (ii) reach an agreement contemplated in subsection (2); or
- (iii) submit a payment schedule in accordance with subsection (5)(a).
- (b) Sections 16(3) and 17(3) and (4) of this Act and section 216 of the Constitution apply, with the necessary changes, to the withholding and stopping of an allocation in accordance with paragraph (a).

(c) The National Treasury may, where it stops an allocation in terms of this section, after consultation with the transferring national officer, determine that a portion of the allocation that will not be spent be reallocated to one or more municipalities, on condition that the allocation will be spent in the financial year or the next financial year.

(7) A municipality must ensure that any allocation made to it in terms of this Act, or 10 by a province or another municipality, that is not reflected in its budget as tabled in accordance with section 16 of the Municipal Finance Management Act, is reflected in its budget to be considered for approval in accordance with section 24 of the Municipal Finance Management Act.

(8) In respect of the Human Settlements Development Grant, the financial and 15 non-financial reporting that is required to be submitted by a municipality with a level three accreditation in terms of the grant framework must relate to the requirements specified in the grant framework.

(9) A municipality with a level three accreditation must submit the required monthly financial and quarterly performance reports to the receiving officer, the transferring 20 national officer and the National Treasury.

#### **Duties of provincial treasuries**

- **29.** (1) The provincial treasury must reflect Schedule 5 allocations separately in the province's appropriation bill or a schedule to its appropriation bill.
- (2) (a) The provincial treasury must, on the same day that its budget is tabled in the 25 provincial legislature, or a later date approved by the National Treasury, but not later than 14 days after this Act takes effect, publish the following in the *Gazette*:
  - (i) The indicative allocation per municipality for every allocation to be made by the province to municipalities from the province's own funds;
  - (ii) the indicative allocation to be made per school in the province;
  - (iii) the indicative allocation to any national or provincial public entity for the implementation of a programme funded by a Schedule 5 allocation on behalf of a province or for assistance provided to the province in implementing such a programme;
  - (iv) the envisaged division of the allocation contemplated in subparagraphs (i) and 35 (ii), in respect of each municipality and school, for the next financial year and the 2013/14 financial year;
  - (v) the conditions and other information in respect of the allocations referred to in subparagraphs (i), (ii) and (iii) to facilitate performance measurement and the use of required inputs and outputs; and
  - (vi) the budget of each hospital in a format determined by the National Treasury.
- (b) The allocations and budgets referred to in paragraph (a) must be deemed to be final if the legislature passes the appropriation bill without any amendments.
- (c) In the event that the legislature amends the appropriation bill, the accounting officer of the provincial treasury must publish amended allocations and budgets in the 45 *Gazette* within 14 days of the legislature passing the appropriation bill, which allocations must align to the appropriation bill as passed by the legislature, and which must be deemed to be final.
- (3) (a) Despite subsection (2) or any law, a provincial treasury may, in accordance with a framework determined by the National Treasury, amend the allocations referred to in subsection (2) or make additional allocations to municipalities that were not published in terms of subsection (1) or (2).
- (b) Any amendments to the allocations published in terms of subsection (2)(a) must be published in the *Gazette* not later than 14 February 2012.
- (c) The allocations referred to in paragraph (a) must be deemed to be final if they are 55 published in the *Gazette* in accordance with paragraph (b).
- (4) (a) A provincial treasury must, as part of its consolidated monthly report in terms of section 32 of the Public Finance Management Act, in the format determined by the National Treasury, report on—
  - (i) actual transfers received by the province from national departments;

5

30

(ii) actual expenditure on such allocations, excluding Schedule 4 allocations, up to the end of that month; and (iii) actual transfers made by the province to municipalities, and actual expendi-

ture by municipalities on such allocations, based on the most up to date information available from municipalities at the time of reporting.

- (b) The report contemplated in paragraph (a) must include reports for each quarter, and be in the format and include the information as may be determined by the National Treasury.
  - (5) A provincial treasury must—
    - (a) ensure-
      - (i) that a payment schedule is agreed between each provincial department and receiving institution referred to in subsection (2)(a); and
      - (ii) that transfers are made promptly in accordance therewith.
    - (b) submit the payment schedules to the National Treasury within 14 days of this Act taking effect.

(6) Should a provincial treasury fail to comply with the requirement to make transfers promptly to a receiving officer in accordance with the agreed payment schedule, as required in terms of subsection (5)(a), the receiving officer concerned may request the provincial treasury to immediately make the transfer to the receiving officer, or to provide written reasons within three business days as to why the transfer has not been 20 made.

(7) Should a provincial treasury fail to either make the transfer requested by the receiving officer or provide reasons within the period stipulated in subsection (6), or if the receiving officer disputes the reasons provided by the provincial treasury as to why the transfer has not been made, then the receiving officer may request the National 25 Treasury to investigate the matter with the provincial treasury concerned.

(8) When the National Treasury is requested to investigate a failure by a provincial treasury to promptly effect transfer to a receiving officer in terms of subsection (7), the National Treasury must investigate the matter, assess any reasons given by the provincial treasury as to why the transfer was not effected, and either direct the 30 provincial treasury to immediately effect the transfer, or provide reasons to the receiving officer concerned confirming why the provincial treasury was correct in not effecting the transfer, and advise the provincial treasury and the receiving officer as to what steps should be taken.

#### **Duties of National Treasury**

- **30.** (1) The National Treasury must, within 14 days of this Act taking effect, submit a notice to all transferring national officers, containing the details of the bank accounts of each province and municipality.
- (2) The National Treasury must, together with the monthly report contemplated in section 32(2) of the Public Finance Management Act, publish a report on actual transfers 40 of all allocations listed in the Schedules referred to in sections 7 and 8 or made in terms of section 24.
- (3) The National Treasury may, in any report it publishes that aggregates reports published by provincial treasuries contemplated in section 71(7) of the Municipal Finance Management Act, and in any report in respect of municipal finances, include a 45 report on the equitable share and conditional allocations provided for in this Act.

#### **CHAPTER 6**

#### **GENERAL**

#### Allocations by public entities to provinces or municipalities

**31.** The accounting officer of a provincial department or municipality that receives 50 funds from a public entity as a grant, sponsorship or donation, must disclose in its financial statements the purpose and amount of such grant, sponsorship or donation received.

10

15

# Liability for costs incurred in violation of principles of cooperative governance and intergovernmental relations

- **32.** (1) An organ of state involved in an intergovernmental dispute regarding any provision of this Act or any division of revenue matter or allocation must, before approaching a court to resolve such dispute, make every effort to settle the dispute with the other organ of state concerned, including exhausting all mechanisms provided for the settlement of disputes in relevant legislation.
- (2) In the event that a dispute is referred back by a court in accordance with section 41(4) of the Constitution, due to the court not being satisfied that the organ of state approaching the court has complied with subsection (1), the expenditure incurred by that 10
- organ of state in approaching the court must be regarded as fruitless and wasteful.

  (3) The amount of any such fruitless and wasteful expenditure must, in terms of a prescribed procedure, be recovered without delay from the person who caused the organ

#### Irregular expenditure

15

- **33.** The following transfers constitute irregular expenditure in terms of the Public Finance Management Act and the Municipal Finance Management Act, as the case may be:
  - (a) A transfer prohibited in terms of section 15(2);

of state not to comply with the requirements of subsection (1).

- (b) a transfer by a transferring national officer to a bank account of a province or 20 municipality that is not—
  - (i) the primary bank account;
  - (ii) in respect of provinces, a corporation for public deposits account; or
  - (iii) in respect of the Gautrain Rapid Rail Link Grant, the dedicated banking account configuration established for the transfer of the Gautrain Rapid Rail Link Grant in accordance with the directive issued by the National Treasury under section 10(2)(*a*) of the Division of Revenue Act, 2006 (Act No. 2 of 2006); or
- (c) any transfer made or spending of an allocation in contravention of this Act or a framework published in terms of this Act.

30

#### Financial misconduct

- **34.** (1) Despite anything to the contrary contained in any law, any serious or persistent non-compliance with a provision of this Act or a framework published in terms of this Act constitutes financial misconduct.
- (2) Section 84 of the Public Finance Management Act and section 171 of the 35 Municipal Finance Management Act apply in respect of financial misconduct contemplated in subsection (1).

## **Delegations and assignments**

- **35.** (1) The Minister may, in writing, delegate any of the powers entrusted to the National Treasury in terms of this Act and assign any of the duties imposed on the 40 National Treasury in terms of this Act, to an official of the National Treasury.
- (2) A delegation or assignment in terms of subsection (1) to an official of the National Treasury—
  - (a) is subject to any limitations or conditions that the Minister may impose;
  - (b) may authorise that official to subdelegate, in writing, the delegated power or 45 assigned duty to another National Treasury official; and
  - (c) does not divest the National Treasury of the responsibility concerning the exercise of the delegated power or the performance of the assigned duty.
- (3) The Minister may confirm, vary or revoke any decision taken by an official as a result of a delegation, subject to any rights that may have vested as a consequence of the decision.
- (4) A Member of the Executive Council responsible for financial matters in a province may, in writing, delegate any of the powers entrusted to the Provincial Treasury of that province in terms of this Act and assign any of the duties imposed on the Provincial Treasury in terms of this Act, to an official of the Provincial Treasury.

(5) Subsections (2) and (3) apply with the necessary changes to a delegation or assignment in terms of subsection (4).

#### **Exemptions**

- **36.** (1) The National Treasury may, on written application by a transferring national officer, province or municipality, exempt such officer, province or municipality in writing from complying with a provision of this Act.
- (2) Any exemption granted in terms of subsection (1) must set out the period and conditions, if any, to which it is subject and must be published in the *Gazette*.

#### Regulations

- **37.** The Minister may, by notice in the *Gazette*, make regulations regarding—
  - (a) anything which must or may be prescribed in terms of this Act; and
  - (b) any ancillary or incidental administrative or procedural matter that it is necessary to prescribe for the proper implementation or administration of this Act.

Repeal of laws 15

- **38.** (1) Subject to subsection (2), the Division of Revenue Act, 2010 (Act No. 1 of 2010), with the exception of sections 28(8) and 38(1), is hereby repealed.
- (2) The repeal of the Division of Revenue Act, 2010 (Act No. 1 of 2010), does not affect any duty or obligation set out in that Act, the execution of which is still outstanding.

20

#### Short title and commencement

**39.** This Act is called the Division of Revenue Act, 2011, and takes effect on 1 April 2011 or the date of publication thereof by the President in the *Gazette*, whichever is the later date.

SCHEDULE 1

EQUITABLE DIVISION OF REVENUE RAISED NATIONALLY AMONG
THE THREE SPHERES OF GOVERNMENT

	Column A Column B		mn B	
Spheres of Government	2011/12	Forward Estimates		
	Allocation	Allocation 2012/13		
	R'000	R'000	R'000	
National <sup>1,2</sup>	566 322 576	624 832 817	689 463 889	
Provincial	288 492 831	305 725 449	323 604 408	
Local	34 107 901	37 573 396	39 960 288	
TOTAL	888 923 308	968 131 662	1 053 028 585	

- 1. National share includes conditional allocations to provincial and local spheres, general fuel levy sharing with metropolitan municipalities, debt service cost and the contingency reserve.
- 2. The direct charges for the provincial equitable share are netted out.

#### **SCHEDULE 2**

# DETERMINATION OF EACH PROVINCE'S EQUITABLE SHARE OF THE PROVINCIAL SPHERE'S SHARE OF REVENUE RAISED NATIONALLY (as a direct charge against the National Revenue Fund)

	Column A	Column B		
Province	2011/12	Forward Estimates		
	Allocation	2012/13	2013/14	
	R'000	R'000	R'000	
Eastern Cape	44 120 028	46 495 024	48 931 829	
Free State	17 520 835	18 430 860	19 363 325	
Gauteng	50 428 480	53 973 066	57 699 363	
KwaZulu-Natal	62 927 556	66 877 612	70 992 966	
Limpopo	36 348 545	38 104 133	39 884 915	
Mpumalanga	23 378 714	24 570 021	25 786 476	
Northern Cape	7 742 909	8 207 056	8 688 325	
North West	19 271 431	20 562 274	21 912 710	
Western Cape	26 754 333	28 505 403	30 344 499	
TOTAL	288 492 831	305 725 449	323 604 408	

SCHEDULE 3

DETERMINATION OF EACH MUNICIPALITY'S EQUITABLE SHARE OF THE LOCAL GOVERNMENT SPHERE'S SHARE OF REVENUE RAISED NATIONALLY

			National Financial Year		
			Column A Column B		n B
			2011/12	Forward Estimates	
Nι	umber	Municipality	Allocation	2012/13	2013/14
			R'000	R'000	R'000
EAST	ERN CA	PE	K 000	K 000	N 000
A	BUF	Buffalo City	583 628	644 542	686 869
A	NMA	Nelson Mandela	656 653	730 416	778 401
В	EC101	Camdeboo	33 092	36 581	38 967
В	EC101	Blue Crane Route	34 998	38 693	41 218
В	EC103	Ikwezi	13 753	15 192	16 183
В	EC104	Makana	59 143	65 410	69 677
В	EC105	Ndlambe	49 522	54 813	58 398
В	EC106	Sundays River Valley	32 105	35 517	37 845
В	EC107	Baviaans	15 153	16 751	17 848
В	EC108	Kouga	41 038	45 585	48 593
В	EC109	Kou-Kamma	25 910	28 642	30 511
C	DC10	Cacadu District Municipality	67 220	69 666	72 317
Total:	Cacadu 1	Municipalities	371 935	406 850	431 557
В	EC121	Mbhashe	105 238	116 503	124 173
В	EC122	Mnquma	136 070	150 518	160 377
В	EC123	Great Kei	28 209	31 184	33 221
В	EC124	Amahlathi	79 616	87 995	93 737
В	EC126	Ngqushwa	55 140	60 965	64 949
В	EC127	Nkonkobe	76 099	84 004	89 451
В	EC128	Nxuba	18 627	20 591	21 935
C	DC12	Amatole District Municipality	530 281	583 693	627 182
Total:	Amatole	Municipalities	1 029 279	1 135 453	1 215 023
В	EC131	Inxuba Yethemba	35 770	39 535	42 109
В	EC131	Tsolwana	21 878	24 165	25 740
В	EC133	Inkwanca	15 748	17 399	18 534
В	EC134	Lukhanji	96 062	106 129	113 038
В	EC135	Intsika Yethu	78 404	86 613	92 256
В	EC136	Emalahleni	61 619	68 112	72 564
В	EC137	Engcobo	66 596	73 753	78 619
В	EC138	Sakhisizwe	34 165	37 799	40 278
C	DC13	Chris Hani District Municipality	325 908	360 052	384 758
Total:	Chris Ha	nni Municipalities	736 151	813 558	867 896
В	EC141	Elundini	64 855	71 762	76 475
В	EC142	Senqu	79 190	87 568	93 295
В	EC143	Maletswai	20 224	22 351	23 809
В	EC144	Gariep	22 167	24 507	26 107
C	DC14	Joe Gqabi District Municipality	149 031	164 710	175 859
Total:	Joe Gqa	bi Municipalities	335 467	370 898	395 545
В	EC153	Ngquza Hill	97 182	107 542	114 611
В	EC154	Port St Johns	59 537	65 862	70 188
В	EC155	Nyandeni	111 925	123 822	131 944
В	EC156	Mhlontlo	87 213	96 486	102 818
В	EC157	King Sabata Dalindyebo	153 086	169 585	180 725
С	DC15	O.R. Tambo District Municipality	416 223	450 392	480 874
Total:	O.R. Tar	nbo Municipalities	925 167	1 013 688	1 081 159
В	EC441	Matatiele	92 449	102 304	109 023
В	EC441 EC442	Umzimvubu	92 949	102 304	109 667
В	EC442 EC443	Mbizana	98 860	109 353	116 530
В	EC152	Ntabankulu	54 929	60 753	64 740
C	DC44	Alfred Nzo District Municipality	265 535	294 916	315 328
		zo Municipalities	604 767	670 232	715 289
Total:	Eastern	Cape Municipalities	5 243 046	5 785 638	6 171 740

SCHEDULE 3

DETERMINATION OF EACH MUNICIPALITY'S EQUITABLE SHARE OF THE LOCAL GOVERNMENT SPHERE'S SHARE OF REVENUE RAISED NATIONALLY

			Natio	National Financial Year		
			Column A	Colum	n B	
	Number	Montalealte	2011/12	Forward Es	timates	
I	Number	Municipality	Allocation	2012/13	2013/14	
			R'000	R'000	R'000	
FRE	E STATE					
A	MAN	Mangaung	546 417	603 089	642 590	
В	FS161	Letsemeng	45 212	49 982	53 241	
В	FS162	Kopanong	79 334	87 704	93 418	
В	FS163	Mohokare	45 632	50 451	53 741	
В	FS171	Naledi	33 288	36 803	39 204	
C	DC16	Xhariep District Municipality	20 629	22 002	23 067	
Tota	l: Xhariep	Municipalities	224 095	246 941	262 671	
В	FS181	Masilonyana	72 352	79 980	85 192	
В	FS182	Tokologo	38 552	42 614	45 393	
В	FS183	Tswelopele	55 333	61 187	65 182	
В	FS184	Matjhabeng	390 659	432 635	460 936	
В	FS185	Nala	120 920	133 667	142 372	
C	DC18	Lejweleputswa District Municipality	93 735	97 203	101 043	
Tota	l: Lejwelep	outswa Municipalities	771 551	847 286	900 117	
_		_				
В	FS191	Setsoto	147 875	163 504	174 167	
В	FS192	Dihlabeng	114 851	127 105	135 413	
В	FS193	Nketoana	69 567	76 929	81 951	
В	FS194	Maluti a Phofung	305 453	338 239	360 402	
В	FS195	Phumelela	49 899	55 167	58 766	
В	FS196	Mantsopa	59 517	37 002	70 096	
C	DC19	Thabo Mofutsanyana District Municipality	72 399	76 038	79 952	
Tota	l: Thabo N	Mofutsanyana Municipalities	819 560	873 983	960 747	
В	FS201	Moqhaka	145 181	160 452	170 892	
В	FS203	Ngwathe	137 311	151 789	161 675	
В	FS204	Metsimaholo	88 125	97 702	104 114	
В	FS205	Mafube	67 075	74 135	78 962	
C	DC20	Fezile Dabi	127 132	131 308	135 790	
		abi Municipalities	564 823	615 385	651 434	
Tota	l: Free Sta	te Municipalities	2 926 447	3 186 684	3 417 559	

SCHEDULE 3

DETERMINATION OF EACH MUNICIPALITY'S EQUITABLE SHARE OF THE LOCAL GOVERNMENT SPHERE'S SHARE OF REVENUE RAISED NATIONALLY

			Nati	National Financial Year		
			Column A	Colum	n B	
	N	Manistration	2011/12	Forward Estimates		
	Number	Municipality	Allocation	2012/13	2013/14	
			R'000	R'000	R'000	
GA	UTENG					
A	EKU	Ekurhuleni	1 644 128	1 828 391	1 949 038	
A	JHB	City of Johannesburg	1 897 561	2 134 780	2 276 247	
A	TSH	City of Tshwane	923 020	1 031 527	1 100 611	
В	GT421	Emfuleni	539 842	597 880	636 986	
В	GT422	Midvaal	44 379	49 238	52 480	
В	GT423	Lesedi	52 626	58 222	62 021	
C	DC42	Sedibeng District Municipality	220 439	227 627	233 903	
Tota	al: Sedibeng	g Municipalities	857 285	932 967	985 390	
В	GT481	Mogale City	189 605	210 233	224 025	
В	GT482	Randfontein	81 638	90 469	96 398	
В	GT483	Westonaria	87 796	97 063	103 392	
В	GT484	Merafong City	167 868	185 935	198 124	
C	DC48	West Rand District Municipality	163 221	168 881	173 109	
Tota	al: West Ra	nd Municipalities	690 128	752 581	795 048	
Tota	al: Gauteng	Municipalities	6 012 123	6 680 246	7 106 335	

SCHEDULE 3

DETERMINATION OF EACH MUNICIPALITY'S EQUITABLE SHARE OF THE LOCAL GOVERNMENT SPHERE'S SHARE OF REVENUE RAISED NATIONALLY

			Nati	National Financial Year		
			Column A	Colum	n B	
			2011/12	Forward Estimates		
	Number	Municipality	Allocation	2012/13	2013/14	
			R'000	R'000	R'000	
12337	AZIII II NIA	TAI				
KW	'AZULU-NA	MIAL				
A	ETH	eThekwini	1 595 941	1 773 889	1 891 834	
В	KZN211	Vulamehlo	31 146	34 452	36 712	
В	KZN212	Umdoni	24 104	26 677	28 423	
В	KZN213	Umzumbe	76 198	84 283	89 808	
В	KZN214	UMuziwabantu	38 288	42 368	45 149	
В	KZN215	Ezinqoleni	23 010	25 457	27 127	
В	KZN216	Hibiscus Coast	76 844	85 416	91 051	
C	DC21	Ugu District Municipality	235 738	260 342	278 485	
Tota	al: Ugu Mur	icipalities	505 328	558 995	596 755	
В	KZN221	uMshwathi	49 180	54 362	57 915	
В	KZN222	uMngeni	30 559	33 938	36 175	
В	KZN223	Mooi Mpofana	19 485	21 553	22 962	
В	KZN224	Impendle	21 087	23 335	24 868	
В	KZN225	Msunduzi	304 835	339 004	361 409	
В	KZN226	Mkhambathini	24 863	27 485	29 282	
В	KZN227	Richmond	25 849	28 573	30 440	
C	DC22	Umgungundlovu District Municipality	286 019	314 348	338 703	
Tota	al: Umgungı	ındlovu Municipalities	761 877	842 598	901 755	
В	KZN232	Emnambithi/Ladysmith	93 368	103 324	110 079	
В	KZN233	Indaka	53 002	58 621	62 458	
В	KZN234	Umtshezi	25 843	28 630	30 510	
В	KZN235	Okhahlamba	55 031	60 875	64 865	
В	KZN236	Imbabazane	57 268	63 321	67 462	
C	DC23	Uthukela District Municipality	229 471	253 506	270 913	
Tota	al:Uthukela	Municipalities	513 982	568 277	606 286	
В		Endumeni	27 416	30 369	32 368	
В	KZN242	*	62 785	69 485	74 053	
В	KZN244	e	61 218	67 731	72 182	
В	KZN245		36 785	40 689	43 354	
C	DC24	Umzinyathi District Municipality	160 451	177 410	189 546	
Tota	al: Umzinya	thi Municipalities	348 655	385 683	411 503	
_	WEDVOES	N d	244400	250 55	200 277	
В		Newcastle	244 400	270 655	288 355	
В		Emadlangeni	12 256	13 549	14 436	
В	KZ254	Dannhauser	42 091	46 545	49 591	
C	DC25	Amajuba District Municipality	88 571	97 343	104 815	
Tota	ai: Amajuba	Municipalities	387 318	428 091	457 198	

SCHEDULE 3

DETERMINATION OF EACH MUNICIPALITY'S EQUITABLE SHARE OF THE LOCAL GOVERNMENT SPHERE'S SHARE OF REVENUE RAISED NATIONALLY

			Na	National Financial Year		
			Column A	Column B		
2011/12		Forward E	stimates			
	Number	Municipality	Allocation	2012/13	2013/14	
			R'000	R'000	R'000	
В	KZN261	eDumbe	32 113	35 522	37 849	
В		UPhongolo	52 612	58 176	61 981	
В		Abaqulusi	69 224	76 515	81 515	
В		Nongoma	62 891	69 561	74 122	
В	KZN266	-	68 274	75 482	80 420	
C	DC26	Zululand District Municipality	234 326	258 934	276 726	
-		Municipalities	519 441	574 191	612 613	
		-				
В	KZN271	Umhlabuyalingana	49 682	54 977	58 595	
В	KZN272	Jozini	64 897	71 806	76 525	
В	KZN273	The Big Five False Bay	12 396	13 701	14 599	
В	KZN274	Hlabisa	38 543	24 972	26 589	
В	KZN275	Mtubatuba	42 041	46 763	49 926	
C	DC27	Umkhanyakude District Municipality	159 548	176 416	188 500	
Tot	al: Umkhan	yakude Municipalities	367 107	388 636	414 735	
В	KZN281	Mfolozi	37 931	41 898	44 633	
В	KZN282	uMhlathuze	161 654	179 231	190 994	
В	KZN283	Ntambanana	17 675	19 514	20 786	
В	KZN284	uMlalazi	69 515	76 827	81 845	
В	KZN285	Mthonjaneni	22 096	24 442	26 043	
В	KZN286	Nkandla	44 648	49 392	52 635	
C	DC28	Uthungulu District Municipality	299 178	328 816	354 199	
Tot	al: Uthungu	lu Municipalities	652 698	720 120	771 135	
В	K7N291	Mandeni	57 058	63 057	67 171	
В		KwaDukuza	65 237	72 636	77 471	
В		Ndwedwe	50 875	56 207	59 874	
В		Maphumulo	43 137	47 728	50 862	
C	DC29	iLembe District Municipality	206 729	228 084	244 242	
_		Municipalities	423 035	467 712	499 620	
В	KZN431	8	44 868	49 640	52 898	
В	KZN432	Kwa Sani	10 517	11 617	12 375	
В	KZN433	Greater Kokstad	42 946	47 560	50 680	
В	KZN434	Ubuhlebezwe	44 405	49 112	52 331	
В	KZN435	Umzimkhulu	75 003	82 979	88 423	
C	DC43	Sisonke District Municipality	182 881	202 113	215 785	
Tot	al: Sisonke I	Municipalities	400 620	443 021	472 492	
Tr.	-1. IZ 77 1	NI_4_1 Nf11_4	C 48C 004	7 151 011	F (3F 03)	
101	aı: KwaZult	ı-Natal Municipalities	6 476 001	7 151 211	7 635 926	

SCHEDULE 3

DETERMINATION OF EACH MUNICIPALITY'S EQUITABLE SHARE OF THE LOCAL GOVERNMENT SPHERE'S SHARE OF REVENUE RAISED NATIONALLY

			Nat	National Financial Year		
			Column A	Column B		
			2011/12	Forward Estimates		
	Number	Municipality	Allocation	2012/13	2013/14	
			R'000	R'000	R'000	
LIN	<b>ЛРОРО</b>					
В	LIM331	Greater Giyani	116 853	129 173	137 610	
В	LIM332		117 473	129 869	138 353	
В	LIM333	Greater Tzaneen	176 879	195 784	208 601	
В	LIM334	Ba-Phalaborwa	53 751	59 394	63 265	
В	LIM335	Maruleng	47 176	52 144	55 552	
С	DC33	Mopani District Municipality	419 718	463 587	495 550	
Tota	al: Mopani l	Municipalities	931 851	1 029 952	1 098 930	
В	LIM341	Musina	27 908	30 858	32 874	
В	LIM341 LIM342		27 908 37 628	41 594	32 874 44 316	
ь В		Thulamela	235 608			
_	LIM343			260 810	277 899	
В		Makhado	212 830	235 552	250 972	
C	DC34	Vhembe District Municipality	431 171	476 629	508 893	
Tota	ai: vnembe	Municipalities	945 145	1 045 443	1 114 955	
В	I IM251	Blouberg	79 413	87 909	93 693	
В	LIM351	Aganang	66 770	73 814	78 641	
В		Molemole	66 171	73 146	77 922	
В		Polokwane	350 705	388 419	413 878	
В			109 337	120 857	128 749	
		Lepelle-Nkumpi				
C	DC35	Capricorn District Municipality	365 229 1 037 625	402 266 1 146 411	431 885 1 224 768	
100	ai: Capricor	n Municipalities	1 03/ 625	1 146 411	1 224 /08	
В	LIM361	Thabazimbi	53 095	58 755	62 602	
В		Lephalale	73 300	79 669	84 829	
В	LIM364	Mookgopong	22 822	25 260	26 918	
В	LIM365	Modimolle	50 075	55 376	58 990	
В	LIM366	Bela-Bela	39 790	44 012	46 885	
В	LIM367	Mogalakwena	225 142	249 257	265 591	
C	DC36	Waterberg District Municipality	87 880	91 087	94 101	
-		rg Municipalities	552 105	603 417	639 917	
В	LIM471	Ephraim Mogale	60 529	66 895	71 260	
В	LIM472	Elias Motsoaledi	114 134	126 125	134 349	
В	LIM473	Makhuduthamaga	126 339	139 682	148 812	
В	LIM474	Fetakgomo	40 562	44 840	47 774	
В	LIM475	Greater Tubatse	114 137	126 220	134 480	
C	DC47	Greater Sekhukhune District Municipality	330 877	365 664	390 926	
Tota	al: Greater S	Sekhukhune Municipalities	786 577	869 427	927 602	
Tot	al: Limnono	Municipalities	4 253 303	4 694 650	5 006 171	
100	ат. сппроро	virumcipanues	4 255 305	4 094 050	5 UUO 1/1	

SCHEDULE 3

DETERMINATION OF EACH MUNICIPALITY'S EQUITABLE SHARE OF THE LOCAL GOVERNMENT SPHERE'S SHARE OF REVENUE RAISED NATIONALLY

			Nat	National Financial Year		
			Column A	Column B		
Ι,	NT I	Manistration	2011/12	Forward Estimates		
I	Number	Municipality	Allocation	2012/13	2013/14	
			R'000	R'000	R'000	
MPU	JMALANG	<b>S</b> A				
В	MP301	Albert Luthuli	141 281	156 282	166 510	
В	MP302	Msukaligwa	93 142	103 000	109 722	
В	MP303	Mkhondo	88 732	98 160	104 590	
В	MP304	Pixley Ka Seme	74 975	82 907	88 318	
В	MP305	Lekwa	69 959	77 364	82 412	
В	MP306	Dipaleseng	39 319	43 465	46 299	
В	MP307	Govan Mbeki	171 429	190 045	202 507	
C	DC30	Gert Sibande District Municipality	246 282	254 200	261 625	
Tota	l: Gert Sib	ande Municipalities	925 119	1 005 423	1 061 984	
В	MP311	Victor Khanye	45 078	49 867	53 124	
В	MP312	Emalahleni	163 854	181 726	193 663	
В	MP313	Steve Tshwete	77 312	85 857	91 509	
В	MP314	Emakhazeni	31 562	34 885	37 159	
В	MP315	Thembisile	196 665	217 431	231 624	
В	MP316	Dr JS Moroka	205 518	227 206	242 036	
C	DC31	Nkangala District Municipality	291 974	301 317	309 985	
Tota	l: Nkangal	a Municipalities	1 011 962	1 098 290	1 159 101	
В	MP321	Thaba Chweu	66 696	73 750	78 563	
В	MP322	Mbombela	282 081	312 447	332 931	
В	MP323	Umjindi	43 164	47 758	50 884	
В	MP324	Nkomazi	234 566	259 743	276 789	
В	MP325	Bushbuckridge	398 491	440 908	469 771	
C	DC32	Ehlanzeni District Municipality	170 414	178 778	185 817	
Tota	ıl: Ehlanzer	ni Municipalities	1 195 411	1 313 385	1 394 754	
Tota	l: Mnumal	anga Municipalities	3 132 492	3 417 098	3 615 839	

SCHEDULE 3

DETERMINATION OF EACH MUNICIPALITY'S EQUITABLE SHARE OF THE LOCAL GOVERNMENT SPHERE'S SHARE OF REVENUE RAISED NATIONALLY

			Nat	National Financial Year		
			Column A	Column B		
			2011/12	Forward E	stimates	
	Number	Municipality	Allocation	2012/13	2013/14	
			R'000	R'000	R'000	
NOI	RTHERN C	'APE				
1,02	KTIIDIK! (					
В	NC061	Richtersveld	10 415	11 507	12 257	
В	NC062	Nama Khoi	28 625	31 609	33 661	
В	NC064	Kamiesberg	10 907	12 041	12 824	
В	NC065	Hantam	17 683	19 543	20 817	
В	NC066	Karoo Hoogland	11 544	12 740	13 568	
В	NC067	Khâi-Ma	10 921	12 053	12 837	
C	DC6	Namakwa District Municipality	29 792	31 219	32 445	
Tota	al: Namakw	a Municipalities	119 887	130 712	138 409	
В	NC071	Ubuntu	15 669	17 313	18 441	
В	NC072	Umsobomvu	26 095	28 836	30 714	
В	NC073	Emthanjeni	30 439	33 642	35 832	
В	NC074	Kareeberg	10 466	11 565	12 319	
В	NC075	Renosterberg	12 912	14 257	15 185	
В	NC076	Thembelihle	12 849	14 194	15 120	
В	NC077	Siyathemba	17 829	19 715	21 003	
В	NC078	Siyancuma	31 726	35 106	37 408	
C	DC7	Pixley Ka Seme District Municipality	24 727	26 159	27 319	
Tota	al: Pixley K	a Seme Municipalities	182 713	200 786	213 341	
В	NC081	Mier	8 620	9 518	10 140	
В	NC082	Kai !Garib	41 556	45 950	48 953	
В	NC083	//Khara Hais	46 121	51 070	54 414	
В	NC084	!Kheis	13 955	15 416	16 422	
В	NC085	Tsantsabane	22 454	24 844	26 473	
В	NC086	Kgatelopele	12 918	14 266	15 194	
	DC8	Siyanda District Municipality	39 818	41 647	43 304	
Tota	al: Siyanda	Municipalities	185 443	202 712	214 900	
В	NC091	Sol Plaatje	132 176	146 577	156 193	
В	NC092	Dikgatlong	40 046	44 285	47 179	
В	NC093	Magareng	25 546	28 238	30 080	
В	NC094	Phokwane	56 719	62 706	66 797	
C	DC9	Frances Baard District Municipality	79 281	86 061	88 994	
Tota	al: Frances	Baard Municipalities	333 769	367 868	389 243	
D	NC451	Machaniana	62.000	70.720	as 00a	
В	NC451	Moshaweng	63 808	70 638	75 287	
В	NC452	Ga-Segonyana	58 219	64 408	68 626	
В	NC453	Gamagara	18 283	20 213	21 532	
C	DC45	John Taolo Gaetsewe District Municipality	50 939	54 196	56 721	
1 Ota	ai: John Ta	olo Gaetsewe Municipalities	191 248	209 455	222 165	
Tat	al. Nouth	Cone Municipalities	1 012 050	1 111 522	1 170 070	
1 Ota	ai: Northeri	n Cape Municipalities	1 013 059	1 111 533	1 178 060	

SCHEDULE 3

DETERMINATION OF EACH MUNICIPALITY'S EQUITABLE SHARE OF THE LOCAL GOVERNMENT SPHERE'S SHARE OF REVENUE RAISED NATIONALLY

			Natio	National Financial Year		
			Column A	Colum	n B	
Number Manishalia		Manistration	2011/12	Forward Estimates		
	Number	Municipality	Allocation	2012/13	2013/14	
			R'000	R'000	R'000	
NO	RTH WEST					
В	NW371	Moretele	138 282	152 916	162 918	
В	NW372	Madibeng	247 326	273 870	291 812	
В	NW373	Rustenburg	231 669	256 655	273 476	
В	NW374	Kgetlengrivier	38 032	42 057	44 806	
В	NW375	Moses Kotane	203 756	225 305	240 017	
C	DC37	Bojanala Platinum District Municipality	232 094	239 987	247 826	
Tota	al: Bojanala	Platinum Municipalities	1 091 159	1 190 789	1 260 855	
В	NW381	Ratlou	59 576	65 873	70 182	
В	NW382	Tswaing	56 186	62 108	66 164	
В	NW383	Mafikeng	109 725	121 500	129 460	
В	NW384	Ditsobotla	69 696	76 999	82 017	
В	NW385	Ramotshere Moiloa	70 458	77 882	82 968	
C	DC38	Ngaka Modiri Molema	357 615	394 150	422 535	
Total: Ngaka Modiri Molema Municipalities		723 256	798 513	853 327		
		<u>=</u>				
В	NW392	Naledi	29 119	32 204	34 305	
В	NW393	Mamusa	28 210	31 172	33 205	
В	NW394	Greater Taung	88 632	97 982	104 392	
В	NW396	Lekwa-Teemane	24 989	27 613	29 413	
В	NW397	NW397	58 504	64 639	68 853	
C	DC39	Dr Ruth Segomotsi Mompati District Municipality	188 347	207 978	179 335	
Tota	al: Dr Ruth	Segomotsi Mompati Municipalities	417 801	461 588	449 503	
В	NW401	Ventersdorp	40 951	45 287	48 247	
В	NW402	Tlokwe	76 801	85 191	90 788	
В	NW403	City of Matlosana	303 560	336 019	357 961	
В	NW404	Maquassi Hills	69 259	76 618	81 631	
C	DC40	Dr Kenneth Kaunda District Municipality	153 622	158 939	155 901	
Tota	al: Dr Kenn	eth Kaunda Municipalities	644 194	702 054	734 528	
Tota	al: North W	est Municipalities	2 876 410	3 152 944	3 298 214	

SCHEDULE 3

DETERMINATION OF EACH MUNICIPALITY'S EQUITABLE SHARE OF THE LOCAL GOVERNMENT SPHERE'S SHARE OF REVENUE RAISED NATIONALLY

			Natio	onal Financial Year	r
			Column A	Colum	n B
			2011/12	Forward Es	timates
]	Number	Municipality	Allocation	2012/13	2013/14
			R'000	R'000	R'000
WES	STERN CA	PE			
A	CPT	City of Cape Town	970 473	1 090 987	1 163 859
В	WC011	Matzikama	32 066	35 495	37 821
В	WC012	Cederberg	23 008	25 447	27 106
В	WC013	Bergrivier	19 898	22 010	23 445
В	WC014	Saldanha Bay	29 296	32 575	34 724
В	WC014	•	22 887	25 392	27 058
C	DC1	West Coast District Municipality	68 652	70 992	72 151
		ast Municipalities	195 808	211 911	222 306
		in the state of th	150 000	222,222	
В	WC022	Witzenberg	40 561	44 855	47 781
В	WC023	Drakenstein	62 476	69 437	74 007
В	WC024	Stellenbosch	36 977	41 252	43 992
В	WC025	Breede Valley	56 804	63 024	67 160
В		Langeberg	45 172	49 943	53 197
C	DC2	Cape Winelands District Municipality	200 739	207 135	210 408
		nelands Municipalities	442 730	475 646	496 545
	ar cupe ***	nounds 1/2umo.pumos	112700		.,,,,,,
В	WC031	Theewaterskloof	46 935	51 892	55 272
В	WC032	Overstrand	31 156	34 666	36 957
В	WC033	Cape Agulhas	14 805	16 402	17 477
В	WC034	Swellendam	16 648	18 425	19 629
C	DC3	Overberg District Municipality	41 692	43 229	44 363
		g Municipalities	151 236	164 615	173 698
100	iii Overber,	5 Hamelpanetes	101 200	101010	170 070
В	WC041	Kannaland	16 840	18 619	19 833
В	WC042	Hessequa	23 227	25 750	27 438
В	WC043	Mossel Bay	36 942	40 915	43 589
В	WC044	George	72 201	80 365	85 706
В	WC045	Oudtshoorn	37 618	41 577	44 281
В	WC047	Bitou	20 415	22 696	24 195
В	WC047	Knysna	26 687	29 726	31 695
C	DC4	Eden District Municipality	122 912	126 954	129 701
		inicipalities	356 842	386 601	406 437
В		Laingsburg	8 239	9 097	9 690
В	WC052	Prince Albert	9 200	10 149	10 809
В	WC053	Beaufort West	28 956	32 081	34 194
C	DC5	Central Karoo District Municipality	11 535	12 304	12 906
Tota	al: Central	Karoo Municipalities	57 931	63 632	67 599
Tota	al: Western	Cape Municipalities	2 175 019	2 393 393	2 530 444
			21.0 019		
	onal Total		34 107 901	37 573 396	39 960 288

SCHEDULE 4

ALLOCATIONS TO PROVINCES TO SUPPLEMENT THE FUNDING OF PROGRAMMES OR FUNCTIONS FUNDED FROM PROVINCIAL BUDGETS

					Column A	Column B	ın B
Vote	Name of allocation	Purpose	Type of allocation	Province	2011/12	Forward Estimates	Stimates
					Allocation	2012/13	2013/14
					R'000	R'000	R'000
Agriculture,	Comprehensive Agricultural Support	To expand the provision of agricultural	General conditional allocation to	Eastern Cape	174 985	195 209	223 523
Forestry and	Programme Grant	support services, promote and facilitate	provinces	Free State	102 932	114 829	131 484
Fisheries	,	agricultural development by targeting		Gauteng	41 173	45 931	52 594
(Vote 26)		smallholder and previously disadvantaged		KwaZulu-Natal	164 691	183 726	210 375
		farmers		Limpopo	154 398	172 243	197 226
				Mpumalanga	102 932	114 829	131 484
				Northern Cape	72 052	80 380	92 039
				North West	133 812	149 277	170 930
				Western Cape	82 346	91 863	105 188
				TOTAL	1 029 321	1 148 287	1 314 843
Basic Education	Basic Education   Education Infrastructure Grant	To help accelerate construction,	General conditional allocation to	Eastern Cape	968 435	1 058 635	1 116 860
(Vote 15)		maintenance, upgrading and rehabilitation provinces	provinces	Free State	418 776	458 483	483 699
		of new and existing infrastructure in		Gauteng	461 011	508 633	536 608
		education; and to enhance capacity to		KwaZulu-Natal	1 158 136	1 270 878	1 340 777
		deliver infrastructure in education		Limpopo	874 897	865 485	913 086
				Mpumalanga	472 881	463 237	488 715
				Northern Cape	289 158	317 450	334 909
				North West	469 967	515 812	544 182
				Western Cape	385 039	424 558	447 909
				TOTAL	5 498 300	5 883 171	6 206 745
Health	(a) Health Infrastructure Grant	To supplement provincial funding of health General conditional allocation to	General conditional allocation to	Eastern Cape	299 754	327 673	345 695
(Vote 16)		infrastructure to accelerate the provision of provinces	provinces	Free State	129 621	141 911	149 716
		health facilities and ensure proper		Gauteng	142 694	157 434	166 093
		maintenance of provincial health		KwaZulu-Natal	358 471	393 367	415 002
		infrastructure		Limpopo	270 802	267 888	282 622
				Mpumalanga	146 368	143 383	151 269
				Northern Cape	89 501	98 258	103 662
				North West	145 466	159 656	168 437
				Western Cape	119 179	131 411	138 638
				TOTAL	1 701 856	1820 981	1 921 134

SCHEDULE 4

ALLOCATIONS TO PROVINCES TO SUPPLEMENT THE FUNDING OF PROGRAMMES OR FUNCTIONS FUNDED FROM PROVINCIAL BUDGETS

					Column A	Column B	ın B
Vote	Name of allocation	Purpose	Type of allocation	Province	2011/12	Forward Estimates	Stimates
					Allocation	2012/13	2013/14
					R'000	R'000	R'000
Health	(b) Health Professions Training and	Support provinces to fund service costs	Nationally assigned function to	Eastern Cape	170 071	178 730	188 560
(Vote 16)	Development Grant	associated with training of health	provinces	Free State	124 444	130 930	138 131
	•	professionals; development and		Gauteng	690 803	725 310	765 202
		recruitment of medical specialists in under-		KwaZulu-Natal	249 917	261 860	276 262
		served provinces; and support and		Limpopo	99 730	103 913	109 628
		strengthen undergraduate and postgraduate		Mpumalanga	80 718	85 208	89 894
		training processes in health facilities		Northern Cape	65 510	68 583	72 356
				North West	88 323	93 522	999 86
				Western Cape	407 794	428 120	451 667
				TOTAL	1 977 310	2 076 176	2 190 366
	(c) National Tertiary Services Grant	To compensate tertiary facilities for the	Nationally assigned function to	Eastern Cape	609 327	660 693	698 110
		additional costs associated with spill over	provinces	Free State	715 204	769 964	800 000
		effects; and to ensure adequate provision		Gauteng	2 759 968	2 933 361	3 100 895
		of tertiary health services for all South		KwaZulu-Natal	1 201 831	1 303 824	1 408 053
		African citizens		Limpopo	267 314	277 314	287 314
				Mpumalanga	91 879	105 970	120 270
				Northern Cape	235 948	245 948	255 948
				North West	194 280	209 280	224 280
				Western Cape	1 973 127	2 182 468	2 494 337
				TOTAL	8 048 878	8 688 822	9 389 207
Higher	Further Education and Training Colleges	To ensure the successful transfer of the	General conditional allocation to	Eastern Cape	627 611	681 826	761 118
Education and	Grant	further education and training colleges	provinces	Free State	291 772	314 461	346 682
Training		function to the national Department of		Gauteng	1 012 089	1 101 754	1 233 994
(Vote 17)		Higher Education and Training		KwaZulu-Natal	754 793	828 470	939 709
		,		Limpopo	490 395	531 141	590 206
				Mpumalanga	320 378	341 147	370 651
				Northern Cape	929 29	75 558	89 875
				North West	236 178	254 511	280 449
				Western Cape	527 117	576 220	649 704
				TOTAL	4 325 989	4 705 088	5 262 388

SCHEDULE 4

ALLOCATIONS TO PROVINCES TO SUPPLEMENT THE FUNDING OF PROGRAMMES OR FUNCTIONS FUNDED FROM PROVINCIAL BUDGETS

					Column A	Column B	ın B
Vote	Name of allocation	Purpose	Type of allocation	Province	2011/12	Forward Estimates	stimates
					Allocation	2012/13	2013/14
					R'000	R'000	R'000
Transport	(a) Provincial Roads Maintenance Grant	To supplement provincial roads	General conditional allocation to Eastern Cape	Eastern Cape	1 034 086	1 215 920	1 312 210
(Vote 37)		investments and support preventative	provinces	Free State	447 165	525 794	567 433
		maintenance on provincial road networks;		Gauteng	566 917	583 226	625 542
		and to ensure provinces implement and		KwaZulu-Natal	1 236 648	1 454 099	1 569 251
		maintain road asset management systems		Limpopo	934 208	1 098 478	1 185 468
				Mpumalanga	1 016 603	1 253 564	1 449 002
				Northern Cape	308 760	363 053	391 803
				North West	501 826	290 062	636 795
				Western Cape	411 141	483 437	521 720
				TOTAL	6 457 354	7 567 638	8 259 224
	(b) Public Transport Operations Grant	To provide supplementary funding towards Nationally assigned function to	Nationally assigned function to	Eastern Cape	166 953	180 461	196 061
		public transport services provided by	provinces	Free State	184 566	195 515	208 162
		provincial departments of transport		Gauteng	1 577 612	1 635 695	1 702 781
				KwaZulu-Natal	773 473	815 611	864 281
				Limpopo	249 498	274 561	303 510
				Mpumalanga	420 099	436 626	455 715
				Northern Cape	37 565	42 715	48 662
				North West	77 211	89 230	103 111
				Western Cape	666 255	690 480	718 460
				TOTAL	4 153 232	4 360 894	4 600 743

SCHEDULE 4

ALLOCATIONS TO MUNICIPALITIES TO SUPPLEMENT THE FUNDING OF FUNCTIONS FUNDED FROM MUNICIPAL BUDGETS

				Column A	Column B	ın B
Vote	Name of allocation	Purpose	City	2011/12	Forward Estimates	stimates
				Allocation	2012/13	2013/14
				R'000	R'000	R'000
Human	Urban Settlements Development Grant	To improve the efficiency and coordination of investments in the built	Buffalo City	423 446	497 908	547 338
Settlements		environment by providing large municipalities with appropriate resources and City of Cape Town	City of Cape Town	824 030	972 615	1 067 485
(Vote 31)		control over the selection and pursuit of investment programmes in the built   City of Johannesburg	City of Johannesburg	1 027 970	1 216 260	1 333 559
		environment	City of Tshwane	891 081	1 053 856	1 155 692
		<u>11.11</u>	Ekurhuleni	1 094 276	1 297 640	1 421 452
		9	eThekwini	1 091 574	1 299 706	1 421 322
			Mangaung	411 995	483 427	531 884
			Nelson Mandela Bay	502 626	588 100	648 074
			TOTAL	6 266 998	7 409 512	8 126 806

SCHEDULE 5

					Column A	Column B	ın B
Vote	Name of allocation	Purpose	Type of allocation	Province	2011/12	Forward Estimates	Stimates
					Allocation	2012/13	2013/14
					R'000	R'000	R'000
Agriculture,	(a) Ilima/Letsema Projects Grant	To assist targeted vulnerable South African Conditional allocation	Conditional allocation	Eastern Cape	40 000	42 000	44 310
Forestry and		farming communities to increase		Free State	52 000	54 600	57 603
Fisheries		agricultural production and improve		Gauteng	20 000	21 000	22 155
(Vote 26)		farming skills		KwaZulu-Natal	000 09	63 000	66 465
		,		Limpopo	40 000	42 000	44 310
				Mpumalanga	40 000	42 000	44 310
				Northern Cape	000 09	63 000	66 465
				North West	40 000	42 000	44 310
				Western Cape	48 000	50 400	53 172
				TOTAL	400 000	420 000	443 100
	(b) Land Care Programme Grant: Poverty	To enhance the sustainable conservation of Conditional allocation	Conditional allocation	Eastern Cape	9 244	16 823	15 866
	Relief and Infrastructure Development	natural agriculture resources through a		Free State	4 622	8 953	8 571
		community-based participatory approach;		Gauteng	4 044	6 246	6 163
		create job opportunities through the		KwaZulu-Natal	9 244	20 304	18 746
		Expanded Public Works programme; and		Limpopo	299 8	20 356	19 562
		to create an enabling environment for		Mpumalanga	5 198	10 958	10 249
		improved food security and poverty relief		Northern Cape	6 355	12 724	12 055
		•		North West	6 932	11 557	10 552
				Western Cape	3 466	7 740	7 233
				TOTAL	57 772	115 661	108 997
Arts and Cultur	Arts and Culture Community Library Services Grant	To transform urban and rural community (	Conditional allocation	Eastern Cape	80 974	79 273	75 633
(Vote 14)		library infrastructure, facilities and services		Free State	47 909	50 304	55 070
		(primarily targeting previously		Gauteng	54 716	57 452	60 611
		disadvantaged communities) through a		KwaZulu-Natal	45 401	48 619	56 297
		recapitalised programme at provincial level		Limpopo	66 497	69 822	73 662
		in support of local government and national		Mpumalanga	66 497	69 822	73 662
		initiatives		Northern Cape	006 69	73 395	77 430
				North West	62 832	65 973	69 601
				Western Cape	48 694	56 129	60 216
				TOTAL	543 420	220 189	602 182

SCHEDULE 5

Mame of allocation  Basic Education  (a) Dinaledi Schools Grant  (Vote 15)  (b) HIV and Aids (Life Skills Education)  Grant  (c) National School Nutrition Programme  Grant	<b>Jocation</b>	Purpose	Type of allocation				
Vote 15)  (b) HIV and Aids (Life S Grant Grant  (c) National School Nutr Grant	rant		type of anocarron	Fronince	2011/12	Forward Estimates	stimates
Vote 15)  (b) HIV and Aids (Life S Grant  (c) National School Nutr  Grant	rant				Allocation	2012/13	2013/14
Vote 15)  (b) HIV and Aids (Life 5)  Grant  (c) National School Nutr  Grant	rant				R'000	R'000	R'000
Vote 15)  (b) HIV and Aids (Life S Grant  (c) National School Nutr Grant		To promote Mathematics and Physical	Conditional allocation	Eastern Cape	8 400	12 000	12 660
(b) HIV and Aids (Life S Grant (c) National School Nutr		Science teaching and learning; to improve		Free State	5 040	7 200	7 596
(b) HIV and Aids (Life S Grant (c) National School Nutr Grant		learner performance in Mathematics and		Gauteng	14 140	20 200	21 311
(b) HIV and Aids (Life S Grant (c) National School Nurt Grant		Physical Science in line with the Action		KwaZulu-Natal	12 320	17 600	18 568
(b) HIV and Aids (Life E Grant (c) National School Nutr		Plan 2014; and to improve teachers content		Limpopo	7 140	10 200	10 761
(b) HIV and Aids (Life E Grant (c) National School Nutr Grant		knowledge of Mathematics and Physical		Mpumalanga	6 440	9 200	9026
(b) HIV and Aids (Life S Grant (c) National School Nutr		Science		Northern Cape	2 380	3 400	3 587
(b) HIV and Aids (Life S Grant (c) National School Nutr				North West	7 420	10 600	11 183
(b) HIV and Aids (Life S Grant  (c) National School Nutr  Grant				Western Cape	6 720	009 6	10 128
(b) HIV and Aids (Life S Grant  Grant  (c) National School Nutr  Grant				TOTAL	20 000	100 000	105 500
Grant (c) National School Nutr Grant	Skills Education)		Conditional allocation	Eastern Cape	34 346	35 322	37 265
(c) National School Nutr		strategy by increasing sexual and reproductive		Free State	11 772	12 561	13 252
(c) National School Nutr		knowledge, skills and appropriate decision		Gauteng	28 175	29 217	30 824
(c) National School Nutr		making among learners and educators; to		KwaZulu-Natal	45 114	46 876	49 445
(c) National School Nutr		mitigate the impact of HIV by providing a		Limpopo	28 088	30 012	31 663
(c) National School Nutr		caring, supportive and enabling environment for		Mpumalanga	16 388	17 486	18 448
(c) National School Nutr		rearriers and educators; and to ensure the		Northern Cape	4 357	4 649	4 905
(c) National School Nutr		provision of a safe, rights-based environment in schools that is free of discrimination stigms		North West	14 700	15 685	16 555
(c) National School Nutr		and any form of sexual harassment/abuse		Western Cape	16 388	17 486	18 448
(c) National School Nutr				TOTAL	199 328	209 294	220 805
Grant	trition Programme	To provide nutritious meals to targeted	Conditional allocation	Eastern Cape	845 166	909 644	959 674
		learners		Free State	244 699	263 367	277 852
				Gauteng	862 605	548 690	578 868
				KwaZulu-Natal	1 070 013	1 151 644	1 214 985
				Limpopo	839 669	892 964	942 077
				Mpumalanga	440 923	474 560	500 661
				Northern Cape	105 116	113 136	119 359
				North West	305 935	329 301	347 412
				Western Cape	227 433	244 784	258 247
				TOTAL	4 578 752	4 928 090	5 199 135

SCHEDULE 5

Vote					Column	Column	Q III
	Name of allocation	Purpose	Type of allocation	Province	2011/12	Forward Estimates	stimates
					Allocation	2012/13	2013/14
					R'000	R'000	R'000
Basic Education (d)	Basic Education (d) Technical Secondary Schools	To recapitalise technical schools to	Conditional allocation	Eastern Cape	37 584	39 464	41 635
(Vote 15) Rec	Recapitalisation Grant	improve the capacity to contribute to skills		Free State	14 428	15 150	15 983
		development and training		Gauteng	30 596	32 127	33 894
		,		KwaZulu-Natal	38 563	40 490	42 717
				Limpopo	27 450	28 822	30 407
				Mpumalanga	18 078	18 983	20 027
				Northern Cape	L99 L	8 052	8 495
				North West	17 015	17 867	18 850
				Western Cape	8 619	9 045	9 542
				TOTAL	200 000	210 000	221 550
Health (a)	(a) Comprehensive HIV and Aids Grant	To enable the health sector to develop an	Conditional allocation	Eastern Cape	864 173	1 014 134	1 222 480
(Vote 16)		effective response to HIV and Aids		Free State	530 440	621 824	750 788
		including universal access to HIV		Gauteng	1 620 673	1 934 057	2 294 326
		counselling and testing; to support the		KwaZulu-Natal	1 889 427	2 246 099	2 675 081
		implementation of the National Operational		Limpopo	624 909	733 963	884 146
		Plan for comprehensive HIV and Aids		Mpumalanga	490 366	278 087	694 264
		treatment and care; and to subsidise in-part		Northern Cape	212 923	247 228	301 557
		funding for antiretroviral treatment		North West	599 437	696 502	848 533
		programme		Western Cape	660 614	743 249	935 489
				TOTAL	7 492 962	8 824 610	10 606 664
(p)	(b) Forensic Pathology Services Grant	To continue the development and provision Conditional allocation	Conditional allocation	Eastern Cape	73 506	1	1
		of adequate forensic pathology services in		Free State	39 451	•	•
		all provinces		Gauteng	996 26	•	•
				KwaZulu-Natal	161 550	•	•
				Limpopo	42 308	•	•
				Mpumalanga	53 114	1	•
				Northern Cape	24 240	•	•
				North West	28 019	1	1
				Western Cape	70 226	1	1
				TOTAL	590 380	•	•

SCHEDULE 5

					Column A	Column B	ın B
Vote	Name of allocation	Purpose	Type of allocation	Province	2011/12	Forward Estimates	stimates
					Allocation	2012/13	2013/14
					R'000	R'000	R'000
Health	(c) Hospital Revitalisation Grant	To provide funding to enable provinces to   Conditional allocation	Conditional allocation	Eastern Cape	382 048	402 678	387 104
(Vote 16)		plan, manage, modernise, rationalise and		Free State	417 883	438 140	412 172
		transform health infrastructure, health		Gauteng	801 965	828 552	760 206
		technology, monitoring and evaluation of		KwaZulu-Natal	547 698	209 995	533 432
		the health facilities in line with national		Limpopo	371 672	392 410	369 152
		policy objectives		Mpumalanga	356 557	378 014	355 081
		1		Northern Cape	406 892	427 263	401 940
				North West	370 074	401 150	377 375
				Western Cape	481 501	501 096	471 397
				TOTAL	4 136 290	4 335 908	4 067 859
Human	Human Settlements Development Grant	To provide funding for the creation of	Conditional allocation	Eastern Cape	2 177 676	2 274 820	2 403 688
Settlements		sustainable human settlements		Free State	913 907	954 326	1 006 814
(Vote 31)				Gauteng	3 804 611	3 970 951	4 186 987
				KwaZulu-Natal	2 769 871	2 891 813	3 050 176
				Limpopo	1 398 914	1 459 839	1 539 788
				Mpumalanga	916 677	957 218	1 009 865
				Northern Cape	322 639	336 906	355 437
				North West	98 376	1 042 529	1 099 868
				Western Cape	1 638 845	1 711 035	1 804 785
				TOTAL	14 941 516	15 599 437	16 457 408
Public Works	(a) Devolution of Property Rate Funds	To facilitate the transfer of property rates	Conditional allocation	Eastern Cape	192 709	202 854	214 360
(Vote 7)	Grant	expenditure responsibility to provinces		Free State	219 916	233 737	249 390
				Gauteng	270 775	284 314	299 951
				KwaZulu-Natal	518 585	699 955	599 319
				Limpopo	34 054	35 757	37 725
				Mpumalanga	73 964	77 647	84 601
				Northern Cape	41 754	44 355	47 303
				North West	160 192	179 984	206 241
				Western Cape	291 281	322 730	351 963
				TOTAL	1 803 230	1 938 047	2 090 853

SCHEDULE 5

					Column A	Column B	ın B
Vote	Name of allocation	Purpose	Type of allocation	Province	2011/12	Forward Estimates	Stimates
					Allocation	2012/13	2013/14
					R'000	R'000	R'000
Public Works	(b) Social Sector Expanded Public Works	To incentivise provincial social sector	Conditional allocation	Eastern Cape	5 070	7 120	9 139
(Vote 7)	Programme Incentive Grant for Provinces   departments identified in the 2011 Social	departments identified in the 2011 Social		Free State	15 586	18 703	22 444
		Sector EPWP Log-frame to increase job		Gauteng	34 848	41 818	50 181
		creation by focusing on the strengthening		KwaZulu-Natal	30 269	32 485	34 354
		and expansion of social service		Limpopo	28 332	33 998	40 798
		programmes that have employment		Mpumalanga	13 407	16 088	19 306
		potential		Northern Cape	13 890	16 669	20 002
		1		North West	42 222	54 229	65 901
				Western Cape	16 734	20 394	23 903
				TOTAL	200 358	241 504	286 028
Sport and	Mass Sport and Recreation Participation	To facilitate mass participation within	Conditional allocation	Eastern Cape	63 570	66 749	70 420
Recreation	Programme Grant	communities and schools through selected		Free State	33 078	34 732	36 642
South Africa		activities, empowerment of communities		Gauteng	71 148	74 705	78 814
(Vote 20)		and schools in conjunction with relevant		KwaZulu-Natal	87 694	92 078	97 142
		stakeholders		Limpopo	53 636	56 318	59 415
				Mpumalanga	38 382	40 301	42 518
				Northern Cape	26 372	27 691	29 214
				North West	35 124	36 880	38 909
				Western Cape	42 964	45 112	47 593
				TOTAL	451 968	474 566	200 667
Transport	Gautrain Rapid Rail Link	To provide for national government	Conditional allocation	Eastern Cape	1	1	•
(Vote 37)		funding contribution to the Gauteng		Free State	•	•	•
		Provincial Government for the construction		Gauteng	5 300	1	1
		of a fully integrated Gautrain Rapid Rail		KwaZulu-Natal	1	1	1
		network		Limpopo	i	i	•
				Mpumalanga	1	1	•
				Northern Cape	1	1	•
				North West	1	1	1
				Western Cape	-		
				TOTAL	2 300	•	

SCHEDULE 6

### SPECIFIC PURPOSE ALLOCATIONS TO MUNICIPALITIES

			Column A	Column B	nn B
Vote	Name of allocation	Purpose	2011/12	Forward Estimates	Stimates
			Allocation	2012/13	2013/14
			R'000	R'000	R'000
RECURRENT GRANTS	FRANTS				
Cooperative	Municipal Systems Improvement Grant	To assist municipalities in building in-house capacity to perform their functions and stabilise	0		
Governance and		institutional and governance systems as required in the Municipal Systems Act and related	219 420	230 096	242 734
Affairs		registation, politics and focal government tunial ound strategy			
(Vote 3)					
National	Local Government Financial Management				
Treasury	Grant	implement the Municipal Finance Management Act (MFMA) As part of strengthening financial and	434 641	479 333	526 086
(Vote 10)		asset management in municipalities, the grant provides funding for water and energy internship			
		programme to graduates in selected water boards and municipalities			
Water Affairs	Water Services Operating Subsidy Grant	To subsidise and build capacity in water schemes owned and/or operated by the Department of Water	i c	000	000
(V OTE 38)		Affairs of by other agencies on behalf of the department and transfer these schemes to local government	200 /94	999 000	420 943
		TOTAL	1 214 855	1 108 429	1 189 765
		TOTAL	CC0 +17 I	774 001 1	1 107 /05
INFRASTRUCTURE GRANTS	TURE GRANTS				
Cooperative	Municipal Infrastructure Grant	To provide specific capital finance for basic municipal infrastructure backlogs for poor households,			
Governance and		micro enterprises and social institutions servicing poor communities	11 443 505	13 914 132	14 6/9 408
Traditional Affairs					
(Vote 3)					
Energy	(a) Integrated National Electrification	To implement the Integrated National Electrification Programme (INEP) by providing capital			
(Vote 29)	Programme (Municipal) Grant	subsidies to municipalities to address the electrification backlog of occupied residential dwellings, the installation of bulk infrastructure and rehabilitation and refurbishment of electricity infrastructure	1 096 612	1 151 443	1 214 //2
		in order to improve quality of supply			
	(b) Electricity Demand Side Management	To implement the Electricity Demand Side Management (EDSM) programme by providing subsidies			
	(Municipal) Grant	to licenced distributors to address EDSM in residential dwellings, communities and municipal infrastructure in order to mitigate the risk of load shedding and supply interruptions	780 000	1	1
National	Neighbourhood Development Partnership	To support neighbourhood development projects that provide community infrastructure and create			
Treasury	Grant	the platform for other public and private sector development, towards improving the quality of life of	750 000	800 000	800 000
(Vote 10)		residents in targeted underserved neighbourhoods (townships generally)			
Transport	(a) Public Transport Infrastructure and	To provide for accelerated planning, construction and improvement of public and non-motorised	1 803 347	1 900 781	5 563 604
(Vote 37)	Systems Grant	transport networks	1+0000+	4 223 /81	+00 505 5
	(b) Rural Transport Services and Infrastructure Grant	To assist rural district municipalities to set up rural road asset management systems, and collect road and traffic data in line with the Road Infrastructure Strateoic Framework for South Africa	35 439	37 382	39 250
. 00 4					
Water Affairs (Vote 38)	Municipal Drought Relief Grant	To provide capital finance for construction of appropriate water infrastructure to alleviate further impacts of drought in Nelson Mandela Bay metropolitan municipality for affected households, micro	450 000	1	1
		enterprises and social institutions			
		TOTAL	18 858 903	20 902 738	22 297 035

SCHEDULE 7

ALLOCATIONS-IN-KIND TO PROVINCES FOR DESIGNATED SPECIAL PROGRAMMES

Antocation 2012/13    R'000   R'000   R'000   Cape   Cape	R'000   Eastern Cape   R'000   Free State   - Cauteng   - Cauten	R'000   Eastern Cape   R'000   Pree State   Cauteng   KwaZulu-Natal   Cauteng   Caut	R'000	R'000	R'000	R'000	R'000	R'000
Eastern Cape Free State	Eastern Cape Free State Gauteng	Eastern Cape Free State Gauteng KwaZulu-Natal	_	_	_	_	_	
			Eastern Cape Free State Gauteng KwaZulu-Natal Limpopo	tern Cape e State tteng aZulu-Natal npopo	r Cape ate g slu-Natal po alanga	oe atal atal tpe	la e	al s
on of water,	ater,			Eas Free Gau Kw Lim Mp	Easterr Free St Gauten KwaZu Limpol Mpuma	Eastern Cap Free State Gauteng KwaZulu-N Limpopo Mpumalang Northern Ca	Eastern Cape Free State Gauteng KwaZulu-Nat Limpopo Mpumalanga Northern Cap North West	Eastern Cape Free State Gauteng KwaZulu-Natal Limpopo Mpumalanga Northern Cape North West Western Cape Unallocated
Eradication of inappropriate school infrastructure; and provisi sanitation and electricity to schools	Eradication of inappropriate school infrastructure; and provision of w sanitation and electricity to schools	Eradication of inappropriate school infrastructure; and provision of water, sanitation and electricity to schools	Eradication of inappropriate school infrastructure; and provision of water, sanitation and electricity to schools	Eradication of inappropriate school infrastructure; and provision of water, sanitation and electricity to schools	Eradication of inappropriate school infrastructure; and provision of water, sanitation and electricity to schools	Eradication of inappropriate school infrastructure; and provision of water, sanitation and electricity to schools	Eradication of inappropriate school infrastructure; and provision of water, sanitation and electricity to schools	Eradication of inappropriate school infrastructure; and provision of water, sanitation and electricity to schools
				(Vote 15) sanitation and electricity to school and electricity to scho				

SCHEDULE 7

ALLOCATIONS-IN-KIND TO MUNICIPALITIES FOR DESIGNATED SPECIAL PROGRAMMES

			Column A	Column B	nn B
Vote	Name of allocation	Purpose	2011/12	Forward Estimates	Estimates
			Allocation	2012/13	2013/14
			R'000	R'000	R'000
National Treasury	Neighbourhood Development Partnership Grant	` _	100 000	80 000	55 000
(Vote 10)		residents in targeted underserved neighbourhoods (townships generally)			
Energy (Vote 29)	(a) Integrated National Electrification Programme (Eskom) Grant	To implement the Integrated National Electrification Programme (INEP) by providing capital subsidies to Eskom to address the electrification backlog of occupied residential dwellings, the installation of bulk infrastructure and rehabilitation and refurbishment of electricity infrastructure in order to improve quality of supply	1 737 812	1 882 057	1 985 570
	(b) Electricity Demand Side Management (Eskom) Grant	To implement the Electricity Demand Side Management (EDSM) programme by providing subsidies to Eskom to address EDSM in residential dwellings and communities in order to mitigate the risk of load shedding and supply interruptions	118 800	1	1
Water Affairs (Vote 38)	(a) Water Services Operating Subsidy Grant	To subsidise and build capacity in water schemes owned and/or operated by the Department of Water Affairs or by other agencies on behalf of the department and transfer these schemes to local government	99 935	1	1
	(b) Regional Bulk Infrastructure Grant	To develop regional bulk infrastructure for water supply to supplement water treatment works at resource development and link such water resource development with the local bulk and local distribution networks on a regional basis cutting across several local municipal boundaries. In the case of sanitation, to supplement regional bulk collection as well as regional waste water treatment works	1 704 140	2 003 217	2 176 274
Human Settlements (Vote 31)	Rural Households Infrastructure Grant	To provide specific capital funding for the eradication of rural water and sanitation backlogs and is targeted at existing households where bulk-dependent services are not viable. The grant also funds training for beneficiaries on health and hygiene practices and how to maintain the facilities provided	231 500	479 500	517 250
		TOTAL	3 992 187	4 444 774	4 734 094

SCHEDULE 8

## INCENTIVES TO PROVINCES TO MEET TARGETS WITH REGARDS TO PRIORITY GOVERNMENT PROGRAMMES

					Column A	Column B	nn B
Vote	Name of allocation	Purpose	Type of allocation	Province	2011/12	Forward Estimates	Estimates
					Allocation	2012/13	2013/14
					R'000	R'000	R'000
Public Works	Public Works   Expanded Public Works Programme	To incentivise provincial departments to Incentive allocation to provinces   Eastern Cape	Incentive allocation to provinces	Eastern Cape	30 431	1	'
(Vote 7)	Incentive Grant for Provinces	increase job creation efforts in		Free State	10 606	1	•
		infrastructure, environment and culture		Gauteng	44 210	1	1
		programmes through the use of labour-		KwaZulu-Natal	129 836	1	•
		intensive methods and the expansion of job		Limpopo	9 545	1	1
		creation in line with the Expanded Public		Mpumalanga	9 486	1	•
		Works Programme (EPWP) guidelines		Northern Cape	4 810	1	1
				North West	5 758	1	•
				Western Cape	22 587	1	'
				Unallocated	•	325 220	401 588
				TOTAL	096 296	325 220	401 588

# INCENTIVES TO MUNICIPALITIES TO MEET TARGETS WITH REGARDS TO PRIORITY GOVERNMENT PROGRAMMES

			Column A	Column B	an B
Vote	Name of allocation	Purpose	2011/12	Forward Estimates	Stimates
			Allocation	2012/13	2013/14
			R'000	R'000	R'000
Public Works (Vote 7)	Expanded Public Works Programme Incentive Grant for Municipalities	To incentivise municipalities to increase job creation efforts in infrastructure, environment and culture programmes through the use of labour-intensive methods and the expansion of job creation in	679 583	829 999	779 496
		line with the Expanded Public Works Programme (EPWP) guidelines			
		TOTAL	679 583	829 999	779 496

SCHEDULE 9

UNALLOCATED PROVISIONS FOR PROVINCES FOR DISASTER RESPONSE

				Column A	Column B	nn B
Vote	Name of allocation	Purpose	Province	2011/12	Forward Estimates	Estimates
				Allocation	2012/13	2013/14
				R'000	R'000	R'000
rative	Cooperative Provincial Disaster Grant	To provide for the immediate release of funds for disaster response	Eastern Cape	1	1	1
nance and			Free State	1	1	•
ional			Gauteng	1	1	•
Affairs			KwaZulu-Natal	'	1	•
-			Limpopo	1	1	1
			Mpumalanga	1	1	1
			Northern Cape	1	1	1
			North West	1	1	•
			Western Cape	•	1	•
			Unallocated	305 000	180 000	190 000
1			TOTAL	305 000	180 000	190 000

### UNALLOCATED PROVISIONS FOR MUNICIPALITIES FOR DISASTER RESPONSE

				Column A	Column B	nn B
Vote	Name of allocation	Purpose		2011/12	Forward Estimates	Estimates
				Allocation	2012/13	2013/14
				R'000	R'000	R'000
Cooperative	Cooperative   Municipal Disaster Grant	To provide for the immediate release of funds for disaster response				
Governance and				470 000	330 000	350 000
Traditional						
Affairs						
(Vote 3)						
-		TOTAL	LAL	470 000	330 000	350 000

### MEMORANDUM ON THE OBJECTS OF THE DIVISION OF REVENUE BILL, 2011

### 1. BACKGROUND

- 1.1 Section 214(1) of the Constitution of the Republic of South Africa, 1996, ("the Constitution") requires that an Act of Parliament must provide for—
  - 1.1.1 the equitable division of revenue raised nationally among the national, provincial and local spheres of government;
  - 1.1.2 the determination of each province's equitable share of the provincial share of that revenue; and
  - 1.1.3 any other allocations to provinces, local government or municipalities from the national government's share of that revenue, and for any conditions on which those allocations may be made.
- 1.2 Section 10 of the Intergovernmental Fiscal Relations Act, 1997 (Act No. 97 of 1997) ("the IGFRA"), requires that, as part of the process of the enactment of the Act of Parliament referred to in paragraph 1.1, each year when the annual budget is introduced, the Minister of Finance must introduce in the National Assembly a Division of Revenue Bill ("the Bill") for the financial year to which that budget relates.
- 1.3 The IGFRA requires that the Bill be accompanied by a memorandum explaining—
  - 1.3.1 how the Bill takes account of each of the matters listed in section 214(2)(a) to (j) of the Constitution;
  - 1.3.2 the extent to which account was taken of any recommendations of the Financial and Fiscal Commission ("the FFC") that were submitted to the Minister of Finance or were raised during consultations with the FFC; and
  - 1.3.3 any assumptions or formulae used in arriving at the respective shares of the three spheres of government and the division of the provincial share between the nine provinces.
- 1.4 In terms of section 7(4) of the Money Bills Amendment Procedure and Related Matters Act, 2009 (Act No. 9 of 2009) ("the MBAPRMA"), when tabling the budget, a report must also be tabled that responds to the recommendations made in the reports by the Parliamentary Committees on Finance on the proposed fiscal framework in the Medium Term Budget Policy Statement ("the MTBPS") and the reports by the Committees on Appropriations regarding the proposed division of revenue and the conditional grant allocations to provinces and local governments as contained in the MTBPS. The report must explain how the Bill and the national budget give effect to, or the reasons for not taking into account, the recommendations contained in the Committee reports.
- 1.5. The memorandum referred to in paragraph 1.3 above will be attached as "Annexure W1" to the Budget Review, and the report referred to in paragraph 1.4 above will be tabled with the budget.
- 1.6 The Bill is introduced in compliance with the requirements of the Constitution, the IGFRA, and the MBAPRMA, as set out in paragraphs 1.1 to 1.4 above.
- 1.7 The allocations contemplated in section 214(1) of the Constitution are set out in eight Schedules to the Bill:

- Schedule 1 contains the equitable shares of the three spheres of government;
- Schedule 2 sets out provincial equitable share allocations;
- Schedule 3 sets out local government equitable share allocations per municipality;
- Schedules 4-8 deal with all other allocations, including conditional grants, unconditional grants, indirect transfers and incentive grants to provinces and local government;
- *Schedule 9* provides for the release of funds to provinces and municipalities for disaster response.

### 2. SUMMARY OF BILL

The following is a brief summary of the Bill:

- Clause 1 contains the relevant definitions;
- *Clause 2* sets out the objects of the Bill, which is essentially the promotion of cooperative governance in intergovernmental budgeting;
- *Clause 3* provides for the equitable division of anticipated revenue raised nationally among the national, provincial and local spheres of government, which is set out in Schedule 1;
- Clause 4 provides for each province's equitable share, which is set out in Schedule 2, and for a payment schedule in terms of which such shares must be transferred;
- *Clause 5* provides for local government's equitable share of revenue and the determination of each municipality's share of that revenue;
- Clause 6 determines what must happen if actual revenue raised falls short or is in excess of anticipated revenue for the financial year, and allows for additional conditional and unconditional allocations from the national government's portion of the equitable share or excess revenue;
- Clause 7 provides for conditional allocations to provinces;
- Clause 8 provides for conditional allocations to municipalities;
- Clauses 9 and 10 set out the duties of a transferring national officer in respect of Schedule 4 to 8 allocations;
- Clauses 11 and 12 set out the duties of a receiving officer in respect of Schedule 4, 5, 6, 8 or 9 allocations;
- *Clause 13* prescribes the duties in respect of annual financial statements and annual reports for 2011/12;
- Clause 14 requires the publication of allocations and conditional grant frameworks in the Government Gazette;
- Clause 15 requires that spending must only be in accordance with the purpose and subject to the conditions set out in the grant frameworks for Schedule 4 to 9 allocations;
- Clauses 16 and 17 provide for the withholding and stopping of allocations;
- Clause 18 provides for the re-allocation of funds;
- Clause 19 provides for the possible conversion of Schedule 6 and Schedule 7 allocations during the course of the financial year, in order to prevent under-spending on the allocation, and for the possible conversion of an allocation in terms of the School Infrastructure Backlogs Grant to become an allocation in terms of the Education Infrastructure Grant;
- Clause 20 provides for the overall management of unspent conditional allocations;
- Clauses 21 and 22 provide for the management and amendment of payment schedules;
- Clause 23 provides for the correction of any allocation transferred in error or fraudulently;
- Clause 24 provides for allocations not listed in the Schedules;
- Clause 25 addresses the implementation of re-demarcations of municipal boundaries:
- Clauses 26 and 27 provide for preparations for the next financial year and expenditure prior to the commencement of the Division of Revenue Act, 2012;
- Clause 28 sets out the duties of municipalities;
- Clause 29 sets out the duties of provincial treasuries;

- Clause 30 sets out the duties of the National Treasury;
- Clauses 31 to 36 provide for general treasury matters such as allocations by public entities to provinces or municipalities, liabilities, unauthorised and irregular expenditure, financial misconduct, delegations and assignments, and exemptions;
- Clauses 37 to 39 provide for regulations, the repeal of laws and the short title.

### 3. ORGANISATIONS AND INSTITUTIONS CONSULTED

The following institutions were consulted on the Bill:

- The Financial and Fiscal Commission;
- the South African Local Government Association; and
- national and provincial departments.

### 4. FINANCIAL IMPLICATIONS TO THE STATE

This memorandum outlines the proposed division of revenue between the three spheres of government, and the financial implications to government are limited to the total transfers to provinces and local government as indicated in the Schedules to the Bill.

### 5. CONSTITUTIONAL IMPLICATIONS

This Bill gives effect to section 214 of the Constitution.

### 6. PARLIAMENTARY PROCEDURE

- 6.1 The State Law Advisers and the National Treasury are of the opinion that this Bill must be dealt with in accordance with the procedure prescribed by section 76(1) of the Constitution, since it provides for legislation envisaged in Chapter 13 of the Constitution, and it includes provisions affecting the financial interests of the provincial sphere of government, as contemplated in section 76(4)(b) of the Constitution.
- 6.2 The State Law Advisers are of the opinion that it is not necessary to refer this Bill to the National House of Traditional Leaders in terms of section 18(1)(a) of the Traditional Leadership and Governance Framework Act, 2003 (Act No. 41 of 2003), since it does not contain provisions pertaining to customary law or customs of traditional communities.

### DIVISION OF REVENUE ATTACHMENTS

		Page
1.	Website "Annexure W1" to the 2011 Budget Review: Explanatory Memorandum to the Division of Revenue	56
2.	Annexure W2: Frameworks for Conditional Grants to Provinces	99
3.	Annexure W3: Frameworks for Conditional Grants to Municipalities	149
4.	Annexure W4: Specific Purpose Allocations to Municipalities (Schedule 6): Recurrent Grants	177
5.	Annexure W5: Infrastructure Grant Allocations to Municipalities (Schedules 4, 6 and 9)	200
6.	Annexure W6: Allocations-In-Kind to Municipalities (Schedule 7)	243
7.	Annexure W7: Incentives to Municipalities to meet Targets with regards to Priority Government Programmes (Schedule 8)	276
8.	Annexure W8: Equitable Share and Total Allocations to Municipalities by National and Municipal Financial Years	288
9.	Appendix W1: Equitable Share Allocations to Municipalities (Equitable Share Formula Allocations + RSC Levies Replacement + Special Contribution towards Councillor Remuneration)	300
10	. Appendix W2: Breakdown of Equitable Share Allocations per Local Municipality per Service for District Municipalities authorised for Services	311
11	. Appendix W3: Breakdown of MIG Allocations per Local Municipality for District Municipalities authorised for Services	323
12	. Appendix W4: Breakdown of Regional Bulk Infrastructure Grant Allocations per Local Municipality per Project	334
13	. Appendix W5: Breakdown of EPWP Social Sector Allocations per Provincial Department	340
14	. Appendix W6: Incentives to Provinces and Municipalities to meet Targets with regards to Priority Government Programmes	342
15	. Appendix W7: Format of Infrastructure Projects List for Provinces that receive the Education Infrastructure Grant, the Health Infrastructure Grant and the Provincial Roads Maintenance Grant as required by Section 9(2)(d) of the Bill	353

### EXPLANATORY MEMORANDUM TO THE DIVISION OF REVENUE

(Website "Annexure W1" to the 2011 Budget Review)

### W1

### **Explanatory memorandum** to the division of revenue

### Background

The allocation of resources to the three spheres of government is a critical step in the budget process, required before national government, nine provinces and 278 municipalities (after the 2011 municipal elections) can determine their own budgets. The allocation process takes into account the powers and functions assigned to the three spheres of government. The process for making this decision is at the heart of cooperative governance as envisaged in the Constitution.

To foster transparency and ensure smooth intergovernmental relations, section 214(1) of the Constitution requires that every year a Division of Revenue Act determine the equitable division of nationally raised revenue between the three spheres of government. The Intergovernmental Fiscal Relations Act (1997) prescribes the process for determining the equitable sharing and allocation of revenue raised nationally. Sections 9 and 10(4) of the act set out the consultation process to be followed with the Financial and Fiscal Commission (FFC), including the process of considering recommendations made with regard to the equitable division of nationally raised revenue.

This explanatory memorandum to the 2011 Division of Revenue Bill fulfils the requirement set out in section 10(5) of the Intergovernmental Fiscal Relations Act that requires the Division of Revenue Bill to be accompanied by an explanatory memorandum detailing how the bill takes account of the matters listed in sections 214(2)(a) to (j) of the Constitution, government's response to the recommendations of the FFC, and any assumptions and formulas used in arriving at the respective divisions among provinces and municipalities. This explanatory memorandum contains six parts:

- Part 1 lists the factors that inform the division of resources between the three spheres of government.
- Part 2 describes the 2011 division of revenue between the three spheres of government.
- Part 3 sets out how the FFC's recommendations on the 2011 division of revenue have been taken into account.
- Part 4 explains the formula and criteria for the division of the provincial equitable share and for conditional grants to provinces.
- Part 5 sets out the formula and criteria for the division of the local government equitable share and conditional grants between municipalities.
- Part 6 summarises issues that will form part of subsequent reviews of provincial and local government fiscal frameworks.

The Division of Revenue Bill and its underlying allocations are the culmination of extensive consultation between national, provincial and local government. The Budget Council deliberated on the matters discussed in this memorandum at several meetings during the year. The approach to local government

allocations was discussed with organised local government at technical meetings with the South African Local Government Association (SALGA), culminating in a meeting of the Budget Forum (Budget Council plus SALGA) on 5 October 2010. An extended Cabinet meeting involving ministers, provincial premiers and the chairperson of SALGA was held on 13 October 2010, and the division of revenue was agreed for the next three years.

### Part 1: Constitutional considerations

Section 214 of the Constitution requires that the annual Division of Revenue Act be enacted only after account is taken of factors in sub-sections (2)(a) to (j) of the Constitution. These include national interest, provision for debt, needs of national government and emergencies, the allocation of resources to provide basic services and meet developmental needs, fiscal capacity and efficiency of the provincial and local spheres, reduction of economic disparities, and promotion of stability and predictability. The constitutional principles taken into account in deciding on the division of revenue are briefly noted below.

### National interest and the division of resources

The national interest is encapsulated by those governance goals that benefit the nation as a whole. The spending priorities for the medium-term expenditure framework (MTEF) are informed by the 12 priorities adopted at the Cabinet lekgotla held from 20 to 22 January 2010, which are derived from the medium-term strategic framework (MTSF). A detailed analysis of how these have been allocated in government budgets can be found in Chapter 4 of the 2010 Medium Term Budget Policy Statement and Chapter 8 of the 2011 *Budget Review*.

### **Provision for debt costs**

The resources shared among the three spheres of government include proceeds from national government borrowing used to fund spending by all spheres. National government provides for the resulting debt costs to protect the integrity and credit reputation of the country.

### National government's needs and interests

The Constitution assigns exclusive and concurrent powers and functions to each sphere of government. National government is exclusively responsible for functions that serve the national interest and are best centralised. National and provincial government have concurrent responsibility for a range of functions. Changes have been made to a number of national transfers to provincial and local government to improve their efficiency, effectiveness and alignment with national strategic objectives.

### Provincial and local government basic services

Provinces and municipalities are assigned key service delivery functions such as school education, health, social development, housing, roads, and provision of electricity, water and municipal infrastructure. They have significant autonomy to allocate resources to meet basic needs and respond to provincial and local priorities, while giving effect to nationally agreed priorities. The division of revenue provides equitable shares to provinces and local government. This year's division of revenue provides additional resources to provinces for the public-sector wage agreements of 2010, to address the occupation-specific dispensation (OSD) agreements in the health and education sectors, and to fund policies on HIV and Aids treatment, agricultural support programmes and infrastructure provision for education and roads. These additions, especially in the case of HIV and Aids treatment and prevention and education infrastructure, build on significant additions to baselines in previous budgets. In the 2010 division of revenue, government reinforced its commitment to free basic services at the municipal level through a substantial increase to the local government equitable share. The 2011 division of revenue

protects these baselines, ensuring that municipalities are able to deal with the cost pressures of providing free basic services due to increased electricity charges.

### Fiscal capacity and efficiency

The Constitution assigns the primary government revenue-raising power to the national sphere. Provinces have limited revenue-raising capacity relative to the resources required to deliver provincial functions that do not lend themselves to self-funding or cost recovery. Local governments finance most of their expenditure through property rates, user charges and fees. It is recognised, however, that rural municipalities raise significantly less revenue than larger urban and metropolitan municipalities. To compensate for this, provinces receive the largest share of nationally raised revenue, and local government a portion that is substantial and will continue to grow over the medium term. The provincial equitable share formula was reviewed in 2010 and the recommendations will be implemented in 2011. A new health component will be introduced and constantly improved to ensure that its objectives are achieved. A review of the local government equitable share is under way. Both reviews should lead to significant changes that result in more efficient and effective funding arrangements.

### **Developmental needs**

Developmental needs are encapsulated in the equitable share formulas for provincial and local government and in specific conditional grants. In particular, the various infrastructure grants and growing capital budgets aim to boost the economic and social development of provinces and municipalities. Developmental needs are accounted for at two levels: first, in the determination of the division between the three spheres, which explains the continued commitment to grow provincial and local government shares of nationally raised revenue, and second, in the determination of the division within each sphere, through the formulas used for dividing national transfers among municipalities and provinces.

### **Economic disparities**

Both the equitable share and infrastructure grant formulas are redistributive towards poorer provinces and municipalities. Government remains committed to investing in economic infrastructure (roads) and social infrastructure (schools, hospitals and clinics) to stimulate economic development and job creation, and address economic and social disparities.

### Obligations in terms of national legislation

While the Constitution confers autonomy on provincial governments and municipalities to determine priorities and allocate budgets, national government retains responsibility for policy development, national mandates and the monitoring of implementation for concurrent functions. New national mandates and priorities result in increased allocations to provincial and local government, in addition to their existing baseline allocations. The 2011 MTEF and division of revenue provides additional funding for OSD agreements in health and education and HIV and Aids prevention and treatment programmes. Resources have also been made available to address backlogs in education infrastructure.

### Predictability and stability

Provincial and local government equitable share allocations are based on estimates of nationally raised revenues. These allocations are protected. In the event that nationally raised revenue falls short of the estimates, the equitable share will not be adjusted downwards. Allocations are assured (voted, legislated and guaranteed) for the first year and are transferred according to a payment schedule. To contribute to longer-term predictability and stability, forward estimates for a further two years are published alongside the annual proposal for appropriations. Changes in the forward estimates or revisions to the equitable share formulas are phased in to ensure minimal disruption.

### Need for flexibility in responding to emergencies

Government has flexibility to respond to emergencies through a contingency reserve that provides a cushion for emergencies and unforeseeable events. The 2011 division of revenue introduces two new conditional grants to allow for the swift allocation and transfer of funds to provinces and municipalities affected by emergencies that could not be provided for through the budget or adjustment budget processes. Sections 16 and 25 of the Public Finance Management Act (PFMA) (1999) make specific provision in relation to allocation of funds to deal with emergency situations while section 30(2) deals with adjustment allocations in respect of unforeseeable and unavoidable expenditure. Section 29 of the Municipal Finance Management Act (MFMA) (2003) allows a municipal mayor to authorise unforeseeable and unavoidable expenditure in an emergency or extraordinary circumstances. The 2011 Division of Revenue Bill introduces special provisions for funding immediate responses to disasters through new disaster conditional grants to provinces and municipalities.

### Part 2: The 2011 division of revenue

In preparation for the 2011 MTEF, all spheres of government have identified cost savings, eliminated non-essential expenditure and are focusing on high-priority programmes. While the economic outlook has improved in the past year, government had to raise debt significantly to ensure budget baselines were protected following the economic downturn of 2008/09. Increases to government expenditure are aimed at specific priorities; however these increases are moderate to ensure that debt costs remain within sustainable levels.

Excluding debt service costs, allocated expenditure to be shared between the three spheres amounts to R812.3 billion, R877.3 billion and R949 billion over each of the MTEF years. These allocations take into account government's spending priorities, the revenue-raising capacity and functional responsibilities of each sphere, and inputs from various intergovernmental forums and the recommendations of the FFC. The local and provincial equitable share formulas are designed and implemented in a manner that ensures desirable, stable and predictable revenue shares, and that economic and fiscal disparities are addressed.

### Government's policy priorities for the 2011 MTEF

Government's major budget priorities over the medium term include:

- Enhancing the quality of basic education and skills development.
- Improving the quality of health care and infrastructure.
- Investing in new infrastructure and proper maintenance of social and economic infrastructure networks.
- Acceleration of job creation.

Government has responded to the recession by maintaining social expenditure and continuing to invest in infrastructure, leading to improved access to basic services, expanded public transport and more schools and hospitals. These investments form part of a long-term commitment to the economy's growth.

Substantial additional resources are allocated to provinces to cover the 2010 wage agreements, the OSD agreements, the carry-through effect of previous agreements in education and health, infrastructure in education and roads, and HIV and Aids treatment.

Given the continuous large-scale rural-urban migration to South Africa's cities, infrastructure grants have been reorganised to develop the urban built environment and upgrade informal settlements.

Table W1.1 shows how the additional allocations are apportioned to the priority areas across the three spheres of government.

Table W1.1 2011 Budget priorities – additional MTEF allocations, 2011/12 – 2013/14

Table W1.1 2011 Budget priorities – additional MTE			1	
R million	2011/12	2012/13	2013/14	Total
Job creation	0.004	0.050		40.007
Job creation, small enterprise development, youth employment	2 301	3 352	4 415	10 067
Economic affairs and infrastructure development				
Public transport	2 107	3 607	4 701	10 415
Rural development and emerging farmer support	622	919	1 303	2 844
Consolidation of economic regulatory capacity	353	177	205	735
Enterprise investment programme	150	200	250	600
Infrastructure projects	_	265	297	562
Green economy	200	300	500	1 000
Broadband ICT: universal access and cost reduction	100	150	200	450
Housing and community amenities				
Human settlements upgrading and municipal services	794	1 618	2 452	4 865
Water infrastructure and services; acid mine drainage response	840	944	666	2 450
Education				
FET college expansion and skills development	2 022	3 337	4 158	9 517
School infrastructure and facilities, improved learner support materials	780	2 315	5 189	8 284
Funza Lushaka teacher bursaries and bursaries for scientific post graduate students	5	272	677	954
Health				
Hospital revitalisation and primary health family care teams	858	1 931	2 730	5 519
HIV and Aids and ARVs	60	560	1 860	2 480
Social protection				
Other (including social grants)	1 164	2 644	5 102	8 910
Public order and safety				
Police personnel expansion and training	100	400	1 283	1 783
General public services				
Municipal disaster grant	470	330	350	1 150
Provincial disaster grant	305	180	190	675
Post-recovery and reconstruction for current floods	600	_		600
Municipal charges	683	964	1 149	2 796
Compensation of employees adjustments				
Wages (including additional personnel)	11 413	13 164	14 806	39 382
Other adjustments	-5 185	-8 057	-8 685	-21 927
Total	20 742	29 573	43 797	94 112

### The fiscal framework

Table W1.2 presents medium-term macroeconomic forecasts for the 2011 Budget. It sets out the growth assumptions and fiscal policy targets on which the fiscal framework is based.

Table W1.2 Medium-term macroeconomic assumptions, 2010/11-2013/14

	2010	)/11	2011	I/12	2012	2/13	2013/14
	2010	2011	2010	2011	2010	2011	2011
R billion	Budget						
Gross domestic product	2 699.9	2 666.9	2 967.6	2 914.9	3 295.7	3 201.3	3 536.0
Real GDP growth	2.9%	3.1%	3.4%	3.6%	3.6%	4.2%	4.4%
GDP inflation	7.1%	5.9%	6.3%	5.5%	7.2%	5.4%	5.8%
National budget framework							
Revenue	643.2	666.6	721.7	729.9	807.9	806.4	904.3
Percentage of GDP	23.8%	25.0%	24.3%	25.0%	24.5%	25.2%	25.6%
Expenditure	818.1	809.9	888.3	888.9	964.3	968.1	1 053.0
Percentage of GDP	30.3%	30.4%	29.9%	30.5%	29.3%	30.2%	29.8%
Main budget balance <sup>1</sup>	-174.9	-143.4	-166.6	-159.1	-156.4	-161.7	-148.7
Percentage of GDP	-6.5%	-5.4%	-5.6%	-5.5%	-4.7%	-5.1%	-4.2%

<sup>1.</sup> A positive number reflects a surplus and a negative number a deficit.

Table W1.3 sets out the division of revenue for the 2011 MTEF, taking new policy priorities into account.

Table W1.3 Division of nationally raised revenue, 2007/08 - 2013/14

	2007/08	2008/09	2009/10	2010/11	2011/12	2012/13	2013/14
		Outcome		Revised	Mediu	m-term esti	mates
R million				estimate			
Debt-service cost	52 877	54 394	57 129	66 570	76 579	90 808	104 036
Non-interest expenditure	488 566	581 560	690 068	743 353	812 345	877 324	948 992
Percentage increase	16.9%	19.0%	18.7%	7.7%	9.3%	8.0%	8.2%
Total expenditure	541 443	635 953	747 197	809 923	888 923	968 132	1 053 029
Percentage increase	15.2%	17.5%	17.5%	8.4%	9.8%	8.9%	8.8%
Unallocated					40	330	530
Contingency reserve	-	_	_	-	4 090	11 405	23 375
Division of available funds							
National departments	242 580	289 236	345 366	359 120	380 154	408 439	439 049
Provinces	207 504	246 836	293 164	323 080	357 929	380 450	404 251
Equitable share	171 054	201 796	236 891	265 139	288 493	305 725	323 604
Conditional grants	36 451	45 040	52 073	57 941	69 436	74 724	80 647
Gautrain Ioan	_	_	4 200	_	_	_	_
Local government	38 482	45 487	51 537	61 152	70 171	77 029	82 317
Equitable share <sup>1</sup>	20 676	25 560	23 845	30 559	34 108	37 573	39 960
General fuel levy sharing	_	_	6 800	7 542	8 573	9 040	9 613
Conditional grants	17 806	19 928	20 892	23 051	27 490	30 416	32 743
Total	488 566	581 560	690 068	743 353	808 254	865 919	925 617
Percentage shares							
National departments	49.7%	49.7%	50.0%	48.3%	47.0%	47.2%	47.4%
Provinces	42.5%	42.4%	42.5%	43.5%	44.3%	43.9%	43.7%
Local government	7.9%	7.8%	7.5%	8.2%	8.7%	8.9%	8.9%

<sup>1.</sup> With effect from 2006/07, the local government equitable share includes compensation for the termination of Regional Services Council (RSC) and Joint Services Board (JSB) levies for metros and district municipalities. From 2009/10 the RSC levies replacement grant will only be allocated to district municipalities.

Table W1.4 shows how additional resources are divided among the three spheres of government. The new priorities and additional allocations are accommodated through shifting of savings towards priorities.

Table W1.4 Changes over baseline, 2011/12 - 2013/141

R million	2011/12	2012/13	2013/14
National departments	9 947	14 682	24 188
Provinces	9 507	13 385	17 280
Local government	1 288	1 506	2 329
Allocated expenditure	20 741	29 573	43 796

<sup>1.</sup> Excludes shifting of savings towards priorities over the MTEF.

Table W1.5 sets out schedule 1 of the Division of Revenue Bill, which reflects the legal division of revenue between the three spheres. In this division, the national share includes all conditional grants to the other two spheres in line with section 214(1) of the Constitution, and the provincial and local government allocations reflect their equitable shares only.

Table W1.5 Schedule 1 of the Division of Revenue Bill, 2011/12 - 2013/14

	2011/12	2012/13	2013/14
	Column A	Colun	nn B
R million	Allocation	Forward e	stimates
National <sup>1, 2</sup>	566 323	624 833	689 464
Provincial	288 493	305 725	323 604
Local	34 108	37 573	39 960
Total	888 923	968 132	1 053 029

<sup>1.</sup> National share includes conditional grants to provinces and local government, general fuel levy sharing with metropolitan municipalities, debt service cost and the contingency reserve.

The 2011 *Budget Review* sets out in detail how constitutional issues and government's priorities are taken into account in the 2011 division of revenue. It focuses on the economic and fiscal policy considerations, revenue issues, debt and financing considerations, and expenditure plans of government. This memorandum should be read with the 2011 *Budget Review*, as aspects of national, provincial and local government financing are discussed in some detail in Chapter 8 of the review.

### Part 3: Response to the recommendations of the FFC

Section 214 of the Constitution and section 9 of the Intergovernmental Fiscal Relations Act (1997) require the FFC to make recommendations in April every year, or soon thereafter, on the division of revenue for the coming budget. The FFC complied with this obligation by tabling its *Submission for the Division of Revenue 2011/12* to Parliament in May 2010. This part of the explanatory memorandum complies with the Constitution and section 10 of the Intergovernmental Fiscal Relations Act by setting out how government has taken the FFC's recommendations into account when determining the division of revenue for the 2011 MTEF.

The 2011/12 recommendations cover four interrelated areas. The first chapter deals with the global economic crisis, fiscal frameworks and coping with vulnerabilities; the second chapter discusses options for social assistance reform during a period of fiscal stress; the third chapter focuses on an effective, efficient and transparent intergovernmental fiscal system and the fourth chapter discusses the intergovernmental fiscal issues in urban public transport.

### Chapter 1: The global economic crisis, fiscal frameworks and coping with vulnerabilities

The FFC recommends that in the short term, government should continue to strive for fiscal consolidation by limiting the growth in entitlement spending to successful programmes, and refocus expenditure on improved service quality. In particular, child support and old-age pension grants should be expanded, and the high access levels to education and health services achieved despite the global

<sup>2.</sup> The direct charges for the provincial equitable share are netted out.

economic crisis must be maintained. Government should also reprioritise expenditure towards repair and maintenance by emphasising existing projects and initiating new ones.

### Government response

Government agrees that the implications of the global economic crisis and reduced fiscal space necessitate fiscal consolidation. Government has refocused existing budgets towards government priorities and more efficient spending. The number of social grant beneficiaries has more than tripled between 2002/03 and 2010/11 and a committee of inquiry is in place to investigate more cost-effective beneficiary payment options for the South African Social Security Agency.

The majority of additions to baseline over the 2011 MTEF target health, education, roads and the eradication of informal settlements. These additions are on top of substantial increases in previous budgets and will enable provinces to increase the quality of health and education services through improved infrastructure, conditions of employment, and provision of medicines. Government is finalising asset registers as it implements the Government Immovable Asset Management Act (2007) to ensure that infrastructure assets are properly maintained.

### Block grant for health, education and social development

The FFC recommends that government should introduce a block grant for education, health and social development over the medium to long term, to fund clearly defined and costed outcomes.

### Government response

As part of the review of the provincial equitable share formula, government considered proposals made by the FFC in 2009, which were included in their recommendations on the 2010/11 division of revenue. These proposals dealt with challenges created by the design of the intergovernmental fiscal system and a number of issues related to the provincial equitable share formula. Government has not fully considered decentralising further fiscal powers to provinces. Thus, government agreed to separate the review of the provincial equitable share formula from the devolution of fiscal powers.

Government considered the proposal on block grants as part of the provincial equitable share review, and it has concluded that block grants are potentially unconstitutional.

### Independent cost effectiveness and quality reviews

Undertake independent cost effectiveness and quality reviews of education, health and the social wage in both the public and private sectors.

### Government response

Government agrees that independent cost effectiveness and quality reviews should be undertaken. The Ministry of Performance Monitoring and Evaluation would perform such reviews in the period ahead.

### Chapter 2: Options for social assistance reform during a period of fiscal stress

The FFC recommends that during a period of fiscal stress, government should protect social assistance expenditure as far as possible during fiscal consolidation. Government should protect the relative simplicity of the social assistance system, especially when contemplating reform options. The FFC also recommends that government pilot conditional cash transfer and workfare programmes on a small scale and evaluate them to expand successful pilots, strengthen non-cash complementary social developmental services by emphasising quality improvements within defined resource limits, and avoid universal income grants.

### Government response

Government supports these recommendations with the exception of the piloting of a workfare programme. Although exploring potential policy initiatives through pilot programmes has many benefits, giving cash benefits to individuals can be problematic. It will be extremely difficult for government to exclude people from the pilot and if it fails it will be complicated to withdraw support.

### Chapter 3: Towards an effective, efficient and transparent intergovernmental fiscal system

The FFC recommends that when introducing and terminating conditional grants, national departments must introduce a mandatory, systematic process for designing and planning individual conditional grants that covers incentive effects, administrative accountability arrangements and stipulates regular review periods and exit strategies of the grant. Government should also ensure there is an independent evaluation of the grant performance at entry, midterm and end of the grant.

### Government response

Guidelines on how to introduce and terminate conditional grants are available and the recommendation that criteria for disbanding the grant should be identified upon introduction is welcomed. It will be beneficial in the long term to develop capacity within national and provincial departments, provincial treasuries and municipalities to perform independent evaluations of grant performance. However, the fiscal system should continue to be responsive to the needs of government. Government should have the flexibility of introducing and terminating conditional grants where there is a clear rationale to do so.

### Allocation criteria of conditional grants

The FFC recommends that government should make the criteria for dividing grant allocations transparent.

### Government response

The allocation criteria for grants are explained in the relevant conditional grant frameworks.

### Emphasise non-financial data reporting

The FFC recommends that government continue to emphasise the importance of non-financial data reporting, linking outer-year allocations to independently evaluated performance information and gazette expected deliverables.

### Government response

Government agrees and will continue to emphasise the importance of non-financial performance reporting. Performance audits at provincial level should bring about substantial improvements to the quality of performance information. Reforms are in place to develop proper reporting systems for financial reporting in municipalities, and once these are in place the focus will shift to non-financial reporting. Government seeks to improve alignment of budgets and measurable objectives through its budget reforms.

### Results-based accountability

The FFC recommends that through incentive grants, national departments must make accounting for delivery a prerequisite for most conditional grants. They should encourage designing grants that

explicitly promote innovation in sub-national governments and strengthen incentives for optimal service delivery.

### Government response

Government supports this recommendation. However, the administrative ability in provincial and local government must be strengthened before incentives for innovation are likely to have the desired effect.

### Budget allocation process

The budget allocation process must specifically follow the conditional grant frameworks and this should be monitored periodically through section 32 of the PFMA and section 71 of the MFMA.

### Government response

Government agrees with the proposal and monitoring is taking place through section 32 of the PFMA and section 71 of the MFMA.

### Improve general performance of municipalities in revenue improvement and collection

The FFC recommends that government adopt standard indicators or early warning systems to measure and detect fiscal stress in municipalities. In addition to those already prescribed, these indicators should be pre-conditions for instigating mandatory provincial interventions. Monthly budget statements should report on actual revenue per source and on the percentage of collected revenue to the total value of billed revenue.

Government should legislate ratios for revenue collection as one of the key performance areas against which to assess overall municipal performance. Collection and coverage ratios must be used as standard indicators to revenue collection performance assessment across municipalities.

Excessive levels of municipal debt must be reduced through constant taxpayer education and incentives to improve the quality of services in general. Local government should be able to issue garnishee orders and the judicial system should have dedicated courts to deal with outstanding municipal accounts until the debt is reduced to acceptable levels.

The FFC recommends that government should make a concerted effort to estimate the fiscal capacity and fiscal effort of municipalities to dispel the perception that certain municipalities will never be financially viable.

Municipalities should have broad revenue improvement programmes that focus on administrative streamlining as well as revenue and expenditure interventions, and efficiency-based interventions. These interventions should be specific to local economic circumstances so that the emphasis is appropriate.

Effective revenue management processes, good financial management and the provision of good quality services should underpin revenue improvement programmes. These should only be conducted when municipalities have maximised the collection of local and outstanding revenue sources and should be subject to empirical tests.

### Government response

National government agrees that improving municipalities' management of their own revenue sources is an essential part of placing local government finances on a more sustainable base, while strengthening local accountability and service delivery. National government's estimates indicate that improving municipalities' management of their revenues has the potential to yield substantially greater returns than the additional funds local government is budgeted to receive through the local government equitable share and conditional grants over the 2011 MTEF.

National government supports the recommendation that each municipality should have a broad revenue improvement programme in place. It is for this reason that national government has emphasised the importance of fostering a culture of payment for local government services through various initiatives, and why National Treasury and the Department of Cooperative Governance and Traditional Affairs are working together to provide practical support to municipalities on initiatives to strengthen their own revenue policies, procedures and processes.

Regarding the FFC's specific proposals for revenue improvement and collection:

National government has an early warning system that monitors municipal finances. This system is anchored by the budget formats prescribed in terms of the municipal budget and reporting regulations, the associated funding compliance assessment (see MFMA Circular 42) and monthly financial reports required in terms of section 71 of the MFMA. This system has proved to be very effective at identifying fiscal stress in municipalities. The real challenge is finding people with the necessary skills and integrity to tackle the identified problems.

The new budget and reporting formats emphasise cash-flow budgeting and reporting, which lays the foundation for closely monitoring actual revenue collections against billed revenues. However, it is not practical to insist on one-to-one reporting of revenue collected against revenue billed – because of the way systems are set up, the manner in which payments are allocated to settle bills, and the impact that debt repayments have on current revenue numbers.

The relevant acts that govern revenue collection by municipalities place positive obligations on municipal councils and municipal managers to put policies, processes and systems in place to ensure sound management of the various own revenue sources. National government is progressively strengthening its ability to monitor compliance with this legislation. The primary challenge is to get municipalities to make it a priority and manage it effectively.

Legally, municipalities may apply to a court for a garnishee order to recover outstanding debts from a customer. However, this is a costly approach and is therefore not widely used. National government believes that at this stage there is far more to be gained from improving the accuracy and completeness of municipal billing systems, and putting in place proper customer relations capacity. Experience has shown that these yield far better results for debt collection, but the idea of creating municipal debt courts can be considered in the medium term.

The notion that certain municipalities will never be financially viable is a misrepresentation of both the design of the local government fiscal framework and the practical reality of local economies. The fiscal framework is intended to ensure an equitable distribution of resources between the rich and poor areas of the country – but it does not absolve municipalities of the responsibility to raise property rates and services charges for the non-poor living within their municipal boundaries. Information on municipal own revenues indicates that many smaller municipalities are failing to do so. National government is investigating the issue of developing measures of fiscal capacity – it is by no means a straightforward task given the impact that traditional land and different modes of service delivery have on many municipalities' rates bases and service revenues.

### Local government equitable share

The FFC recommends that the institutional component and revenue-raising correction component are reformed, so that the institutional component assists poor municipalities and the step structure of the differentiated tax mechanism of the revenue-raising correction component is removed. The current method of using actual property rates and own revenues collected to calculate the correction is problematic as poor fiscal effort and reporting is used as a measure for additional funding, which could

be seen as a contradiction of section 227(2) of the Constitution. The FFC recommends that this practice is replaced with alternative methods of revenue prediction.

### Government response

Government agrees that changes are required to improve the way the local government equitable share is allocated between municipalities. To assess whether the local government equitable share formula results in an equitable division among individual municipalities (horizontal division) as stipulated in sections 214 and 227 of the Constitution, the formula should be analysed as a whole – rather than subcomponents in isolation. Several adjustments are made to the local government equitable share formula this year and further changes will be made after a full review of the formula in 2011 (discussed in part six of this annexure).

The institutional component has been changed in 2011, adjusting the amount allocated to municipalities in terms of their poverty rate. This means that poorer municipalities will receive increased allocations. Details of this change are set out in part five of this annexure.

The revenue-raising correction component was introduced in 2005 following the FFC's recommendation that equitable share allocations should be adjusted to take account of the ability of municipalities to raise their own revenues. The original model used to predict municipal revenue-raising capacity became distorted over time and was replaced in 2008 with a simple model that uses reported revenue raised in previous years to predict future revenue-raising capacity. Although the Municipal Property Rates Act (2004) has been amended to take effect by not later than 1 July 2011, stipulating that municipalities must implement valuation rolls, the system is not yet at a stage where property rates revenue estimates can inform municipal equitable share allocations. In the absence of accurate property valuation rolls for all municipalities and with no official data on economic activity at municipal level, it is very difficult to construct a model that would provide fair estimates of municipalities' capacity to raise their own revenues.

During 2010, National Treasury has worked together with officials from the FFC, SALGA and the Department of Cooperative Governance and Traditional Affairs to try to develop an alternative model. Due to the lack of adequate data, these attempts did not succeed in producing a model that could provide plausible predictions of the revenue-raising capacity of all municipalities. Using these models in the formula would have produced large distortions in allocations to municipalities and would risk contravening section 214(2)(i) of the Constitution that requires that allocations "take into account the desirability of stable and predictable allocations of revenue shares." The current method used for predicting municipalities' capacity to raise own revenues is therefore the fairest and most accurate method available at present and cannot be changed this year.

Government agrees to the recommendation to remove the stepped structure from the revenue-raising correction and a smooth curve has been used instead to calculate the differentiated "revenue correction" rate applied to each municipality. As outlined in part five of this annexure, the value of the revenue-raising correction has also been reduced in this MTEF to compensate for demographic changes not reflected in the formula, due to the use of 2001 Census data. The total value of the revenue-raising correction component of the formula will be reduced, giving it less of an impact on the allocations to individual municipalities.

### Regionalising municipal services: the electricity distribution industry in South Africa

The FFC does not support the blanket regionalisation approach proposed in the 17th amendment to the Constitution. The electricity distribution industry (EDI) restructuring process should consider an approach that allows for differences in performance.

The FFC made a number of recommendations with respect to the EDI restructuring and establishment of regional electricity distributors.

### Government response

The FFC recommendations were proposed before government resolved not to continue with the restructuring and establishment of the regional electricity distributors. The FFC's recommendations are therefore moot.

### Chapter 4: Intergovernmental fiscal issues in urban public transport

### Urban public transport

The FFC recommends that the Passenger Rail Agency of South Africa (PRASA) and cities should ensure that investment projects on rail and roads infrastructure are aligned and coordinated for optimal use of resources.

### Government response

The regulation function for public transport is being developed in municipalities to support the alignment and coordination of investment and planning for public transport services. Changes to funding flows to improve coordination include the transfer of the *public transport operations grant* in the 2011 Division of Revenue Act. Talks are also under way with the national Department of Transport to discuss the transfer of rail operational subsidies to cities, rather than directly to PRASA, to ensure integrated network design and management.

### Municipal Land Transport Fund

The FFC recommends that government make an immediate decision on the funding for the Municipal Land Transport Fund, as delays could negatively affect the financial position of affected municipalities.

### Government response

The National Land Transport Act (NLTA) (2009) requires the creation of a municipal land transport fund in each municipality. Municipalities can deposit grants from national and provincial governments into the Municipal Land Transport Fund, as well as user charges from transport services and revenue from local taxes. If municipalities need to raise additional local taxes for the fund, they can apply to the Minister of Finance to authorise such a tax. To date no such request has been received.

### Funding for commuter rail sector

The FFC recommends that PRASA should ensure that funding for investment on the commuter rail sector prioritises corridors already identified as A and B in the National Rail Plan.

### Government response

Government supports the proposal.

### Comprehensive review of the urban form

The FFC recommends a comprehensive review conducted by relevant stakeholders into the costs associated with current urban form in a selection of major South African cities. This review will improve the efficiency of land-use patterns. The current mechanisms and basis for distributing transport subsidies should be reviewed by the Department of Transport, National Treasury and other key stakeholders to promote the efficiency of urban transport and land-use systems, taking equity and distributional effects on households into account.

### Government response

Government recognises the need to review fiscal and financing arrangements for large cities, particularly in light of recent policy development resulting in the transfer of public transport, human settlements and land management functions to cities. This requires the development of new fiscal instruments, monitoring and capacity-support arrangements. The introduction of the new *urban settlements development grant* to address informal settlements and accelerate urban land release should help to improve the efficiency of land-use patterns. The current system of transport subsidies is not linked to ridership levels in cities, nor does it cover the full cost of ridership, thus placing ridership risks with the cities. Government will consider possible further changes to the financing of urban housing and public transport. A key challenge is to improve service delivery while breaking down the apartheid settlement patterns that continue to marginalise the poor in cities.

Cities will require capacity support to take on these new roles. Government looks forward to FFC analysis and recommendations on key fiscal issues, options and risks regarding this issue.

### National Land Transport Act

The FFC recommends that government examines the potential financial implications of the NLTA on municipalities, and identifies dedicated funding streams for public transport.

### Government response

Funding has been allocated to the Department of Transport for the implementation of the NLTA over the MTEF. National Treasury will review funding options to support the creation of regulatory capacity at local level. Funding also needs to be shifted from provinces to local government, as the act transfers the function from provinces to municipalities.

### National Household Travel Survey

The FFC recommends that the Department of Transport should regularly update the South African National Household Travel Survey.

### Government response

Government supports the recommendation. Funding has been allocated to the Department of Transport to begin updating the National Household Travel Survey in 2011/12.

### Part 4: Provincial allocations

Sections 214 and 227 of the Constitution require that an equitable share of nationally raised revenue be allocated to provincial government to enable it to provide basic services and perform the other functions allocated to this sphere.

An amount of R40.2 billion is added to the provincial baseline over the next three years. The provincial equitable share baselines are revised upwards by R30.1 billion and conditional grants are increased by R10.1 billion. National transfers to provinces increase from R323.1 billion in 2010/11 to R357.9 billion in 2011/12. Over the three-year period, provincial transfers will grow at an average annual rate of 7.8 per cent to R404.3 billion in 2013/14.

Table W1.6 below sets out the total transfers to provinces for the 2011/12 financial year, which amounts to R357.9 billion, with R288.5 billion allocated to the provincial equitable share and R69.4 billion to conditional grants, which includes an unallocated R305 million for the *provincial disaster grant*, but

does not include an indirect transfer to provinces of R700 million for the *school infrastructure backlogs* grant.

Table W1.6 Total transfers to provinces, 2011/12

R million	Equitable share	Conditional grants	Total transfers
Eastern Cape	44 120	8 896	53 016
Free State	17 521	4 976	22 497
Gauteng	50 428	14 665	65 094
KwaZulu-Natal	62 928	13 314	76 241
Limpopo	36 349	6 912	43 261
Mpumalanga	23 379	5 197	28 576
Northern Cape	7 743	2 473	10 216
North West	19 271	4 541	23 813
Western Cape	26 754	8 156	34 910
Unallocated	_	305	305
Total	288 493	69 436	357 929

### Provincial equitable share

At 80.6 per cent of national transfers to provinces in 2011/12, the equitable share constitutes the main source of revenue for meeting provincial expenditure responsibilities. The proposed revisions of R7.8 billion, R10.3 billion, and R12 billion bring the equitable share allocations to R288.5 billion in 2011/12, R305.7 billion in 2012/13, and R323.6 billion in 2013/14. These revisions result in the provincial equitable shares increasing 8.8 per cent between 2010/11 and 2011/12, and 6.9 per cent over the MTEF in nominal terms.

### Policy priorities underpinning equitable share revisions

The revisions to baseline equitable share allocations provide for personnel and policy adjustments. The personnel adjustments provide for the 2010 wage agreements and additional resources to provinces to deal with outstanding OSD costs in education and new and outstanding OSD costs in health. Policy adjustments seek to improve access to health services and to help provinces stabilise their health systems and prepare for the introduction of national health insurance.

### Review of the provincial equitable share formula

### Background

Section 227 of the Constitution entitles provinces to an equitable share of nationally raised revenue to enable them to provide basic services and perform their functions. Section 214 of the Constitution requires that before providing for the equitable division of revenue through the Division of Revenue Act, government must consider a number of factors listed in Section 214(2)(a) to (j), which provide a framework to guide revisions to the provincial equitable share formula.

In 2004 it was decided that national government should take responsibility for social assistance. These funds were taken out of the provincial equitable share and the formula was reviewed and some minor changes were made.

An objective redistributive formula is used to divide the equitable share among provinces. In 2006 the Budget Council called for a comprehensive review of the provincial equitable share formula to address concerns that it was not flexible enough to deal with expenditure responsibilities driving provincial budgets. The review was conducted in two phases; the first phase was concluded by the FFC in 2009. The FFC identified the policy imperatives that should underpin the review and proposed options for

reform of the provincial fiscal powers and the provincial fiscal framework. National Treasury lead the second phase of the review and the recommendations were presented to Budget Council on 4 October 2010. Government responds to the FFC's recommendations relating to the formula and the provincial division of revenue, but has not considered FFC's proposed changes to the fiscal powers of provinces.

Based on the review, a number of reforms will be introduced to the provincial equitable share formula for the 2011 Budget. The structure of the provincial equitable share (six components) remains unchanged. A new health component will be introduced and the weights of the education, health and basic components will be revised. The unconditional nature of the equitable share formula, the oversight and support responsibilities of national departments in charge of concurrent functions and the discretion of provincial governments have not been affected by these changes.

### Formula incentives

The formula consists of six components. The components' respective weights should be considered together, along with the interrelated functions that provinces perform, taking into account the autonomy provinces have regarding the way in which they address government priorities.

Of the six components, only the institutional component and the economic activity component, which account for 6 per cent of the provincial equitable share, are not informed by demographic data. Therefore 94 per cent of the equitable share is influenced by population movements and the characteristics of provincial populations.

The education component encourages enrolment at schools, but due to the weighting of the subcomponents, penalises provinces that allow retention and repetition. The value of any province's health component, even in its new form, will not change significantly if the demand for health services in the province falls. This creates an incentive for provinces to manage their overall health care and social programmes well and improve the health of their population, reducing the burden on the health system and freeing up resources for other purposes.

The poverty component ensures the formula is redistributive and the economic component creates an incentive for provinces to create favourable economic conditions. The economic component is weighted according to the limited economic functions provinces are required to perform. The fixed institutional component ensures all provinces receive administrative funding, and due to the fixed value of this component, provinces that are administratively efficient are able to free up resources for service delivery.

### Improving the formula's ability to deal with provincial expenditure responsibilities

The provincial equitable share is agreed to during the vertical division of revenue and the amount allocated to it is an indication of the value placed on the functions performed by provinces relative to the responsibilities of the two other spheres of government. The role of the formula is to ensure that each province receives its share of the provincial equitable share.

The provincial equitable share formula, as a blunt instrument, facilitates robust democratic engagement and provincial priority setting, which is consistent with the principles of the intergovernmental fiscal system. Resource allocation and budgeting decisions are made at the provincial level. Thus, poor resource and budgeting decisions cannot be attributed to the formula.

The intergovernmental fiscal system and implementation of concurrent functions can be improved through the strengthening of two key institutions. First, national departments responsible for concurrent functions must ensure that the policy standards for their sector are in place and legislated, so that provinces can align their activities with these standards. By requiring compliance with norms and standards, national departments allow provinces to choose how to achieve compliance, which encourages efficiency and innovation. Second, the capacity of provincial treasuries needs to be strengthened so that the cost implications of policies can be fully understood and monitored. A capacitated provincial treasury

enables the provincial executive to prepare and defend a balanced budget that addresses government priorities.

# Changes to the provincial equitable share formula

A number of new components to the formula were considered, but all options were constrained by the availability of regularly updated official demographic data sets.

A new social development component was considered, but as the key dependent variable was poverty, it was agreed that it would be more efficient to simply use Statistics South Africa (StatsSA) poverty estimates, used in the existing poverty component. As the poverty component already exists it was decided that a social development component was unnecessary. A number of possibilities were considered for the education formula, particularly the costed norms approach often promoted by the FFC. However, this approach is too subjective, and as the incentives created by the existing component are supported it was left unchanged.

Since the last review of the provincial equitable share formula, the expenditure shares of education and health have changed. Health and education are still the two largest expenditure items on provincial budgets. Total enrolment has stabilised and is starting to fall off, however the pressure on health budgets continues and the relative share of these sectors has changed. The average expenditure on health and education as a proportion of total provincial expenditure for the period 2007/08 to 2009/10 was used to estimate new weights for these components.

The health sector proposed a new formula for the health component and it is discussed in detail below.

# The equitable share formula

The formula is reviewed and updated with new data annually. For the 2011 MTEF, the equitable share formula has been updated with data from the 2010 mid-year population estimates, 2010 Education School Realities and output data from the health sectors. The risk-adjusted capitation index is based on data from the Risk Equalisation Fund, the 2008 gross domestic product by region (GDP-R) and the 2005 Income and Expenditure Survey. The impact of these updates on the provincial equitable shares is to be phased-in over three years (2011/12 to 2013/14).

Because the formula is largely population driven, the allocations it generates capture shifts in population across provinces, leading to changes in the relative demand for public services.

#### Phasing-in of the formula

To mitigate the impact of the new data updates on provincial equitable shares, and the new health component, the new shares are phased in over the MTEF. Table W1.7 shows the revised weighted provincial equitable shares for the period 2010/11 to 2013/14.

Table W1.7 Implementation of the equitable share weights, 2011/12 - 2013/14

	2010/11	2011/12	2012/13	2013/14		
	Weighted shares	2011 MTEF weighted shares 3-year phasing				
Percentage						
Eastern Cape	15.5%	15.3%	15.2%	15.1%		
Free State	6.1%	6.1%	6.0%	6.0%		
Gauteng	17.3%	17.5%	17.6%	17.8%		
KwaZulu-Natal	21.7%	21.8%	21.9%	21.9%		
Limpopo	12.8%	12.7%	12.5%	12.3%		
Mpumalanga	8.2%	8.1%	8.0%	8.0%		
Northern Cape	2.7%	2.7%	2.7%	2.7%		
North West	6.6%	6.7%	6.7%	6.8%		
Western Cape	9.1%	9.2%	9.3%	9.4%		
Total	100.0%	100.0%	100.0%	100.0%		

# Summary of the structure of the formula

The formula, shown in Table W1.8 below, consists of six components that capture the relative demand for services between provinces and take into account specific provincial circumstances. The components of the formula are neither indicative budgets nor guidelines as to how much should be spent on those functions in each province or by provinces collectively. Rather, the education and health components are weighted broadly in line with historical expenditure patterns to provide an indication of relative need. Provincial executive councils have discretion regarding the determination of departmental allocations for each function, taking into account the priorities that underpin the division of revenue.

Table W1.8 Distributing the equitable shares by province, 2011 MTEF

	Education	Health	Basic share	Poverty	Economic activity	Institu- tional	Weighted average
	48%	27%	16%	3%	1%	5%	100%
Eastern Cape	16.7%	14.1%	13.5%	16.7%	7.5%	11.1%	15.1%
Free State	5.6%	5.9%	5.7%	5.9%	5.2%	11.1%	6.0%
Gauteng	15.5%	20.2%	22.4%	15.7%	33.1%	11.1%	17.8%
KwaZulu-Natal	23.1%	22.4%	21.3%	22.9%	16.4%	11.1%	21.9%
Limpopo	13.9%	10.6%	10.9%	14.3%	7.2%	11.1%	12.3%
Mpumalanga	8.4%	7.0%	7.2%	8.6%	7.6%	11.1%	8.0%
Northern Cape	2.2%	2.3%	2.2%	2.5%	2.3%	11.1%	2.7%
North West	6.3%	7.0%	6.4%	7.5%	6.5%	11.1%	6.8%
Western Cape	8.3%	10.5%	10.4%	6.0%	14.3%	11.1%	9.4%
Total	100.0%	100.0%	100.0%	100.0%	100.0%	100.0%	100.0%

For the 2011 Budget, the weight of the education component changes from 51 per cent to 48 per cent, the weight of the health component changes from 26 per cent to 27 per cent and the basic component changes from 14 per cent to 16 per cent. The new weights for the education and health components are aligned with their expenditure shares, excluding expenditure on conditional grants over the past three years. The basic share is a residual component, changing to capture the balance in the changes mentioned above. The formula components are set out as follows:

- An *education share* (48 per cent) based on the size of the school-age population (ages 5-17) and the number of learners (Grade R to 12) enrolled in public ordinary schools.
- A *health share* (27 per cent) based on a combination of risk-adjusted capitation index for the population, which takes into account the health risks associated with the demographic profile of the population and the relative share of case loads in hospitals. The risk-adjusted capitation is given a 25 per cent weighting and the case load (output component) is given a 75 per cent weighting.

- A basic share (16 per cent) derived from each province's share of the national population.
- An institutional component (5 per cent) divided equally between the provinces.
- A poverty component (3 per cent) reinforcing the redistributive bias of the formula.
- An economic output component (1 per cent) based on GDP-R data.

#### Education component

The education component is intended to enable provinces to fund school education, which amounts to about 90 per cent of provincial education spending. The formula uses school-age population (ages 5-17), based on Census 2001, and actual enrolment data drawn from the 2010 Education School Realities to reflect relative demand for education, with each element assigned a weight of 50 per cent. Table W1.9 shows the impact data updates on the education component shares.

Table W1.9 Impact of changes in school enrolment on the education component shares

	Age cohort	2009	2010	Changes in	Weighted	l average	Difference
	5 - 17	School	School	enrolment			in weighted
Learner		enrolment	enrolment		2010 MTEF	2011 MTEF	average
numbers					10.00/		2 1221
Eastern Cape	2 151 992	2 076 400	2 052 386	-24 014	16.8%	16.7%	-0.13%
Free State	760 486	656 754	654 704	-2 050	5.6%	5.6%	-0.02%
Gauteng	1 931 719	1 939 231	1 974 066	34 835	15.4%	15.5%	0.11%
KwaZulu-Natal	3 013 243	2 816 974	2 806 988	-9 986	23.2%	23.1%	-0.08%
Limpopo	1 798 862	1 707 280	1 706 401	-879	13.9%	13.9%	-0.03%
Mpumalanga	1 074 972	1 035 469	1 036 432	963	8.4%	8.4%	-0.01%
Northern Cape	280 975	267 709	269 392	1 683	2.2%	2.2%	0.00%
North West	826 218	741 892	759 114	17 222	6.2%	6.3%	0.06%
Western Cape	1 094 565	973 136	1 000 616	27 480	8.2%	8.3%	0.10%
Total	12 933 032	12 214 845	12 260 099	45 254	100.0%	100.0%	_

# Health component

The previous health component was based on population with medical aid, using the 2009 General Household Survey. Mid-year population estimates are used to update the subcomponent "people without medical aid." People without medical aid were weighted four times as much as people with medical aid.

Government has adopted a new health component that will be used in the 2011 division of revenue. This component combines a risk-adjusted capitation index based on the Risk Equalisation Fund (REF) and output data from public hospitals to estimate each province's share of the health component. These two methods work together to balance needs (risk-adjusted capitation) and demands (output component).

Cost curves have been developed from the REF using data on consumption of health services in the private sector. These curves are based on health services used by people on medical aid and disaggregated by age and gender. For the purposes of the health formula, these curves have been weighted for HIV and Aids and pregnancy. The resulting curves are used to estimate an age- and gender-adjusted weighted population for each district and province. The risk-adjusted capitation is estimated using demographic data of the non-insured population in each district and province drawn from the basic population estimates in the General Household Survey (2009). In the absence of data on health consumption patterns in the public sector, the health formula has had to rely on REF data.

Output data or workload data (patient-day equivalent) has been provided by the District Health Information System (DHIS) for 2007/08 to 2009/10. Hospitals are divided by level of care, based on classifications used in the DHIS. A normative estimate of the care mix per hospital level has been developed, and average unit costs for this mix of care have been formulated, so that the relative cost of

services provided at the different levels can be estimated. The estimated level of care is based on a modified version of the National Planning Framework's bed mix by hospital and standard bed-to-population ratios. This approach is normative to deal with the various types of hospitals in provinces.

The unit costs used are derived from a combination of actual data (BAS and DHIS) and previous information comparing normative costs for different hospital types. Each patient-day equivalent is multiplied by a normative costing to estimate a normative health budget per district and aggregated to the provincial level. This is used to estimate a total health budget and each province's share of the output component is based on their share of the total normative budget.

Table W1.10 compares the weighted shares for the health component using the old formula with the new formula. The first column shows the weighted shares in 2010 and what they would be in 2011 using the updated General Household Survey population estimates. The first two columns on the left show the weighted share for 2010 and what the shares would be in 2011 with data updates if the health formula was not changed. The weighted share 2011 column is estimated by weighting the risk-adjusted component (75 per cent) and the output component (25 per cent). These weights will be reviewed as more data becomes available.

Table W1.10 New health component weighted shares

	with G mid-year <sub>l</sub>	nares updated HS and population nates	Weighted shares 2011 MTEF	Difference in weighted shares
Percentage	2010 MTEF	2011 MTEF	25% output, 75% risk adjusted	
Eastern Cape	14.0%	14.1%	14.1%	0.03%
Free State	5.9%	5.6%	5.9%	0.38%
Gauteng	19.9%	20.8%	20.2%	-0.62%
KwaZulu-Natal	22.2%	22.1%	22.4%	0.27%
Limpopo	11.3%	11.7%	10.6%	-1.10%
Mpumalanga	7.5%	7.4%	7.0%	-0.41%
Northern Cape	2.4%	2.2%	2.3%	0.11%
North West	6.7%	6.5%	7.0%	0.47%
Western Cape	10.1%	9.6%	10.5%	0.88%
Total	100.0%	100.0%	100.0%	_

# Poverty component

The poverty component introduces a redistributive element within the formula and is assigned a weight of 3 per cent. The poor population consists of persons who fall in quintiles 1 and 2 based on the 2005 Income and Expenditure Survey. Each province's share is then expressed as the percentage of the poor population residing in that province, where the population figure is drawn from the 2010 mid-year population estimates. Table W1.11 shows the poverty quintiles of the Income and Expenditure Survey, basic component value and the weighted share of the poverty component per province.

Table W1.11 Comparison of current and new poverty component weighted shares

	IES	Curi	rent (2010 M <sup>-</sup>	ΓEF)	Ne	New (2011 MTEF)		
	Survey 2005 (Q1+Q2)	Basic compo- nent value	Poor population	Weighted shares	Basic compo- nent value	Poor population	Weighted shares	in weighted shares
Eastern Cape	49.8%	6 649	3 314	16.7%	6 744	3 361	16.7%	0.00%
Free State	41.7%	2 902	1 211	6.1%	2 825	1 178	5.9%	-0.25%
Gauteng	28.1%	10 757	3 025	15.3%	11 192	3 147	15.7%	0.40%
KwaZulu-Natal	43.2%	10 449	4 511	22.8%	10 646	4 596	22.9%	0.11%
Limpopo	52.9%	5 227	2 763	13.9%	5 440	2 875	14.3%	0.37%
Mpumalanga	47.7%	3 607	1 720	8.7%	3 618	1 725	8.6%	-0.09%
Northern Cape	44.9%	1 148	515	2.6%	1 104	496	2.5%	-0.13%
North West	46.9%	3 224	1 513	7.6%	3 201	1 502	7.5%	-0.16%
Western Cape	23.1%	5 357	1 237	6.2%	5 224	1 206	6.0%	-0.24%
Total		49 321	19 809	100.0%	49 991	20 087	100.0%	-

# Economic activity component

The economic activity component is a proxy for provincial tax capacity and is assigned a weight of 1 per cent. For the 2011 MTEF, 2008 GDP-R data is used. Table W1.12 shows the impact of the revised weighted shares of the economic activity component. The right-hand column shows changes as a result of relative growth of provincial contributions to GDP.

Table W1.12 Current and new economic activity component weighted shares

	Current (20	10 MTEF)	New (2011 MTEF)		Difference in
	GDP-R, 2007 (R million)	Weighted shares	GDP-R, 2008 (R million)	Weighted shares	weighted shares
Eastern Cape	155 520	7.8%	170 502	7.5%	-0.31%
Free State	108 892	5.4%	119 317	5.2%	-0.22%
Gauteng	668 926	33.5%	755 391	33.1%	-0.39%
KwaZulu-Natal	324 216	16.2%	373 662	16.4%	0.14%
Limpopo	138 163	6.9%	164 150	7.2%	0.28%
Mpumalanga	138 732	6.9%	172 587	7.6%	0.62%
Northern Cape	44 159	2.2%	52 681	2.3%	0.10%
North West	129 872	6.5%	148 219	6.5%	-0.01%
Western Cape	290 607	14.5%	327 314	14.3%	-0.21%
Total	1 999 087	100.0%	2 283 822	100.0%	_

# Institutional component

The institutional component recognises that some costs associated with running a provincial government and providing services are not directly related to the size of a province's population. It is therefore distributed equally between provinces. It constitutes 5 per cent of the total equitable share, of which each province receives 11.1 per cent.

# Basic component

The basic component is derived from the proportion of each province's share of the national population. Following the review of the provincial equitable share formula, the weight of this component increases from 14 per cent to 16 per cent to balance the total change in weights for the education and health components. For the 2011 MTEF, population data are drawn from the 2010 mid-year population estimates. Table W1.13 shows the impact of the revised weighted shares of the basic component.

Table W1.13 Impact of the changes in population on the basic component shares

Population (thousand)	2009 Mid- year population estimates	2010 Mid- year population estimates	Population change	% population change	Current (2010 MTEF)	New (2011 MTEF)	Difference in weighted shares
Eastern Cape	6 649	6 744	95	1.4%	13.5%	13.5%	0.01%
Free State	2 902	2 825	-78	-2.7%	5.9%	5.7%	-0.23%
Gauteng	10 757	11 192	435	4.0%	21.8%	22.4%	0.58%
KwaZulu-Natal	10 449	10 646	196	1.9%	21.2%	21.3%	0.11%
Limpopo	5 227	5 440	212	4.1%	10.6%	10.9%	0.28%
Mpumalanga	3 607	3 618	11	0.3%	7.3%	7.2%	-0.08%
Northern Cape	1 148	1 104	-44	-3.8%	2.3%	2.2%	-0.12%
North West	3 224	3 201	-24	-0.7%	6.5%	6.4%	-0.14%
Western Cape	5 357	5 224	-133	-2.5%	10.9%	10.4%	-0.41%
Total	49 321	49 991	671	1.4%	100.0%	100.0%	_

# **Conditional grants to provinces**

There are four types of provincial conditional grants. Schedule 4 sets out general grants that supplement various programmes partly funded by provinces, such as infrastructure and central hospitals. Transfer and spending accountability arrangements differ, as more than one national or provincial department may be responsible for different outputs expected from the grant. Schedule 5 grants fund-specific responsibilities for both the transferring and receiving of provincial accounting officers. A schedule 8 grant, introduced in 2009/10, is intended to provide provinces (and municipalities) with an incentive to meet or exceed prescribed targets. A schedule 9 grant, introduced in 2011/12, provides for the swift allocation and transfer of funds to a province to assist it in dealing with a disaster.

# Changes to conditional grant framework

The 2011 MTEF introduces two new conditional grants: the *school infrastructure backlogs grant* and the *provincial disaster grant*. The *school infrastructure backlogs grant* is a short-term grant introduced to deal with the backlog in inappropriate structures and access to basic services in education. To improve alignment of provincial infrastructure transfers with sector needs, the *infrastructure grant to provinces* has been restructured. The portion of the grant that was earmarked for health is now allocated to the new *health infrastructure grant*. The portion that was earmarked for roads is now put into the *provincial roads maintenance grant*, which will be administered by the Department of Transport. The portion of the grant earmarked for education is now in the new *education infrastructure grant*.

Table W1.14 shows the revisions to provincial conditional grants, which provide for technical, policy and inflation adjustments. Including additions funded from savings and after accounting for shifts from provincial conditional grants to municipal grants, revisions to conditional grant baseline allocations total R2.2 billion, R3.4 billion and R5.9 billion over the MTEF, and bring the new conditional grant baselines to R69.4 billion in 2011/12, R74.7 billion in 2012/13 and R80.6 billion in 2013/14. This excludes the *school infrastructure backlogs grant*, which is an indirect transfer.

Table W1.14 Revisions to provincial conditional grant baseline allocations,

2011/12 – 2013/14 <sup>1</sup>				
R million	2011/12	2012/13	2013/14	2011 MTEF
Technical adjustments	-2 578	-3 232	-3 821	-9 631
Health	_	-620	-1 154	-1 774
Forensic pathology services	_	-620	-654	-1 274
Hospital revitalisation	_	_	-500	-500
National Treasury	-13 091	-14 008	-14 778	-41 877
Infrastructure grant to provinces	-13 091	-14 008	-14 778	-41 877
Revised to:	13 091	14 008	14 778	41 877
Education infrastructure	5 498	5 883	6 207	17 588
Health infrastructure	1 702	1 821	1 921	5 444
Provincial roads maintenance	5 891	6 303	6 650	18 845
Human Settlements	-2 246	-2 339	-2 468	-7 053
Human settlements development	-2 246	-2 339	-2 468	-7 053
Public Works	-332	-273	-199	-804
Expanded public works programme incentive for provinces	-533	-515	-485	-1 532
Social sector expanded public works programme incentive for provinces	200	242	286	728
Savings effected on conditional grants	-71	-45	-54	-170
Human Settlements	-35	_	_	-35
Human settlements development	-35	_	_	-35
Health	-36	-45	-54	-135
Hospital revitalisation	-36	-45	-54	-135
Additions to baselines	2 227	3 433	5 852	11 512
Agriculture, Forestry and Fisheries	50	175	275	500
Comprehensive agricultural support programme	50	120	230	400
Land care programme: Poverty relief and infrastructure development	-	55	45	100
Cooperative Governance and Traditional Affairs	305	180	190	675
Provincial disaster grant	305	180	190	675
Higher Education and Training	354	536	864	1 754
Further education and training colleges	354	536	864	1 754
Health	310	560	2 110	2 980
Comprehensive HIV and Aids	60	60	1 360	1 480
National tertiary services	250	500	750	1 500
Public Works	641	718	804	2 163
Devolution of property rate funds	641	718	804	2 163
Transport	566	1 264	1 609	3 440
Provincial roads maintenance	566	1 264	1 609	3 440
Indirect transfers	700	2 315	5 189	8 204
Basic Education	700	2 315	5 189	8 204
School infrastructure backlogs	700	2 315	5 189	8 204

<sup>1.</sup> Some national shifts and savings are not recorded in provincial conditional grant baselines.

Table W1.15 provides a summary of conditional grants by sector for the 2011 MTEF. More detailed information, including the framework and formula for each grant, is provided in Annexure W2 of the 2011 Division of Revenue Bill. The frameworks provide the conditions for each grant, the outputs expected, the allocation criteria used for dividing each grant between provinces, a summary of the audit outcome in 2009/10 and any other material issues to be addressed.

Table W1.15 Conditional grants to provinces, 2010/11 - 2013/14

R million	2010/11	2011/12	2012/13	2013/14
Agriculture, Forestry and Fisheries	1 167	1 487	1 684	1 867
Agricultural disaster management	50	_	_	_
Comprehensive agricultural support programme	862	1 029	1 148	1 315
Ilima/Letsema projects	200	400	420	443
Land care programme: Poverty relief and infrastructure development	55	58	116	109
Arts and Culture	513	543	571	602
Community I brary services	513	543	571	602
Basic Education	7 107	10 546	11 331	11 954
Dinaledi schools	-	70	100	106
Education infrastructure	3 206	5 498	5 883	6 207
HIV and Aids (life skills education)	188	199	209	221
National school nutrition programme	3 663	4 579	4 928	5 199
Technical secondary schools recapitalisation	50	200	210	222
Cooperative Governance and Traditional Affairs	214	305	180	190
Provincial disaster	_	305	180	190
Provincial infrastructure disaster relief	214	_	_	_
Health	20 483	23 948	25 746	28 175
Comprehensive HIV and Aids	6 052	7 493	8 825	10 607
Forensic pathology services	557	590	_	_
Health infrastructure	840	1 702	1 821	1 921
Health professions training and development	1 865	1 977	2 076	2 190
Hospital revitalisation	3 771	4 136	4 336	4 068
National tertiary services	7 398	8 049	8 689	9 389
Higher Education and Training	3 804	4 326	4 705	5 262
Further education and training colleges	3 804	4 326	4 705	5 262
Human Settlements	13 033	14 942	15 599	16 457
Housing disaster relief	134	_	_	_
Human settlements development	12 899	14 942	15 599	16 457
Public Works	2 181	2 271	2 505	2 778
Devolution of property rate funds	1 865	1 803	1 938	2 091
Expanded public works programme incentive for provinces	259	267	325	402
Social sector expanded public works programme incentive for provinces	57	200	242	286
Sport and Recreation South Africa	426	452	475	501
Mass sport and recreation participation programme	426	452	475	501
Transport	9 013	10 616	11 929	12 860
Gautrain rapid rail link	438	5	_	_
Overload control	11	_	_	-
Provincial roads maintenance	4 700	6 457	7 568	8 259
Public transport operations	3 863	4 153	4 361	4 601
Total	57 941	69 436	74 724	80 647
Indirect transfers	_	700	2 315	5 189
School infrastructure backlogs	_	700	2 315	5 189

# Agriculture grants

The *comprehensive agricultural support programme* aims to provide support for newly established and emerging farmers. Included in this grant is the extension recovery programme, which focuses on improving extension services through training programmes and providing equipment for extension officers. The grant also targets farm infrastructure and provides support for dipping, fencing and

rehabilitation of viable irrigation schemes. An amount of R400 million is added to this grant over the MTEF.

The *land care programme grant: poverty relief and infrastructure development* aims to increase productivity and the sustainable use of natural resources. Provinces are encouraged to use this grant to create jobs through the expanded public works programme. An amount of R282 million is allocated over the medium term.

The *Ilima/Letsema projects grant* is intended to boost food production. The grant is aimed at assisting previously disadvantaged South African farming communities to achieve an increase in agricultural production and receives R1.3 billion over the MTEF.

# Arts and culture grant

The community library services grant is administered by the Department of Arts and Culture. The purpose of the grant is to enable South Africans to gain access to knowledge and information that will improve their socioeconomic situation. The grant is allocated to the relevant provincial department and either administered by that department or through a service-level agreement with municipalities. This grant is allocated R1.7 billion over the MTEF.

# Basic education grants

The Department of Basic Education administers the *national school nutrition programme grant, the Dinaledi schools grant,* the *technical secondary schools recapitalisation grant* and the *HIV and AIDS (life skills education) grant.* The new *school infrastructure backlogs grant* and the *education infrastructure grant* will be administered by the department over the 2011 MTEF.

The national department will administer the school infrastructure backlogs grant – a grant-in-kind for provinces to ensure a national, coordinated and high-impact approach to eradicating backlogs in inappropriate structures and access to basic services at schools. The grant has been allocated R8.2 billion over the MTEF during which all backlogs will be eradicated and the grant will cease to exist at the end of this period.

The new *education infrastructure grant* is the portion of the *infrastructure grant to provinces* that was earmarked for education, and has been allocated R17.6 billion over the MTEF. The best-practice planning principles established through the *infrastructure grant to provinces* will continue to be implemented through this grant.

The Department of Basic Education will coordinate the implementation of these infrastructure grants to ensure provinces manage their entire education asset stock efficiently and effectively. Key to the success of these grants is the institutionalisation of best-practice infrastructure procurement practices that lead to the packaging of many small- and medium-sized infrastructure projects into single projects, which are delivered through a single contracting process.

The *national school nutrition programme* seeks to improve nutrition of poor school children, enhance active learning capacity and improve attendance in schools. An amount of R14.7 billion is allocated to this grant over the MTEF.

*The technical secondary schools recapitalisation grant* provides for equipment and facilities in technical high schools. This grant came into effect in 2010/11 and R632 million is allocated to it over the MTEF.

The *Dinaledi schools grant* starts in 2011/12. The grant will support Dinaledi schools to improve teaching in mathematics and sciences. The grant is allocated R276 million over the MTEF.

The HIV and Aids (life skills education) grant provides for life skills training, sexuality and HIV and Aids education in primary and secondary schools and is fully integrated into the school system, with

learner and teacher support material provided for grades 1 to 9. This grant is allocated R629 million over the MTEF.

# Cooperative governance grants

The new conditional *provincial disaster grant* is introduced in the 2011 MTEF. This grant will be administered by the National Disaster Management Centre in the Department of Cooperative Governance and Traditional Affairs as an unallocated grant to provincial government. A new schedule and clauses have been inserted into the Division of Revenue Act to create special provisions for this grant, enabling the National Disaster Management Centre to disburse disaster response funds immediately after a disaster is declared, without the need for the transfers to first be gazetted. Over the MTEF, R675 million is available for disbursement through this grant.

## Health grants

The *national tertiary services grant* aims to provide strategic funding to enable provinces to plan, modernise, and transform tertiary hospital service delivery, in line with national policy objectives. The grant operates in 26 hospitals across the nine provinces, concentrated in urban Gauteng and the Western Cape. Consequently, the Western Cape and Gauteng receive the largest shares of the grant as they provide the largest proportion of these high-level, sophisticated services for the benefit of the health sector countrywide. The grant has been increased by R1.5 billion over the MTEF to provide for OSD agreements for doctors and implementation of norms and standards in hospitals.

The health portion of the *infrastructure grant to provinces* will be transferred through the new *health infrastructure conditional grant*. The capacity-building programmes that supported the *infrastructure grant to provinces* will continue and the national department will support provinces to implement best-practice planning and project implementation processes through this grant, which has been allocated R5.4 billion over the MTEF.

The *hospital revitalisation grant* plays a key role in the large-scale transformation and modernisation of infrastructure and equipment in hospitals. This grant remains separate to the *health infrastructure grant* to enable the national Department of Health to manage projects funded through this grant closely. Taking into account revisions to the baseline of this grant, R12.5 billion is allocated over the MTEF. During 2011/12, work will begin on merging the *health infrastructure grant* and the *hospital revitalisation grant* into a single consolidated source of infrastructure funding for health.

The *health professions training and development grant* funds the training of health professionals, and the development and recruitment of medical specialists. It enables the shifting of teaching activities from central to regional and district hospitals. This grant is allocated R6.2 billion over the medium term.

The *comprehensive HIV* and *Aids grant* enables the health sector to develop a specific response to HIV and Aids. In addition to prevention programmes, the grant supports specific interventions that include voluntary counselling and testing, prevention of mother-to-child transmission, post-exposure prophylaxis and home-based care. In addition to substantial increases to this grant over the 2010 MTEF and additions to the provincial equitable share over the 2011 MTEF for HIV and Aids programmes, R1.5 billion is added to this grant over the 2011 MTEF to fund the higher-than-expected demand for treatment and prevention programmes. The grant is allocated R26.9 billion over the MTEF.

The *forensic pathology services grant* assists with the transfer of medico-legal mortuaries from the South African Police Service to the health sector and supports the provision of comprehensive forensic pathology services for the criminal justice system. This grant will be phased into the provincial equitable share from 2012/13 and is allocated R590 million in 2011/12.

# Higher education and training grants

The further education and training colleges grant was introduced in 2010/11 to protect spending on these colleges by provinces while the legislative processes required to shift this function to national government are completed. An amount of R1.8 billion is added to this grant over the MTEF to cover the cost of wage agreements, carry-through effects of OSD agreements and to provide for increased enrolment.

# Human settlements grants

The human settlements development grant facilitates the establishment of habitable, stable and sustainable human settlements in which all citizens have access to social and economic amenities. From 2011/12 the portion of this grant that should go to cities for internal infrastructure to houses will be taken out of the grant and added to the urban settlements development grant, formerly the municipal infrastructure cities grant. The baseline of the human settlements development grant has accordingly been revised to R47 billion over the MTEF. As more municipalities with large urban centres are able to take on these responsibilities, they will join the urban settlements development grant and their portion of the human settlements development grant will be transferred. This should accelerate the eradication and formalisation of informal settlements. In cases where municipalities are accredited in terms of the Housing Act (1997), the municipalities will receive the grant directly from national government.

# Public works grants

The devolution of property rate funds grant ensures that provinces take over the responsibility of paying property rates and municipal charges on properties that were administered by national government on their behalf. An amount of R2.2 billion is added to this grant over the MTEF, as more information about property ownerships and municipal rates has become available.

The expanded public works programme incentive grant for provinces provides incentives to provinces to increase spending on labour-intensive programmes. It is awarded to provinces based on the number of work opportunities they create through specific programmes. The baseline of this grant has been revised to provide for the social sector expanded public works programme incentive grant for provinces and is in line with provincial department's capacity to perform. Over the MTEF, R994 million is allocated to this grant. In the 2010 Budget, the Department of Public Works introduced a subsidy grant: the expanded public works programme for the social sector. That grant has been restructured into an incentive grant for the social sector to increase employment in non-profit organisations working in this sector and to improve the reach and quality of their services. The grant has been allocated R728 million over the MTEF.

# Transport grants

The Department of Transport is allocated R5.3 million in 2011/12, through the *gautrain rapid rail link grant*, as a final contribution to the construction of the Gautrain Rapid Rail Link to cover the cost of any foreign exchange losses. This is the last year of this grant.

The *public transport operations grant* subsidises commuter bus services. The payment of bus subsidies to operators was previously funded on an agency arrangement between national and provincial government, and this grant enables government to take greater responsibility in ensuring contractual obligations are met. This grant will amount to R13.1 billion over the MTEF.

The portion of the infrastructure grant to provinces earmarked for provincial roads has been allocated to the provincial roads maintenance grant, which comes into effect in 2011/12. An amount of R3.4 billion is added to the earmarked portions from the infrastructure grant to provinces over the MTEF to enable provinces to expand their maintenance activities and to cover the cost of rehabilitation work created by coal haulage in Mpumalanga and Gauteng. The provincial roads maintenance grant will require

provinces to follow best-planning practices according to road asset management systems and to keep these systems updated regularly. The grant has been allocated R22.3 billion over the MTEF.

# Sports and Recreation grants

The Department of Sports and Recreation administers the *mass sport and recreation participation* programme grant to encourage mass sports participation within communities and schools through selected activities. This grant is allocated R1.4 billion over the MTEF.

# Part 5: Local government fiscal framework and allocations

A number of changes are made to the fiscal framework and allocations to local government this year. Revisions have been made to conditional grants to differentiate between and better respond to the needs of rural and urban municipalities. The formula used to allocate the equitable share to municipalities has also been adjusted to direct more funds towards poorer municipalities. These changes will be built on after a review of the local government fiscal framework to be conducted during 2011.

This section outlines what transfers are made to local government and how these funds are distributed between municipalities. Funds raised by national government are transferred to municipalities through conditional and unconditional grants. These funds help municipalities to meet their constitutional mandate to deliver basic services and meet the public service needs of all their residents, while promoting local economic development. National transfers to municipalities are published to enable them to plan fully for their coming 2011 budgets, and to promote better accountability and transparency by ensuring that all national allocations are included in municipal budgets.

Table W1.16 Revisions to direct and indirect transfers to local government,

2011/12 – 2013/14				
	2011/12	2012/13	2013/14	2011 MTEF Total revisions
R million				revisions
Technical adjustments	1 463	953	1 050	3 465
Direct transfers	1 552	1 453	1 566	4 572
Municipal infrastructure grant	-493	-600	-633	-1 725
Urban settlements development grant	2 739	2 939	3 101	8 778
Neighbourhood development patnership grant	-440	-382	-447	-1 270
Municipal systems improvement grant	-5	-6	-6	-17
Expanded public works programme incentive grant to provinces for the infrastructure sector	-428	-498	-448	-1 374
Water services operating subsidy grant	181	_	_	181
Indirect transfers	-90	-500	-517	-1 107
Rural households infrastructure grant	-119	-271	-274	-663
Water services operating subsidy grant	100	_	_	100
Regional bulk infrastructure grant	-71	-230	-243	-544
Additions to baselines	2 037	2 691	3 843	8 571
Direct transfers	1 937	2 307	3 375	7 619
Equitable share	168	339	678	1 185
Urban settlements development grant	396	662	1 008	2 067
Public transport infrastructure and systems grant	378	875	1 212	2 465
Rural transport services and infrastructure grant	24	26	27	77
Municipal drought relief grant	450	_	_	450
Municipal disaster grant	470	330	350	1 150
Financial management grant	50	75	100	225
Indirect transfers	100	384	468	952
Regional bulk infrastructure grant	100	384	468	952

The 2011 MTEF provides for an additional R1.2 billion for the local government equitable share over the MTEF, which results in a growth for unconditional allocations to municipalities over the period, from R30.6 billion in 2010/11 to R40 billion in 2013/14, at an average annual rate of 9.4 per cent. This growth follows several years of significant increases that saw the local government equitable share grow 971 per cent in the decade between 2001/02 and 2011/12 and more than double the proportion of the total national budget it accounts for – growing by an annual average rate of 30.1 per cent. This growth rate could not be sustained indefinitely and will level off over the MTEF. Changes to the local government equitable share in this budget are focused on adjustments to improve the allocative efficiency of the formula. Including additions funded from savings, an amount of R7.6 billion is added to direct transfers and R952 million is added to indirect transfers over the MTEF.

Table W1.17 Transfers to local government, 2007/08 – 2013/14

	2007/08	2008/09	2009/10	2010/11	2011/12	2012/13	2013/14
		Outcome		Revised	Mediu	m-term esti	mates
R million				estimate			
Direct transfers	38 483	45 487	51 538	61 152	70 171	77 029	82 317
Equitable share	20 676	25 560	23 845	30 559	34 108	37 573	39 960
General fuel levy sharing with metros	-	-	6 800	7 542	8 573	9 040	9 613
Conditional grants	17 807	19 927	20 893	23 052	27 490	30 416	32 743
Infrastructure	16 290	18 562	18 812	20 972	25 596	28 642	30 774
Capacity building and other	1 517	1 365	2 081	2 080	1 894	1 774	1 969
Indirect transfers	1 884	2 307	2 997	3 095	3 992	4 445	4 734
Infrastructure	1 334	1 928	2 754	2 947	3 892	4 445	4 734
Capacity building and other	550	379	243	148	100	_	_
Total	40 367	47 794	54 535	64 247	74 164	81 474	87 051

# The local government equitable share

The primary role of the local government equitable share is to distribute local government's share of nationally raised revenue, supplementing municipal own revenues, to assist municipalities in providing basic services to poor households.

# Equitable share formula

Local government's share of nationally raised revenue is allocated between municipalities using a formula that takes account of the different demographics and service levels in municipalities. The equitable share formula ensures that each municipality receives a share that allows it to meet its basic service obligations, taking account of both the operational costs of providing those services and administrative and governance costs incurred in running a municipality. Allocations are corrected to account for the varying ability of municipalities to raise their own revenues.

# Changes to the formula

Government recognises that the current equitable share formula could be improved on and intends to introduce a new equitable share formula for local government after a thorough review of the local government fiscal framework over the medium term (more details on this review are provided in part six of this annexure). However, it is likely that this new formula will only be introduced after the data from the 2011 Census is made available, and it may only affect allocations in 2014/15. Several adjustments have been made to the current formula to ensure that some of the flaws are addressed over the MTEF. In particular, adjustments to the formula are made to ensure that a greater proportion of funds are allocated to municipalities in the poorest areas of the country to improve service delivery, and changes are made to the basic services, institutional and revenue-raising capacity correction components of the formula.

To ensure stability in municipal allocations, municipalities are guaranteed to receive at least 90 per cent of the indicative allocation for 2011/12 published in the 2010 Division of Revenue Act and 100 per cent of their 2010/11 allocation. The majority of municipalities (and all local municipalities), poor municipalities in particular, will receive more than their guaranteed amounts.

The adjustments to the formula are described in detail in the subsections that follow. Their net effect is to increase the equitable share allocations to the 70 poorest municipalities by an average of 6.7 per cent and to the 21 district municipalities that provide water and sanitation services by an average of 5 per cent. These increases are in addition to the increases in funds added to the local government equitable share, which benefits all municipalities.

The structure and components of the formula are summarised in the box below:

# Structure of the local government equitable share formula

Grant =  $BS + D + I - R \pm C$ 

where

BS is the basic services component

D is the development component

I is the institutional support component

R is the revenue-raising capacity correction and

C is a correction and stabilisation factor

#### The basic services component

The purpose of this component is to assist municipalities in providing basic services to poor households and with meeting municipal health service needs for all. For each subsidised basic service there are two levels of support: a full subsidy for poor households that are connected to municipal services, and a partial subsidy for poor households that are not yet connected to municipal networks. In the past, the allocation for un-serviced households was set at a third of the cost of the subsidy to serviced households. In the 2011 MTEF this has been increased to 45 per cent of the value of subsidy to serviced households. This increase acknowledges that progress has been made in connecting more poor households to services, increasing the service costs to municipalities. It is not possible to adjust the number of serviced households accurately until the next census results are released, so a general increase to the value of allocations against un-serviced households has been made to help cover these additional costs. This has a significant impact on municipalities in the poorest parts of the country, as they tend to have the highest service backlogs.

The characteristics of the basic services component are:

- Supporting poor households earning less than R800 per month in 2001 prices.
- Distinguishing between poor households connected to services and those that are not connected to services and may be provided with alternatives.
- Recognising water reticulation, sanitation, refuse removal and electricity reticulation as core municipal services.
- Providing for municipal health services to all households (through funding allocated to district and metropolitan municipalities).

## The basic services component

BS=[water subsidy 1\*poor with water + water subsidy 2\*poor without water] +

[sanitation subsidy 1\*poor with sanitation + sanitation subsidy 2\*poor without sanitation] +

[refuse subsidy 1\*poor with refuse + refuse subsidy 2\*poor without refuse] +

[electricity subsidy 1\*poor with electricity + electricity subsidy 2\*poor without electricity] +

[municipal health services\*total number of households]

# The development component

This component is currently inactive. The review of the local government fiscal framework which will commence in 2011 will consider how best the formula can respond to the development needs of the different types and categories of municipalities.

## The institutional support component

The average low- or medium-capacity municipality (typically in rural areas or small towns), spends more than half of its own revenue on administrative and governance costs, leaving a reduced portion available for the provision of basic services to residents. Given the existing capacity challenges in these municipalities, the institutional support component of the equitable share formula offers assistance in meeting some of these requirements, providing a supplement to augment, but not fully cover, institutional costs.

The institutional support component has been adjusted in the 2011 formula to take account of the level of poverty in a municipality and its relative ability to fund administrative and governance costs from own revenue. Previously, this component was largely determined by the population size of a municipality. The adjusted formula still reflects the relative sizes of different municipalities, but this is now adjusted by their poverty rate.

# The institutional support component

I = Base allocation + [allocation per councillor \* number of seats] \* [poverty factor]

Where the values used in the formula are:

I = R550 000 + [R54 000\* councillors] \* [% of households in poverty + 17%]

The base allocation is an amount that will go to every municipality. The second term of this formula recognises that administrative costs go up with the size of a municipality and the ability of a municipality to fund these costs from their own revenue is lower the greater the proportion of its residents that are poor. This second term incorporates two elements; an allocation per councillor that reflects the relative size of a municipality (councillor numbers are determined by the population of a municipality) and a poverty factor calculated as the proportion of poor households in a municipality (poor households divided by total households). The municipality with the highest proportion of poor households receives a poverty factor of 100 per cent (the poorest municipality has 83 per cent of its households below the R800-a-month poverty line according to 2001 prices), so 17 per cent is added to the proportion of poor households in each municipality to calculate the poverty factor.

This component (together with the special support for councillor remuneration to poor municipalities provided outside of the equitable share formula) provides sufficient resources for municipalities to pay their councillors' salaries and a significant portion of their administrative costs without having to use the funds allocated through the basic services component.

The number of seats that will be recognised for purposes of the formula is determined by the Minister of Cooperative Governance and Traditional Affairs for elections and composition.

# The revenue-raising capacity correction

To account for the varying fiscal capacities of municipalities, the formula must account for each municipality's ability to raise revenue for the purposes of fulfilling its constitutional mandate. This component therefore takes into account income from property rates and the fuel levy sharing with metropolitan municipalities. In the absence of proper information on property valuation rolls across the spectrum of municipalities and as an interim measure, previous property rate collections between 2004/05 and 2006/07 have been used as a basis for determining future capacity to collect income from this source.

The formula does not look at changes in the levels of revenue collection after 2006/07, to avoid penalising municipalities that have improved their revenue collection efforts. The projected capacity of a municipality to raise revenue from property rates is assumed to be the average of past revenue collection grown to reflect the impact of inflation. In the case of fuel levy sharing with metropolitan municipalities, the revenue-raising capacity correction is calculated using the allocations published for the MTEF.

In order to achieve greater horizontal equity in the allocation system and to acknowledge the revenue-raising constraints faced by smaller municipalities, a differentiated "revenue correction" rate on property rates income is applied. The applicable revenue correction rate for a municipality is based on the level of per capita own operating revenue (based on 2004/05 to 2006/07 figures), and own operating revenue is the difference between past actual total operating revenue and income from grants and subsidies.

The revenue correction rates range from 1 per cent for municipalities with the lowest operating revenue per capita to 7 per cent for those municipalities with the highest operating revenue per capita. The correction rate applied to each municipality's predicted revenue from property rates is calculated using the following formula (with a 7 per cent maximum cut-off for municipalities with operating revenue per capita above R2 500):

# "Revenue correction rate" = 1 + 6/2500 \* [Operating revenue per capita]

The application of revenue-raising capacity correction in the local government equitable share formula means that municipalities are expected to use between 1 per cent and 7 per cent of the revenue they raise from property rates to top-up the funds provided through the equitable share.

District municipalities do not collect property rates, so the revenue-raising capacity correction component of the formula is applied as a flat "tax" of 6 per cent of the value of the *regional services council RSC / Joint Services Board (JSB) levy replacement grant*, allocated to each district. This grant is an unconditional allocation that replaces the major source of own revenue for district municipalities prior to 2006.

There have been two changes in this component of the formula for the 2011 MTEF. The previous stepped taxation structure for property rates (in which municipalities were placed into eight bands with one revenue correction rate applying to all municipalities in each band) has been replaced with the smoothed curve structure described above. This is fairer to municipalities that were on the outer edges of the bands in the previous formula, as they will now have their own revenue correction rate. In addition, the rate of revenue correction has been reduced for all municipalities. Previously, this rate ranged from 1.5 per cent to 9.5 per cent, now it ranges from 1 per cent to 7 per cent.

These changes reduce the impact of this component on the final allocations to municipalities by 12 per cent. The revised component takes account of the substantial migration to more prosperous municipalities since 2001. These municipalities are funding the provision of services to larger numbers of poor residents through cross-subsidisation from their own revenue. Reducing the revenue correction rate in this component will free up more funds within these municipalities for cross-subsidisation.

# Stabilising constraint

With the publication of three-year budget allocations, a guarantee mechanism is applied to the indicative outer-year baseline amounts with the aim of ensuring that municipalities are given what was indicated in the previous MTEF, as far as this is possible, given overall budget constraints and the need to amend the formula to increase allocations to poorer municipalities. In the 2010 MTEF, the applicable guarantees on the allocations are 100 per cent for 2010/11 and 90 per cent for 2011/12. In the schedules of the 2011 Division of Revenue Act, the applicable guarantees are 100 per cent for the 2011/12 allocations, 90 per cent for the 2012/13 allocations, with no guarantee on the indicative 2013/14 allocations published.

To deal with these constraints, municipalities are divided into two groups: municipalities that require a "top-up" in order to meet the stabilising constraints and those that do not. The total size of the top-up is calculated and deducted from those that do not require a top-up amount in proportion to the "surplus."

All district management areas (DMAs) have been incorporated into local municipalities as part of the redemarcation of municipal boundaries that comes into effect with the 2011 local government elections. Previously, district municipalities received the equitable share funds allocated on the basis of households in the DMAs. As these district municipalities are no longer responsible for providing services in the DMAs, the guarantees on their equitable share allocations (described above) were applied after subtracting the amounts previously allocated to them for the DMAs.

# Other considerations in applying the formula

The formula outlined above has to be rescaled to make allowance for intricacies in the allocation process. In particular, powers and functions must be taken into account, and the overall budget must balance.

#### Powers and functions

Local government is divided into categories A, B and C.<sup>1</sup> The division of powers and functions between local and district municipalities differs – and this is also true between the different local municipalities within the same district. In order to deal with these differences, the formula has to ensure that the allocations made in terms of the basic services component go to the municipality that is authorised to perform that function. To enhance transparency in the budget process, local government equitable share and *municipal infrastructure grant (MIG)* allocations to district municipalities are published per unauthorised local municipality in the relevant district municipality.

# Balancing allocations

The horizontal division of allocations made between municipalities depends on the size of the overall allocation made to the local government sphere, usually decided through a separate consultative process to determine the equitable share of nationally raised revenue for each of the three spheres of government (the vertical division). As there is no guarantee that allocations made in terms of the horizontal division add up precisely to the amount allocated to the local government equitable share, such allocations need to be adjusted to fit within the constraints outlined above.

<sup>1</sup> Category A are metropolitan municipalities, category B are local municipalities and category C are district municipalities.

## Rescaling of the BS, D and I components

The simplest way of making the system balance is to rescale the BS, D and I components to the available budget, and the formula actually becomes:

# $Grant = adjustment factor*(BS + D + I) - R \pm C$

This adjustment factor is calculated to ensure that the system balances.

#### Measurement issues

The integrity of the data is as important as the set of equations in determining whether the allocations meet the constitutional requirement of equity. Although extensive work has been undertaken to try update the data used in the formula, Census 2001 remains the only source of data that is reliable down to municipal level for population, income and service access data. Data for the number of councillors per municipality is provided by the Independent Electoral Commission and the Municipal Demarcation Board, and data on property rates collected between 2004/05 and 2006/07 is sourced from the reports that municipalities submit to National Treasury in terms of section 71 of the MFMA.

# a) Poverty

Household income is used to estimate poverty at a municipal level, as it allows for a cross-tabulation of poverty against servicing levels. The majority (over 90 per cent) of funds allocated through the formula are based on the service delivery needs of poor households.

# b) Changes to municipal boundaries

The Municipal Demarcation Board announced in September 2010 that a series of municipal boundary changes would come into effect with the local government elections in 2011. Buffalo City and Mangaung will shift from local municipalities to metropolitan municipalities, and Metsweding district municipality and its local municipalities will be incorporated into the Tshwane metropolitan municipality. Several local municipalities will shift into different district municipalities, some municipalities will merge, and certain wards will move from one municipality to another. All district management areas will be eliminated. These sparsely populated areas were previously serviced by district municipalities, but will now be incorporated into local municipalities. To reflect these changes in the allocations for the 2011 MTEF, the 2001 Census data used to calculate the equitable share has been updated by StatsSA.

# c) Servicing levels

The basic services subsidy for poor households is a key determinant in the current formula. There is no accurate data on these service costs across all municipalities, and so these allocations are based on estimates on the relative costs of services and the amount of funds available. As outlined in the basic services section, it is now assumed that providing alternative services to households that did not have services when Census 2001 was conducted is 45 per cent of the cost of providing full services. After the adjustment factor and other components are applied, the actual subsidies per basic service made available through the equitable share are set out in table W1.19.

Table W1.18 Number of poor households

	Serviced	Unserviced
Service	households	households
Electricity	3 079 340	2 456 443
Water	3 322 295	2 213 488
Sanitation	3 260 814	3 274 969
Refuse	2 176 923	3 358 860

Source: 2001 Census

Table W1.19 Actual average monthly basic services subsidies per poor household

		•		•	•	
Monthly	Serviced households			Households not connected to services		
Rand	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14
Electricity	188.04	208.31	222.05	84.62	93.76	99.92
Water	125.36	138.84	148.03	56.41	62.53	66.61
Sanitation	125.36	138.82	148.03	56.41	62.52	66.61
Refuse	125.36	138.82	148.03	56.41	62.52	66.61
Total	564.12	624.79	666.15	253.85	281.32	299.77

#### Other unconditional allocations

# RSC/JSB levies replacement grant

Prior to 2006, district municipalities raised levies on local businesses within their districts through either an RSC levy or a JSB levy. This source of revenue was replaced in 2006/07 with the RSC/JSB levies replacement grant, which was allocated to all district and metropolitan municipalities, based on the amounts they had previously collected through the levies. In the 2011 MTEF, the RSC/JSB levies replacement grant is grown by 9 per cent a year for municipalities authorised for water and sanitation functions and 3 per cent for unauthorised municipalities, acknowledging the very different service responsibilities of these district municipalities.

The redemarcation that will come into effect with the 2011 local government elections will see two district municipalities disestablished and the boundaries of a further five district municipalities changed substantially. Alfred Nzo, Xhariep and Thabo Mofutsanayana district municipalities will expand to include additional local municipalities. The RSC levy replacement grant to these district municipalities will be increased. The value of this increase will be calculated as a portion of the RSC levy replacement grant of the district municipality that each local municipality used to form part of, in proportion to their share of that district's population. OR Tambo and Amatole districts will both reduce in size but their RSC levy replacement grant will not be changed. Motheo and Metsweding districts will both be disestablished; their RSC levy replacement grant will be returned to the local government equitable share and allocated to all municipalities through the formula. Both new metropolitan municipalities (Mangaung and Buffalo City) will receive funds from the fuel levy sharing with metropolitan municipalities and Tshwane metropolitan municipality's allocation will be increased to account for its incorporation of Metsweding district (details in part six of this annexure).

# Special support for councillor remuneration

Councillors' salaries are subsidised in poor municipalities. This support is calculated separately to the local government equitable share and is additional to the governance costs allocation provided in the institutional support component. The level of support provided to each municipality is published in the government gazette issued by the Minister of Cooperative Governance and Traditional Affairs, determining the upper limits of salaries, allowances and benefits of members of municipal councils. The gazette classifies municipal councils into six grades based on their total income and population size. Special support is provided to the lowest three grades of municipal councils (the smallest and poorest municipalities).

# Conditional grants to local government

Conditional grants to local government aim to eradicate backlogs and build institutional financial capacity in local government. The total value of conditional grants directly transferred to local government, including the water operating subsidy, increase from R27.5 billion in 2011/12, to R30.4 billion in 2012/13 and R32.7 billion in 2013/14.

Conditional grant allocations to local government are being reconfigured to increase differentiation between the funding of urban and rural municipalities. The most significant change to be introduced in 2011 is the creation of a new *urban settlements development grant* for metropolitan municipalities to fund the upgrading of informal settlements. This means that metropolitan municipalities no longer receive allocations through the *MIG*. Several other grants also provide for specific rural and urban challenges, including the *rural transport infrastructure grant* and *rural households infrastructure grant*, which fund rural municipalities while the *public transport infrastructure and systems grant* funds public transport projects in large cities. A new *municipal disaster grant* has also been introduced to enable the speedy allocation and transfer of funds to municipalities affected by disasters.

# Infrastructure conditional grants to local government

National transfers for infrastructure, including indirect or in-kind allocations to entities executing specific projects in municipalities, amount to R29.5 billion, R33.1 billion and R35.5 billion for each of the 2011 MTEF years.

Table W1.20 Infrastructure transfers to local government, 2007/08 – 2013/14

	2007/08	2008/09	2009/10	2010/11	2011/12	2012/13	2013/14
		Outcome		Revised	Mediu	m-term est	imates
R million				estimate			
Direct transfers	16 290	18 562	18 812	20 972	25 596	28 642	30 774
Municipal infrastructure grant	6 967	6 968	8 788	9 515	11 444	13 914	14 679
Urban settlements development grant	2 950	3 590	4 285	4 855	6 267	7 410	8 127
National electrification programme	462	589	914	1 020	1 097	1 151	1 215
Public transport infrastructure and systems grant	1 174	2 920	2 418	3 699	4 803	5 000	5 564
Neighbourhood development partnership grant	41	182	506	1 030	750	800	800
2010 FIFA World Cup stadiums development grant	4 605	4 295	1 661	302	_	_	_
Rural transport services and infrastructure grant	_	9	10	10	35	37	39
Electricity demand side management	_	_	175	220	280	_	_
Municipal disaster grant	-	-	-	_	470	330	350
Municipal drought relief grant	90	9	54	320	450	_	_
Indirect transfers	1 334	1 928	2 754	2 947	3 892	4 445	4 734
National electrification programme	973	1 148	1 478	1 720	1 738	1 882	1 986
Neighbourhood development partnership grant	61	54	90	125	100	80	55
Regional bulk infrastructure grant	300	450	612	893	1 704	2 003	2 176
Backlogs in water and sanitation at clinics and schools	-	186	350	_	-	-	-
Backlogs in the electrification of clinics and schools	_	90	149	_	_	_	_
Electricity demand-side management	_	_	75	109	119	_	_
Rural households infrastructure grant	-	_	-	100	232	480	517
Total	17 624	20 490	21 566	23 919	29 488	33 087	35 508

# Municipal infrastructure grant

The largest infrastructure transfers are through the *MIG*, which supports government's objective of expanding service delivery and alleviating poverty. The *MIG* funds the provision of infrastructure to provide basic services, roads and social infrastructure for poor households in all non-metropolitan municipalities. *MIG* previously also included the *MIG* cities grant – a schedule 4 allocation to

metropolitan municipalities, but from 2011/12 MIG cities forms part of the baseline of the new urban settlements development grant.

The MIG allocations are based on a formula with a vertical and horizontal division. The vertical division allocates resources to sectors or other priority areas, and the horizontal division is based on a formula that takes account of poverty, backlogs, and municipal powers and functions. There are five main components of the formula, as demonstrated in the box below. A constant component of R5 million ensures that a reasonable minimum allocation is made to poor municipalities.

$$MIG_{(F)} = C + B + P + E + N + M$$

# C Constant to ensure increased minimum allocation for poor municipalities (This allocation is made to all municipalities)

B Basic residential infrastructure (new and rehabilitation of existing ones)

Proportional allocations for water supply and sanitation, electricity, roads and "other" (street lighting and solid waste removal)

- P Public municipal service infrastructure (new and rehabilitation of existing ones)
  - E Allocation for social institutions and micro-enterprises infrastructure
    - N Allocation to all nodal municipalities
  - M Negative or positive allocation related to past performance of each municipality relative to grant conditions

The total *MIG* allocations grow to R11.4 billion, R13.9 billion and R14.7 billion over the 2011 MTEF. This represents real growth of 28 per cent during the period. Amounts of R493 million, R600 million and R633 million have been removed from the *MIG* baseline and added to the previous *MIG cities* baseline to create the *urban settlements development grant*. Table W1.21 shows the weighted share per sector and the respective amounts that flow through the vertical division of the *MIG* funds.

Table W1.21 Municipal infrastructure grant allocations per sector, 2010/11 – 2013/14

	_	-	· ·	
	2010/11	2011/12	2012/13	2013/14
Weights		Α	djusted weights	
Municipal infrastructure grant (a)				
Special municipal infrastructure fund and mar	nagement (b)			
Ring-fenced allocation: Eradication of the bucket sanitation system (c)				
Bulk infrastructure (d)				
Municipal infrastructure grant (formula)	(a)-(b)	(a)-(b)-(c)-(d)	(a)-(b)-(c)-(d)	(a)-(b)-(c)-(d)
B Component	75.0%	75.0%	75.0%	75.0%
Water and sanitation	72.0%	72.0%	72.0%	72.0%
Electricity	0.0%	0.0%	0.0%	0.0%
Roads	23.0%	23.0%	23.0%	23.0%
Other	5.0%	5.0%	5.0%	5.0%
P Component	15.0%	15.0%	15.0%	15.0%
E Component	5.0%	5.0%	5.0%	5.0%
N Component	5.0%	5.0%	5.0%	5.0%

# Urban settlements development grant

This is a new grant introduced in the 2011 division of revenue. It is allocated to metropolitan municipalities to supplement their capital budgets, enabling them to better leverage their resources to

develop sustainable human settlements. The grant funds the provision of basic municipal services to new housing projects and will allow municipalities to plan and budget for both services and the construction of housing as they attain authorisation for the human settlements function. The grant is created with funds previously allocated to these cities through the *MIG cities grant* and the internal infrastructure portion of the provincial *human settlements development grant*, as well as additional allocations of R2.1 billion over the MTEF. The total *urban settlements development grant* is allocated R6.2 billion in 2011/12, R7.4 billion in 2012/13 and R8.1 billion in 2013/14.

# The public transport infrastructure and systems grant

This grant is administered by the Department of Transport, and aims to provide passenger transport networks in cities, particularly public transport and non-motorised transport infrastructure. This includes the provision of bus rapid transit systems. The grant has an allocation of R4.8 billion in 2011/12, to R5 billion in 2012/13 and R5.6 billion in 2013/14.

# Rural transport services and infrastructure grant

This grant aims to improve rural transport infrastructure, and will fund the collection of accurate data on the condition of rural roads in 2011/12, in line with the Road Infrastructure Strategic Framework for South Africa. The grant will support rural district municipalities in establishing databases on the condition and usage of all the municipal roads in their area, so that the spending of infrastructure funds (from the *MIG* and elsewhere) can be properly planned. The grant has an allocation of R35 million in 2011/12, R37 million in 2012/13 and R39 million in 2013/14.

# Neighbourhood development partnership grant

The grant supports the development of community infrastructure and aims to attract private-sector investment that improves the quality of life in townships. The grant is administered by National Treasury and is allocated R850 million in 2011/12, to R880 million in 2012/13 and R855 million in 2013/14 for both the technical assistance (indirect) and capital (direct) grant.

# Integrated national electrification programme

Government plans to spend R9.1 billion over the next three years on its national electrification programme, to sustain the progress made in supplying electricity to poor households. Of this amount, R3.5 billion will be spent by municipalities directly and R5.6 billion by Eskom on behalf of municipalities. This programme was instrumental in the connection of 80 per cent of all households in the country to the national electricity grid, as reported in the 2007 Community Survey.

# Electricity demand-side management grant

The grant aims to improve energy efficiency demand-side management in residential dwellings and commercial buildings, reducing energy consumption. The grant has been allocated R399 million in 2011/12, and it is scheduled to end after the 2011/12 financial year. A review of the grant's performance will be conducted during 2011 and will inform any decision on whether to extend the life of the grant.

## Regional bulk infrastructure grant

This grant aims to provide regional bulk water and sanitation across several municipal boundaries. In the case of sanitation, it supplements regional bulk collection and regional wastewater treatment works. The grant has an allocation of R1.7 billion in 2011/12, to R2.0 billion in 2012/13 and R2.2 billion in 2013/14.

# Municipal disaster grant

A new conditional grant for disasters is introduced in the 2011 MTEF. This grant will be allocated to the National Disaster Management Centre in the Department of Cooperative Governance and Traditional Affairs as an unallocated grant to local government. A new schedule and clauses have been inserted into the Division of Revenue Act to create special provisions for this grant that enable the immediate disbursement of disaster response funds after a disaster is declared, without the need to gazette the transfers. Over the MTEF, R1.2 billion is available for disbursement through this grant.

# Drought relief grant

R450 million is allocated in the 2011/12 financial year to provide assistance to the Nelson Mandela Metropolitan Municipality for drought relief.

# Capacity-building and other current transfers

The *capacity-building grants* aim to assist municipalities in building management, planning, technical, budgeting and financial management skills. The current MTEF expands the capacity-support programme to assist weaker or poorer municipalities, particularly with the implementation of financial management reforms. Total allocations for *capacity-building grants* amount to R2 billion in 2011/12, R1.8 billion in 2012/13 and R2 billion in 2013/14.

Table W1.22 Capacity building and other current transfers to local government, 2007/08 – 2013/14

2007/08 - 2013/14							
	2007/08	2008/09	2009/10	2010/11	2011/12	2012/13	2013/14
				Revised			
R million				estimate			
Direct transfers	1 517	1 365	2 081	2 080	1 894	1 774	1 969
Municipal systems improvement grant	200	200	200	212	219	230	243
Restructuring grant	530	-	-	_	_	_	_
Financial management grant	145	180	300	365	435	479	526
2010 FIFA World Cup host city operating grant	-	-	508	210	_	-	-
Water services operating subsidy grant	642	985	871	670	561	399	421
Expanded public works programme municipal incentive grant	_	_	202	623	680	666	779
Indirect transfers	550	379	243	148	100	_	_
Financial management grant: DBSA	53	50	-	_	_	_	-
Water services operating subsidy grant	497	329	243	148	100	-	-
Total	2 067	1 744	2 324	2 228	1 994	1 774	1 969

The *financial management grant* under the National Treasury vote, funds the modernisation of financial management, including building in-house municipal capacity to implement multi-year budgeting, linking integrated development plans to budgets, producing quality and timely in-year and annual reports, and generally supporting municipalities in the implementation of the MFMA. Total allocations for the *financial management grant* amount to R1.4 billion over the three year cycle.

# Expanded public works programme incentive grant for municipalities

This grant encourages municipalities to hire more people in public works projects. The grant is allocated R680 million in 2011/12, to R666 million in 2012/13 and R779 million in 2013/14.

# Water services operating subsidy

The water services operating subsidy is a grant with schedule 6 (direct) and schedule 7 (in-kind) components used to fund water schemes. The grant covers staff-related costs and direct operating and maintenance costs, while provision is also made for the refurbishment of infrastructure. The allocation per municipality is based on the operational budget for each scheme and the funding requirements identified and agreed on in the transfer agreement. These schemes were administered by the Department of Water Affairs prior to 1994 and are now being transferred to municipalities. At the end of 2009/10, 59 agreements had been signed, 4 903 staff transferred and 1 643 schemes (including rudimentary schemes) with a total asset value of about R6.4 billion transferred to municipalities. In the 2011 MTEF, R1.5 billion is allocated for the water services operating grant (direct and indirect transfers), to enable the transfer of staff to water schemes. It is a transitional grant and is expected to be phased out over time.

# Part 6: Future work on provincial and municipal fiscal frameworks

# Refinement of the provincial fiscal framework

The new formula for the health component of the provincial equitable share formula is one of the first steps towards the introduction of national health insurance. The health formula will be improved as more information on the provincial consumption of health services becomes available. These improvements will complement the implementation of the national health insurance system.

# Review of the local government fiscal framework

Local government is South Africa's youngest sphere of democratic government, the system of wall-to-wall, democratically elected local municipalities is only just over a decade old. This is an appropriate time to evaluate the fiscal framework's performance in supporting local government. Government has already achieved significant success in creating a system of intergovernmental transfers that is stable, predictable and transparent. Building on this foundation in the second decade of democratic local government, the fiscal framework can do more to promote the efficient and equitable delivery of services. Towards this end, an extensive review process will be undertaken in 2011 that may lead to significant changes in the future shape of the local government fiscal framework.

Although municipalities have made significant strides in building their institutions and delivering services, they have failed to achieve their full potential, and finances are a contributing factor. Government will review the fiscal system and identify reforms to create the right incentives for more effective local government in the future.

For reforms to be successful, a differentiated approach to local government needs to be adopted and conditional grants (separate grants for rural and urban municipalities) are likely to play a major role in future. National Treasury will conduct a full review of the equitable share formula during 2011/12, with the aim of introducing a new formula in time for the release of the Census 2011 data. The formula review will include an examination of municipal services and their costs, different municipal functions, and how the data used in the formula could be updated between censuses. The changes that follow the review processes in 2011 have the potential to substantially alter the nature and effectiveness of the local government fiscal framework.

# **Municipal Property Rates Act**

The Municipal Property Rates Act regulates the power of municipalities to impose rates on properties in accordance with section 229(1)(a) of the Constitution. Income derived from municipal property rates is an important own revenue source.

The original four-year transitional period given to municipalities to implement the Municipal Property Rates Act (up to 1 July 2009) was extended by two years (up to 1 July 2011) through a legislative amendment to the act in 2009 to allow those municipalities that had failed to implement new valuation rolls to continue to use existing valuation rolls and supplementary valuation rolls until 30 June 2011. There are eight municipalities that are expected to implement valuation rolls in terms of the act for the first time on 1 July 2011.

The Department of Cooperative Governance and Traditional Affairs intends to introduce further amendments to the act in 2011/12 to improve its implementation and minimise legal ambiguities. In addition, a number of the proposed amendments make provision that property rating by municipalities is undertaken in the national interest.

# **Municipal Fiscal Powers and Functions Act**

The Municipal Fiscal Powers and Functions Act (MFPFA) of 2007 provides for the authorisation of taxes, levies and duties that municipalities may impose under section 229 of the Constitution. The MFPFA does not deal with property rates or municipal tariff charges and fees.

# Authorisation of taxes that existed prior to the act in terms of section 12

In terms of section 12 of the act, a municipality had to apply to the Minister of Finance by 9 September 2009 for the authorisation of an existing tax. All municipalities submitted applications to the Minister of Finance by the deadline stipulated. These applications were analysed and municipalities, SALGA, FFC and appropriate national departments were advised of the preliminary rulings by the Minister of Finance. Out of 155 applications received from 55 municipalities, only 19 in 17 municipalities potentially complied with the criteria of a municipal tax. The affected municipalities gave extensive feedback, and this will result in some adjustments to the preliminary determinations. To conclude this process, draft regulations need to be published for public comment and submitted to Parliament. Based on feedback received from these processes, the Minister of Finance will gazette the final determinations.

# Application for a new municipal tax in terms of section 5 of the act

Section 5 of the MFPFA requires that a municipality, group of municipalities or organised local government apply to the Minister of Finance for the authorisation of a municipal tax, levy or duty, other than property rates, before imposing such a tax. An application for a municipal tax must set out the reasons for the proposed tax and how the revenue from the tax will be used. An application for a new municipal tax is more likely to succeed if it is proven that the municipality's own revenue is insufficient to meet service delivery needs.

To date, National Treasury has received one application that complies with the requirements stipulated in section 5 of the MFPFA. The application is for the introduction of a rural-based development levy in areas where the municipality faces difficulties in implementing valuation rolls. Similar applications were identified during the verification process of taxes that existed prior to the act. Authorisation of this tax needs to be informed by the criteria of good municipal tax, and whether it can run alongside the property rates system and be applicable to other municipalities.

# Regulation of surcharges in terms of section 8 of the act

In terms of section 8 of the MFPFA, the Minister of Finance may prescribe compulsory national norms and standards for imposing municipal surcharges, including maximum municipal surcharges that may be imposed by municipalities. Surcharges currently form part of a tariff (when regulations are introduced in terms of section 8 of the act, these will be split). Due to the interrelationship between tariffs and surcharges, it is important that National Treasury's regulatory processes regarding surcharges be aligned

to the regulatory processes of sector departments regarding municipal tariffs. The National Electricity Regulator of South Africa is currently in the process of introducing an economic regulation framework in metropolitan municipalities, which will be subsequently rolled out in other smaller municipalities. The regulation of municipal tariff practices regarding water and waste are moving at a much slower pace, therefore the introduction of norms and standards for municipal surcharges will only be over the medium to long term.

# Sharing of the general fuel levy

The sharing of the general fuel levy with metropolitan municipalities was introduced in the 2009 Budget as the primary replacement to the former RSC and JSB levies, in addition to the VAT reforms introduced in 2006. The sharing of the general fuel levy is a direct charge and is formalised annually through a government gazette under schedule 1 of the 2009 Taxation Laws Amendment Act.

To facilitate the smooth transition from the RSC levy replacement grant system to the sharing of the general fuel levy system, implementation has been phased-in over three years. In 2011/12, metropolitan municipalities receive 25 per cent of the former RSC levy replacement grant and 75 per cent of the sharing of the general fuel levy. Allocations in 2012/13 will be based on fuel sales only. The 2011/12 allocations will include two additional metropolitan municipalities that will be introduced after the 2011 local government elections. The fuel levy data has also been updated for those existing metropolitan municipalities whose boundaries change after the 2011 local government elections.

# **ANNEXURE W2:**

# FRAMEWORKS FOR CONDITIONAL GRANTS TO PROVINCES

# Annexure W2: Frameworks for Conditional Grants to Provinces

# Detailed frameworks on Schedules 4, 5, 7, 8 and 9 grants to provinces

# Introduction

This annexure provides a brief description of the framework for the grants set out in Schedules 4, 5, 7, 8 and 9 of the 2011 Division of Revenue Bill. The following are key areas considered for each grant:

- Strategic goal and purpose of the grant
- Outcome statements and outputs of the grant
- Priority outcome(s) of government that the grant primarily contributes to
- Conditions of the grant (additional to what is required in the Bill)
- Criteria for allocation between provinces
- Rationale for funding through a conditional grant
- Past performance
- The projected life of the grant
- 2011 MTEF allocations
- The payment schedule
- Responsibilities of transferring national department and receiving provincial departments
- Process for approval of business plans for 2012/13

The attached frameworks are not part of the Division of Revenue Bill, but are published in order to provide more information on each grant to parliament, legislatures, municipal councils, officials in all three spheres of government and the public. Once the 2011 Division of Revenue Bill is enacted, these frameworks will be gazetted in terms of the Act.

The financial statements and annual reports for 2011/12 will report against the Division of Revenue Act, Division of Revenue Amendment Act and their schedules, and the grant frameworks as gazetted in terms of the Act. Such reports must cover both financial and non-financial performance, focusing on the outputs achieved.

# AGRICULTURE, FORESTRY AND FISHERIES GRANTS

	Comprehensive Agriculture Support Programme Grant
Transferring department	Agriculture, Forestry and Fisheries (Vote 26)
Strategic goal	To create a favourable and supportive agricultural service environment for the farming community, including subsistence, smallholder and commercial farmers
Grant purpose	To expand the provision of agricultural support services, promote and facilitate agricultural development by targeting smallholder and previously disadvantaged farmers
Outcome statements	Improved quality of comprehensive post settlement support services
	Broadened access to agricultural support for subsistence, smallholder and previously
	disadvantaged farmers
	• Improved knowledge and information on farming within smallholder and previously
	disadvantaged farmers
	Improved efficacy of extension and advisory services for farmers
	Increased and sustainable employment within the agriculture sector
	Established previously disadvantaged entrepreneurs within the agricultural sector
	Improved farming efficiency
	Improved household food production
Outputs	• Number of subsistence, smallholder and commercial farmers supported through the
	Comprehensive Agricultural Support Programme (CASP)
	Number of youth and women farmers supported through CASP  Number of CASP bareficiency trained on farming motheds.
	<ul> <li>Number of CASP beneficiaries trained on farming methods</li> <li>Number of CASP beneficiaries selling produce to markets</li> </ul>
	Number of permanent and seasonal jobs created
	<ul> <li>Number of extension personnel recruited and maintained in the system</li> </ul>
	Number of extension officers upgrading qualifications in various institutions
Priority outcome(s) of	Outcome 7: Vibrant, equitable and sustainable rural communities with food security for all
government that this grant	Outcome 4: Decent employment through inclusive economic growth
primarily contributes to	
Details contained in the	Outcome indicators
business plan	Output indicators
	• Inputs
	Key activities
~ ***	Monitoring and evaluation plan
Conditions	• The grant must be allocated to projects in terms of the approved provincial CASP business
	plans
	<ul> <li>Provincial department must confirm human resources capacity to implement the provincial CASP business plan by 1 April 2011</li> </ul>
	<ul> <li>Provinces must inform the transferring national officer of any changes from plans and</li> </ul>
	allocations approved by the transferring national officer within 30 days of such change
	• The business plans must be signed off by head of the provincial department responsible for
	agriculture in collaboration with the head of the Provincial Treasury
Allocation criteria	• The formula used to allocate funds is a weighted average of the following variables:
	competitive CASP performance
	land area (ha)
	restituted land delivered
	redistributed land delivered
D	benchmarks on production
Reasons not incorporated in equitable share	<ul> <li>A conditional grant enables the national department to coordinate and align agriculture support programmes with national priorities</li> </ul>
Past performance	2009/10 audited financial outcomes
r ast performance	Allocated and transferred R715 million to provinces
	• Of the total available R739 million (including provincial roll overs), R687 million
	(93 per cent) was spent
	2009/10 service delivery performance
	Extension recovery programme:
	217 extension officers recruited nationally
	646 extension officers registered for qualification upgrading
	2 346 extension officers trained in skills programmes
	2 244 extension officers supported with ICT equipment
	A total of 26 266 beneficiaries were supported and 706 projects were completed

	Comprehensive Agriculture Support Programme Grant
Projected life	The grant will be reviewed during 2011/12
MTEF allocations	• 2011/12: R1 029 million, 2012/13: R1 148 million and 2013/14: R1 315 million
Payment schedule	• 20 per cent: 21 April 2011, 35 per cent: 22 July 2011, 35 per cent: 7 October 2011 and
	10 per cent: 11 January 2012
Responsibilities of the	Responsibilities of the national department
transferring national officer and receiving officer	<ul> <li>Agree on outputs and targets with provincial departments in line with grant objectives</li> <li>Review and revise norms and standards for the implementation of the grant during the Intergovernmental Technical Committee on Agriculture and Land Affairs – Quarterly Review Meetings (ITCAL-QRM) workshops held quarterly</li> <li>Provide the guidelines and criteria for the development and approval of business plans</li> <li>Monitor implementation through project site visits and provide support on monthly and quarterly basis</li> <li>Submit quarterly performance reports to National Treasury within 45 days after the end of each quarter</li> <li>Convene the National Assessment Panel</li> </ul>
	Responsibilities of the provincial departments
	<ul> <li>Participate in the National Assessment Panel</li> <li>Provinces to implement the CASP pillars according to the approved business plans otherwise notify the national department in writing about deviations before implementation can take place</li> <li>Monitor and evaluate the impact of the implementation of CASP in achieving the programme goals</li> </ul>
Process for approval of the 2012/13 business plans	<ul> <li>Provide the provincial departments with business plans format guidelines, criteria and outputs as prescribed by National Treasury by 31 May 2011</li> <li>Submission of provincial business plans to the national department by 30 September 2011</li> <li>Engagement with provinces on submitted business plans during October 2011 prior to National Assessment Panel</li> </ul>
	• Evaluation and recommendation of business plans by NAP between November 2011 and January 2012
	<ul> <li>Funding agreement submitted to provinces by end of February 2012</li> <li>Agreement signed by HOD of provincial departments of agriculture, head of provincial treasuries and CASP coordinators and submitted to the national department by 23 March 2012</li> <li>Approval of Business plans by Director-General of the national department before 30 March 2012</li> </ul>
	<ul> <li>Inform provinces of approval of business plans by 2 April 2012</li> <li>Approval by the Director-General regarding DORA 2012 business planning process compliance during March 2012 and sent to National Treasury by 16 April 2012</li> </ul>

	Ilima/Letsema Projects Grant
Transferring department	Agriculture, Forestry and Fisheries (Vote 26)
Strategic goal	To reduce poverty through increased food production initiatives
Grant purpose	To assist targeted vulnerable South African farming communities to increase agricultural production and improve farming skills
Outcome statements	Increased agricultural production efficiency for targeted groups
	Increased agricultural production for the targeted group      Improved food and dustion at both bounded and national level.
	<ul> <li>Improved food production at both household and national level</li> <li>Improved ability of targeted households to cope with high food prices</li> </ul>
Outputs	Number of hectares (ha) supported by the programme
Outputs	Number of farmers trained
	Number of tons produced within agricultural development corridors
	Number of beneficiaries supported
	Number of newly established irrigate infrastructure plants
	Number of hectares (ha) of rehabilitated and expanded irrigation schemes
Priority outcome(s) of	Outcome 7: Vibrant, equitable and sustainable rural communities with food security for all
government that this grant primarily contributes to	Outcome 4: Decent employment through inclusive economic growth
<b>Details contained in the</b>	Outcome indicators
business plan	Outputs indicators
	• Inputs
	Key activities
	Monitoring and evaluation     Dislocard with a time.
	<ul><li>Risks and mitigations</li><li>Exit strategies</li></ul>
Conditions	The Ilima/Letsema grant must be allocated in terms of the approved provincial Land and
Conditions	Agrarian Reform Programme (LARP) business plans and must support Comprehensive
	Rural Development Programme (CRDP) sites in provinces
	• Provincial Departments must confirm human resources capacity to implement
	Ilima/Letsema business plans by 1 April 2011
	• Provinces must inform the national transferring officer of any changes from plans and
	allocations approved by national transferring officer within 30 days of such change
	The business plans must be signed off by the Head of Department (HOD) of Agriculture in collaboration with the head of the Provincial Treasury
Allocation criteria	The formula used to allocate funds is a weighted average of LARP priority areas and targeted areas of increased food production
Reason not incorporated in	• The funding originated with the special poverty allocations made by national government
equitable share	for a specific purpose and a conditional grant enables government to ensure specific
D 4 6	projects are targeted
Past performance	2009/10 audited financial outcomes
	<ul> <li>Allocated and transferred R50 million to provinces</li> <li>Of the total available R70.9 million (including roll overs), R66.9 million (94.3 per cent) was</li> </ul>
	spent
	2009/10 service delivery performance
	• 1 053 jobs were created
	• 12 967 beneficiaries/households were supported
	8 029 hectares planted with seeds
	3 irrigation schemes were rehabilitated
Projected life	Grant will be reviewed during 2011/12
MTEF allocations	• 2011/12: R400 million, 2012/13: R420 million and 2013/14: R443 million
Payment schedule	• 20 per cent: 21 April 2011, 35 per cent: 22 July 2011, 35 per cent: 7 October 2011 and 10 per cent: 11 January 2012
Responsibilities of the	Responsibilities of the national department
transferring national officer	Agree on outputs and targets with provincial departments in line with grant objectives      Paving and ravious parms and standards for the implementation of the grant during
and receiving officer	• Revise and review norms and standards for the implementation of the grant during
	Intergovernmental Technical Committee on Agriculture and Land Affairs (ITCAL)- Letsema workshops held quarterly
	<ul> <li>Provide the guidelines and criteria for the development and approval of business plans</li> </ul>
	Monitor implementation through project site visits and provide support on monthly and
	quarterly basis
	• Submit quarterly performance reports to National Treasury within 45 days after the end of
	each quarter
	Convene the National Assessment Panel (NAP)

Ilima/Letsema Projects Grant				
	Responsibilities of the provincial departments			
	Provinces to report monthly (financial) and quarterly (non-financial) on the progress of the projects			
	Participate in the National Assessment Panel			
	Provinces to implement the projects according to the approved business plans and notify DAFF in writing about deviations before implementation of projects can take place			
	Monitor projects implementation and evaluate the impacts of projects in achieving			
	Ilima/Letsema goals			
Process for approval of the 2012/13 business plans	<ul> <li>Provide provincial departments with business plan format guidelines, criteria and outputs as prescribed by National Treasury by 31 May 2011</li> </ul>			
•	Submission of provincial and individual Ilima/Letsema business plans by provinces on 30 September 2011			
	• Engagement with provinces on submitted business plans during October 2011 prior to NAP			
	Evaluation and recommendation of business plans by NAP during November 2011			
	• Sent Funding Agreements to provinces by January/February 2012 to be signed by the Heads			
	of Departments, Chief Financial Officers, and CASP/Ilima/Letsema Coordinators			
	Approval of business plans by the Minister before 30 March 2012			
	• Inform provinces of approval of the business plan March/April 2012			

	Land Care Programme Grant: Poverty Relief and Infrastructure Development
Transferring department	Agriculture, Forestry and Fisheries (Vote 26)
Strategic goal	To promote conservation of agricultural natural resources to optimise production
Grant purpose	• To enhance the sustainable conservation of natural agriculture resources through a community-based
	participatory approach
	To create job opportunities through the Expanded Public Works programme  To create job opportunities through the Expanded Public Works programme
0-4	<ul> <li>To create an enabling environment for improved food security and poverty relief</li> <li>Improved protection and utilization of agricultural natural resources</li> </ul>
Outcome statements	Improved protection and utilization of agricultural natural resources     Improved Land Care ethics and high level of public awareness
	Sustained and efficient use of water resources in agricultural production
	Management of soil degradation
	Making arable lands available for agricultural production and grazing
	Effective management of run-off and protection of arable land
Outputs	9 233 youths successfully attending organized Junior Land Care initiatives
	<ul> <li>28 695 jobs created on Land Care projects including Conservation Agriculture practices</li> <li>8 297 ha of land alien invasive plants eradicated</li> </ul>
	<ul> <li>8 297 ha of land alien invasive plants eradicated</li> <li>75 per cent to 55 per cent reduction of degraded land in South Africa</li> </ul>
	1 382 ha of grazing area improved
	34 755 ha of cultivated land rehabilitated and or protected
	• 798 structures of gabions constructed
	598 ha of wetlands protected
	3 327 people with improved capacity and skill levels benefiting from capacity building initiatives
Priority outcome(s) of	Outcome 7: Vibrant, equitable and sustainable rural communities with food security for all
government that this	Outcome 10: Environmental assets and natural resources that are well protected and continually enhanced
grant primarily contributes to	
	Outroma in Harton
Details contained in the	Outcome indicators     Outputs indicators
business plan	• Inputs
	Key activities
	Monitoring and evaluation
	Risk and mitigation
	Exit strategy
Conditions	Provinces must confirm capacity to implement projects and operational funding by March 2010
	<ul> <li>Provincial department's Annual Performance Plans for 2011/12 must clearly indicate measurable objectives and performance targets as per approved business plans with the Department of Agriculture,</li> </ul>
	Forestry and Fisheries (DAFF)
Allocation criteria	The allocation criteria uses an index comprising of nodes, land capability, poverty, degradation and land
	size derived from the following sources:
	- nodes are the ISRDP nodes and URP (CoGTA)
	<ul> <li>poverty = per cent share in poverty gap (Human Science Research Council)</li> <li>degradation = ha X 100 000 - (National land cover 2000)</li> </ul>
	- size = number of ha per province - (new boundaries from Municipal Demarcation Board of SA,
	2008)
Reason not incorporated	• The funding originated with the special poverty allocations made by National Government for a specific
in equitable share	purpose
Past performance	2009/10 audited financial outcomes
	Allocated and transferred R51 million to provinces
	• Of the total available of R54 million to provinces (including provincial rollovers), expenditure was
	R52 million (96.7 per cent)  2009/10 service delivery performance
	169 Junior Care management sub-programme implemented
	24 695 jobs created on Land Care projects
	8 297 ha of land alien invasive plants eradicated
	1 382 ha of grazing area improved
	559 ha of wetlands protected
	39 water sources developed or protected against over-utilization
D	798 structures of gabions were constructed     Operains and subject to review.
Projected Life	Ongoing and subject to review
MTEF allocations	• 2011/12: R58 million, 2012/13: R116 million and 2013/14: R109 million
unventions	

	Land Care Programme Grant: Poverty Relief and Infrastructure Development
Payment schedule	• 10 per cent: 21 April 2011, 35 per cent: 22 July 2011, 35 per cent: 7 October 2011 and 20 per cent:
	11 January 2012
Responsibilities of the	Responsibilities of the national department
transferring national	Agree on outputs and targets with provincial departments in line with grant objectives for 2012/13
officer and receiving	Review norms and standards for the implementation of the grant during ITCAL-Letsema workshops held
officer	quarterly
	Provide the guidelines and criteria for the development and approval of business plans
	Monitor implementation through project site visits and provide support quarterly or as required
	Submit quarterly performance reports to National Treasury within 45 days after the end of each quarter
	Responsibilities of the provincial departments
	• Provinces to report monthly (financial) and quarterly (non-financial) on the progress of the projects
	• Provinces to implement the projects according to the approved business plans, any deviation should first
	be communicated to DAFF in writing and approved by DAFF before implementation
	Monitor projects implementation and evaluate the impacts of projects in achieving Land Care goals
	Provinces to report EPWP projects to public works department
Process for approval of	Provide provincial departments with business plan format guidelines, criteria and outputs as prescribed by
2012/13 business plans	National Treasury by 31 May 2011
2012/10 Submess Plans	Submission of provincial and individual Land Care business plans by provinces on 30 September 2011
	• Engagement with provinces on submitted business plans during October 2011 prior to National
	Assessment Panel (NAP)
	Evaluation and recommendation of business plans by NAP during November 2011
	• Interactions with provinces requested by NAP to correct the business plans accordingly prior Ministerial
	approval and re-submit by 30 November 2011, only if necessary
	• Send Funding Agreements to provinces by January/February 2012 to be signed by the Heads of
	Departments, Chief Financial Officers and Land Care Coordinators
	Approval of business plans by the Director-General before 31 March 2012

# ARTS AND CULTURE GRANT

	Community Library Services Grant
Transferring department	Arts and Culture (Vote 14)
Strategic goal	To enable the South African society to gain access to knowledge and information that will improve their socio-economic situation
Grant purpose	<ul> <li>To transform urban and rural community library infrastructure, facilities and services (primarily targeting previously disadvantaged communities) through a recapitalised programme at provincial level in support of local government and national initiatives</li> </ul>
Outcome statements	<ul> <li>Improved coordination and collaboration between national, provincial and local government on library services</li> <li>Transformed and equitable library and information services delivered to all rural and urban communities</li> <li>Improved library infrastructure and services that reflect the specific needs of the</li> </ul>
	<ul> <li>communities they serve</li> <li>Improved staff capacity at urban and rural libraries to respond appropriately to community knowledge and information needs</li> <li>Improved culture of reading</li> </ul>
Outputs	<ul> <li>Signed agreements between national, provincial and local government on the planning, management and maintenance of community libraries</li> <li>500 000 library materials (books, periodicals, toys etc) purchased</li> <li>Library Information and Communication Technology (ICT) infrastructure and systems using open source software in all provinces</li> <li>Library material and services for the visually impaired at community libraries in three provinces</li> <li>14 new libraries structures</li> <li>75 upgraded and maintained library structures</li> <li>Additional community library staff appointed</li> <li>Capacity building programmes</li> </ul>
Priority outcome(s) of government that this grant primarily contributes to	Outcome 12: An efficient, effective and development oriented public service and an empowered, fair and inclusive citizenship
Details contained in the business plan	<ul> <li>Outcome indicators</li> <li>Output indicators</li> <li>Inputs</li> <li>Key activities</li> </ul>
Conditions	<ul> <li>The provincial business plans must be developed in accordance with identified priority areas</li> <li>This funding must not be used as a replacement funding for provinces</li> <li>Provinces may use a maximum of 5 per cent of the total amount allocated to them for capacity building and provincial management of the grant. The detail of how these funds will be used must be included in the respective business plans</li> <li>Service level agreements determining reporting protocols must be signed with receiving municipalities</li> </ul>
Allocation criteria	The distribution formula is based on an evaluation report for 2009/10 as well as an impact assessment study conducted by the department which identified community library needs and priorities
Reason not incorporated in equitable share	<ul> <li>This funding is intended to address backlogs and disparities in the ongoing provision and maintenance of community library services across provinces and enable national department to provide strategic guidance and alignment with national priorities</li> </ul>
Past performance	<ul> <li>2009/10 audited financial outcomes</li> <li>Allocated and transferred R441 million to provinces</li> <li>Of the total available R471 million (including provincial rollovers), R418 million (88.7 per cent) was spent</li> <li>2009/10 service delivery performance</li> <li>43 libraries upgraded</li> <li>7 libraries built</li> </ul>
Projected life	The projected life will be informed by ongoing evaluation reports
MTEF allocations	• 2011/12: R543 million, 2012/13: R571 million and 2013/14: R602 million
Payment schedule	• Four instalments (21 April 2011, 29 July 2011, 28 October 2011 and 31 January 2012)
Responsibilities of the transferring national officer and receiving officer	Responsibilities of the national department     Identify risks and challenges     Monitor and evaluate implementation     Evaluate annual performance of the grant for the 2010/11 financial year, for submission to National Treasury     Submit monthly financial and quarterly performance reports to National Treasury     Determine outputs and targets for 2012/13 with provincial departments

Community Library Services Grant			
	Responsibilities of the provincial departments		
	Monitoring and evaluation of the programme and municipalities		
	Establish library governance structures in all provinces		
	<ul> <li>Submit monthly financial and quarterly performance reports including quarterly expenditure reports of municipalities to Department of Arts and Culture (DAC)</li> </ul>		
Process for approval of 2012/13	<ul> <li>Provinces to submit draft business plans to DAC by 30 September 2011</li> </ul>		
business plans	• DAC to evaluate provincial business plans and provide feedback to provinces by 30 November 2011		
	<ul> <li>Provinces to submit final provincial business plans to DAC by January 2012</li> </ul>		
	DAC approves business plans and submits them to National Treasury by 16 March 2012		

# **BASIC EDUCATION GRANTS**

Transferring department	Dinaledi Schools Grant  Basic Education (Vote 15)
Strategic goal	To improve the participation and performance of learners in Mathematics and Physical
Strategic goar	Science in line with the National Strategy for Mathematics, Science and Technology Education (NSMSTE)
Grant purpose	<ul> <li>To promote Mathematics and Physical Science teaching and learning</li> <li>Improve learner performance in Mathematics and Physical Science in line with the Action Plan 2014</li> </ul>
	Improve teachers content knowledge of Mathematics and Physical Science
Outcome statements	Continually increased performance of Mathematics and Physical Science learning and teaching in underprivileged schools
Outputs	• Up to 7 Textbooks for each learner in Grades 8, 9, 10, 11 and 12 learners as is required to address shortages
	Mobile Science laboratories to 300 Dinaledi schools without labs and science kits to 500 Dinaledi Schools
	Mathematics kits for 500 Dinaledi schools
	<ul> <li>300 ICT laboratories in Dinaledi schools without ICT Labs</li> <li>Televisions that can receive education TV Broadcasts and solutions installed in 500</li> </ul>
	Dinaledi schools
	<ul> <li>50 computers (specify) in each of the 500 schools</li> <li>Mathematics, Physical Science and English FAL teaching and learning software at 500</li> </ul>
	Dinaledi Schools
	2000 teachers trained on content knowledge in mathematics , physical science and English FAL
	<ul> <li>Distribution to 500 Dinaledi schools of a diagnostic tool to assess, measure and address learning and teaching deficiencies in mathematics, physical science and English FAL</li> <li>Capacity development of principals in 90 Dinaledi schools achieving below 60% pass</li> </ul>
	rate in the NSC examinations  Grade 8, 9 and 10 learners in 500 Dinaledi schools trained and supported to participate in
7.1.1	mathematics and science Olympiads
Priority outcome(s) of government that this	Outcome 1: Improved quality of basic education
grant primarily contributes to	
Details contained in the	Outcome indicators
business plan	Output indicators
	• Inputs
	Key activities     Right management plan
Conditions	<ul> <li>Risk management plan</li> <li>Provinces may only procure through suppliers identified by the Department of Basic</li> </ul>
Condition	Education (DBE). Provinces must submit procurement plans for materials purchased for Dinaledi Schools to the DBE by 6 May 2011 for approval. The payment of the second instalment of this grant is contingent upon receipt and approval by DBE of these procurement plans
	Provinces must distribute budget allocation in terms of the needs analysis contained in the provincial business plans
	Provinces must appoint project managers to support provincial administration in Dinaledi Schools who will also be responsible for District Development and SMT mentoring Support
	Provinces must prove to the (transferring national officer) DBE that they have the necessary capacity to implement the grant effectively
Allocation criteria	Dinaledi Schools will have to satisfy the following criteria:
	- At least 60 per cent of the learners enrolled in Mathematics and a maximum of 40 per cent of these may be enrolled for Mathematical Literacy
	- At least 60 per cent of learners offering Mathematics should achieve 50 per cent and above  At least 20 learners should have passed Mathematics/Physical Science at 50 per cent
	- At least 20 learners should have passed Mathematics/Physical Science at 50 per cent and above  There is an improvement in enrolment and performance in Mathematics/Physical
	<ul> <li>There is an improvement in enrolment and performance in Mathematics/Physical Science year on year at 50 per cent and above</li> <li>There should be more than 50 Grade 12 learners enrolled</li> </ul>
	- Serve disadvantaged communities
	The minimum criteria of being a Dinaledi School must be adhered to as other schools
	may be considered for inclusion if the current schools do not meet performance standards

Disolali Sahaala Cuunt	
Daggar not incompared	Dinaledi Schools Grant
Reason not incorporated in equitable share	• The Dinaledi Schools Project is a pilot project and will be used as a model for other
_	schools to improve performance in mathematics and physical science
Past performance	2009/10 audited financial outcomes
	New grant
	2009/10 service delivery performance
	New grant
Projected life	Until 2013/14, subject to review
Payment schedule	• Four instalments (20 April 2011, 15 July 2011, 28 October 2011 and 27 January 2012)
MTEF allocations	• 2011/12: R70 million, 2012/13: R100 million and 2013/14: R105 million
Responsibilities of the	Responsibilities of the national department
transferring national	• The DBE will monitor the performance of the provincial departments quarterly and will
officer and receiving	evaluate in June 2011 and in November 2011
officer	Develop a Monitoring and Evaluation Plan
	Develop and submit approved national business plans to National Treasury
	Evaluate, approve and submit provincial business plans to National Treasury
	Manage, monitor and support the programme implementation in provinces and evaluate
	provincial capacity to implement the grant
	• Consolidate and submit quarterly performance reports to National Treasury within 45
	days after the end of each quarter
	• Evaluate performance of the conditional grant and submit an evaluation report to
	National Treasury by 29 July 2011
	Enter into agreements with suppliers of materials for Dinaledi schools
	Responsibilities of the provincial departments
	• Comply with the minimum requirements of the transferring national officer (DBE)
	monitoring and evaluation plan
	<ul> <li>Develop and submit approved business plans to DBE</li> </ul>
	Appoint competent project managers to monitor and provide support to districts/regions
	and schools
	Submit approved quarterly financial and narrative reports to DBE
	• Report to DBE on the human resource capacity available in the province to implement
	the grant
	• Evaluate the performance of the conditional grant and submit evaluation reports to DBE
	by 31 May 2011
	• Submit a quarterly performance report to DBE within 30 days after the end of each
	quarter
Process for approval of	• Consultation with district officials, finance sections of provincial treasuries and National
2012/13 business plans	Treasury
	• Provinces submit first draft business plans and report on capacity to implement the grant
	to the DBE by 28 October 2011
	DBE evaluates draft business plans and sends comments to provinces by
	15 November 2011 and provides feedback to the provinces by 15 December 2011
	Provinces submit final business plans to DBE by 28 January 2012
	Director-General approves national and provincial business plans by 2 April 2012

Education Infrastructure Grant	
Transferring department	Basic Education (Vote 15)
Strategic goal	To supplement provinces to fund provincial education infrastructure
Grant purpose	To help accelerate construction, maintenance, upgrading and rehabilitation of new and
	existing infrastructure in education
	To enhance capacity to deliver infrastructure in education
Outcome statements	Improved service delivery by provincial departments as a result of improved and increased stock of schools infrastructure
	Aligned and coordinated approach to infrastructure development at the provincial level
	Improved education infrastructure expenditure patterns
Outputs	New schools and additional libraries and laboratories constructed
	Existing school infrastructure upgraded and rehabilitated
	New and existing schools maintained  Outcome 1. Language descriptions described.
Priority outcome(s) of government that this grant primarily contributes to	Outcome 1: Improved quality of basic education
Details contained in the	• This grant uses a User Asset Management Plan (U-AMP), which contains the following:
business plan	<ul> <li>Levels of service</li> <li>Demand and need determination</li> </ul>
	- Project lists for 2011/12 and 2012/13
	- Financial summary
	- Organisational and support plan
Conditions	<ul> <li>Provincial Education Departments (PEDs) must submit tabled User Asset Management Plans (U-AMPs), (formerly Infrastructure Plans) with prioritised project lists that are signed-off for the 2011 MTEF by 8 April 2011 to the national Department of Basic Education (DBE) and the relevant Provincial Treasuries. This deliverable is required for</li> </ul>
	the transfer of the first instalment of the grant
	• The flow of the second instalment depends on the receipt by DBE of fourth quarter infrastructure reports for the 2010/11 financial year and final list of projects captured on the IBM by 21 April 2011. These projects appear to a positive data the relevant Provincial.
	the IRM by 21 April 2011. These reports must be submitted to the relevant Provincial Treasury  • The flow of the third instalment is dependent upon receipt by DBE of the draft 2012/13
	U-AMPs including the initial list of prioritised projects as captured in the Infrastructure Programme Management Plan (IPMP). The 2012/13 project list must be drawn from the prioritised project list for the MTEF tabled in 2011/12. Changes to the MTEF prioritised list of projects must be motivated to the DBE
	The flow of the third, fourth and fifth instalments are conditional upon submission and approval of signed-off quarterly infrastructure reports and education project assessments forms
	<ul> <li>Quarterly reports on the implementation of infrastructure projects via the Infrastructure Reporting Model (IRM) are required in addition to monthly In-Year Monitoring expenditure reports. Client departments must enter into service delivery agreements (SDAs) with their Implementing Agents. The development or review of the SDA should continue in parallel with the development of the IPMP</li> </ul>
	Provincial Education Capacity plans must be approved by the DBE
	All immovable asset management and maintenance responsibilities as prescribed by the Government Immovable Asset Management Act of 2007 (GIAMA) and the Infrastructure Delivery Management Toolkit must be adhered to
	PEDs must provide School Governing Bodies with maintenance guidelines to conduct minor maintenance
	Major maintenance function must be budgeted and planned for at the provincial level and carried out through the district department. This must be captured in the U-AMP
Allocation criteria	Allocations for 2011/12 and 2012/13 are based on historical division within the previous Infrastructure Grant for Provinces
	The allocation criteria for 2013/14 will be reviewed and therefore allocations currently shown may change  The living for the control of
Reason not incorporated in equitable share	Funding infrastructure through a conditional grant enables the national department to ensure the delivery and maintenance of education infrastructure is coordinated and efficient
Past performance	2009/10 pre-audited financial outcomes
	• New grant
	2009/10 service delivery performance  New grant
Projected life	<ul> <li>New grant</li> <li>To be reviewed by DBE after five years</li> </ul>
110jecteu me	- 10 be reviewed by DDD after five years

Education Infrastructure Grant	
MTEF allocations	• 2011/12: R5 498 million, 2012/13: R5 883 million and 2013/14: R6 207 million
Payment schedule	• 12.5 per cent: 19 April 2011, 12.5 per cent: 27 May 2011, 25 per cent: 16 September 2011, 25 per cent: 18 November 2011, and 25 per cent: 13 January 2012
Responsibilities of the	Responsibilities of the national department
transferring national officer and receiving officer	<ul> <li>DBE will visit selected infrastructure sites in provinces</li> <li>DBE and National Treasury will support provinces to improve infrastructure delivery capacity and systems</li> <li>DBE must provide guidance to provinces in planning and prioritisation and evaluate U-AMPs and prioritised projects that provinces develop and submit</li> <li>DBE to report on quarterly performance on infrastructure delivery to the Council of Education Ministers (CEM) and to National Treasury</li> <li>DBE will provide provinces with a framework for capacity development plans</li> <li>DBE will convene and chair meetings of the National Steering Committee during which assessments will be made of projects to be supplemented by this grant and projects funded by the School Infrastructure Backlogs Grant to ensure that the funding from the grants is directed to projects that are aligned with respective grant objectives</li> <li>Submit quarterly performance reports to National Treasury and National Council of Provinces within 45 days after the end of each quarter</li> </ul>
	Responsibilities of the provincial departments
	<ul> <li>Provinces must have an Infrastructure Projects Progress Review committee to oversee, monitor and report on the full physical and financial development cycle of all funded projects, and report per prescribed reporting formats and Infrastructure Delivery Management Toolkit</li> <li>Provincial departments must attend and participate in meetings of the national steering committee and submit relevant project information within timeframes as reasonably requested by the National Department</li> <li>Convene and chair the Provincial Progress Review Meetings</li> <li>Compile and submit comprehensive projects progress report to DBE, Provincial Treasury and National Treasury</li> <li>PEDs must table U-AMPs together with draft strategic and annual performance plans</li> </ul>
Process for approval of 2012/13 business plans	<ul> <li>Submission of draft U-AMPs and prioritised list of projects for 2012/13 by PEDs to DBE by 12 August 2011, or a later date as may be determined by DBE</li> <li>Submission of IPMPs for 2012/13 by Client Departments to Implementing Agents by 2 September 2011</li> <li>Implementing Department(s) or Agent(s) must submit the Infrastructure Programme Implementation Plans (IPIP) for 2012/13 to Client departments by 30 November 2011. Copies of IPIPs and SDAs must be submitted to DBE at the same time</li> </ul>

	HIV and Aids (Life Skills Education) Grant
Transferring department	Basic Education (Vote 15)
Strategic goal	To enhance awareness programmes offered by schools to prevent and mitigate the
	impact of HIV
	Increased knowledge, skills and confidence amongst learners and educators to take self-
	appropriate sexual and reproductive health decisions
	To increase access to sexual and reproductive health services including HIV services for learners and educators
Grant purpose	To support South Africa's HIV prevention strategy by increasing sexual and
Grant pur pose	reproductive knowledge, skills and appropriate decision making among learners and
	educators
	• To mitigate the impact of HIV by providing a caring, supportive and enabling
	environment for learners and educators
	To ensure the provision of a safe, rights-based environment in schools that is free of
•	discrimination, stigma and any form of sexual harassment/abuse
Outcome statements	Educators receive in-service training on sexual and reproductive health including HIV
	• Sexual and reproductive health education, including HIV is a mandatory, timetabled
	and assessed subject delivered in all South African schools primarily through the Life Orientation/ Skills subject
	Every South African school has a communicated plan in place to increase access to
	sexual and reproductive health and HIV services for learners and educators
	Barriers to retention and achievement in school for learners who are HIV affected or
	infected are mitigated by implementing pro-poor policies
	Age-appropriate sexual and reproductive health and HIV-related life skills are delivered
	through co-curricular means in all South African schools
	Schools, districts and provinces have integrated all components of the HIV and Aids life skills programme into their core work, evaluation and reporting systems
Outputs	Master trainers trained in the integration of sexual and reproductive health (SRH)
o arpais	programmes into the school curriculum
	Educators trained to implement SRH programmes for learners and to protect themselves
	from HIV
	School Management Teams (SMT) trained to develop school implementation plans
	focusing on keeping young people in school; ensuring that SRH education is
	implemented for all learners in schools; and ensuring access to SRH services
	Co-curricular activities on SRH implemented in schools     Care and support programmes implemented for learners and educators
	Care and support programmes impremented for learners and educators     Curriculum and Assessment Policy Statement (CAPS) compliant material including
	material for learners with barriers to learning printed and distributed to schools
	Advocacy events hosted with learners, educators and school communities
	Monitoring and support visits conducted at district and school levels
Priority outcome(s) of	Outcome 1: Improved quality of basic education
government that this	
grant primarily	
contributes to Details contained in	Outcome indicators
the business plan	Outcome indicators     Output indicators
the business plan	Inputs
	Key activities
Conditions	As agreed to at the interprovincial meeting of the National Department of Basic
	Education and Provincial Education Departments on 3 and 4 December 2010, provinces
	must distribute the grant allocation in accordance with the following weights for the key
	performance areas. The allocation must be spent on the specific activities as described
	in the draft version of this conditional grant framework e-mailed by DBE to provinces
	on 14 December 2010: - training and development (25 per cent)
	- co-curricular activities (20 per cent)
	- care and support (20 per cent)
	- Learning and Teaching Support Material (15 per cent)
	- Advocacy and Communication (10 per cent)
	- Management, support, monitoring and evaluation (10 per cent)

HIV and Aids (Life Skills Education) Grant	
	The above percentages are guidelines and may only be deviated from if a deviation is motivated to and approved by the national accounting officer. Deviations should only be requested to deal with critical challenges and trends in the epidemic specific to the province
	Provinces must report on the implementation of their business plans in their quarterly reports and deviation from the business plans and use of the grant on activities not described in the draft conditional grant framework will result in the withholding of funds
	Provincial Education Departments must ensure that they have the necessary capacity and skills to manage the grant
Allocation criteria	The education component of the equitable share formula as explained in Annexure W1 to the 2011 Division of Revenue Bill is used to allocate the grant amongst provinces
Reason not incorporated in equitable share	To enable the DBE to provide overall guidance, to ensure congruency, coherence and alignment with the government's HIV and Aids and STI National Strategic Plan 2007-2011. This enables DBE to exercise an oversight role in the implementation of the HIV and Aids Life Skills Programme in schools
Past performance	2009/10 audited financial outcomes
_	• From the total allocation of R183 million (including roll-overs), provinces spent R169
	million (91.9 per cent)
	2009/10 service delivery performance
	46 867 learners and 4 500 educators trained as peer educators
	25 609 educators trained in care and support
	• 23 273 educators teaching other learning areas trained in life skills
	1 069 district officials trained as master trainers  253 000
	• 353 000 sets of Learning and Teaching Support Material delivered to 15 000 schools
	115 363 school principals, educators, learners and parents reached through advocacy     activities.
	activities  • 4 937 schools reached through monitoring and support
Projected life	<ul> <li>4 937 schools reached through monitoring and support</li> <li>The grant will be reviewed on an ongoing basis to respond to the nature of the pandemic</li> </ul>
•	
MTEF allocations	• 2011/12: R199 million, 2012/13: R209 million and 2013/14: R221 million
Payment schedule	Four instalments: 7 April 2011, 29 July 2011, 28 October 2011 and 31 January 2012
Responsibilities of transferring national	Responsibilities of the national department  Identify risks and challenges impacting on provincial implementation
officer and receiving	Develop risk management strategies to address the risks
officer	Ensure synergy with national strategies and processes aimed at reducing HIV infection
	in the country
	Agree on outputs and targets with provincial departments in line with grant objectives and national imperatives for 2012/13 by 30 September 2011
	Provide evidence-based guidance for the development of business plans based on monitoring and findings from international and national research
	Monitor implementation of the programme and provide support to provinces  Output  Description of the programme and provide support to provinces
	Develop partnerships with key stakeholders  Find the conditional partnership with the conditio
	Evaluate performance of the conditional grant and submit an evaluation report to National Treasury by 29 July 2011
	Submit quarterly performance reports to National Treasury within 45 days after the end of each quarter
	Responsibilities of the provincial departments
	Ensure synergy with national strategies and processes aimed at reducing HIV infection and all other related issues
	Identify risks and challenges impacting on implementation
	Develop risk management strategies and implementation plans to address the risks
	Submit monthly, quarterly and annual performance reports to DBE in line with DoRA and PFMA
	Agree with the DBE on outputs and targets to ensure effective implementation of the
	programme
	Monitor implementation of the programme and provide support to districts and schools
	Evaluate and submit a provincial evaluation report on the performance of conditional grant to the DBE by 31 May 2011
	grain to the DDE by 31 May 2011

HIV and Aids (Life Skills Education) Grant	
	Submit quarterly performance reports to DBE within 30 days after the end of each quarter
Process for approval of 2012/13 business plans	Communication and meeting with provinces to inform targets for the next financial year by 30 September 2011
·	<ul> <li>Provinces submit draft business plans to DBE for evaluation by 30 November 2011</li> <li>The DBE evaluates provincial business plans by 9 December 2011</li> <li>Comments sent to provinces to amend the plans by 10 January 2012</li> <li>Provinces submit amended, signed plans to DBE by 29 February 2012</li> <li>DBE secures the Director-General's approval of provincial business plans by 2 April 2012</li> </ul>

	National School Nutrition Programme Grant
Transferring department	Basic Education (Vote 15)
Strategic goal	To enhance learning capacity and to improve access to education
Grant purpose	To provide nutritious meals to targeted learners
Outcome statements	Enhanced learning capacity and improved access to education
Outputs	Nutritious meals served to learners
Priority outcome(s) of	Outcome 1: Improved quality of basic education
government that this grant	
primarily contributes to	
Details contained in	Outcome indicators     Outcome indicators
the business plan	Output indicators     Inputs
	The wo
	<ul><li>Key activities</li><li>Risk Management Plan</li></ul>
Conditions	Develop national and provincial business plans
Conditions	<ul> <li>A minimum of 95 per cent of the grant must be used for school feeding at both secondary and primary</li> </ul>
	schools. A maximum of 5 per cent of the grant may be used for administration and other activities
	Provinces must adhere to the following minimum feeding requirements:
	— provide nutritious meals to all learners in quintile 1 to 3 primary and secondary schools (as per
	gazetted national quintiles) as well as identified special schools on all school days
	<ul> <li>cost per meal per learner in primary schools as well as identified special schools at an average of</li> </ul>
	R2.46 and in secondary schools at an average R3.36, inclusive of cooking fuel and honorarium
	a minimum honorarium of R640 per person per month, in line with a food handler to learner ratio of 1:200. A ratio of 1:120 is recommended for schools where learner enrolment is 250 or fewer
	comply with recommended food specifications and approved menu
	<ul> <li>fresh fruit/vegetables must be served daily and vary between green and yellow/red on a weekly basis</li> </ul>
	— a variety of protein food must be served per week in line with approved menu options. Soya should
	not be served more than twice a week
	<ul> <li>pilchards should be served at least once a week. High quality protein products can replace pilchard in</li> </ul>
	areas where it is not socially acceptable. In areas where fresh milk/maas is unavailable, only whole
	powdered milk may be used
	<ul> <li>meals should be served to learners by 10h00</li> <li>The 10 May 2011 budget transfer (as per payment schedule) is for cooking facilities, equipment and</li> </ul>
	utensils for quintile 1-3 primary schools as per equipment specifications provided by the Department
	<ul> <li>Provinces must promote sustainable food production and nutrition education</li> </ul>
	• Provincial business plans will be approved in line with the above minimum requirements and
	available resources. The following variations may be approved by the Transferring National Officer
	based on achievements and/or critical challenges in each province
	<ul> <li>feeding days reduced to a minimum of 193 days</li> </ul>
	— feeding cost below the above stated minimum requirements, which provide meals with
	maximum nutritional value as per menu specifications  — number of learners that exceed the gazetted quintiles, but only if the province is able to meet
	minimum standards of the programme for all learners in gazetted quintiles
	<ul> <li>serving of processed vegetables or fruit in remote areas</li> </ul>
	feeding time later than 10h00 under special provincial circumstances such as provisioning of
	school breakfast and circumstances beyond control
Allocation criteria	• The distribution formula is poverty based in accordance with the poverty distribution table used in the
	National Norms and Standards for School Funding as gazetted by the Minister of Education on
	6 November 2009
Reason not incorporated in	• The National School Nutrition Programme (NSNP) is a government programme for poverty
equitable share	alleviation, specifically initiated to uphold the rights of children to basic food and education. The
	conditional grant enables the Department of Basic Education (DBE) to play an oversight role in the implementation of all the NSNP activities in schools
Past performance	2009/10 audited financial outcomes
1 ast periormanee	Allocated and transferred R2 394 million to provinces
	<ul> <li>Of the total available of R2 536 million (including provincial rollovers), R2 497 million (99 per cent)</li> </ul>
	was spent
	2009/10 service delivery performance
	The programme was successfully extended to Quintile 1 secondary schools for the first time in
	April 2009
	Meals were provided to 6 181 574 learners in 18 384 Quintiles 1 to 3 public primary schools and
	943 699 learners in 1 961 Quintile 1 public secondary schools, thus reaching a total of 7 125 273
	learners in 20 345 schools nationally
	• 19 643 recipe books entitled 'Mnandi 4 Sure' were printed and distributed to schools
	5 868 vegetable gardens were maintained to teach school communities about food production and to supplement school meals
Duningted life	
Projected life	• Given the need for the grant and negative impact of factors associated with poor nutrition has on
	education, the grant will be necessary for a number of years. However it will be subject to review at a

	National School Nutrition Programme Grant
	time agreed to by the Department of Basic Education and the National Treasury
Payment schedule	<ul> <li>The payment schedule will be in five instalments and in line with respective provincial procurement models as follows:         <ul> <li>Provinces that transfer funds directly to schools (Free State, Gauteng, North West and Northern Cape): (07 April 2011, 10 May 2011, 14 June 2011, 14 September 2011 and 09 December 2011)</li> <li>Provinces that procure from service providers on behalf of schools: (07 April 2011, 10 May 2011, 17 August 2011, 28 October 2011 and 12 January 2012)</li> </ul> </li> </ul>
MTEF allocations	• 2011/12: R4 579 million, 2012/13: R4 928 million and 2013/14: R5 199 million
Responsibilities of the	Responsibilities of the national department
transferring national officer	<ul> <li>Develop and submit approved national business plans to National Treasury</li> </ul>
and receiving officer	<ul> <li>Evaluate, approve and submit provincial business plans to National Treasury</li> </ul>
	<ul> <li>Manage, monitor and support programme implementation in provinces</li> </ul>
	Ensure compliance with reporting requirements and NSNP guidelines
	<ul> <li>Consolidate and submit quarterly performance reports to National Treasury within 45 days after the end of each quarter</li> </ul>
	<ul> <li>Evaluate performance of the conditional grant and submit an evaluation report to National Treasury annually by 31 July</li> </ul>
	Responsibilities of the provincial departments
	<ul> <li>Develop and submit approved business plans to DBE</li> </ul>
	<ul> <li>Monitor and provide support to districts/regions/area project offices and schools</li> </ul>
	Submit approved quarterly financial and narrative reports to DBE
	<ul> <li>Monitor and evaluate implementation of the programme according the departments monitoring and evaluation plan</li> </ul>
	Provide human resource capacity at all relevant levels
	• Evaluate the performance of the conditional grant annually and submit evaluation reports to the DBE by 31 May
	Submit quarterly performance reports to DBE after the end of each quarter
Process for approval of	<ul> <li>Planning meeting between DBE and provincial departments by 29 July 2011</li> </ul>
2012/13 business plans	<ul> <li>Consultation with district officials, provincial treasuries, provincial finance sections and National Treasury</li> </ul>
	Provinces submit first draft business plans to DBE by 15 September 2011
	<ul> <li>Inter-provincial meeting held in September 2011 to consult provinces on the 2012/13 conditional grant framework</li> </ul>
	DBE evaluates draft business plans and sends comments to provinces by 13 October 2011
	Provinces submit final business plans to DBE by 27 January 2012
	Director-General approves national and provincial business plans by 2 April 2012

Technical Secondary Schools Recapitalisation Grant	
Transferring department	Basic Education (Vote 15)
Strategic goal	To improve conditions of technical schools and modernise them to meet the teaching requirements of learners in the technical fields and increase the number of suitably qualified and technically skilled graduates from these schools
Grant purpose	To recapitalise up to 200 technical schools to improve the capacity to contribute to skills development and training in the country by:     building workshops at technical schools to support the technical subject offerings refurbishing or re-designing workshops in technical schools to comply with safety laws and regulations and to meet minimum industry standards buying and installing new machinery and equipment consistent with the technical subjects that are offered in technical schools training and up-skilling teachers at technical schools to acquire new trends, practical skills, and developments in their technical subjects.
Outcome statements	Skills Development Strategy supported by training young people in relevant technical skills
Outputs	<ul> <li>32 new workshops to be built to support the technical subject offerings</li> <li>147 existing workshops refurbished, upgraded and re-designed to comply with safety laws and regulations and minimum industry standards</li> <li>Equipment bought, delivered and installed at 191 workshops</li> <li>578 Technical schools teachers trained in practical skills of the subject content delivery</li> </ul>
Priority outcome(s) of government that this grant primarily contributes to	Outcome 1: Improved quality of basic education
Details contained in the business plan	<ul> <li>Outcome indicators</li> <li>Output indicators</li> <li>Inputs</li> <li>Key activities</li> <li>Risk management plan</li> </ul>
Conditions	<ul> <li>Three year provincial recapitalisation plans for technical schools must be submitted to and approved by the Department of Basic Education (DBE)</li> <li>Technical schools should develop detailed yearly (operational) business plans that must demonstrate how the approved funding would be spent</li> <li>Deviations of between 2 per cent and 5 per cent from category allocations in business plans must be authorised by the Director-General: Basic Education</li> <li>Should the entire recapitalisation process not be completed within the stipulated timeframe, an additional six (6) months will be added to complete the project. If the funds are not completely utilized, it will be redirected to other schools within the same province</li> <li>Recapitalisation business plans assessed against:</li></ul>
Allocation criteria  Reason not incorporated in equitable share	<ul> <li>The distribution of funds is based on the needs identified through the outcome of the capacity audit conducted in October/November 2009</li> <li>Technical schools are not proportionally distributed across the provinces</li> </ul>
Past performance	2009/10 audited financial outcomes  • New grant
	2009/10 service delivery performance  New grant
Projected life	It is envisaged that the project will be completed in three years (2011/12-2013/14)

Technical Secondary Schools Recapitalisation Grant	
MTEF allocations	• 2011/12: R200 million, 2012/13: R210 million and 2013/14: R222 million
Payment schedule	• Four instalments (13 April 2011, 15 July 2011, 28 October 2011 and 27 January 2012)
Responsibilities of the	Responsibilities of the national department
transferring national officer	Develop and submit approved national business plans to National Treasury
and receiving officer	Evaluate, approve and submit provincial business plans to National Treasury
	Manage and support programme implementation
	Develop and distribute planning, monitoring and evaluation guidelines
	Ensure compliance with reporting requirements
	Evaluate the performance of the conditional grant and submit an evaluation report to National Treasury by 29 July 2011
	Establish and strengthen partnerships with relevant stakeholders
	Submit quarterly performance reports to National Treasury within 45 days after the end of each quarter
	Responsibilities of the provincial departments
	Develop and submit approved business plans to DBE
	Manage and implement the programme in line with the Division of Revenue Act (DoRA) and the
	Public Finance Management Act (PFMA)
	Monitor and provide support to districts/regions and schools
	Provide human resource capacity at all relevant levels
	Evaluate the performance of conditional grant and submit evaluation reports to DBE by 31 May 2011
	Establish and strengthen partnerships with relevant stakeholders
	Submit quarterly performance reports to DBE within 30 days after the end of each quarter
Process for approval of 2012/13 business plans	Provinces to submit first draft of the consolidated provincial business plans to DBE for appraisal by 27 January 2012
pasiness pasis	• The DBE team will meet to evaluate the consolidated business plans by 10 February 2012
	• The comments on the business plans will be sent to provinces for amendments by 17 February 2012
	Provinces will be required to submit the provincially approved amended business plans to DBE by 8 March 2012
	DBE will approve the final business plans by 26 March 2012
	Approved business plans by Director-General: Basic Education are submitted to the National Treasury on 2 April 2012

TD 6 1 1 4	School Infrastructure Backlogs Grant  Basic Education (Vote 15)
Transferring department	The eradication of inappropriate education structures and backlogs in basic services
Strategic goal	Eradication of inappropriate education structures and backlogs in basic services      Eradication of inappropriate school infrastructure
Grant purpose	Provision of water, sanitation and electricity to schools
Outcome statements	Improved access to enabling learning and teaching environments
Outputs	Eradication and replacement of 395 mud schools
	• 1 307 schools provided with water
	<ul> <li>536 schools provided with sanitation</li> <li>1 434 schools provided with electricity</li> </ul>
Priority outcome(s) of	Outcome 1: Improved quality of basic education
government that this grant primarily contributes to	
Details contained in the	Outcome indicators
business plan	Output indicators
	• Inputs
	<ul><li>Key activities</li><li>Procurement plan</li></ul>
	Monitoring and evaluation plan
Conditions	This is a grant-in-kind administered by the national Department of Basic Education (DBE) that may
	be transferred to a province through the Education Infrastructure Grant if the province is able to
	<ul> <li>demonstrate, through a proven track record, that it has the capacity to implement the projects</li> <li>DBE must submit a Programme Implementation Plan which includes detailed project plans,</li> </ul>
	procurement plans and procurement strategies, projected cash flow schedules and plans detailing
	the monitoring and evaluation of project implementation of projects funded through this grant to
	National Treasury for sign off before requesting the first drawings on this grant
	Provincial departments must appoint Programme Planning and Monitoring Teams (PPMT) that will assist the DBE plan and oversee the implementation of projects
	The DBE must submit monthly project cash flow reports to National Treasury 7 days after the end
	of each month that show how actual payments and cash flow reconciles with the projected cash
	flow schedule and explain any deviations from the original projected cash flow
	Provincial departments of education must report on the progress of projects funded through this
	grant in their annual report and describe how the schools have been considered in their future planning
Allocation criteria	The grant allocation is based on the distribution of inappropriate structures, schools without access
	to water, sanitation and electricity
	Final allocations will be based on the finalised project implementation plan of the DBE as approved by the National Steering Committee (NSC)
Reason not incorporated in	This is a specific purpose grant to deal with backlogs in education infrastructure and will be
equitable share	administered by the national department to achieve maximum impact in the shortest time possible
Past performance	2009/10 audited financial outcomes  • New grant
	2009/10 service delivery performance
	New grant
Projected life	Backlogs should be dealt with by 2013/14
MTEF allocations	• 2011/12: R700 million, 2012/13: R2 315 million and 2013/14: R5 189 million
Payment schedule	Payments will be made according to invoices as per the projected cash flow schedules of projects approved by NSC
Responsibilities of the	Responsibilities of the national department
transferring national officer	DBE will convene and chair meetings of the NSC during which projects assessed for funding
and receiving officer	through this grant and those that will be supplemented by the Education Infrastructure Grant to
	Provinces will be approved to ensure that the funding from the grants is directed to projects that are aligned with respective grant objectives
	The DBE must agree in writing with the provinces on projects that the DBE will administer on
	behalf of the province
	Develop a Sector Procurement Strategy and procurement strategy for this grant in terms of the practice guide prescribed by the Infrastructure Delivery Management Toolkit. The department will
	develop a procurement strategy for this grant that will lead to the quickest possible achievement of
	the grant objectives which may require the clustering of projects across provincial boundaries
	Submit approved project list with business plan to National Treasury
	Manage, monitor and support programme implementation in provinces    Transport   Tran
	Ensure compliance with reporting requirements and adherence to projected cash flow schedules

#### **School Infrastructure Backlogs Grant** Consolidate and submit quarterly reports to National Treasury and National Council of Provinces (NCOP) within 45 days after the end of each quarter Conduct site visits to selected projects on a monthly basis to assess performance Create the necessary organisational structures and build capacity within the department to oversee and monitor the implementation of the grant In consultation with National Treasury convene the NSC that will approve Project Implementation Plans. The DBE must ensure this committee meetings at least once a month is provided with sufficiently detailed reports to assess project implementation and projected cash flow schedules reconciled to the end of the month preceding the monthly meeting Responsibilities of the provincial departments Establish PPMTs that will provide support to the DBE when implementing projects funded by this Submission of business plans for 2012/13 projects by 30 September 2011 by DBE to National Process for approval of 2012/13 business plans Submission of implementation plans for 2012/13 projects by 10 February 2012 by DBE to National Treasury

## COOPERATIVE GOVERNANCE AND TRADITIONAL AFFAIRS GRANT

	Provincial Disaster Grant
Transferring department	Cooperative Governance and Traditional Affairs (Vote 3)
Strategic goal	To pro-actively respond to the immediate needs after a disaster has occurred
Grant purpose	To provide for the immediate release of funds for disaster response
Outcome statements	Immediate consequences of disasters are alleviated
Outputs	The disbursement of this grant is contingent on disasters occurring and therefore no outputs can be specified in advance
Priority outcome(s) of government that this grant this grant primarily contributes to	Outcome 12: An efficient, effective and development orientated public service and an empowered, fair and inclusive citizenship
Details contained in the business plan	This grant uses the National Disaster Management Centre (NDMC) Provincial Disaster Grant Template which will include the following:     Copy of the classification letter in terms of the Disaster Management Act     Copy of declaration of disaster in terms of the Disaster Management Act     Number of people affected     Items to be purchased     Support received from NGOs and business     Funds required for disaster response
Conditions	<ul> <li>Funds from this grant can only be used to purchase the items described in NDMC Provincial Disaster Grant Template and will only be released on request of a provincial department and verification of a declared disaster as per the Disaster Management Act</li> <li>Provinces must authorise expenditure on this grant through an adjustments budget if the expenditure occurs prior to the tabling of the provinces adjustment budget or through a finance bill following the tabling of the annual report of the province for 2011/12</li> <li>Funds can only be released after a disaster has been declared in terms of the Disaster Management Act</li> </ul>
Allocation criteria	The grant is allocated based on the location of the declared disasters and an assessment of immediate needs
Reason not incorporated in equitable share	This grant caters for responses to unforeseen disasters
Past performance	2009/10 audited financial outcomes  New grant introduced in 2011/12  2009/10 service delivery performance  New grant introduced in 2011/12
Projected life	This grant is expected to continue over the medium term but will be subject to review
MTEF allocations	• 2011/12: R305 million, 2012/13: R180 million and 2013/14: R190 million
Payment schedule	Transfers are made subject to approval by National Treasury
Responsibilities of transferring national officer and receiving officer	<ul> <li>Responsibilities of the national department</li> <li>Advise Provincial Disaster Management Centres (PDMCs) about the existence of the grant and how grant funding must be applied for</li> <li>Perform assessments of disasters and verification of applications for funding as per the requirements of the Disaster Management Act</li> <li>Seek approval from National Treasury for disbursement of funds to provinces and provide written advice on the timing of disbursements to provinces and transfer these funds to provinces within five days of drawing the funds from the National Revenue Fund</li> <li>Notify the relevant provincial treasury of a transfer at least one day before transfer and transfer the funds no later than five days after notification</li> <li>Notify the relevant PDMC of a transfer and reason for transfer within one day of the transfer of funds to provinces</li> <li>Build relationships and establish the necessary communication channels with relevant national departments to ensure the country has a coordinated disaster response approach</li> <li>Provide a performance report to National Treasury within 45 days after the end of the quarter during which funds are spent         Provide National Treasury and the relevant provincial treasury with written notification of the transfer within 14 days of a transfer of this grant     </li> <li>Responsibilities of the provincial departments</li> <li>Provide a performance report within 30 days after the end of the quarter during which funds are spent to the NDMC and relevant PDMC</li> <li>Initiate requests for disaster funding when appropriate</li> </ul>
Process for approval of 2012/13 business plans	Not applicable

# **HEALTH GRANTS**

	Comprehensive HIV and Aids Grant
Transferring department	Health (Vote 16)
Strategic goal	The implementation of the National Strategic Plan 2007 – 2011 and the National Operational Plan for Comprehensive Care, Management and Treatment of HIV and Aids and Sexually Transmitted Infections (STIs)
Grant purpose	To enable the health sector to develop an effective response to HIV and Aids including
	universal access to HIV Counselling and Testing (HCT)
	To support the implementation of the National Operational Plan for comprehensive HIV
	and Aids treatment and care
	To subsidise in-part funding for antiretroviral treatment programme
Outcome statements	Improved coordination and collaboration in the implementation of comprehensive HIV
	and Aids grant between national, provincial and local government
	<ul> <li>Improved quality of HIV and Aids services including access to:</li> </ul>
	<ul> <li>HIV Counselling and Testing (HCT)</li> </ul>
	<ul> <li>Antiretroviral Treatment (ART)</li> </ul>
	<ul> <li>Home and Community-based Care (HCBC)</li> </ul>
	<ul> <li>Prevention of TB in HIV positive and Prevention of Mother-to-Child-Transmission (PMTCT)</li> </ul>
	<ul> <li>Medical Male Circumcision (MMC)</li> </ul>
	Improved health workers' capacity at the three levels of care to ensure quality service
	delivery to South Africans
	Reduced HIV incidence and prevalence
Outputs	• Number of public health facilities offering ART services, and number of patients initiated
	on ART
	HCBC beneficiaries; active caregivers receiving stipends
	<ul> <li>Male and female condoms distributed among high risk population and at high</li> </ul>
	transmission sites
	Pregnant mothers tested for HIV
	PMTCT babies tested for PCR
	Nevirapine dose to baby coverage
	PMTCT dual therapy coverage
	TB and HIV support:
	<ul> <li>HIV positive client screened for TB</li> </ul>
	<ul> <li>HIV positive eligible TB sufferers started on IPT (INH Preventive Therapy)</li> </ul>
	Government health facilities offering HCT services
	Active lay counsellors on stipends
	People offered counselling and testing for HIV
	<ul> <li>Health facilities offering MMC services and MMC conducted and adverse events</li> </ul>
	reported
Priority outcome(s) of	Outcome 2: A long and healthy life for all South Africans
government that this	
grant primarily	
contributes to	
Details contained in the	Outcome indicators
business plan	Output indicators
	• Inputs
	Key activities
	Risk management plans
Conditions	• The following priority areas must be supported through the grant: 1. ART related
	interventions; 2. Home Community Based Care (HCBC); 3. Condom distribution and
	related interventions; 4. PMTCT; 5. Programme Management Strengthening (PM); 6.
	Regional Training and Quality Assurance Centres (RTC); 7. TB/HIV and SDC; 8 HCT;
	9. MMC
	• Provinces must submit monthly financial reports (IYM) and the monthly break-down
	report per sub-programme to the National Department of Health by the 15th of the
	following month using standard formats as determined by the national department. An
	electronic version and/or a faxed hard copy signed by the provincial grant receiving
	manager, chief financial officer and the Head of Department must be submitted
Allogation switswis	Provinces must indicate all sources of funding for the programme in their business plans     Appropriate LIV/ providences estimated share of Aids pages populations next demonstrates.
Allocation criteria	Antenatal HIV prevalence, estimated share of Aids cases, populations post-demarcation

	Comprehensive HIV and Aids Grant
Reason not incorporated	HIV and Aids is a high national priority and there is therefore a need for a coordinated
in equitable share	response for the country as a whole
Past performance	2009/10 audited financial performance
	Allocated and transferred R4 376 million to provinces
	Of the total available R4 378 million (including provincial roll overs), R4 310 million
	(98.4 per cent) was spent
	2009/10 service delivery performance
	8 652 Lay counsellors trained and providing services at service points
	• 4 624 facilities were providing HCT services
	• 5 069 people received counselling and 86 per cent (4 349) were tested for HIV,
	1 192 pregnant women were tested for HIV
	• 1 121 patients who had access to HCBC services by the end of March 2010
	495 accredited ART service points in operation
	• 1 007 patients were on ART
	• 454 intervention high transmission sites in operation
	• 73 per cent of new-born babies received Nevirapine
	95 per cent of PHC facilities offer PMTCT services
Projected life	Ongoing in line with National Strategic Plan on HIV and Aids
MTEF allocations	• 2011/12: R7 493 million, 2012/13: R8 825 million and 2013/14: R10 607 million
Payment schedule	Monthly instalments based on the approved payment schedule
Responsibilities of the	Responsibilities of the national department
transferring national	Visit provinces to monitor implementation and provide support
officer and receiving	Submit quarterly performance reports to National Treasury within 45 days after the end
officer	of the quarter
	Improve demand planning for the treatment programme
	Meet with National Treasury to review performance of the grant
	Responsibilities of the provincial departments
	• Quarterly performance output reports to be submitted within 30 days following the
	reporting period using standard formats as determined by the national department.
	Submit an electronic version and/or faxed hard copy signed by the provincial grant
	receiving manager and the chief financial officer
	Clearly indicate measurable objectives and performance targets as agreed with the
	national department in provincial departmental plans for 2012/13 and over the MTEF
	Submit risk management plans with final business plans
Process for approval of	National and Provincial Departments of Health to sign and certify, provincial business
2012/13 business plans	plans by 28 February 2012

	Forensic Pathology Services Grant
Transferring department	Health (Vote 16)
Strategic goal	To ensure provision of impartial professional forensic evidence for the criminal justice system concerning death due to unnatural causes
Grant purpose	To continue the development and provision of adequate forensic pathology services in all provinces
Outcome statements	Comprehensive Forensic Pathology Service (FPS) in all provinces
Outputs	New forensic pathology facilities built, refurbished and equipped
	Human resource organograms filled with qualified personnel
Priority outcome(s) of	<ul> <li>Operational standards for mortuaries published and implemented</li> <li>Outcome 2: A long and healthy life for all South Africans</li> </ul>
government that this grant primarily contributes to	Cutcome 2.11 long and neutring into 101 and bodies 1 interests
Details contained in the business plan	<ul><li>Outcome indicators</li><li>Output indicators</li></ul>
business plan	Inputs
	Key activities
Conditions	Grant administration should be coordinated and monitored by the Programme Manager in consultation with provincial Grant Managers
	<ul> <li>Provincial Health Departments should ensure a complete ring-fencing of the grant to facilitate an</li> </ul>
	expeditious procurement of goods and services and filling of vacant posts.
	Provincial Health Departments should provide copies of a Business Case for every mortuary under
	<ul> <li>construction</li> <li>The Chief Financial Officer/Delegate and Public Works delegate/s need to be present at all provincial</li> </ul>
	visit meetings
Allocation criteria	In accordance with the National Project Plan, as modified for demarcation and inflation
Reason not incorporated in	• The grant funds the transfer of the service from the South African Police Services to the national
the equitable share	Department of Health (DoH) to establish an integrated Forensic Pathology Service. It will be phased into the provincial equitable share in 2012/13
Past performance	2009/10 audited financial outcomes
T dist periorimuno	Allocated and transferred R502 million to provinces
	• Of the total available R527 million (including provincial roll overs), R642 million (122 per cent) was
	spent 2009/10 service delivery performance
	Mortuaries completed: 13
	Holding Facilities completed: 9
	<ul> <li>Vehicles procured: 46</li> <li>Computers and Laptops procured: 89</li> </ul>
	Employees recruited:
	Pathologists: 4
	Forensic Officers: 48
Projected life	Support Staff: 17  • The grant ends in 2011/12 when it is phased into the provincial equitable share
	2011/12: R590 million
MTEF allocations	Monthly instalments
Payment schedule	Responsibilities of the national department
Responsibilities of the transferring national	<ul> <li>Monitor the provision of the service by visiting provinces once per year or as the needs require</li> </ul>
officer and receiving officer	Coordinating the bi-annual meetings with the National Forensic Pathology Services Committee
, , , , , , , , , , , , , , , , , , ,	• Ensure that the FPS Code of Guidelines Manual is used at all mortuaries as a procedure manual and for
	<ul> <li>developing Standard Operating Procedures customised for the provinces</li> <li>Submit quarterly performance reports to National Treasury 45 days after the end of each quarter</li> </ul>
	Submit annual evaluation reports to National Treasury by 31 July 2011
	Complete the designated mortuary building and upgrading programme
	Convene meetings with CFOs, HODs and all other stakeholders during provincial visits to strengthen the capacity of the grant administration and monitoring mechanisms
	Responsibilities of the provincial departments
	Submit the annual evaluation report for the grant to DoH by 31 May 2011
	Report any deviation from the prescripts of service    Applies of the prescript of the baldway to Dell's his ground prescript visits.
	<ul> <li>Invite other provincial stakeholders to DoH's bi-annual provincial visits</li> <li>Submit quarterly performance reports within 30 days after the end of each quarter</li> </ul>
Process for approval of	Not applicable as the grant will be phased into the provincial equitable share
2012/13 business plans	
	1

	Health Infrastructure Grant
Transferring department	Health (Vote 16)
Strategic goal	Provinces plan, manage, maintain and transform health infrastructure in line with national policy objectives
Grant purpose	To supplement provincial funding of health infrastructure to accelerate the provision of health facilities and ensure proper maintenance of provincial health infrastructure
Outcome statements	Improved accessibility to and quality of health services
	<ul> <li>Quality, modernised and well maintained health infrastructure</li> <li>Improved organisational capacity in provinces for infrastructure asset management planning and implementation</li> </ul>
Outputs	Number of health facilities, planned, designed, constructed, maintained and operationalized
Priority outcome(s) of	Outcome 2: A long and healthy life for all South Africans
government that this grant primarily contributes to	Sates and 2011 19 in the analysis of the sates and sates
Details contained in the business plan	This grant uses a User Asset Management Plan (U-AMP), which includes the following information: immovable assets requirements acquisition plan refurbishment plan repairs and maintenance
Conditions	<ul> <li>Provincial Departments of Health must submit tabled U-AMPs (formerly Infrastructure Plans) with prioritised project lists that are signed-off by HOD for the 2011 MTEF by 14 April 2011 to the national Department of Health (DoH). Tabled U-AMPs must include the projects listed in the approved Infrastructure Programme Implementation Plans (IPIPs) for 2011/12 and sector procurement strategy as per guideline in the Infrastructure Delivery Management System (IDMS) Toolkit. The flow of the first instalment of this grant is dependent upon receipt by the DoH of this project list and satisfaction that the list meets required detailed and alignment with infrastructure planning information submitted in 2010/11 or earlier. These plans must also be submitted to the relevant provincial treasury</li> <li>The flow of the third quarter transfers is dependent upon receipt by DoH of the draft 2012/13 U-AMPs including the initial list of prioritised projects as captured in the Infrastructure Programme Management Plan (IPMP). Changes to the MTEF prioritised list of projects must be approved in conjunction with DoH</li> <li>Provinces must adhere to the Infrastructure Alignment Model in terms of programme implementation and in year reporting. Quarterly performance reports must be submitted to the National Council of Provinces within 45 days after the end of each quarter</li> <li>The provinces must put in place a Provincial Progress Review Committee as approved by the National Health Council for monitoring and oversight for both the Health Infrastructure Grant and the Hospital Revitalisation Grant</li> <li>Provinces must submit a list of projects in the required format to DoH within 14 days after the 2011 Division of Revenue Act comes into effect. Provinces must also submit quarterly performance reports, drawn from their infrastructure reporting model to the National Council of Provinces within 45 days after the end of each quarter. Failure to comply with these conditions may result in the withholding of fun</li></ul>
Allocation criteria	Allocations are based on the formula that was used for the Infrastructure Grant to Provinces
Reason not incorporated in equitable share	Infrastructure funding is protected through conditional grants to ensure that national priorities are addressed and best planning practices are complied with
Past performance	2009/10 audited financial outcomes  New grant
	2009/10 service delivery performance
Projected life	New grant     This grant will be merged with the Hospital Revitalisation Grant to create a consolidated source of health infrastructure funding once the necessary conditions are in place and is expected to continue past the end of the 2011 MTEF due to the scale of need for ongoing infrastructure spending in health and oversight of this spending
MTEF allocations	• 2011/12: R1 702 million, 2012/13: R1 821 million and 2013/14: R1 921 million
Payment schedule	Instalments as per approved payment schedule
Responsibilities of	Responsibilities of the national department
transferring national officer and receiving officer	<ul> <li>Provide guidelines and monitor the development and approval of provincial U-AMPs including project list and procurement strategies</li> <li>Perform on-site monitoring of the implementation of projects and approved adherence to the project list and procurement strategy</li> <li>Support and oversee provincial progress review meetings in all provinces</li> </ul>
	<ul> <li>Convene and chair quarterly National Progress Review meetings</li> <li>Collect project lists from provinces 14 days after the 2011 Division of Revenue Act comes into effect and publish the project list in a gazette within 28 days after the 2011 Division of Revenue Act comes into effect</li> <li>Submit quarterly performance reports to National Treasury and National Council of Provinces within 45 days after the end of each quarter</li> </ul>

Health Infrastructure Grant	
	Responsibilities of the provincial departments
	• Ensure that the provincial Infrastructure Projects Progress Review committee sit every month to oversee, monitor and report on the full physical and financial development cycle of all health infrastructure projects implemented by the province that comply with Infrastructure Delivery Improvement Programme (IDIP) guidelines
	<ul> <li>Submit quarterly performance reports, drawn from the infrastructure reporting model, to the DoH and National Council of Provinces within 30 days after the end of each quarter</li> <li>Ensure that all health infrastructure projects are aligned and lead to best practice infrastructure planning and</li> </ul>
	<ul> <li>project implementation and effective and efficient management and sustainability of the province's health assets</li> <li>Submit a list of projects in the required format to DoH within 14 days after the 2011 Division of Revenue Act comes into effect</li> </ul>
Process for approval of 2012/13 business plans	Provinces must submit IPIPs and U-AMPs signed off by the head of provincial department of health with project lists for 2012/13 to DoH by the end of February 2012 for approval

	Health Professions Training and Development Grant
Transferring department	Health (Vote 16)
Strategic goal	To contribute to the appropriate and adequate training of health professions nationally through the
Strategic goal	provision of quality education and development in designated health facilities in South Africa
Grant purpose	Support provinces to fund service costs associated with training of health professionals
• •	• Development and recruitment of medical specialists in under-served provinces (Eastern Cape, Limpopo,
	Mpumalanga, Northern Cape and North West)
	Support and strengthen undergraduate and postgraduate training processes in health facilities  Of the last of
Outcome statements	Medical specialists developed in targeted provinces (Mpumalanga, Limpopo, Eastern Cape, North West and Northern Cape)
	<ul> <li>Strengthened undergraduate and post graduate training processes in designated health facilities</li> </ul>
Outputs	Number and composition of health sciences students by province and training institution
Outputs	(Under and Post graduate students, Registrars and Specialists)
	Number of students per discipline and per training institution
	Number of students supported on the service platform
	Number of supervisors supporting students on the service platform
Priority outcome(s) of	Outcome 2: A long and healthy life for all South Africans
government that this grant primarily contributes to	
	Outcome indicators
Details contained in the	<ul><li>Outcome indicators</li><li>Output indicators</li></ul>
business plan	• Inputs
	Key activities
Conditions	Business plans must be submitted in the approved format by 28 February 2011
	• The training platform and re-sourcing thereof need to be developed after consultation with the appropriate
	Health Science Institutions
Allocation criteria	Allocation is based on the distribution of medical students
Reason not incorporated in equitable share	Expansion and shifting of location of training activities requires national coordination
	2009/10 audited financial outcomes
Past performance	Allocated and transferred R1 760 million to provinces. Total available amounts to R1 780 million, which
	includes provincial rollovers
	2009/10 service delivery performance
	Provincial achievements in training and development by discipline:
	medical students and professional nurse students: 20 157
	registrars: 791
	specialists: 260
Projected life	registrars/specialists involved in outreach services: 185
Projected life	registrars/specialists involved in outreach services: 185  • A review of the grant started during 2010/11 and will continue in 2011/12
MTEF allocations	registrars/specialists involved in outreach services: 185  • A review of the grant started during 2010/11 and will continue in 2011/12  • 2011/12: R1 997 million, 2012/13: R2 076 million and 2013/14: R2 190 million
MTEF allocations Payment schedule	registrars/specialists involved in outreach services: 185  • A review of the grant started during 2010/11 and will continue in 2011/12  • 2011/12: R1 997 million, 2012/13: R2 076 million and 2013/14: R2 190 million  • Monthly instalments
MTEF allocations Payment schedule Responsibilities of the	registrars/specialists involved in outreach services: 185  • A review of the grant started during 2010/11 and will continue in 2011/12  • 2011/12: R1 997 million, 2012/13: R2 076 million and 2013/14: R2 190 million  • Monthly instalments  Responsibilities of the national department
MTEF allocations Payment schedule Responsibilities of the transferring national officer	registrars/specialists involved in outreach services: 185  • A review of the grant started during 2010/11 and will continue in 2011/12  • 2011/12: R1 997 million, 2012/13: R2 076 million and 2013/14: R2 190 million  • Monthly instalments  Responsibilities of the national department  • Provide the guidelines and criteria for the development and approval of business plans
MTEF allocations Payment schedule Responsibilities of the	registrars/specialists involved in outreach services: 185  • A review of the grant started during 2010/11 and will continue in 2011/12  • 2011/12: R1 997 million, 2012/13: R2 076 million and 2013/14: R2 190 million  • Monthly instalments  Responsibilities of the national department  • Provide the guidelines and criteria for the development and approval of business plans  • Monitor implementation and provide support to provincial departments
MTEF allocations Payment schedule Responsibilities of the transferring national officer	registrars/specialists involved in outreach services: 185  • A review of the grant started during 2010/11 and will continue in 2011/12  • 2011/12: R1 997 million, 2012/13: R2 076 million and 2013/14: R2 190 million  • Monthly instalments  Responsibilities of the national department  • Provide the guidelines and criteria for the development and approval of business plans  • Monitor implementation and provide support to provincial departments  • Annual support visits to the provinces
MTEF allocations Payment schedule Responsibilities of the transferring national officer	registrars/specialists involved in outreach services: 185  • A review of the grant started during 2010/11 and will continue in 2011/12  • 2011/12: R1 997 million, 2012/13: R2 076 million and 2013/14: R2 190 million  • Monthly instalments  Responsibilities of the national department  • Provide the guidelines and criteria for the development and approval of business plans  • Monitor implementation and provide support to provincial departments  • Annual support visits to the provinces  • Submit quarterly performance reports to National Treasury within 45 days after the end of each quarter  • Submit an annual evaluation report on the grant performance to the National Treasury by 30 July 2011
MTEF allocations Payment schedule Responsibilities of the transferring national officer	registrars/specialists involved in outreach services: 185  • A review of the grant started during 2010/11 and will continue in 2011/12  • 2011/12: R1 997 million, 2012/13: R2 076 million and 2013/14: R2 190 million  • Monthly instalments  Responsibilities of the national department  • Provide the guidelines and criteria for the development and approval of business plans  • Monitor implementation and provide support to provincial departments  • Annual support visits to the provinces  • Submit quarterly performance reports to National Treasury within 45 days after the end of each quarter  • Submit an annual evaluation report on the grant performance to the National Treasury by 30 July 2011  Responsibilities of the provincial departments
MTEF allocations Payment schedule Responsibilities of the transferring national officer	registrars/specialists involved in outreach services: 185  • A review of the grant started during 2010/11 and will continue in 2011/12  • 2011/12: R1 997 million, 2012/13: R2 076 million and 2013/14: R2 190 million  • Monthly instalments  Responsibilities of the national department  • Provide the guidelines and criteria for the development and approval of business plans  • Monitor implementation and provide support to provincial departments  • Annual support visits to the provinces  • Submit quarterly performance reports to National Treasury within 45 days after the end of each quarter  • Submit an annual evaluation report on the grant performance to the National Treasury by 30 July 2011  Responsibilities of the provincial departments  • Submit an annual evaluation report on the grant performance to the national Department of Health (DoH)
MTEF allocations Payment schedule Responsibilities of the transferring national officer	registrars/specialists involved in outreach services: 185  • A review of the grant started during 2010/11 and will continue in 2011/12  • 2011/12: R1 997 million, 2012/13: R2 076 million and 2013/14: R2 190 million  • Monthly instalments  Responsibilities of the national department  • Provide the guidelines and criteria for the development and approval of business plans  • Monitor implementation and provide support to provincial departments  • Annual support visits to the provinces  • Submit quarterly performance reports to National Treasury within 45 days after the end of each quarter  • Submit an annual evaluation report on the grant performance to the National Treasury by 30 July 2011  Responsibilities of the provincial departments  • Submit an annual evaluation report on the grant performance to the national Department of Health (DoH) by 31 May 2011
MTEF allocations Payment schedule Responsibilities of the transferring national officer	registrars/specialists involved in outreach services: 185  • A review of the grant started during 2010/11 and will continue in 2011/12  • 2011/12: R1 997 million, 2012/13: R2 076 million and 2013/14: R2 190 million  • Monthly instalments  Responsibilities of the national department  • Provide the guidelines and criteria for the development and approval of business plans  • Monitor implementation and provide support to provincial departments  • Annual support visits to the provinces  • Submit quarterly performance reports to National Treasury within 45 days after the end of each quarter  • Submit an annual evaluation report on the grant performance to the National Treasury by 30 July 2011  Responsibilities of the provincial departments  • Submit an annual evaluation report on the grant performance to the national Department of Health (DoH) by 31 May 2011
MTEF allocations Payment schedule Responsibilities of the transferring national officer	registrars/specialists involved in outreach services: 185  • A review of the grant started during 2010/11 and will continue in 2011/12  • 2011/12: R1 997 million, 2012/13: R2 076 million and 2013/14: R2 190 million  • Monthly instalments  Responsibilities of the national department  • Provide the guidelines and criteria for the development and approval of business plans  • Monitor implementation and provide support to provincial departments  • Annual support visits to the provinces  • Submit quarterly performance reports to National Treasury within 45 days after the end of each quarter  • Submit an annual evaluation report on the grant performance to the National Treasury by 30 July 2011  Responsibilities of the provincial departments  • Submit an annual evaluation report on the grant performance to the national Department of Health (DoH) by 31 May 2011  • Quarterly reporting by provinces on the number of students enrolled by discipline, level and training
MTEF allocations  Payment schedule  Responsibilities of the transferring national officer and receiving officer	registrars/specialists involved in outreach services: 185  • A review of the grant started during 2010/11 and will continue in 2011/12  • 2011/12: R1 997 million, 2012/13: R2 076 million and 2013/14: R2 190 million  • Monthly instalments  Responsibilities of the national department  • Provide the guidelines and criteria for the development and approval of business plans  • Monitor implementation and provide support to provincial departments  • Annual support visits to the provinces  • Submit quarterly performance reports to National Treasury within 45 days after the end of each quarter  • Submit an annual evaluation report on the grant performance to the National Treasury by 30 July 2011  Responsibilities of the provincial departments  • Submit an annual evaluation report on the grant performance to the national Department of Health (DoH) by 31 May 2011  • Quarterly reporting by provinces on the number of students enrolled by discipline, level and training institution using the prescribed format must be submitted to DoH within 30 days after the end of each quarter  • Submit monthly financial reports to DoH
MTEF allocations Payment schedule Responsibilities of the transferring national officer	registrars/specialists involved in outreach services: 185  • A review of the grant started during 2010/11 and will continue in 2011/12  • 2011/12: R1 997 million, 2012/13: R2 076 million and 2013/14: R2 190 million  • Monthly instalments  Responsibilities of the national department  • Provide the guidelines and criteria for the development and approval of business plans  • Monitor implementation and provide support to provincial departments  • Annual support visits to the provinces  • Submit quarterly performance reports to National Treasury within 45 days after the end of each quarter  • Submit an annual evaluation report on the grant performance to the National Treasury by 30 July 2011  Responsibilities of the provincial departments  • Submit an annual evaluation report on the grant performance to the national Department of Health (DoH) by 31 May 2011  • Quarterly reporting by provinces on the number of students enrolled by discipline, level and training institution using the prescribed format must be submitted to DoH within 30 days after the end of each quarter

	Hospital Revitalisation Grant
Transferring department	Health (Vote 16)
Strategic goal	To enable provinces to plan, manage, modernise, rationalise and transform the health infrastructure in line with national policy objectives
Grant purpose	<ul> <li>To provide funding to enable provinces to plan, manage, modernise, rationalise and transform health infrastructure, health technology, monitoring and evaluation of the health facilities in line with national policy objectives</li> <li>Supplement expenditure on health infrastructure delivered through public-private partnerships</li> </ul>
Outcome statements	<ul> <li>Improved accessibility and service delivery in health</li> <li>Provision of quality and modernised health infrastructure</li> </ul>
Outputs	Number of hospitals funded to upgrade, rebuilt and fully commissioned as per approved 2011/12 Project Implementation Plans (PIP)
Priority outcome(s) of government that this grant primarily contributes to	Outcome 2: A long and healthy life for all South Africans
Details contained in the business plan	This grant uses a PIP, which includes the following:     outcome indicators     output indicators     inputs     key activities
Conditions	Provinces must implement projects in line with the Project Implementation Plans, as guided by the Project Implementation Manual (PIM) for 2011/12 that has been be approved by the national Department of Health (DoH) for the 2011/12 financial year
	<ul> <li>With the exception of funding for costs incurred for planning, all projects commencing construction in 2011/12 must have business cases and Initial Project Implementation Plans (IPIPs) approved before funds can be released for such projects</li> <li>All the new projects should follow Peer Review stages as per prescribed in Reporting Format incorporated in the Project Implementation Manual (PIM) 2011/12</li> <li>Provinces must submit draft 2012/13 U-AMPs including the initial list of prioritised projects as captured in the Infrastructure Programme Management Plan (IPMP) to DoH by 30 September 2011. The monthly instalment following this deadline is dependent on receipt of this information</li> <li>Monthly reports for all projects funded in 2011/12 financial year in this grant must be submitted through the prescribed comprehensive reporting as prescribed by 2011/12 PIM and Infrastructure Reporting Model (IRM). This does not replace the reporting in the In-Year Monitoring (IYM) expenditure reports</li> <li>Provincial health departments must strengthen grant management by appointing a complete Revitalisation Team as guided by DoH and ensure the core infrastructure management structure approved by National Health Council (NHC) for such grants is in place</li> <li>Provinces must put in place Provincial Progress Review Committees as approved by the NHC for monitoring and oversight of the Health Infrastructure Grant and the Hospital Revitalisation Grant</li> <li>Provinces may not award a tender to commence construction over the approved MTEF</li> <li>Provinces may not use this grant on projects that were not part of approved 2011/12 PIP unless written approval to such effect is granted by DoH</li> </ul>
Allocation criteria	Allocations based on provincial projected cash flow figures per project as per prescribed format per contract, which include budgets on infrastructure, health technology, organisational development and quality improvement
Reason not incorporated in equitable share	This grant funds large projects requiring substantial capital investment. Their size, complexity and national strategic importance is suited to dedicated and closely managed funding
Past performance	2009/10 audited financial performance  Allocated R3 370 million and transferred R2 989 million to provinces  Of the total available R3 495 million (including provincial roll overs), R2 550 million (73 per cent) was spent  2009/10 service delivery performance  Five hospitals were completed
Projected life	This grant will be merged with the Health Infrastructure Grant to create a consolidated source of health infrastructure funding once the necessary conditions are in place and is expected to continue past the end of the 2011 MTEF due to the scale of need for ongoing infrastructure spending in health and oversight of this spending
MTEF allocations	• 2011/12: R4 136 million, 2012/13: R4 336 million and 2013/14: R4 068 million
Payments schedule	Monthly instalments as per approved payment schedule

#### **Hospital Revitalisation Grant** Responsibilities of the national department Responsibilities of Provide guidelines and monitor the development and approval of provincial infrastructure plan, business transferring national case, project brief, design and Initial Project Implementation Plan officer and receiving Conducting peer reviews per project at stages highlighted in the prescribed Comprehensive Progress Report officer format as prescribed in the 2011/12/ PIM Perform on-site monitoring of the implementation of projects Support and oversee provincial progress review meetings in all provinces Convene and chair quarterly National Progress Review meetings Approve all plans funded under the grant before the first transfer Submit quarterly performance reports to National Treasury and National Council of Provinces (NCOP) within 45 days after the end of each quarter Collect project lists from provinces 14 days after the 2011 Division of Revenue Act (DoRA) comes into effect and publish the project list in a gazette within 28 days after the 2011 DoRA comes into effect Responsibilities of the provincial departments Convene Infrastructure Projects Progress Review committees to oversee, monitor and report on the full physical and financial development cycle of all health infrastructure projects Report as per prescribed comprehensive reporting format required within the 2011/12 PIM and Infrastructure Development Improvement Programme (IDIP) guidelines Provincial departments must participate in the national Progress Review committee meetings and report on project implementation of all health projects Submit monthly financial reports to DoH within 15 days after the end of each month Submit quarterly performance reports to DoH and NCOP within 30 days after the end of each quarter Convene and chair the Provincial Progress Review Meetings Ensure that all health infrastructure projects are aligned and lead to best practice infrastructure planning and project implementation and effective and efficient management and sustainability of the province's health assets Provinces submit their draft PIP to DoH covering all four components of the Hospital Revitalisation Grant by Process for approval of 2012/13 business plans 28 February 2011 Business cases and Initial Project Implementation Plan for projects potentially starting construction in 2012/13 will be approved by DoH and communicated to provinces by 30 June 2011 Provinces to submit Project Cost Template per Contract together with the PIP covering all components of Provinces final signed 2012/13 annual PIPs by 28 February 2012 to DoH

Transferring department	National Tertiary Services Grant  • Health (Vote 16)
Strategic goal	• To enable provinces to plan, modernise, rationalise and transform the tertiary hospital service delivery
~	platform in line with national policy objectives including improving access and equity
Grant purpose	<ul> <li>To compensate tertiary facilities for the additional costs associated with spill over effects</li> <li>Ensure adequate provision of tertiary health services for all South African citizens</li> </ul>
Outcome statements	Provision of modernised and transformed tertiary services that allows for improved access and equity
Outputs	Provision of designated national tertiary service levels in 26 hospitals/complexes as agreed between the
7.1.1	Province and the national Department of Health (DoH)
Priority outcome(s) of government that this grant	Outcome 2: A long and healthy life for all South Africans
primarily contributes to	
Details contained in the	This grant uses the Service Level Agreement (SLA) which contains the following:
business plan	definitions of tertiary services funded by the grant
	designated services funded by the National Tertiary Services Grant (NTSG), by facility and by province
	summary of inpatient separations, inpatient days, day patient separations, outpatient first visits,
	outpatient follow up visits
	monitoring and reporting
	validation and revision of data deviations or changes to tertiary services
	referral responsibilities
Conditions	Completion of SLA and business plan in the prescribed format signed by each provincial department or
	receiving officer, and the transferring officer by 28 February 2011
	<ul> <li>The grant must contribute a maximum of 85 per cent of each tertiary hospital/complex budget</li> <li>Provinces must gazette allocations to individual hospitals/complexes as per SLA and be hospital/complex</li> </ul>
	• Provinces must gazette allocations to individual hospitals/complexes as per SLA and be hospital/complex specific
	<ul> <li>Institutions receiving the grant must report on expenditure and patient activity monthly to the provincial</li> </ul>
	department
	• Provinces must maintain a separate budget for each of the 26 hospitals/complexes
	• Province must communicate in writing to each benefiting hospital/complex the allocation made, with the agreed upon service specifications which must be rendered. For monitoring purposes this information
	must be supplied to the DoH by 29 April 2011
Allocation criteria	Distribution of cost of designated tertiary services as determined by the ongoing reviews of patient
	<ul><li>activity</li><li>Approved plans for the modernisation of tertiary services</li></ul>
Reason not incorporated in	<ul> <li>Approved plans for the modernisation of tertiary services</li> <li>There are significant cross boundary flows and spill effects associated with tertiary services that are not</li> </ul>
equitable share	affected by provincial boundaries due to their specialised nature. Tertiary hospitals are a national asset
	requiring collective agreement and management
Past performance	2009/10 audited financial outcomes  Allocated and transferred R6 614 million to provinces, total available amounts to R6 708 million
	(including provincial roll overs)
	2009/10 service delivery performance
	Provincial tertiary services performance was measured against the Service Level Agreements and the
	total patient activity rendered is as follows:  551 436 inpatient separations
	3 087 857 inpatient days
	2 216 183 day patient separations
	868 614 outpatient first visits
Projected life	<ul> <li>1 555 207 outpatient follow up visits</li> <li>Support for tertiary services will continue because of the cross boundary flows and spill over effects. The</li> </ul>
1 rojecteu me	• Support for tertiary services will continue because of the cross boundary flows and spill over effects. The planning of the service configuration and the basis for the calculation of the grant will be constantly
	reviewed
MTEF allocations	• 2011/12: R8 049 million, 2012/13: R8 689 million and 2013/14: R9 389 million
Payment schedule	Monthly instalments as per approved payment schedule    Degraphibilities of the notional deposits and the notional de
Responsibilities of the transferring national officer	Responsibilities of the national department  Revise and assess framework for SLAs prior to transfer of funds
and receiving officer	<ul> <li>Monitor expenditure and patient activity and provide on-site support to facilities and provinces</li> </ul>
-	Conduct support visits to the provinces and hospitals/complexes
	Submit quarterly performance reports to National Treasury within 45 days after the end of each quarter
	Responsibilities of the provincial departments  Provide detail (by cotogony level and number) and costs of personnel applicad in the hearitals and
	<ul> <li>Provide detail (by category, level and number) and costs of personnel employed in the hospitals and complexes</li> </ul>
	<ul> <li>Provide detailed expenditure on major cost drivers in goods and services</li> </ul>
	Provide detailed break downs of capital expenditure
Process for approval of	• Completion of SLA and an approved business plan, in the prescribed format, signed by each receiving
2012/13 business plans	officer, and the transferring officer by 28 February 2012

# HIGHER EDUCATION AND TRAINING GRANT

	Further Education and Training Colleges Grant
Transferring department	Higher Education and Training (Vote 17)
Strategic Goal	The successful transfer of the further education and training (FET) colleges function to the national Department of Higher Education and Training (DHET)
Grant purpose	To ensure the successful transfer of the FET College function to the Department of Higher Education and Training
Outcome statements	FET Colleges offer approved programmes in support of Skills Development according to the Funding Norms for FET Colleges
Outputs	<ul> <li>Enrolment of National Curriculum (Vocational) Programmes as set out in college enrolment target planning</li> <li>Enrolments in Report 191 Programmes in line with the Report 191</li> <li>Expanding ICT for teaching and learning utilising connectivity norms</li> <li>Continue Implementing MIS systems for the delivery of transversal MIS services</li> <li>Implementation of the Funding Norms for FET Colleges</li> <li>Refurbishment, maintenance and repairs of infrastructure and equipment to support the delivery of approved programmes</li> <li>New Infrastructure</li> </ul>
Priority outcome(s) of government that this grant primarily contributes to	Outcome 5: A skilled and capable workforce to support an inclusive growth path
Details contained in the business plan	<ul> <li>Provincial Implementation Protocol signed by the Minister and the Members of Executive Committee (MECs) of the Provincial Department of Education</li> <li>Provincial Implementation Protocols signed by the Director-General and the Heads of Provincial Education Departments (PEDs)</li> </ul>
Conditions	<ul> <li>The funding of each of the following outputs will depend on the priority set for each college within available funding</li> <li>The national enrolment plan linked to funding norms will be used as a guideline for allocating the grant to each college. Any upward deviation from these enrolments must be funded by the college or entity causing such deviation</li> <li>On receipt of a conditional grant transfer for a college, the PEDs must transfer the relevant portion of the funds to the respective colleges within 14 days of receipt thereof. The date and proof of transfer to a college must be confirmed with the DHET within 2 days thereafter</li> <li>A portion of the grant to be withheld per province, calculated on the basis of 6 per cent of the total compensation of employees pending the finalisation of the labour negotiations in the Education Labour Relations Council (ELRC) and General Public Service Sector Bargaining Council (GPSSBC)</li> </ul>
Allocation criteria	<ul> <li>The basis for the allocation is the allocation for the FET Colleges as per programme 5 in PDEs as set out in the 2009/10 MTEF</li> <li>Allocations to each FET College must be informed by the funding norms for FET Colleges</li> </ul>
Reason not incorporated in equitable share	This grant protects funding for FET colleges while legislative processes required to shift the FET College function to the national DHET is completed
Past performance	2009/10 Audited financial outcomes New grant 2009/10 Service delivery performance New grant
Projected life	The projected life will be determined by the legislative process that needs to take place to give effect to the shifting of the FET College function to the Department of Higher Education and Training.
MTEF allocations	• 2011/12: R4 326 million, 2012/13: R4 705 million and 2013/14: R5 262 million
Payment schedule	Eleven instalments
Responsibilities of the transferring national officer and receiving officer	<ul> <li>Responsibilities of the national department</li> <li>Inform and provide proof of transfer of grants to PEDs</li> <li>Convene an Inter-provincial committee to oversee the implementation of the grant</li> <li>Provide framework for the development of College Operational and Strategic Plans</li> <li>Monitor the grant according to approved College Operational Plans and budgets</li> <li>Consolidate and submit Quarterly Performance Reports to National Treasury within 45 days after the end of the quarter being reported on</li> <li>To monitor the utilisation of the grant against the set outcomes and to take appropriate action if cases of non-compliance are discovered</li> <li>Calculate the programme based funding per college based on the Funding Norms for FET Colleges each year and recommend the transfer of this to the relevant FET College</li> <li>Establish provincial level institutional support to FET Colleges</li> <li>Evaluate performance of the conditional grant and submit an evaluation report to National Treasury</li> </ul>

Further Education and Training Colleges Grant	
	Responsibilities of the provincial departments
	Transfer grant allocations to colleges within 14 days after transfer of funds by Provincial Treasury and confirm within 2 days of transfer:  - whether or not the transfer was received  - whether the grants were transferred to colleges  Support the process of giving effect to the Provincial Implementation Protocol with the DHET
	Ensure provincial officials who are currently supporting FET College functions to continue such support
	Comply with reporting requirements as may be set from time to time
	Evaluate performance of the conditional grant and submit an evaluation report to DHET annually by 31 July
Process for approval of 2012/13 business plans	The currently signed Implementation Protocols between the Minister and MECs, as well as the DG and Heads of Education Departments will remain in effect until such time that amendments to the protocols are made.

# **HUMAN SETTLEMENTS GRANT**

	Human Settlements Development Grant
Transferring department	Human Settlements (Vote 31)
Strategic goal	The creation of sustainable human settlements that enables an improved quality of household life
Grant purpose	To provide funding for the creation of sustainable human settlements
Outcome statements	The facilitation and provision of basic infrastructure, top structures and basic social and economic amenities that contribute to the creation of sustainable human settlements
Outputs	Financial interventions and measures that improve access to human settlement development and the
	property market  Number of informal settlement households upgraded
	Number of social and rental housing units developed
	Hectares of well located land and property acquired and developed
	<ul> <li>Number of Rural Housing units developed</li> <li>Number of serviced sites developed and provided</li> </ul>
Details contained in the	Number of serviced sites developed and provided     Outcome indicators
business plan	Outputs
•	Key Activities
	Monitoring and Reporting
Priority outcome(s) of government that this grant primarily contributes to	Outcome 8: Sustainable human settlements and improved quality of household life
Conditions	Funds for this grant will only be released upon:     receipt of signed off provincial business plans supported by a project list per housing programme that indicate the readiness of projects for implementation, including cash flow projections report and compliance certificates
	provision of allocations to municipalities receiving the Urban Settlements Development Grant for publishing in a government gazette as required by section 10(8) of the 2011 Division of Revenue Act
	Transfers to provinces will be contingent on their performance as assessed in reports submitted through the Housing Subsidy System (HSS) for project and programme administration
	<ul> <li>Provinces may, if a proven need exists, utilise up to 5 per cent of the provincial allocation for the Operational Capital Budget Programme (OPSCAP) to support the implementation of the approved national provincial housing and accredited municipal programmes and priorities</li> </ul>
	<ul> <li>The Minister for Human Settlements may identify and approve a project as a priority project upon pronouncement by the State President, Cabinet, the Minister and/or Human Settlements MinMec</li> <li>A national priority project will satisfy one or more of the following conditions:</li> </ul>
	the project promotes a national development interest including poverty eradication, sustainable development and/or dignity of communities and citizens
	the project promotes the targets and outputs contained in Outcome 8
	the project promotes good practices in human settlement development the approval of the project will result in the alleviation of an emergency and/or a life threatening situation
	• To expedite the completion of the high impact projects the provincial departments of human settlements must allocate the stipulated amounts from the provincial grant allocation per project as
	indicated below: Gauteng; Khutsong Resettlement project: R105 million Gauteng; Doornkop: R200 million
	Gauteng; Diepsloot: R100 million
	Eastern Cape; Duncan Village: R120 million
	Gauteng; Sweet Waters: R50 million
	KwaZulu-Natal; Cornubia: R132 million Limpopo; Lephalale: R318.8 million
	Western Cape; Drommedaris: R68.6 million
	All new projects must form part of the Performance and Delivery Agreements signed in terms of
	Outcome 8, Provincial Multi-year Housing Plans, National, Provincial and Local Spatial Development Frameworks and Human Settlement Sector Plans and comply with the Housing Code and readiness criteria for implementation
	<ul> <li>and readiness criteria for implementation</li> <li>Provinces must make appropriate budget allocations to the National Upgrading Support</li> </ul>
	Programme to improve capacity to upgrade informal settlement households
	The targets per province and municipality accredited to level one and two accreditation must be consistent with the outputs and targets contained in the Delivery Agreements between the Minister, MECs and where appropriate with Mayors
	Where targets are revised and/or budget shifted a revised business plan must be submitted to National Department within 30 days after Adjusted Estimates of National Expenditure

	Human Settlements Development Grant
Allocation criteria	The grant is apportioned into a 80/20 split as follows:     80 per cent of the proposed 80/20 split is allocated to provinces in terms of the sector approved.
	formula
	20 per cent portion is allocated by the national department to supplement the funding of the
	social and rental housing, informal settlement upgrading, land and national priority programmes
	The provincial allocations to municipalities accredited to level one and two accreditation will
	be made on the sector approved formula basis that is consistent with the Accreditation Framework
	An interim allocation basis will consider the current approved projects budgets running in the municipalities accredited to level one and two
Reason not incorporated in equitable share	A conditional grant enables the national department to provide effective oversight, ensure compliance with the housing code and direct portions of the grant to accredited municipalities
Past performance	2009/10 audited financial outcomes
	Allocated and transferred R12 442 million to provinces
	Of the total available of R12 442 million, R12 250 million (98.5 per cent) was spent
	2009/10 service delivery performance
	• 161 854 housing units completed
	64 362 serviced sites completed
Projected life	It is a long term grant of which the exact life span cannot be stipulated as the government has an obligation to assist the poor with the provision of human settlements
MTEF allocations	• 2011/12: R14 942 million, 2012/13: R15 599 million and 2013/14: R16 457 million
Payment schedule	Monthly instalments as per the approved payment schedule
Responsibilities of the	Responsibilities of the national department
transferring national officer and receiving	Monitor the provincial and municipal performance on grant, financial and non-financial, and control systems related to the human settlements conditional grant
officer	Provide support to provinces and accredited municipalities with regards to human settlement delivery as maybe required
	Undertake structured and other visits as is necessary to provinces and municipalities
	Facilitate regular strategic interaction between national, provincial departments of human settlements and accredited municipalities
	Submit an annual evaluation report for 2010/11 to National Treasury by 29 July 2011
	• Submit quarterly performance reports to National Treasury within 45 days after the end of each
	<ul> <li>quarter</li> <li>Provide systems (Housing Subsidy System) that support the administration of the human settlement</li> </ul>
	<ul> <li>delivery process</li> <li>Comply with the responsibilities of the national transferring officer outlined in the Division of</li> </ul>
	Revenue Act (DoRA)
	Responsibilities of the provincial departments
	Submit 2010/11 annual evaluation reports to the national department by 31 May 2011
	<ul> <li>Support accredited municipalities in carrying out the functions delegated as per the Accreditation Framework</li> </ul>
	Utilize the Housing Subsidy System (HSS) for the administration of all the human settlement
	delivery processes
	Ensure effective and efficient utilization of the Housing Subsidy System by municipalities
	Comply with the responsibilities of the receiving officer outlined in the DORA
	Comply with the terms and conditions of the national performance agreements and provincial and
	local delivery agreements
	<ul> <li>Submit quarterly reports on funds allocated and utilised on programmes and projects in respect of OPSCAP</li> </ul>
Process for approval of	• First draft provincial business plans to be submitted to the national department by
2012/13 business plans	15 November 2011
<b>F</b>	• Submit final provincial business plans project lists including each flow projections and compliance
	<ul> <li>Submit final provincial business plans, project lists including cash flow projections and compliance certificates to the national department by 15 February 2012</li> <li>Submit approved 2012/13 provincial and national plan to National Treasury by</li> </ul>

### **PUBLIC WORKS GRANTS**

	Devolution of Property Rate Funds Grant
Transferring department	Public Works (Vote 7)
Strategic goal	• To enable provincial accounting officers to be fully accountable for their expenditure and payment of
	provincial property rates
Grant purpose	To facilitate the transfer of property rates expenditure responsibility to provinces
Outcome statements	• Provinces take full responsibility for property rates with respect to property owned and deemed to be
	owned by provincial departments
Outputs	• Payment of property rates for provincially-owned and deemed to be owned properties as per the
	verified invoices submitted by municipalities
Priority outcome(s) of	• Outcome 12: An efficient, effective and development oriented public service and an empowered, fair
government that this grant	and inclusive citizenship
primarily contributes to	N. P. 11
Details contained in the business plan	Not applicable
Conditions	Provinces must limit payments to invoices submitted by municipalities for property rates
Conditions	<ul> <li>Provinces must mint payments to invoices submitted by municipanties for property rates</li> <li>Provinces must ensure that sufficient provision is made within their baselines to accommodate future</li> </ul>
	commitments pertaining to provincially-owned and deemed to be owned properties
Allocation criteria	Funds are allocated per province based on the determination/calculations submitted by the national
Amocation criteria	Department of Public Works (DPW)
	The determinations/calculations are assessed against:
	property lists extracted from the national department's assets register of properties which were
	paid during 2005/06 financial year
	based on inputs by the provinces and the national departments as well as inflationary adjustments
Reason not incorporating in	To ensure that provinces are not adversely affected during the division of funds based on the number of
equitable share	properties in that specific province
	To ensure an effective transition of the function from national to provincial government
Past performance	2009/10 audited financial outcomes
	Allocated and transferred R1 350 million to provinces
	• Of the total available R1 486 million (including provincial roll overs), R1 214 million (81.7 per cent)
	was spent
	2009/10 service delivery performance
	Payments made for provincially-owned properties
Projected life	Subject to review in the 2011/12 financial year
MTEF allocations	• 2011/12: R1 803 million, 2012/13: R1 938 million and 2013/14: R2 091 million
Payment schedule	Three instalments ( 08 July 2011, 07 October 2011 and 20 January 2012)
Responsibilities of the	Responsibilities of the national department
national transferring officer	DPW shall be responsible for payment of outstanding property rates arrears of list of properties
and receiving officer	<ul> <li>devolved to the provinces prior to 1 April 2008</li> <li>Oversee the transfer of the function and skills development of the provincial receiving officers</li> </ul>
	<ul> <li>Monitor expenditure and performance of provinces monthly and quarterly</li> <li>DPW shall be responsible to apply for vesting of properties into the name of National Government</li> </ul>
	through the national Department of Rural Development and Land Reform (RDLR)
	Monitor and evaluate the devolution process
	Submit quarterly performance reports within 45 days after the end of each quarter to National Treasury
	Responsibilities of the provincial departments
	Provinces must validate payment claims and accounts from municipalities in line with Treasury
	Regulations
	Submit quarterly performance reports to DPW within 30 days after the end of each quarter
	Provincial departments are responsible to apply for vesting of properties into the name of provincial
	government through RDLR
	Provincial departments are responsible for requesting any additional funding from National Treasury
	through DPW
	Provincial departments are responsible for requesting roll over funds from National Treasury through
	Provincial Treasuries
Process for approval of	Not applicable
2012/13 business plans	

	Expanded Public Works Programme Incentive Grant for Provinces
Transferring department	Public Works (Vote 7)
Strategic goal	To increase the amount of full-time equivalent (FTE) jobs created through the use of labour-intensive methods by provincial departments
Grant purpose	To incentivise provincial departments to increase job creation efforts in infrastructure, environment and culture programmes through the use of labour-intensive methods and the expansion of job creation in line with the Expanded Public Works Programme (EPWP) guidelines
Outcome statements	Improved quality of life of unemployed people and increased social stability through engaging the
	unemployed in paid and productive activities  Reduced levels of poverty
	Contribute towards increased levels of unemployment
	Improved opportunities for sustainable work through experience and learning gained
Outputs	Increased number of people employed and receiving income through the EPWP
	<ul> <li>Increased average duration of the work opportunities created</li> <li>Increased income per EPWP beneficiary</li> </ul>
Priority outcome(s) of	Outcome 4: Decent employment through inclusive growth
government that this grant primarily contributes to	Successive in December and Sugar mentalities growth
Details contained in the	This grant uses incentives agreements which contains the following:
business plan	FTE thresholds for each eligible provincial department FTE target for each eligible provincial department
Conditions	Eligible provincial departments must sign a standard incentive agreement with the national Department
	of Public Works (DPW)
	Provincial departments must report on all projects for which they are claiming the incentive via DPW's EPWP Reporting systems
	Reports must be loaded on the EPWP reporting system within 22 days after the end of every quarter in
	order for performance to be assessed for an incentive payment 45 days after the end of each quarter
	Provincial departments must maintain beneficiary and payroll records as specified in the Audit Requirements in the EPWP Incentive Manual
	Once received by the eligible provincial department, the incentive grant must be used for continuing or expanding job creation programmes. A maximum of 5 per cent of the incentive received can be used for project management based capacity, which must be project linked and on contract
	To be paid out an incentive allocation, the eligible public body must:
	sign an incentive agreement
	report on EPWP performance quarterly within required timeframes perform above the quarterly threshold
Allocation criteria	To be eligible for an incentive allocation in 2011/12, a provincial department must have:
	reported EPWP performance by 22 April 2010
	met the threshold FTEs to qualify  • A nominal allocation of R680 000 is provided when:
	public bodies report after 22 April 2010 but before 22 October 2010
	public bodies do not meet the minimum threshold
	The nominal allocation can be accessed if public bodies meet their threshold in-year  The common state of the common stat
	• Incentive allocations are based on a targeted number of FTEs each provincial department should create above the set minimum threshold. Incentive allocations are indicative and the actual incentive payout is
	based on actual quarterly performance reported above the quarterly threshold during 2011/12
	• The indicative incentive allocation from underperforming provincial departments may be re-allocated to
	over performing provincial departments by Public Works  • Eligible provincial departments that exceed their targets may be paid an incentive in excess of their
	published indicative incentive allocation subject to the availability of funds
Reason not incorporated in equitable share	This grant is intended to incentivise and reward performance on the EPWP. This incentive will be paid out based on the performance of provincial departments
Past performance	2009/10 audited financial outcomes
	Payments on the EPWP infrastructure incentive grant was made to the following provinces:     Eastern Cape: R18 million
	- Eastern Cape: K18 million - KwaZulu-Natal: R85 million
	- Western Cape: R 14 million
	2009/10 service delivery performance
	<ul> <li>129 193 work opportunities were reported from provincial departments in the infrastructure sector</li> <li>36 601 full time equivalent jobs were reported from provincial departments in the infrastructure sector</li> </ul>
Projected life	Grant continues until the end of the 2013/14 financial year subject to review

	Expanded Public Works Programme Incentive Grant for Provinces
MTEF allocations	• 2011/12: R267 million, 2012/13: R325 million and 2013/14: R402 million
Payment schedule	• Four instalments per annum (16 May 2011, 15 August 2011, 15 November 2011 and 15 February 2012)
Responsibilities of the transferring national officer and receiving officer	Responsibilities of the national department of public works  Determine eligible provincial departments and set minimum thresholds, performance targets and indicative incentive allocations for eligible municipalities based on reported performance  Provide provincial departments with the EPWP Incentive Manual outlining the conditions and obligations of the incentive as well as the audit requirements  Support provincial departments to apply the EPWP Guidelines to project design  Support provincial departments to report using the EPWP reporting systems  Monitor the performance of provincial departments and report on quarterly progress against targets  Submit to National Treasury quarterly progress against performance targets; indicative incentive amounts earned by provincial departments; and a revised payment schedule within 45 days after the end of the quarter  Conduct sample audits on a continuous basis  Audit the final performance of provinces after the closure of the financial year  Disburse incentives on a quarterly basis to eligible provincial departments (via provincial treasuries) based on performance in the previous quarter once the quarterly thresholds have been exceeded  Responsibilities of the provincial treasuries  Appropriate the indicative incentive allocation to eligible provincial departments as indicated in the annual Division of Revenue Act (DoRA) and the Gazette on allocations and frameworks for provinces  Disburse the incentive to the eligible provincial departments in accordance with the quarterly disbursement instructions of National Public Works  Monitor the spending of the incentive on job creation programmes and provide spending reports on the grant to Public Works as per DoRA
	Responsibilities of the eligible provincial departments
	Sign the standard incentive agreement with National Public Works agreeing to comply with the
	<ul> <li>conditions and obligations of the incentive grant before receiving any incentive payment</li> <li>Report all projects to be taken into account when assessing performance into the EPWP reporting system and update progress quarterly in accordance with the reporting requirements in the incentive agreement</li> </ul>
Process for approval of 2012/13 business plans	<ul> <li>Provincial departments must report performance on EPWP projects for the 2010/11 financial year by the 21 April 2011</li> <li>A late eligibility provision is applicable for late reporting: Public bodies that report after this date but before 21 October 2011 will qualify for only a nominal incentive allocation of R680 000</li> </ul>
	<ul> <li>National Public Works will distribute the standard incentive agreement by 31 March 2011</li> <li>Eligible provincial departments must sign the standard incentive agreement with National Public Works and agree to comply with the conditions and obligations of the incentive grant by 29 April 2011</li> </ul>

Socia	l Sector Expanded Public Works Programme Incentive Grant for Provinces
Transferring department	Public Works (Vote 7)
Strategic goal	To increase job creation through the expansion of Social Sector Expanded Public Works Programme (EPWP) programmes
Grant purpose	To incentivise provincial social sector departments identified in the 2011 Social Sector EPWP Log-frame to increase job creation by focusing on the strengthening and expansion of social service programmes that have employment potential
Outcome statements	<ul> <li>Improved service delivery to communities by expanding the reach and quality of social services</li> <li>Improved quality of life of unemployed people through employment creation and increased income</li> <li>Reduced levels of poverty</li> <li>Contribute towards decreased levels of employment</li> <li>Improved opportunities for sustainable work through the experience and learning gained</li> <li>Strengthened capacity of non-government delivery partners through increased access to funds for training, wages and administration</li> </ul>
Outputs	<ul> <li>Increased number of people employed and receiving income through the EPWP</li> <li>Increased duration of the work opportunities created</li> <li>Increased number of households and beneficiaries to which services are provided</li> <li>Increased income per EPWP beneficiary</li> </ul>
Priority outcome(s) of government that this grant primarily contributes to	<ul> <li>Outcome 4: Decent employment through inclusive economic growth</li> <li>Outcome 5: A skilled and capable workforce to support an inclusive growth path</li> <li>Outcome 7: Vibrant, equitable and sustainable rural communities and food security for all</li> </ul>
Details contained in the business plan	<ul> <li>Outcome indicators</li> <li>Output indicators</li> <li>Inputs</li> <li>Key activities</li> </ul>
Allocation criteria	<ul> <li>Provincial departments must submit to the national Department of Public Works (DPW) signed-off EPWP targets and budgets for the 2011/12 financial year as part of the EPWP annual log frame planning process by 31 March 2011</li> <li>Provincial departments must submit to DPW signed-off implementation plans on how to achieve these EPWP targets by 31 March 2011</li> <li>Provincial departments must report both conditional grant and equitable share EPWP expenditure on the monthly In-Year-Monitoring (IYM) tool in accordance with section 32 of the PFMA</li> <li>Provincial departments must sign a incentive agreement with DPW by 31 March 2011 to comply with the conditions and obligations of the grant</li> <li>Reports must be loaded on the EPWP Web-based Reporting System within 15 days after the end of every quarter</li> <li>Provincial departments must adhere to the Audit Requirements stipulated in an EPWP Incentive Manual</li> <li>The Incentive Grant allocation must be used to expand job creation programmes in the Social Sector</li> <li>The Incentive Grant allocation must be used to fund the following priority areas:         <ul> <li>To provide stipends to unpaid volunteers at R60 per day</li> <li>To expand Social Sector EPWP programmes as identified in the EPWP Social Sector Log-frame</li> <li>To create additional work opportunities</li> </ul> </li> <li>A minimum of 80 per cent of the incentive allocation must used to pay stipends or wages</li> <li>To be eligible for an incentive allocation in 2011/12, a provincial department must have:</li> </ul>
Allocation criteria	<ul> <li>To be eligible for an incentive allocation in 2011/12, a provincial department must have:         <ul> <li>reported EPWP performance by 22 April 2010 for an incentive allocation to be calculated based on 2009/10 performance</li> <li>must have met at least 35 per cent of their Full Time Equivalent (FTE) target for the 2009/10 financial year</li> </ul> </li> <li>Each provincial department's performance is assessed against a set of EPWP performance indicators to determine the size of the incentive allocations for those years. These are:         <ul> <li>Targeted number of FTE per provincial department</li> <li>Beneficiary profile consisting of 2 per cent persons with disabilities</li> <li>Beneficiary profile consisting of 40 per cent youth</li> <li>Beneficiary profile consisting of 55 per cent female beneficiaries</li> <li>10 per cent of days worked spent in training</li> <li>Average duration of 100-day work opportunities</li> <li>Average Minimum Daily Wage of R60 per person day of work</li> </ul> </li> <li>Incentive allocations to each provincial department are based on their past performance score [%] x the number of FTEs created x FTE Reward [R60 per day x 230 days]</li> <li>Provincial grant allocations for the two outer years are indicative and will be revised based on the performance of each province</li> </ul>

Social	Sector Expanded Public Works Programme Incentive Grant for Provinces
Reason not incorporated in equitable share	The incentive allocation is based on the performance of programmes in a prior financial year and use of the allocation is specifically earmarked for EPWP programme expansion
Past performance	2009/10 audited financial outcomes
	• New grant
	2009/10 service delivery performance
	<ul> <li>New grant</li> <li>Ongoing subject to review</li> </ul>
Projected life	
MTEF allocations	• 2011/12: R200 million, 2012/13: R242 million and 2013/14: R286 million
Payment schedule	• Four instalments (6 May 2011, 8 July 2011, 7 October 2011 and 6 January 2012)
Responsibilities of the	Responsibilities of the national department of public works
transferring national officer	• Determine the eligibility of provincial departments, set job creation targets and performance
and receiving officer	measures and calculate incentive allocations
	Draft an Incentive Manual that will provide provincial departments with standard information on the rules of the incentive programme, its application, monitoring and evaluation information
	and audit regulations
	<ul> <li>Draft an Incentive Agreement outlining the requirements of the Incentive Grant and ensure that</li> </ul>
	each provincial department signs the agreement by 31 March 2011
	Reach agreement with national sector departments on their roles in ensuring effective
	implementation of the Incentive Grant by 31 March 2011
	Support provincial departments to develop plans to meet job creation targets
	Support the sector to collect the required data, align monitoring and reporting frameworks and to      Support and law entered and the EDWA With Bland Support
	<ul> <li>report on key outputs on the EPWP Web Based System</li> <li>Monitor the performance of provincial departments and the use of the incentive grant against the</li> </ul>
	conditions in the framework and report to National Treasury on quarterly progress
	<ul> <li>Audit the final performance of provincial departments after the closure of the financial year</li> </ul>
	Responsibilities of provincial departments
	Identify the employment and expansion potential of departmental Social Sector EPWP
	programmes and develop plans for maximising job creation and service delivery expansion
	Submit and obtain approval for the required programme expansion plans to DPW to show how targets will be achieved
	Sign the standard incentive agreement with DPW agreeing to comply with the conditions and obligations of the incentive grant before receiving any incentive payment
	<ul> <li>Report EPWP performance onto the EPWP Web Based System and update progress quarterly in</li> </ul>
	accordance with the reporting requirements in the incentive agreement
	• Provide data on the use of the Incentive Grant on a quarterly basis in the format and manner
	prescribed by DPW
	Maintain beneficiary and payroll records as specified in the Audit Requirements of the EPWP
Process for approval of	• Provincial departments report performance on Social Sector EPWP programmes for the 2010/11
2012/13 business plans	financial year by the 15 April 2011  Performance in 2010/11 will determine the terrests and incentive ellocations for 2012/13
	<ul> <li>Performance in 2010/11 will determine the targets and incentive allocations for 2012/13</li> <li>Provincial departments participate in the planning exercise in January each year and submit their</li> </ul>
	business plans and targets to DPW during this process in the format required
	DPW to distribute the Incentive Agreement in March every year
	Provincial departments sign the Incentive Agreement with DPW by 30 March 2012 and agree to
	comply with the conditions and obligations of the incentive grant

# SPORT AND RECREATION SOUTH AFRICA GRANT

	Mass Sport and Recreation Participation Programme Grant
Transferring department	Sport and Recreation South Africa (Vote 20)
	Increased and sustained participation in sport and recreation development continuum
Strategic goal	
Grant purpose	• To facilitate mass participation within communities and schools through selected activities, empowerment of communities and schools in conjunction with relevant stakeholders
Outcome statements	Increased and sustained participation in sport through mass participation
Outcome statements	Mass participation in schools and communities
	Mass participation in sports programmes amongst marginalised groups
Outputs	Improved sector capacity/increased capacity to deliver Mass Participation in communities and schools
_	Sustainable mass participation structures established
	Schools and communities supported through the mass participation programmes
	Marginalised groups supported to participate in mass participation programmes
Priority outcome(s) of government that this grant primarily contributes to	Outcome 12: An efficient, effective and development oriented public service and an empowered, fair and inclusive citizenship
Details contained in the	Outcome indicators
business plans	Output indicators
	• Inputs
Conditions	Key activities  Provincial department of the formula and
Conditions	<ul> <li>Provincial departments responsible for sport are required to enter into formal agreement with Sport and Recreation South Africa (SRSA) after approval of business plans prior to the start of the financial year</li> </ul>
	<ul> <li>Provinces must appoint permanent staff on their establishments for the programme in consultation with</li> </ul>
	the national department. The establishment must not be more than 6 per cent of the total grant
	requested
	• Each province must have risk management plans by 30 March 2011
	• Each province must have a sustainability plan for projects to ensure they will be self-sufficient after 3
	<ul> <li>years</li> <li>Provincial department's Annual Performance Plan for 2011/12 and over the MTEF to clearly indicate</li> </ul>
	measurable objectives and performance targets of the conditional grant as agreed with the national
	department
Allocation criteria	• Funds are distributed among provinces on a baseline allocation of R10 million, a needs analysis, performance and the provincial equitable share formula
Reason not incorporated in equitable share	A conditional grant is necessary to ensure national coordination, monitoring and facilitation
Past performance	2009/10 audited financial outcomes
	Allocated and transferred R402 million     Of the state and transferred R402 million     Page 2 million (62 ( conserv) many)
	Of the total available R412 million (including provincial roll overs), R382 million (92.6 per cent) was spent
	2009/10 service delivery performance
	Number of people trained in sport and development: 14 679
	Number of Mass mobilisation campaigns held: 45
	• Number of people participating in the programme: 4 520 000 people participating in schools and
D : 4 11:6	community clubs
Projected life MTEF allocations	Ongoing subject to review as agreed with National Treasury  2011/12 PA52 million 2012/12 PA75 million and 2012/14 P501 million
Payment schedule	<ul> <li>2011/12: R452 million, 2012/13: R475 million and 2013/14: R501 million</li> <li>Four instalments (29 April 2011, 29 July 2011, 31 October 2011 and 30 January 2012)</li> </ul>
Responsibilities of the	Responsibilities of the national department
transferring national	Submit the 2010/11 annual evaluation report to National Treasury by 29 July 2011
officer and receiving	• Agree on outputs and targets with provincial departments in line with grant objective for 2012/13 by
officer	15 September 2011
	Provide the guidelines and criteria for the development and approval of business plans  Marier in all approval deposits and approval of business plans
	<ul> <li>Monitor implementation and provide support</li> <li>Submit quarterly performance reports to National Treasury within 45 days after the end of each quarter</li> </ul>
	• Submit quarterly performance reports to National Treasury within 45 days after the end of each quarter
	• Ensure that all the conditional grant practice notes issued by National Treasury are adhered to
	• Ensure that all the conditional grant practice notes issued by National Treasury are adhered to <b>Responsibilities of the provincial departments</b>
	• Ensure that all the conditional grant practice notes issued by National Treasury are adhered to  Responsibilities of the provincial departments
	<ul> <li>Ensure that all the conditional grant practice notes issued by National Treasury are adhered to</li> <li>Responsibilities of the provincial departments</li> <li>Submit the 2010/11 annual evaluation report to SRSA by 31 May 2011</li> </ul>

Mass Sport and Recreation Participation Programme Grant	
	Provinces to ensure that capacity exist to manage the grant and there's a grant manager responsible for
	the grant
	Provinces must include the organisational capacity on the programme
Process for approval of	Provinces to provide draft business plans to SRSA by 14 October 2011
2012/13 business plans	SRSA evaluates draft business plan by 15 November 2011
	Comments sent to provinces by 30 November 2011
	Provinces to submit revised business plans to SRSA by 15 December 2011
	SRSA to approve revised business plans by 13 February 2012
	HODs submit signed business plans to SRSA 15 March 2012
	SRSA to sign project implementation agreements and business plans with Provincial HODs by
	5 April 2012
	SRSA submit approved business plan to National Treasury by 13 April 2012

## TRANSPORT GRANTS

	Gautrain Rapid Rail Link
Transferring department	Transport (Vote 37)
Strategic goal	To develop a fully integrated Rapid Rail Link based on a north-south spine between Tshwane and Johannesburg and an east-west spine between OR Tambo International Airport (ORTIA) and Sandton
Grant purpose	To provide for national government funding contribution to the Gauteng Provincial Government for the construction of a fully integrated Gautrain Rapid Rail network
Outcome statements	Reduction in road traffic congestion in the corridors between Tshwane and Johannesburg and between ORTIA and Sandton
	Integration of the Gautrain Rapid Rail Link into to broader public transport network in Gauteng
Outputs	The completion of the civil works and operational systems of the Gautrain Rapid Rail Link according to the specifications and milestones agreed between the Gauteng Province and the Concessionaire in the Public Private Partnership (PPP) Concession Agreement (CA)
	Implementation of the Gautrain Strategic Integration Plan as approved by Cabinet in December 2005
Priority outcome(s) of government that this grant primarily contributes to	Outcome 6: An efficient, competitive and responsive economic infrastructure network
Details contained in the business plan	This grant uses a PPP Concession Agreement
Conditions	<ul> <li>The conditional grant is to be used towards the payment of the provincial contribution for the completion of the Gautrain, as specified in the CA</li> <li>The Province's rights and obligations in the CA will be managed by the Gauteng Management</li> </ul>
	Agency (GMA), established through provincial legislation as a Schedule 3C public entity in terms of the Public Finance Management Act (PFMA)  • The GMA will provide the national Department of Transport(DoT) with an annual projection of payment in terms of the Milestone completion schedule in the CA, which indicates the projected
	dates on which each General Milestone and each Key Milestone payment will be due and the quantum thereof payable from the conditional grant  Interim payments and the final payment made by the Province to the Concessionaire in terms of the
	Concession Agreement throughout the development period (payable according to the agreed General and Key Milestone completion schedules) will be made by the province, drawing from the conditional grant payments received from the national DoT in such a manner that the amounts expended on the project include any adjustment necessary to compensate for foreign exchange fluctuations as agreed by the National Treasury as well as escalation over and above the inflation rate set for the Project by the South African Reserve Bank (SARB)
	• Failure by the province to make payments to the Concessionaire within the stipulated 10 business days will result in the Province incurring interest on each overdue sum
	<ul> <li>The national DoT will advise the SARB of the annual payment schedule required for the Gautrain</li> <li>The GMA will advise the Gauteng Treasury to draw required funds from the SARB's Gautrain holding account and to effect payment to the Concessionaire</li> </ul>
	The annual payment schedule lodged with the SARB may be amended from time to time according to revisions to the annual cash flow forecasts provided by the GMA
	The Gauteng Department of Roads and Transport must ensure the efficient and effective functioning of the Gautrain Public Transport Integration Committee comprising of the three metros (Tshwane, Johannesburg and Ekurhuleni), DoT and the Passenger Rail Agency of South Africa with the primary task of developing a Gautrain Integration Implementation Plan that will respond to the:
	Rail Plan corridor alignment and the Gautrain implementation parameters proposed new commuter rail links including station precinct developments, specifically
	Pretoria, Johannesburg, Hatfield and Rhodesfield as well as park and ride and feeder facilities a single architecture integrated ticket system and the roll out of fare integration strategy to all public transport users and public transport service providers effected in the three Metropolitan Authorities
	Operating License Strategies of the three Metropolitan Authorities Operating Contracts with metropolitan bus services, subsidised bus services, Metrorail and
Allocation criteria	recapitalised taxi operators and their service specifications  The Grant is based on a financial model specifically developed for this project

	Gautrain Rapid Rail Link
Reason not incorporated in equitable share	<ul> <li>The conditional grant is made for a specific, large public transport infrastructure project being undertaken by the Gauteng Province, as endorsed by Cabinet. The total cost of the project is unaffordable to the Province within the limits of its equitable share. It has therefore been agreed that fifty percent of the capital costs of the project will be borne by the Province (through a combination of funds from its equitable share and a borrowing agreement with National Treasury), and fifty percent will be borne by national government and made available to the Province through a conditional grant</li> <li>Cabinet has endorsed the project as a key strategic PPP project of national significance, with potential to stimulate investment in infrastructure and the economy, and to provide opportunities for public transport restructuring and integration</li> </ul>
Past performance	2009/10 audited financial outcomes
	Allocated and transferred R2 977 million to provinces
	Of the total available allocation, 100 per cent was spent
	2009/10 service delivery performance
	• Service delivery is measured in terms of the Annual Performance Evaluation submitted by the GMA, based inter alia on Milestones certified by the Independent Certifier
	The Concessionaire reported actual total project progress to be at 89 per cent compared to a planned progress of 94 per cent
Projected life	• The grant comes to an end in 2011/12
MTEF allocations	• 2011/12: R5.3 million
Payment schedule	<ul> <li>Transfer payments will be made based on an annual transfer schedule provided to the SARB by the national DoT, according to annual cash flow forecasts provided by the Gautrain Management Agency, which may be amended from time to time by the National Treasury</li> </ul>
Responsibilities of the	Responsibilities of the national department
transferring national officer	DOT ensures that transfer payments are made to the Gauteng Province in accordance with the
and receiving officer	approved payment schedule and certified Milestones
	• Ensures that expenditure from the Division of Revenue Act (DoRA) grant is in terms of the
	allocated funds and based on the approved payment schedule
	Submits reports as required in terms of the DoRA to National Treasury  Output  Description:
	Facilitates the integration of Gautrain with the public transport system in line with the Cabinet resolution
	Responsibilities of the provincial departments
	The Gautrain Management Agency (GMA) on behalf of the Gauteng Provincial Government manages the Public-Private Partnership (PPP) Agreement and reports on progress and expenditure to DoT in accordance with the above-mentioned conditions.
	• In line with the DoRA and the PFMA requirements, the Gauteng Province will submit reports to the DoT at the end of each quarter, detailing:
	interim payments and the final payment made to the Concessionaire according to the General and Key Milestone payment schedules in that quarter. These reports will be supported by copie of the Interim Certificates and (when applicable) the Final Certificate issued by the Independen Certifier, which effected each payment made in the previous quarter in terms of the Concession Agreement progress as it relates to Gautrain Integration Implementation Plan
	These reports will be submitted by the Gauteng Province to the DoT in the first week of January,     April, July and October each year of the development period
Process for approval of 2012/13 business plans	Not applicable

	Provincial Roads Maintenance Grant
Transferring department	• Transport (Vote 37)
Strategic goal	To ensure efficient investment in provincial roads
Grant purpose	To supplement provincial roads investments and support preventative, routine and emergency
	maintenance on provincial road networks
Outcome statements	Ensure provinces implement and maintain road asset management systems  In any and life group of growing in larged and decreased which appreciate acceptance of growing in larged and decreased which appreciate acceptance of growing in larged and decreased which appreciate acceptance of growing in larged and decreased which appreciate acceptance of growing in larged and decreased which appreciate acceptance of growing in larged and decreased which acceptance of growing in larged acceptance of growing in
Outcome statements	Increased lifespan of provincial roads and decreased vehicle operating costs on provincial road networks
Outputs	Collected traffic volumes data and pavement condition data
Outputs	Updated road asset management systems
	Squared kilometres of of preventative, routine and emergency maintenance work
	Percentage of roads for which condition of roads changed from poor and very poor to at least fair
	Weighbridges
	Percentage of roads in good or very good condition maintained in that condition
	Number of EPWP jobs created
	Number of S3 students provided with experiential internships
	Number of emerging contractor opportunities created
Priority outcome(s) of	Outcome 6: An efficient, competitive and responsive economic infrastructure network
government that this grant	
primarily contributes to	
Details contained in the	This grant uses a Road Asset Management Plan, which contains the following details:
business plan	- levels of service
	<ul> <li>network condition and traffic volumes</li> <li>project lists for 2011/12 to 2013/14</li> </ul>
	- project lists for 2011/12 to 2013/14 - financial summary
	- organisational and support plan
Conditions	Provincial departments must submit quarterly infrastructure reports to the national Department of
	Transport (DoT) that comply with the infrastructure reporting model and toolkit
	• Final list of projects must be captured on the Infrastructure Reporting Model and submitted to the
	relevant Provincial Treasury and Department of Transport by the 21 April 2011
	• A detailed Asset Management Plan, that is compliant with the requirements of the Government
	Immovable Assets Management Act (2007) and based on the Road Asset Management System must
	be submitted by 31 August 2011 to DoT and National Treasury by 30 November 2011
	All immovable asset management and maintenance responsibilities as prescribed by the  Common Management and maintenance responsibilities as prescribed by the
	Government Immovable Asset Management Act, 2007 (GIAMA), and the Infrastructure Delivery Management Toolkit must be adhered to
	The payment of the second instalment of this grant is dependent on receipt by DoT of the quarterly
	performance reports for the fourth quarter of 2010/11. The third instalment is dependent on receipt
	by DoT of the first quarter report for 2011/12 and the final instalment is dependent on receipt of the
	second quarterly performance reporting for 2011/12
	• Expenditure of this grant must be in accordance with maintenance requirements as identified by a
	road asset management system, set within the relevant budget limit, to improve conditions of the
	roads and extend the lifespan of road infrastructure
	• Allocations for 2013/14 will be conditional on updated systems and data. Data on roads and bridges
	needs to be updated every two and five years respectively
	Projects for which tendered contracts have been entered into or funding committed before  21 Marsh 2011 will be avanted from the conditions above enert from the guertally greating.
	31 March 2011 will be exempted from the conditions above apart from the quarterly reporting
	<ul> <li>requirements on the infrastructure reporting model</li> <li>Mpumalanga and Gauteng provinces must allocate the following amounts to coal haulage projects:</li> </ul>
	<ul> <li>Mpumalanga and Gauteng provinces must allocate the following amounts to coal haulage projects:</li> <li>Mpumalanga: R511 million in 2011/12, R659 million in 2012/13 and R808 million in 2013/14</li> </ul>
	- Mpuniatanga. R311 hillion in 2011/12, R039 hillion in 2012/13 and R008 hillion in 2013/14 - Gauteng: R74 million in 2011/12, R4.4 million in 2012/13 and R0.9 million in 2013/14
Allocation criteria	Criteria will include performance in relation to key performance indicators as agreed to between
<del></del>	DoT and the provinces
	• Allocations are retained at 45 per cent of the Infrastructure Grant to Provinces in relation of
	proportional allocations shown in the 2010 Division of Revenue Act for 2011/12 and 2012/13
	Allocation criteria from 2013/14 onwards will be based on the extent of the provincial road
	network, the traffic volumes, the visual condition indices on the network and geo-climatic and
	topographic factors. Key performance indicators will be developed during 2011/12 which will be
	used to link allocations of this grant to performance
	• The funding for the coal haulage road network to Gauteng and Mpumalanga is subject to separate
D	allocation criteria based on the programme schedule as submitted to DoT and National Treasury
Reason not incorporated in	This grant is intended to ensure that provinces give priority to road infrastructure maintenance and
equitable share	promote efficiency in road investment

Provincial Roads Maintenance Grant	
Past performance	2009/10 audited financial outcomes
	New grant
	2009/10 service delivery performance
	New grant
Projected life	The grant is ongoing, but will be subject to periodic review
MTEF allocations	• 2011/12: R6 457 million, 2012/13: R7 568 million and 2013/14: R8 259 million
Payment schedule	According to payment schedule approved by National Treasury
Responsibilities of the	Responsibilities of the national department
transferring national officer and receiving officer	Assess and evaluate all provinces road asset management systems and ensure these are used to plan and prioritise maintenance work according the Road Infrastructure Strategic Framework for South Africa
	Provide support to provinces to enable them to report on asset management system data that is compatible with the national system
	<ul> <li>Maintain standards for Pavement Management Systems (TRH22) and ensure standardized data collection in relation to visual conditioning indices (TMH9,THM12 and M3-1), traffic volumes and vehicle operating costs</li> </ul>
	<ul> <li>Monitor expenditure and performance in accordance with road asset management systems</li> <li>Verify that primary data sources are updated</li> </ul>
	Submit quarterly performance reports to National Treasury and National Council of Provinces within 45 days after the end of each quarter
	• Prescribe national principles, requirements, guidelines, frameworks and national norms and standards that must be applied uniformly by provinces as per the National Land Transport Act (Act 5 of 2009)
	• Liaise with tertiary institutions for the placement of S3 students
	Responsibilities of the provincial departments
	• Update monthly expenditure reporting in terms of Section 40(4)(c) of the Public Finance Management Act and through the Infrastructure Reporting Model
	Submit quarterly performance reports within 30 days after the end of each quarter to the relevant provincial treasury and National Treasury, including reporting on:     visual conditioning index
	- visual conditioning index - traffic volumes
	- lifespan of the road (based on original design life, traffic and weather patterns)
Process for approval of 2012/13	Provinces submit Road Asset Management Systems priority projects in the form of a User Asset
business plans	Management Plan by 30 November 2011

	Public Transport Operations Grant
Transferring department	Transport (Vote 37)
	Subsidisation of road based public transport services
Strategic goal	
Grant purpose	To provide supplementary funding towards public transport services provided by provincial departments of transport
<b>Outcomes statements</b>	The provision of public transport services in terms of contracts which are kilometre based that are supportive of intermodal efficiencies in public transport and affordable to the users of the services
Outputs	Subsidy per trip operated
_	Subsidy per km operated
	Subsidy per passenger
	Subsidy per vehicle
	Number of vehicles subsidised
	Number of scheduled trips     Number of trips are set of
	Number of trips operated     Total number of possible in suggest
	<ul> <li>Total number of penalties incurred</li> <li>Monetary value of penalties (including escalation)</li> </ul>
	Passengers per kilometre operated
	Passengers per trip operated
	Employees per vehicle
Priority outcome(s) of	Outcome 6: An efficient, competitive and responsive economic infrastructure network
government that this grant primarily contributes to	
Details contained in	Not applicable
business plans	
Conditions	• The conditional grant is the national contribution to subsidised service contracts entered into by the
	provincial departments of transport and public transport operators for the provision of affordable subsidised services
	All contracts concluded must be done as per relevant legislation and in compliance with the Public
	Transport Strategy
	Designs and operators' business plans detailing subsidised services will have to be approved by the Public Transport Integration Committee comprising of the three spheres of Government to ensure alignment with Integrated Public Transport Networks (IPTNs). Where an Intermodal Planning Committee is established at municipal level, in terms of the National Land Transport Act no. 5 of 2009, the functions of the two committees must be consolidated to ensure integration of planning, services and modes
	<ul> <li>Supervision, monitoring and or verification must be done to certify the correctness of the operator's</li> </ul>
	claim in terms of km of services provided and report to Department of Transport (DoT) monthly
	• If the operating licence function is transferred to any municipality before the 2011/12 adjusted budget,
	the appropriate portion of the grant will also be devolved to the municipality. This devolvement must include all services in that city at once. To have the money devolved in the adjusted budget the municipality will have to have received the function by 1 September 2011. Should the function be devolved later than that the money will only be shifted in 2012/13. The municipality, province and
Allocation criteria	<ul> <li>operators will have to make transitional arrangements to ensure payments to operators</li> <li>The 2011/12 to 2013/14 allocations are based on 2009 Division of Revenue Act (DoRA) allocation baseline plus percentage of additional budget per year based on an allocation formula as developed by DoT recognising the following factors:</li> </ul>
	DoT recognising the following factors: population weighted by density
	GDP per province
	employment in province
	poverty population
	passengers transported
Reason not incorporated in equitable share	Subsidies are earmarked for the provision of public transport services
Past performance	2009/10 audited financial outcomes
periorinanee	Allocated and transferred R3 532 million to provinces
	• Of the total available of R3 532 million (including provincial rollovers), R3 523 million (99.3 per cent)
	was spent

	Public Transport Operations Grant
	2009/10 service delivery performance
	Summary of Monthly Averages for Key Outputs and Performance for subsidised services 2009/10     Number of vehicles subsidised: 6 676
	Number of routes subsidised: 8 224 Number of vehicle kms subsidised: 22 760 344 Subsidy/vehicle: 44 193 Subsidy/passenger: 10
	Subsidy/kilometre operated: 13 Kilometres operated/vehicle: 3 410 Passengers/vehicle: 4 615 Passengers/trip operated: 54 Passenger revenue/kilometre: 11 Passenger revenue/trip operated: 432
Projected life	Staff/vehicle: 2  • Subject to the devolution of funds to local government as part of the operationalisation of the National Land Transport Act (NLTA)
MTEF allocations	• 2011/12: R4 153 million, 2012/13: R4 361 million and 2013/14: R4 601 million
Payment schedule	Twelve (12) monthly payments/instalments according to payment schedule approved by National Treasury
Responsibilities of the transferring National officer and receiving officer	<ul> <li>Responsibilities of the national department</li> <li>Develop and coordinate the necessary contracting documents to be used in subsidising public transport services</li> <li>Provide the guidelines and criteria for the development of business plans for services to be subsidised</li> <li>Develop norms and standards for the creation of the IPTNs with all spheres of government</li> <li>Advise contracting authorities regarding the design of contracted services</li> <li>Maintain national database with key performance indicators of public transport services</li> <li>Improve efficiencies of public transport spending</li> <li>Responsibilities of the provincial department</li> <li>Any contractual agreement entered into by a provincial department in relation to this grant will be the responsibility of the provincial department</li> <li>Verify accuracy of monthly claims</li> <li>Ensure that contracted operators' certified claims are paid within thirty (30) calendar days from the date of receipt</li> <li>Submit monthly performance reports to DoT within 25 calendar days of the month following the operation and quarterly performance reports within 30 days after the end of each quarter using the reporting format developed by DoT</li> <li>Utilise supervision, monitoring and or verification to certify the correctness of the operator's subsidy claims in terms of km of services provided and report to DoT monthly</li> <li>Ensure alignment of IPTNs with national policy, legislation and other guidelines and/or standards</li> <li>Provinces must assist municipalities in the process of devolving the operating licence function as set</li> </ul>
Process for approval of 2012/13 business plans	out in the National Land Transport Act  Not applicable

### **ANNEXURE W3:**

### FRAMEWORKS FOR CONDITIONAL GRANTS TO MUNICIPALITIES

### Annexure W3: Frameworks for Conditional Grants to Municipalities

Detailed frameworks on Schedules 4, 6, 7, 8 and 9 grants to municipalities

### **Introduction**

This annexure provides a brief description for each grant in Schedules 4, 6, 7, 8 and 9 of the 2011 Division of Revenue Bill. The following are key areas considered for each grant:

- Strategic goal and purpose of the grant
- Outcome statements and outputs of the grant
- Priority outcome(s) of government that the grant primarily contributes to
- Conditions of the grant (additional to what is required in the Bill)
- Criteria for allocation between municipalities
- Rationale for funding through a conditional grant
- Past performance
- The projected life of the grant
- 2011 MTEF allocations
- The payment schedule
- Responsibilities of transferring national department and receiving municipalities
- Process for approval of business plans for 2012/13/Process for approval of 2012 MTEF allocations

The attached frameworks are not part of the Division of Revenue Bill, but are published in order to provide more information on each grant to parliament, legislatures, municipal councils, officials in all three spheres of government and the public. Once the 2011 Division of Revenue Bill is enacted, these frameworks will be gazetted in terms of the Act.

The financial statements and annual reports for 2011/12 will report against the Division of Revenue Act, Division of Revenue Amendment Act and their schedules, and the grant frameworks as gazetted in terms of the Act. Such reports must cover both financial and non-financial performance, focusing on the outputs achieved.

### COOPERATIVE GOVERNANCE AND TRADITIONAL AFFAIRS GRANTS

	Municipal Disaster Grant
Transferring department	Cooperative Governance and Traditional Affairs (Vote 3)
Strategic goal	To pro-actively respond to immediate needs after a disaster has occurred
Grant purpose	To provide for the immediate release of funds for disaster response
Outcome statements	Immediate consequences of disasters are alleviated
Outputs	<ul> <li>Victims of disasters are supplied with immediate relief</li> <li>The impact of disasters minimised</li> </ul>
Priority outcome(s) of government that this grant	• Outcome 12: An efficient, effective and development orientated public service and an empowered, fair and inclusive citizenship
primarily contributes to	Outcome 9: A responsive, accountable, effective and efficient local government system
Details contained in the	This grant uses the template developed by the National Disaster Management Centre (NDMC) which
business plan	must include the following:  Copy of the classification letter in terms of the Disaster Management Act Copy of declaration of disaster in terms of the Disaster Management Act Number of people affected Items to be purchased or that have already been purchased by municipalities with relevant proof Support received from Non-Government Organisations (NGOs) and local business Contribution by the municipality (both financially or in-kind) Funds required for disaster response
Conditions	<ul> <li>Funds from this grant can only be used to fund the items as described in the NDMC guideline and will only be released on request of a provincial government through the Provincial Disaster Management Structure</li> <li>Funds can only be released after a disaster has been declared in terms of the Disaster Management Act</li> <li>Municipalities must fund a portion of the costs of the disaster response from their own budget or prove that they are not able to do so. Municipalities with large revenue raising capacity will be expected to fund 25 per cent of the response, mid-sized municipalities will be expected to fund 10 per cent and municipalities with limited or no revenue raising capacity may be funded in full</li> </ul>
Allocation criteria	<ul> <li>The grant is allocated based on the location of declared disasters and an assessment of immediate needs</li> <li>The proportion of a municipality's disaster response costs funded by the grant will be determined on a case-by-case basis</li> </ul>
Reason not incorporated in equitable share	This grant caters for responses to unforeseen disasters
Past performance	<ul> <li>2009/10 audited financial outcomes</li> <li>New grant introduced in 2011/12</li> </ul>
	2009/10 service delivery performance
Projected life	New grant introduced in 2011/12  This great is appropriate continue country and will be subject to review.
	This grant is expected to continue over the medium term and will be subject to review  2011/12 P.750   W.   2012/12 P.250   W.    2011/12 P.750   W.
MTEF allocations	• 2011/12: R470 million, 2012/13: R330 million and 2013/14: R350 million
Payment schedule	Transfers are made subject to approval by National Treasury
Responsibilities of the transferring national officer and receiving officer	<ul> <li>Responsibilities of national department</li> <li>Advise municipalities about the existence of the grant and how grant funding can be applied for</li> <li>Develop a guideline on the items that will qualify for funding through this grant</li> <li>Establish procedures for funding items already purchased by municipalities</li> <li>Together with the affected municipalities and provinces, conduct preliminary assessments of disaster impacts to verify the applications for funding as per the requirements of the Disaster Management Act</li> <li>Seek approval from National Treasury for disbursement of funds to municipalities and provide written advice on the timing of disbursements to municipalities and transfer these funds to municipalities within five days of drawing the funds from the National Revenue Fund</li> <li>Notify the relevant municipality of a transfer at least one day before transfer and transfer the funds no later than five days after notification</li> <li>Notify the relevant Provincial Disaster Management Centre (PDMC) of a transfer and reason for transfer within one day of the transfer of funds to municipalities</li> <li>Build relationships and establish the necessary communication channels with relevant national departments to ensure the country has a coordinated disaster response approach</li> <li>Provide a performance report to National Treasury within 45 days after the end of the quarter during which funds are spent</li> <li>Provide National Treasury and the relevant Provincial Treasury with written notification of the transfer within 14 days of a transfer of this grant</li> <li>Responsibilities of municipalities</li> <li>Provide a performance report to the NDMC and relevant PDMC within 30 days after the end of the quarter during which funds are spent</li> <li>Initiate requests for dispeter funding when appropriete</li> </ul>
Process for approval of 2012 MTEF allocations	<ul> <li>Initiate requests for disaster funding when appropriate</li> <li>Not applicable</li> </ul>

	Municipal Infrastructure Grant
Transferring department	Cooperative Governance and Traditional Affairs (Vote 3)
Strategic goal	Subsidise the capital costs of providing basic services to poor households: priority must be given to meeting the basic infrastructure needs of poor households through the provision of appropriate bulk, connector and internal infrastructure in key services
	Distribute funding for municipal infrastructure in an equitable, transparent and efficient manner which supports a co-ordinated approach to local development and maximises developmental outcomes
	Assist in enhancing the developmental capacity of municipalities through supporting multi-year planning and budgeting systems
	Provide a mechanism for the co-ordinated pursuit of national policy priorities with regard to basic municipal infrastructure programmes, while avoiding the duplication and inefficiency associated with sectoral fragmented grants
Grant purpose	The grant is intended to provide specific capital finance for basic municipal infrastructure backlogs for poor households, micro enterprises and social institutions servicing poor communities
Outcome statements	Facilitate access to basic services infrastructure
Outputs	Number of additional poor households receiving basic water and sanitation services per annum
	<ul> <li>Number of additional kilometres of municipal roads developed</li> <li>Number of additional sport and recreation facilities servicing poor communities developed</li> </ul>
	Number of work opportunities created using Expanded Public Works Programme (EPWP) guidelines for above outputs
Priority outcome(s) of government that this grant	Outcome 9: A responsive, accountable, effective and efficient local government system
primarily contributes to	Project description (specific details, e.g. number of kilometres of roads to be constructed)
Details contained in the business plan	Project description (specific details, e.g. number of knothedes of loads to be constituted)     Project funding (Municipal Infrastructure Grant (MIG) funding, private as well as municipal own funding)
	Project category (B-Component, P-component, E-component or N-component) in terms of MIG allocation formula
	Population benefiting from the project (the grant is biased to rural poor households)
	<ul> <li>Project location (this includes GPS coordinates of the project)</li> <li>Operation and maintenance budget projections</li> </ul>
	Employment generation of the projects
	• Specific indicators (depending on level of service to be provided, e.g. VIP or full water borne sanitation)
	<ul> <li>Projects approved by council that also appear in the municipal three year capital budget or Integrated Development Plan (IDP)</li> <li>Sector departments' recommendations where applicable</li> </ul>
Conditions	Prioritise basic residential infrastructure for water, sanitation, refuse removal, streets lighting, solid
	waste, connector and internal bulk infrastructure, and other municipal infrastructure like roads, sport and recreation and community facilities in line with the MIG policy framework (2004) and/or other government sector policies existed before the amalgamation of various grants into MIG
	• Funds can be used for new or upgrading basic bulk and connector component of residential infrastructure as a result of the formalisation of settlements subject to compliance with sector policy and compliance on condition that backlogs as at 2001 has been addressed
	Compliance with Chapter 5 of the Municipal Systems Act (2000). Infrastructure investment and
	delivery must be based on an IDP, strengthened through comprehensive infrastructure planning, that provides a medium to long-term framework for sustainable human settlements and is in accordance with the principles of the National Spatial Development Perspective, Provincial Growth and
	Development Strategies and municipal council resolutions  Municipalities must adhere to labour-intensive construction methods in terms of EPWP guidelines
	Compliance with the Division of Revenue Act, including additional reporting requirements on spending and projects as approved by sector departments
	Municipalities to ensure appropriate programme and project planning and implementation readiness
	<ul> <li>Receiving officers must comply with sector conditions, norms, standards and legislation</li> <li>Funds can be used for basic new needs for upgraded and formalised settlements subject to compliance</li> </ul>
	<ul> <li>Funds can be used for basic new needs for upgraded and formalised settlements subject to compliance with sector policy and compliance</li> <li>A municipality receiving MIG must table a three year capital budget as part of its budget for the</li> </ul>
	2010/11 financial year in accordance with the MFMA, unless exempted in terms of that Act
Allocation criteria	Part 5 of Annexure W1 spells out the MIG formula in detail. The formula incorporates backlog and poverty data
	• The MIG must be transferred directly to a category B or C municipality that has the powers and functions referred to in section 84 of the Municipal Structures Act, to enable the municipality to provide municipal infrastructure in respect of these powers and functions.
	<ul> <li>municipal infrastructure in respect of those powers and functions</li> <li>The MIG allocation for a category B or C municipality may be transferred to the category C municipality or province within whose jurisdiction the municipality is situated if, in the assessment of the transferring national officer in consultation with the National Treasury, the municipality does not:         <ul> <li>have sufficient expenditure capacity to effectively carry out the infrastructure programmes</li> </ul> </li> </ul>
	- adhere to good financial governance practices

	Municipal Infrastructure Grant
Reason not incorporated in equitable share	This is a specific purpose grant with conditions, objectives and distribution criteria different from that of the equitable share
Past performance	2009/10 audited financial outcome     Allocated R11 077 million to municipalities with R9 019 million (78.9 per cent) spent by the end of the 2009/10 municipal financial year     2009/10 service delivery performance
	<ul> <li>Cumulative households benefiting from MIG by end June 2010: water 1 232 643, sanitation 703 121, storm water 166 151, solid waste 312 424, street/community lighting 391 164</li> <li>21 990.51km of roads developed</li> </ul>
	A cumulative total of 3 934 SMMEs and 2 436 BEE companies were utilised in the implementation of MIG projects
Projected life	The programme will continue up to 2013 subject to review
MTEF allocations	• 2011/12: R11 444 million, 2012/13: R13 914 million and 2013/14: R14 679 million
Payment schedule	Transfers are made in accordance with a payment schedule approved by the National Treasury
Responsibilities of the transferring national officer and receiving	Responsibilities of national departments  Submit a report within 20 days after the end of each month to National Treasury and other national departments that have responsibilities relating to the grant. The report must include:
officer	<ul> <li>expenditure and non-financial performance information on programmes funded by an allocation</li> <li>an explanation of any material problems experienced by municipalities regarding an allocation that has been received and a summary of the interventions or steps taken to deal with such problems</li> <li>Department of Cooperative Governance and Traditional Affairs (CoGTA) administers the municipal infrastructure grant and co-ordinates its activities with all stakeholders through appropriate structures</li> <li>Department of Water Affairs and the Department of Human Settlements to support and monitor municipalities to prepare and implement Water Services Development Plans (WSDPs), and monitor and oversee progress on water and sanitation projects implemented through the MIG grant</li> <li>Department of Public Works to monitor compliance with the EPWP</li> <li>Sport and Recreation South Africa (SRSA) to assist municipalities with planning of sport and recreation facilities and monitor implementation</li> <li>Each sector, national or provincial department of local government will be expected to fulfil a sectoral and provincial monitoring role on relevant sectoral and provincial outputs</li> <li>CoGTA coordinates monitoring systems and the overall programme implementation</li> </ul>
	Responsibilities of provincial departments  Coordinating municipal reports  Providing and coordinating support and capacity to municipalities  Submit reports to national CoGTA  Monitor project implementation in collaboration with sectors and submit site visit reports to CoGTA
	<ul> <li>Monitor compliance with provincial legislation and alignment to Provincial Growth and Development Strategies through project registration</li> <li>Ensure that municipalities implement projects in accordance with the municipal IDPs and council resolutions</li> <li>Monitor performance of municipal Project Management Units and recommend relevant sanctions for under-performance to CoGTA</li> <li>The provincial department must provide final recommendations to receiving officer by 30 September 2011</li> </ul>
	Responsibilities of municipalities     The receiving officer of MIG must, by 2 August 2011, submit all the project registrations forms for the projects to be implemented in 2011/12 to the provincial department responsible for local government     Municipalities must submit to the national department by 28 October 2011, detailed project implementation plans for all the projects to be implemented in the 2012/13 financial year. Such plans should include timelines regarding project designs, initiation of procurement, and Environmental Impact Assessment (EIA) approvals
Process for approval of 2012/13 business plans	The receiving officer of a MIG must, by 27 May 2011, submit all technical reports to the sector departments responsible for water services, sport and recreation, roads and transport for all projects to be implemented in 2012/13  The responsible sector department must evaluate reports and provide final recommendations to the receiving officer by 29 July 2011

Strategic goal  An efficient and developmental sphere of local government capable of delivering services to local communities  To assist municipalities in building in-house capacity to perform their functions and stabilise institutional and governance systems as required in the Municipal Systems Act (MSA) and related legislation, policies and local government turnaround strategy  A responsive, accountable, effective and efficient local government system  Strengthen the ward participation systems in local government system  Number of municipalities improving their viability and management through targeted support interventions  Number of municipalities improving their viability and management through targeted support interventions  Number of municipalities with strengthened administrative systems enabling effective implementation of the ward participation system  Number of municipalities with information systems that support local government legislation  Number of municipalities with information systems that support effective service delivery  Number of municipalities with information systems that support effective service delivery environment (specific municipalities with information systems that support legislation  Number of municipalities with information systems that support effective service delivery environment (specific municipalities and developed turnaround strategies and are implementing the strategies  Priority outcome(s) of government that this grant of municipalities were supported internations of prioritised measurable outputs  Allocation criteria  Allocations of a signed activity plan in a prescribed format with detailed budgets and timeframes for the implementation of prioritised measurable outputs  Mulceation criteria  Reasons not incorporated in emancipalities must adhere to the local government turnaround strategy  Allocations are made to selected municipalities turnaround strategy  Allocations are made to selected municipalities with R186 million (94.3 per cent) spent by the end of the 2009/		Municipal Systems Improvement Grant
Communities	Transferring department	Cooperative Governance and Traditional Affairs (Vote 3)
institutional and governance systems as required in the Municipal Systems Act (MSA) and related legislation, policies and local government unraround strategy.  Outcome statements  Polity in the control of municipalities improving their viability and management through targeted support interventions.  Number of municipalities improving their viability and management through targeted support interventions.  Number of municipalities with strengthened administrative systems enabling effective implementation of the ward participation systems.  Number of municipalities with strengthened administrative systems enabling effective implementation of the ward participation system.  Number of municipalities improving their viability and management through targeted support interventions.  Number of municipalities with strengthened administrative systems enabling effective implementation of the ward participation system.  Number of municipalities with information systems that support local government legislation.  Number of municipalities with information systems that support effective service delivery.  Number of municipalities with information systems that support effective service delivery.  Number of municipalities with information systems that support effective service delivery.  Number of municipalities with developed turnaround strategies and are implementing the strategies of the surface of the surf	Strategic goal	
Duttome statements  A responsive, accountable, effective and efficient local government ystem  Strengthen the ward participation systems in local government  Effective implementation of municipalities upworing their viability and management through targeted support interventions  Number of municipalities with strengthened administrative systems enabling effective implementation of the ward participation system  Number of municipalities with strengthened administrative systems enabling effective implementation of the ward participation system  Number of municipalities with information systems that support local government legislation  Number of municipalities with information systems that support local government legislation  Number of municipalities with information systems that support effective service delivery  Number of municipalities with information systems that support effective service delivery  Number of municipalities with developed turnaround strategies and are implementing the strategies  Priority outcome(s) of covernment that this grant or function of prioritised measurable outputs  Conditions  Outcome 9. A responsive, accountable, effective and efficient local government system or sex sex services of the implementation of prioritised measurable outputs  Numicipalities must after to the local government turnaround strategy  Allocation criteria  A allocation sare made to selected municipalities based on previous expenditure performance and assessed priority needs  Part performance  Part	Grant purpose	
A responsive, accountable, effective and efficient local government system		
Strengthen the ward participation systems in local government Effective implementation of municipalities improving their viability and management through targeted support interventions  Number of municipalities with strengthened administrative systems enabling effective implementation of the ward participation system  Number of municipalities with strengthened administrative systems enabling effective implementation of the ward participation systems that support local government legislation  Number of municipalities with information systems that support effective service delivery  Number of municipalities with information systems that support effective service delivery  Number of municipalities with information systems that support effective service delivery  Number of municipalities with evolve and efficient local government system or interventions.  Priority outcome(s) of outcome indicators outcomes and expensive, accountable, effective and efficient local government system or interventions.  Outcome indicators  Output indicators  Output indicators  Number of municipalities and evolve and efficient local government system or the business plan  Output indicators  Submission of a signed activity plan in a prescribed format with detailed budgets and timeframes for the implementation of prioritised measurable outputs  Municipalities must athere to the local government turnaround strategy  Allocation criteria  Easons not incorporated in expensive expensive to the local government strategy  Past performance  Past performance  Past performance  Past performance  Past performance  2009/10 andited financial outcomes  Allocated R200 million to municipalities with R186 million (94.3 per cent) spent by the end of the 2009/10 municipal financial year  2009/10 arcticle delivery performance expense required in terms of local government legislation  1 allocated R200 million to municipalities with R186 million (94.3 per cent) spent by the end of the 2009/10 municipal financial year  2009/10 acticle financial outcomes  Alloca	Outcome statements	
Dutputs    Effective implementation of municipality-specific turnaround strategies   Number of municipalities improving their viability and management through targeted support interventions   Number of municipalities with strengthened administrative systems enabling effective implementation of the ward participation system   Number of municipalities implementing by-laws, policies and systems that support floctive service delivery   Number of municipalities with information systems that support floctive service delivery of municipalities that developed turnaround strategies and are implementing the strategies   Number of municipalities and developed turnaround strategies and are implementing the strategies   Outcome 9: A responsive, accountable, effective and efficient local government system   Outcome 9: A responsive, accountable, effective and efficient local government system   Outcome 9: A responsive, accountable, effective and efficient local government system   Outcome 9: A responsive, accountable, effective and efficient local government system   Outcome 9: A responsive, accountable, effective and efficient local government system   Outcome 9: A responsive, accountable, effective and efficient local government system   Outcome 9: A responsive, accountable, effective and efficient local government system   Outcome 9: A responsive, accountable, effective and efficient local government system   Outcome 9: A responsive, accountable, effective and efficient local government system   Outcome 9: A responsive, accountable, effective and efficient local government system   Outcome 10: A system	Succine statements	
Number of municipalities improving their viability and management through targeted support interventions		
Number of municipalities with strengthened administrative systems enabling effective implementation of the ward participation system   Number of municipalities implementing by-laws, policies and systems that support local government legislation   Number of municipalities that developed turnaround strategies and are implementing the strategies   Priority outcome(s) of government that this grant primarily contributes to	Outputs	Number of municipalities improving their viability and management through targeted support
implementation of the ward participation system  Number of municipalities implementing by-laws, policies and systems that support local government legislation  Number of municipalities with information systems that support effective service delivery  Number of municipalities that developed turnaround strategies and are implementing the strategies  Outcome 9: A responsive, accountable, effective and efficient local government system potentials contained in the pusiness plan  Outcome indicators  Every activities  Inputs  Conditions  Outcome indicators  Submission of a signed activity plan in a prescribed format with detailed budgets and timeframes for the implementation of prioritised measurable outputs  Submission of a signed activity plan in a prescribed format with detailed budgets and timeframes for the implementation of prioritised measurable outputs  Allocation criteria  Allocations are made to selected municipalities based on previous expenditure performance and assessed priority needs  Reasons not incorporated in exputable share  Past performance  The grant is conditional and aimed at building the capacity of municipalities to implement sound institutional and governance systems required in terms of local government legislation  The grant is conditional and aimed at building the capacity of municipalities to implement sound institutional and governance systems required in terms of local government legislation  2009/10 audited financial outcomes  Allocated R200 million to municipalities with R186 million (94.3 per cent) spent by the end of the 2009/10 municipal financial year  2009/10 service delivery performance  2009/10 service able to fully implement the Municipal Property Rates Act  68 municipalities were supported by the grant to review policies  230 municipalities were supported by the grant to review policies  230 municipalities were supported by the grant to review policies  230 municipalities will practice (GRAP)  Projected life  Transfer to be made in July 2011 in accordance with payment schedule		
Number of municipalities implementing by-laws, policies and systems that support local government legislation Number of municipalities with information systems that support effective service delivery Number of municipalities that developed turnaround strategies and are implementing the strategies Outcome 9: A responsive, accountable, effective and efficient local government system obtails contained in the outsiness plan  Outcome indicators Output indicators Submission of a signed activity plan in a prescribed format with detailed budgets and timeframes for the implementation of prioritised measurable outputs Municipalities must adhere to the local government turnaround strategy Allocations are made to selected municipalities based on previous expenditure performance and assessed priority needs  The grant is conditional and aimed at building the capacity of municipalities to implement sound institutional and governance systems required in terms of local government legislation  2009/10 audited financial outcomes Allocated R200 million to municipalities with R186 million (94.3 per cent) spent by the end of the 2009/10 service delivery performance  2009/10 s		
legislation  Number of municipalities with information systems that support effective service delivery  Number of municipalities that developed turnaround strategies and are implementing the strategies  Outcome 9: A responsive, accountable, effective and efficient local government system  Details contained in the pusiness plan  Outcome indicators  Output indicators  Submission of a signed activity plan in a prescribed format with detailed budgets and timeframes for the implementation of prioritised measurable outputs  Municipalities must adhere to the local government turnaround strategy  Allocation criteria  Seasons not incorporated in equitable share  Past performance  Past performance  Past performance  Allocated R200 million to municipalities with R186 million (94.3 per cent) spent by the end of the 2009/10 audited financial outcomes  Allocated R200 million to municipalities with R186 million (94.3 per cent) spent by the end of the 2009/10 sumicipalities were supported with this grant to establish effective ward systems for public and community participation  1 33 municipalities were supported by the grant to review policies  2 200 municipalities were supported by the grant to review policies  2 30 municipalities were supported by the grant to review during the 2011/12 financial year  Projected life  Transfers to be made in July 2011 in accordance with payment schedule approved by National Treasury  Responsibilities of the ransferring national officer and receiving officer  Nonitoring of expenditure on the grant  Visits to slow spending and non-reporting municipalities to determine challenges and to provide required support  Implementation of the monthly expenditure report tracking system  Analysis of monthly expenditure report tracking system  Analysis of municipalities  Submit quarterly and annual performance reports to National Treasury  Responsibilities of municipalities  Submit quarterly and annual performance reports to National Treasury  Responsibilities of municipalities  Submit activity plans in th		
Priority outcome(s) of povernment that this grant primarily contributes to Details contained in the pusiness plan  Outcome 9: A responsive, accountable, effective and efficient local government system pusiness plan  Outcome indicators  Output indicators  Number of municipalities with a developed turnaround strategies and are implementing the strategies of the pusiness plan  Outcome indicators  Output indicators  New activities Inputs  Conditions  Submitsion of a signed activity plan in a prescribed format with detailed budgets and timeframes for the implementation of prioritised measurable outputs  Municipalities must adhere to the local government turnaround strategy  Mulcation criteria  Reasons not incorporated in equitable share  Past performance  Past performance  Allocation are made to selected municipalities based on previous expenditure performance and assessed priority needs  Reasons not incorporated in equitable share  Past performance  Allocation and governance systems required in terms of local government legislation  Payment schedule  Allocated R200 million to municipalities with R186 million (94.3 per cent) spent by the end of the 2009/10 municipalities were supported with this grant to establish effective ward systems for public and community participation  133 municipalities were supported by the grant to review policies  230 municipalities were supported by the grant to review policies  230 municipalities were supported by the grant to review during the 2011/12 financial year  MTEF allocations  Payment schedule  Transfers to be made in July 2011 in accordance with payment schedule approved by National Treasury  Responsibilities of the transferring national officer and receiving officer  Monitoring of expenditure on the grant  Visits to slow spending and non-reporting municipalities to determine challenges and to provide required support  Implementation of the monthly expenditure report tracking system  Analysis of monthly expenditure reports from municipalities  Submit quarterly and annual perfo		
Priority outcome(s) of covernment that this grant primarily contributes to Details contained in the business plan  Potatils contained in the business plan  Conditions  Condit		
powerment that this grant orimarily contributes to Details contained in the pusiness plan  - Outcome indicators - Output indicators - Ney activities - Inputs  - Submission of a signed activity plan in a prescribed format with detailed budgets and timeframes for the implementation of prioritised measurable outputs - Municipalities must adhere to the local government turnaround strategy  - Allocation criteria - Allocations are made to selected municipalities based on previous expenditure performance and assessed priority needs  - Reasons not incorporated in equitable share - Past performance  - Past performance - Allocated R 200 million to municipalities with R186 million (94.3 per cent) spent by the end of the 2009/10 service delivery performance - 2009/10 service delivery performance - 200 municipalities were supported with this grant to establish effective ward systems for public and community participation - 133 municipalities were supported with this grant to establish effective ward systems for public and community participation - 133 municipalities were supported by the grant to review policies - 230 municipalities were supported by the grant to review policies - 230 municipalities received support to identify, verify and finalise fixed asset registers in line with the Generally Recognised Accounting Practice (GRAP)  - The grant continues over the MTEF period and will be reviewed during the 2011/12 financial year  - WITEF allocations - Payment schedule - Transfers to be made in July 2011 in accordance with payment schedule approved by National Treasury  - Responsibilities of the ransferring national officer and receiving officer - Monitoring of expenditure on the grant - Visits to slow spending and non-reporting municipalities - Monitoring of expenditure on the grant - Visits to slow spending and non-reporting municipalities - Monitoring of expenditure on the grant - Visits to slow spending and non-reporting municipalities - Monitoring of expenditure on the grant - Visits to slow spending and non-reporting		
Details contained in the pusiness plan  - Outcome indicators - Output indicators - Ney activities - Inputs  - Submission of a signed activity plan in a prescribed format with detailed budgets and timeframes for the implementation of prioritised measurable outputs - Municipalities must adhere to the local government turnaround strategy  - Allocations are made to selected municipalities based on previous expenditure performance and assessed priority needs - The grant is conditional and aimed at building the capacity of municipalities to implement sound institutional and governance systems required in terms of local government legislation  - Past performance  - Allocated R200 multilion to municipalities with R186 million (94.3 per cent) spent by the end of the 2009/10 audited financial outcomes - Allocated R200 municipalities were supported with this grant to establish effective ward systems for public and community participation - 133 municipalities were supported with this grant to establish effective ward systems for public and community participation - 133 municipalities were supported by the grant to review policies - 230 municipalities were supported by the grant to review policies - 230 municipalities received support to identify, verify and finalise fixed asset registers in line with the Generally Recognised Accounting Practice (GRAP)  - Projected life - The grant continues over the MTEF period and will be reviewed during the 2011/12 financial year  - Witst to slow spending and non-reporting municipalities to determine challenges and to provide required support - Minimal properties of the ransferring national officer and receiving officer - Monitoring of expenditure on the grant - Visits to slow spending and non-reporting municipalities - Monitoring of expenditure on the grant - Visits to slow spending and non-reporting municipalities - Minimal activity plans in the prescribed format	Priority outcome(s) of	Outcome 9: A responsive, accountable, effective and efficient local government system
Outcome indicators  Output indicators  (Key activities Inputs  Conditions  Submission of a signed activity plan in a prescribed format with detailed budgets and timeframes for the implementation of prioritised measurable outputs  Municipalities must adhere to the local government turnaround strategy  Allocation criteria  Allocations are made to selected municipalities based on previous expenditure performance and assessed priority needs  Reasons not incorporated in equitable share  Past performance  Allocation and a aimed at building the capacity of municipalities to implement sound institutional and governance systems required in terms of local government legislation  2009/10 audited financial outcomes  Allocated R200 million to municipalities with R186 million (94.3 per cent) spent by the end of the 2009/10 municipal financial year  2009/10 service delivery performance  2000 municipalities were supported with this grant to establish effective ward systems for public and community participation  1 33 municipalities were supported by the grant to review policies  2 30 municipalities were supported by the grant to review policies  2 30 municipalities were supported by the grant to review policies  2 30 municipalities of the grant sound and a service of the grant sound in terms of local government legislation  Payment schedule  Transferring national officer  Responsibilities of the grant sound and payment schedule approved by National Treasury  Responsibilities of the grant sound and payment schedule approved by National Treasury  Responsibilities of mational department  Visits to slow spending and non-reporting municipalities to determine challenges and to provide required support  Implementation of the monthly expenditure report tracking system  Analysis of monthly expenditure reports from municipalities  Submit quarterly and annual performance reports to National Treasury  Responsibilities of municipalities  Submit activity plans in the prescribed format		
Output indicators		Outcome indicators
Key activities   Inputs		
Conditions  - Submission of a signed activity plan in a prescribed format with detailed budgets and timeframes for the implementation of prioritised measurable outputs  - Municipalities must adhere to the local government turnaround strategy  - Allocations are made to selected municipalities based on previous expenditure performance and assessed priority needs  - Reasons not incorporated in equitable share  - Past performance  - The grant is conditional and aimed at building the capacity of municipalities to implement sound institutional and governance systems required in terms of local government legislation  - Allocation and an agovernance systems required in terms of local government legislation  - Allocated R200 million to municipalities with R186 million (94,3 per cent) spent by the end of the 2009/10 service delivery performance  - 2009/10 municipal financial year  - 2009/10 service delivery performance  - 2009 municipalities were supported with this grant to establish effective ward systems for public and community participation  - 133 municipalities were supported by the grant to review policies  - 230 municipalities were supported by the grant to review policies  - 230 municipalities received support to identify, verify and finalise fixed asset registers in line with the Generally Recognised Accounting Practice (GRAP)  - The grant continues over the MTEF period and will be reviewed during the 2011/12 financial year  - WITEF allocations  - 2011/12: R219 million, 2012/13: R230 million and 2013/14: R243 million  - Transfers to be made in July 2011 in accordance with payment schedule approved by National Treasury  - Responsibilities of national department  - Wisits to slow spending and non-reporting municipalities to determine challenges and to provide required support  - Monitoring of expenditure on the grant  - Visits to slow spending and non-reporting municipalities  - Submit quarterly and annual performance reports to National Treasury  - Responsibilities of municipalities  - Submit quarterly and annua		*
the implementation of prioritised measurable outputs  Municipalities must adhere to the local government turnaround strategy  Allocation criteria  Allocations are made to selected municipalities based on previous expenditure performance and assessed priority needs  The grant is conditional and aimed at building the capacity of municipalities to implement sound institutional and governance systems required in terms of local government legislation  2009/10 audited financial outcomes  Allocated R200 million to municipalities with R186 million (94.3 per cent) spent by the end of the 2009/10 service delivery performance  2009/10 service		
Allocation criteria  Allocations are made to selected municipalities based on previous expenditure performance and assessed priority needs  Reasons not incorporated in equitable share  Past performance  Allocated R200 million to municipalities with R186 million (94.3 per cent) spent by the end of the 2009/10 audited financial outcomes  Allocated R200 million to municipalities with R186 million (94.3 per cent) spent by the end of the 2009/10 service delivery performance  2009/10 service delivery performance  2009/10 service delivery performance  2000 municipalities were supported with this grant to establish effective ward systems for public and community participation  133 municipalities were able to fully implement the Municipal Property Rates Act  68 municipalities were supported by the grant to review policies  230 municipalities were supported by the grant to review policies  230 municipalities received support to identify, verify and finalise fixed asset registers in line with the Generally Recognised Accounting Practice (GRAP)  Projected life  The grant continues over the MTEF period and will be reviewed during the 2011/12 financial year  MTEF allocations  Payment schedule  Transfers to be made in July 2011 in accordance with payment schedule approved by National Treasury  Responsibilities of the transferring national officer and receiving officer  MREADOR AND	Conditions	
Allocation criteria  • Allocations are made to selected municipalities based on previous expenditure performance and assessed priority needs  • The grant is conditional and aimed at building the capacity of municipalities to implement sound institutional and governance systems required in terms of local government legislation  2009/10 audited financial outcomes • Allocated R200 million to municipalities with R186 million (94.3 per cent) spent by the end of the 2009/10 service delivery performance • 2009/10 service delivery performance • 200 municipalities were supported with this grant to establish effective ward systems for public and community participation • 133 municipalities were supported by the grant to review policies • 230 municipalities received support to identify, verify and finalise fixed asset registers in line with the Generally Recognised Accounting Practice (GRAP)  Projected life • The grant continues over the MTEF period and will be reviewed during the 2011/12 financial year  MTEF allocations  Payment schedule • Transfers to be made in July 2011 in accordance with payment schedule approved by National Treasury  Responsibilities of the transferring national officer and receiving officer  Which is to slow spending and non-reporting municipalities to determine challenges and to provide required support  MITEF allocations  • Monitoring of expenditure on the grant • Visits to slow spending and non-reporting municipalities to determine challenges and to provide required support  Monitoring of expenditure reports from municipalities  • Analysis of monthly expenditure report tracking system  • Analysis of monthly expenditure reports to National Treasury  Responsibilities of municipalities  • Submit activity plans in the prescribed format		
Reasons not incorporated in equitable share  Past performance  2009/10 audited financial outcomes  Allocated R200 million to municipalities with R186 million (94.3 per cent) spent by the end of the 2009/10 municipalities were supported with this grant to establish effective ward systems for public and community participation  133 municipalities were supported with this grant to review policies  68 municipalities were supported by the grant to review policies  133 municipalities were supported by the grant to review policies  230 municipalities received support to identify, verify and finalise fixed asset registers in line with the Generally Recognised Accounting Practice (GRAP)  Projected life  The grant continues over the MTEF period and will be reviewed during the 2011/12 financial year  MTEF allocations  Payment schedule  Transfers to be made in July 2011 in accordance with payment schedule approved by National Treasury  Responsibilities of the transferring national officer and receiving officer  Which is to slow spending and non-reporting municipalities to determine challenges and to provide required support  Monitoring of expenditure on the grant  Visits to slow spending and non-reporting municipalities to determine challenges and to provide required support  Implementation of the monthly expenditure reports from municipalities  Submit quarterly and annual performance reports to National Treasury  Responsibilities of municipalities  Submit activity plans in the prescribed format		
Past performance  **The grant is conditional and aimed at building the capacity of municipalities to implement sound institutional and governance systems required in terms of local government legislation  **2009/10 audited financial outcomes**  **Allocated R200 million to municipalities with R186 million (94.3 per cent) spent by the end of the 2009/10 service delivery performance**  **2009/10 service deli	Allocation criteria	
Past performance  Allocated R 200 million to municipalities with R186 million (94.3 per cent) spent by the end of the 2009/10 municipal financial year  2009/10 service delivery performance  200 municipalities were supported with this grant to establish effective ward systems for public and community participation  133 municipalities were able to fully implement the Municipal Property Rates Act  68 municipalities were supported by the grant to review policies  230 municipalities received support to identify, verify and finalise fixed asset registers in line with the Generally Recognised Accounting Practice (GRAP)  Projected life  The grant continues over the MTEF period and will be reviewed during the 2011/12 financial year  MTEF allocations  Payment schedule  Responsibilities of the transferring national officer and receiving officer  Responsibilities of national department  Wisits to slow spending and non-reporting municipalities to determine challenges and to provide required support  Monitoring of expenditure on the grant  Visits to slow spending and non-reporting municipalities  Monitoring of expenditure reports from municipalities  Monitoring of expenditure reports from municipalities  Submit quarterly and annual performance reports to National Treasury  Responsibilities of municipalities  Submit activity plans in the prescribed format	Reasons not incorporated in	* *
Allocated R200 million to municipalities with R186 million (94.3 per cent) spent by the end of the 2009/10 municipal financial year  2009/10 service delivery performance     200 municipalities were supported with this grant to establish effective ward systems for public and community participation     133 municipalities were able to fully implement the Municipal Property Rates Act     68 municipalities were supported by the grant to review policies     230 municipalities received support to identify, verify and finalise fixed asset registers in line with the Generally Recognised Accounting Practice (GRAP)  Projected life     The grant continues over the MTEF period and will be reviewed during the 2011/12 financial year  MTEF allocations Payment schedule  Transfers to be made in July 2011 in accordance with payment schedule approved by National Treasury  Responsibilities of the transferring national officer and receiving officer  MRESPONSIBILITIES of national department  MRESPONSIBILITIES of municipalities	equitable share	
2009/10 municipal financial year 2009/10 service delivery performance  2000 municipalities were supported with this grant to establish effective ward systems for public and community participation  133 municipalities were able to fully implement the Municipal Property Rates Act  68 municipalities were supported by the grant to review policies  230 municipalities received support to identify, verify and finalise fixed asset registers in line with the Generally Recognised Accounting Practice (GRAP)  Projected life  The grant continues over the MTEF period and will be reviewed during the 2011/12 financial year  2011/12: R219 million, 2012/13: R230 million and 2013/14: R243 million  Payment schedule  Transfers to be made in July 2011 in accordance with payment schedule approved by National Treasury  Responsibilities of national department  Monitoring of expenditure on the grant  Visits to slow spending and non-reporting municipalities to determine challenges and to provide required support  Implementation of the monthly expenditure report tracking system  Analysis of monthly expenditure reports from municipalities  Submit quarterly and annual performance reports to National Treasury  Responsibilities of municipalities  Submit activity plans in the prescribed format	Past performance	
2009/10 service delivery performance  200 municipalities were supported with this grant to establish effective ward systems for public and community participation  133 municipalities were able to fully implement the Municipal Property Rates Act  68 municipalities were supported by the grant to review policies  230 municipalities received support to identify, verify and finalise fixed asset registers in line with the Generally Recognised Accounting Practice (GRAP)  Projected life  The grant continues over the MTEF period and will be reviewed during the 2011/12 financial year  2011/12: R219 million, 2012/13: R230 million and 2013/14: R243 million  Payment schedule  Transfers to be made in July 2011 in accordance with payment schedule approved by National Treasury  Responsibilities of the transferring national officer and receiving officer  When the transfer in patient of the monthly expenditure on the grant  Nonitoring of expenditure on the grant  Visits to slow spending and non-reporting municipalities to determine challenges and to provide required support  Implementation of the monthly expenditure report tracking system  Analysis of monthly expenditure reports from municipalities  Submit quarterly and annual performance reports to National Treasury  Responsibilities of municipalities  Submit activity plans in the prescribed format		
200 municipalities were supported with this grant to establish effective ward systems for public and community participation     133 municipalities were able to fully implement the Municipal Property Rates Act     68 municipalities were supported by the grant to review policies     230 municipalities received support to identify, verify and finalise fixed asset registers in line with the Generally Recognised Accounting Practice (GRAP)  Projected life  The grant continues over the MTEF period and will be reviewed during the 2011/12 financial year  2011/12: R219 million, 2012/13: R230 million and 2013/14: R243 million  Payment schedule  Transfers to be made in July 2011 in accordance with payment schedule approved by National Treasury  Responsibilities of the transferring national officer and receiving officer  Monitoring of expenditure on the grant  Visits to slow spending and non-reporting municipalities to determine challenges and to provide required support  Implementation of the monthly expenditure report tracking system  Analysis of monthly expenditure reports from municipalities  Submit quarterly and annual performance reports to National Treasury  Responsibilities of municipalities  Submit activity plans in the prescribed format		
community participation  133 municipalities were able to fully implement the Municipal Property Rates Act  68 municipalities were supported by the grant to review policies  230 municipalities received support to identify, verify and finalise fixed asset registers in line with the Generally Recognised Accounting Practice (GRAP)  Projected life  The grant continues over the MTEF period and will be reviewed during the 2011/12 financial year  MTEF allocations  Payment schedule  Transfers to be made in July 2011 in accordance with payment schedule approved by National Treasury  Responsibilities of the transferring national officer and receiving officer  Monitoring of expenditure on the grant  Monitoring of expenditure on the grant  Wisits to slow spending and non-reporting municipalities to determine challenges and to provide required support  Implementation of the monthly expenditure report tracking system  Analysis of monthly expenditure reports from municipalities  Submit quarterly and annual performance reports to National Treasury  Responsibilities of municipalities  Submit activity plans in the prescribed format		· · · · · · · · · · · · · · · · · · ·
• 68 municipalities were supported by the grant to review policies     • 230 municipalities received support to identify, verify and finalise fixed asset registers in line with the Generally Recognised Accounting Practice (GRAP)  Projected life  The grant continues over the MTEF period and will be reviewed during the 2011/12 financial year  2011/12: R219 million, 2012/13: R230 million and 2013/14: R243 million  Transfers to be made in July 2011 in accordance with payment schedule approved by National Treasury  Responsibilities of the transferring national officer and receiving officer  Which is a solow spending and non-reporting municipalities to determine challenges and to provide required support  Implementation of the monthly expenditure report tracking system  Analysis of monthly expenditure reports from municipalities  Submit quarterly and annual performance reports to National Treasury  Responsibilities of municipalities  Submit activity plans in the prescribed format		
the Generally Recognised Accounting Practice (GRAP)  The grant continues over the MTEF period and will be reviewed during the 2011/12 financial year  MTEF allocations  Payment schedule  Transfers to be made in July 2011 in accordance with payment schedule approved by National Treasury  Responsibilities of the transferring national officer and receiving officer  Monitoring of expenditure on the grant  Visits to slow spending and non-reporting municipalities to determine challenges and to provide required support  Implementation of the monthly expenditure report tracking system  Analysis of monthly expenditure reports from municipalities  Submit quarterly and annual performance reports to National Treasury  Responsibilities of municipalities  Submit activity plans in the prescribed format		
Projected life  The grant continues over the MTEF period and will be reviewed during the 2011/12 financial year  Payment schedule  Transfers to be made in July 2011 in accordance with payment schedule approved by National Treasury  Responsibilities of the transferring national officer and receiving officer  Which is to slow spending and non-reporting municipalities to determine challenges and to provide required support  Implementation of the monthly expenditure report tracking system  Analysis of monthly expenditure reports from municipalities  Submit quarterly and annual performance reports to National Treasury  Responsibilities of municipalities  Submit activity plans in the prescribed format		
### Analysis of monthly expenditure reports from municipalities    Submit quarterly and annual performance reports to National Treasury    Submit activity plans in the prescribed format    Submit activity plans in the prescribed format   Payment 2013/14: R243 million   Responsibilition and 2013/14: R243 millio	Projected life	
Payment schedule  Responsibilities of the transferring national officer and receiving officer  When the distribution is a continuation of the transferring national officer and receiving officer  Responsibilities of national department  Monitoring of expenditure on the grant  Visits to slow spending and non-reporting municipalities to determine challenges and to provide required support  Implementation of the monthly expenditure report tracking system  Analysis of monthly expenditure reports from municipalities  Submit quarterly and annual performance reports to National Treasury  Responsibilities of municipalities  Submit activity plans in the prescribed format		
Responsibilities of the transferring national officer and receiving officer  **Monitoring of expenditure on the grant*  **Visits to slow spending and non-reporting municipalities to determine challenges and to provide required support*  **Implementation of the monthly expenditure report tracking system*  **Analysis of monthly expenditure reports from municipalities*  **Submit quarterly and annual performance reports to National Treasury*  Responsibilities of municipalities*  **Submit activity plans in the prescribed format*		·
<ul> <li>Monitoring of expenditure on the grant</li> <li>Visits to slow spending and non-reporting municipalities to determine challenges and to provide required support</li> <li>Implementation of the monthly expenditure report tracking system</li> <li>Analysis of monthly expenditure reports from municipalities</li> <li>Submit quarterly and annual performance reports to National Treasury</li> <li>Responsibilities of municipalities</li> <li>Submit activity plans in the prescribed format</li> </ul>		
<ul> <li>Visits to slow spending and non-reporting municipalities to determine challenges and to provide required support</li> <li>Implementation of the monthly expenditure report tracking system</li> <li>Analysis of monthly expenditure reports from municipalities</li> <li>Submit quarterly and annual performance reports to National Treasury</li> <li>Responsibilities of municipalities</li> <li>Submit activity plans in the prescribed format</li> </ul>	Responsibilities of the	
required support  Implementation of the monthly expenditure report tracking system  Analysis of monthly expenditure reports from municipalities  Submit quarterly and annual performance reports to National Treasury  Responsibilities of municipalities  Submit activity plans in the prescribed format	transferring national officer	
<ul> <li>Implementation of the monthly expenditure report tracking system</li> <li>Analysis of monthly expenditure reports from municipalities</li> <li>Submit quarterly and annual performance reports to National Treasury</li> <li>Responsibilities of municipalities</li> <li>Submit activity plans in the prescribed format</li> </ul>	and receiving officer	
<ul> <li>Analysis of monthly expenditure reports from municipalities</li> <li>Submit quarterly and annual performance reports to National Treasury</li> <li>Responsibilities of municipalities</li> <li>Submit activity plans in the prescribed format</li> </ul>		1 11
<ul> <li>Submit quarterly and annual performance reports to National Treasury</li> <li>Responsibilities of municipalities</li> <li>Submit activity plans in the prescribed format</li> </ul>		
Responsibilities of municipalities  • Submit activity plans in the prescribed format		
, , , , , , , , , , , , , , , , , , , ,		• Submit monthly expenditure reports, in the prescribed format, 20 working days after the end of
every month	D	
	Process for approval of 2012/13 business plan	
2012/13 business plan  - activity plan format guidelines, criteria and outputs sent to municipalities by 28 January 2012 - submission of business/activity plans by municipalities by February/March 2012	2012/13 Dusiness Plan	
- appraisal and approval of business/activity plans by April/May 2012		

### PUBLIC WORKS GRANT

Transferring department	Expanded Public Works Programme Incentive Grant for Municipalities     Public Works (Vote 7)
Strategic goal	
Strategic goar	To increase the amount of full-time equivalent jobs created by municipalities through the use of labour-intensive methods
Grant purpose	To incentivise municipalities to increase job creation efforts in infrastructure, environment and
	culture programmes through the use of labour-intensive methods and the expansion of job creation in
	line with the Expanded Public Works Programme (EPWP) guidelines
Outcome statements	Improved quality of life of unemployed people and increased social stability through engaging the
	unemployed in paid and productive activities
	Reduced levels of poverty
	Contribute to increased levels of employment
Outputs	Improved opportunities for sustainable work through experience and learning gained    Improved opportunities for sustainable work through experience and learning gained
Outputs	<ul> <li>Increased number of people employed and receiving income through the EPWP</li> <li>Increased average duration of the work opportunities created</li> </ul>
	Increased average dutation of the work opportunities created     Increased income per EPWP beneficiary
	Increase in the number of municipalities reporting on EPWP
	Increased total number of person days of employment created
Priority outcome(s) of	Outcome 4: Decent employment through inclusive growth
government that this grant	
primarily contributes to	
Details contained in the	This grant uses an incentive agreement which contains the following:
business plans	Full Time Equivalent (FTE) thresholds for each eligible municipality
	FTE target for each eligible municipality
Conditions	Eligible municipalities must sign a standard incentive agreement with the national Department of
	Public Works (DPW)
	Municipalities must report on all projects for which they are claiming the incentive via national DPW's EPWP reporting systems
	• Reports must be loaded on the EPWP reporting systems within 22 days after the end of every quarter
	in order for performance to be assessed for an incentive payment 45 days after the end of each
	quarter
	Municipalities must maintain beneficiary and payroll records as specified in the audit requirements in
	the EPWP Incentive Manual
	• Once received by the eligible municipality, the incentive grant must be used for continuing or
	expanding job creation programmes. Municipalities may use a maximum of 5 per cent of the grant to
	fund on-site management capacity that is contract-based and required to manage data capturing and
	on-site management costs related to the use of labour intensive methods. The incentive may not be used to hire permanent staff
Allocation criteria	Incentive allocations for each municipality are based on the targeted number of FTEs each
	municipality must create above the set minimum threshold. The targeted number of FTEs for each
	municipality is based on the allocated infrastructure grant budget and performance in the 2009/10
	financial year
	• Incentive allocations are indicative - the actual incentive payout is based on actual quarterly
	performance reported above the quarterly threshold during 2011/12
	To be eligible for an incentive allocation in 2011/12, a municipality must have:    To be eligible for an incentive allocation in 2011/12, a municipality must have:
	reported EPWP performance by 21 April 2010 created a minimum number of FTEs (met the threshold) to qualify - rural municipalities have a
	zero minimum threshold in 2010/11
	A nominal allocation of R680 000 is provided when:
	municipalities report after 21 April 2010 but before 21 October 2010
	municipalities do not meet the minimum threshold
	municipalities exceed the threshold but qualify for an incentive of less than R680 000
	The nominal allocation can then be accessed if public bodies meet their threshold in-year
	To be paid out an incentive allocation, a municipality must:
	sign an incentive agreement
	report on EPWP performance to national DPW quarterly within required timeframes
	perform above the quarterly threshold
	• The indicative incentive allocation from underperforming municipalities may be re-allocated to over-
	performing municipalities by DPW
	• Eligible municipalities that exceed their targets may be paid an incentive in excess of their published

	Expanded Public Works Programme Incentive Grant for Municipalities
Reason not incorporated in equitable share	This grant is intended to incentivise and reward performance on the EPWP
Past performance	2009/10 audited financial outcomes
	• 47 of the 68 eligible municipalities earned the incentive and were paid a total of R114 million (57 per
	cent of the R202 million allocated) from the incentive
	2009/10 service delivery performance
	• 127 222 work opportunities were reported by 127 municipalities
D 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	40 793 full time equivalent jobs were reported by 127 municipalities
Projected life	Grant continues until the end of the 2013/14 financial year, subject to review
MTEF allocations	• 2011/12: R680 million, 2012/13: R666 million and 2013/14: R779 million
Payment schedule	Transfers will be made in accordance with a payment schedule agreed with National Treasury
Responsibilities of the	Responsibilities of the national department of public works
transferring national officer and receiving officer	Determine eligible municipalities and set minimum thresholds, performance targets and indicative incentive allocations for eligible municipalities based on reported performance
	• Provide municipalities with the EPWP Incentive Manual outlining the conditions and obligations of
	the incentive as well as the audit requirements
	Support municipalities to apply the EPWP Guidelines to project design
	Support municipalities to report using the EPWP reporting systems
	Monitor the performance of municipalities and report on quarterly progress against targets
	Submit to National Treasury quarterly progress against performance targets; incentive amounts earned by municipalities; and a revised payment schedule within 45 days after the end of the quarter
	Conduct sample audits on a continuous basis
	Audit the final performance of municipalities after the closure of the financial year
	• Disburse incentives on a quarterly basis to eligible municipalities based on performance in the
	previous quarter, if the quarterly thresholds have been exceeded
	Responsibilities of eligible municipalities
	Sign the standard incentive agreement with the national DPW
	Report all projects to be taken into account when assessing performance into the EPWP reporting
	systems and update progress quarterly in accordance with the reporting requirements in the incentive
Process for approval of 2012	agreement  Musicialities west and and agreement and EDWD arrives for the 2010/11 formula was been
MTEF allocations	• Municipalities must report performance on EPWP projects for the 2010/11 financial year by 21 April 2011. Performance in 2010/11 as well as the allocated infrastructure budgets will determine
WILL anocations	the targets and allocations for 2012/13
	A late eligibility provision is applicable for late reporting: municipalities that report after this date but
	before 21 October 2011 will qualify for only a nominal incentive allocation of R680 000
	National DPW will distribute the standard incentive agreement by end March every year
	• Eligible municipalities must sign the standard incentive agreement with national DPW and agree to
	comply with the conditions and obligations of the incentive grant at the beginning of the municipal
	financial year

### NATIONAL TREASURY GRANTS

	Local Government Financial Management Grant
Transferring department	National Treasury (Vote 10)
Strategic goal	To secure sound and sustainable management of the fiscal and financial affairs of municipalities
Grant purpose	To promote and support reforms in financial management by building capacity in municipalities to implement the Municipal Finance Management Act (MFMA)
	As part of strengthening financial and asset management in municipalities, the grant provides funding for water and energy internship programme to graduates in selected water boards and municipalities.
Outcome statements	Improved capacity in financial and asset management of municipalities
	<ul> <li>Progressive improvement in the quality of reporting for municipalities</li> </ul>
	<ul> <li>Appropriately skilled financial management officers appointed in municipalities consistent with the competency regulations</li> </ul>
	Appropriately skilled engineers, technicians and artisans appointed in municipalities consistent with
	<ul> <li>the infrastructure competency regulations</li> <li>Municipalities have processes and procedures in place to provide quality reports and are publishing</li> </ul>
	financial information on a regular basis
	<ul> <li>Continuous improvement in audit outcomes</li> <li>Improved revenue, expenditure, asset and liability management</li> </ul>
	Improved revenue, experientalic, asset and habitiny management     Improved overall compliance with implementing the MFMA
Outputs	Improved and sustained skills development including the appointment of at least five interns
	supporting the implementation of financial management reforms in municipalities focusing on the gaps identified in MFMA implementation plans
	Upgrading of IT systems to deliver reports required for financial management improvement and improve the quality of data
	Preparation and implementation of multi-year budgets meeting uniform norms and standards
	Assist in the implementation of supply chain reforms, accounting reforms, producing quality and timely financial statements
	• Improved and sustained skills development including the appointment of water and energy interns in selected municipalities supporting asset management reforms in such municipalities by focusing on the gaps identified in, amongst others, audit reports
	Assist in the preparation and implementation of financial recovery plans
	<ul> <li>Progressive improvements in audit outcomes</li> <li>Improvements to internal and external reporting on budgets, finances, Service Delivery Budget and</li> </ul>
	Implementation Plan (SDBIP), in-year and annual reports  Implementation of the MFMA
Priority outcome(s) of	Outcome 9: A Responsive, accountable, effective and efficient local government system
government that this grant primarily contributes to	• Outcome 12: An efficient, effective and development oriented public service and empowered, fair and inclusive citizenship
Details contained in the	This grant uses an implementation plan which contain the following:
business plan	<ul><li>Outcomes indicators</li><li>Output indicators</li></ul>
	Key activities
	– Inputs
Conditions	• Establishment of a Budget and Treasury Office (BTO) with positions filled by appropriately qualified
	<ul> <li>personnel and appointment of at least five interns over a multi-year period</li> <li>Appoint interns through the water and energy internship programme in selected municipalities and</li> </ul>
	water boards
	Establishment of Supply Chain Management and Internal Audit units
	Ongoing review, revision, and submission of MFMA implementation plans to National Treasury to
	address weaknesses in financial management
	• Acquisition of a financial management system that can produce multi-year budgets, in-year reports,
	<ul> <li>SDBIP, annual reports and automation of financial management practices</li> <li>Utilise funding to support the training of municipal officials in financial management reforms in</li> </ul>
	<ul> <li>support of Competency Regulations, Gazette 29967 of June 2007</li> <li>Preparation and submission of annual financial statements for audits and implement changes required</li> </ul>
	<ul> <li>to address audit findings</li> <li>Provide technical support to municipalities in financial management and the transference of skills to</li> </ul>
	municipal officials
	<ul> <li>Where appropriate, the preparation of a financial recovery plan and the implementation thereof</li> <li>Transfers will be dependent on timely reporting and appropriate spending levels</li> </ul>

	Local Government Financial Management Grant
Allocation criteria	Funds allocated to municipalities to assist in the implementation of financial management reforms, attendance at accredited training and capacity building programmes on financial management     Additional support to selected municipalities in the employment of an appropriately skilled, experienced and suitably qualified chief financial officer
	Funds allocated to selected municipalities to improve asset management through the employment of technical interns on water and electricity amounting to R50 million in 2011/12, R75 million in 2012/13 and R100 million in 2013/14
	To address special requests linked to financial reforms as pilot initiatives for wider application to all municipalities
	• Render support to national and provincial departments for administration of the programme, not exceeding 2.5 per cent of annual allocations
Reason not incorporated in equitable share	To provide direct support to develop municipal financial management and technical capacity in water and energy to lend assistance for the implementation of the MFMA and regulations
Past performance	2009/10 audited financial outcomes
	Allocated and transferred R300 million to 283 municipalities of which R256 million (85.2 per cent) was spent by the end of 2009/10 municipal financial year
	No adverse audit issues were identified
	2009/10 service delivery performance  Weakshape with manifest officials have been undertaken cones the country on the implementation of
	Workshops with municipal officials have been undertaken across the country on the implementation of the regulations providing for minimum competencies
	Facilitation of training providers accreditation with Local Government Sector Education and Training Authority (LGSETA)
	Generally Recognised Accounting Practice (GRAP) and Municipal Budget formats and reporting regulations training was provided to municipal officials
	Over 1200 graduate finance interns have been appointed in municipalities with the objective of increasing finance management capacity
	Internship workshops have been concluded in a number of provinces
	180 municipalities have submitted their Annual Reports for 2008/09      Define of account along for annual ideal things and a financial distance.
	<ul> <li>Drafting of recovery plans for municipalities under financial distress</li> <li>Issues around spending of the grant are addressed at the quarterly MFMA co-ordination meetings</li> </ul>
	which are held with relevant stakeholders in support of provincial and municipal efforts in implementing the financial reforms
	Technical advisors have been placed in two provincial treasuries (KwaZulu-Natal and Eastern Cape) and other support was offered to municipalities in Limpopo and Eastern Cape provinces
	<ul> <li>Total registrations for the MFMA DVD Interactive learning programme is at 7 443 officials</li> <li>Training was provided to over 150 officials on the application and implementation of the MFMA</li> </ul>
	Asset Transfer Regulations
	220 MFMA implementation plans submitted
	• Effectiveness and efficiency of the BTOs in 283 municipalities assessed and feedback provided to 17 large municipalities, while consolidated data on the 266 municipalities provided to provincial treasuries to strengthen their oversight role and provide the relevant feedback to their delegated
	municipalities  • Support and guidance through the MFMA mailbox provided to municipalities on implementation and
	<ul> <li>interpretation of the reforms</li> <li>Training of municipal councillors on governance and oversight undertaken upon request</li> </ul>
Projected life	The programme is designed to support and implement the MFMA and associated reforms. The grant forms part of government's broader capacity building initiative and focuses on building in-house municipal capacity
MTEF allocations	• 2011/12: R435 million, 2012/13: R479 million and 2013/14: R526 million
Payment schedule	The grant will be disbursed during July 2011 and January 2012
Responsibilities of the	Responsibilities of the national department
transferring national	Monitoring and management of the programme  Transfer for the description of the programme of the programme of the programme.
officer and receiving officer	<ul> <li>Transfer funds to support administration of the grant and to municipalities to assist implementation of the MFMA and its supporting regulations</li> <li>Undertake ongoing monitoring in all municipalities</li> </ul>
	Responsibilities of the municipalities
	Submit monthly reports as per the requirements contained in the Division of Revenue Act
	Transfer allocations to water boards and licenced energy providers, where technical interns are based, if paid by water boards and/or Eskom
	• In the case where interns are based at the municipality during the training period, monthly reports must be submitted on a monthly basis by the municipality as per the requirements contained in the Division of Revenue Act or as prescribed by National Treasury
Process for approval of	Ongoing review, revision and submission of implementation plans to address weaknesses in financial
2012 MTEF allocations	and asset management
	The programme is based on the MFMA implementation plans of municipalities

	Neighbourhood Development Partnership Grant
Transferring department	National Treasury (Vote 10)
Strategic goal	To improve quality of life of residents in township areas through the creation of economically viable and sustainable neighbourhoods
Grant purpose	To support neighbourhood development projects that provide community infrastructure and create the platform for other public and private sector development, towards improving the quality of life of residents in targeted underserved neighbourhoods (townships generally)
Outcome statements	Deliver targeted and sustainable capital improvements
	Crowd in private and public investments and partners into target areas
	Increase municipal/local capacity for ongoing township development
Outputs	Number of municipalities granted award status
•	Establish long-range strategies and targeted plans (programmes) for township regeneration
	Catalytic township development projects are scoped, designed and implemented
	Partnerships and funding are leveraged into the programmes and/or target areas
	Township development knowledge is generated and disseminated
Priority outcome(s) of	Outcome 8: Create sustainable human settlements and improve the quality of household life
government that this grant contributes to	Outcome 9: Responsive, accountable, effective and efficient local government system
Details contained in the	Outcome indicators
business plan	Output indicators
	Key activities
	• Inputs
Conditions	<ul> <li>Compliance with terms of funding agreement signed between municipality and national department</li> <li>Approval by the National Treasury of business plans and project plans which are aligned with Neighbourhood Development Partnership Grant (NDPG) objectives and guides</li> </ul>
	Programmes are aligned to municipal Integrated Development Plans (IDPs), and there is a council resolution supporting the award and/or programme to be implemented
	Receiving officer to submit a payment schedule with budgets and timeframes for programme implementation by March annually
	Submission and approval of required monthly and quarterly reports
	Evidence of partnerships and leveraged funding into programme/target area
	<ul> <li>Prioritisation of projects and approaches which demonstrate explicit response to government-wide priorities, including the promotion of labour-absorbing investments, green technology, and youth development</li> </ul>
Allocation criteria	Allocations are made to 57 qualifying municipalities who demonstrated the need for township developments that catalyse commercial and social upliftment. The grant is currently not open for new
Reason not incorporated in	<ul> <li>applications</li> <li>To directly facilitate neighbourhood / township development in selected municipal areas, and to</li> </ul>
equitable share	investigate the catalytic role of public funding for spatially targeted development
Past performance	2009/10 audited financial outcomes
2 day personalise	R551 million allocated in Schedule 6 direct transfers to municipalities, R508 million transferred and R505 million (99.3 per cent) spent by municipalities
	R110 million allocated in Schedule 7 indirect transfers to municipalities, R90 million transferred
	2009/10 service delivery performance
	• 57 municipalities granted award status, 52 of which are in receipts of technical assistance funding for
	programme planning and preparation (Schedule 7 grant), and 32 (5 new in period) have embarked on construction or implementation (Schedule 6 grant) to date
	<ul> <li>100 townships directly targeted by NDPG investment</li> </ul>
	8 municipalities meeting township regeneration strategy requirements
	1 Business Plan approved for NDPG programme implementation
	• 32 projects under construction
	• 3 projects in completion (exit and review) stage
	• 7 programmes with identified partnerships and funding leveraged
	• 2 (6 day) learning/training events delivered to a total of 95 NDPG officials and township development stakeholders through the Training for Township Renewal Initiative
	4 knowledge generation products supported (Township Renewal Sourcebook, Township Transformation Timeline, Impact of township retail study, Value capture of transport interchanges study)
Projected life	This is a ten year programme where the first year of actual implementation was 2008
,	2 and a con your programme where the first your of detail implementation was 2000

	Neighbourhood Development Partnership Grant
MTEF allocations	Direct transfers (Schedule 6)
	2011/12: R750 million, 2012/13: R800 million and 2013/14: R800 million
	Allocation in-kind (Schedule 7)
	2011/12: R100 million, 2012/13: R80 million and 2013/14: R55 million
Payment schedule	Transfers in accordance with a payment schedule approved by National Treasury
Responsibilities of the	Responsibilities of national department
transferring national	Determine the criteria for evaluating requests for grants from municipalities
officer and receiving officer	Determine the grant allocations for the future MTEF periods
	Monitor, manage and evaluate financial and non-financial performance
	Oversee and enforce objectives and conditions of this grant
	Responsibilities of municipalities
	<ul> <li>Compile and submit monthly and quarterly expenditure and progress reports in line with NDPG requirements and as stipulated in the Division of Revenue Act</li> </ul>
	Provide adequate human resources capacity for the successful coordination and implementation of NDPG
	Coordinate the development of programmes and plans aligned with the grant objectives against which performance will be assessed
	Establish additional clear indicators for outputs and outcomes against which performance may be further assessed
	Manage and monitor implementation of programmes according to the approved business and project plans, and ensuring sound financial management and value for money
	<ul> <li>Maintain accurate and current grant and performance information as specified in NDPG management information formats and systems</li> </ul>
	Engage stakeholders so as to develop partnerships and mainstream the township development agenda in municipal and other relevant growth and development plans
Process for approval of	• NDPG business plans are typically approved on a multi-year basis. The process for approval of
2012/13 business plans	business plans and annual allocations against these plans are based on:
	<ul> <li>approved project plans and submission of payment schedules in July 2011</li> </ul>
	<ul> <li>ongoing progress reporting, and performance review where necessary</li> </ul>
	Municipal provisional allocations will be finalised by October 2011

### **ENERGY GRANTS**

	Electricity Demand Side Management (Eskom) Grant
Transferring department	• Energy (Vote 29)
Strategic goal	To reduce electricity consumption by promoting energy efficient practices
Grant purpose	To implement the Electricity Demand Side Management (EDSM) programme by providing subsidies to Eskom to address EDSM in residential dwellings and communities in order to mitigate the risk of load shedding and supply interruptions
Outcomes statements	Amount of energy saved in W/Wh
	A reduction in the demand for electricity
	Increased awareness
	Energy management capability created
Outputs	Amount of energy saved
Priority outcome(s) of government that this grant primarily contributes to	Outcome 10: Environmental assets and natural resources that are well protected and continually enhanced
Details contained in the	Outcome indicators
business plans	Outputs indicators
	Key activities
	• Inputs
Conditions	Expenditure should be guided by the approved business plans
	EDSM programme must be implemented in line with a municipality's Integrated Development Plan
	<ul> <li>Measurement and verification must be done in line with the standards set by the South African Bureau of Standards</li> </ul>
Allocation criteria	Its targeted to high energy consuming municipalities
Reason not incorporated in equitable share	This is a specific conditional transfer in support of the EDSM programme
Past performance	2009/10 audited financial outcomes
	Transferred amount is R75 million
	2009/10 service delivery performance
	Installation of solar water heaters started and will be completed in April 2011
Projected life	The grant will continue until 2011/12 financial year and is subject to review
MTEF allocations	• 2011/12: R119 million
Payment schedule	Transfers are made according to a payment schedule approved by National Treasury
Responsibilities of the	Responsibilities of the national department
national transferring and	Performance review report to be submitted in June 2011 to National Treasury
receiving officer	Responsibilities of municipalities
	Submit business plans to Department of Energy (DOE)
	Submit monthly and quarterly reports to DOE and National Treasury
Process for approval of 2012/13 business plan	The grant is subject to review and the outcome thereof will inform its continuation into 2012/13

	Electricity Demand Side Management (Municipal) Grant
Transferring department	• Energy (Vote 29)
Strategic goal	To reduce electricity consumption by promoting energy efficient practices
Grant purpose	To implement the Electricity Demand Side Management (EDSM) programme by providing subsidies to licenced distributors to address EDSM in residential dwellings, communities and municipal infrastructure in order to mitigate the risk of load shedding and supply interruptions
Outcome statements	<ul> <li>Amount of energy saved in W/Wh</li> <li>A reduction in the demand for electricity</li> <li>Increased awareness</li> <li>Energy management capability created</li> </ul>
Outputs	Amount of energy saved
Priority outcome(s) of government that this grant primarily contributes to	Outcome 10: Environmental assets and natural resources that are well protected and continually enhanced
Details contained in the business plans	Outcome indicators     Outputs indicators     Key activities     Inputs
Conditions	<ul> <li>Expenditure should be guided by the approved business plans</li> <li>EDSM programme must be implemented in line with a municipality's Integrated Development Plan</li> <li>Measurement and verification must be done in line with the standards set by the South African Bureau of Standards</li> </ul>
Allocation criteria	Applications in the form of business plans from licensed municipal distributors are evaluated based on:     targeting areas with high electricity usage     competency and the ability to implement and manage projects
Reason not incorporated in equitable share	This is a specific conditional transfer in support of the EDSM programme
Past performance	2009/10 audited financial outcomes     Allocated R175 million and R171 million was transferred to municipalities of which R86 million (50.3 per cent) spent by the end of the 2009/10 municipal financial year  2009/10 service delivery performance     There are 19 Municipalities involved in the programme, of these, 11 completed projects
Projected life	<ul> <li>There are 19 Mulincipanties involved in the programme, of these, 11 completed projects</li> <li>The grant was introduced for a three year period, ending 2011/12, subject to review</li> </ul>
MTEF allocations	The grant was introduced for a time eyear period, ending 2011/12, subject to review     2011/12: R280 million
Payment schedule	Transfers are made in accordance to a payment schedule approved by National Treasury
Responsibilities of the	Responsibilities of the national department
transferring national officer	Performance review report to be submitted in June 2011 to National Treasury
and receiving officer	Responsibilities of municipalities
Ŭ	Submit business plans to Department of Energy (DOE)
	Submit monthly and quarterly reports to DOE and National Treasury
Process for approval of 2012/13 business plan	The grant is subject to review and the outcome thereof will inform its continuation into 2012/13

	Integrated National Electrification Programme (Eskom) Grant
Transferring department	• Energy (Vote 29)
Strategic goal	To reduce the backlogs of un-electrified households and clinics
Strategic goar	Funding of bulk infrastructure to ensure constant supply of electricity
Grant purpose	To implement the Integrated National Electrification Programme (INEP) by providing capital subsidies to Eskom to address the electrification backlog of occupied residential dwellings, the installation of bulk infrastructure and rehabilitation and refurbishment of electricity infrastructure in order to improve quality of supply
Outcome statements	<ul> <li>A reduction in household and clinic electrification backlogs</li> <li>Universal access to electricity and improvement in distribution infrastructure reliability</li> </ul>
Outputs	The number of connections to households and clinics per annum
	The number of bulk infrastructure installations
	Implementation of labour intensive methods on electrification projects and the number of jobs created
Priority outcome(s) of government that this grant primarily contributes to	Outcome 9: A Responsive, accountable, effective and efficient local government system
Details contained in	Outcome indicators
business plans	Output indicators
	Key activities
	• Inputs
Conditions	<ul> <li>Plans need to have under gone pre-engineered projects feasibility approval</li> <li>Project prioritized by municipalities in terms of their IDP and supporting letters provided</li> </ul>
	<ul> <li>Project prioritized by municipalities in terms of their IDP and supporting letters provided</li> <li>Municipalities to comply with requirements to provide approved bulk project in their business plans</li> </ul>
Allocation criteria	<ul> <li>Allocations to Eskom are made on behalf of municipalities based on applications from Eskom for non-</li> </ul>
Reason not incorporated in equitable share Past performance	in the equitable share  2009/10 audited financial outcomes
•	R1 492 million was allocated and transferred to Eskom
	2009/10 service delivery performance
	100 379 household connections and 6 substations were completed
Projected life	The grant will continue until the backlogs are eradicated
MTEF allocations	• 2011/12: R1 738 million, 2012/13: R1 882 million and 2013/14: R1 986 million
Payment schedule	Transfers are made monthly in accordance with a payment schedule approved by National Treasury
Responsibilities of the	Responsibilities of the national department
transferring national	Agree with municipalities on outputs and targets     Continuously manifestimal amountains.
officer and receiving	Continuously monitor implementation     Provide directive for central coordination for bulk infrastructure
officer	<ul> <li>Provide directive for central coordination for bulk infrastructure</li> <li>Approve submissions for refurbishment of critical infrastructure</li> </ul>
	Responsibilities of Eskom
	20 Amps per household connection and applicable supply for clinic connections
	Report to Department of Energy and National Treasury on monthly expenditure for the grant
Dunning for or	Ensure that all planned projects are in line with the municipal IDP and priority list
Process for approval of 2012/13 business plan	Ensure that all planned projects are in line with the indiricipal fibr and priority list     Ensure that planned projects are feasible and went through the pre-engineering process

	Integrated National Electrification Programme (Municipal) Grant
Transferring department	Energy (Vote 29)
Strategic goal	<ul> <li>To reduce the backlogs of un-electrified households and clinics</li> <li>Funding of bulk infrastructure to ensure constant supply of electricity</li> </ul>
Grant purpose	To implement the Integrated National Electrification Programme (INEP) by providing capital subsidies to municipalities to address the electrification backlog of occupied residential dwellings, the installation of bulk infrastructure and rehabilitation and refurbishment of electricity infrastructure in order to improve quality of supply
Outcome statements	A reduction in household and clinic electrification backlogs     Universal access to electricity and improvement in distribution infrastructure reliability
Outputs	<ul> <li>The number of connections to households and clinics per annum</li> <li>The number of bulk infrastructure installations</li> <li>Implementation of labour intensive methods on electrification projects and the number of jobs created</li> </ul>
Priority outcome(s) of government that this grant primarily contributes to	Outcome 6: An efficient, competitive and responsive economic infrastructure network Outcome 9: A Responsive, accountable, effective and efficient local government system
Details contained in the business plan	<ul> <li>Outcome indicators</li> <li>Output indicators</li> <li>Key activities</li> <li>Inputs</li> </ul>
Conditions	<ul> <li>Adhere to the labour intensive construction methods in terms of the Expanded Public Works Programme (EPWP) guidelines for activities such as trenching, planting of poles, etc</li> <li>Register master plans for bulk infrastructure with INEP and abide by the directives of the department regarding the central planning and co-ordination for such bulk infrastructure. Use INEP funds for the refurbishment of critical infrastructure, only upon submission of a project plan which must be approved by the national department</li> <li>Utilise own funding if subsidy is insufficient – top–up funding must be available</li> <li>Minimum suite of supply of 1KVA, ADMD, 20 Amp per household and maximum of 16 KVA per clinic connection</li> </ul>
Allocation criteria	Applications from licensed municipal distributors based on:     high backlogs     rural bias     nodal zones     existence of households for connections projects     past performance     integration with other programmes such as Urban Renewal Programme, other infrastructure programmes like Breaking New Ground (BNG), housing, etc     the financial, technical and staff capabilities to distribute electricity, to expand and maintain the networks     consultation with communities in terms of Integrated Development Plan (IDP) process     ensuring that universal access objectives are fast tracked     new and upgrading of bulk infrastructure projects related to (i) future electrification and (ii) where distribution network reliability adversely impacts economic activity     Infrastructure which is in a state of disrepair, unsafe and which prohibits further connections     Informal settlements where service delivery has been prioritised
Reason not incorporated in equitable share	This is a specific conditional capital transfer for electrification of households and clinics not catered for in the equitable share
Past performance	2009/10 audited financial outcome:  R932 million was allocated and R914 million was transferred to municipalities with R806 million (88.2 per cent) spent by the end of the 2009/10 municipal financial year  2009/10 service delivery performance  48 752 households were electrified and 3 sub-stations were completed
Projected life	Grant continues until the universal access is achieved and subject to government review
MTEF allocations	• 2011/12: R1 097 million, 2012/13: R1 151 million and 2013/14: R1 215 million
Payment schedule	Transfers are made in accordance with a payment schedule approved by National Treasury
Responsibilities of the	Responsibilities of national department
transferring national	Agree with municipalities on outputs and targets  Continuously monitor implementation and provide support to municipalities.
officer and receiving officer	Continuously monitor implementation and provide support to municipalities     Verify reports from municipalities
	Verify reports from municipalities  Responsibilities of municipalities
	Responsibilities of municipalities  • Ensure that projects are implemented in line with what is reflected in the IDP of the municipality  • Report correctly on the management of this grant
Process for approval of 2012/13 business plans	Application forms are sent to municipalities and evaluation of all applications/proposals received from municipalities by July 2011

### **HUMAN SETTLEMENTS GRANTS**

	Rural Households Infrastructure Grant
Transferring department	Human Settlements (Vote 31)
Strategic goal	To eliminate backlogs in water and sanitation in rural households
Grant purpose	• The grant is intended to provide specific capital funding for the eradication of rural water and sanitation backlogs and is targeted at existing households where bulk-dependent services are not viable. The grant also funds training for beneficiaries on health and hygiene practices and how to maintain the facilities provided
Outcome statements	Increase the number of rural households provided with non-bulk-dependent sanitation and water facilities
	<ul> <li>Encourage the use of Community Based Organisations, Non-Governmental Organisations and Public Entities in delivering facilities and on-site training to rural households on how to sustain the maintenance of infrastructure</li> <li>Alleviate poverty by implementing programmes through labour-based methods</li> </ul>
	Pursue on-site solutions, and allowing people to take charge of the improvement of their wellbeing
Outputs	<ul> <li>Number of rural households provided with access to on-site sanitation and non-bulk-dependent water facilities</li> <li>Number of jobs created in accordance with Expanded Public Works Programme (EPWP) guidelines</li> <li>Number of people trained in on-site technologies and maintenance of facilities</li> <li>Number of people reached by health and hygiene awareness training</li> </ul>
Priority outcome(s) of	<ul> <li>Number of people reached by health and hygiene awareness training</li> <li>Outcome 7: Vibrant, equitable and sustainable rural communities and food security for all</li> </ul>
government that this grant primarily contributes to	Outcome 7. Vibrant, equitable and sustainable fural communities and food security for an
Details contained in the	This grant uses an implementation framework which contains the following:
business plan	<ul><li>outcome indicators</li><li>outputs indicators</li></ul>
	- key activities
	- inputs
Conditions	• Target the provision for rural on-site sanitation and water facilities to households not intended for
	<ul> <li>connector services</li> <li>Implementation of this grant has to be done in consultation with the Water Services Authorities (WSAs)</li> </ul>
	responsible for an area
	• Specific service level agreements with the respective WSAs have to be entered into with the national
	programme manager
	<ul> <li>Agreements include joint planning as well as acceptance of obligation by the recipient municipality regarding facilitation of operation and maintenance of facilities provided</li> </ul>
	The design of sanitation facilities have to be within acceptable norms and standards
	<ul> <li>The implementation of the programme should include training of communities on their understanding of their responsibilities with the outcomes of the programme and health and hygiene awareness training</li> <li>Proper procurement procedures must be followed</li> </ul>
Allocation criteria	Households in municipalities identified as rural households where water and sanitation needs exist and
	connector solutions are not appropriate
Reason not incorporated in	• This is a special purpose grant with specific objectives and distribution criteria and is not implemented
equitable share	by municipalities
Past performance	Outputs have to be measured specifically  2009/10 audited financial outcomes
i ast periormance	• Grant introduced in 2010/11
	2009/10 service delivery performance
	• Grant introduced in 2010/11
Projected life	The programme is subject to meeting targets to eradicate rural backlogs by 2014
MTEF allocations	• 2011/12: R232 million, 2012/13: R480 million and 2013/14: R517 million
Payment schedule	Transfers are made in accordance with a payment schedule agreed to by National Treasury
Responsibilities of the	Responsibilities of the national department
transferring national	Administer the implementation of the programme
officer and receiving officer	
	<ul> <li>Continuously monitor implementation and provide support to municipalities and households</li> <li>Submit monthly and quarterly reports to National Treasury</li> <li>Verify reports from municipalities</li> </ul>
	verny reports from municipanties

### **Rural Households Infrastructure Grant** Responsibilities of Water Services Authorities (WSAs) Determine through municipal participatory structures the most appropriate technology per project area Municipality will be responsible for selection of the project areas that provide total coverage as per available funds (e.g. whole ward, village etc) The ownership of assets and schemes developed through the project will not necessarily reside with either of the parties to the service level agreement. However, the municipality shall be responsible for maintenance of the installed infrastructure WSAs will be part of the project steering committee, and will be part of problem solving, progress monitoring, evaluation and reporting Facilitate the introduction of the project team to ensure smooth running of the project to benefiting communities The WSAs shall adopt a Water Services Development Plan, as part of its Integrated Development Plan, as prescribed by the Water Services Act, and ensure that all projects included in the schedule of projects are identified and prioritized in accordance with its Water Services Development Plan The WSAs must ensure that groundwater protocol has been conducted to manage the potential of groundwater contamination by the on-site sanitation and is attached to the agreement Process for approval of Based on current concentrations of backlogs in rural sanitation as a priority, the Department of Human 2012 MTEF allocations Settlements must determine the municipalities and specific communities where funding will be spent. Similarly, in respect to the eradication of rural household water backlogs, allocations will be determined in consultation with the Department of Water Affairs. In both cases, allocations will be made in consultation with the respective WSAs

Tuonafaurina danautmant	Urban Settlements Development Grant
Transferring department Strategic goal	Human Settlements (Vote 31)  The Urban Settlement Development Grant (USDG) scales to support the development of sustainable.
Strategic goai	The Urban Settlement Development Grant (USDG) seeks to support the development of sustainable human settlements and improved quality of life for households through accelerating the provision of serviced land with secure tenure for low-income households in large urban areas by supplementing municipal resources
Grant purpose	To improve the efficiency and coordination of investments in the built environment through:     Providing large municipalities with appropriate resources and control over the selection and pursuit of investment programmes in the built environment
Outcome statements	Enhanced sustainability of human settlements and improved quality of household life in urban areas as measured by (measurement units in brackets):
	• Increased availability of serviced urban land parcels (reduction in average production cost and price of land)
	<ul> <li>Increased access to suitable shelter (reduction in real costs of rental for a well-located housing unit)</li> <li>Increased average density of cities (city-wide average dwelling units per hectare)</li> </ul>
	• Increase security of tenure irrespective of ownership, or rental, formal or informal structures security (number of new freehold tenure units transferred to beneficiaries, rental agreements signed and legal protection provided)
Outputs	<ul> <li>The number of proclaimed and serviced land units produced per annum</li> <li>The number of serviced land units transferred to eligible beneficiaries per annum (number of households receiving secure tenure)</li> </ul>
	<ul> <li>Number of hectares of municipal land proclaimed, serviced and released for human settlement</li> <li>Reduction in the number of informal settlements (number informal settlements upgraded)</li> </ul>
	<ul> <li>Number of households in informal settlements provided secure tenure and basic services either in-situ or elsewhere</li> <li>Identification and development of land within strategically located areas for mixed income and use</li> </ul>
	residential purposes  Number of households receiving support in basic services per annum over the grant period, including
	<ul> <li>water, sanitation, solid waste and electricity services and bulk and connector infrastructure</li> <li>Average number of working days taken from application to approval of development applications by category (township establishment; building plan approval)</li> </ul>
	The number of brownfield development approvals within the existing urban fabric (subdivision, rezoning, redevelopment)
	<ul> <li>Ratio of annual municipal own investments in land development relative to annual USDG expenditure</li> <li>Ratio of total land development investment (USDG and own financing) to Human Settlements Development Grant (HSDG)</li> </ul>
	<ul> <li>Ratio of total value of building completed to total municipal capital expenditure</li> <li>Number of local areas (suburbs) benefiting from financing from USDG, Public Transport Infrastructure and Systems(PTIS) and Neighbourhood Development Partnership (NDP) grants</li> </ul>
	<ul> <li>Timely submission of annual plan</li> <li>Timely submission of in-year and annual report</li> </ul>
	Capital expenditure performance (variance of budgeted to actual capital expenditure)     The number of households with adequate shelter
	<ul> <li>Number of new affordable rental units delivered</li> <li>Number of informal rental upgrades</li> </ul>
Priority outcome(s) of government that this grant	Outcome 8: Sustainable human settlements and an improved quality of household life
primarily contributes to	
Details contained in business plan	This grant uses a performance framework which contains the following;     City specific problem analysis
	<ul> <li>Human settlements strategy and targets</li> <li>Public transportation and spatial planning linkages</li> </ul>
	<ul> <li>Strategies for informal settlements, Greenfield and Brownfield development</li> </ul>
	<ul> <li>Financing strategy and instruments</li> </ul>
Conditions	<ul> <li>Subsidy and service standards</li> <li>Multi-year performance targets must be agreed with the national Department of Human Settlements and</li> </ul>
Conditions	where required the relevant provincial departments, based on the strategic capital investment plans and
	<ul> <li>programme of the municipality</li> <li>Conditions associated with the grant must be restricted to output and outcome performance of the</li> </ul>
	<ul> <li>overall municipal capital expenditure programme</li> <li>Funds may be transferred to a municipality only if the municipality has satisfied the criteria set out in</li> </ul>
	the grant policy framework; and has submitted a draft performance framework by 30 March 2011 and the final approved framework by 7 June 2011, and human settlement and built environment performance framework which complies with the requirements set out in the grant policy framework to the national Department of Human Settlements, along with proof that the performance framework and the
	<ul> <li>performance targets contained therein have been ratified by a resolution of the municipal council</li> <li>This programme must:         <ul> <li>prioritise residential infrastructure for water, sanitation, refuse removal, streets lighting, solid</li> </ul> </li> </ul>
	waste, connector and bulk infrastructure, and roads which support the planning, funding and development of human settlements

Than Settlements Development Grant
A municipality may utilise grant funds for:
the acquisition, development or re-development and release of land with secure tenure the capital costs of the basic services infrastructure at bulk, connection and internal levels for po households  he rapid upgrading of informal settlements  A municipality must ensure that: grant funds are used for the purpose of land development that supports the integrated developme of human settlements  poor households receive the final benefit of the grant expenditures are consistent with the national and provincial human settlement strategic objective and goals  Municipalities must comply with grant conditions associated with the transfer and outlined in it performance framework including agreed performance targets  transfer of the first instalment depends on the submission and approval of a performance framework including agreed performance targets  transfer of the 2 <sup>rol.</sup> , 3 <sup>rol</sup> and 4 <sup>th</sup> instalment will be conditional upon submission and approval of support of the proper support to municipalities of the provide support of municipalities of the provide support to municipalities of the provide support of the
- the capital costs of the basic services infrastructure at bulk, connection and internal levels for pohouseholds  - the rapid upgrading of informal settlements  - A municipality must ensure that: - grant funds are used for the purpose of land development that supports the integrated developme of human settlements - poor households receive the final benefit of the grant - expenditures are consistent with the national and provincial human settlement strategic objective and goals  - Municipalities must comply with grant conditions associated with the transfer and outlined in performance framework including timely and full compliance with reporting requirements as outline below: - transfer of the first instalment depends on the submission and approval of a performance framework including agreed performance targets - transfer of the 2 <sup>14</sup> , 3 <sup>14</sup> and 4 <sup>18</sup> instalment will be conditional upon submission and approval signed-off quarterly reports - a report on performance against the targets provided for in the framework and in conjunction with the reporting requirements of section 71 of the Municipal Finance Management Act must be submitted to National Treasury - Non compliance to the above conditions can result in the funds being withheld or re-allocated - The base allocation is derived from the Municipal Infrastructure Grant formula explained in part 5 or annexure W1 in the bill - The formula incorporates backlog and poverty-weighted data - This is a new grant that was introduced in 2011 MTEF - 2009/10 service delivery performance - This is a new grant that was introduced in 2011 MTEF - 2009/10 service delivery performance - This is a new grant that was introduced in 2011 MTEF - The programme will continue up to 2014 subject to review - 2011/12: Re 267 million, 2012/13: R7 410 million and 2013/14: R8 127 million - 20 quarterly transfers in terms of a payment schedule approved by National Treasury by 15 December 2011 - Provide support to municipalities with regards to human settlement delivery as may be required Unde
- the rapid upgrading of informal settlements A municipality must ensure that: - grant funds are used for the purpose of land development that supports the integrated developme of human settlements - poor households receive the final benefit of the grant - expenditures are consistent with the national and provincial human settlement strategic objective and goals - Municipalities must comply with grant conditions associated with the transfer and outlined in performance framework including timely and full compliance with reporting requirements as outline below: - transfer of the first instalment depends on the submission and approval of a performance framework including agreed performance targets - transfer of the 2 <sup>nd</sup> , 3 <sup>nd</sup> and 4 <sup>nd</sup> instalment will be conditional upon submission and approval signed-off quarterly reports - a report on performance against the targets provided for in the framework and in conjunction with the reporting requirements of section 71 of the Municipal Finance Management Act must be submission or the profit of the profit of the Municipal Finance Management Act must be submission and approval signed-off quarterly reports - a report on performance against the targets provided for in the framework and in conjunction with the reporting requirements of section 71 of the Municipal Finance Management Act must be submission and approval signed of the profit of the Municipal Infrastructure Grant formula explained in part 5 of annexure W1 in the bill - The base allocation is derived from the Municipal Infrastructure Grant formula explained in part 5 of annexure W1 in the bill - This is a supplementary infrastructure grant with conditions, objectives and distribution criteria, (e.g. 2009/10 audited financial outcomes - This is a new grant that was introduced in 2011 MTEF - 2009/10 service delivery performance - This is a new grant that was introduced in 2011 MTEF - Projected life - The programme will continue up to 2014 subject to review - MTEF allocations - Responsibilities of the national depart
A municipality must ensure that:   grant funds are used for the purpose of land development that supports the integrated development of human settlements   grant funds are used for the purpose of land development that supports the integrated development of human settlements   grant funds are used for the purpose of land development that supports the integrated development of human settlement strategic objective and goals   Municipalities must comply with grant conditions associated with the transfer and outlined in a performance framework including timely and full compliance with reporting requirements as outline below:   transfer of the first instalment depends on the submission and approval of a performance framework including agreed performance targets   transfer of the 2", 3" and 4" instalment will be conditional upon submission and approval signed-off quarterly reports   a report on performance against the targets provided for in the framework and in conjunction with the reporting requirements of section 71 of the Municipal Finance Management Act must be submitted to National Treasury   Non compliance to the above conditions can result in the funds being withheld or re-allocated   The base allocation is derived from the Municipal Infrastructure Grant formula explained in part 5 or annexure W1 in the bill   The formula incorporates backlog and poverty-weighted data   This is a supplementary infrastructure grant with conditions, objectives and distribution criteria, (e.g. 2009/10 service delivery performance   This is a new grant that was introduced in 2011 MTEF   2009/10 service delivery performance   This is a new grant that was introduced in 2011 MTEF   2009/10 service delivery performance   This is a new grant that was introduced in 2011 MTEF   2009/10 service delivery performance on financial, non-financial, and control systems related to the grant of the performance of infancial, non-financial, and control systems related to the grant provide systems (Housing Subsidy System) that support the adminis
- grant funds are used for the purpose of land development that supports the integrated developme of human settlements - poor households receive the final benefit of the grant - expenditures are consistent with the national and provincial human settlement strategic objective and goals  • Municipalities must comply with grant conditions associated with the transfer and outlined in a performance framework including timely and full compliance with reporting requirements as outline below: - transfer of the first instalment depends on the submission and approval of a performance framework including agreed performance targets - transfer of the 2°°, 3°° and 4°° instalment will be conditional upon submission and approval signed-off quarterly reports - a report on performance against the targets provided for in the framework and in conjunction with the reporting requirements of section 71 of the Municipal Finance Management Act must be submitted to National Treasury - Non compliance to the above conditions can result in the funds being withheld or re-allocated  Allocation criteria - The base allocation is derived from the Municipal Infrastructure Grant formula explained in part 5 or annexure W1 in the bill - The formula incorporates backlog and poverty-weighted data - This is a supplementary infrastructure grant with conditions, objectives and distribution criteria, (e.g. 2009/10 audited financial outcomes - This is a new grant that was introduced in 2011 MTEF - 2009/10 service delivery performance - This is a new grant that was introduced in 2011 MTEF - 2009/10 service delivery performance - This is a new grant that was introduced in 2011 MTEF - 2009/10 service delivery performance - This is a new grant that was introduced in 2011 MTEF - 2009/10 service delivery performance - This is a new grant that was introduced in 2011 MTEF - 2009/10 service delivery performance - This is a new grant that was introduced in 2011 MTEF - 2009/10 service delivery performance on financial, non-financial, and control systems related to t
of human settlements  - poor households receive the final benefit of the grant  - expenditures are consistent with the national and provincial human settlement strategic objective and goals  • Municipalities must comply with grant conditions associated with the transfer and outlined in its performance framework including timely and full compliance with reporting requirements as outline below:  - transfer of the first instalment depends on the submission and approval of a performance framework including agreed performance targets  - transfer of the 2 <sup>nd</sup> , 3 <sup>nd</sup> and 4 <sup>th</sup> instalment will be conditional upon submission and approval of signed-off quarterly reports  - a report on performance against the targets provided for in the framework and in conjunction with the reporting requirements of section 71 of the Municipal Finance Management Act must be submitted to National Treasury  • Non compliance to the above conditions can result in the funds being withheld or re-allocated and the reporting requirements of section 71 of the Municipal Finance Management Act must be submitted to National Treasury  • Non compliance to the above conditions can result in the funds being withheld or re-allocated from the Municipal Infrastructure Grant formula explained in part 5 or annexure W1 in the bill  • The base allocation is derived from the Municipal Infrastructure Grant formula explained in part 5 or annexure W2 in the bill  • This is a supplementary infrastructure grant with conditions, objectives and distribution criteria, (e.g. 2009/10 audied financial outcomes  • This is a new grant that was introduced in 2011 MTEF  2009/10 service delivery performance  • This is a new grant that was introduced in 2011 MTEF  2009/10 service delivery performance of apayment schedule approved by National Treasury  • Quarterly transfers in terms of a payment schedule approved by National Treasury  • Monitor municipal performance on financial, non-financial, and control systems related to the grant provide support to municipalities of the nati
- poor households receive the final benefit of the grant - expenditures are consistent with the national and provincial human settlement strategic objective and goals  Municipalities must comply with grant conditions associated with the transfer and outlined in a performance framework including timely and full compliance with reporting requirements as outline below:  - transfer of the first instalment depends on the submission and approval of a performance framework including agreed performance targets  - transfer of the 2 <sup>nd</sup> , 3 <sup>nd</sup> and 4 <sup>nd</sup> instalment will be conditional upon submission and approval of signed-off quarterly reports  - a report on performance against the targets provided for in the framework and in conjunction with the reporting requirements of section 71 of the Municipal Finance Management Act must be submitted to National Treasury  • Non compliance to the above conditions can result in the funds being withheld or re-allocated  **The base allocation is derived from the Municipal Infrastructure Grant formula explained in part 5 or annexure W1 in the bill  • The formula incorporates backlog and poverty-weighted data  **Reason not incorporated in equitable share  **Past performance**  **Past performance**  **Past performance**  **Deviloaudited financial outcomes**  **This is a new grant that was introduced in 2011 MTEF  2009/10 suddied financial outcomes  **This is a new grant that was introduced in 2011 MTEF  2019/12 Re 267 million, 2012/13: R7 410 million and 2013/14: R8 127 million  **Payment schedule**  **Responsibilities of the national department**  **Quarterly transfers in terms of a payment schedule approved by National Treasury  **Provide support to municipalities with regards to human settlement delivery as may be required  **Undertake structured and/or necessary visits to municipalities on a regular basis  **Pastitute strategic planning, funding and implementation interaction between national and provincia department in evaluation report on the 2010/11 municipal annual reports to Nati
- expenditures are consistent with the national and provincial human settlement strategic objective and goals  - Municipalities must comply with grant conditions associated with the transfer and outlined in in performance framework including timely and full compliance with reporting requirements as outline below:  - transfer of the first instalment depends on the submission and approval of a performance framework including agreed performance targets  - transfer of the first instalment depends on the submission and approval of a gentlement of quarterly reports  - a report on performance against the targets provided for in the framework and in conjunction with the reporting requirements of section 71 of the Municipal Finance Management Act must be submitted to National Treasury  - Non compliance to the above conditions can result in the funds being withheld or re-allocated  **Allocation criteria**  **Allocation criteria**  **Allocation criteria**  **Proformance in equitable share**  **Past performance**  **Past performance**  **Past performance**  **Past performance**  **Past performance**  **Projected life**  **Projected life**  **Projected life**  **Projected life**  **This is a new grant that was introduced in 2011 MTEF**  **2009/10 service delivery performance**  **Projected life**  **This is a new grant that was introduced in 2011 MTEF**  **Projected life**  **Projected life**  **This is a new grant that was introduced in 2011 MTEF**  **Projected life**  **MTEF allocations**  **Payment schedule**  **Responsibilities of the antional department**  **Amment schedule**  **Responsibilities of the antional department**  **Amment schedule**  **Provide support to municipalities with regards to human settlement delivery as may be required**  **Undertake structured and/or necessary visits to municipalities on a regular basis**  **Submit an evaluation report on the 2010/11 municipal annual reports to National Treasury by 15 December 2011**  **Provide systems (Housing Subsidy System) that support the administration of t
### Allocation criteria  ### Allocation criter
Municipalities must comply with grant conditions associated with the transfer and outlined in in performance framework including timely and full compliance with reporting requirements as outline below:  transfer of the first instalment depends on the submission and approval of a performance framework including agreed performance targets  transfer of the 2 <sup>nd</sup> , 3 <sup>nd</sup> and 4 <sup>th</sup> instalment will be conditional upon submission and approval of grant performance against the targets provided for in the framework and in conjunction with the reporting requirements of section 71 of the Municipal Finance Management Act must be submissed to National Treasury  Non compliance to the above conditions can result in the funds being withheld or re-allocated to annexure W1 in the bill  The base allocation is derived from the Municipal Infrastructure Grant formula explained in part 5 or annexure W1 in the bill  The formula incorporates backlog and poverty-weighted data  This is a supplementary infrastructure grant with conditions, objectives and distribution criteria, (e.g. backlogs on infrastructure) different from that of the equitable share  Past performance  Past performance  This is a new grant that was introduced in 2011 MTEF  2009/10 service delivery performance  This is a new grant that was introduced in 2011 MTEF  2009/10 service delivery performance  This is a new grant that was introduced in 2011 MTEF  2011/12: R6 267 million, 2012/13: R7 410 million and 2013/14: R8 127 million  Payment schedule  Responsibilities of transferring national officer and receiving officer  Monitor municipal performance on financial, non-financial, and control systems related to the grant Provide support to municipalities with regards to human settlement delivery as may be required Undertake structured and/or necessary visits to municipalities on a regular basis  Submit an evaluation report on the 2010/11 municipal annual reports to National Treasury by 15 December 2011  Provide systems (Housing Subsidy System) that support the administration
performance framework including timely and full compliance with reporting requirements as outline below:  - transfer of the first instalment depends on the submission and approval of a performance framework including agreed performance targets - transfer of the 2 <sup>th</sup> , 3 <sup>th</sup> and 4 <sup>th</sup> instalment will be conditional upon submission and approval signed-off quarterly reports - a report on performance against the targets provided for in the framework and in conjunction with the reporting requirements of section 71 of the Municipal Finance Management Act must I submitted to National Treasury  Non compliance to the above conditions can result in the funds being withheld or re-allocated  **Allocation criteria**  The base allocation is derived from the Municipal Infrastructure Grant formula explained in part 5 of annexure W1 in the bill  The formula incorporates backlog and poverty-weighted data  **Compliance to the above conditions can result in the funds being withheld or re-allocated  This is a supplementary infrastructure grant with conditions, objectives and distribution criteria, (e.g. 2009/10 service delivery performance  **This is a new grant that was introduced in 2011 MTEF  2009/10 service delivery performance  This is a new grant that was introduced in 2011 MTEF  2009/10 service delivery performance  This is a new grant that was introduced in 2011 MTEF  2011/12: R6 267 million, 2012/13: R7 410 million and 2013/14: R8 127 million  Payment schedule  Responsibilities of transferring national  provide support to municipalities in terms of a payment schedule approved by National Treasury  Responsibilities of the national department  Monitor municipal performance on financial, non-financial, and control systems related to the grant municipal ties with regards to human settlement soft numan settlement son a regular basis  Submit an evaluation report on the 2010/11 municipal annual reports to National Treasury by 15 December 2011  Provide systems (Housing Subsidy System) that support the administration of the human s
below:  - transfer of the first instalment depends on the submission and approval of a performanc framework including agreed performance targets  - transfer of the 2 <sup>nd</sup> , 3 <sup>nd</sup> and 4 <sup>th</sup> instalment will be conditional upon submission and approval signed-off quarterly reports  - a report on performance against the targets provided for in the framework and in conjunction with the reporting requirements of section 71 of the Municipal Finance Management Act must be submitted to National Treasury  • Non compliance to the above conditions can result in the funds being withheld or re-allocated annexure W1 in the bill  • The base allocation is derived from the Municipal Infrastructure Grant formula explained in part 5 or annexure W1 in the bill  • The formula incorporates backlog and poverty-weighted data  * This is a supplementary infrastructure grant with conditions, objectives and distribution criteria, (e.g. backlogs on infrastructure) different from that of the equitable share  * Past performance  * This is a new grant that was introduced in 2011 MTEF  * 2009/10 service delivery performance  • This is a new grant that was introduced in 2011 MTEF  * Projected life  * The programme will continue up to 2014 subject to review  * MTEF allocations  * 2011/12: R6 267 million, 2012/13: R7 410 million and 2013/14: R8 127 million  * Payment schedule  * Responsibilities of transferring national officer and receiving officer  * Monitor municipal performance on financial, non-financial, and control systems related to the grant provide support to municipalities with regards to human settlement delivery as may be required  • Undertake structured and/or necessary visits to municipalities on a regular basis  • Submit an evaluation report on the 2010/11 municipal annual reports to National Treasury by 15 December 2011  • Provide systems (Housing Subsidy System) that support the administration of the human settlement delivery process  • Convene the evaluation committee for assessing the built environment plans  * Responsibilities of
framework including agreed performance targets transfer of the 2 <sup>pd</sup> , 3 <sup>pd</sup> and 4 <sup>th</sup> instalment will be conditional upon submission and approval signed-off quarterly reports a report on performance against the targets provided for in the framework and in conjunction with the reporting requirements of section 71 of the Municipal Finance Management Act must be submitted to National Treasury  Non compliance to the above conditions can result in the funds being withheld or re-allocated  The base allocation is derived from the Municipal Infrastructure Grant formula explained in part 5 or annexure W1 in the bill The formula incorporates backlog and poverty-weighted data This is a supplementary infrastructure grant with conditions, objectives and distribution criteria, (e.g. backlogs on infrastructure) different from that of the equitable share  Past performance This is a new grant that was introduced in 2011 MTEF  2009/10 service delivery performance This is a new grant that was introduced in 2011 MTEF  2011/12: R6 267 million, 2012/13: R7 410 million and 2013/14: R8 127 million  Payment schedule  Responsibilities of transferring national officer and receiving officer  Responsibilities of the national department Monitor municipal performance on financial, non-financial, and control systems related to the grant Provide support to municipalities with regards to human settlement delivery as may be required Undertake structured and/or necessary visits to municipalities Facilitate strategic planning, funding and implementation interaction between national and provincia departments of human settlements and accredited municipalities on a regular basis Submit an evaluation report on the 2010/11 municipal annual reports to National Treasury by 15 December 2011 Provide systems (Housing Subsidy System) that support the administration of the human settlement delivery process Conven the evaluation committee for assessing the built environment plans  Responsibilities of the provincial departments Province and coordinate with the
signed-off quarterly reports  a report on performance against the targets provided for in the framework and in conjunction with the reporting requirements of section 71 of the Municipal Finance Management Act must be submitted to National Treasury  Non compliance to the above conditions can result in the funds being withheld or re-allocated  Non compliance to the above conditions can result in the funds being withheld or re-allocated to the above conditions can result in the funds being withheld or re-allocated annexure W1 in the bill  The base allocation is derived from the Municipal Infrastructure Grant formula explained in part 5 or annexure W1 in the bill  The formula incorporates backlog and poverty-weighted data  This is a supplementary infrastructure grant with conditions, objectives and distribution criteria, (e.g. backlogs on infrastructure) different from that of the equitable share  2009/10 audited financial outcomes  This is a new grant that was introduced in 2011 MTEF  2009/10 service delivery performance  This is a new grant that was introduced in 2011 MTEF  2009/10 service delivery performance  This is a new grant that was introduced in 2011 MTEF  2011/12: R6 267 million, 2012/13: R7 410 million and 2013/14: R8 127 million  Quarterly transfers in terms of a payment schedule approved by National Treasury  Responsibilities of the national department  Monitor municipal performance on financial, non-financial, and control systems related to the grant of the provide support to municipalities with regards to human settlement delivery as may be required  Undertake structured and/or necessary visits to municipalities  Monitor municipal performance on financial, non-financial, and control systems related to the grant departments of human settlements and accredited municipalities on a regular basis  Submit an evaluation report on the 2010/11 municipal annual reports to National Treasury by 15 December 2011  Provide systems (Housing Subsidy System) that support the administration of the human settlement del
signed-off quarterly reports  a report on performance against the targets provided for in the framework and in conjunction with the reporting requirements of section 71 of the Municipal Finance Management Act must be submitted to National Treasury  Non compliance to the above conditions can result in the funds being withheld or re-allocated  Non compliance to the above conditions can result in the funds being withheld or re-allocated to the above conditions can result in the funds being withheld or re-allocated annexure W1 in the bill  The base allocation is derived from the Municipal Infrastructure Grant formula explained in part 5 or annexure W1 in the bill  The formula incorporates backlog and poverty-weighted data  This is a supplementary infrastructure grant with conditions, objectives and distribution criteria, (e.g. backlogs on infrastructure) different from that of the equitable share  2009/10 audited financial outcomes  This is a new grant that was introduced in 2011 MTEF  2009/10 service delivery performance  This is a new grant that was introduced in 2011 MTEF  2009/10 service delivery performance  This is a new grant that was introduced in 2011 MTEF  2011/12: R6 267 million, 2012/13: R7 410 million and 2013/14: R8 127 million  Quarterly transfers in terms of a payment schedule approved by National Treasury  Responsibilities of the national department  Monitor municipal performance on financial, non-financial, and control systems related to the grant of the provide support to municipalities with regards to human settlement delivery as may be required  Undertake structured and/or necessary visits to municipalities  Monitor municipal performance on financial, non-financial, and control systems related to the grant departments of human settlements and accredited municipalities on a regular basis  Submit an evaluation report on the 2010/11 municipal annual reports to National Treasury by 15 December 2011  Provide systems (Housing Subsidy System) that support the administration of the human settlement del
- a report on performance against the targets provided for in the framework and in conjunction wither reporting requirements of section 71 of the Municipal Finance Management Act must be submitted to National Treasury  Non compliance to the above conditions can result in the funds being withheld or re-allocated  The base allocation is derived from the Municipal Infrastructure Grant formula explained in part 5 or annexure W1 in the bill  The formula incorporates backlog and poverty-weighted data  Reason not incorporated in equitable share  Past performance  Past performance  2009/10 audited financial outcomes  This is a supplementary infrastructure grant with conditions, objectives and distribution criteria, (e.g. beacklogs on infrastructure) different from that of the equitable share  2009/10 audited financial outcomes  This is a new grant that was introduced in 2011 MTEF  2009/10 service delivery performance  This is a new grant that was introduced in 2011 MTEF  Projected life  The programme will continue up to 2014 subject to review  MTEF allocations  Payment schedule  Responsibilities of transferring national officer and receiving officer  Responsibilities of the national department  Monitor municipal performance on financial, and control systems related to the grant of Provide support to municipalities with regards to human settlement delivery as may be required  Undertake structured and/or necessary visits to municipalities  Facilitate strategic planning, funding and implementation interaction between national and provincia departments of human settlements and accredited municipalities on a regular basis  Submit an evaluation report on the 2010/11 municipal annual reports to National Treasury be 15 December 2011  Provide systems (Housing Subsidy System) that support the administration of the human settlement delivery process  Convene the evaluation committee for assessing the built environment plans  Responsibilities of municipalities  Responsibilities of municipalities
the reporting requirements of section 71 of the Municipal Finance Management Act must be submitted to National Treasury  Non compliance to the above conditions can result in the funds being withheld or re-allocated annexure W1 in the bill  The base allocation is derived from the Municipal Infrastructure Grant formula explained in part 5 of annexure W1 in the bill  The formula incorporates backlog and poverty-weighted data  Reason not incorporated in equitable share  Past performance  2009/10 audited financial outcomes  This is a new grant that was introduced in 2011 MTEF  2009/10 service delivery performance  This is a new grant that was introduced in 2011 MTEF  2009/10 service delivery performance  This is a new grant that was introduced in 2011 MTEF  Projected life  The programme will continue up to 2014 subject to review  MTEF allocations  2011/12: R6 267 million, 2012/13: R7 410 million and 2013/14: R8 127 million  Payment schedule  Responsibilities of the national department  Monitor municipal performance on financial, and control systems related to the grant provide support to municipalities with regards to human settlement delivery as may be required  Undertake structured and/or necessary visits to municipalities  Facilitate strategic planning, funding and implementation interaction between national and provincia departments of human settlements and accredited municipalities on a regular basis  Submit an evaluation report on the 2010/11 municipal annual reports to National Treasury by 15 December 2011  Provide systems (Housing Subsidy System) that support the administration of the human settlement delivery process  Convene the evaluation committee for assessing the built environment plans  Responsibilities of the provincial departments  Promote and coordinate with the relevant municipalities, the planning, funding and implementation human settlements and built environment programmes  Responsibilities of municipalities
submitted to National Treasury  Non compliance to the above conditions can result in the funds being withheld or re-allocated  The base allocation is derived from the Municipal Infrastructure Grant formula explained in part 5 or annexure W1 in the bill  The formula incorporates backlog and poverty-weighted data  Reason not incorporated in the funds being withheld or re-allocated  This is a supplementary infrastructure grant with conditions, objectives and distribution criteria, (e.g. backlogs on infrastructure) different from that of the equitable share  Past performance  2009/10 audited financial outcomes  This is a new grant that was introduced in 2011 MTEF  2009/10 service delivery performance  This is a new grant that was introduced in 2011 MTEF  2009/10 service delivery performance  This is a new grant that was introduced in 2011 MTEF  2009/10 service delivery performance  This is a new grant that was introduced in 2011 MTEF  2009/10 service delivery performance  This is a new grant that was introduced in 2011 MTEF  2009/10 service delivery performance  The programme will continue up to 2014 subject to review  MTEF allocations  Payment schedule  Responsibilities of payment schedule approved by National Treasury  Responsibilities of payment schedule approved by National Treasury  Responsibilities of the national department  Monitor municipal performance on financial, non-financial, and control systems related to the grant of the provide support to municipalities with regards to human settlement delivery as may be required  Undertake structured and/or necessary visits to municipalities on a regular basis  Submit an evaluation report on the 2010/11 municipal annual reports to National Treasury by 15 December 2011  Provide systems (Housing Subsidy System) that support the administration of the human settlement delivery process  Convene the evaluation committee for assessing the built environment plans  Responsibilities of the provincial departments  Promote and coordinate with the relevant municipalities, the
Non compliance to the above conditions can result in the funds being withheld or re-allocated  The base allocation is derived from the Municipal Infrastructure Grant formula explained in part 5 o annexure W1 in the bill The formula incorporates backlog and poverty-weighted data  This is a supplementary infrastructure grant with conditions, objectives and distribution criteria, (e.g. backlogs on infrastructure) different from that of the equitable share  Past performance  O09/10 audited financial outcomes This is a new grant that was introduced in 2011 MTEF  2009/10 service delivery performance This is a new grant that was introduced in 2011 MTEF  Projected life The programme will continue up to 2014 subject to review  Outerly transfers in terms of a payment schedule approved by National Treasury  Responsibilities of transferring national officer and receiving officer  Provide support to municipalities with regards to human settlement delivery as may be required Undertake structured and/or necessary visits to municipalities Facilitate strategic planning, funding and implementation interaction between national and provincia departments of human settlements and accredited municipalities on a regular basis Submit an evaluation report on the 2010/11 municipal annual reports to National Treasury by 15 December 2011 Provide systems (Housing Subsidy System) that support the administration of the human settlement delivery process Convene the evaluation committee for assessing the built environment plans  Responsibilities of the provincial departments Promote and coordinate with the relevant municipalities, the planning, funding and implementation human settlements and built environment programmes  Responsibilities of municipalities Promote and coordinate with the relevant municipalities, the planning, funding and implementation human settlements and built environment programmes
**Allocation criteria**  **The base allocation is derived from the Municipal Infrastructure Grant formula explained in part 5 of annexure W1 in the bill  **The formula incorporates backlog and poverty-weighted data  **This is a supplementary infrastructure grant with conditions, objectives and distribution criteria, (e.g. backlogs on infrastructure) different from that of the equitable share  **Past performance**  **2009/10 audited financial outcomes**  **This is a new grant that was introduced in 2011 MTEF**  **2009/10 service delivery performance**  **This is a new grant that was introduced in 2011 MTEF**  **2009/10 service delivery performance**  **This is a new grant that was introduced in 2011 MTEF**  **Projected life**  **MTEF allocations**  **2011/12: R6 267 million, 2012/13: R7 410 million and 2013/14: R8 127 million**  **Quarterly transfers in terms of a payment schedule approved by National Treasury**  **Responsibilities of the national department**  **Monitor municipal performance on financial, non-financial, and control systems related to the grant of the provincial department of the national and provincial department of human settlements and accredited municipalities on a regular basis of human settlements and accredited municipalities on a regular basis of human settlements and accredited municipalities on a regular basis on the evaluation report on the 2010/11 municipal annual reports to National Treasury by 15 December 2011  **Provide systems (Housing Subsidy System) that support the administration of the human settlement delivery process on convene the evaluation committee for assessing the built environment plans human settlements and built environment programmes  **Responsibilities of the provincial departments*  **Promote and coordinate with the relevant municipalitie
annexure W1 in the bill The formula incorporates backlog and poverty-weighted data This is a supplementary infrastructure grant with conditions, objectives and distribution criteria, (e.g. backlogs on infrastructure) different from that of the equitable share  2009/10 audited financial outcomes This is a new grant that was introduced in 2011 MTEF  2009/10 service delivery performance This is a new grant that was introduced in 2011 MTEF  2009/10 service delivery performance This is a new grant that was introduced in 2011 MTEF  2011/12: R6 267 million, 2012/13: R7 410 million and 2013/14: R8 127 million Payment schedule Responsibilities of the national department  Monitor municipal performance on financial, non-financial, and control systems related to the grant Provide support to municipalities with regards to human settlement delivery as may be required Undertake structured and/or necessary visits to municipalities Facilitate strategic planning, funding and implementation interaction between national and provincia departments of human settlements and accredited municipalities on a regular basis Submit an evaluation report on the 2010/11 municipal annual reports to National Treasury by 15 December 2011 Provide systems (Housing Subsidy System) that support the administration of the human settlement delivery process Convene the evaluation committee for assessing the built environment plans  Responsibilities of municipalities  Responsibilities of municipalities Responsibilities of municipalities Responsibilities of municipalities Responsibilities of municipalities
Reason not incorporated in equitable share  Past performance  2009/10 audited financial outcomes  • This is a new grant that was introduced in 2011 MTEF  2009/10 service delivery performance  • This is a new grant that was introduced in 2011 MTEF  2009/10 service delivery performance  • This is a new grant that was introduced in 2011 MTEF  2009/10 service delivery performance  • This is a new grant that was introduced in 2011 MTEF  Projected life  MTEF allocations  Payment schedule  Responsibilities of transferring national officer and receiving officer  Honitor municipal performance on financial, non-financial, and control systems related to the grant Provide support to municipalities with regards to human settlement delivery as may be required  • Undertake structured and/or necessary visits to municipalities on a regular basis  • Submit an evaluation report on the 2010/11 municipal annual reports to National Treasury by 15 December 2011  • Provide systems (Housing Subsidy System) that support the administration of the human settlement delivery process  • Convene the evaluation committee for assessing the built environment plans  Responsibilities of municipalities  • Promote and coordinate with the relevant municipalities, the planning, funding and implementation human settlements and built environment programmes
Past performance Projected life MTEF allocations Payment schedule Responsibilities of transferring national officer and receiving officer Provide support to municipalities with regards to human settlements and accredited municipalities on a regular basis Submit an evaluation report on the 2010/11 municipal annual reports to National Treasury by 15 December 2011 Provide systems (Housing Subsidy System) that support the administration of the human settlement delivery process Convene the evaluation committee for assessing the built environment plans Responsibilities of the provincial departments and built environment programmes Provide support to municipal performance on financial, non-financial, and control systems related to the grant provide systems (Housing Subsidy System) that support the administration of the human settlement delivery process Convene the evaluation committee for assessing the built environment plans Responsibilities of the provincial departments Provide systems (Housing Subsidy System) that support the administration of the human settlement and built environment programmes Responsibilities of municipalities Responsibilities of the provincial departments Provide systems and built environment programmes Responsibilities of municipalities Responsibilities of municipalities Responsibilities of municipalities
backlogs on infrastructure) different from that of the equitable share  2009/10 audited financial outcomes This is a new grant that was introduced in 2011 MTEF  2009/10 service delivery performance This is a new grant that was introduced in 2011 MTEF  2019/10 service delivery performance This is a new grant that was introduced in 2011 MTEF  2019/10 service delivery performance This is a new grant that was introduced in 2011 MTEF  2011/12: R6 267 million, 2012/13: R7 410 million and 2013/14: R8 127 million  2011/12: R6 267 million, 2012/13: R7 410 million and 2013/14: R8 127 million  2011/12: R6 267 million, 2012/13: R7 410 million and 2013/14: R8 127 million  2011/12: R6 267 million, 2012/13: R7 410 million and 2013/14: R8 127 million  2011/12: R6 267 million, 2012/13: R7 410 million and 2013/14: R8 127 million  2011/12: R6 267 million, 2012/13: R7 410 million and 2013/14: R8 127 million  2011/12: R6 267 million, 2012/13: R7 410 million and 2013/14: R8 127 million  2011/12: R6 267 million, 2012/13: R7 410 million and 2013/14: R8 127 million  2011/12: R6 267 million, 2012/13: R7 410 million and 2013/14: R8 127 million  2011/12: R6 267 million, 2012/13: R7 410 million and 2013/14: R8 127 million  2011/12: R6 267 million, 2012/13: R7 410 million and 2013/14: R8 127 million  2011/12: R6 267 million, 2012/13: R7 410 million and 2013/14: R8 127 million  2011/12: R6 267 million, 2012/13: R7 410 million and 2013/14: R8 127 million  2011/12: R6 267 million, 2012/13: R7 410 million and 2013/14: R8 127 million  2011/12: R6 267 million, 2012/13: R7 410 million and 2013/14: R8 127 million  2011/12: R6 267 million, 2012/13: R7 410 million and 2013/14: R8 127 million  2011/12: R6 267 million, 2012/13: R7 410 million and 2013/14: R8 127 million  2011/12: R6 267 million, 2012/13: R7 410 million and 2013/14: R8 127 million  2011/12: R6 267 million, 2012/13: R7 410 million and 2013/14: R8 127 million  2011/12: R6 267 million, 2012/13: R7 410 million and 2013/14: R8 127 million  2011/12: R6 267 million, 2013/14: R8 127 million  20
This is a new grant that was introduced in 2011 MTEF  2009/10 service delivery performance  This is a new grant that was introduced in 2011 MTEF  Projected life The programme will continue up to 2014 subject to review  MTEF allocations Payment schedule Responsibilities of transferring national officer and receiving officer  Responsibilities of the national department  Monitor municipal performance on financial, non-financial, and control systems related to the grant Provide support to municipalities with regards to human settlement delivery as may be required Undertake structured and/or necessary visits to municipalities Facilitate strategic planning, funding and implementation interaction between national and provincia departments of human settlements and accredited municipalities on a regular basis Submit an evaluation report on the 2010/11 municipal annual reports to National Treasury by 15 December 2011 Provide systems (Housing Subsidy System) that support the administration of the human settlement delivery process Convene the evaluation committee for assessing the built environment plans  Responsibilities of the provincial departments Promote and coordinate with the relevant municipalities, the planning, funding and implementation human settlements and built environment programmes  Responsibilities of municipalities
2009/10 service delivery performance This is a new grant that was introduced in 2011 MTEF  Projected life The programme will continue up to 2014 subject to review  MTEF allocations Payment schedule Responsibilities of transferring national officer and receiving officer  Monitor municipal performance on financial, non-financial, and control systems related to the grant of the provide support to municipalities with regards to human settlement delivery as may be required underpartments of human settlements and accredited municipalities on a regular basis  Submit an evaluation report on the 2010/11 municipal annual reports to National Treasury by 15 December 2011  Provide systems (Housing Subsidy System) that support the administration of the human settlement delivery process Convene the evaluation committee for assessing the built environment plans  Responsibilities of municipalities
• This is a new grant that was introduced in 2011 MTEF  Projected life  ### The programme will continue up to 2014 subject to review  ### The programme will continue up to 2014 subject to review  ### 2011/12: R6 267 million, 2012/13: R7 410 million and 2013/14: R8 127 million  ### Quarterly transfers in terms of a payment schedule approved by National Treasury  ### Responsibilities of the national department  ### Monitor municipal performance on financial, non-financial, and control systems related to the grant  ### Onliver and receiving officer  ### Monitor municipal performance on financial, non-financial, and control systems related to the grant  ### Onliver municipal performance on financial, non-financial, and control systems related to the grant  ### Undertake structured and/or necessary visits to municipalities  ### Facilitate strategic planning, funding and implementation interaction between national and provincial departments of human settlements and accredited municipalities on a regular basis  ### Submit an evaluation report on the 2010/11 municipal annual reports to National Treasury by 15 December 2011  ### Provide systems (Housing Subsidy System) that support the administration of the human settlement delivery process  ### Convene the evaluation committee for assessing the built environment plans  ### Responsibilities of the provincial departments  ### Promote and coordinate with the relevant municipalities, the planning, funding and implementation of human settlements and built environment programmes  ### Responsibilities of municipalities  #### Responsibilities of municipalities  #### Promote and coordinate with the relevant municipalities, the planning, funding and implementation of human settlements and built environment programmes  ##### Responsibilities of municipalities  ####### Responsibilities of municipalities  ###################################
Projected life MTEF allocations Payment schedule Responsibilities of transferring national officer and receiving officer  MTEF allocations Provide support to municipalities with regards to human settlement delivery as may be required Undertake structured and/or necessary visits to municipalities on a regular basis Facilitate strategic planning, funding and implementation interaction between national and provincial departments of human settlements and accredited municipalities on a regular basis Submit an evaluation report on the 2010/11 municipal annual reports to National Treasury by 15 December 2011 Provide systems (Housing Subsidy System) that support the administration of the human settlement delivery process Convene the evaluation committee for assessing the built environment plans  Responsibilities of municipalities Responsibilities of municipalities Responsibilities of municipalities Responsibilities of municipalities
Payment schedule Responsibilities of transferring national officer and receiving officer  Monitor municipal performance on financial, non-financial, and control systems related to the grant of undertake structured and/or necessary visits to municipalities  Pacilitate strategic planning, funding and implementation interaction between national and provincial departments of human settlements and accredited municipalities on a regular basis  Submit an evaluation report on the 2010/11 municipal annual reports to National Treasury by 15 December 2011  Provide systems (Housing Subsidy System) that support the administration of the human settlement delivery process  Convene the evaluation committee for assessing the built environment plans  Responsibilities of municipalities  Promote and coordinate with the relevant municipalities, the planning, funding and implementation of human settlements and built environment programmes  Responsibilities of municipalities
Payment schedule Responsibilities of transferring national officer and receiving officer  Monitor municipal performance on financial, non-financial, and control systems related to the grant of the provide support to municipalities with regards to human settlement delivery as may be required  Undertake structured and/or necessary visits to municipalities  Facilitate strategic planning, funding and implementation interaction between national and provincial departments of human settlements and accredited municipalities on a regular basis  Submit an evaluation report on the 2010/11 municipal annual reports to National Treasury by 15 December 2011  Provide systems (Housing Subsidy System) that support the administration of the human settlement delivery process  Convene the evaluation committee for assessing the built environment plans  Responsibilities of the provincial departments  Promote and coordinate with the relevant municipalities, the planning, funding and implementation of human settlements and built environment programmes  Responsibilities of municipalities
Responsibilities of transferring national officer and receiving officer  Monitor municipal performance on financial, non-financial, and control systems related to the grant  Provide support to municipalities with regards to human settlement delivery as may be required  Undertake structured and/or necessary visits to municipalities  Facilitate strategic planning, funding and implementation interaction between national and provincial departments of human settlements and accredited municipalities on a regular basis  Submit an evaluation report on the 2010/11 municipal annual reports to National Treasury by 15 December 2011  Provide systems (Housing Subsidy System) that support the administration of the human settlement delivery process  Convene the evaluation committee for assessing the built environment plans  Responsibilities of the provincial departments  Promote and coordinate with the relevant municipalities, the planning, funding and implementation of human settlements and built environment programmes  Responsibilities of municipalities
<ul> <li>Monitor municipal performance on financial, non-financial, and control systems related to the grant Provide support to municipalities with regards to human settlement delivery as may be required</li> <li>Undertake structured and/or necessary visits to municipalities</li> <li>Facilitate strategic planning, funding and implementation interaction between national and provincial departments of human settlements and accredited municipalities on a regular basis</li> <li>Submit an evaluation report on the 2010/11 municipal annual reports to National Treasury by 15 December 2011</li> <li>Provide systems (Housing Subsidy System) that support the administration of the human settlement delivery process</li> <li>Convene the evaluation committee for assessing the built environment plans</li> <li>Responsibilities of the provincial departments</li> <li>Promote and coordinate with the relevant municipalities, the planning, funding and implementation of human settlements and built environment programmes</li> <li>Responsibilities of municipalities</li> </ul>
<ul> <li>Provide support to municipalities with regards to human settlement delivery as may be required</li> <li>Undertake structured and/or necessary visits to municipalities</li> <li>Facilitate strategic planning, funding and implementation interaction between national and provincial departments of human settlements and accredited municipalities on a regular basis</li> <li>Submit an evaluation report on the 2010/11 municipal annual reports to National Treasury by 15 December 2011</li> <li>Provide systems (Housing Subsidy System) that support the administration of the human settlement delivery process</li> <li>Convene the evaluation committee for assessing the built environment plans</li> <li>Responsibilities of the provincial departments</li> <li>Promote and coordinate with the relevant municipalities, the planning, funding and implementation of human settlements and built environment programmes</li> <li>Responsibilities of municipalities</li> </ul>
<ul> <li>Undertake structured and/or necessary visits to municipalities</li> <li>Facilitate strategic planning, funding and implementation interaction between national and provincial departments of human settlements and accredited municipalities on a regular basis</li> <li>Submit an evaluation report on the 2010/11 municipal annual reports to National Treasury by 15 December 2011</li> <li>Provide systems (Housing Subsidy System) that support the administration of the human settlement delivery process</li> <li>Convene the evaluation committee for assessing the built environment plans</li> <li>Responsibilities of the provincial departments</li> <li>Promote and coordinate with the relevant municipalities, the planning, funding and implementation of human settlements and built environment programmes</li> <li>Responsibilities of municipalities</li> </ul>
<ul> <li>Facilitate strategic planning, funding and implementation interaction between national and provincial departments of human settlements and accredited municipalities on a regular basis</li> <li>Submit an evaluation report on the 2010/11 municipal annual reports to National Treasury by 15 December 2011</li> <li>Provide systems (Housing Subsidy System) that support the administration of the human settlement delivery process</li> <li>Convene the evaluation committee for assessing the built environment plans</li> <li>Responsibilities of the provincial departments</li> <li>Promote and coordinate with the relevant municipalities, the planning, funding and implementation human settlements and built environment programmes</li> <li>Responsibilities of municipalities</li> </ul>
departments of human settlements and accredited municipalities on a regular basis  Submit an evaluation report on the 2010/11 municipal annual reports to National Treasury by 15 December 2011  Provide systems (Housing Subsidy System) that support the administration of the human settlement delivery process  Convene the evaluation committee for assessing the built environment plans  Responsibilities of the provincial departments  Promote and coordinate with the relevant municipalities, the planning, funding and implementation of human settlements and built environment programmes  Responsibilities of municipalities
<ul> <li>Submit an evaluation report on the 2010/11 municipal annual reports to National Treasury by 15 December 2011</li> <li>Provide systems (Housing Subsidy System) that support the administration of the human settlement delivery process</li> <li>Convene the evaluation committee for assessing the built environment plans</li> <li>Responsibilities of the provincial departments</li> <li>Promote and coordinate with the relevant municipalities, the planning, funding and implementation human settlements and built environment programmes</li> <li>Responsibilities of municipalities</li> </ul>
15 December 2011  Provide systems (Housing Subsidy System) that support the administration of the human settlement delivery process  Convene the evaluation committee for assessing the built environment plans  Responsibilities of the provincial departments  Promote and coordinate with the relevant municipalities, the planning, funding and implementation human settlements and built environment programmes  Responsibilities of municipalities
<ul> <li>Provide systems (Housing Subsidy System) that support the administration of the human settlement delivery process</li> <li>Convene the evaluation committee for assessing the built environment plans</li> <li>Responsibilities of the provincial departments</li> <li>Promote and coordinate with the relevant municipalities, the planning, funding and implementation human settlements and built environment programmes</li> <li>Responsibilities of municipalities</li> </ul>
delivery process  Convene the evaluation committee for assessing the built environment plans  Responsibilities of the provincial departments  Promote and coordinate with the relevant municipalities, the planning, funding and implementation human settlements and built environment programmes  Responsibilities of municipalities
<ul> <li>Convene the evaluation committee for assessing the built environment plans</li> <li>Responsibilities of the provincial departments</li> <li>Promote and coordinate with the relevant municipalities, the planning, funding and implementation human settlements and built environment programmes</li> <li>Responsibilities of municipalities</li> </ul>
Promote and coordinate with the relevant municipalities, the planning, funding and implementation human settlements and built environment programmes  Responsibilities of municipalities
human settlements and built environment programmes  Responsibilities of municipalities
Responsibilities of municipalities
• Submit 2010/11 annual reports to the national department on or before 30 September 2011 and
National Treasury by 15 December 2011
Ensure that the municipal planning, funding and implementation of human settlement and but the stripped of the stripped o
environment development are consistent and aligned with national and provincial strategic and annu
<ul> <li>performance plans</li> <li>Support the national department and province in carrying out the functions delegated as per the</li> </ul>
Constitution, legislation, policy and procedure in relation to human settlements development and the
built environment
Utilise the housing subsidy system for the administration of all the human settlement delivery processe
Comply with the terms and conditions of the national performance agreements and provincial and loc
delivery agreements on the human settlements development outcome
Submit quarterly reports on funds allocated and utilised on programmes and projects in respect of the submit quarterly reports on funds allocated and utilised on programmes and projects in respect of the submit quarterly reports on funds allocated and utilised on programmes and projects in respect of the submit quarterly reports on funds allocated and utilised on programmes and projects in respect of the submit quarterly reports on funds allocated and utilised on programmes and projects in respect of the submit quarterly reports on funds allocated and utilised on programmes and projects in respect of the submit quarterly reports on funds allocated and utilised on programmes and projects in respect of the submit quarterly reports on funds allocated and utilised on programmes and projects in respect of the submit quarterly reports on the submit quarterly reports of the submit qu
grant
Process for approval of  • Municipalities must submit a draft comprehensive and credible funding proposals in a prescribed
2012 MTEF allocations format by 31 March 2011 and final by 7 June 2011, and demonstrate eligibility for the grant in terms of
prescribed requirements

### TRANSPORT GRANTS

	Public Transport Infrastructure and Systems Grant
Transferring department	• Transport (Vote 37)
Strategic goal	To support the Public Transport Strategy (PTS) and Action Plan in promoting the provision of accessible, reliable and affordable Integrated Rapid Public Transport Network (IRPTN) services in the major cities of South Africa in line with the National Land Transport Act (NLTA)
Grant purpose	To provide for accelerated planning, construction and improvement of public and non-motorised transport networks
Outcome statements	Improved public transport network services that are formal, scheduled and well managed and which are accessible to an increasing percentage of the population of major cities
Outputs	<ul> <li>Public transport infrastructure includes development of dedicated lanes, upgrade of network routes, stations, depots and control centres and the fare system</li> <li>Public transport services include buses, security, station management, ticketing and contributions towards the economic rights of existing operators and workers</li> <li>IRPTN plans which should be approved at municipal level and contain:         <ul> <li>network operational plans including universal design access plans</li> <li>engineering and architectural designs</li> <li>vehicle and technology plans</li> <li>institutional and operator business plans and financial, marketing and communication plans for the network services</li> </ul> </li> </ul>
Priority outcome(s) of government that this grant primarily contributes to	Outcome 6: An efficient, competitive and responsive infrastructure network
Details contained in the business plans	This grant uses Integrated Rapid Public Transport Network (IRPTN) operational plans
Conditions	<ul> <li>The allocation of Public Transport Infrastructure and Systems (PTIS) funds must be aligned with the Integrated Transport Plan (ITP) and its IRPTN components as approved by the relevant municipal council</li> <li>From the start of operations, IRPTN systems must recover all the direct operating costs of contracted operators from fare revenue, other local funding sources and, if applicable, from any Public Transport Operations Grant contributions. These direct operational costs include fuel, labour and vehicle maintenance. City-wide networks must ultimately also recover the capital costs of vehicles</li> <li>If buses are bought with grant funds and are used by contracted operators, the municipality must retain ownership</li> <li>Cities are required to establish specialist capacity to manage and monitor IRPTN system contracts and operations as well as to plan future expansions of the network. This capacity must be in place in advance of the first IRPTN operator commencing with service provision to the public</li> <li>Up to R10 million in the 2011/12 financial year may be used to prepare for the public transport regulatory function (see Responsibilities of Municipalities). In future years, funding must be sourced from municipal sources</li> </ul>
Allocation criteria	<ul> <li>The grant is focused on IRPTN implementation in up to 12 cities in accordance with Phases 1 and 2 of the Public Transport Strategy and Action Plan</li> <li>Budget requests will be evaluated in accordance with the outputs of a municipal IRPTN operational plan which specifies the infrastructure, systems and transitional costs of serving a defined number of passenger trips per day by a given fleet of IRPTN vehicles running on a defined amount of exclusive IRPTN infrastructure (including IRPTN stations, feeder stops, depots and exclusive lanes)</li> </ul>
Reason not incorporated in equitable share	Public transport network investment needs are not provided for in the local government equitable share allocations
Past performance	<ul> <li>2009/10 audited financial outcomes</li> <li>Allocated and transferred R2 418 million to municipalities with R4 279 million (176.9 per cent) spent by the end of the of 2009/10 municipal financial year</li> <li>2009/10 service delivery performance</li> <li>43 vehicles were delivered to the City of Cape Town and 24 vehicles to Nelson Mandela Bay municipality</li> <li>The Phase 1a trunk service supported by complimentary and feeder services is operating in the City of Johannesburg. It is 25.5 km with 33 stations and carrying a peak of 34 000 passengers/day</li> <li>City of Cape Town has constructed 17 km of West Coast route and 17 stations between the Civic Centre and Bayside. An 18.5 km Airport to Civic Centre trunk service is currently operating</li> <li>Nelson Mandela Bay has constructed 8.45 km of bus way at an average cost of R34.6 million per kilometre</li> <li>Other cities that have completed operational plans (Tshwane, Polokwane and Rustenburg) have spent most of the PTIS funds allocated up to June 2010 on infrastructure and services for the FIFA World Cup. Tshwane spent over R600 million on road and non-motorised transport infrastructure. eThekwini spent R332 million on strategic projects including R65 million on Warwick Junction, R30 million on electronic ticketing and R20 million on an inner-city distribution system. Match-day operations cost</li> </ul>

Public Transport Infrastructure and Systems Grant
• The grant is expected to continue up to at least 2020 in support of the Public Transport Strategy of 2007
• 2011/12: R4 803 million, 2012/13: R5 000 million and 2013/14: R5 564 million
Transfers are made in accordance with an agreed payment schedule by National Treasury
Responsibilities of the national department  Disburse PTIS funds and monitor PTIS expenditure  Monitor IRPTN implementation progress in line with the National Land Transport Act and the Public Transport Strategy  Verify reports from municipalities by conducting at least one site visit per annum  Allocate the funds based on stated priorities through a Joint PTIS Committee comprising the Department of Transport and National Treasury  Evaluate the performance of the grant annually  Responsibilities of municipalities  Ensure that projects are implemented in line with what is reflected in the integrated development plan of the municipality  Reporting is done correctly on the management of this grant and all relevant DoRA requirements are adhered to  Provide budget proposals for the PTIS, IRPTN funding that are based on sound operational plans  Compile and submit data that indicates the efficiency and effectiveness of planned and actual IRPTN services including:  number of weekday passenger trips carried on IRPTN systems  change, relative to the previous year, in the number and percentage of households within 500 metres of IRPTN access points (stations and feeder service stops)  number of kilometres of bi-directional, exclusive lanes in operation in IRPTN systems  number of IRPTN network stations and feeder service stops in operation  planned/actual capital expenditure per IRPTN passenger kilometre  planned/actual operational expenditure per IRPTN passenger kilometre  size of the formal IRPTN vehicle fleet in operation (per contracted trunk, complementary and feeder vehicle type) as well as passenger trips per trunk/complementary vehicle per weekday  provide detailed information on the actual costs of procuring IRPTN inputs including those for infrastructure, systems and transitional and regulatory items  establish a dedicated IRPTN project team to implement the system in the development phase and subsequently once IRPTN services to the public have commenced  establish a specialist capacity to manage and monitor oper
<ul> <li>Municipalities will be requested to submit budget proposals that are based on sound IRPTN operational plans by 15 July 2011</li> </ul>
These requests will be evaluated by a Joint PTIS committee comprising the Department of Transport
<ul> <li>and National Treasury</li> <li>Municipal provisional allocations will be finalised by 30 November 2011</li> </ul>

	Rural Transport Services and Infrastructure Grant
Transferring department	Transport (Vote 37)
Strategic goal	Ensure efficient and effective investment in rural roads through development of Road Asset Management Systems (RAMS)
Grant purpose	To assist rural district municipalities to set up rural road asset management systems, and collect road and traffic data in line with the Road Infrastructure Strategic Framework for South Africa (RISFSA)
Outcome statements	<ul> <li>Improve data on rural roads to guide infrastructure investments</li> <li>Reduce vehicle operating costs and extend the lifespan of rural roads</li> </ul>
Outputs	<ul> <li>Collection of road inventory data including condition assessment and traffic data</li> <li>Setting up pavement and bridge management systems compatible with national standards</li> </ul>
Priority outcome(s) of government that this grant primarily contributes to	Outcome 6: An efficient, competitive and responsive economic infrastructure network     Outcome 7: Vibrant, equitable and sustainable rural communities and food security
Details contained in the business plan	Not applicable as 2011 MTEF allocations are meant to be used to collect data on rural roads
Conditions	<ul> <li>Data must be collected and presented in the format prescribed by RISFSA</li> <li>Data collection should use labour intensive methods that comply with Expanded Public Works Programme (EPWP) guidelines</li> <li>All data collected must be made available to the national Department of Transport (DoT), South African National Roads Agency Limited (SANRAL) and the relevant provincial roads authorities</li> <li>Systems developed to record data must be compatible with Department of Transport specifications</li> </ul>
Allocation criteria	Amount equally shared among 21 rural district municipalities
Reason not incorporated in equitable share	This is a specific purpose grant mainly for the provision of systems to collect rural road and traffic data
	<ul> <li>Allocated and transferred R9.8 million to Chris Hani District municipality and R7.1 million (72. 4 per cent) was spent by the end of the 2009/10 municipal financial year</li> <li>2009/10 service delivery performance</li> <li>The allocations for this grant were spent towards design and construction of an intermodal public</li> </ul>
Projected life	transport facility in Chris Hani District Municipality  The grant has a life span up to 2013/14
MTEF allocations	2011/12: R35 million, 2012/13: R37 million and 2013/14: R39 million
Payment schedule	Transfers are made in accordance with a payment schedule approved by National Treasury
Responsibilities of the transferring national officer and receiving officer	Responsibilities of national department  Monitoring implementation of RAMS together with provincial road authorities  Data integrity will be checked by DoT and provincial road authorities  Provide guidance on sustainable RAMS operation and standards
	Facilitate training to municipalities and assist them to acquire RAMS from SANRAL     Check the quality of data captured on municipalities' RAMS
	<ul> <li>Responsibilities of municipalities</li> <li>Municipalities must make provision to maintain RAMS after the lifespan of the grant</li> <li>Data for all rural roads to be updated within two years</li> <li>Ensure human capacity at municipalities for the operation of RAMS is built</li> <li>Road quality data on RAMS will be a conditionality for Municipal Infrastructure Grant (Roads) in 2012/13</li> </ul>
Process for approval of 2012 MTEF allocations	<ul> <li>Municipalities must submit business plans to the Department of Transport by 30 September 2011</li> <li>Business plans must contain the following:         <ul> <li>the extent of the road network in the municipality</li> <li>the condition of the network in the municipality</li> <li>the status of the municipality's RAMS</li> <li>the proportion of municipal roads with updated data captured on its RAMS</li> </ul> </li> <li>The Department of Transport together with provincial roads authorities will evaluate the business plans and review technical reports by 15 November 2011</li> </ul>

### WATER AFFAIRS GRANTS

	Regional Bulk Infrastructure Grant
Transferring department	Water Affairs (Vote 38)
Strategic goal	Facilitate achieving water targets through successful execution and implementation of regional bulk infrastructure projects
Grant purpose	To develop regional bulk infrastructure for water supply to supplement water treatment works at resource development and link such water resource development with the local bulk and local distribution networks on a regional basis cutting across several local municipal boundaries
	<ul> <li>In the case of sanitation, to supplement regional bulk collection as well as regional waste water treatment works</li> </ul>
Outcome statements	<ul> <li>Access to water supply enabled through the establishment of regional bulk infrastructure</li> <li>Proper waste water management and disposal will be effected through establishment of regional sanitation infrastructure (bulk sewer pipelines and waste water treatment works)</li> </ul>
Outputs	<ul> <li>Number of implementation readiness plans finalised for regional bulk projects</li> <li>Number of regional bulk projects initiated</li> </ul>
	<ul> <li>Number of projects completed</li> <li>Number of people or households being impacted due to a new regional bulk system</li> </ul>
	Number of villages/areas/municipalities to benefit from the regional bulk system
	Number of jobs created
Priority outcome(s) of	Outcome 5: A skilled and capable workforce to support an inclusive growth path
government that this grant	Outcome 6: An efficient, competitive and responsive economic infrastructure network
primarily contributes to	
Details contained in the	Outcome indicators
business plan	Output indicators
	Key activities     Inputs
Conditions	<ul> <li>Inputs</li> <li>The emphasis is only on the social component of regional bulk water and sanitation services</li> </ul>
Conditions	<ul> <li>The emphasis is only on the social component of regional bulk water and samutation services</li> <li>This grant can be used to build enabling infrastructure required to connect water resources over significant distances with bulk and reticulation system</li> </ul>
	The need for a bulk infrastructure solution must be confirmed and accepted
	• Financing plan with associated/co-funding options and agreements must be in place prior to implementation
	<ul> <li>All sources of funding must be outlined in the business plan</li> <li>Regional Bulk Infrastructure Grant (RBIG) allocations will be transferred to Water Services Authorities (WSA) or institutions which have capacity to deliver the service</li> </ul>
	<ul> <li>Department of Water Affairs must register the asset in their own register before transferring it to the WSA</li> </ul>
	The transfer of infrastructure will be done on condition that the new owner has capacity to operate and maintain the scheme  Output  Description:
	Projects will be assessed individually, no blanket approach allocation method will be utilised  All projects must be aligned with and referenced to Integrated Development Plan (IDP) and Water  All projects must be aligned with and referenced to Integrated Development Plan (IDP) and Water
	All projects must be aligned with and referenced to Integrated Development Plan (IDP) and Water Services Development Plans as well as a detailed plan which shows alignment of RBIG and Municipal Infrastructure Grant (MIG) projects
Allocation criteria	Allocations are made on a project basis and must take into account the conditions listed above
Reason not incorporated in equitable share	Regional bulk projects are closely linked to water resource development which is a Water Affairs competency
	Bulk water supply systems frequently cross municipal boundaries and so cannot be funded at municipal level
Past performance	2009/10 audited financial outcome
	• Allocated R571 million was transferred to the national Department of Water Affairs (DWA) for
	implantation of this grant  2009/10 service delivery performance
	<ul> <li>Projects completed (6), in the following local municipalities:</li> </ul>
	- Mbhashe LM with 4 901 people benefitting
	- Umzumbe LM with 21 100 people benefitting
	- Mthonjaneni/ Nkandla LM with 26 327 people benefitting
	- Nkomazi LM with 3 977 people benefiting
	- Kareeberg LM with 1 924 people benefiting
	- Renosterberg LM with 13 553 people benefiting
	• 32 Projects in construction phase
	18 Projects are in design or tender phase
	48 Projects are in feasibility stage

	Regional Bulk Infrastructure Grant
Projected life	The life span of the grant depends on the eradication of the water services backlogs
MTEF allocations	• 2011/12: R1 704 million, 2012/13: R 2 003 million and 2013/14: R2 176 million
Payment schedule	Transfers are made in accordance with a payment schedule approved by National Treasury
Responsibilities of the	Responsibilities of national department
transferring national	The national department must:
officer and receiving officer	<ul> <li>Ensure every municipality benefiting from a specific project or scheme, is invited to participate in the feasibility study</li> </ul>
	<ul> <li>Enter into agreements with municipalities regarding the construction, ownership, funding arrangements and operation and maintenance of proposed infrastructure prior to the commencement of construction</li> </ul>
	Make quarterly visits to projects
	- Provide detailed information on the selection criteria and conditions for the grant
	<ul> <li>Monitor implementation by water services institutions (municipalities or water boards)</li> </ul>
	Direct implementation where capacity gaps exist
	- Ensure that service level agreements as well as Memorandums of Understanding are in place
	between Water Services Providers and Water Services Authorities
	Responsibilities of WSA/ Institutions which schemes will be transferred to
	Submission of quarterly reports and annual reports by municipalities
	Ensure that projects are appropriately linked to the municipality's IDP
	Ensure that the municipality is able to provide the reticulation services required to provide households
	with access to the water provided through the bulk infrastructure funded by this grant
Process for approval of	Implementation readiness studies will be evaluated and approved if the feasibility studies comply with
2012/13 business plans	the RBIG implementation framework and criteria
	At a regional level a committee comprising of representatives from the DWA, National Treasury,
	Department of Cooperative Governance and Traditional Affairs, and the South African Local
	Government Association meet and approve and prioritise projects
	At national level projects are allocated a budget by DWA, based on the implementation framework
	guidelines and National Treasury will be informed of the decisions taken

	Municipal Drought Relief Grant
Transferring department	Water Affairs (Vote 38)
Strategic goal	Ensuring access to water supply in the declared and gazetted drought disaster area of Nelson Mandela Bay metropolitan municipality
Grant purpose	To provide capital finance for construction of appropriate water infrastructure to alleviate further impacts of drought in Nelson Mandela Bay metropolitan municipality for affected households, micro enterprises and social institutions
Outcome statements	<ul> <li>All citizens in drought declared area have access to water supply</li> <li>Impacts of drought alleviated in the municipality</li> </ul>
Outputs	Number of households, micro enterprises and social institutions receiving water
Details contained in the business plans	<ul> <li>The number of households, micro enterprises and social institutions, without access to basic services that will be impacted</li> <li>Type of drought intervention infrastructure developed and registered in municipal asset register</li> </ul>
	Detailed design approved by a professional engineer, schedule of quantities, activity-based budget and scheme layout
	Project implementation plan and deliverables
Duionity outo(-) -f	Alignment of the project outcome with the municipality's long-term water services planning  Outcome (A. A. efficient account titing and propagation account infraction to the project outcome.)
Priority outcome(s) of government that this grant primarily contributes to	Outcome 6: An efficient, competitive and responsive economic infrastructure network
Conditions	Funds must be spent in accordance with:
	- municipal drought implementation plans submitted to the national Department of Water Affairs (DWA)
	- technical reports developed by the metropolitan municipality and assessed by DWA
	Monthly cash flow projections and progress reports must be submitted to DWA and presented to provincial and national coordination and management structures
	Affected municipalities must:     account to DWA for the allocated funds on a monthly basis by the 10th of every month as per Division of Revenue Act (DoRA)
	- adhere to the approved drought programme and agreed cash flow budgets
	<ul> <li>reflect all assets created under the special drought relief funds in the municipal asset register</li> <li>use drought relief funds to provide new or refurbished critical infrastructure, only upon submission of a</li> </ul>
	technical report which must be approved and regulated by DWA
	Operation and maintenance plans specific to drought intervention infrastructure must be incorporated and funded through normal municipal processes
Allocation criteria	Allocation based on application received and extent of need
Reason not incorporated in equitable share	This is a specific grant with objectives and criteria different from that of the equitable share
Past performance	2009/10 audited financial outcomes
-	New allocation to Nelson Mandela Bay metropolitan municipality
	2009/10 service delivery performance
	Not applicable
Projected life	One year
MTEF allocations	• 2011/12: R450 million
Payment schedule	Transfers are made in accordance with a payment schedule approved by National Treasury
Responsibilities of the	Responsibilities of the national department
transferring national	Administer the grant and coordinate all stakeholders through the appropriate structures
officer and receiving officer	Monitor the overall programme implementation    Description of programme implementation
UIIICU	Responsibilities of municipality
<b>T</b>	No responsibilities additional to standard DoRA requirements
Process for approval of 2012 MTEF allocations	Not applicable

	Water Services Operating Subsidy Grant
Transferring department	Water Affairs (Vote 38)
Strategic goal	<ul> <li>To ensure effective, efficient and sustainable service delivery by all Water Services Authorities (WSA) or municipalities</li> </ul>
Grant purpose	To subsidise and build capacity in water schemes owned and/or operated by the Department of Water Affairs or by other agencies on behalf of the department and transfer these schemes to local government
Outcomes statements	To ensure that transferred schemes are fully functional and operated by skilled personnel to ensure optimal service delivery by WSA
	<ul> <li>To ensure that funds transferred to WSA are utilised for the intended purpose and that there is proper accountability by both receiving and transferring institutions</li> </ul>
Outputs	<ul> <li>Operating outputs as defined in the business plan for each individual scheme:         <ul> <li>transfer workable and fully functional schemes to municipalities in accordance with the Water Services Act</li> <li>build capacity and skills to run schemes</li> </ul> </li> <li>Transfer outputs as outlined in the policy framework and business plan:         <ul> <li>schemes refurbished to standards outlined in terms of the agreed policy framework</li> <li>sustainability assessments completed per scheme or group of schemes to be transferred</li> <li>Water services authority/provider has developed sufficient capacity in line with funding requirements</li> <li>number of transferred schemes with adequate capacity and skills to function optimally</li> </ul> </li> </ul>
Details contained in the business plans	<ul> <li>Outcome indicators</li> <li>Output indicators</li> <li>Key activities</li> <li>Inputs</li> </ul>
Priority outcome(s) of government that this grant primarily contributes to	Outcome 9 : Responsive, accountable, effective and efficient local government system
Conditions	<ul> <li>The operating and transfer subsidy is a grant-in-kind until the effective date of transfer</li> <li>The operating subsidy (grant-in-kind) will cover staff related costs (HR component), the direct operating and maintenance costs (O component), the refurbishment costs and will facilitate the transfer of schemes</li> <li>All receiving municipalities and providers will be required to conclude formal transfer agreements</li> <li>This grant must be used to build the necessary capacity to operate, maintain and further develop the infrastructure to be taken over by the relevant institution</li> <li>Reporting on the grant will be done on a monthly basis</li> <li>The transferring national department may adjust the allocation to a municipality, subject to approval from National Treasury, to reflect: <ul> <li>the actual personnel allocation payable to a municipality as a result of the number of staff transferred to the municipality</li> <li>the actual operating allocation payable to a municipality, as informed by the percentage or portion of assets transferred to a municipality in respect of assets shared across municipal boundaries</li> </ul> </li> </ul>
Allocation criteria	<ul> <li>Allocation as agreed in the transfer agreement, in line with a business plan for the project and agreed upon with National Treasury</li> <li>Schedule 6 will cover human resources (compensation of employees), operations and maintenance (goods and services) and refurbishment</li> <li>Schedule 7 will cover human resources (compensation of employees) and operations and maintenance (goods and services)</li> </ul>
Reason not incorporated in equitable share	To establish functional, viable and sustainable capacity in municipalities to provide water services and will be incorporated into the equitable share subject to an assessment of capacity and the timelines in business plans
Past performance	<ul> <li>2009/10 audited financial outcomes</li> <li>Schedule 6 direct transfers to municipalities, allocated R871million and transferred R849 million to municipalities with R862 million (98.9 per cent) spent by the end of the 2009/10 municipal financial year</li> <li>Schedule 7 allocation in kind for municipalities, allocated R130 million</li> <li>2009/10 service delivery performance</li> <li>59 agreements signed, 4 903 staff transferred in total, 1 051 staff transferred during 2009/10. Total number of schemes transferred is 1 643 this includes 1 155 rudimentary schemes and 488 schemes with a total asset value of approximating R6.4 billion</li> <li>Currently 1 181 schemes have been refurbished at a cost of R973 million</li> </ul>
	Of 18 337 components targeted, 15 427 have been refurbished

	Water Services Operating Subsidy Grant
Projected life	Subject to the outcome of a review of the grant in 2011/12
MTEF allocations	Direct transfers to municipalities (Schedule 6)     2011/12: R561 million, 2012/13: R399 million and 2013/14: R421 million     Allocation in kind to municipalities/WSA (Schedule 7)     2011/12: R100 million
Payment schedule	Payments on direct transfers will be made on a quarterly basis and based on performance, as agreed to in the transfer agreement for each specific scheme in a municipality/WSA and also according to a payment schedule approved by National Treasury
Responsibilities of the	Responsibilities of national department
national transferring officer and receiving officer	<ul> <li>Agree with municipalities on outputs and targets</li> <li>Continuously monitor implementation and provide support to municipalities to enable independence in managing schemes</li> <li>Report to National Treasury on a monthly basis about the performance of the grant (in the prescribed format determined by National Treasury) to ensure compliance with the grant conditions</li> <li>Implement a comprehensive information monitoring and evaluation system</li> </ul>
	Submit monthly and quarterly progress reports on expenditure in a format agreed by National Treasury     Submit a comprehensive review of the current status of the subsidy
	Responsibilities of municipalities and WSAs
	Municipalities or WSA will submit monthly and quarterly reports in the format prescribed by National Treasury and DWA
	Submit of operating and maintenance plans as well as refurbishment plans in compliance with the standard set by DWA
Process for approval of	2012/13 business plans to be signed and formalised by December 2011
2012/13 business plan	<ul> <li>Specific support to enhance ability and capacity to enable schemes function optimally</li> <li>Break-down of cost of refurbishment and implications</li> <li>A transfer agreement and addendum in place which contains revised allocations and actual funds to</li> </ul>
	be transferred  A comprehensive assessment of the status report on management of the WSAs

### ANNEXURE W4

# SPECIFIC PURPOSE ALLOCATIONS TO MUNICIPALITIES (SCHEDULE 6): RECURRENT GRANTS

## (National and Municipal Financial Years)

ANNEXURE W4
SPECIFIC PURPOSE RECURRENT GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULE 6) 1 OF 2

			Local Cos	Local Covernment Enemois Monocoment Cront	oiol Monogom	ont Cront			Muni	Minicinal Systems Improvement Grant	Jaomono	- mont	
		Natio	National Financial Year	Year	Munic	Municipal Financial Year	Year	Natio	National Financial Year	Year	Munic	Municipal Financial Year	Year
Category	ory Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
EASTERN CAPE	CAPE												
4 4	BUF Buffalo City NMA Nelson Mandela Bay	1 450 1 250	1 500 1 250	1 500	1 450 1 250	1 500	1500						
В	EC101 Camdeboo	2 000	2 000	2 000	2 000	2 000	2 000	790	800	850	790	800	850
В		1 450	1 500	1 500	1 450	1 500	1 500	790	800	850	790	800	850
В	EC103 Ikwezi	1 450	1 500	1 500	1 450	1 500	1 500	790	800	850	790	800	850
В		1 450	1 500	1 500	1 450	1 500	1 500	790	800	850	790	800	850
В Е		2 000	2 000	2 000	2 000	2 000	2 000	790	800	850	790	800	820
В	_	1 500	1 500	1 500	1 500	1 500	1 500	790	800	820	790	800	820
B		1 500	1 250	1 500	1 500	1 250	1 500	790	800	820	790	800	820
B		1 450	1 500	1 500	1 450	1 500	1 500	790	800	820	790	800	820
B		1 250	1 500	1 500	1 250	1 500	1 500	790	800	850	790	800	850
ŭ	DC10 Cacadu District Municipality	1 250	1 500	1 500	1 250	1 500	1 500	790	1 000	1 000	790	1 000	1 000
Total: Cac	Fotal: Cacadu Municipalities	15 300	15 750	16 000	15 300	15 750	16 000	7 900	8 200	8 650	7 900	8 200	8 650
В	EC121 Mbhashe	1 500	1 500	1 500	1 500	1 500	1 500	790	800	006	790	800	006
В	EC122 Mnquma	1 450	1 500	1 500	1 450	1 500	1 500	790	800	006	790	800	006
B I		1 500	1 500	1 500	1 500	1 500	1 500	790	800	006	790	800	006
В		1 500	1 500	1 500	1 500	1 500	1 500	790	800	006	790	800	006
В		1 450	1 500	1 500	1 450	1 500	1 500	790	800	006	790	800	006
В		1 450	1 500	1 500	1 450	1 500	1 500	790	800	006	790	800	006
В		1 450	1 500	1 500	1 450	1 500	1 500	790	800	006	790	800	006
C	DC12 Amathole District Municipality	1 250	1 373	1 500	1 250	1 373	1 500	790	1 000	0006	790	1 000	006
Total: Ama	Fotal: Amathole Municipalities	11 550	11 873	12 000	11 550	11 873	12 000	6 320	0099	7 200	6 320	9 6 600	7 200
B E	EC131 Inxuba Yethemba	1 500	1 500	1 500	1 500	1 500	1 500	790	800	850	790	800	850
В	EC132 Tsolwana	1 250	1 500	1 500	1 250	1 500	1 500	790	800	850	790	800	820
В	EC133 Inkwanca	2 000	1 500	1 500	2 000	1 500	1 500	790	800	850	790	800	850
	EC134 Lukhanji	1 500	1 500	1 500	1 500	1 500	1 500	790	800	850	790	800	820
		1 500	1 500	1 500	1 500	1 500	1 500	790	800	850	790	800	820
В	EC136 Emalahleni	1 500	1 500	1 500	1 500	1 500	1 500	790	800	850	790	800	820
В		1 450	1 500	1 500	1 450	1 500	1 500	790	800	850	790	800	820
В		1 500	1 500	1 500	1 500	1 500	1 500	790	800	850	790	800	820
C C	DC13 Chris Hani District Municipality	1 250	1 500	1 500	1 250	1 500	1 500	190	1 000	850	790	1 000	820
Total: Chr.	Fotal: Chris Hani Municipalities	13 450	13 500	13 500	13 450	13 500	13 500	7 110	7 400	7 650	7 110	7 400	7 650

ANNEXURE W4
SPECIFIC PURPOSE RECURRENT GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULE 6) 1 OF 2

		Local Gov	Local Government Financial Management Grant	cial Manageme	nt Grant			Mun	icipal Systems	Municipal Systems Improvement Grant	Frant	
	Nati	National Financial Year	Year	Munic	Municipal Financial Year	Year	Natio	National Financial Year	Year	Munic	Municipal Financial Year	Year
	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14
Category	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)
B EC141 Elundini	1 500	1 500	1 500	1 500	1 500	1 500	790	800	006	790	800	006
B EC142 Sengu	1 450	1 500	1 500	1 450	1 500	1 500	790	800	006	790	800	006
B EC143 Maletswai	1 500	1 500	1 500	1 500	1 500	1 500	790	800	006	790	800	006
B EC144 Gariep	1 450	1 500	1 500	1 450	1 500	1 500	790	800	006	790	800	006
C DC14 Ukhahlamba District Municipality	1 250	1 250	1 500	1 250	1 250	1 500	790	800	006	790	800	006
Total: Ukhahlamba Municipalities	7 150	7 250	7 500	7 150	7 250	7 500	3 950	4 000	4 500	3 950	4 000	4 500
B EC153 Ngquza Hill	1 450	1 500	1 500	1 450	1 500	1 500	790	800	850	790	800	820
B EC154 Port St Johns	2 000	1 500	1 500	2 000	1 500	1 500	790	800	850	790	800	850
B EC155 Nyandeni	1 450	1 500	1 500	1 450	1 500	1 500	790	800	850	790	800	850
B EC156 Mhlontlo	1 450	1 500	1 500	1 450	1 500	1 500	790	800	850	790	800	850
B EC157 King Sabata Dalindyebo	1 450	1 500	1 500	1 450	1 500	1 500	790	800	850	790	800	820
C DC15 OR Tambo District Municipality	1 250	1 500	1 500	1 250	1 500	1 500	790	800	850	790	800	850
Total: O.R.Tambo Municipalities	9 050	000 6	000 6	9 050	0006	0006	4 740	4 800	5 100	4 740	4 800	5 100
B EC441 Matatiele	1 450	1.500	1 500	1 450	1 500	1 500	790	800	850	790	800	850
B EC442 Umzimvubu	1 450	_	1 500	1 450	1 500	1 500	790	800		790	800	850
	1 450	1 500	1 500	1 450	1 500	1 500	790	800		790	800	820
B EC152 Ntabankulu	1 500	1 500	1 500	1 500	1 500	1 500	790	800	850	790	800	820
C DC44 Alfred Nzo District Municipality	1 250	1 500	1 500	1 250	1 500	1 500	790	800	850	790	800	850
Total: Alfred Nzo Municipalities	7 100	005 L	7 500	7 100	7 500	7 500	3 950	4 000	4 2 5 0	3 950	4 000	4 250
Total: Eastern Cape Municipalities	006 99	67 623	68 250	99	67 623	68 250	33 970	35 000	37 350	33 970	35 000	37 350
									•			

ANNEXURE W4
SPECIFIC PURPOSE RECURRENT GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULE 6) 1 OF 2

			Local Gove	Local Government Financial Management Grant	cial Manageme	ent Grant			Muni	cipal Systems I	Municipal Systems Improvement Grant	rant	
		Nation	National Financial Year	/ear	Munic	Municipal Financial Year	Year	Natio	National Financial Year	Year	Munic	Municipal Financial Year	Year
Category	Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
FREE STATE													
A MAN	Mangaung	1 450	1 500	1 500	1 450	1 500	1 500						
B FS161	Letsemeng	1 450	1 500	1 500	1 450	1 500	1 500	790	800	850	790	800	850
		1 450	1 500	1 500	1 450	1 500	1 500	790	800	850	790	800	850
B FS163	Mohokare	1 500	1 500	1 500	1 500	1 500	1 500	790	800	850	790	800	850
		1 450	1 500	1 500	1 450	1 500	1 500	790	800	850	790	800	850
C DC16	Xhariep District Municipality	1 250	1 250	1 250	1 250	1 250	1 250	790	800	850	790	008	850
Total: Xhariep Municipalities	Aunicipalities	7 100	7 250	7 250	7 100	7 250	7 250	3 950	4 000	4 250	3 950	4 000	4 250
B FS181	Masilonyana	1 250	1 500	1 500	1 250	1 500	1 500	790	800	850	790	800	850
		1 450	1 500	1 500	1 450	1 500	1 500	790	800	850	790	800	850
B FS183	Tswelopele	1 450	1 500	_	1 450	1 500	1 500	840	840	850	840	840	850
B FS184	Matjhabeng	1 450	1 500	1	1 450	1 500	1 500	790	800	850	790	800	850
B FS185	Nala	1 500	1 500	1	1 500	1 500	1 500	790	800	850	790	800	850
C DC18	Lejweleputswa District Municipality	1 250	1 250	1 250	1 250	1 250	1 250	790	1 000	850	790	1 000	850
Total: Lejwelepu	Total: Lejweleputswa Municipalities	8 350	8 750	8 750	8 350	8 750	8 750	4 790	5 040	5 100	4 7 9 0	5 040	5 100
B FS191	Setsofo	1 450	1.500	1 500	1 450	1 500	1 500	790	800	850	790	800	850
		1 450	1 500	1 500	1 450	1 500	1 500	790	800	850	790	800	850
B FS193		1 450	1 500	1 500	1 450	1 500	1 500	790	800	850	790	800	850
B FS194	. Maluti a Phofung	1 450	1 500	1 500	1 450	1 500	1 500	790	800	850	790	800	850
B FS195	Phumelela	1 300	1 500	1 500	1 300	1 500	1 500	790	800	850	790	800	850
B FS196	Mantsopa	1 450	1 500	1 500	1 450	1 500	1 500	790	800	850	790	800	850
C DC19		1 250	1 250	1 250	1 250	1 250	1 250	790	1 000	850	790	1 000	850
Total: Thabo Mc	Total: Thabo Mofutsanyana Municipalities	008 6	10 250	10 250	008 6	10 250	10 250	5 530	5 800	5 950	5 530	5 800	5 950
B FS201	Moqhaka	1 450	1 500	1 500	1 450	1 500	1 500	790	800	850	790	800	850
B FS203	Ngwathe	1 450	1 500	1 500	1 450	1 500	1 500	2007	800	850	790	800	820
B FS204		1 450	1 500	1 500	1 450	1 500	1 500	790	800	850	790	800	850
B FS205	Mafube	1 450	1 500	1 500	1 450	1 500	1 500	790	800	850	790	800	820
C DC20	Fezile Dabi District Municipality	1 250	1 250	1 250	1 250	1 250	1 250	790	1 000	1 000	790	1 000	1 000
Total: Fezile Dabi Municipalities	bi Municipalities	7 050	7 250	7 250	7 050	7 250	7 250	3 950	4 200	4 400	3 950	4 200	4 400
Total: Free State Municipalities	Municipalities	33 750	35 000	35 000	33 750	35 000	35 000	18 220	19 040	19 700	18 220	19 040	19 700

ANNEXURE W4
SPECIFIC PURPOSE RECURRENT GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULE 6) 1 OF 2

			Local Gov	Local Government Financial Management Grant	icial Manageme	nt Grant			Mun	icipal Systems l	Municipal Systems Improvement Grant	rant	
		Natio	National Financial Year	Year	Munic	Municipal Financial Year	Year	Natio	National Financial Year	Year	Munic	Municipal Financial Year	Year
7	M	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14
Category	Municipainty	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)
GAUTENG													
A EKU Ekurhuleni	ï	1 250	1 250	1 250	1 250	1 250	1 250						
	City of Johannesburg	1 250	1 250	1 250	1 250	1 250	1 250						
A TSH City of Tshwane	hwane	5 250	5 000	5 000	5 250	5 000	5 000						
B GT421 Emfuleni		1 250	1 250	1 250	1 250	1 250	1 250	800	1 000	1 000	800	1 000	1 000
B GT422 Midvaal		1 250	1 250	1 250	1 250	1 250	1 250	800	1 000	1 000	800	1 000	1 000
B GT423 Lesedi		1 250	1 250	1 250	1 250	1 250	1 250	800	1 000	1 000	800	1 000	1 000
C DC42 Sedibeng 1	DC42 Sedibeng District Municipality	1 250	1 250	1 250	1 250	1 250	1 250	800	1 000	1 000	800	1 000	1 000
Total: Sedibeng Municipalities	es	2 000	2 000	2 000	2 000	2 000	2 000	3 200	4 000	4 000	3 200	4 000	4 000
O closed Message		030	050	0201	0301	0301	030	000	000	-	6	-	-
GH491	lily	1 250	1 250	1 250	1 250	1 250	1 250	000	1 000	0001	000	000 -	000 -
B G1482 Kandrontein	lli.	1 250	1 250	1 250	1 250	1 250	1 250	008	1 000	1 000	900	000 -	1 000
G1483		057 1	062.1		0.00	067 1	007 1	900	1 000	000 1	000	000 1	000 1
B GT484 Merafong	City	1 250	1 500	-	1 250	1 500	1 500	800	1 000	1 000	800	1 000	1 000
C DC48 West Rand	West Rand District Municipality	1 250	1 250	1 250	1 250	1 250	1 250	800	1 000	1 000	800	1 000	1 000
Total: West Rand Municipalities	ities	6 250	002 9	9 200	6 250	002 9	0 200	4 000	2 000	2 000	4 000	2 000	5 000
Total: Gauteng Municipalities	Se	19 000	19 000	19 000	19 000	19 000	19 000	7 200	0006	0006	7 200	000 6	000 6

ANNEXURE W4
SPECIFIC PURPOSE RECURRENT GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULE 6) 1 OF 2

			Local Gove	Local Government Financial Management Grant	ial Managemo	ent Grant			Mun	icipal Systems	Municipal Systems Improvement Grant	rant	
		National	National Financial Year	ar	Munic	Municipal Financial Year	Year	Natio	National Financial Year	Year	Munic	Municipal Financial Year	Year
Category Municipality	. 2011. (R'06	12 ))	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
KWAZULU-NATAL													
A ETH eThekwini		1 250	1 500	1 250	1 250	1 500	1 250						
B KZN711 Virlamehlo		1 500	0051	1 750	1 500	1 500	1 750	007	008	000	092	OUS	000
B KZN213 v maniemo		1 450	1 500	1 750	1 450	1 500	1 750	06/	008	006	067	008	006
		1 500	1 500	1 750	1 500	1 500	1 750	790	800	006	790	800	006
		1 450	1 500	1 750	1 450	1 500	1 750	790	800	006	790	800	006
		1 500	1 500	1 750	1 500	1 500	1 750	790	800	0006	790	800	006
B KZN216 Hibiscus Coast		1 450	1 500	1 750	1 450	1 500	1 750	790	800	900	790	008	006
otal: Ugu Munici	1	10000	10 250	11 750	10 100	10 250	11 750	5 740	5 800	6 400	5 740	2 800	6 400
		0	000				1	ī	000	0	i i	0	000
KZN221		1 500	1 500	1 750	1 500	1 500	1 750	790	008	006	06/	008	006
KZN222		1 450	1 500	1 /50	1 450	1 500	1 /50	750	800	006	790	008	006
B KZN223 Mpotana		500	1 500	1 750	1 500	1 500	1 750	06/	008	006	06/	008	006
KZN225		1 441	1 500	1 750	141	1 500	1 750	790	800	006	790	800	006
		1 500	1 500	1 750	1 500	1 500	1 750	790	800	006	790	800	006
KZN227		1 500	1 500	1 750	1 500	1 500	1 750	790	800	006	790	800	006
C DC22 uMgungundlovu District Municipality		1 250	1 250	1 250	1 250	1 250	1 250	790	1 000	1 000	790	1 000	1 000
Total: uMgungundlovu Municipalities		11 641	11 750	13 500	11 641	11 750	13 500	6 320	009 9	7 300	6 3 2 0	0099	7 300
B KZN232 Emnambethi-Ladysmith		1 450	1 500	1 750	1 450	1 500	1 750	790	800	006	790	800	006
		1 500	1 500	1 750	1 500	1 500	1 750	790	800	006	790	800	006
		1 450	1 500	1 750	1 450	1 500	1 750	790	800	006	790	800	006
		1 450	1 500	1 750	1 450	1 500	1 750	790	800	006	790	800	006
B KZN236 Imbabazane		1 500	1 500	1 750	1 500	1 500	1 750	790	008	006	790	008	006
DC23 otal:Uthukela Mu		8 600	8 750	10 000	8 600	8 750	10 000	4 740	5 000	5 500	4740	2 000	5 500
B KZN341 Endumeni		1 450	1 500	1 750	1.450	1 500	1 750	002	008	000	007	008	000
B KZN242 Nouthu		1 500	1 500	1 750	1 500	1 500	1 750	790	008	006	790	008	006
		1 450	1 500	1 750	1 450	1 500	1 750	790	800	006	790	800	006
B KZN245 Umvoti		1 450	1 500	1 750	1 450	1 500	1 750	790	800	006	790	800	006
C DC24 Umzinyathi District Municipality		1 250	1 250	1 250	1 250	1 250	1 250	790	1 000	1 000	790	1 000	1 000
Total: Umzinyathi Municipalities		7 100	7 250	8 250	7 100	7 250	8 250	3 950	4 200	4 600	3 950	4 200	4 600
B KZN252 Newcastle		1 450	1 500	1 750	1 450	1 500	1 750	790	800	850	790	800	850
		1 500	1 500	1 750	1 500	1 500	1 750	790	800	850	790	800	850
++		1 250	1 250	1 750	1 250	1 250	1 750	790	008	850	790	800	850
C DC25 Amajuba District Municipality		1 250	1 500	1 500	1 250	1 500	1 500	790	1 000	1 000	790	1 000	1 000
Total: Amajuba Municipalities		2 450	5 750	6 750	5 450	5 750	6 750	3 160	3 400	3 550	3 160	3 400	3 550

ANNEXURE W4
SPECIFIC PURPOSE RECURRENT GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULE 6) 1 OF 2

		Local Gov	ernment Finan	Local Government Financial Management Grant	nt Grant			Muni	Municipal Systems Improvement Grant	[mprovement C	rant	
	Natio	National Financial Year	ear	Munici	Municipal Financial Year	Year	Natio	National Financial Year	Year	Munic	Municipal Financial Year	Year
Category Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
B KZN261 eDumbe	1 450	1 500	1 750	1 450	1 500	1 750	790	1 000	006	790	1 000	0006
B KZN262 UPhongolo	1 450	1 500	1 750	1 450	1 500	1 750	1 000	1 000	006	1 000	1 000	006
B KZN263 Abaqulusi	1 450	1 500	1 750	1 450	1 500	1 750	1 000	1 000	006	1 000	1 000	006
KZN265	1 450	1 500	1 750	1 450	1 500	1 750	790	800	006	790	800	0006
B KZN266 Ulundi	1 450	1 500	1 750	1 450	1 500	1 750	1 000	1 000	006	1 000	1 000	0006
C DC26 Zululand District Municipality	1 250	1 250	1 250	1 250	1 250	1 250	1 000	1 000	1 000	1 000	1 000	1 000
Total: Zululand Municipalities	8 500	8 750	10 000	8 500	8 750	10 000	5 580	2 800	5 500	5 580	5 800	5 500
	,	,	1	,	,	,	i	0	6	Î	6	6
KZNZ/I	1 450	1 500	05/ 1	1 450	1 500	1 /50	06/	800	006	06/	008	006
	1 450	1 500	1 750	1 450	1 500	1 750	790	800	006	190	800	006
	1 450	1 500	1 750	1 450	1 500	1 750	790	800	006	790	800	0006
B KZN274 Hlabisa	1 450	1 500	1 750	1 450	1 500	1 750	790	800	006	790	800	006
KZN275	1 250	1 500	1 750	1 250	1 500	1 750	790	800	006	790	800	006
C DC27 Umkhanyakude District Municipality	1 250	1 250	1 250	1 250	1 250	1 250	790	1 000	1 000	790	1 000	1 000
Total: Umkhanyakude Municipalities	8 300	8 750	10 000	8 300	8 750	10 000	4 740	2 000	2 500	4 7 4 0	2 000	5 500
	4	1	1	4			1		4 4	1	4	4
KZNZ81	1 450	1 500	1 /50	1 450	1 500	1 /50	06/	008	006	06/	008	006
KZN282	1 450	1 500	1 750	1 450	1 500	1 750	790	800	006	790	800	006
KZN283	1 450	1 500	1 750	1 450	1 500	1 750	790	800	006	190	800	006
	1 450	1 500	1 750	1 450	1 500	1 750	790	800	006	190	800	006
B KZN285 Mthonjaneni	1 450	1 500	1 750	1 450	1 500	1 750	790	800	006	790	800	0006
B KZN286 Nkandla	1 500	1 500	1 750	1 500	1 500	1 750	790	800	006	790	800	006
C DC28 uThungulu District Municipality	1 250	1 250	1 250	1 250	1 250	1 250	790	1 000	1 000	790	1 000	1 000
Total: uThungulu Municipalities	10 000	10 250	11 750	10 000	10 250	11 750	5 530	2 800	6 400	5 530	2 800	6 400
B KZN291 Mandeni	1 450	1 500	1 750	1 450	1 500	1 750	790	1 000	850	790	1 000	850
	1 450	1 500	1 750	1 450	1 500	1 750	790	800	850	790	800	850
B KZN293 Ndwedwe	1 450	1 500	1 750	1 450	1 500	1 750	790	800	850	790	800	850
B KZN294 Maphumulo	1 450	1 500	1 750	1 450	1 500	1 750	790	800	850	790	800	820
C DC29 iLembe District Municipality	1 250	1 250	1 250	1 250	1 250	1 250	790	1 000	850	790	1 000	820
Total: iLembe Municipalities	7 050	7 250	8 250	7 050	7 250	8 250	3 950	4 400	4 250	3 950	4 400	4 2 5 0
B KZN431 Income	1 450	1 500	1 750	1 450	1 500	1 750	062	800	850	790	008	058
	1 500	1 500	1 750	1 500	1 500	1 750	790	008	850	062	008	850
KZN433	1 250	1 500	1 750	1 250	1 500	1 750	062	800	850	790	800	850
KZN434	1 250	1 500	1 750	1 250	1 500	1 750	062	008	850	790	008	850
B KZN435 Umzimkhulu	1 450	1 500	1 750	1 450	1 500	1 750	062	008	850	790	008	850
	1 250	1 250	1 250	1 250	1 250	1 250	790	1 000	006	790	1 000	006
Total: Sisonke Municipalities	8 150	8 750	10 000	8 150	8 750	10 000	4 740	2 000	5 150	4 740	2 000	5 150
Total: KwaZulu-Natal Municinalities	86 141	000 68	101 500	86 141	89 000	101 500	48 450	51 000	54 150	48 450	51 000	54 150

ANNEXURE W4
SPECIFIC PURPOSE RECURRENT GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULE 6) 1 OF 2

			Local Gov	Local Government Financial Management Grant	cial Managemo	ent Grant			Mun	icipal Systems	Municipal Systems Improvement Grant	rant	
		Nation	National Financial Year	l'ear	Munic	Municipal Financial Year	Year	Natio	National Financial Year	Year	ojun <sub>W</sub>	Municipal Financial Year	ear
Category	Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
LIMPOPO													
LIM331	Greater Giyani	1 250	1 500	1 500	1 250	1 500	1 500	790	800	006		800	006
	Greater Letaba	1 250	1 500	1 500	1 250	1 500	1 500	790	800	006	790	800	006
LIM333	Greater Tzaneen	1 250	1 500	1 500	1 250	1 500	1 500	790	800	006		800	006
LIM334	Ba-Phalaborwa	1 250	1 500	1 500	1 250	1 500	1 500	790	800	006		800	006
LIM335	Maruleng	1 500	1 500	1 500	1 500	1 500	1 500	790	800	006	790	800	006
C DC33 Mo	Mopani District Municipality	1 250	1 250	1 500	1 250	1 250	1 500	790	1 000	1 000	790	1 000	1 000
Total: Mopani Municipalities	palities	7 750	8 750	0006	7 750	8 750	0006	4 740	2 000	5 500	4 7 4 0	2 000	5 500
LIM341	Musina	1 250	1 250	1 500	1 250	1 250	1 500	790	800	006	062	008	006
LIM342	Mutale	1 250	1 500	1 500	1 250	1 500	1 500	062	800	006	062	800	006
LIM343	Thulamela	1 250	1 500	1 500	1 250	1 500	1 500	790	800	006	790	800	006
B LIM344 Makhado	khado	1 250	1 500	1 500	1 250	1 500	1 500	790	800	006	790	800	006
DC34	Vhembe District Municipality	1 250	1 250	1 250	1 250	1 250	1 250	790	1 000	1 000	790	1 000	1 000
Total: Vhembe Municipalities	palities	6 250	7 000	7 250	6 250	7 000	7 250	3 950	4 200	4 600	3 950	4 200	4 600
LIM351	Blouberg	1 500	1 500	1 500	1 500	1 500	1 500	790	800	006		800	006
	Aganang	1 250	1 500	1 500	1 250	1 500	1 500	790	800	006	790	800	006
B LIM353 Molemole	lemole	1 250	1 500	1 500	1 250	1 500	1 500	790	800	006	790	800	006
	okwane	1 250	1 500	1 500	1 250	1 500	1 500	790	800	006	790	800	006
B LIM355 Lepelle-Nkumpi	oelle-Nkumpi	1 250	1 500	1 500	1 250	1 500	1 500	790	800	006	790	800	006
C DC35 Cap	Capricorn District Municipality	1 250	1 250	1 250	1 250	1 250	1 250	790	1 000	1 000	190	1 000	1 000
Total: Capricorn Municipalities	icipalities	7 750	8 750	8 750	7 750	8 750	8 750	4 740	5 000	5 500	4740	2 000	5 500
R I IM361 Thahazimhi	hazimbi	1 500	1 500	1 500	1 500	1 500	1 500	062	008	006		008	006
B LIM362 Lep	Lephalale	1 250	1 500	1 500	1 250	1 500	1 500	062	800	006	062	800	006
LIM364	okgopong	1 250	1 500	1 500	1 250	1 500	1 500	790	800	006		800	006
	Modimolle	1 250	1 500	1 500	1 250	1 500	1 500	790	800	006		800	006
B LIM366 Bela Bela	a Bela	1 250	1 500	1 500	1 250	1 500	1 500	790	800	006		800	006
LIM367	Mogalakwena	1 250	1 500	1 500	1 250	1 500	1 500	790	800	006	790	800	006
C DC36 Wat	Waterberg District Municipality	1 250	1 250	1 250	1 250	1 250	1 250	790	1 000	1 000	190	1 000	1 000
Total: Waterberg Municipalities	nicipalities	000 6	10 250	10 250	000 6	10 250	10 250	5 530	2 800	6 400	5 530	2 800	6 400
LIM471	Ephraim Mogale	1 250	1 500	1 500	1 250	1 500	1 500	790	800	006		800	006
LIM472	Elias Motsoaledi	1 250	1 500	1 500	1 250	1 500	1 500	790	800	006		800	006
LIM473	Makhuduthamaga	1 500	1 500	1 500	1 500	1 500	1 500	790	800	006		800	006
LIM474	Fetakgomo	1 500	1 500	1 500	1 500	1 500	1 500	790	800	006		800	006
	Greater Tubatse	1 250	1 500	1 500	1 250	1 500	1 500	790	800	006	790	800	006
C DC47 Gre	Greater Sekhukhune District Municipality	1 250	1 250	1 250	1 250	1 250	1 250	790	800	1 000	790	800	1 000
Total: Greater Sekhukhune Municipalities	thune Municipalities	8 000	8 750	8 750	8 000	8 750	8 750	4 740	4 800	5 500	4 7 4 0	4 800	5 500
Total: Limpopo Municipalities	cipalities	38 750	43 500	44 000	38 750	43 500	44 000	23 700	24 800	27 500	23 700	24 800	27 500

ANNEXURE W4
SPECIFIC PURPOSE RECURRENT GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULE 6) 1 OF 2

		Local Gov	Local Government Financial Management Grant	cial Managem	ent Grant			Mun	Municipal Systems Improvement Grant	Improvement (	Frant	
	Nati	National Financial Year	Year	Munic	Municipal Financial Year	Year	Nati	National Financial Year	Year	Muni	Municipal Financial Year	Year
Category Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
MPUMALANGA												
B MP301 Albert Luthuli	1 250	1 250	1 500	1 250	1 250	1 500	790	800	850	790	800	850
B MP302 Msukaligwa	1 250	1 500	1 500	1 250	1 500	1 500	790	800	850	790	800	850
B MP303 Mkhondo	1 250	_	1 500	1 250	1 500	1 500	790	800	850	790	800	850
B MP304 Pixley Ka Seme	1 250	_	1 500	1 250	1 500	1 500	790	800	850	790	800	850
B MP305 Lekwa	1 250	1 250	1 500	1 250	1 250	1 500	790	800	850	790	800	850
B MP306 Dipaleseng	1 250	_	1 500	1 250	1 500	1 500	790	800	850	790	800	850
B MP307 Govan Mbeki	1 250	1 500	1 500	1 250	1 500	1 500	790	800	850	790	800	850
C DC30 Gert Sibande District Municipality	1 250	1 250	1 250	1 250	1 250	1 250	1 000	1 000	1 000	1 000	1 000	1 000
Total: Gert Sibande Municipalities	10 000	11 250	11 750	10 000	11 250	11 750	6 530	0099	0569	6 530	009 9	056 9
B MP311 Victor Khanye	1 250	1 500	1 500	1 250	1 500	1 500	1 200	1 200	006	1 200	1 200	006
B MP312 Emalahleni	1 250	1 500	1 500	1 250	1 500	1 500	790	800	850	790	800	850
B MP313 Steve Tshwete	1 250	1	1 500	1 250	1 500	1 500	790	1 000	850	790	1 000	850
B MP314 Emakhazeni	1 250	1 500	1 500	1 250	1 500	1 500	790	800	850	790	800	850
B MP315 Thembisile	1 250	1 250	1 500	1 250	1 250	1 500	790	800	850	790	800	850
B MP316 Dr JS Moroka	1 250	1 250	1 500	1 250	1 250	1 500	790	800	850	790	800	850
C DC31 Nkangala District Municipality	1 250	1 250	1 500	1 250	1 250	1 500	1 000	1 000	1 000	1 000	1 000	1 000
Total: Nkangala Municipalities	8 750	9 750	10 500	8 750	9 750	10 500	6 150	6 400	6 150	6 150	6 400	6 150
B MP321 Thaba Chweu	1 250	1 500	1 500	1 250	1 500	1 500	790	800	006	790	800	006
B MP322 Mbombela	1 250	1	1 500	1 250	1 500	1 500	790	800	006	790	800	006
B MP323 Umjindi	1 250	1 500	1 500	1 250	1 500	1 500	790	800	006	790	800	006
B MP324 Nkomazi	1 250	1 500	1 500	1 250	1 500	1 500	790	800	006	790	800	006
B MP325 Bushbuckridge	1 250	1 500	1 500	1 250	1 500	1 500	790	800	006	790	800	006
C DC32 Ehlanzeni District Municipality	1 250	1 500	1 500	1 250	1 500	1 500	1 000	1 000	1 000	1 000	1 000	1 000
Total: Ehlanzeni Municipalities	7 500	000 6	000 6	7 500	000 6	0006	4 950	2 000	5 500	4 950	2 000	5 500
Total: Mpumalanga Municipalities	26 250	30 000	31 250	26 250	30 000	31 250	17 630	18 000	18 600	17 630	18 000	18 600

ANNEXURE W4
SPECIFIC PURPOSE RECURRENT GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULE 6) 1 OF 2

1,500   1,750   1,450   1,500   1,75				Local Gov	Local Government Financial Management Grant	cial Manageme	ant Grant			Mun	icipal Systems	Municipal Systems Improvement Grant	rant	
1, 10,   1			Natio	nal Financial	Year	Munic	ipal Financial	Year	Nati	onal Financial	Year	Munic	ipal Financial	Year
1450   1500   1750   1450   1500   1750	Category	Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
pullity    1	NORTHERN CAPE													
1450   1500   1750   1450   1500   1750	NC061	ichtersveld	1 450	1 500	1 750	1 450	1 500	1 750	790			062	800	006
1450   1500   1750   1450   1500   1750	NC062	ama Khoi	1 450	1 500	1 750	1 450	1 500	1 750	790			790	800	006
1450   1500   1750   1500   1750   1500   1750	NC064	amiesberg	1 450	1 500	1 750	1 450	1 500	1 750	790			790	800	006
1250   1260   1750   1260   1750	NC065	antam	1 450	1 500	1 750	1 450	1 500	1 750	790			790	800	006
1250   1250	NC066	aroo Hoogland	1 250	1 500	1 750	1 250	1 500	1 750	790			790	800	006
1450   1250   1150   1250   1150   1250   1150	NC067	hâi-Ma	1 250	1 500	1750	1 250	1 500	1750	790	•	•	790	008	006
Auticipality 1 450 1 500 1 750 1 450 1 750	Total: Namakwa Min	amakwa District Municipality nicipalities	052 1	10 250	11 750	052 1	10 250	11 750	5 530			08/	2 800	6 400
1450   1500   1770   1450   1500   1770   1450   1500   1770   1750	Total: Mailian wa Min	merpanies	000	007 01	06/11	Occ /	00701	00/11	OCC C	000 0	0	OCC C	2000	0
1450   1500   1750   1450   1500   1750   1450   1500   1750   1450   1500   1750   1450   1500   1750   1450   1500   1750	NC071	buntu	1 450	1 500	1 750	1 450	1 500	1 750	790	800		790	800	006
1450   1500   1750   1450   1500   1750   1450   1500   1750   1450   1500   1750   1450   1500   1750   1450   1500   1750	NC072	msobomvu	1 450	1 500	1 750	1 450	1 500	1 750	790	800		790	800	006
1450   1500	NC073	mthanjeni	1 450	1 500	1 750	1 450	1 500	1 750	790			790	800	006
1450   1500   1750   1450   1500   1750   1450   1500   1750   1450   1500   1750   1450   1500   1750   1450   1500   1750   1750   1450   1500   1750	NC074	areeberg	1 450	1 500	1 750	1 450	1 500	1 750	1 200	1		1 200	1 200	006
1 450   1 500   1 750   1 500   1 750   1 500   1 75	NC075	enosterberg	1 450	1 500	1 750	1 450	1 500	1 750	790			790	800	006
1450   1500   1750   1450   1500   1750   1450   1500   1750	NC076	hembelihle	1 500	1 500	1 750	1 500	1 500	1 750	790			190	800	006
Auticipality         1 450         1 500         1 750         1 450         1 500         1 750         1 450         1 500         1 750         1 500         1 750	NC077	iyathemba	1 450	1 500	1 750	1 450	1 500	1 750	790	800		790	800	006
Authorizability         1250	NC078	iyancuma	1 450	1 500	1 750	1 450	1 500	1 750	790	800		790	800	006
15.900   13.250   15.260   15.260   15.260   15.250   79.30   8.000   8.200   79.30   14.50   14.50   15.20	C DC7 P.	ixley Ka Seme District Municipality	1 250	1 250	1 250	1 250	1 250	1 250	1 200	1 200	1 000	1 200	1 200	1 000
1450   1500   1750   1450   1500   1750   1450   1500   1750	Total: Pixley Ka Sem	ne Municipalities	12 900	13 250	15 250	12 900	13 250	15 250	7 930	8 000	8 200	7 930	8 000	8 200
1450   1500   1750   1450   1500   1750   1500   1750   1500   1750   1500   1750   1500   1750   1500   1750   1500   1750   1500   1750   1500   1750   1500   1750   1500   1750   1750   1500   1750	100018		1 450	1 500	037.1	1 450	1,500	750	005	000	000	100	000	900
1450   1500   1750   1450   1500   1750   1450   1500   1750	NC081		1 450	1 500	1 750	1 450	1 500	1 750	260	900		06/	900	906
Hity 1450 1500 1750 1450 1500 1750 7790 800 7790 7790 1790 7790 1790 7790 1790 7790 1790 1	NC082	Kan Gand	1 450	1 500	1 750	1 450	1 500	1 750	1 200	_		1 200	1 200	006
lity   1450   1500   1750   1450   1500   1750   17	NC084	Khala mas (heis	1 450	1 500	1 750	1 450	1 500	1 750	790			067	008	006
lity   1500   1500   1750   1500   1500   1750   1500   1750   17	NC085	santsahane	1 450	1 500	1 750	1 450	1 500	1 750	790			062	800	006
lity         1 250         2 250	NC086	gatelopele	1 500	1 500	1 750	1 500	1 500	1 750	062			062	800	006
10 000   10 250   11 750   14 50   10 250   17 750   14 50   17 750   17	DC8	ivanda District Municipality	1 250	1 250	1 250	1 250	1 250	1 250	790	-	_	790	1 000	1 000
Inicipality 1450 1500 1750 1450 1500 1750 790 800 900 790 1790 1450 1500 1750 1750 1750 790 800 900 790 1790 1500 1500 1750 1500 1750 1500 1750 175	Total: Siyanda Muni	cipalities	10 000	10 250	11 750	10 000	10 250	11 750	5 940	9		5 940	6 2 0 0	6 400
1450   1500   1750   1450   1500   1750   1750   1750   7790   800   790   7														
1450   1500   1750   1450   1500   1750   1450   1500   1750	NC091	ol Plaatje	1 450	1 500	1 750	1 450	1 500	1 750	790	800		790	800	006
1500   1500   1750   1500   1750   1500   1750   1750   790   800   79	NC092	ikgatlong	1 450	1 500	1 750	1 450	1 500	1 750	790			190	800	006
Linicipality 1500 1500 1500 1500 1500 1500 1700 1700	NC093	lagareng	1 500	1 500	1 750	1 500	1 500	1 750	790				800	006
1,200	NC094	norwane concess Doord District Municipality	1 250	1 250	1 250	1 350	1 250	1 250	790	-	-	-	200	900
trict Municipality 1 560	Total: Frances Baard	1 Municipalities	7 150	7 250	8 250	7 150	7 250	8 250	4 360	4 400			4 400	4 600
1430   1500   1750   1430   1500   1750   1430   1500   1750   1430   1500   1750   1430   1500   1750   1430   1500   1750	NOAEI	-	037	002	040	1450	1	0 10	000	000			000	000
trict Municipality 1 560	NC451	Iosnaweng	1 450	1 500	1 /30	1 450	1 500	1 /30	06/	008			900	906
trict Municipality         1 500         1 700         1 700         1 700         1 700         790           trict Municipality         1 250         1 250         1 250         1 250         1 250         1 250         1 000         1 000         1 000         790           trict Municipality         5 600         5 750         6 500         5 750         6 500         3 160         3 400         3 700         790           trict Municipality         5 600         5 750         6 500         5 750         6 500         3 400         3 700         700           trict Municipality         45 700         45 700         45 700         45 700         3 400         3 700         7 800	NC452	a-Segonyana	1 450	1 500	1 /50	1 450	1 500	1 /50	790				800	006
1.20	NC453	amagara	1 450	1 500	1 750	1 450	1 500	1 750	96/	-	-		800	006
5 600 5 750 6 500 5 750 6 500 3 160 3 400 3 700 3 160 3 400 3 700 3 160 3 400 3 700 3 160 3 400 3 700 3 160 3 400 3 700 3 160 3 400 3 700 3 160 3 400 3 160 3 400 3 40 3 40 3 40 3 40 3 40 3 40 3	C DC45 JC	ohn Taolo Gaetsewe District Municipality	1.250	1 250	1 250	1 250	1 250	1 250	06/			06/	1 000	1 000
45 700 46 750 51 51 500 46 750 51 500 76 070 77 800 26 070 76 070	Total: John Taolo Ga	aetsewe Municipalities	2 600	5 750	9 200	2 600	5 750	9 200	3 160	3 400	3 700	3 160	3 400	3 7 0 0
45 700 45 700 45 750 53 500 53 500 53 500 75 78 00 75 78 78 00 70 75 78 78 00 78 75 78 78 00 78 00 78 00 78 00														
	Total: Northern Can	e Municipalities	45 200	46 750	53 500	45 200	46 750	53 500	26 920	27 800	29 300	26 920	27 800	29 300

ANNEXURE W4
SPECIFIC PURPOSE RECURRENT GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULE 6) 1 OF 2

		Local Gov	Local Government Financial Management Grant	cial Managem	ent Grant			Mun	icipal Systems	Municipal Systems Improvement Grant	rant	
	Natic	National Financial Year	Year	Munic	Municipal Financial Year	Year	Nati	National Financial Year	Year	Munic	Municipal Financial Year	Year
Category Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
NORTH WEST												
B NW371 Moretele	1 250	1 500	1 500	1 250	1 500	1 500	1 000	1 000	006	1 000	1 000	006
B NW372 Madibeng	1 500	1 500	1 500	1 500	1 500	1 500	790	1 000	006	790	1 000	006
B NW373 Rustenburg	1 250	1 500	1 500	1 250	1 500	1 500	790	1 000	006	790	1 000	006
B NW374 Kgetlengrivier	1 250	1 500	1 500	1 250	1 500	1 500	790	1 000	006	790	1 000	006
B NW375 Moses Kotane	1 250	1 500	1 500	1 250	1 500	1 500	1 000	1 000	006	1 000	1 000	006
C DC37 Bojanala Platinum District Municipality	1 250	1 250	1 250	1 250	1 250	1 250	790	1 000	1 000	790	1 000	1 000
Total: Bojanala Platinum Municipalities	7 750	8 750	8 750	7 750	8 750	8 750	5 160	0009	5 500	5 160	000 9	5 500
B NW381 Ration	1 500	1 500	1,500	1.500	1 500	1.500	190	008	006	190	008	006
	1 250	1 500	1 500	1 250	1 500	1 500	790	800	006	790	800	006
B NW383 Mafikeng	2 000	1 500	1 500	2 000	1 500	1 500	830	1 000	1 000	830	1 000	1 000
B NW384 Ditsobotla	1 250	1 500	1 500	1 250	1 500	1 500	1 200	1 500	1 000	1 200	1 500	1 000
B NW385 Ramotshere Moiloa	1 500	1 500	2 086	1 500	1 500	2 086	1 000	800	006	1 000	800	006
C DC38 Ngaka Modiri Molema District Municipality	1 250	1 250	1 250	1 250	1 250	1 250	1 000	1 000	1 000	1 000	1 000	1 000
Total: Ngaka Modiri Molema Municipalities	8 750	8 750	9 336	8 750	8 750	9 336	5 610	5 900	5 700	5 610	2 900	5 700
B NW392 Naledi	1 500	1 500	1 500	1 500	1 500	1 500	790	800	006	790	800	006
NW393	1 250	1 500	1 500	1 250	1 500	1 500	790	800		790	800	850
B NW394 Greater Taung	1 250	1 500	1 500	1 250	1 500	1 500	790	800		790	800	850
B NW396 Lekwa-Teemane	1 500	1 500	1 500	1 500	1 500	1 500	790	800	850	790	800	850
	2 000	3 000	3 000	2 000	3 000	3 000						
C DC39 Dr Ruth Segomotsi Mompati District Municipality	1 250	1 250	1 250	1 250	1 250	1 250	790	1 000		190	1 000	1 000
Total: Dr Ruth Segomotsi Mompati Municipalities	8 750	10 250	10 250	8 750	10 250	10 250	3 950	4 200	4 450	3 9 5 0	4 200	4 4 5 0
B NW401 Ventersdorp	1 250	1 500	1 500	1 250	1 500	1 500	790	800	870	790	800	870
B NW402 Tlokwe	1 250	1 500	1 500	1 250	1 500	1 500	790	800	006	790	800	006
B NW403 City of Matlosana	1 250	1 500	1 500	1 250	1 500	1 500	790	800	006	790	800	006
B NW404 Maquassi Hills	1 250	1 500	1 500	1 250	1 500	1 500	790	800	006	790	800	006
C DC40 Dr Kenneth Kaunda District Municipality	1 250	1 250	1 250	1 250	1 250	1 250	790	1 000	1 000	790	1 000	1 000
Total: Dr Kenneth Kaunda Municipalities	6 250	7 250	7 250	6 250	7 250	7 250	3 950	4 200	4 570	3 950	4 200	4 570
Total: North West Municipalities	31 500	35 000	785 28	31 500	35,000	35 586	18 670	20 300	20 220	18 670	20 300	20.220

ANNEXURE W4
SPECIFIC PURPOSE RECURRENT GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULE 6) 1 OF 2

			Local Gov	Local Government Financial Management Grant	cial Manageme	nt Grant			Muni	cipal Systems	Municipal Systems Improvement Grant	rant	
			National Financial Year	Year	Munic	Municipal Financial Year	Year	Natio	National Financial Year	Year	Munici	Municipal Financial Year	/ear
Category	Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
WESTERN CAPE													
A CPT City of Cape Town		1 250	1 250	1 250	1 250	1 250	1 250						
B WC011 Matzikama		1 250	1 250	1 250	1 250	1 250	1 250	790	800	006	790	800	006
B WC012 Cederberg		1 250	1 250	1 250	1 250	1 250	1 250	790	800	006	790	800	0006
WC013		1 250	1 250	1 250	1 250	1 250	1 250	790	800	006	790	800	0006
		1 250	1 250	1 250	1 250	1 250	1 250	790	800	006	790	800	006
2	:	1 250	1 250	1 250	1 250	1 250	1 250	790	800	006	790	008	006
Total: West Coast District Municipality  Total: West Coast Municipalities	Municipality	7 500	7 500	7 500	7 500	7 500	7 500	4 740	800	5500	4 740	800	2 500
4													
		1 250	1 250	1 250	1 250	1 250	1 250	1 200	1 000	1 000	1 200	1 000	1 000
WC023		1 250	1 250	1 250	1 250	1 250	1 250	790	800	006	790	800	006
WC024		1 250	1 250	1 250	1 250	1 250	1 250	790	800	006	790	800	0006
		1 250	1 250	1 250	1 250	1 250	1 250	790	800	006	790	800	006
B WC026 Langeberg C DC2 Cane Winelands District Municipality	rict Municipality	1 250	1 250	1 250	1 250	1 250	1 250	790	800	006	790	008	006
otol. Cone Winel	aret maner party	002 1	002 7	002 1	002 1	0027	0027	5 150	2000	000 1	2150	2000	009 1
10tal: Cape Wiletains Municipalities  R WC031 Theewsterchoof		000 /	005 1	200	0000 /	2005 /	2006 /	1 200	0001	2 000	0001	007 5	2 000
WC032		1 250	1 250	1 250	1 250	1 250	1 250	007	008	000	007	008	000
		1 250	1 250	1 250	1 250	1 250	1 250	790	008	006	790	008	006
WC034		1 250	1 250	1 250	1 250	1 250	1 250	790	856	006	790	856	006
DC3	unicipality	1 250	1 250	1 250	1 250	1 250	1 250	790	1 000	1 000	790	1 000	1 000
Total: Overberg Municipalities		6 250	9 200	9 200	6 250	9 200	9 200	4 360	4 456	4 700	4360	4 4 5 6	4 700
WC041		1 250	1 250	1 250	1 250	1 250	1 250	1 720	1 500	1 014	1 720	1 500	1 014
WC042		1 250	1 250	1 250	1 250	1 250	1 250	790	800	006	790	800	006
WC043		1 250	1 250	1 250	1 250	1 250	1 250	790	800	0006	790	800	0006
B wC044 George		1 250	1 250	1 250	1 250	1 250	1 250	790	008	006	790	008	006
		1 250	1 250	1 250	1 250	1 250	1 250	06/	008	006	790	00%	006
WC048		1 250	1 250	1 250	1 250	1 250	1 250	06/	008	006	790	008	006
DC4	pality	1 250	1 250	1 250	1 250	1 250	1 250	790	1 000	1 000	790	1 000	1 000
Total: Eden Municipalities		10 000	10 000	10 000	10 000	10 000	10 000	7 250	7 300	7 414	7 250	7 300	7 414
B WC051 Laingsburg		1 500	1 500	1 500	1 500	1 500	1 500	790	800	006	790	800	006
WC052		1 250	1 250	1 250	1 250	1 250	1 250	790	800	006	790	800	006
m	:	1 250	1 250	1 250	1 250	1 250	1 250	790	800	006	790	800	006
C DC5 Central Karoo District Municipality	ct Municipality	1 250	1 250	1 250	1 250	1 250	1 250	06/	1 000	1 000	06/	1 000	1 000
Total: Central Karoo Municipalities		5 250	5 250	5 250	5 250	5 250	5 250	3 160	3 400	3 700	3 160	3400	3 700
		on the	000 00	000 00	i i	000.00	000 00	977.70	74.4.46	7 20 26	077.74	74.40	1,000
Total: Western Cape Municipalities		37.750	38 000	38 000	37.750	38 000	38 000	24 660	25 IS6	26 914	74 000	921 22	26 914
Unallocated:		20 000	75 000	100 000									
National Total		434 641	478 873	526 086	384 641	403 873	426 086	219 420	230 096	242 734	219 420	230 096	242 734
* ************************************			2	2					; ;	!!!!		) )	1

ANNEXURE W4
SPECIFIC PURPOSE RECURRENT GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULE 6) 2 OF 2

Category  ASTERN CAPE  BUF  NMA  EC101  EC102  EC103  EC104  EC104	Municipality  Municipality  Buffalo City Nelson Mandela Bay  Camdeboo Blue Crane Route Ikwezi Makana Ndharba Sundays River Valley Baviaans Kouga	Nationa   201/12   (R000)     1 680	National Financial Year 7/2 2012/13 20, 00) (R'000) (R') 1 680 3 076	(R'000) 5 000	Inancial Year   Municipal Fir   1273   2013/14   2011/12   2012   1000   (R'000)   (R'000)   (R'000)	Municipal Financial Year ///12 2012/13 201. ////////////////////////////////////	Year 2013/14 (R'000)	Nati 2011/12	National Financial Year           12         2012/13         20           13         (R'000)         (R	2013/14 (R'000)	2011/12 (R'000)	Municipal Financial Year  12 2012/13 20  (R'000) (R	Year 2013/14 (R'000)
ASTERN CAPE BUF NMA EC101 EC102 EC103 EC104 EC104 EC105	ipality  o City  Mandela Bay  boo  rane Route  a  be be be so River Valley  ms  annma	. 08	3 076	2013/14 (R'000) 5 000	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13	2013/14 (R'000)
ASTERN CAPE BUF NMA EC101 EC102 EC103 EC103 EC104 EC104	o City Mandela Bay boo rane Route a be bos selections with a selection of the selection of	089	3 076	2 000				(K 000)		( )		(00001)	
BUF NMA EC101 EC102 EC103 EC104 EC104	o City Mandela Bay bbo rane Route a be bo sys River Valley annna	089	3 076	2 000									
NMA EC101 EC102 EC103 EC104 HC105	n Mandela Bay boo rane Route a be bo sys River Valley annna				1 680	3 076	5 000	3 130		9 6 500	3 130	4 576	6 500
EC101 EC102 EC103 EC104	boo Trane Route a be ys River Valley nns							1 250	1 250		1 250	1 250	1 250
EC102 EC103 EC104 EC105	rane Route a a be sk River Valley annma							062.6	2 800	2 850	2 790	2 800	2 850
EC103 EC104	a be ys River Valley ns amma							2 2 4 0	2 1	2 1	2 240	2 300	2 350
EC104	a be ys River Valley ns amma							2 2 4 0			2 240	2 300	2 350
EC105	be ys River Valley ns amma							2 2 4 0			2 240	2 300	2 350
COLOS	ys River Valley us amma							2 790			2 790	2 800	2 850
	ns amma							2 2 9 0			2 290	2 300	2 350
EC107	amma							2 2 9 0			2 290	2 050	2 350
B EC108 Kouga	amma							2 2 4 0	2 300	2 350	2 240	2 300	2 350
B EC109 Kou-kamma								2 040	2 300		2 040	2 300	2 350
C DC10 Cacadu	Cacadu District Municipality							2 040	2 500	2 500	2 040	2 500	2 500
Total: Cacadu Municipalities	ities							23 200	23 950	24 650	23 200	23 950	24 650
ŗ								0000			0000	000	6
EC121	ne e							2 2 2 0			2 2 2 0	2 300	2 400
EC122	na							2.240			2 240	2 300	2 400
EC123	Kei							2 290			2 290	2 300	2 400
EC124	athi							2 290			2 290	2 300	2 400
EC126	hwa							2 2 4 0			2 240	2 300	2 400
	ope							2 2 4 0			2 240	2 300	2 400
								2 2 4 0			2 240	2 300	2 400
C DC12 Amathol	Amathole District Municipality	9 333	11 842	2 000	9 333	11 842	5 000	11 373			11 373	14 2 15	7 400
Total: Amathole Municipalities	alities	9 333	11 842	2 000	9 333	11 842	5 000	27 203	30 315	24 200	27 203	30 315	24 200
B EC131 Inxuba N	Inxuba Yethemba							2 290	2 300	2 350	2 290	2 300	2 350
B EC132 Tsolwana	ana							2 040	2 300	2 350	2 040	2 300	2 350
B EC133 Inkwanca	nca							2 790	2 300		2 790	2 300	2 350
B EC134 Lukhanji	iţiu							2 2 9 0	2 300		2 290	2 300	2 350
B EC135 Intsika Yethu	Yethu							2 2 9 0			2 290	2 300	2 350
	hleni							2 290			2 290	2 300	2 350
EC137	po							2 2 4 0			2 240	2 300	2 350
EC138	izwe							2 290			2 290	2 300	2 350
C DC13 Chris Ha	Chris Hani District Municipality	7 512	9 242	5 000	7 512	9 242	5 000	9 552			9 552	11 742	7 350
Total: Chris Hani Municipalities	ipalities	7 512	9 242	2 000	7 512	9 242	5 000	28 072	30 142	26 150	28 072	30 142	26 150

ANNEXURE W4
SPECIFIC PURPOSE RECURRENT GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULE 6) 2 OF 2

			Water S	ervices Opera	Water Services Operating Subsidy Grant	Grant				SUB-TOTAL: RECURRENT	RECURREN	L	
		Nation	National Financial Year	Year	Municip	Municipal Financial Year	Year	Nati	National Financial Year	Year	junjyj	Municipal Financial Year	Year
Conference	Memipiesalite	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	21/1102	2012/13	2013/14
Category	Municipality	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)
B EC141	Elundini							2 290	2 300	2 400	2 290	2 300	2 400
B EC142	Sengu							2 2 4 0	2 300	2 400	2 240	2 300	2 400
B EC143	Maletswai							2 2 9 0	2 300	2 400	2 290	2 300	2 400
B EC144	Gariep							2 2 4 0	2 300	2 400	2 240	2 300	2 400
C DC14								2 040	2 050	2 400	2 040	2 050	2 400
Total: Ukhahlami	Fotal: Ukhahlamba Municipalities							11 100	11 250	12 000	11 100	11 250	12 000
B EC153	Ngquza Hill							2 2 4 0	2 300	2 350	2 240	2 300	2 350
B EC154	Port St Johns							2 790	2 300	2 350	2 790	2 300	2 350
B EC155	Nyandeni							2 2 4 0	2 300	2 350	2 240	2 300	2 350
B EC156	Mhlontlo							2 2 4 0	2 300	2 350	2 240	2 300	2 350
B EC157	King Sabata Dalindyebo							2 2 4 0	2 300	2 350	2 240	2 300	2 350
C DC15	O R Tambo District Municipality	15 569	17 268	2 000	15 569	17 268	2 000	17 609	19 568	7 350	17 609	19 568	7 350
Total: O.R. Tambo Municipalities	o Municipalities	15 569	17 268	2 000	15 569	17 268	2 000	29 359	31 068	19 100	29 359	31 068	19 100
B EC441								2 240				2 300	2 350
B EC442	Umzimvubu							2 240	2 300	2 350	2 240	2 300	2 350
B EC443	Mbizana							2 240	2 300	2 350	2 240	2 300	2 350
B EC152	Ntabankulu							2 290	2 300	2 350	2 290	2 300	2 350
C DC44	Alfred Nzo District Municipality	0909	6906	2 000	090 9	6906	2 000	8 100	11 369	7 350	8 100	11 369	7 350
Total: Alfred Nzo Municipalities	Municipalities	090 9	6906	2 000	090 9	6906	2 000	17 110	20 569	16 750	17 110	20 569	16 750
Total: Eastern Ca	Fotal: Eastern Cape Municipalities	40 154	50 497	25 000	40 154	50 497	25 000	140 424	153 120	130 600	140 424	153 120	130 600

ANNEXURE W4
SPECIFIC PURPOSE RECURRENT GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULE 6) 2 OF 2

			Water	Water Services Operating Subsidy Grant	ating Subside	Grant				SIR-TOTAL	STIB-TOTAL: BECTIREENT	F	
		Natic	National Financial Year	Year	Munic	Municipal Financial Year	l Year	Nati	National Financial Year	Year	Muni	Municipal Financial Year	Year
Category	y Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
FREE STATE	Э.												
A MA	MAN Mangaung							1 450	1 500	1 500	1 450	1 500	1 500
B FS1	FS161 Letsemeng							2 240	2 300	2 350	2 240	2 300	2 350
								2 2 4 0	2 300	2 350		2 300	2 350
B FS1	FS163 Mohokare							2 2 9 0	2 300	2 350		2 300	2 350
B FS1								2 2 40	2 300	2 350		2 300	2 350
Lotel View	DC 16 Anariep District Municipality							2 040	2 050	2 100		2 050	2 100
I otal: Xhari	Lotal: Xhariep Municipalities							050 11	067 11	11 500	00011	067 11	000 11
B FS1	FS181 Masilonvana							2 040	2 300	2 350	2 040	2 300	2 350
B FS1								2 2 4 0	2 300	2 350		2 300	2 350
B FS1	FS183 Tswelopele							2 2 9 0	2 340	2 350		2 3 4 0	2 350
B FS1	FS184 Matjhabeng							2 2 4 0	2 300	2 350	2 240	2 300	2 350
B FS1	FS185 Nala							2 2 9 0	2 300	2 350		2 300	2 350
C DC	DC18 Lejweleputswa District Municipality							2 040	2 250	2 100	2 040	2 2 5 0	2 100
Total: Lejwe	Fotal: Lejweleputswa Municipalities							13 140	13 790	13 850	13 140	13 790	13 850
B FS1	FS191 Setsoto							2 240	2 300	2 350	2 240	2 300	2 350
B FS1	FS192 Dihlabeng							2 240	2 300	2 350	2 240	2 300	2 350
B FS1	FS193 Nketoana							2 240	2 300	2 350	2 240	2 300	2 350
		7 661		7 000	7 661		7 000	9 901	2 300	9 350		2 300	9 350
B FS1	FS195 Phumelela							2 090	2 300	2 350		2 300	2 350
B FS1								2 240	2 300	2 350		2 300	2 350
C DC	DC19 Thabo Mofutsanyana District Municipality							2 040	2 250	2 100		2 250	2 100
Total: Thabo	Fotal: Thabo Mofutsanyana Municipalities	7 661		7 000	7 661		7 000	22 991	16 050	23 200	22 991	16 050	23 200
B FS2	FS201 Moqhaka							2 240		2 350		2 300	2 350
B FS2	FS203 Ngwathe							2 240		2 350		2 300	2 350
B FS2	FS204 Metsimaholo	784			784			3 024	2 300	2 350		2 300	2 350
B FS2	FS205 Mafube							2 240	2 300	2 350	2 240	2 300	2 350
C DC	DC20 Fezile Dabi District Municipality							2 040	2 250	2 250		2 2 5 0	2 250
Total: Fezile	Fotal: Fezile Dabi Municipalities	784			784			11 784	11 450	11 650	11 784	11 450	11 650
Total Free S	Total: Free State Municipalities	8 445		7 000	8 445		7 000	60 415	54 040	61 700	60 415	54 040	002 19

ANNEXURE W4
SPECIFIC PURPOSE RECURRENT GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULE 6) 2 OF 2

			Water S	Water Services Operating Subsidy Grant	ting Subsidy	Grant				SUB-TOTAL: RECURRENT	RECURREN	T	
		Natio	National Financial Year	Year	Munici	Municipal Financial Year	Year	Natic	National Financial Year	Year	Muni	Municipal Financial Year	Year
Cotogory	Cotonow Municipality	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14
Category	wantelpainty	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)
GAUTENG													
1								010	0.00	0.00	0.00		0.00
A EKU								1.250	002.1	062.1	1.250	1.250	002.1
A JHB	City of Johannesburg							1 250	1 250	1 250	1 250	1 250	1 250
A TSH	City of Tshwane	22 601		15 000	22 601		15 000	27 851	5 000	20 000	27 851	5 000	20 000
B GT421	GT421 Emfuleni							2 050	2 250	2 250	2 050	2 2 5 0	2 250
B GT422	GT422 Midvaal							2 050	2 250	2 250	2 050	2 2 5 0	2 250
B GT425	Lesedi							2 050	2 250	2 250	2 050	2 2 5 0	2 250
C DC42	DC42 Sedibeng District Municipality							2 050	2 250	2 250	2 050	2 2 5 0	2 250
Total: Sedibeng	Fotal: Sedibeng Municipalities							8 200	000 6	000 6	8 200	0006	000 6
B GT481	Mogale City							2 050	2 250	2 250	2 050	2 2 5 0	2 250
B GT482								2 050	2 250	2 250	2 050	2 2 5 0	2 250
B GT483								2 050	2 250	2 250	2 050	2 2 5 0	2 250
B GT484	1 Merafong City							2 050	2 500	2 500	2 050	2 500	2 500
C DC48								2 050	2 250	2 250	2 050	2 2 5 0	2 250
Total: West Ra	Fotal: West Rand Municipalities							10 250	11 500	11 500	10 250	11 500	11 500
Total Canteno	Total: Ganteng Municinalities	22,601		15 000	22,601		15,000	48 801	28 000	43 000	48 801	28 000	43 000

ANNEXURE W4
SPECIFIC PURPOSE RECURRENT GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULE 6) 2 OF 2

			D 7- /11							CITE TOTAL : DECLIDED	TOTAL		
	•	Notice	Water Service	water Services Operating Substay Grant	Ving Subsidy (	Municipal Financial Voca	Voor	Stolk	SUB-	Von	KECUKKEN	Municipal Emondial Von	7,000
		2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14
Category	Municipality	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)
KWAZULU-NATAL	ATAL												
А ЕТН	eThekwini			3 000			3 000	1 250	1 500	4 250	1 250	1 500	4 250
								2 290	2 300	2 650	2 290	2 300	2 650
								2 240	2 300	2 650	2 240	2 300	2 650
								2 2 9 0	2 300	2 650	2 290	2 300	2 650
	_							2 2 4 0	2 300	2 650	2 240	2 300	2 650
B KZN215								2 2 9 0	2 300	2 650	2 290	2 300	2 650
C DC21	16 Hibiscus Coast Ugu District Municipality			2 000			2 000	2 250	2 250	7 250	2 250	2 250	7 250
Total: Ugu Municipalities	nicipalities			2 000			2 000	15 840	16 050	23 150	15 840	16 050	23 150
R KZN221	1 nNchwarhi							0 2 900	2 300	2 650	2 290	2 300	059 6
B KZN222								2 240	2 300	2 650	2 240	2 300	2 650
								2 290	2 300	2 650	2 290	2 300	2 650
B KZN224								2 290	2 300	2 650	2 290	2 300	2 650
								2 231	2 300	2 650	2 231	2 300	2 650
								2 290	2 300	2 650	2 290	2 300	2 650
_								2 290	2 300	2 650	2 290	2 300	2 650
C DC22	uMgungundlovu District Municipality							2 040	2 250	2 250	2 040	2 2 5 0	2 250
Total: uMgung	Total: uMgungundlovu Municipalities							17 961	18 350	20 800	17 961	18 350	20 800
		_						2 240	2 300	2 650	2 240	2 300	2 650
								2 290	2 300	2 650	2 290	2 300	2 650
B KZN234								2 240	2 300	2 650	2 240	2 300	2 650
B KZNZ35								2 240	2 300	2 650	2 240	2 300	2 650
	oo intoapazane Intentala District Municipality			000 0			000 6	2 290	2 250	050 7	067 7	2 250	050 7
Total:Uthukela	ΙΞ			2 000			2 000	13 340	13 750	17 500	13 340	13 750	17 500
B KZN241	11 Endumeni							2 240	2 300	2 650	2 240	2 300	2 650
B KZN242	12 Nguthu							2 290	2 300	2 650	2 290	2 300	2 650
								2 240	2 300	2 650	2 240	2 300	2 650
								2 240	2 300	2 650	2 240	2 300	2 650
C DC24	Umzinyathi District Municipality			2 000			2 000	2 040	2 250	4 250	2 040	2 250	4 250
Total: Umzinya	Total: Umzinyathi Municipalities			2 000			2 000	11 050	11 450	14 850	11 050	11 450	14 850
B KZN252	52 Newcastle	12 910		2 000	12 910		2 000	15 150	2 300	4 600	15 150	2 300	4 600
								2 290	2 300	2 600	2 290	2 300	2 600
B KZN254				- 000			- 000	2 040	2 050	2 600	2 040	2 050	2 600
C DC25	Amajuba District Municipality	9		2 000			2 000	2 040	2 500	4 500	2 040	2 500	4 500
Total: Amajuba	Total: Amajuba Municipalities	12 910		4 000	12 910		4 000	21 520	9 150	14 300	21 520	9 150	14 300

ANNEXURE W4
SPECIFIC PURPOSE RECURRENT GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULE 6) 2 OF 2

			Woton C	onoing Journal	Woton Sourions Openating Subsider Crons				SIR TOTAL BEGINDENT	DECTIDEN	E	
		Nation	National Financial Year	Year	Municipal Financial Year	ancial Year	Nati	National Financial Year	Year	Muni	Municipal Financial Year	Year
Category	Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 2012/13 (R'000) (R'000)	/13 2013/14 00) (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
19UNZA d	onumbo						0,0,0			0770	003.0	0.59.0
CACNTY B							2 4 50	2 500	2 650	0 4 2 0 2 4 50	2 500	2 650
							2450			2.450	2.500	2 650
B KZN265							2 2 40		2 650	2 240	2300	2 650
							2 4 50		2 650	2 450	2 500	2 650
				5 000		5 000			7 250	2 250	2 250	7 250
Total: Zululand Municipalities	Municipalities			2 000		2 000			20 500	14 080	14 550	20 500
									0		000	6
B KZN2/1							2 240		2 650	2 240	2 300	7 650
							2 240	2 300	2 650	2 240	2 300	2 650
	The big 3 raise bay						2 2 40			2 240	2 300	0697
B KZN275							2 240			2 040	2 300	2 650
				2 000		5 000			7 250	2 040	2 2 50	7 250
Total: Umkhany	Total: Umkhanyakude Municipalities			2 000		2 000			20 500	13 040	13 750	20 500
									6			
B KZN281	1 MIOIOZI						2 240	2 300	2 650	2 240	2 300	7 650
							2 240			2 240	2 300	050 7
B KZN283							2 240		2 650	2 240	2 300	2 650
							2 240			2.240	2 300	2 650
B KZN285							2 240		2 650	2 240	2 300	2 650
B KZN286				000		100	2 290	2300	2 650	2 290	2300	2 650
UC28	u i nunguiu District Municipality			000 c		2 000			007 /	2 040	0677	007 /
Total: uThungu	Total: uThungulu Municipalities			2 000		2 000	15 530	16 050	23 150	15 530	16 050	23 150
B KZN291	Mandeni						2 240	2 500	2 600	2 240	2.500	2,600
B KZN292							2 240		2	2 240	2 300	2 600
B KZN293							2 240		2 600	2 240	2 300	2 600
B KZN294	4 Maphumulo						2 240	2 300	2 600	2 240	2 300	2 600
C DC29	iLembe District Municipality			2 000		2 000			4 100	2 040	2 2 5 0	4 100
Total: iLembe Municipalities	<b>Junicipalities</b>		_	2 000		2 000	11 000	11 650	14 500	11 000	11 650	14 500
D CZNIA21	in control						0,000	7 300	009 (	0770	2300	009 0
	Z. Kwa Sani						2 290		2 600	2 290	2 300	2 600
							2.040			2.040	2.300	2.600
							2 040		2 600	2 040	2 300	2 600
B KZN435							2 240		2 600	2 240	2 300	2 600
C DC43	Sisonke District Municipality	322			322		2 362	2 250	2 150	2 362	2 2 5 0	2 150
Total: Sisonke Municipalities	Aunicipalities	322			322		13 212	13 750	15 150	13 212	13 750	15 150
Total: KwaZulu	Total: KwaZulu-Natal Municipalities	13 232		33 000	13 232	33 000	147 823	140 000	188 650	147 823	140 000	188 650

ANNEXURE W4
SPECIFIC PURPOSE RECURRENT GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULE 6) 2 OF 2

		Weter		time Culterial	,				SILE TOTAL . BECIEBERS	DECTIDEN	F	
	Natio	Water Service National Financial Vear	Water Services Operating Subsidy Grant	Municin	Subsitity Graffit Municipal Financial Vear	Vear	Nati	National Financial Vear	Vear	MECURINE	Municinal Financial Vear	Vear
	C1/110C	2012/13	2013/14	2011/10	2012/13	2013/14	C1/110C	2012/13	2013/14	C1/11/0C	2012/13	2013/14
Category Municipality	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)
LIMPOPO												
B I IM 331 Greater Givani							2 040	2 300	2 400	2 040	2 300	2 400
LIM332							2 040	2 300	2 400	2 040	2 300	2 400
							2 040	2 300	2 400	2 040	2 300	2 400
B LIM334 Ba-Phalaborwa							2 040	2 300	2 400	2 040	2 3 0 0	2 400
							2 290	2 300	2 400	2 290	2 3 0 0	2 400
C DC33 Mopani District Municipality	829 69	60 909		829 69	609 09	55 000	71 718	62 859	57 500	71 718	62 859	57 500
Total: Mopani Municipalities	829 69			829 69	609 09	55 000	82 168	74 359	005 69	82 168	74 359	69 500
D ITM241 Mension							070	030.0	,	0,000	030 0	0 400
LIM341							2 040	2 050	2 400	2 040	2 050	2 400
LIM342							2 040	2 300	2 400	2 040	2 300	2 400
LIM343							2 040	2 300	2 400	2 040	2 300	2 400
B LIM344 Makhado	012 031			012 631	122 004	90 00	2 040	2 300	2 400	2 040	2300	2 400
Ž	152 018	132 004	80 045	152 618	132 004	80 045	124 020	133 204	101 705	167 818	134 234	101 705
R I IM351 Blomborn							0000	2 300	2 400	0000	2 300	2 400
							2 040	2 300	2 400	2 040	2 300	2 400
							2 040	2 300	2 400	2 040	2 300	2 400
B LIM354 Polokwane	19 342	8 5 9 8	10 000	19 342	8 5 9 8	10 000	21 382	10 898	12 400	21 382	10 898	12 400
B LIM355 Lepelle-Nkumpi							2 040	2 300	2 400	2 040	2 3 0 0	2 400
C DC35 Capricom District Municipality	42 620	23 641	10 000	42 620	23 641	10 000	44 660	25 891	12 250	44 660	25 891	12 250
Total: Capricorn Municipalities	61 962		20 000	61 962	32 239	20 000	74 452	45 989	34 250	74 452	45 989	34 250
B LIM361 Thabazimbi							2 290	2 300	2 400	2 290	2300	2 400
	3 827			3 827			5 867	2 300	2 400	5 867	2 300	2 400
B LIM364 Mookgopong							2 040	2 300	2 400	2 040	2 300	2 400
LIM365							2 040	2 300	2 400	2 040	2 300	2 400
LIM366	130			130			2 170	2 300	2 400	2 170	2 300	2 400
	11 144	6.009	2 000	11 144	6 0 0 9	2 000	13 184	8 379	7 400	13 184	8 3 7 9	7 400
C DC36 Waterberg District Municipality							2 040	2 250	2 250	2 040	2 2 5 0	2 250
Total: Waterberg Municipalities	15 101	6209	2 000	15 101	6 0 0 9	2 000	29 631	22 129	21 650	29 631	22 129	21 650
B LIM471 Ephraim Mogale							2 040	2 300	2 400	2 040	2 300	2 400
B LIM472 Elias Motsoaledi							2 040	2 300	2 400	2 040	2 3 0 0	2 400
LIM473							2 290	2 300	2 400	2 290	2 300	2 400
B LIM474 Fetakgomo							2 290	2 300	2 400	2 290	2 300	2 400
LIM475							2 040	2 300	2 400	2 040	2 300	2 400
C DC47 Greater Sekhukhune District Municipality	38 855			38 855	38 851	40 000	40 895	40 901	42 250	40 895	40 901	42 250
Total: Greater Sekhukhune Municipalities	38 855	38 851	40 000	38 855	38 851	40 000	51 595	52 401	54 250	51 595	52 401	54 250
Total: Limpopo Municipalities	338 214	269 782	209 945	338 214	269 782	209 945	400 664	338 082	281 445	400 664	338 082	281 445

ANNEXURE W4
SPECIFIC PURPOSE RECURRENT GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULE 6) 2 OF 2

		Wat	Water Services Operating Subsidy Grant	rating Subsidy	Grant				SUB-TOTAL: RECURRENT	RECURREN	T	
	Nat	National Financial Year	ial Year	Munici	Municipal Financial Year	Year	Natio	National Financial Year	Year	Muni	Municipal Financial Year	Year
Category Municipality	2011/12	2012/13	3 2013/14 (P'000)	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14 (P'000)
	(VOCO)	(000 V)	+	(M000 N)	(NOO V)	(000 V)	(000 V)	(000 V)	(M 000)	(NOON)	(A 000)	(a) (A)
MPUMALANGA												
3 MP301 Albert Luthuli	6 963		6 963 5 000	6 963	6 963	2 000	9 003	9 013	7 350	9 003	9 013	7 350
3 MP302 Msukaligwa							2 040	2 300	2 350	2 040	2 300	2 350
3 MP303 Mkhondo							2 040	2 300	2 350	2 040	2 300	2 350
3 MP304 Pixley Ka Seme							2 040	2 300	2 350	2 040	2 300	2 350
3 MP305 Lekwa							2 040	2 050	2 350	2 040	2 050	2 350
3 MP306 Dipaleseng							2 040	2 300	2 350	2 040	2 300	2 350
3 MP307 Govan Mbeki							2 040	2 300	2 350	2 040	2 300	2 350
DC30 Gert Sibande District Municipality	lity						2 2 5 0	2 250	2 250	2 250	2 2 5 0	2 250
Fotal: Gert Sibande Municipalities	6 963		6 963 5 000	6 963	6 963	5 000	23 493	24 813	23 700	23 493	24 813	23 700
3 MP311 Victor Khanve							2 450	2 700	2 400	2 450	2 700	2 400
							2 040	2 300	2 350	2 040	2 300	2 350
3 MP313 Steve Tshwete							2 040	2 500	2 350	2 040	2 500	2 350
3 MP314 Emakhazeni							2 040	2 300	2 350	2 040	2 300	2 350
3 MP315 Thembisile	8889		88 5 000	988 9	6 888	2 000	8 928	8 938	7 350	8 928	8 6 3 8	7 350
3 MP316 Dr JS Moroka	28 785		17 458 15 000	28 785	17 458	15 000	30 825	19 508	17 350	30 825	19 508	17 350
DC31 Nkangala District Municipality							2 250	2 250	2 500	2 250	2 2 5 0	2 500
Fotal: Nkangala Municipalities	35 673	73 24 346	46 20 000	35 673	24 346	20 000	50 573	40 496	36 650	50 573	40 496	36 650
3 MP321 Thaba Chweu	059	20		059			2 690	2 300	2 400	2 690	2 300	2 400
3 MP322 Mbombela	14 501	10 501	01 14 000	14 501	10 501	14 000	16 541		16 400		12 801	16 400
3 MP323 Umjindi							2 040		2 400	2 040	2 300	2 400
3 MP324 Nkomazi	868		85 5 000	8 985	8 985	2 000	11 025	11 285		11 025	11 285	7 400
3 MP325 Bushbuckridge	27 489		16364 25000	27 489	16364	25 000	29 529	18 664	27 400	29 529	18 664	27 400
C DC32 Ehlanzeni District Municipality							2 250	2 500	2 500	2 250	2 500	2 500
Fotal: Ehlanzeni Municipalities	51 625	25 35 850	50 44 000	51 625	35 850	44 000	64 075	49 850	58 500	64 075	49 850	58 500
Fotal: Mpumalanga Municipalities	94 261	67 159	65 000	94 261	62 129	000 69	138 141	115 159	118 850	138 141	115 159	118 850

ANNEXURE W4
SPECIFIC PURPOSE RECURRENT GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULE 6) 2 OF 2

		Woton C	Ononio Ononio	Woton Courioss Oronoting Subsider Cunt	,,,			3	TE TOTAL.	SIB TOTAL: BECIBBENT	_	
	Natic	National Financial Year	Year	Municipal Financial Year	Financial Y	ear	Natio	National Financial Year	/ear	Muni	Municipal Financial Year	Year
Category Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 20 (R'000) (F	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
NORTHERN CAPE												
B NC061 Richtersveld							2 240	2 300	2 650	2 240	2 300	2 650
NC062							2 2 4 0	2 300	2 650	2 240	2 300	2 650
NC064							2 2 4 0	2 300	2 650	2 240	2 300	2 650
B NC06 Kano Hooland							2 040	2 300	2 650	2 240	2 300	2 650
							2 040	2 300	2 650	2 040	2 300	2 650
DC6							2 040	2 250	2 250	2 040	2 2 2 5 0	2 250
Total: Namakwa Municipalities							15 080	16 050	18 150	15 080	16 050	18 150
B NC071 Thuntu							2.240	2,300	2,650	2.240	2,300	2,650
							2 240	2 300	2 650	2 240	2 300	2 650
NC073							2 240	2 300	2 650	2 240	2 300	2 650
NC074							2 650	2 700	2 650	2 650	2 700	2 650
NC075							2 240	2 300	2 650	2 240	2 300	2 650
							2 290	2 300	2 650	2 290	2 300	2 650
NC077							2 240	2 300	2 650	2 240	2 300	2 650
NC078							2 240	2 300	2 650	2 240	2 300	2 650
otol. Div							000 00	2 450	23.450	20 630	2 450	2 250
TOTAL TAILS AND DELIE TRANSPORTED							000 07	007	001	200	007	200
NC081							2 240	2 300	2 650	2 240	2 300	2 650
NC082							2 650	2 700	2 650	2 650	2 700	2 650
NC083							2 240	2 300	2 650	2 240	2 300	2 650
NC084							2 240	2 300	2 650	2 240	2 300	2 650
NC085							2 240	2 300	2 650	2 240	2 300	2 650
B NC086 Kgatelopele							2 290	2 300	2 650	2 290	2300	2 650
otal: Siyanda Mu							15 940	16 450	18 150	15 940	16 450	18 150
NC091							2 240	2 300	2 650	2 240	2 300	2 650
B NC092 Dikgatlong							2 240	2 300	2 650	2 240	2300	2 650
NC094	1 357			1 357			3 647	2 300	2 650	3 647	2 300	2 650
				3			2 450	2 450	2 250	2 450	2 4 50	2 250
otal: Frances Baa	1 357			1 357			12 867	11 650	12 850	12 867	11 650	12 850
B NC451 Moshaweng	3 875			3 875			6 115	2 300	2 650	6115	2 300	2 650
NC452	1 481			1 481			3 721	2 300	2 650	3 721	2 300	2 650
		11 562			11 562		2 240	13 862	2 650	2 240	13 862	2 650
C DC45 John Taolo Gaetsewe District Municipality							2 040	2 250	2 250	2 040	2 250	2 250
Total: John Taolo Gaetsewe Municipalities	5 356	11 562		5 356	11 562	Ī	14 116	20 712	10 200	14 116	20 712	10 200
Total: Northern Cape Municipalities	6713	11 562		6 713	11 562	Ī	78 833	86 112	82 800	78 833	86 112	82 800
	_		-									

ANNEXURE W4
SPECIFIC PURPOSE RECURRENT GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULE 6) 2 OF 2

			Water S	Water Services Operating Subsidy Grant	ting Subsidy (	Grant			02	SUB-TOTAL:	SUB-TOTAL: RECURRENT	L	
		Nation	National Financial Year	Year	Municip	Municipal Financial Year	Year	Natio	National Financial Year	Year	Muni	Municipal Financial Year	Year
Category	Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
NORTH WEST													
B NW371	Moretele	3 205			3 205			5 455	2 500	2 400	5 455	2 500	2 400
		3 910		15 000	3 910		15 000	6 200	2 500	17 400	6 200	2 500	17 400
B NW373		909		2 500	909		2 500	2 646	2 500	4 900	2 646	2 500	4 900
B NW374								2 040	2 500	2 400	2 040	2 500	2 400
B NW375	Moses Kotane	2 542		2 500	2 542		2 500	4 792	2 500	4 900	4 792	2 500	4 900
C DC37	Bojanala Platinum District Municipality							2 040	2 250	2 250	2 040	2 2 5 0	2 250
Total: Bojanala	Total: Bojanala Platinum Municipalities	10 263		20 000	10 263		20 000	23 173	14 750	34 250	23 173	14 750	34 250
B NW381	Ratlon							2 290	2 300	2 400	2 290	2 300	2 400
B NW382	Tswaing							2 040	2 300	2 400	2 040	2 300	2 400
B NW383								2 830	2 500	2 500	2 830	2 500	2 500
B NW384								2 450	3 000	2 500	2 450	3 000	2 500
B NW385	Samotshere Moiloa							2 500	2 300	2 986	2 500	2 300	2 986
C DC38	Ngaka Modiri Molema District Municipality	17 530		15 000	17 530		15 000	19 780	2 250	17 250	19 780	2 2 5 0	17 250
Total: Ngaka M	Fotal: Ngaka Modiri Molema Municipalities	17 530		15 000	17 530		15 000	31890	14 650	30 036	31 890	14 650	30 036
B NW392	Naledi							2 2 9 0	2 300	2 400	2 290	2 300	2 400
B NW393	Mamusa							2 040	2 300	2 350	2 040	2 300	2 350
B NW394	Greater Taung							2 040	2 300	2 350	2 040	2 300	2 350
B NW396	5 Lekwa-Teemane							2 290	2 300	2 350	2 290	2 300	2 350
B NW397								2 000	3 000	3 000	2 000	3 000	3 000
C DC39	Dr Ruth Segomotsi Mompati District Municipality	6 991		15 000	6 991		15 000	9 031	2 250	17 250	9 031	2 2 5 0	17 250
Total: Dr Ruth	Fotal: Dr Ruth Segomotsi Mompati Municipalities	6 991		15 000	6 991		15 000	19 691	14 450	29 700	19 61	14 450	29 700
B NW401	Ventersdorp							2 040	2 300	2 370	2 040	2 300	2 370
B NW402	Tlokwe							2 040	2 300	2 400	2 040	2 300	2 400
B NW403	3 City of Matlosana							2 040	2 300	2 400	2 040	2 300	2 400
B NW404	Haquassi Hills							2 040	2 300	2 400	2 040	2 300	2 400
C DC40	Dr Kenneth Kaunda District Municipality							2 040	2 250	2 250	2 040	2 2 5 0	2 250
Total: Dr Kenne	Fotal: Dr Kenneth Kaunda Municipalities							10 200	11 450	11 820	10 200	11 450	11 820
Total: North We	Total: North West Municipalities	34 784		20 000	34 784		20 000	84 954	55 300	105 806	84 954	55 300	105 806
													1

ANNEXURE W4
SPECIFIC PURPOSE RECURRENT GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULE 6) 2 OF 2

			Water Sc	Water Services Onerating Subsidy Grant	ting Subsidy (	Grant				SUB-TOTAL: RECURRENT	RECURREN	_	
		Nation	National Financial Year	Year	Municip	Municipal Financial Year	Year	Natio	National Financial Year	Year	Muni	Municipal Financial Year	Year
Category	Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
WESTERN CAPE	36												
A CPT	City of Cape Town							1 250	1 250	1 250	1 250	1 250	1 250
B WC011	Matzikama							2 040	2 050	2 150	2 040	2 050	2 150
B WC012	Cederberg	193		4 000	193		4 000	2 233	2 050	6 150	2 233	2 050	6 150
								2 040	2 050	2 150	2 040	2 050	2 150
B WC014								2 040	2 050	2 150	2 040	2 050	2 150
C DC1	Swaruand West Coast District Municipality	303		4 000	303		4 000	2 343	2 050	6 250	2 343	2 050	6 250
Total: West Coas	Total: West Coast Municipalities	496		8 000	496		8 000	12 736	12 300	21 000	12 736	12 300	21 000
B WC022	Witzenberg							2 450	2 250	2 250	2 450	2 2 5 0	2 250
B WC023								2 040	2 050	2 150	2 040	2 050	2 150
B WC024	Stellenbosch							2 040	2 050	2 150	2 040	2 050	2 150
								2 040	2 050	2 150	2 040	2 050	2 150
B WC026	Langeberg Cane Winelands District Municipality							2 040	2 050	2 150	2 040	2 050	2 150
Total. Come Win	class Wincianus Districtivimino parity							040.7	13 700	12 100	12,650	13 700	12 100
10tal: Cape Win	Lotal: Cape Winelands Municipalities							12 050	17 /00	001 CI	000 71	17 / 00	13 100
B WC031	Theewaterskloof							2 450	2 500	2 500	2 450	2 500	2 500
	Overstrand							2 040	2 050	2 150	2 040	2 050	2 150
B WC033	Cape Agulhas							2 040	2 050	2 150	2 040	2 050	2 150
B WC034	Swellendam Overberg District Municipality							2 040	2 106	2 150	2 040	2 106	2 150
Total: Overberg Municipalities	Municipalities							10 610	10 956	11 200	10 610	10 956	11 200
B WC041	Kannaland							2 970	2 750	2 264	2 970	2.750	2 264
								2 040	2 050	2 150	2 040	2 050	2 150
								2 040	2 050	2 150	2 040	2 050	2 150
B WC045	George	1 807		000	1 804		000	2 040	2 050	2 150	2 040	2 050	2 150
	Bitou	1001		P .	1		8	2 040	2 050	2 150	2 040	2 050	2 150
								2 040	2 050	2 150	2 040	2 050	2 150
C DC4	Eden District Municipality							2 040	2 250	2 250	2 040	2 2 5 0	2 250
Total: Eden Municipalities	nicipalities	1 894		4 000	1 894		000	19 144	17 300	21 414	19 144	17 300	21 414
B WC051	Laingsburg							2 290	2 300	2 400	2 290	2 300	2 400
	Prince Albert							2 040	2 050	2 150	2 040	2 050	2 150
B WC053	Beaufort West							2 040	2 050	2 150	2 040	2 050	2 150
Total. Control K	Total: Control Kono Municipalities							040 7	062.2	050 8	2 040	057 7	050 8
Total: Central N	aroo Muncipanues							0 410	0000	05% Q	0410	nco o	06% 6
		0000		900 61	0000		000	000	744 67	1000	00047	220 62	1
Total: Western (	Total: Western Cape Municipalities	2 390		12 000	2 390		12 000	94 800	63 156	76 914	04 800	63 156	76 914
Unallocated:								20 000	75 000	100 000			
National Total		560 794	399 000	420 945	560 794	399 000	420 945	1 214 855	1 107 969	1 189 765	1 164 855	1 032 969	1 089 765

## ANNEXURE WS

## INFRASTRUCTURE GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULES 4, 6 AND 9)

(National and Municipal Financial Years)

ANNEXURE W5
INFRASTRUCTURE GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULES 4 AND 6) 1 OF 4

		Muni	Municipal Infrastructure	ture Grant				Urban S	ettlements Dev	Urban Settlements Development Grant			Neigh	Neighbourhood Development Partnership Grant (Capital Grant)	elopment Part	tnership Grant	(Capital Gran	G G
	Natio	National Financial Year	ar		cial >		National	National Financial Year		Municip	Municipal Financial Year	ear	Nation	National Financial Year	ear	Munic	Municipal Financial Year	ear
Category Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 .: (R'000) (	2011/12 (R000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
EASTERN CAPE																		
A BUF Buffalo City A NMA Nelson Mandela Bay							423 446 502 626	497 908 588 100	547 338 648 074	423 446 502 626	497 908 588 100	547 338 648 074	000 09	5 000	10 000	000 09	5 000	10 000
EC101	12 973	15 772	16640	12 973	15 772	16 640												
	8 446	10 268	10 833	8 446	10 268	10 833												
B EC104 Makana B EC105 Ndlambe	24 312	29 560	31 185	24 312	29 560	31 185							15 000	18 000	13 071	15 000	18 000	13 071
EC106	19 504	23 714	25018	19 504	23 714	25 018												
B EC107 Baviaans B EC108 Kouga	10 447	12 701	13 400 29 505	10 447 23 002	12 701 27 967	13 400 29 505												
	14 643	17 804	18 784	14 643	17 804	18 784												
151	152 547	185 474	195 675	152 547	185 474	195 675							15 000	18 000	13 071	15 000	18 000	13 071
B EC121 Mbhashe	32 809	39 893	42 087	32 809	39 893	42 087												
BC122	44 363	53 941	26907	44 363	53 941	26 907												
B EC123 Great Kei B EC124 Amahlathi	11 892	14 460	15255	77 884	14 460	15 255												
EC124	17 257	20 983	22 137	17 257	20 983	22 137												
B EC127 Nkonkobe	21 693	26 377	27 828	21 693	26 377	27 828												
B EC128 Nxuba C DC12 Amathole District Municipality	321 323	390 699	12 403	9 669	390 699	12 403							34 000	40 000	20 000	34 000	40 000	20 000
e	481 891	585 933	618159	481891	585 933	618159							34 000	40 000	20 000	34 000	40 000	50 000
B EC131 Inxuba Yethemba	10 904	13 259	13 988	10 904	13 259	13 988												
	9 631	11 710	12354	9 631	11 710	12 354												
B EC133 Inkwanca	8 236	10 015	10 565	8 236	10 015	10 565									0000			0000
	25 935	31 534	33 269	25 935	31 534	33 269									000 6			000 6
	19 217	23 366	24 651	19 217	23 366	24 651									10 000			10 000
B EC138 Sakhisizwe	13 431	16 330	17 228	13 431	16 330	17 228												
C DC13 Chris Hani District Municipality	284 604	346 051	365 084	284 604	346051	365 084												
Total: Chris Hani Municipalities	420 060	510 750	538 841	420 060	510750	538 841									19 000			19 000
	22 466	27 317	28 819	22 466	27 317	28 819												
B EC142 Sengu	23 102	28 090	29 635	23 102	28 090	29 635							2 900			2 900		
	9 928	12 072	12736	9 928	12 072	12 736												
C DC14 Ukhahlamba District Municipality Total: IIkhahlamba Municipalitise	143 957	175 038	184 665	143 957	175 038	184 665	+	+	$\dagger$	+	$\dagger$	$\dagger$	2 900	$\dagger$	1	2 900		
HILL IN COLUMN IN	i i	000 000	0,70	0	000 000	27												
B EC154 Port St Johns	20 254	24 626	25 981	20254	24 626	25 981												
	32 771	39 846	42 038	32 771	39 846	42 038												
B EC157 King Sabata Dalindyebo	28 338 46 279	34 456 56 271	59 365	46 279	54 450 56 271	59 365												
DC15	562 773	684 281	721916	562 773	684 281	721916												
Total: O.R.Tambo Municipalities	721 959	877 833	926114	721 959	877 833	926114			+	+		1	1			$\dagger$		
B EC441 Matatiele	28 791	35 008	36933	28 791	35 008	36 933												
	29 445	35 802	37771	29 445	35 802	37 771												
BC152	19 664	23 909	25 224	19 664	23 909	25 224												
Z	417 203	507 279	535 180	417 203	507 279	535 180												
Total: Eastern Cape Municipalities	2 405 728	2 925 122	3 086 004	2 405 728	2 9 2 5 1 2 2	3 086 004	926 072	1 086 008	1 195 412	926 072	1 086 008	1 195 412	111 900	123 000	128 071	111 900	123 000	128 071

ANNEXURE W5
INFRASTRUCTURE GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULES 4 AND 6) 1 OF 4

		Mur	Municipal Infrastructure Grant	ture Grant				Urban	Settlements D	Urban Settlements Development Grant	ant		Neigh	bourhood Dev	Neighbourhood Development Partnership Grant (Capital Grant)	nership Gran	(Capital Gra	nt)
	Nati	National Financial Year		Munici	pal Financial 1	ear	Nation	National Financial Year	ar	Munic	Municipal Financial Year	Year	Nation	National Financial Year	ear	Munic	Municipal Financial Year	Year
	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14
Category Municipality	(R'000)	(R000)		(R'000)	(R'000) (R'000) (R'000)	(R'000)	(R000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)
EE STATE																		
MAN Mangaung							411 995	483 427	531 884	411 995	483 427	531 884						
	01001	27	22.260	01001	27	23.260												
FS161 Edselleng FS162 Konanony	19 390	23 576	25 539	19390	23 576	24 873												
	15 533	2000	10003	15 533	20001	10 073												
FS163 Monokare FS164 Naledi	13 332	17 203	18150	14 149	17 203	18 150												
DC16 Xhariep District Municipality																		
al: Xhariep Municipalities	67 281	81 805	86 305	67 281	81 805	86 305												
	000			00000		700 00												
FS181 Masilonyana	30 322	36 868	38896	30 322	36 868	38 896												
	17 855			1/855		22 903												
FS183 Tswelopele	25 015			25015		32 088												
FS184 Matjhabeng	164 896			164896	61	211526							2 000	000 9	7 000	2 000	0009	7 000
	45 642			45 642	55 497	58 549												
DC18 Lejweleputswa District Municipality																		
al: Lejweleputswa Municipalities	283 731	344 988	363 963	283 731	344 988	363 963							5 000	000 9	2 000	2 000	000 9	7 000
FS191 Seisoto	62.963			62.963	76.557	80 768												
	TD 6.47			12647	51.855	5.07.02							4000	000	2,000	7 000	2000	7,000
FS 193 Netcana	28 044	34 098	35 973	28044	34 098	35 973							200	000	80	1	900	000 /
	171 336			171336	208 329	219 787												
	21 947			21947	26 686	28 154												
	22 773			22 773	27 689	29 212												
DC19 Thabo Mofutsanyana District Municipality																		
al: Thabo Mofutsanyana Municipalities	349 711	425 214	448 600	349 711	425 214	448 600							4 000	2 000	2 000	4 000	2 000	7 000
FS201 Moqhaka	38 661		49 592	38 661	47 007	49 592												
	41 582		53.340	41.582	50.560	53 340												
FS204 Metsimaholo	38 843	47 228	49 826	38 843	47 228	49 826												
	21 303		27 326	21 303	25 901	27 326												
DC20 Fezile Dabi District Municipality																		
al: Fezile Dabi Municipalities	140 388	170 696	180 085	140 388	170 696	180 085												
al: Free State Municipalities	841 111	1 022 704	1 078 952	841111	1 0 22 7 04	1 078 952	411 995	483 427	531 884	411 995	483 427	531 884	0006	11 000	14 000	000 6	11 000	14 000

ANNEXURE W5
INFRASTRUCTURE GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULES 4 AND 6) 1 OF 4

		Muni	Municipal Infrastructure Gran	ture Grant				Urban	Urban Settlements Development Grant	velopment Gra	nt nt		Neigl	bourhood Dev	Neighbourhood Development Partnership Grant (Capital Grant)	tnership Gra	nt (Capital Gra	III)
	Nation	National Financial Year	ar	Munici	unicipal Financial Year	/ear	Nation	National Financial Year	ar	Munici	Municipal Financial Year	ear	Nation	National Financial Year	'ear	Muni	Municipal Financial Year	Year
	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14
Category Municipanty	(R'000)	(R'000)	(R'000)	(R000)	(R000)	(R000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)
GAUTENG																		
A EKU Ekurhuleni							1 094 276	1 297 640	1 421 452	1 094 276	1 297 640	1 421 452	20 000	15 000	10 000	20 000	15 000	10 000
							1 027 970	1216260	1 333 559	1 027 970	1 216 260	1 333 559	000 09	000 09	50 000	000 09	000 09	20 000
A TSH City of Tshwane							891 081	1 053 856	1 155 692	891081	1 053 856	1 155 692	45 000	20 000	000 59	45 000	20 000	65 000
B GT421 Emfuleni	126 985	154 402	162 894	126 985	154 402	162 894												
B GT422 Midvaal	22 845	27 776	29 304	22 845	27 776	29 304												
	23 139	28 135	29 683	23 139	28 135	29 683												
C DC42 Sedibeng District Municipality													14 000	25 000	25 000	14 000	25 000	25 000
Fotal: Sedibeng Municipalities	172 969	210 313	221 880	172 969	210 313	221 880							14 000	25 000	25 000	14 000	25 000	25 000
B GT481 Mogale City	80957	98 436	103 850	80 957	98 436	103 850							20 000	30 000	30 000	20 000	30 000	30 000
B GT482 Randfontein	30 284	36 822	38 847	30 284	36 822	38 847												
B GT483 Westonaria	54184	65 882	905 69	54 184	65 882	905 69												
B GT484 Merafong City	61137	74 337	78 425	61 137	74 337	78 425												
C DC48 West Rand District Municipality													2 000	10 000	10 000	2 000	10 000	10 000
Total: West Rand Municipalities	226 563	275 477	290 628	226 563	275 477	290 628							25 000	40 000	40 000	25 000	40 000	40 000
Total: Gauteng Municipalities	399 532	485 790	512 509	399 532	485 790	512 509	3 013 327	3 567 756	3 910 703	3 0 13 3 27	3 567 756	3 910 703	164 000	190 000	190 000	164 000	190 000	190 000

ANNEXURE W5
INFRASTRUCTURE GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULES 4 AND 6) 1 OF 4

		Munic	Municipal Infrastructure	ure Grant				Urban	Urban Settlements Development Grant	velopment Gr	m		Neigh	bourhood Dev	velopment Par	tnership Gran	Neighbourhood Development Partnership Grant (Capital Grant)	nt)
	Natio	National Financial Year		Municip	ial 🕽	ear	Nationa	National Financial Year	ar	Munic	Municipal Financial Year	Year	Nation	National Financial Year	ear	Muni	Municipal Financial Year	Year
Category Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
KWAZULU-NATAL																		
A ETH eThekwini							1 091 574	1 299 706	1 421 322	1 091 574	1 299 706	1 421 322	85 100	000 06	50 000	85 100	00006	50 000
B KZN211 Vulamehlo	13 526	16 447	17 351	13 526	16 447	17 351												
KZN212	13 028	15 840	16 711	13 028	15 840	16 711												
B KZN213 Umzumbe	24 655	29 978	31 627	24 655	29 978	31 627												
KZN214	15 128	18 394	19 405	15 128	18 394	19 405												
	11 125	13 527	14 271	11 125	13 527	14 271												
9	25 379	30 859	32 556	25 379	30 859	32 556												
Total: Ugu Municipalities	341888	415 702	438 565	341 888	415 702	438 565												
KZN221	15171	18 447	19 462	15 171	18 447	19 462												
	13347	16 228	17 120	13 347	16 228	17 120												
	899 6	11 756	12 402	899 6	11 756	12 402												
	8 871	10 786	11 379	8 871	10 786	11 379												
	126455	153 758	162 215	126 455	153 758	162 215							4 000	9 000	20 000	4 000	000 9	20 000
	10258	12 472	13 157	10 258	12 472	13 157												
B KZN22/ Kichmond C DC32 "Memorandlern Diensich Municipalite	13500	16 414	17.317	13 500	10.4 39.5	17.317												
- 18	363 131	344 346	110 127	000 000	344 346	363 100							4 000	0000	00000	4000	0000	000 00
Total: unigungungovu ivi unicipanites	171 607	244 740	001 COC	171 607	34 40	nor cac		İ	İ	İ			000 +	000 0	70,000	000+	0000	000 07
B KZN232 Emnambethi-Ladysmith	25 356	30 830	32 526	25 356	30 830	32 526							12 000	13 000	5 500	12 000	13 000	5 500
B KZN233 Indaka	16 770	20 391	21 513	16 770	20 391	21 513												
KZN234	10 510	12 779	13 482	10 510	12 779	13 482							4 000	3 000	1 000	4 000	3 000	1 000
B KZN235 Okhahlamba	19 153	23 287	24 568	19 153	23 287	24 568												
	17 281	21 013	22 168	17 281	21 013	22 168												
C DC23 Uthukela District Municipality	150740	183 286	193 367	150 740	183 286	193 367												
Total: Uthukela Municipalities	239 812	291 587	307 624	239 812	291 587	307 624							16 000	16 000	6 500	16 000	16 000	6 500
B KZN241 Endumeni	10.251	12 463	13 149	10 251	12 463	13 149												
B KZN242 Nquthu	21 020	25 558	26 963	21 020	25 558	26 963												
B KZN244 Msinga	21 981	26 727	28 197	21 981	26 727	28 197												
B KZN245 Umvoti	14 732	17 912	18 897	14 732	17 912	18 897												
C DC24 Umzinyathi District Municipality	161 942	196 907	207 737	161 942	196 907	207 737												
Total: Umzinyathi Municipalities	229 926	279 567	294 944	229 926	219 567	294 944												
	74 224	90 249	95 213	74 224	90 249	95 213							12 000	13 000	25 000	12 000	13 000	25 000
	8 624	10 486	11 062	8 624	10 486	11 062												
B KZN254 Dannhauser	15337	18 648	19 673	15 337	18 648	19 673												
12	1120 306	30 109	170 013	130 307	30 109	170 013		Ì					13 000	13 000	00000	12,000	13,000	25,000
rotan zanajuba satumujpanues	200	**** YOT	1/0 040	10 / CO	10.7	1/0 0/1							- AAA	Trans	2000	7 0000	To ooo	A

ANNEXURE W5
INFRASTRUCTURE GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULES 4 AND 6) 1 OF 4

		Minni	Municipal Infrastructure Gra	fure Great		_		Ilrhan	Tirban Settlements Development Grant	velonment Gr	ant		oioN	bhourhood De	velonment Per	Neighbourhood Develonment Partnershin Grant (Canital Grant)	(Conited Gre	ut)
	Natio	National Financial Year	ar ar	Mimici	nt unicinal Financial Vear	ear	Nation	National Financial Year	ar ar	Minic	Municinal Financial Year	Vear	Natio	National Financial Year	Venopment 1 an	Munici	Municipal Financial Year	Year
	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14
Category Municipality	(R'000)	(R'000)	(R'000)	(R000)	(R000)	(R000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R000)	(R'000)	$\dashv$	(R'000)	(R'000)
		4 600			907	100												
_	12 /46	15 498	100 01	12 /40	15 498	100 01												
KZN262	16 793	20 4 19	21 542	16 793	20 4 19	21 542												
	20 975	25 504	26 906	20 975	25 504	26 906												
	20 868	25 373	26 769	20 868	25 373	26 769							10 000	10 000	10000	10 000	10000	10 000
KZN266	22 011	297.90	28 234	22 011	26.762	28.734												
2000	201100	227 220	010100	222 100	020	010100												
DC20 Zululand District Municipality	227 100	770 137	616 167	227 100	201 0/2	616 167							0000	000	0000	0000	0 0 0	000
Total: Zululand Municipalities	320 493	389 688	411 121	320 493	389 688	411 121							10 000	10 000	10 000	10 000	10 000	10 000
	000001	100	100	- 000 01	- 100													
	19 908	24.205	755 537	19 908	24 205	75 53/												
KZN272 Jozini	24 172	29 391	31 008	24 172	29 391	31 008												
	9 234	11 228	11 845	9.234	11 228	11 845												
VZNOTA Hlakies	0 60	11 0/3	12 600	600	11 0/3	12 600												
	440 6	200.00	12 000	779 6	20000	12 000												
_	1/193	70.607	/ SO 77	661 / 1	706.07	75 05/												
DC27 Umkhanyakude District Municipality	186 942	227 304	239 806	186 942	227 304	239 806												
Total: Umkhanyakude Municipalities	267 273	324 978	342 852	267 273	324 978	342 852												
		e e	0	i i	000	0												
	14 4/5	1/ 600	18 208	14 4/5	1/ 600	18 268												
	71 404	86 820	91 595	71 404	86 820	91 595												
KZN283 Ntambanana	10 193	12 393	13 075	10 193	12 393	13 075												
KZN284 Umlalazi	22 284	27 095	28 585	22 284	27 095	28 585												
	10.761	13 083	13 803	10 761	13 083	13 803												
	17 887	21 748	22 944	17 887	21 748	22 944												
	160933	195 680	206 442	160 933	195 680	206 442												
18	920208	374 419	305 012	307 036	374 419	395 012												
an armangana managamace	000 100	2	-	000 100		-												
KZN291 Mandeni	21 152	25 7 19	27 134	21 152	25 719	27 134							12 000	13 000	15 000	12 000	13 000	15 000
	28075	35 231	37 160	28 075	35 231	37 160							15,000	8 000		15,000	0008	
	30 104	24 552	25.00	20,00	24 552	35 004							000 01	9000	000 10	000 01	0000	000.10
Winzby indwedwe	16107	24 333	2 304	100 77	24 333	10,00							000 01	00000	71 000	10,000	9 000	71 000
_	16881	50 525	21 654	16 881	20 272	21 654												
DC29 iLembe District Municipanty	14 / 801	11/6/1	189 396	14/801	11/6/1	189 596												
Fotal: iLembe Municipalities	235 003	285 740	301 456	235 003	285 740	301 456							37 000	29 000	36 000	37 000	29 000	36 000
KZN431 Inewe	16 569	20 146	21 254	16 569	20 146	21 254												
	7.887	0850	10 117	7 887		10.117												
	1500.51	10.365	20 430	15 007		20 430												
NAMES OF GREATER NOKSUAGE	12 921	500 61	20 450	126 CI		20 4 30												
	17 409	21 167	22 331	17 409		22 331												
	39 688	48 256	50 910	39 688		50 910							2 000	000 9		2 000	000 9	
DC43 Sisonke District Municipality	136610	166 105	175 240	136 610		175 240												
Fotal: Sisonke Municipalities	234 089	284 628	300 283	234 089	284 628	300 283							2 000	000 9		2 000	000 9	
W. C. Water Manager and C.	200 003 0	210004	2 222 640	2 500 002	2100 047	073 040	1 000 574	1 200 700	1 431 333	1001	1 200 200	1 421 222	001 001	000 021	147 500	100 100	000001	1.47 500
Total: KwaZulu-Natal Municipalities	166 9667	3 160 047	3 333 847	7 238 337	3 160 047	3 333 849	1 091 5/4	1 299 700	776 176 1	10915/4	1 299 706	1 471 344	169 100	170 000	147 500	169 100	170000	Ť

ANNEXURE WS
INFRASTRUCTURE GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULES 4 AND 6) 1 OF 4

		Muni	Municipal Infrastructure Gr	cture Grant		-		Urban	Urban Settlements Development Grant	velopment Gr.	ant		Neig	hbourhood De	velopment Par	Neighbourhood Development Partnership Grant (Capital Grant)	t (Capital Gra	ıç
	Nation	National Financial Year	sar	Munici	Aunicipal Financial Year	Year	Nations	National Financial Year	ar	Munic	Municipal Financial Year	Year	Natio	National Financial Year	Year	Munic	Municipal Financial Year	Year
Category Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
LIMPOPO																		
	29 950	36 416	38 419	53	36 416	38 419												
	32 997	40 121	42 328		40 121	42 328												
LIM333	46 712	56 798	59 922	94	56 798	59 922							14 000	10 000	20 000	14 000	10 000	20 000
LIM334 Ba-Phalaborwa	17 129	20 827	21 972	17	20 827	21 972							7 000		12500		15 000	12 500
LIM335 Maruleng	24 874	30 245	31 908	22	30 245	31 908												
DC33 Mopani District Municipality	263 219	320 050	337 652	263	320 050	337 652												
151	414882	504 456	532 201	414	504 456	532 201							21 000	25 000	32 500	21 000	25 000	32 500
	12 039	14 638	15 443	12 039	14 638	15 443												
	13 995	17 016	17 952	13 995	17 016	17 952												
	61 295	74 529	78 628	61 295	74 529	78 628							25 000	29 000	30 000	25 000	29 000	30 000
LIM344 Makhado	55 562	67 558	71 274	55 562	67 558	71 274												
DC34 Vhembe District Municipality	296276	360 245	380 058	296 276	360 245	380 058												
Total: Whembe Municipalities	439 169	533 986	563 355	439 169	533 986	563 355							25 000	29 000	30 000	25 000	29 000	30 000
LIM351 Blombero	25 477	30.977	32 680	25 477	30 977	32 680												
	21 739	26.433	77 887	21 739	26.433	77.887												
	19716	23 972	25 290	19 716	23 972	25 290												
	182 607	222 032	23.4.24.4	182 607	22 032	234 244							17 000	18 640	20 000	17 000	18 640	20.070
	00 201	37 017	36 929	00 701	374 017	36 939							-		(1) (1		1001	
	173914	211.462	223 063	173 914	21.1462	223 093												
15	452169	549 793	580 032	452 169	549 793	580 032							17 000	18 649	29 929	17 000	18 649	29 929
	34 750	42 252	44 575	34 750	42 252	44 575												
	35 907	43 659	46 060	35 907	43 659	46 060												
	13 536	16 458	17 364	13 536	16 458	17 364												
	34 157	41 531	43 815	34 157	41 531	43 815												
	15 518	18 868	19 906	15 518	18 868	19 906												
LIM367 Mogalakwena	113 222	137 668	145 239	113 222	137 668	145 239							28 000	12 000	5 000	28 000	12 000	5 000
DC36 Waterberg District Municipality																		
Total: Waterberg Municipalities	247 090	300 436	316 960	247 090	300 436	316 960							28 000	12 000	5 000	28 000	12 000	2 000
LIM471 Enhaim Mosale	17.871	21 729	22,924	17.871	21 729	22 924												
	29 037	35 306	37 248	29 037	35 306	37 248												
	34 159	41 533	43 817	34 159	41 533	43.817												
LIM474	15 407	18 734	19 764	15 407	18 734	19 764												
LIM475	36 311	151 44	46.580	36.311	151 44	46.580												
DC47	344 208	418 525	441 544	344 208	418 525	441 544												
Total: Greater Sekhukhune Municipalities	476994	879 978	611 877	476 994	816 615	611 877												
Total: Limnono Municipalities	2.030.304	2 468 649	2.604.425	2.030.304	2 468 649	2 604 425							01 000	84 640	07 470	01000	84649	07 470

ANNEXURE W5 INFRASTRUCTURE GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULES 4 AND 6) 1 OF 4

		Min	Municinal Infrastructure Gr	cture Great				Trhan	. Sottlemente D	Urban Settlements Desclonment Grant	on t		Noio	Neighbourhood Development Partnershin Grant (Canital Grant)	evelonment Pe	artnorchin Gr	ont (Canital G	ront
	Acid	V lobourough Lond	Too.	Manie	to Disconded	Vector	Notice	V logonough los	, and the same and	Ment	Menicinal Financial Vern	Veen	Notice of	Tool Electroid	Veen	To discount of the last	defined Pieces	ol Voon
	2011/12	National Financial Leaf	2013/14	2011/12	77   2012/13   201	2013/14	2011/12	7 2012/13 Fear	2013/14	1 C1/11/0C	2012/13	2013/14	2011/12		2013/14	C1/11/0C	72 2012/13 2017	2013/14
Category Municipality	(R'000)	(R'000)	(R'000)	(R'000)	(R000)	(R000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R000)	(R'000)	(R'000)	(R'000)	(R'000)
PUMALANGA																		
MP301 Albert Luthuli	63 670	77 416	81 674	63.670	77.416	81 674												
	33285		42.697	33 285	40 471	42.697												
	48 703		62 476		59 219	62 476												
	26 576		34 090		32 313	34 090												
	35877		46 022		43 622	46 022												
MP306 Dipaleseng	17881	21 742	22 937		21 742	22 937												
MP307 Govan Mbeki DC30 Gert Sibande District Municipality	75755	92 111	771 76	75 755	92 111	771 76												
1.2	301 748	708 99E	387 074	301 748	366 894	387 074												
MP311 Victor Khanye	20 521	24 951	26 324	20 521	24 951	26 324												
MP312 Emalahleni	78 122		100 214	78	686 46	100 214												
MP313 Steve Tshwete	30 957	37 640	39 711		37 640	39 711							15 000	10 000	20 000	15 000	10000	20 000
MP314 Emakhazeni	13 131	15 966	16 844	13 131	15 966	16 844												
MP315 Thembisile	89 106	108 345	114 304	89 106	108 345	114 304												
MP316 Dr JS Moroka	91325	111 042	117 149	91 325	111 042	117 149												
DC31 Nkangala District Municipality																		
tal: Nkangala Municipalities	323 163	392 935	414 546	323 163	392 935	414 546							15 000	10 000	20 000	15 000	10 000	20 000
MP321 Thaba Chweu	26 690	32 452	34 237	26 690	32 452	34 237												
	155031	188 503	198 870	155	188 503	198 870							9 200			6500		
MP323 Umjindi	22 407		28 743	22	27 245	28 743												
	112 208	136 434		112	136 434	143 938												
MP325 Bushbuckridge	235 834			235	286 751	302 523												
DC32 Ehlanzeni District Municipality																		
tal: Ehlanzeni Municipalities	552 171	671 386	708 312	552 171	671 386	708 312							002 9			6 500		
tal: Mpumalanga Municipalities	1177 082	1 431 215	1 509 931	1 177 082	1 431 215	1 509 931							21 500	10 000	20 000	21500	10 000	20 000

ANNEXURE W5
INFRASTRUCTURE GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULES 4 AND 6) 1 OF 4

		Mun	Municipal Infrastructure Grant	ture Grant				Urban S.	Urban Settlements Development Grant	elopment Gra	1		Neigh	De De De	Neighbourhood Development Partnership Grant (Capital Grant)	tnership Gran	t (Capital Grai	9
	Natio	National Financial Year	ear	Municit	funicipal Financial Year	Vear	Nationa	National Financial Year	_	Municit	Municipal Financial Year	ear	Nation	National Financial Year	Year	Munic	Municipal Financial Year	Year
Category Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000) (	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
NORTHERN CAPE																		
NC061	7 113	8 648	9 124	7 113	8 648	9 124												
NC062	13 263	16 126	17 013	13 263	16 126	17 013												
	9 634	11 714	12 358	9 634	11 714	12 358												
NC065	10 505	12 773	13 475	10 505	12 773	13 475												
NC066	9 383	11 408	12 035	9 383	11 408	12 035												
B INCOO Khai-Ma C DC6 Namakwa District Municipality	0.480	10 /48	11 559	8 840	IO /48	11 559												
I A	58739	71 417	75 345	58 739	71 417	75 345												
B NC071 Ubuntu	9 488	11 537	12 171	9 488	11 537	12 171												
NC072	11765	14 305	15 092	11 765	14 305	15 092												
NC073	13 306	16 178	17 068	13 306	16 178	17 068												
NC074	7 892	9 596	10 124	7 892	9 596	10 124												
B NC075 Renosterberg	8 170	9 934	10 480	8 170	9 934	10 480												
	11186	13 600	14 349	11 186	13 600	14 349												
B NC077 Siyathemba	10246	12 458	13 144	10 246	12 458	13 144												
B NC078 Siyancuma C DC7 Biylov Ka Same Dietrict Municipality	16910	20 561	21 692	16 910	20 201	21 692												
2	88 965	108 170	114 119	88 965	108 170	114 119							İ					
•																		
B NC081 Mier	9 385	11 411	12 038	9 385	11 411	12 038												
	17.293	21 02/	22 183	17.293	21 02/	22 183												
NC083	11 434	13 907	14 667	11 434	13 902	14 667												
NC085	11888	14 454	15 249	11 888	14 454	15 249												
B NC086 Kgatelopele	7 27 T	9 395	9 9 11	727 T	9 395	9 911												
C DC8 Siyanda District Municipality																		
Total: Siyanda Municipalities	77 338	94 033	99 205	77 338	94 033	99 205												
B NC091 Sol Plantie	45363	55 157	58 190	45 363	55 157	58 190							10 000	15 000	15 000	10 000	15 000	15 000
NC092	21118	25 677	27 089	21 118	25 677	27 089												
NC093	10 798	13 129	13 851	10 798	13 129	13 851												
_	21735	26 428	27 881	21 735	26 428	27 881												
C DC9 Frances Baard District Municipality																		
Total: Frances Baard Municipalities	99 015	120 391	127 012	99 015	120 391	127 012							10 000	15 000	15 000	10 000	15 000	15 000
	41128	20 008	52 758	41 128	50 008	52 758												
NC452	45 474	55 292	58 333	45 474	55 292	58 333		_	_		_		5 000	10 000	10 000	5 000	10 000	10 000
	14 2 4 9	17 325	18 278	14 249	17 325	18 278												
C DC45 John Taolo Gaetsewe District Municipality																		
Total: John Taolo Gaetsewe Municipalities	100 852	122 624	129 369	100 852	122 624	129 369							2 000	10 000	10 000	2 000	10 000	10 000
Total: Northern Cane Municipalities	424 909	516 634	545 049	424 909	516 634	545 049							15 000	25 000	25 000	15 000	25,000	25 000

ANNEXURE W5
INFRASTRUCTURE GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULES 4 AND 6) 1 OF 4

		Muni	Municipal Infrastructure G	ture Grant				Urban	Urban Settlements Development Grant	velopment Gra	ınt		Neigh	bourhood Dev	elopment Part	mership Grant	Neighbourhood Development Partnership Grant (Capital Grant)	
		National Financial Year		Munici	Municipal Financial Year	Year	Nations	National Financial Year	ar	Munica	Municipal Financial Year	Year	Nation.	National Financial Year	ear	Munici	Municipal Financial Year	ear
Category Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R000)	2012/13 (R'000)	2013/14 (R000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
NORTH WEST																		
B NW371 Moretele	85713	104 218	109 950	85 713	104 218	109 950												
	163 940	199 336	210 299	163 940	199 336	210 299												
	166947	202 992	214 156	166 947	202 992	214 156							4 000	10 000	10 000	4 000	10 000	10 000
B NW374 Kgetlengrivier	16 600	20 184	21 294	16 600	20 184	21 294												
	102 056	124 090	130 915	102 056	124 090	130 915												
C DC37 Bojanala Platinum District Municipality																		
Total: Bojanala Platinum Municipalities	535 256	650 820	686 615	535 256	650 820	686 615							4 000	10 000	10 000	4 000	10 000	10 000
B NW381 Botton	18.073	21 075	23 183	18 073	270 10	23 183												
	20068	25 495	26.897	20 068	25 495	26.897												
	35381	43 019	45 385	35 381	43 019	45 385												
	25 393	30 876	32 574	25 393	30 876	32 574												
B NW385 Ramotshere Moiloa	22 588	27 465	28 976	22 588	27 465	28 976							10 000	15 000	2 000	10 000	15 000	2 000
C DC38 Ngaka Modiri Molema District Municipality	167 463	203 619	214 818	167 463	203 619	214 818												
Total: Ngaka Modiri Molema Municipalities	289 867	352 449	371 834	289 867	352 449	371 834							10 000	15 000	2 000	10 000	15 000	2 000
MW 30 NAV	0000	14 601	15 404	000 21	109 71	15.404												
	13 800	100 +1	105.51	12 800	100 41	105.61												
NW394	26 975	32 799	34 603	26 975	32 799	34 603							28 000	20 000	15 000	28 000	20 000	15 000
	13179	16 025	16 906	13 179	16 025	16 906												
	24613	29 926	31 572	24 613	29 926	31 572												
C DC39 Dr Ruth Segomotsi Mompati District Municipality	91 585	111 358	117 483	91 585	111 358	117 483												
Total: Dr Ruth Segomotsi Mompati Municipalities	182 160	221 487	233 669	182 160	221 487	233 669							28 000	20 000	15 000	28 000	20 000	15 000
B NW401 Ventersdorp	20 046	24 374	25 715	20 046	24 374	25 715												
B NW402 Tlokwe	34 191	41 573	43 859	34 191	41 573	43 859							4 000	000 6	15 000	4 000	000 6	15 000
	100 609	122 332	129 060	100 609	122 332	129 060							2 000	10 000	20 000	2 000	10 000	20 000
	28416	34 551	36 451	28 416	34 551	36 451												
C DC40 Dr Kenneth Kaunda District Municipality																		
Total: Dr Kenneth Kaunda Municipalities	183 262	222 829	235 084	183 262	222 829	235 084							000 6	19 000	35 000	0006	19 000	35 000
Total: North West Municipalities	1 190 545	1 447 585	1 527 202	1 190 545	1 447 585	1 527 202							51 000	64 000	000 29	21 000	64 000	92 000
		-	1	-				-	-			1		-	-	-	-	1

ANNEXURE WS
INFRASTRUCTURE GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULES 4 AND 6) 1 OF 4

National Property   Nati			Muni	Municipal Infrastructure Grant	ture Grant				Urban	Urban Settlements Development Grant	velopment Gr	ant		Neig	bourhood De	velopment Pa	Neighbourhood Development Partnership Grant (Capital Grant)	t (Capital Gra	Qui
WEATHER PARTIES AND STATES AND S		Natio	nal Financial Ye	ar	Munici	val Financial	Year	Nations	d Financial Ye	ar	Munic	ipal Financial	Year	Nation	nal Financial Y	rear	Munic	Municipal Financial Year	Year
With Exercise Transmission of Control C		2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R000)	2013/14 (R000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
Cut of Cape Transment   1281   18.02   19.02   18.02   19.02	WESTERN CAPE																		
Legacine Conferential Conferent								824 030	972 615	1 067 485	824 030	972 615	1 067 485	77 500	84 226	80 000	77 500	84 226	80 000
Condendation         1234         15.59         16.47         9.78         16.47         15.89         16.47         17.84         15.89         16.47         15.89         16.47         15.89         16.47         15.89         16.47         15.89         16.47         15.89         16.47         15.89         16.47         15.89         16.47         15.89         16.47         15.89         16.47         15.89         16.47         15.89         16.47         15.89         16.47         15.89         16.47         15.89         16.47	WC011	15 321	18 628	19 652	15 321	18 628	19 652							3 000	1 000		3 000	1 000	
State   Stat	WC012	12814	15 580	16 437	12 814	15 580	16 437												
Statistical Bigs Bigs Statistical Bigs Statistical Bigs Statistical Bigs Bigs Statistical Bigs Bigs Bigs Bigs Bigs Bigs Bigs Bigs	WC013	9 727	11 827	12 477	9 727	11 827	12 477												
Secretary Control Multicipality   11-94   14-559   15-59   11-94   14-559   15-59   11-94   14-559   15-59   11-94   14-559   15-59   11-94   14-59   15-59   11-94   14-59   15-59   11-94   14-59   15-59   11-94   14-59   11-94   14-59   11-94   14-59   11-94   14-59   11-94   14-59   11-94   14-59   11-94   14-59   11-94   14-59   11-94   14-59   11-94   14-59   11-94   14-59   11-94   14-59   11-94   14-59   11-94   14-59   11-94   14-59   11-94   14-59	WC014	15164	18 437	19 451	15 164	18 437	19 451												
Numerical Properties   1829   27   104   28   28   28   28   28   28   28   2	WC015	11974	14 559	15 359	11 974	14 559	15 359												
Particular   Par	Total: West Coast Municipalities	64 999	79 030	83 376	64 999	79 030	83 376		T				İ	3 000	1 000		3 000	1 000	
Machine House, part of the control																			
Definition Definition (Controlled Controlled	WC022	18 2 2 9	22 164	23 383	18 229	22 164	23 383												
Stationary   1, 24, 24, 24, 24, 24, 24, 24, 24, 24, 24	WC023	28919	35 162	37 096	28 919	35 162	37 096												
Proceeding	WC024	20955	25 480	26 881	20 955	25 480	26 881												
Langetwight begins Municipality (1972) 19424 194	WC025	25 482	30 983	32 688	25 482	30 983	32 688							0000	00,0	0000		000	0000
1871   1872   1873   1874   1875		15149	18 4 19	19 432	15 149	18 419	19 432							7 000	3 000	7,000	7,000	3 600	7 000
Theywarenekhod Coverstand State	Total: Cape Winelands Municipalities	108 735	132 209	139 480		132 209	139 480							2 000	3 600	2 000	2 000	3 600	2 000
Coveration of 377 16 587 17 92 18 97	B WC031 Theewaterskloof	21577	26 235	27 678	21 577	26 235	27 678												
Currey Againsts         8 479         10 308         10 675         8 179         10 308         10 675         11 151         11 765         10 308         10 675         11 151         11 765         11 151         11 765         11 176	WC032	13 971	16 987	17 971	13 971	16 987	17 92 1							4 200			4200		
Sweldendisch Endige in Swelding Swelding Swelding Swelding Swelding Swelding Swelding Swelding Swelding State (468)         68 239         61 53 198         64 681         68 239         65 239	WC033	8 479	10 308	10 875	8 479	10 308	10 875										3		
Overherg District Municipality  S. 51 198	WC034	9 172	11 151	11 765	9 172	11 151	11 765												
rg Municipalities         53 198         64 681         68 29         \$3.198         64 681         68 29         \$3.198         64 681         68 29         \$3.198         64 681         68 29         \$3.198         64 681         68 29         \$3.198         64 681         68 29         \$3.108         \$3	C DC3 Overberg District Municipality																		
Kannal and Hescopan         9 494         11 543         12 178         9 494         11 543         12 178         9 494         11 543         12 178         9 494         11 543         12 178         9 494         11 543         12 178         12 1835         12 183	Total: Overberg Municipalities	53 198	64 681	68 239	53 198	64 681	68 239							4 200			4 200		
Hessequa Hessequa Hessequa Hessequa Hessequa Hessequa Horszan Hessequa Horszan Hessequa Horszan Horsza	WC041	9 494	11 543	12 178	9 494	11 543	12 178												
Mosel Bay         14143         17196         18142         17196         18142         17196         18142         17196         18142         17196         18142         17196         18142         17196         1811         1417         46575         3639         44147         46575         3639         44147         46575         3639         44147         46575         3639         44147         46575         3639         44147         46575         3639         44147         46575         3639         14431         17546         18511         14431         17546         18511         14431         17546         18511         14431         17546         18511         14431         17546         15497         1569         1569         18511         14431         17546         1569	WC042	10 551	12 829	13 535	10 551	12 829	13 535												
George         44 147         14 57 5 36 30         44 147         46 57 5 36 30         44 147         46 57 5 36 30         44 147         46 57 5 36 30         44 147         17 346         18 51 1 1 4 4 4 1 1 1 1 1 1 1 1 1 1 1 1 1	WC043	14 143	17 196	18 142	14 143	17 196	18 142												
Graph Condition         14431         17 546         18 511         17 546         18 511         17 546         18 511         17 546         18 511         17 546         18 511         17 546         18 511         17 546         18 511         17 546         18 511         17 549         15 497         18 500         15 497         18 500         15 497         18 500 <th< td=""><td>WC044</td><td>36309</td><td>44 147</td><td>46 575</td><td>36 309</td><td>44 147</td><td>46 575</td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td><td></td></th<>	WC044	36309	44 147	46 575	36 309	44 147	46 575												
Righten         Holosope         19281         14 689         15 497         15 407         15 600         16 600         16 600         16 600         16 60	WC045	14 431	17 546	18 511	14 431	17 546	18 511							000 6	9 825		0006	9 825	
Second Particle   1933   24.256   1933   24.256   25.569   1933   24.256   25.569		12 081	14 689	15 497	12 081	14 689	15 497							14 000	15 000	11 000		15 000	11 000
Laingsburg         6 735         8 214         8 665         6 735         8 214         8 665         6 735         8 214         8 665         6 735         8 214         8 665         6 735         8 214         8 665         6 735         8 214         8 665         6 735         8 214         8 665         6 735         8 214         8 665         6 735         8 214         8 665         6 735         8 214         8 665         6 735         8 214         8 665         6 735         8 214         8 665         6 735         8 214         8 665         6 735         8 214         8 665         6 735         8 214         8 665         6 735         8 214         8 665         6 735         8 214         8 665         6 735         8 214         8 665         6 735         8 214         8 665         8 675         8 214         8 665         8 675         8 214         8 665         8 675         8 214         8 665         8 675         8 214         8 214         8 675         8 214         8 214         8 675         8 214         8 214         8 214         8 214         8 214         8 214         8 214         8 214         8 214         8 214         8 214         8 214         8 214         8 214		19933	24.236	75 569	19 953	24.236	75 569							2 000	8 000	20 000	000 c	8 000	20 000
Laingsbrug         6 755         8 214         8 665         6 755         8 214         8 665         705         7 055         8 778         9 050         7 055         8 778         9 050         7 055         8 778         9 050         7 055         8 778         9 050         7 055         8 778         9 050         7 055         8 778         9 050         7 055         8 778         9 050         7 050	Total: Eden Municipalities	116 943	142 187	150 007	116 943	142 187	150 007							28 000	32 825	31 000	28 000	32 825	31 000
Frame Shugh Frame Albert Four Control West Control Municipalities 375 38 456 386 481 487 375 388 456 386 481 487 375 388 456 386 481 487 375 388 456 386 481 487 375 388 456 386 481 487 375 388 456 386 481 487 375 388 456 386 481 487 375 375 375 375 375 375 375 375 375 37	130071	991	1100	2000	331.7	6100	377 0												
Beaufort West         17 673         21 487         22 669         17 673         21 487         26 69         76 69         76 69         700         <	WC052	7.055	120	050.0	7.055	17.0	050 6												
Central Kamo District Municipality  Karon Municipalities  1. Cape Municipaliti	WC053	17 673	21 487	22 669	17 673	21 487	22 669							2 800	700		2 800	200	
Karvo Municipalities         31 484         38 279         40 384         38 279         40 384         38 279         40 384         70 40 384         70 60         700	DCS		i			i	l							ı					
1 Cape Municipalities 375 358 456 386 481 487 874 036 972 615 1 067 485 824 030 972 615 1 067 485 117 500 122 351	Total: Central Karoo Municipalities	31 484	38 279	40 384		38 279	40 384							2 800	200		2 800	200	
Cape Municipalities 375 358 456 386 481 487 824 030 972 615 1 067 485 824 030 972 615 1 067 485 117 500 122 351																			
	Total: Western Cape Municipalities	375 358	456 386	481 487	375 358	456 386	481 487	824 030	972 615	1 067 485	824 030	972 615	1 067 485	117 500	122 351	113 000	117 500	122 351	113 000
	Unallocated:																		
11 443 505 13 914 131 14 679 408 6 266 998 7 409 512 8 126 816 6 266 998 7 409 512 8 126 806 750 000 800 000	National Total	11 443 505	13 914 131	14 679 408	11 443 505	13 914 131	14 679 408	6 266 998	7 409 512	8 126 806	6 266 998	7 409 512	8 126 806	750 000	800 000	800 000	750 000	800 000	800 000

ANNEXURE W5
INFRASTRUCTURE GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULES 4 AND 6) 2 OF 4

	77.8	Public Tra	Public Transport Infrastructure and	ucture and System	Systems Grant			Rural Transport Services and Infrastructure Grant	t Services and	Infrastructur	e Grant	$\parallel$	17.10	Municipal Drought Relief Grant	ught Relief Gran	t .	
	2011/12	National Financial Year	2013/14	Mumci	Mumcipal Financial Year	2013/14	National	2012/13   26	13/14 2/	nicipal Financi.	L	2013/14	National Fina	ancial Year	Mumci	Mumcipal Financial Year	7013/14
Category Municipality	(R'000)	(R'000)	(R'000)	2011/12 (R'000)	(R'000)	(R'000)	(R'000) (	(R'000) (R	(R'000) (A	(R'000) (R'	_	$\dashv$	(R'000) (R'000)	(R'000)	(R'000)	(R'000)	(R'000)
EASTERN CAPE																	
A BUF Buffalo City A NMA Nelson Mandela Bay	180 000 340 000	184 781 400 000	320 000	180 000	184 781 400 000	320 000						7	450 000		450 000		
B         EC101         Camdeboo           B         EC102         Blue Came Route           B         EC103         Ikwezi           B         EC104         Makana           B         EC105         Modambee           B         EC105         Sundays West Yalley																	
B   EC107 Baviaans   B   EC108 Kouga   B   EC109 Kou-kamma   C   DC10   Cacada District Municipality																	
Total: Cacadu Municipalities																	
B EC121 Mbhashe B EC122 Mnquma B EC122 Great Kei B EC124 Amahlathi B EC126 Ngqushwa B EC127 Nkonkobe B EC127 Nkonkobe C DC12 Amatholo Dixtrict Municipality								9	698		982	698					
otal: Amathole							1 688	1 780	1 869	1 688	1 780	1 869					
B   EC131   Inxuba Yethemba   B   EC132   Tsolvana   B   EC133   Inxuba Yethemba   B   EC134   Lukhanji   B   EC134   Lukhanji   B   EC135   Emalaheni   B   EC136   Emalaheni   B   EC137   Engodo   B   EC137   Engodo   C   DC13   Chris Han District Municipality   DC13   Chris Han District Municipality							889	2%	998		98	698					
otal: Chris Ha							1 688	1 780	1 869	1 688	1 780	1 869					
B EC141 Elundini B EC142 Sengu B EC143 Maletswai B EC144 Garde C DC14 URhahhmba District Municipality							1 688	1 780	1 869	1 688	1 780	1 869					
Total: Ukhahlamba Municipalities							1 688	1 780	1 869	1 688	1 780	1 869					
B EC153 Ngquza Hill B EC154 Port St Johns B EC155 Nyandeni B EC156 Myandeni B EC157 King Subata Dalindyebo C DC15 OR Tambo District Municipality							1 688	1 780	1 869	1 688	1 780	1 869					
Total: O.R.Tambo Municipalities							1 688	1 780	1869	1 688	1 780	1 869					
B EC441 Mataticle   B EC442 Umzimvubu   B EC443 Mbizana   B EC152 Nabanshulu   C DC44 Alfred Na Obstrict Municipality   Total: Alfred Na Municipalities							1 688	1 780 1 780	1869	1 688	1 780 1 780	1 869 1 869					
Total: Eastern Cape Municipalities	520 000	584 781	1 020 000	520 000	584 781	1 020 000	8 438	8 900	9 345	8 438	8 900	9 345 4	450 000		450 000		

ANNEXURE WS
INFRASTRUCTURE GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULES 4 AND 6) 2 OF 4

		Public Tra	msport Infrastr	Public Transport Infrastructure and Systems Grant	18 Grant			Rural Trans	Rural Transport Services and Infrastructure Grant	and Infrastruc	ture Grant			Mu	unicipal Droug	Municipal Drought Relief Grant	#	
	Nati	National Financial Year	Year	Munici	ipal Financial Year	ear	Natio	National Financial Year	Year N	Municipal Financial Year	ıncial Year		Nation	National Financial Year	Year	Munic	Municipal Financial Year	Year
Category Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
FREE STATE																		
A MAN Mangaung	15 000	15 000	15 000	15 000	15 000	15 000												
B FS161 Letsemeng B FS167 Konstone																		
FS163 FS164																		
C DC16 Xhariep District Municipality																		
Total: Xhariep Municipalities																		
FS181																		
B FS182 Tokologo																_		
C DC18 Lejweleputswa District Municipality Total: Leiweleputswa Municipalities	 						1	$\dagger$		$\dagger$	1			T				
Coverage of the control of the contr																		
FS191																		
B FS192 Dihlabeng																		
																_		
FS195																_		
B FS196 Mantsopa																		
Ž																		
B FS201 Modhaka																		
FS203																		
B FS204 Metsimaholo																_		
DC20																_		
l'a																		
Total: Free State Municipalities	15 000	15 000	15 000	15 000	15 000	15 000												

ANNEXURE WS
INFRASTRUCTURE GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULES 4 AND 6) 2 OF 4

			Public Trans	sport Infrastru	Public Transport Infrastructure and Systems Grant	ns Grant			Rural Trans	Rural Transport Services and Infrastructure Grant	and Infrastruc	ture Grant			Mu.	nicipal Droug.	Municipal Drought Relief Grant	#	
		Nation	National Financial Year	ar	Munici	Municipal Financial Year	ear ear	Natio	National Financial Year		Municipal Financial Year	ncial Year		Nation	National Financial Year	Year	Munici	Municipal Financial Year	Year
Costonomic	Municipality	2011/12	2012/13	2013/14	(000'9) CI/IIOC	2012/13	2013/14	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14
Category	Category Municipanty	(R'000)	(R'000)	(R'000)	2011/12 (N 000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)
0.000																			
GAUTENG																			
A EKU	Ekurhuleni	20 000	20 000	20 000	20 000	20 000	20 000												
A JHB	City of Johannesburg	1 700 000	1 530 000	1 530 000	1 700 000	1.	1 530 000												
A TSH	City of Tshwane	200 000	800 000	850 000	200 000	800 000	850 000												
- E	D(6.10)																		
B GT421	Midwell																		
D CT422																			
C DC42																			
Total: Sedibeng	Total: Sedibeng Municipalities																		
B GT481	Mogale City																		
B GT482	Randfontein																		
B GT483	Westonaria																		
B GT484	Merafong City																		
C DC48	West Rand District Municipality																		
Total: West Ran	Total: West Rand Municipalities																		
Total: Gauteng	Total: Gauteng Municipalities	1 920 000	2 350 000	2 400 000	1 920 000	2 350 000	2 400 000												

ANNEXURE WS
INFRASTRUCTURE GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULES 4 AND 6) 2 OF 4

Maintenant   Mai			Public Tr	ansport Infrast	Public Transport Infrastructure and Systems Grant	ms Grant			Rural Trans	ort Services a	Rural Transport Services and Infrastructure Grant	ire Grant			Mu	nicipal Droug	Municipal Drought Relief Grant	Ħ	
Section   Sect		Nati	onal Financial	Year			Year	Natio	nal Financial Y		unicipal Finan	cial Year		Nation.	al Financial 3	Year	Munic	ipal Financial	Year
215 COO   1100 COO   1106 GO4   235 COO   1100 COO   1106 GO4   1100 COO   1106 GO4   1100 COO   1106 GO4   1100 COO   1106 GO4   1100 COO   1106 GO4   1100 COO		2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)		2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)		2012/13 R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
inicipality  inici	KWAZULU:NATAL																		
Inscipality  65 000 85 000 65 000 85 000 1 688 1 780 1	ЕТН	205 000	100 000				108 604												
micipality  65 000 85 000 25 000 85 000 25 000 85 000 1 688 1 780 1 869 1 688 1 780  District Municipality  64 000 85 000 25 000 85 000 25 000 1 688 1 780 1 869 1 688 1 780  Matter Municipality  65 000 85 000 25 000 85 000 1 688 1 780 1 869 1 688 1 780  Matter Municipality  65 000 85 000 25 000 85 000 1 688 1 780 1 869 1 688 1 780  Matter Municipality  65 000 85 000 1 688 1 780 1 688 1 780  Matter Municipality  66 000 85 000 1 688 1 780 1 688 1 780  Matter Municipality  66 000 1 688 1 780 1 688 1 780  Matter Municipality  66 000 1 688 1 780 1 688 1 780  Matter Municipality  66 000 1 688 1 780 1 688 1 780  Matter Municipality  66 000 1 688 1 780 1 688 1 780  Matter Municipality  66 000 1 688 1 780 1 688 1 780  Matter Municipality  67 000 1 688 1 780 1 688 1 780  Matter Municipality  68 000 1 688 1 780 1 688 1 780  Matter Municipality  69 00 1 688 1 780  Matter Municipality  60 000 1 688 1 780  Matter Municipality  60 000 1 688 1 780  Matter Municipality  60 000 1 688 1 780  Matter Municipality  60 000 1 688 1 780  Matter Municipality  60 000 1 688 1 780  Matter Municipality  60 000 1 688 1 780  Matter Municipality  60 000 1 688 1 780  Matter Municipality  60 000 1 688 1 780  Matter Municipality  60 000 1 688 1 780  Matter Municipality  60 000 1 688 1 780  Matter Municipality  60 000 1 688 1 780  Matter Municipality  60 000 1 688 1 780  Matter Municipality  60 000 1 688 1 780  Matter Municipality  Matter Municipalit	LTCNZA																		
District Municipality  65 000 85 000 25 000 85 000 25 000 1 688 1 780 1 689 1 780  District Municipality  65 000 85 000 25 000 25 000 25 000 1 688 1 780 1 689 1 780  District Municipality  10 088 1 780 1 689 1 780  10 088 1 780 1 688 1 780  10 088 1 780 1 688 1 780  10 088 1 780 1 688 1 780  10 088 1 780 1 688 1 780  10 088 1 780 1 688 1 780  10 088 1 780 1 688 1 780  10 088 1 780 1 688 1 780  10 088 1 780 1 688 1 780  10 088 1 780 1 688 1 780  10 088 1 780 1 688 1 780  10 088 1 780 1 688 1 780  10 088 1 780 1 688 1 780  10 088 1 780 1 688 1 780	KZN212																		
Integrality  65 000  85 000  25 000  65 000  85 000  25 000  85 000  1 688  1 780  1 780  1 688  1 780  1 780  1 780  1 780  1 7	KZN213																		
micpality  65 000 85 000 65 000 85 000 1 688 1 780 1 688 1 780  District Municpality  65 000 85 000 25 000 65 000 1 688 1 780  altities  District Municpality  65 000 85 000 25 000 65 000 1 688 1 780  altities  Absumith  Thumicpality  Thumic	KZN214																		
Interpolity    1688   1780   1889   1780   1	KZN215																		
District Municipality  C65 000  S8 000	DC21							1 688	1 780	1 869	1 688	1 780	1 869						
District Municipality  65 000  85 000  85 000  85 000  85 000  85 000  85 000  1 688  1 780  1 689  1 780  1 689  1 780  1 689  1 780  1 689  1 780  1 689  1 780  1 689  1 780  1 689  1 780  1 689  1 780  1 689  1 780  1 689  1 780  1 689  1 780  1 688  1 780	Total: Ugu Municipalities							1 688	1 780	1 869	1 688	1 780	1869						
District Municipality   GS 000   SS 000   SS 000   SS 000   SS 000   SS 000   SS 000   SS 000   SS 000   SS 000   SS 000   I GSS   I 780   I 780   I GSS   I 780   I																			
District Municipality  65 000 85 000 25 000 85 000 25 000 1 688 1 780 1 869 1 688 1 780  Alysmith  Althreicpality  65 000 85 000 25 000 85 000 1 688 1 780 1 869 1 688 1 780  Althreicpality  65 000 85 000 25 000 1 688 1 780 1 869 1 688 1 780  Althreicpality  65 000 85 000 25 000 1 688 1 780 1 869 1 688 1 780  65 000 85 000 25 000 1 688 1 780 1 869 1 688 1 780  65 000 85 000 1 688 1 780 1 688 1 780  65 000 85 000 1 688 1 780 1 688 1 780  65 000 85 000 1 688 1 780  65 000 85 000 1 688 1 780  65 000 85 000 1 688 1 780  65 000 85 000 1 688 1 780  65 000 85 000 1 688 1 780  65 000 1 68	KZN221																		
abitative Municipality  abitative Admicipality  cf. 6000  sf. 6000	KZN222																		
District Municipality  Librarie Municipality	KZN223																		
District Municipality  65 000  85 000  85 000  10 88 000  10 88 000  10 88 000  10 88 01780  10 88 1780	KZN225	000 59	85,000																
abistrict Municipality  abilities  abistrict Municipality  abistrict Municipal	KZN226	8	000 00																
District Municipality  cd. smith	KZN227																		
Aysmith dispatify 65 000 85 000 65 000 85 000 1 688 1 780 1 680 1	C DC22 uMgungundlovu District Municipality							1 688	1 780	1 869	1 688	1 780	1 869						
thysmith  The Municipality  Th	Total: uMgungundlovu Municipalities	92 000	85 000				25 000	1 688	1 780	1 869	1 688	1 780	1869						
t Municipality  Tric Municipalit	VZND33																		
th Municipality 1 688 1 780 1 869 1 869 1	KZN232																		
trict Municipality 1 688 1 780	KZN234																		
trict Municipality trict Municip	KZN235																		
1 (88)   1	KZN236																		
trict Municipality 1688 1780 1889 1780 1880 1780 1880 1780 1880 1780 1880 1780 1880 1780 1880 1780 1880 1780 1880 1780 1880 1780 1880 1780 1880 18	C DC23 Uthukela District Municipality							1 688	1 780	1 869	1 688	1 780	1 869						
trict Municipality  text Municip	Total:Uthukela Municipalities							1 688	1 780	1 869	1 688	1 780	1 869						
trict Municipality 1688 1780 1869 1688 1869 1869 1869 1869 1869 1869	KZN241																		
rict Municipality 1688 1780 1869 1688 1780 1869 1688 1780 1869 1688 1780 1869 1688 1780 1869 1688 1780 1869 1688 1780 1869 1688 1780 1869 1688 1780 1869 1688 1780 1869 1688 1780 1869 1688 1780 1869 1688 1780 1869 1688 1780 1869 1688 1780 1869 1688 1780 1868 1780 1869 1688 1860 1860 1860 1860 1860 1860 1860	KZN242																		
trict Municipality 1688 1780 1869 1688 1780 1869 1688 1780 1869 1688 1780 1860 1580 1580 1580 1580 1580 1580 1580 158	KZN244																		
trict Municipality  start Municipality  text Municipality  text Municipality  text Municipality  1 688 1 780 1 688 1 780  1 688 1 780  1 688 1 780  1 688 1 780  1 688 1 780  1 688 1 780	KZN245																		
tes	C DC24 Umzinyathi District Municipality							1 688	1 780	1 869	1 688	1 780	1 869						
t Municipality 1688 1780 1869 1688 1780 1869 1688 1780	Total: Umzinyathi Municipalities							1 688	1 780	1 869	1 688	1 780	1 869						
:t Municipality 1688 1780 1869 1688 1780 1869 1688 1780																			
t Municipality 1869 1869 1688 1780 1869 1688 1860 1860 1860 1860 1860 1860 1860	KZN252																		
t Municipality 1869 1869 1869 1889 1780 1869 1889 1780 1869 1889 1780	KZN253																		
1780 1869 1688 1780	C DC25 Amajuba District Municipality							1 688	1 780	1 869	1 688	1 780	1 869						
000 4	Total: Amajuba Municipalities							1 688	1 780	1 869	1 688	1 780	1 869		_				

ANNEXURE W5
INFRASTRUCTURE GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULES 4 AND 6) 2 OF 4

		Public Tr	ansport Infrastr	Public Transport Infrastructure and Systems Grant	ms Grant			Rural Transi	Rural Transport Services and Infrastructure Grant	nd Infrastruct	ure Grant			Muni	Municipal Drought Relief Grant	t Relief Gran		
	Nati	National Financial Year	Year	Munic	ipal Financial Year	/ear	Nation	National Financial Year	ear M	Municipal Financial Year	icial Year		National	National Financial Year	ear	Municip	d Financial	ear
Category Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 2 (R'000) (	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
d thunds																		
KZN263																		
KZN265																		
C DC26 Zululand District Municipality							1 688	1 780	1 869	1 688	1 780	1 869						
Total: Zululand Municipalities							1 688	1 780	1 869	1 688	1 780	1 869						
D VZNI71 IImhlahumilianan																		
VZNOTO																		
KZNZ73																		
C/ZNZ/3							9	000	0,00	9	000	9						
C DC2/ Umknanyakude District Municipality							1 088	1 /80	1 869	1 088	1 /80	1 809						
Total: Umkhanyakude Municipalities							1 688	1 780	1 869	1 688	1 780	1 869						
POOLYGE																		
KZN282																		
KZN283																		
KZN284																		
KZN285																		
B KZN286 Nkandla																		
C DC28 uThungulu District Municipality							1 688	1 780	1 869	1 688	1 780	1 869						
12							1 688	1 780	1869	1 688	1 780	1 869						
B KZN291 Mandeni																		
KZN292																		
221777																		
b NZN294 Maphumulo							000	1	1000	007	100	0,01						
C DC29 ILEMDe District Municipality							000 1	00/1	600 1	000 1	00/1	1 009						
Total: iLembe Municipalities							1 688	1 780	1 869	1 688	1 780	1 869						
D VZNIA21 Isomo																		
D VZN432 V S																		
KZN432																		
KZN434																		
C DC43 Sisonke District Municipality							1 688	1 780	1 869	1 688	1 780	1 869						
Total: Sisonke Municipalities							1 688	1 780	1 869	1 688	1 780	1 869						
Total Vene Zule Notel Maniples	000 020	105 000	133 604	000 020	105 000	133 604	16 076	17 001	10,500	16 076	17 001	10.500	1	t	t	t	t	T
10tal: NWaZuliu-matai mumicipanites	200 0/7				1000 001	122 004	10/001	1/ 001	10 070	0/001	1/ 001	100001		_	_		_	_

ANNEXURE WS
INFRASTRUCTURE GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULES 4 AND 6) 2 OF 4

		Public Transi	ort Infrastruc	Public Transport Infrastructure and Systems Grant	is Grant			Rural Trans	port Services	Rural Transport Services and Infrastructure Grant	ire Grant			Municipal	Municipal Drought Relief Grant	rant	
	Nationa	National Financial Year	_	Munici	Municipal Financial Year	/ear	Natio	National Financial Year	ear N	Municipal Financial Year	П	Н	National F	National Financial Year	Mu	Municipal Financial Year	al Year
Category Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 2 (R'000) 2	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 : (R'000)	2013/14 (R'000)	2011/12 20 (R'000) (R	2012/13 2013/14 (R'000) (R'000)	14 2011/12 ) (R'000)	2012/13 (R'000)	2013/14 (R'000)
LIMPOPO																	
B LIM331 Greater Giyani																	
LIM333																	
LIM334																	
B LIM335 Maruleng							000	7	0,0	9	000	9,0					
C DC35 Mopan District Municipanty							1 000	08/1	1 809	1 000	08/1	1 809					
Total: Mopani Municipalities			1				1 688	1.780	1869	1 688	1.780	1 869					
D 17M241 Musica																	
LTM343																	
1 TM344																	
DC34							1 688	1 780	1 869	1 688	1 780	1 869					
Total: Vhembe Municipalities							1 688	1 780	1 869	1 688	1 780	1 869					
LIM352																	
LIM353	1	000	0	0	0	0											
LIM354	55 347	30 000	25 000	55 347	30 000	25 000											
B LIM355 Lepelle-Nkumpi							007	7	0,01	9	000	0,01					
Total: Carricorn Municipalities	55 347	30 000	25,000	55 247	30 000	000 50	1 689	1 780	1 860	1 688	1 780	1 860					
Total: Capricorn Municipannes	/#C CC	20 000	000 C7	140.00	20 000	000 67	000 1	1 /00	1 00%	1 000	1 /00	1 003					
D 110261 Thebesier																	
LIM361																	
LIM364																	
COSMIT																	
DC36																	
Ę																	
LIM471																	
LIM473																	
LIM474																	
							000	0			000	9					
C DC4/ Greater Sekhukhune District Municipality							1 688	08/ 1	1 869	1 688	08/ 1	1 869					
Total: Greater Sekhukhune Municipalities							1 688	1 780	1869	1 688	1 780	1869					
Total: Limnono Municipalities	55.347	30 000	25 000	55.347	30 000	25 000	052.9	7 120	7 476	6750	7 120	7 476					

ANNEXURE W5
INFRASTRUCTURE GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULES 4 AND 6) 2 OF 4

		Public Tra	nsport Infrastr	Public Transport Infrastructure and Systems Grant	ns Grant			Rural Trans	Rural Transport Services and Infrastructure Grant	nd Infrastru	ture Grant			Mı	Municipal Drought Relief Grant	ght Relief Gra	Ħ	
	Natio	National Financial Year	ear	Munici	Municipal Financial Year	Year	Nation	National Financial Year		Municipal Financial Year	ıncial Year		Natio	National Financial Year	Year	Munic	Municipal Financial Year	Year
Category Municipality	2011/12	2012/13		2011/12 (R'000)	2012/13	2013/14	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14
	(8,000)	(R000)	(R.000)		(R.000)	(R.000)	(R.000)	(R.000)	(8.000)	(8.000)	(R.000)	(R'000)	(R.000)	(R.000)	(R000)	(R.000)	(R.000)	(R000)
MPUMALANGA																		
100001																		
B MP301 Albert Luthuli																		
MP302																		
MP305																		
MP306																		
C DC30 Gert Sibande District Municipality																		
Total: Gert Sibande Municipalities																		
MP311																		
MP312																		
MP313																		
MP314																		
MP315																		
C DC31 Nkangala District Municipality																		
Total: Nkangala Municipalities																		
B MP321 Thaba Chweii																		
MP322	45 000	20 000	20 000	45 000	20 000	20 000												
MP323																		
B MP325 Bushbuckridge																		
C DC32 Ehlanzeni District Municipality																		
Total: Ehlanzeni Municipalities	45 000	20 000	20 000	45 000	20 000	20 000												

ANNEXURE W5
INFRASTRUCTURE GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULES 4 AND 6) 2 OF 4

		Public Tra	nsport Infrast	Public Transport Infrastructure and Systems Grant	ms Grant			Rural Trans	Rural Transport Services and Infrastructure Grant	and Infrastru	cture Grant			M	unicipal Droug	Municipal Drought Relief Grant	ant	
	Nati	National Financial Year	/ear	Munic	Municipal Financial Year	Year	Nation	National Financial Year	Year N	Municipal Financial Year	ancial Year		Natio	National Financial Year	Year	Munic	Municipal Financial Year	Year
Category Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
NORTHERN CAPE																		
NC061																		
B NC064 Kamiesberg																		
NC065																		
B NC066 Karoo Hoogland																		
DC6																		
N.S																		
B NC071 Ubuntu																		
NC072																		
NC074																		
NC075																		
B NC0/6 Thembelinle																		
NC078																		
DC7																		
\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\\																		
NICOGOI MISS.																		
NC083																		
NC084																		
NC085																		
B NC086 Kgatelopele C DC8 Siyanda District Municipality																		
Total: Siyanda Municipalities																		
B NC091 Sol Plaatie																		
NC093																		
B NC094 Phokwane																		
1																		
Total: Frances Daard Municipannes								İ										
NC451																		
NC452																		
B NC453 Gamagara C DC45 Iohn Taolo Gaersewe District Municipality																		
otal: John Tao																		
Total: Northern Cane Municipalities																		

ANNEXURE W5
INFRASTRUCTURE GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULES 4 AND 6) 2 OF 4

		Public Tra	usport Infrastr	Public Transport Infrastructure and Systems Grant	s Grant			Rural Trans	Rural Transport Services and Infrastructure Grant	nd Infrastruc	ture Grant			Munic	cipal Drough	Municipal Drought Relief Grant	#	
	Nation	National Financial Year	ear	Municip	Municipal Financial Year	Year .	Natio	National Financial Year		Fina	ncial Year		National	ial Y		Munici	Municipal Financial Year	Year
Category Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 2 (R'000) (	2011/12 2 (R'000) (	2012/13 2 (R'000) (	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
NORTH WEST																		
B NW371 Moretee B NW372 Madibeng B NW373 Rustenburg B NW374 Kgetlengriver B NW375 Moses Kotane	178 000	235 000	350 000	178 000	235 000	350 000												
	178 000	235 000	350 000	178 000	235 000	350 000												
B NW381 Ratiou B NW382 Tswaing B NW383 Maffkeng B NW384 Ditsobotia B NW384 Randoshee Molioa C DC'38 Noaka Modri Molema District Municipality								130	69%	89	780	698						
12							1 688	1 780	1 869	1 688	1 780	1 869						
B NW392 Naledi B NW393 Mamusa B NW394 Greater Taung B NW395 Lekwa-Teomane B NW397 NW397 C DC39 Dr Ruth Segonotsi Mompati District Municipality							1 688	1 780	1 869	1 688	1 780	1 869						
Total: Dr Ruth Segomotsi Mompati Municipalities  B NW401 Ventersdorp  B NW402 Tlokwe B NW403 Ciry of Antotean B NW404 Ciry of Antotean B NW404 Momenses Hils							1 688	1 780	1 869	1 688	1 780	1 869						
Total: North West Municipalities	178 000	235 000	350 000	178 000	235 000	350 000	3375	3 560	3 738	3 375	3 560	3 738	+	+	$\parallel$			

ANNEXURE W5
INFRASTRUCTURE GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULES 4 AND 6) 2 OF 4

			Public Tra	asport Infrastr	Public Transport Infrastructure and Systems Grant	ms Grant			Rural Transport Services and Infrastructure Grant	ort Services au	nd Infrastruc	ture Grant			Mu	inicipal Droug	Municipal Drought Relief Grant	ıţ	
		Natio	National Financial Year	ear	Munic	Municipal Financial Year	ear.	Nation	ja Z		Fina	ncial Year		Natio	National Financial Year	Year	Munici	Municipal Financial Year	Year
Category	Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R 000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R 000)	2013/14 (R 000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
WESTERN CAPE	PE																		
A CPT	City of Cape Town	1 800 000	1 580 000	1 600 000	1 800 000	1 580 000	1 600 000												
B WC011	Matzikama Codortorro																		
WC013																			
B WC014	Saldanha Bay Swortland																		
DCI	Swattanu West Coast District Municipality																		
Total: West Cog	Total: West Coast Municipalities																		
WC022	Witzenberg																		
WC023	Drakenstein																		
B WC024	Stellenbosch Breede Vallev																		
WC026	Langeberg																		
C DC2	Cape Winelands District Municipality				Ī		1			1		1		1	1				
Total: Cape Wi	Total: Cape Winelands Municipalities	†	1	1		]	1	+	1	+	$\dagger$	$\dagger$	1	1	†				
WC031	Theewaterskloof																		
WC032	Overstrand																		
	Cape Agulhas																		
WC034	Swellendam Overherg District Municipality																		
Total: Overberg	Fotal: Overberg Municipalities		T																
B WC041	Kamaland																		
	Hessequa																		
WC043	Mossel Bay																		
	George																		
WC043	Oudtshoom Biou																		
WC048	Knysna																		
C DC4 Eden Distric	Eden District Municipality																		
							İ												
B WC051	Laingsburg																		
WC053	Beaufort West																		
C DC5	Central Karoo District Municipality												1						
Total: Central F	Fotal: Central Karoo Municipalities																		
Total: Western	Fotal: Western Cape Municipalities	1 800 000	1 580 000	1 600 000	1 800 000	1 580 000	1 600 000												
Unallocated:																			
National Total		4 803 347	4 999 781	5 563 604	4 803 347	4 999 781	5 563 604	35 439	37 382	39 250	35 439	37 382	39 250	450 000			450 000		

ANNEXURE W5
INFRASTRUCTURE GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULE 6) 3 OF 4

		Integrated National Electrification Programme (Municipal) Grant	al Electrification	Programme (M	(unicipal) Gran	, t		E	lectricity Der	nand Side Ma	Electricity Demand Side Management (Municipal) Grant	unicipal) Gra	nt
		National Financial Year	l Year	)	Munici	Municipal Financial Year	Year	Natio	National Financial Year	Year	Municipal Financial Year	nancial Year	1
Category	Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
EASTERN CAPE													
A BUF	Buffalo City Nelson Mandela Bav	21 000	20 000	20 000	21 000 25 000	20 000	20 000	4 000			4 000		
g 10101		1000	000		1000	000							
		29	2 000	468	29	2 000	468						
B EC104	Makana Ndlamba	0000	10.000	000 8	0000	10 000	000 8						
					1	8							
		2 000			2 000								
B EC109													
C DC10	Cacadu District Municipality												
Total: Cacadu Municipalities	cipalities	5 029	16 000	8 4 68	5 029	16 000	8 468						
		000,000	000 81	000	00,00	000 01	000						
B EC121	Monasne	20 680	18 000	30,000	70 680	15 000	15,000						
		2 000	000 61	000 51	2 000	000 CT	000 CT						
		1			l I								
B EC128													
Total: Amotholo Municipalities	Amathole District Municipality	33 680	33 000	000 23	33 680	33 000	000 27						
Total: Alliathore Mui	ncipanties	25 000	000 CC	000 00	25 000	000 CC	000 00						
B EC131	Inxuba Yethemba	1 000	10 000	16 000	1 000	10 000	16 000						
B EC132	Tsolwana	2 000		5 000	2 000		5 000						
B EC133	Inkwanca												
	Lukhanji	1 000			1 000								
B EC135	Intsika Yethu												
	Emalahleni	4 480	7 000	5 000	4 480	7 000	5 000						
		24 000	000 09	000 09	24 000	000 09	000 09						
B EC138	Sakhisizwe	4 000		000 9	4 000		000 9						
Tatal Chin Hand	Chilis Haili Disulct Municipanty	700 700	000	000 00	000	000	000 00						
Total: Chris Hani Municipannes	unicipanues	00 400	// NW	72 UUU	00 490	/ / 000	72 UUU						

ANNEXURE W5
INFRASTRUCTURE GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULE 6) 3 OF 4

	Integrated National Electrification Programme (Municipal) Grant	nal Electrification	n Programme (M	(unicipal) Gra	nt		I	Sectricity Der	mand Side Ma	Electricity Demand Side Management (Municipal) Grant	(unicipal) Gra	mt
	National Financial Year	al Year		Munic	Municipal Financial Year	Year	Natio	National Financial Year	Year	Municipal Fi	Municipal Financial Year	
Catemory Municipality	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14
	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)
B EC141 Elundini	1 000		15 000	1 000		15 000						
B EC142 Sengu												
B EC143 Maletswai	3 000	2 000	2 000	3 000	2 000	2 000						
	1 000	3 000	8 000	1 000	3 000	8 000						
C DC14 Ukhahlamba District Municipality												
Total: Ukhahlamba Municipalities	2 000	2 000	25 000	2 000	2 000	25 000						
B EC153 Ngquza Hill	000 6	20 000	20 000	000 6	20 000	20 000						
B EC154 Port St Johns												
B EC155 Nyandeni	10 000	20 000	40 000	10 000	20 000	40 000						
B EC156 Millontlo	1 000	30 000	85 000	1 000	30 000	85 000						
B EC157 King Sabata Dalindyebo	10 000	10 000	8 000	10 000	10 000	8 000						
C DC15 O R Tambo District Municipality												
Total: O.R.Tambo Municipalities	30 000	80 000	153 000	30 000	80 000	153 000						
B EC441 Matatiele			15 000			15 000						
B EC442 Umzimvubu			15 000			15 000						
B EC443 Mbizana	20 000	20 001	20 000	20 000	20 001	20 000						
B EC152 Ntabankulu	15 000	20 000	10 000	15 000	20 000	10 000						
C DC44 Alfred Nzo District Municipality												
Total: Alfred Nzo Municipalities	35 000	40 001	000 09	35 000	40 001	000 09						
Total: Eastern Cane Municipalities	220 189	291 001	443 468	220 189	291 001	443 468	24 000			24 000		

ANNEXURE W5
INFRASTRUCTURE GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULE 6) 3 OF 4

	Integrated National Electrification Programme (Municipal) Grant	nal Electrification	n Programme (A	Aunicipal) Gra	nt		E	lectricity Der	nand Side Ma	Electricity Demand Side Management (Municipal) Grant	unicipal) Gra	ıţ
	National Financial Year	ıl Year		Munic	Municipal Financial Year	Year	Natio	National Financial Year	Year	Municipal Financial Year	nancial Year	
Category Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
FREE STATE												
A MAN Mangaung	20 000	15 000	13 000	20 000	15 000	13 000	5 000			5 000		
B FS161 Letsemeng	594			594								
FS162	800	200	963	800	200	963						
FS163	627	200		627	200							
B FS164 Naledi C DC16 Xhariep District Municipality												
Ŀ	2 021	1 000	963	2 021	1 000	963						
FS181 FS182												
B FS183 Tswelopele B FS184 Matjhabeng	2 286	200	200	2 286	200	200						
B FS185 Nala C DC18 Leiwelenutswa District Municipality												
otal: Lejweleputswa M	2 286	200	500	2 286	200	200						
B FS191 Setsoto												
FS192												
FS194	19 600	20 000	20 000		20 000	20 000						
B FS195 Phumelela	2 640			2 640								
DC19												
Total: Thabo Mofutsanyana Municipalities	23 050	20 000	20 000	23 050	20 000	20 000						
B FS201 Moqhaka	3 3 60	2 000	4 000	3 360	2 000	4 000						
B FS203 Ngwathe	5 183	5 000	7 000		2 000	7 000						
B FS204 Metsimaholo	2 000	5 000	13 000		2 000	13 000						
	12 100	5 000	8 000	12 100	2 000	8 000						
C DC20 Fezile Dabi District Municipality												
Total: Fezile Dabi Municipalities	22 643	17 000	32 000	22 643	17 000	32 000						
Total: Free State Municipalities	20 000	53 500	66 463	70 000	53 500	66 463	5 000			5 000		

ANNEXURE WS
INFRASTRUCTURE GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULE 6) 3 OF 4

		Integrated National Electrification Programme (Municipal) Grant	al Electrification	Programme (N	(unicipal) Gra	Ħ		⊡	lectricity Der	nand Side Ma	nagement (M	Electricity Demand Side Management (Municipal) Grant	=
		National Financia	ial Year		Munici	Municipal Financial Year	/ear	Natio	National Financial Year	Year	Municipal Financial Year	nancial Year	
		2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14
Category Municipanity	.y	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)
CATTENG													
A EKU Ekurhuleni		100 000	30 000	53 669	100 000	30 000	53 669	27 000			27 000		
A JHB City of Johannesburg	nnesburg	30 982	20 000	000 09	30 982	20 000	000 09	27 000			27 000		
A TSH City of Tshwane	vane	21 000	000 09	65 000	21 000	000 09	65 000	25 000			25 000		
B GT421 Emfuleni		16 256			16 256			9 000			9 000		
		2 600			2 600								
		3 000			3 000								
C DC42 Sedibeng District Municipality	istrict Municipality												
Total: Sed beng Municipalities		21 856			21 856			000 9			000 9		
B GT482 Randfontein		3 762		2 200	3 762		2 200						
B GT483 Westonaria													
B GT484 Merafong City	ity	8 400			8 400								
C DC48 West Rand I	West Rand District Municipality												
Total: West Rand Municipalities		12 162		2 200	12 162		2 200						
Total: Ganteng Municipalities		186 000	140 000	180 869	186 000	140 000	180 869	000 58			85 000		

ANNEXURE W5
INFRASTRUCTURE GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULE 6) 3 OF 4

	Integrated Nati	Integrated National Electrification Programme (Municipal) Grant	n Programme (N	Iunicipal) Gra	ıt l		EI .	ectricity Den	and Side Ma	Electricity Demand Side Management (Municipal) Grant	micipal) Gran	t
	National Financial Year	cial Year		Munici	Municipal Financial Year	Year	Nation	National Financial Year	Year	Municipal Financial Year	nancial Year	
Category Municipality	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	2011/12	2012/13	3/14	2011/12	2012/13	2013/14
KWAZULU-NATAL	(000 11)	(000 W)	(000 W)	(00 W)	(000 V)	(000 V)	(000 W)	(000 W)	(000 W)	(000 W)	(000 V)	(00 V)
A ETH eThekwini	30 500	35 000	20 000	30 500	35 000	20 000	25 000			25 000		
KZN211 KZN212	2 000 2 300	15 000	4 536	2 000	15 000	4 536						
B KZN213 Umzumbe B KZN214 uMuziwabantu B KZN215 Ezinqolweni B KZN216 Hbiscus Coast C DC21 Ugu District Municipality	2 520	40 000	18 000	2 520	40 000	18 000						
Total: Ugu Municipalities	6 8 2 0	25 000	22 536	6 820	25 000	22 536						
B KZN221 uMshwathi B KZN222 uMngeni B KZN223 Mpofana B KZN224 Impendle B KZN225 Msunduzi B KZN225 Msunduzi B KZN227 Richmond C DC22 uMgungundlovu District Municipality	5 436 7 000 4 000	3 000	1360	5 436 7 000 4 000	3 000	1 360	4 000			4 000		
Total: uMgungundlovu Municipalities	16 436	3 000	1 360	16 436	3 000	1 360	4 000			4 000		
B KZN232 Emnambethi-Ladysmith B KZN233 Indaka B KZN234 Umshezi B KZN235 Okhahlamba B KZN236 Imbabazane C DC22 Ithtubabazane	15 000 7 000 7 4 500 7 480 6 150	8 000	7 560	15 000 7 000 4 500 7 480 6 150	8 000	7 560	4 000			4 000		
otal:Uthukela Munici	40 130	8 000	8 220	40 130	8 000	8 220	4 000			4 000		
B KZN241 Endumeni B KZN242 Nquthu B KZN244 Msinga B KZN245 Umvoii	5 000 7 000 6 000	30 000		5 000 7 000 6 000	30 000							
otal: Umzinyathi Mun	18 000	40 000		18 000	40 000							
KZN252 KZN253 KZN254	000 8	7 000	7 358	8 000	7 000	7 358						
C DC25 Amajuba District Municipality Total: Amajuba Municipalities	16 000	7 000	7 358	16 000	7 000	7 358						

ANNEXURE W5
INFRASTRUCTURE GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULE 6) 3 OF 4

		Integrated National Electrification Programme (Municipal) Grani	nal Electrification	n Programme (M	(unicipal) Gra	nt		E	lectricity Den	nand Side Ma	Electricity Demand Side Management (Municipal) Grant	unicipal) Gra	ınt
		National Financial Year	ıl Year		Munic	Municipal Financial Year	Year	Natio	National Financial Year	Year	Municipal Financial Year	nancial Year	
Category	Municipality	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14
		(000 V)	(000 W)	(000 V)	(000 V)	(000 V)	(M 000)	(N 000)	(N 000)	(w 000)	(000 V)	(N 000)	(000 W)
KZN261	eDumbe			10 000			10 000						
B KZN262	UPhongolo	8 000	15 000	000 6	8 000	15 000	000 6						
KZN263	Abaqulusi	13 773	20 000	8 000	13 773	20 000	8 000						
B KZN265	Nongoma		40 000			40 000							
B KZN266	Ulundi	8 000	5 000	2 490	8 000	5 000	2 490						
DC26	Zululand District Municipality												
Total: Zululand Municipalities	ipalities	29 773	80 000	29 490	29 773	80 000	29 490						
KZN271	Umhlabuyalingana	8 000	15 000	14 000	8 000	15 000	14 000						
B KZN272	Jozini	8 000	10 000	15 000	8 000	10 000	15 000						
KZN273	The Big 5 False Bay	000 9	30 000		000 9	30 000							
KZN274	Hlabisa	12 000	30 000	15 000	12 000	30 000	15 000						
KZN275	Mtubatuba												
C DC27	Umkhanyakude District Municipality												
Total: Umkhanyakude Municipalities	Municipalities	34 000	82 000	44 000	34 000	82 000	44 000						
KZN281	Mfolozi												
KZN282	uMhlathuze												
	Ntambanana	16 000		26 400	16 000		26 400						
KZN284	Umlalazi	099		2 000	099		2 000						
KZN285	Mthonjaneni	10 000			10 000								
2	Nkandla	12 000	20 000	10 000	12 000	20 000	10 000						
C DC28	uThungulu District Municipality												
Total: uThungulu Municipalities	icipalities	38 660	20 000	38 400	38 660	20 000	38 400						
. COLUMN													
KZN291	Mandeni												
KZN292	KwaDukuza	2 000			2 000								
KZN293	Ndwedwe		10 000			10 000							
KZN294	Maphumulo	10 000	15 000	10 000	10 000	15 000	10 000						
C DC29	iLembe District Municipality												
Total: iLembe Municipalities	alities	12 000	25 000	10 000	12 000	25 000	10 000						
KZN431	Тисти	000 9	20 000	2 000	0009	000 02	000 \$						
	Kwa Sani												
KZN433	Greater Kokstad												
	Uhuhlebezwe												
KZN435	Umzimkhulu	10 000	30 000	30 000	10,000	30 000	30 000						
DC43	Sisonke District Municipality	200	200	000 00	200	8	00000						
otal: Sisonke Municip	alities	16 000	20 000	35 000	16 000	20 000	35 000						
Total: KwaZulu-Natal Municipalities	Municipalities	258 319	408 000	216 364	258 319	408 000	216 364	33 000			33 000		

ANNEXURE WS
INFRASTRUCTURE GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULE 6) 3 OF 4

	Integrated National Electrification Programme (Municipal) Grant	nal Electrification	n Programme (N	funicipal) Gra	ıt l		H	lectricity Den	nand Side Ma	Electricity Demand Side Management (Municipal) Grant	unicipal) Gra	nt
	National Financial Year	al Year	)	Munici	Municipal Financial Year	Year	Natio	National Financial Year	Year	Municipal Financial Year	nancial Year	
Category Municipality	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14
	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)
LIMPOPO												
	4 000	16 000	12 000	4 000	16 000	12 000						
LIM332												
LIM333	2 000	2 000	10 000	2 000	2 000	10 000						
LIM334	4 000	2 000	000 9	4 000	2 000	0009						
C DC33 Mopani District Municipality												
Total: Mopani Municipalities	13 000	23 000	28 000	13 000	23 000	28 000						
B LIM341 Musina												
LIM342	3 000			3 000								
	40 000	20 000	40 000	40 000	20 000	40 000						
B LIM344 Makhado	3 500	10 000	15 000	3 500	10 000	15 000						
C DC34 Vhembe District Municipality												
Total: Vhembe Municipalities	46 500	000 09	25 000	46 500	000 09	25 000						
	3 000	3 500	2 000	3 000	3 500	2 000						
LIM353												
LIM354	12 000		10 000	12 000		10 000	12 000			12 000		
B LIM355 Lepelle-Nkumpi	3 000		2 500	3 000		2 500						
C DC35 Capricorn District Municipality												
Total: Capricorn Municipalities	18 000	3 500	17 500	18 000	3 500	17 500	12 000			12 000		
	1 600	3 000	2 000	1 600	3 000	2 000						
LIM362												
LIM364	2 000	2 000		2 000	2 000							
LIM365												
LIM366												
LIM367	000 9			000 9								
C DC36 Waterberg District Municipality												
Total: Waterberg Municipalities	009 6	8 000	2 000	009 6	8 000	2 000						
R 1 1M471 Ephroim Mocole												
LIM472												
1 IM/172	7 500		10000	0037		10,000						
1 TM474			000 01	000		00001						
1 IM475	19 800	10 000	19 000	19 800	10.000	19 000						
DC47		200	000	000	2	000						
Total: Greater Sekhukhune Municipalities	27 300	10 000	29 000	27 300	10 000	29 000						
	977	001	001		901	002.767	000			000		
Total: Limpopo Municipalities	114 400	104 500	134 500	114 400	104 500	134 500	12 000			12 000		

ANNEXURE WS
INFRASTRUCTURE GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULE 6) 3 OF 4

	Integrated National	nal Electrification	Electrification Programme (Municipal) Grant	unicipal) Gra	mt		1	Electricity De	mand Side Ma	Electricity Demand Side Management (Municipal) Grant	unicipal) Gra	ınt
	National Financial	al Year	)	Munic	Municipal Financial Year	Year	Natio	National Financial Year	l Year	Municipal Financial Year	nancial Year	
Category Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
MPUMALANGA												
B MP302 Msukaligwa B MP303 Mkhondo	7 236	5 100	2 100	7 236	5 100	2 100						
MP304												
B MP305 Lekwa	1 389			1 389								
B MP306 Dipaleseng	1 533			1 533								
	5 122	10 000	10 000	5 122	10 000	10 000						
C DC30 Gert Sibande District Municipality												
Total: Gert Sibande Municipalities	15 281	15 100	12 100	15 281	15 100	12 100						
B MP311 Victor Khanye												
B MP312 Emalahleni	18 629	9 9 9 9 9	9 9 9 9 9 9	18 629	9 300	6 300						
B MP313 Steve Tshwete	1 440	10 000	10 000	1 440	10 000	10 000						
B MP314 Emakhazeni	634			634								
B MP315 Thembisile	1 813	100	100	1 813	100	100						
C DC31 Nkangala District Municipality												
Total: Nkangala Municipalities	22 516	16 400	16 400	22 516	16 400	16 400						
B MP321 Thaba Chweu	1 442			1 442								
	10 034	5 500	5 000	10 034	5 500	5 000	8 000			8 000		
B MP323 Umjindi	12 784	8 500	8 500	12 784	8 500	8 500						
B MP324 Nkomazi	19 383	4 500	10 000	19 383	4 500	10 000						
B MP325 Bushbuckridge	5 560	15 000	20 000	5 560	15 000	20 000						
C DC32 Ehlanzeni District Municipality												
Total: Ehlanzeni Municipalities	49 203	33 200	43 500	49 203	33 500	43 500	0008			8 000		
Total: Manmalanga Municipalifias	000 28	000 29	72 000	87 000	000 59	72 000	000 8			8 000		

ANNEXURE W5
INFRASTRUCTURE GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULE 6) 3 OF 4

	Integrated National Electrification Programme (Municipal) Grant	nal Electrificatio	n Programme (N	[unicipal] Grar			E	lectricity Den	nand Side Ma	nagement (Mı	Electricity Demand Side Management (Municipal) Grant	=
	National Financial Year	al Year		Munici	Municipal Financial Year	Year	Nation	National Financial Year	Year	Municipal Financial Year	nancial Year	
Category Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
NORTHERN CAPE												
B NC061 Richtersveld B NC062 Name Khoi	75101		8	10.156		000						
NC064	001.01		000 0	00101		000 0						
NC065	1 513			1 513								
NC066	2 500			2 500								
B NC067 Khāi-Ma C DC6 Namakwa District Municinality												
otal: Namakwa Munic	14 169		8 000	14 169		8 000						
B NC071 Thuntu												
	1 170		280	1 170		280	000 09			000 09		
NC073	260		540	260		540						
B NC074 Kareeberg												
NC076	2 500			2 500								
NC077												
NC078	2 964		1 733	2 964		1 733						
C DC7 Pixley Ka Seme District Municipality												
Total: Pixley Ka Seme Municipalities	6 8 9 4		2 553	6 894		2 553	000 09			000 09		
TOO'S MESSAGE												
NC083	1 829	350	310	1 829	350	310						
NC084												
NC085												
NC086												
C DC8 Siyanda District Municipality	000 1	020	310	000	020	916						
Total: Siyanda Muncipannes	1 029	066	OIC	1 029	Occ	OTC						
B NC091 Sol Plaatje	18 003	10 000	5 000	18 003	10 000	2 000						
NC092												
NC093		,	,			,						
B NC094 Phokwane C DC9 Frances Baard District Municipality	2 184	163	163	2 184	163	163						
otal: Frances Baard N	20 187	10 163	5 163	20 187	10 163	5 163						
B NICASI Machamana												
NC452												
NC453	3 533			3 533								
C DC45 John Taolo Gaetsewe District Municipality												
Total: John Taolo Gaetsewe Municipalities	3 533			3 533								
Total: Northern Cape Municipalities	46 612	10 513	16 026	46 612	10 513	16 026	000 09			000 09		
												1

ANNEXURE W5
INFRASTRUCTURE GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULE 6) 3 OF 4

	Integrated National Electrification Programme (Municipal) Grant	nal Electrificatio	n Programme (N	funicipal) Gra	nt		E	lectricity Der	Electricity Demand Side Management (Municipal) Grant	magement (M	unicipal) Gra	nt
	National Financia	al Year		Munic	Municipal Financial Year	Year	Natio	National Financial Year	Year	Municipal Financial Year	nancial Year	
Category Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
NORTH WEST												
B NW371 Moretele B NW372 Madibeng B NW373 Rustenburg B NW374 Kgetlengrivier B NW375 Moses Kotane C DC37 Bojanala Platinum District Municipality	9 202 14 400	7 000	12 000	9 202	7 000	12 000	000 6			000 6		
otal: Bojanala Platinu	23 602	7 000	12 000	23 602	7 000	12 000	000 6			000 6		
B NW381 Ratiou B NW382 Tswaing B NW383 Maffkeng B NW384 Disobotta B NW384 Ramotshere Moiloa C DC38 Ngaka Modiri Molema District Municipality	11 693	10 000		11 693	10 000							
Total: Ngaka Modiri Molema Municipalities	11 693	10 000		11 693	10 000							
B NW392 Naledi B NW393 Mamusa B NW394 Greater Taung B NW395 Akwa-Teemane	4 320	2 000	4 000	4 320	2 000	4 000						
C DC39 Dr Ruth Segomotsi Mompati District Municipality												
Total: Dr Ruth Segomotsi Mompati Municipalities	4 320	2 000	4 000	4 320	2 000	4 000						
	2 886	10 000		2 886	10 000							
B NW403 City of Matlosana B NW404 Maquassi Hills C DC40 Dr Kenneth Kaunda District Municipality	8 719 780			8 719 780								
Total: Dr Kenneth Kaunda Municipalities	12 385	10 000		12 385	10 000							
Total North World Wood deathir	000 62	000 000	000 31	000 62	90 92	000 71	000 0			000		
Total: North West Municipalities	22 000	29 000	16 000	22 000	29 000	16 000	9			9000		

ANNEXURE W5
INFRASTRUCTURE GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULE 6) 3 OF 4

	Integrated National Electrification Programme (Municipal) Grant	nal Electrification	n Programme (M	(unicipal) Gra	nt		Ē	ectricity Den	and Side Ma	Electricity Demand Side Management (Municipal) Grant	nicipal) Gran	nt
	National Financial Year	ıl Year		Munic		Year	اتجا	National Financial Year	Year	Municipal Financial Year	ancial Year	ıı
Category Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
WESTERN CAPE												
A CPT City of Cape Town	5 000	20 000	20 000	5 000	20 000	20 000	20 000			20 000		
B WC011 Matzikama	2 592	2 000	2 000	2 592	2 000	2 000						
WC012 WC013	7 900			7 900								
B WC015 Swartland C Note Count District Municipality	2 000	800	1 983	2 000	800	1 983						
otal: West Coast Mun	10 392	2 800	3 983	10 392	2 800	3 983						
WC022 WC023	2 000	5 815	2 000	2 000	5 815	2 000						
4 10 70	1 500	06	06	1 500	06	06						
Cape Winelands District Municipality  Total: Cape Winelands Municipalities	9 9 9	5 905	5 090	9 500	5 905	5 090						
B WC031 Theewaterskloof B WC032 Overstrand	1 000		3,639	1 000		3,639						
WC033 WC034	2 000			2 000								
C DC3 Overberg District Mumcipality  Total: Overberg Municipalities	2 000		3 639	2 000		3 639						
B WC041 Kannaland							4 000			4 000		
B WC042 Hessequa	3 700	1 000	5 300	3 700	1 000	5 300	4 000			4 000		
WC044	5 349	1 225	2 000	5 349	1 225	2 000	4 000			4 000		
B WC045 Oudtshorm	2 000	1 000	1 800	2 000	1 000	1 800	4 000			4 000		
WC048	1 300	1 000	1 452	1 300	1 000	1 452	-			-		
C DC4 Eden District Municipality  Total: Eden Municipalities	15 200	6 225	22 052	15 200	6 225	22 052	4 000 2 <b>4 000</b>			4 000 24 000		
B WC051 Laingsburg B WC052 Prince Albert B WC053 Beaufort West	20 000	15 000	14 318	20 000	15 000	14 318						
Total: Central Karoo Municipalities	20 000	15 000	14 318	20 000	15 000	14 318						
Total: Western Cape Municipalities	62 092	49 930	69 082	62 092	49 930	69 082	44 000			44 000		
Unallocated:												
National Total	1 096 612	1 151 443	1 214 772	1 096 612	1 151 443	1 214 772	280 000			280 000		

ANNEXURE WS
INFRASTRUCTURE GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULE 9) 4 OF 4

				Municipal D	Municipal Disaster Grant				CIR	TOTAL·INER	SIIB.TOTAI - INERASTRIICTIIRE		
		Natio	National Financial Year	I Year	Muni	Municipal Financial Year	l Year	Nati	National Financial Year	ar	Munic	Municipal Financial Year	ear
Category	Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
EASTERN CAPE	E E												
A BU	BUF Buffalo City NMA Nelson Mandela Bay							628 446 1 397 626	707 689	897 338 1 404 074	628 446 1 397 626	707 689	897 338 1 404 074
B EC	EC101 Camdeboo							13 973	19 772	16 640	13 973	19 772	16 640
								16 801	22 393	21 982	16 801	22 393	21 982
B EC	EC103 Ikwezi							8 446	10 268	10 833	8 446	10 268	10 833
B EC	EC104 Makana							39 312	47 560	44 256	39 312	47 560	44 256
								24 449	37 294	36 795	24 449	37 294	36 795
	EC106 Sundays River Valley							19 504	23 714	25 018	19 504	23 714	25 018
								10 447	12 701	13 400	10 447	12 701	13 400
B ECI	EC108 Kouga							25 002	27 967	29 505	25 002	27 967	29 505
B ECI								14 643	17 804	18 784	14 643	17 804	18 784
C DC	DC10 Cacadu District Municipality												
Total: Cacadu Municipalities	unicipalities							172 576	219 474	217 214	172 576	219 474	217 214
								400	000	000	6	000	too
								53 489	5/893	780 76	53 489	5/893	780 76
								54 363	68 941	71 907	54 363	68 941	71 907
								13 892	14 460	15 255	13 892	14 460	15 255
								22 884	27 824	29 355	22 884	27 824	29 355
B EC	EC126 Ngqushwa							17 257	20 983	22 137	17 257	20 983	22 137
B EC	EC127 Nkonkobe							21 693	26 377	27 828	21 693	26 377	27 828
B EC	EC128 Nxuba							699 6	11756	12 403	699 6	11 756	12 403
C DC	DC12 Amathole District Municipality							357 011	432 479	464 056	357 011	432 479	464 056
Total: Amathole Municipalities	Municipalities							550 259	660 713	735 028	550 259	660 713	735 028
B EC	EC131 Inxuba Yethemba							11 904	23 259	29 988	11 904	23 259	29 988
B EC	EC132 Tsolwana							11 631	11710	17 354	11 631	11 710	17 354
B EC	EC133 Inkwanca							8 236	10015	10 565	8 236	10 015	10 565
B EC	EC134 Lukhanji							27 480	32 197	42 968	27 480	32 197	42 968
B EC	EC135 Intsika Yethu							25 935	31 534	33 269	25 935	31 534	33 269
B EC	EC136 Emalahleni							23 697	30366	39 651	23 697	30 366	39 651
B ECI	EC137 Engcobo							75 621	86 288	87 734	75 621	86 288	87 734
B ECI	EC138 Sakhisizwe							17 431	16330	23 228	17 431	16 330	23 228
C DC	DC13 Chris Hani District Municipality							286 291	347 831	366 953	286 291	347 831	366 953
Total: Chris Hani Municipalities	i Municipalities							488 227	589 530	651 710	488 227	589 530	651 710

ANNEXURE W5
INFRASTRUCTURE GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULE 9) 4 OF 4

			Municipal D	Municipal Disaster Grant				SUB-	TOTAL: INFR	SUB-TOTAL: INFRASTRUCTURE		
	Nati	National Financial Year	Year	Muni	Municipal Financial Year	ıl Year	Natio	National Financial Year	ar	Munic	Municipal Financial Year	Year
Category Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
B EC14 Elundini							23 466	27.317	43 819	23 466	27.317	43 819
							26 002	28 090	29 635	26 002	28 090	29 635
							15 614	17 336	18 180	15 614	17 336	18 180
							10 928	15 072	20 736	10 928	15 072	20 736
C DC14 Ukhahlamba District Municipality							145 645	176 818	186 534	145 645	176 818	186 534
Total: Ukhahlamba Municipalities							221 655	264 633	298 904	221 655	264 633	298 904
B EC153 Ngquza Hill							40 543	58 353	60 462	40 543	58 353	60 462
							20 254	24 626	25 981	20 254	24 626	25 981
B EC155 Nyandeni							42 771	59 846	82 038	42 771	59 846	82 038
							29 338	64 456	121 351	29 338	64 456	121 351
B EC157 King Sabata Dalindyebo							56 279	66 271	67 365	56 279	66 271	67 365
DC15 O R Tambo District Municipality							564 461	686 061	723 785	564 461	686 061	723 785
Fotal: O.R.Tambo Municipalities							753 646	959 613	1 080 983	753 646	959 613	1 080 983
BC441 Matatiele							28 791	35 008	51 933	28 791	35 008	51 933
							31 157	37 883	54 967	31 157	37 883	54 967
B EC443 Mbizana							49 445	55 802	57 771	49 445	55 802	177 77
B EC152 Nabankulu							34 664	43 909	35 224	34 664	43 909	35 224
DC44 Alfred Nzo District Municipality							309 834	376 458	397 154	309 834	376 458	397 154
Total: Alfred Nzo Municipalities							453 891	549 060	597 049	453 891	549 060	597 049
Total: Eastern Cape Municipalities							4 666 326	5 018 812	5 882 300	4 666 326	5 018 812	5 882 300

ANNEXURE WS
INFRASTRUCTURE GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULE 9) 4 OF 4

			Municipal Disaster Grant	isaster Grant				SUS	TOTAL: INFE	SUB-TOTAL: INFRASTRUCTURE		
	Nai	National Financial Year	Year	Munic	Municipal Financial Year	ıl Year	Nai	National Financial Year	ear	Munic	Municipal Financial Year	Year
Category Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
EE STATE												
MAN Mangaung							451 995	513 427	559 884	451 995	513 427	559 884
FN161 Tetcemena							18 804				22 142	23 350
							20 190	24 142	25 836	20 190	24 1 42	25 836
							16 158				19 384	19 923
FS164 Naledi PC16 Xhorion Dietrict Municipality							14 149				17 203	18 150
1.5							69 302	82 805	87 268	69 302	82 805	87 268
4												
							30 322				36 868	38 896
							17 855	21 709	22 903	17 855	21 709	22 903
							25 015				30 415	32 088
FS184 Matjhabeng							172 182				206 999	219 026
							45 642	55 497	58 549		55 497	58 549
DC18 Lejweleputswa District Municipality												
al: Lejweleputswa Municipalities							291 017	351 488	371 463	291 017	351 488	371 463
FS191 Setsoto							62 963				76 557	80 768
							46 647				56 855	61 707
							28 044		35 973	28 044	34 098	35 973
							194 936				228 329	239 787
							24 587	26 686			26 686	28 154
							23 583	27 689			27 689	29 212
DC19 Thabo Mofutsanyana District Municipality												
al: Thabo Mofutsanyana Municipalities							376 761	450 214	475 600	376 761	450 214	475 600
FS201 Moqhaka							42 021	49 007			49 007	53 592
FS203 Ngwathe							46 765				55 560	60 340
							40 843	52 228	62 826	40 843	52 228	62 826
FS205 Mafube							33 403				30 901	35 326
DC20 Fezile Dabi District Municipality												
al: Fezile Dabi Municipalities							163 031	187 696	212 085	163 031	187 696	212 085
al: Free State Municipalities							1 352 106	1 585 631	1 706 299	1 352 106	1 585 631	1 706 299

ANNEXURE WS
INFRASTRUCTURE GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULE 9) 4 OF 4

			Municipal	Municipal Disaster Grant				CITE	SIB TOTAL : INEPASTBLICTIBE	ASTRICTIBE	-	
			Transcribar D	isaster Grant				ace.	TOTAL HALL	MOLOGICA		
	Natio	National Financial Year	Year	Munic	Municipal Financial Year	l Year	Nat	National Financial Year	ear	Munic	Municipal Financial Year	/ear
Category Municipality	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	20/11/12	2012/13 (R'000)	2013/14	2011/12	2012/13	2013/14
	(M 000)	(K 000)	(K 000)	(N 000)	(x 000)	(K 000)	(N 000)		(NOO)	(K 000)	(K 000)	(K 000)
GAUTENG												
A EKU Ekurhuleni							1 261 276	1 362 640	1 505 121	1 261 276	1 362 640	1 505 121
A JHB City of Johannesburg							2 845 952	2 856 260	2 973 559	2 845 952	2 856 260	2 973 559
A TSH City of Tshwane							1 182 081	1 963 856	2 135 692	1 182 081	1 963 856	2 135 692
									9			
							149 241	154 402	162 894	149 241	154 402	162 894
B GT422 Midvaal							25 445	27 776	29 304	25 445	27 776	29 304
B GT423 Lesedi							26 139	28 135	29 683	26 139	28 135	29 683
C DC42 Sedibeng District Municipality							14 000	25 000	25 000	14 000	25 000	25 000
Total: Sedibeng Municipalities							214 825	235 313	246 880	214 825	235 313	246 880
B GT481 Mogale City							100 957	128 436	133 850	100 957	128 436	133 850
B GT482 Randfontein							34 046	36 822	41 047	34 046	36 822	41 047
B GT483 Westonaria							54 184	65 882	905 69	54 184	65 882	905 69
B GT484 Merafong City							69 537	74 337	78 425	69 537	74 337	78 425
C DC48 West Rand District Municipality							5 000	10 000	10 000	2 000	10 000	10 000
Total: West Rand Municipalities							263 725	315 477	332 828	263 725	315 477	332 828
Total: Gauteng Municipalities							5 767 859	6 733 546	7 194 081	5 767 859	6 733 546	7 194 081

ANNEXURE W5
INFRASTRUCTURE GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULE 9) 4 OF 4

			Municipal I	Municipal Disaster Grant				SUB	-TOTAL: INFR	SUB-TOTAL: INFRASTRUCTURE		
	Nati	National Financial Year	Year	Muni	Municipal Financial Year	Year	Nat	National Financial Year	ear	Munic	Municipal Financial Year	Year
Category Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
KWAZULU-NATAL												
A ETH eThekwini							1 437 174	1 524 706	1 599 926	1 437 174	1 524 706	1 599 926
11014222							1		Ţ	i i		i i
							15 526			15 526	16 44 /	17 351
KZN212							15 328	30 840	21 247	15 328	30 840	21 247
B KZN213 Unzumbe							24 655			15 128	18 394	19 405
KZN215							11 125		14 271	11 125	13 527	14 271
							27 899		50 556	27 899	70 859	50 556
otal: Hon M							350 395	472.482	462 970	350 395	472 482	462.970
Toda: Ogu Muncipanuco							20000	101 111	01/104	00000	701	01/20
KZN221							15 171		19 462	15 171	18 447	19 462
KZN222							13 347		17 120	13 347	16 228	17 120
KZN223							899 6		12 402	899 6	11 756	12 402
							14 307		12 739	14 307	13 786	12 739
KZN225							199 455		207 215	199 455	244 758	207 215
KZN226							17.258	12 472	13 15/	17.258	12 4 / 2	13 15/
							17.500 87.538		17.317	17 500	106 166	11 996
Total: uMgungundlovu Municipalities							374 245		411 409	374 245	440 026	411 409
							56 356			56 356	51 830	45 586
							23 770	20 391	21 513	23 770	20 391	21 513
KZN234							19 010			19 010	15 779	15 142
KZN235							26 633		24 568	26 633	23 287	24 568
B KZN256 Imbabazane							23 431	21 013	22 168	152 431	21 013	195 236
otal: Uthukela Munici							301 629		324 213	301 629	317 367	324 213
B KZN241 Endumeni							15 251	12 463	13 149	15 251	12 463	13 149
							21 020			21 020	55 558	26 963
KZN244							28 981			28 981	36 727	28 197
KZN245							20 732			20 732	17 912	18 897
C DC24 Umzinyathi District Municipality							163 630		209 606	163 630	198 687	209 606
Total: Umzinyathi Municipalities							249 614	321 347	296 813	249 614	321 347	296 813
B KZN252 Newcastle							94 224	110 249	127 571	94 224	110 249	127 571
							8 624			8 624	10 486	11 062
B KZN254 Dannhauser							23 337	18 648	19 673	23 337	18 648	19 673
2.5							169 084		213.040	169 084	101 271	213.040
communication reading for the communication of the							100 001		210 010	100 001	11217	200

ANNEXURE W5
INFRASTRUCTURE GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULE 9) 4 OF 4

				Municipal D	Municipal Disaster Grant				CILE	2-TOTAI.INEE	STIB_TOTAL: INEPASTBLICTIBE		
		Nationa	National Financial Year	Year	Muni	Municipal Financial Year	d Year	Nat	National Financial Year	ear rest	Munic	Municipal Financial Year	Year
Category Municipality	2011/12 (R'000)		2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
B VZN761 oDumbo								317 716				15 408	26.351
								24 793				35 419	30 542
KZN263								34 748	45 504	34 906	34 748	45 504	34 906
KZN265								30 868				75 373	36 769
KZN266								30 011				31 762	30 724
DC26								228 787	277 912			277 912	293 188
Total: Zululand Municipalities								361 954	481 468			481 468	452 480
B VZN771 Hahlohuvolinoono								900 22	30.705			30 205	30 537
								27 170	39 203			39 203	46,009
KZN272								32 172	195 951	11 8/5	32 172	195 951	11 845
LCN7X								21 822	41 228			41 228	009 27
KZN275								17 195	20 907	22 057		20 907	22 057
DC27								188 630	229 084			229 084	241 675
Total: Umkhanyakude Municipalities								302 961	411 758			411 758	388 721
1001/277								200	007 21			000	0,2
								14 4/5	1/ 600			17 600	18 208
KZN282								71 404				86 820	91 595
KZN283								26 193	12 393	39 475	26 193	12 393	39 475
KZN284								22 944				27 095	30 585
KZN285								20 761	13 083			13 083	13 803
9								29 887	41 748			41 748	32 944
C DC28 uThungulu District Municipality								162 621	197 460			197 460	208 311
Total: uThungulu Municipalities								348 283	396 199	435 281	348 283	396 199	435 281
B KZN291 Mandeni								33 152				38 719	42 134
B KZN292 KwaDukuza								45 975			45 975	43 231	37 169
B KZN293 Ndwedwe								30 194	42 553		30 194	42 553	46 904
KZN294								26 881	35 525			35 525	31 654
C DC29 iLembe District Municipality								149 488	181 491			181 491	191 465
Total: iLembe Municipalities								285 690	341 520	349 325	285 690	341 520	349 325
B KZN431 Inewe								22 569	40 146			40 146	26 254
KZN432								7 887	6 589	10 117	7 887	685 6	10 117
KZN433								15 927	19 365			19 365	20 430
B KZN434 Ubuhlebezwe								17 409	21 167	22 331		21 167	22 331
B KZN435 Umzimkhulu								54 688	84 256			84 256	80 910
C DC43 Sisonke District Municipality								138 297	167 885			167 885	177 109
Total: Sisonke Municipalities								256 776	342 408	337 152	256 776	342 408	337 152
Total: KwaZulu-Natal Municipalities								4 437 806	5 240 554	5 271 330	4 437 806	5 240 554	5 271 330

ANNEXURE W5
INFRASTRUCTURE GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULE 9) 4 OF 4

				Municipal Disaster Grant	isaster Grant				SCB SCB	FIGIAL: INF	SUB-TOTAL: INFRASTRUCTURE	<b>=</b>	
		Nat	National Financial Year	l Year	Muni	Municipal Financial Year	d Year	Na	National Financial Year		inuM Muni	Municipal Financial Year	/ear
Category	Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
6													
APOPO													
LIM331	Greater Giyani							33 950					50 419
LIM332	Greater Letaba							32 997	40 121	42 328	32 997	40 121	42 328
LIM333	Greater Tzaneen							65 712					89 922
LIM334	Ba-Phalaborwa							28 129					40 472
LIM335	Maruleng							24 874					31 908
DC33	Mopani District Municipality							264 907					339 521
al: Mopani Municipalities	palities							450 570	554 236	594 570	450 570	554 236	594 570
1 134241	Mercini							12 030	14 620	Ch h 31	020 61	14.620	15 443
14CM1741	Mushid							12 037				14 030	C++ CI
LIM342	Mutale							566 91				17 016	266 / 1
LIM343	Thulamela							126 295				153 529	148 628
LIM344	Makhado							290.66	866 //	80.274	290.65	86677	80 2 / 4
DC34	V hembe District Municipality							196 / 67			196 / 67	362 023	381 927
al: Vhembe Municipalities	palities							512 356	624 766	650 224	512 356	624 766	650 224
I IM351	Rloubera							75 A 77					37 680
1 CCMIT	Access							21 120					77 000
LIMISSZ	Aganang							60, 17	20 433		65/17	20 433	72 987
LIM353	Molemole							19716	,	25 290			25 290
LIM354	Polokwane							2/8 954	. •		278 954	. •	299 173
LIM355	Lepelle-Nkumpi							31 718					39 338
DC35	Capricom District Municipality							175 601		224 962		213 242	224 962
al: Capricorn Municipalities	icipalities							556 204	603 722		556 204	603 722	654 330
								0					1
LIM361	Thabazimbi							36 350			36 350		49 575
LIM362	Lephalale							35 907	43 659			43 659	46 060
LIM364	Mookgopong							15 536					17 364
LIM365	Modimolle							34 157					43 815
LIM366	Bela Bela							15 518					19 906
LIM367	Mogalakwena							147 222	149 668	150 239	147 222	149 668	150 239
DC36	Waterberg District Municipality												
al: Waterberg Municipalities	icipalities							284 690	320 436	326 960	284 690	320 436	326 960
LTM471	Enhraim Mogale							17871	21 729				22 924
L IM472	Epinann mogano Flias Motsoaledi							79 037	35 306	37 248	79 037	35 306	37 248
1 DA172	Mathematical and the second and the							41.650					57 0 17
LIM4/3	Makindunianaga							41 039					10 00
LIM4/4	Fetakgomo							15 40/				18 / 34	19 /64
LIM475	Greater Tubatse							56 111				54 151	65 580
DC47	Greater Sekhukhune District Municipality							345 896			345 896	420 305	443 413
d: Greater Sekhuk	al: Greater Sekhukhune Municipalities							505 982	591 758	642 746	505 982	591 758	642 746
								200 000	0.00.010		000000	0,00,00	000000

ANNEXURE W5
INFRASTRUCTURE GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULE 9) 4 OF 4

				Municipal D	Municipal Disaster Grant				SUE	S-TOTAL: INFF	SUB-TOTAL: INFRASTRUCTURE	E)	
		Nati	National Financial Year	Year	Muni	Municipal Financial Year	ıl Year	Nat	National Financial Year	ear	Munic	Municipal Financial Year	Year
Category Municipality		2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
MINIMATANICA					(	( )		)		(		()	
Mromatanga													
B MP301 Albert Luthuli								63 670	77 416		63 670	77 416	81 674
B MP302 Msukaligwa								40 521	45 571			45 571	44 797
B MP303 Mkhondo								48 703	59 219	62 476		59 219	62 476
B MP304 Pixley Ka Seme								26 576	32 313			32 313	34 090
B MP305 Lekwa								37 266	43 622			43 622	46 022
B MP306 Dipaleseng								19 415	21 742	22 937	19 415	21 742	22 937
B MP307 Govan Mbeki								80 877	102 111	107 177	80 877	102 111	107 177
C Gert Sibande District Municipality	y												
Total: Gert Sibande Municipalities								317 029	381 994	399 174	317 029	381 994	399 174
MP311								20 521	24 951	26 324	20 521	24 951	26 324
B MP312 Emalahleni								96 751	101 289			101 289	106 514
								47 397	57 640			57 640	69 711
B MP314 Emakhazeni								13 765	15 966	16 844	13 765	15 966	16 844
B MP315 Thembisile								90 920	108 445	114 404	90 920	108 445	114 404
B MP316 Dr JS Moroka								91 325	111 042	117 149	91 325	111 042	117 149
C DC31 Nkangala District Municipality													
Total: Nkangala Municipalities								360 679	419 335	450 946	390 679	419 335	450 946
									0.5			0.5	0 70
MF321								20 133	32 432	34 237		32 432	34 237
MIT 3 2 2								224 303	214 003		224 303	214 003	070 577
MP323								191 68	55 /45			55 /45	37.243
MP324								131 591	140 934	153	131	140 934	153 938
								241 394	301 751	322 523	241 394	301 751	322 523
C DC32 Ehlanzeni District Municipality													
Total: Ehlanzeni Municipalities								660 874	724 886	771 812	660 874	724 886	771 812
Total: Manuelanda Municipalities								1 338 582	1 526 215	1 621 031	1 338 582	1 526 215	1 621 031

ANNEXURE WS
INFRASTRUCTURE GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULE 9) 4 OF 4

			Municipal D	Municipal Disaster Grant				SUB	TOTAL: INFR	SUB-TOTAL: INFRASTRUCTURE		
	Nati	National Financial Year	l Year	imM	cip	l Year		National Financial Year	ar	Munic	Municipal Financial Year	ear
Category Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
NORTHERN CAPE												
B NC061 Richtersveld							7 113	8 648	9 124	7 113	8 648	9 124
							23 419		25 013	23 419	16 126	25 013
NC064							9 634		12 358	9 634	11 714	12 358
NC065							12 018		13 475	12 018	12 773	13 475
B NC067 Khái-Ma							11 883	11 408	12 033	8 840	10 748	12 035
DC6												
Total: Namakwa Municipalities							72 908	71 417	83 345	72 908	71 417	83 345
B NC071 Illumen							0 488	11 537	17 1 7 1	0.488	11 537	12 171
NC071							77 935		15 372	77 935	14 305	15 372
							13 566		17 608	13 566	16 178	27.5 CI
NC074							7 892		10 124	7 892	9656	10 124
NC075							8 170		10 480	8 170	9 934	10 480
NC076							13 686		14 349	13 686	13 600	14 349
B NC077 Siyathemba							10 246		13 144	10 246	12 458	13 144
B NC078 Siyancuma							19 874		23 425	19 874	20 561	23 425
C DC7 Pixley Ka Seme District Municipality												
Total: Pixley Ka Seme Municipalities							155 859	108 170	116 672	155 859	108 170	116 672
NICOOL ME.							900		0000	300	117	0000
							9 383		12 038	9 383	21 027	72 193
NC082							21 440	24 195	25 165	21 440	24 195	25 165
NC084							11 434		14 667	11 434	13 902	14 667
							11 888		15 249	11 888	14 454	15 249
NC086							T2T T		9 911	T 27 T	9 395	9 911
C DC8 Siyanda District Municipality  Total: Siyanda Municipalities							79 167	94 383	99 515	79 167	94 383	99 515
											3	
B NC091 Sol Plaatje							73 366		78 190	73 366	80 157	78 190
							21 118		27 089	21 118	25 677	27 089
NC093							10 798	13 129	13 851	10 798	13 129	13 851
B NC094 Phokwane							23 919		28 044	23 919	26 591	28 044
T. 4.1. T							120.001	145 54	201 101	120.003	1.45 554	76. 64.
A Otal: Frances Daard Municipannes							707 671		C/1 /+I	707 671	140 041	14/ 1/2
							41 128	50 008	52 758	41 128	50 008	52 758
NC452							50 474		68 333	50 474	65 292	68 333
B NC453 Gamagara C DC45 Iohn Taolo Gaersewe District Municinality							17 783	17 325	18 278	17 783	17 325	18 278
otal: John Taolo Gaet							109 385	132 624	139 369	109 385	132 624	139 369
Total: Northam Cana Municipalities						T	546 521	777 177	586 075	546 521	771 525	220 985
TOTAL INDIANCIA CAPE INTUINCIPALINES							170 040	1+1 7cc	20000	170 040	141 700	20000

ANNEXURE WS
INFRASTRUCTURE GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULE 9) 4 OF 4

			Municipal Disaster Gran	isaster Grant				SUB	-TOTAL: INFR	SUB-TOTAL: INFRASTRUCTURE		
	Nat	National Financial Year	Year	Muni	Municipal Financial Year	l Year	Nat	National Financial Year		Munic	Municipal Financial Year	ear
Category Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
NORTH WEST												
B NW371 Morestele							95 713	104 218	100 050	95 713	816 101	100 050
I/CWNI							01/00	104 710	006 601	61, 62,	017 +01	006 601
NW372							173 142	199 336	210 299	173 142	199 336	210 299
B NW373 Rustenburg							372 347	454 992	586 156	372 347	454 992	586 156
B NW374 Kgetlengrivier							16 600	20 184	21 294	16 600	20 184	21 294
B NW375 Moses Kotane							102 056	124 090	130 915	102 056	124 090	130 915
DC37												
١Ē							749 858	902 820	1 058 615	749 858	902 820	1 058 615
•												
B NW381 Ratiou							18 073	21 975	23 183	18 073	21 975	23 183
B NW382 Tswaing							32 661	25 495	26 897	32 661	25 495	26 897
B NW383 Mafikeng							35 381	53 019	45 385	35 381	53 019	45 385
B NW384 Ditsobotla							25 393	30 876	32 574	25 393	30 876	32 574
B NW385 Ramotshere Moiloa							32 588	42 465	33 976	32 588	42 465	33 976
C DC38 Ngaka Modiri Molema District Municipality							169 151	205 400	216 688	169 151	205 400	216 688
Total: Ngaka Modiri Molema Municipalities							313 248	379 229	378 703	313 248	379 229	378 703
B NW392 Naledi							16 329	16 601	19 404	16 329	16 601	19 404
B NW393 Mamusa							13 800	16 778	17 701	13 800	16 778	17 701
NW394							54 975	52 799	49 603	54 975	52 799	49 603
B NW396 Lekwa-Teemane							13 179	16 025	16 906	13 179	16 025	16 906
B NW397 NW397							24 613	29 926	31 572	24 613	29 926	31 572
C DC39 Dr Ruth Segomotsi Mompati District Municipality							93 272	113 138	119 352	93 272	113 138	119 352
Total: Dr Ruth Segomotsi Mompati Municipalities							216 168	245 267	254 538	216 168	245 267	254 538
B NW401 Ventersdorp							22 932	24 374	25 715	22 932	24 374	25 715
B NW402 Tlokwe							38 191	60 573	58 859	38 191	60 573	58 859
B NW403 City of Matlosana							114 329	132 332	149 060	114 329	132 332	149 060
B NW404 Maquassi Hills							29 196	34 551	36 451	29 196	34 551	36 451
C DC40 Dr Kenneth Kaunda District Municipality												
Total: Dr Kenneth Kaunda Municipalities							204 647	251 829	270 084	204 647	251 829	270 084
Total: North West Municipalities							1 483 921	1 779 145	1 961 940	1 483 921	1 779 145	1 961 940

ANNEXURE WS
INFRASTRUCTURE GRANT ALLOCATIONS TO MUNICIPALITIES (SCHEDULE 9) 4 OF 4

			Municinal D	Municinal Disaster Grant				SIIS	SIIB-TOTAL: INFRASTRICTIBE	ASTRICTIRE	r.	
	Nat	National Financial Year	l Year	Munic	Municipal Financial Year	Year		National Financial Year	ear	Munic	Municipal Financial Year	ear
Category Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
WESTERN CAPE												
A CPT City of Cape Town							2 726 530	2 656 841	2 767 485	2 726 530	2 656 841	2 767 485
B WC011 Matzikama							20 913	21 628	21 652	20 913	21 628	21 652
B WC012 Cederberg							15 614	15 580	16 437	15 614	15 580	16 437
B WC013 Bergnivier							9 727	11 827	12 477	9 727	11 827	12 477
WC014							20 164	19 237	21 434	20 164	19 237	21 434
B WC015 Swartland							11 974	14 559	15 359	11 974	14 559	15 359
otal: West C							78 391	82 830	87.360	78 391	82 830	87 360
WC022							18 229	22 164	23 383	18 229	22 164	23 383
WC023							30 919	40 977	42 096	30 919	40 977	42 096
WC024							20 955	25 480	26 881	20 955	25 480	26 881
							26 982	30 983	32 688	26 982	30 983	32 688
B WCUZb Langeberg C DC2 Cane Winelands District Municipality							20 149	22 109	77 277	20 149	22 109	77 277
uds							117 235	141 713	146 570	117 235	141 713	146 570
B WC031 Theewaterskloof							22 577	26 235		22 577	26 235	27 678
WC032							20 171	16 987	21.560	20 171	16 987	21.560
WC033							8 479	10 308		8 479	10 308	10 875
WC034							11 172	11 151	11 765	11 172	11 151	11 765
C DC3 Overberg District Municipality												
Total: Overberg Municipalities							62 398	64 681	71 878	62 398	64 681	71 878
B WC041 Kannaland							13 494	11 543		13 494	11 543	12 178
B WC042 Hessequa							18 251	13 829	18 835	18 251	13 829	18 835
WC043							20 143	18 196		20 143	18 196	25 642
B WC044 George							45 658	45 372	48 575	45 658	45 372	48 575
WC047							32.081	30 689	30 497	32.081	30 689	30 497
WC048							26 233	33 236	47 021	26 233	33 236	47 021
DC4							4 000			4 000		
Total: Eden Municipalities							184 143	181 237	203 059	184 143	181 237	203 059
B WC051 Laingsburg							6 755	8 2 1 4		6 755	8 214	8 665
WC052							7 055	8 5 7 8	9 050	7 055	8 578	9 050
WC053							40 473	37 187	.,	40 473	37 187	36 987
C DC5 Central Karoo District Municipality												
Total: Central Karoo Municipalities							54 284	53 979	54 702	54 284	53 979	54 702
Total: Western Cape Municipalities							3 222 980	3 181 281	3 331 054	3 222 980	3 181 281	3 331 054
Unallocated:	470 000	330 000	350 000	470 000	330 000	350 000	470 000	330 000	350 000	470 000	330 000	350 000
National Total	470 000	330 000	350 000	470 000	330 000	350 000	25 595 901	28 642 249	30 773 841	25 595 901	28 642 249	30 773 841
	-							-			-	

## ANNEXURE W6

## ALLOCATIONS-IN-KIND TO MUNICIPALITIES (SCHEDULE 7)

## (National and Municipal Financial Year)

ANNEXURE W6
ALLOCATIONS-IN-KIND TO MUNICIPALITIES (SCHEDULE 7) 1 OF 3

		Reg	Regional Bulk Inf	frastructure Grant	ant			Water	Water Services Onerating Subsidy	rating Subsig	ų.			Rural	Rural Households Infrastructure Grant	frastructure (	rant	
	Natio	nal Financial	Year		ipal Financial	Year	Nations	National Financial Year	sar	Munici	pal Financial	Year	Nation	National Financial Year	Year	Munici	Municipal Financial Year	Year
Category Municipality	2011/12 (R'000)	2012/13 (R'000)	(17/12 2012/13 2013/14 (000) (R'000) (R'000)	201 (R'0	1/12 2012/13 2013/14 00) (R'000) (R'000)	2013/14 (R'000)	2010/11 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2010/11 (R'000)	VII 2011/12 2012/13 00) (R'000) (R'000)	2012/13 (R 000)	2010/11 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2010/11 (R'000)	2011/12 (R'000)	2012/13 (R'000)
EASTERN CAPE																		
A BUF Buffalo City A NMA Nelson Mandela Bay																		
B EC101 Camdeboo B EC102 Blue Crame Route B FC103 Ibuseri																		
EC104 EC105																		
EC106 EC107	16 000	8 300	11 000	16 000	8 300	11 000												
EC108 EC109				-		6												
C DC10 Cacadu District Municipatity  Total: Cacadu Municipalities	23 000	25 300	42 000	23 000	25 300	42 000												
BC121 BC122													4 500	13 000	10 000	4 500	13 000	10 000
													4 500 4 500	000 6	10 000	4 500 4 500	000 6	10 000
BC127 BC128	1			1	6	9												
C DC12 Amathole District Municipality Total: Amathole Municipalities	55 700	33 200 33 200	10 000 10 000	55 700 55 700	33 200 33 200	10 000 10 000		$\parallel$	$\parallel$				18 000	44 000	40 000	18 000	44 000	40 000
B EC131 Inxuba Yethemba B EC132 Tsolwana B EC133 Inkwanca																		
BC134 BC135													4 500	7 000	4 000	4 500	7 000	4 000
EC136 EC137 EC138													4 500	000 6	10 000	4 500	000 6	10 000
C DC13 Chris Hani District Municipality  Total: Chris Hani Municipalities	112 000	179 350	210 000 210 000	112 000 112 000	179 350 179 350	210 000 210 000		$\parallel$	$\parallel$				000 6	16 000	14 000	0006	16 000	14 000
B EC141 Elundini B EC142 Senqu													4 500	000 6	10 000	4 500	000 6	10 000
ᄪ													000 6	18 000	20 000	0006	18 000	20 000
B EC153 Ngquza Hill B EC154 Port St. Johns B EC155 Nyandeni B EC156 Milhortho B EC157 Kirpe Salvan Diindvebo													4 500 4 500 4 500	000 6 000 6	8 000 8 000 8 000	4 500 4 500 4 500	000 6 000 6	000 8 000 8
15	4 120	000 9	10 000	4 120 4 120	000 9	10 000							13 500	27 000	24 000	13 500	27 000	24 000
B EC441 Maraticle B EC442 Unzimvubu B EC443 Mirzana													4 500 4 500 4 500	000 6	10 000 8 000 8 000	4 500 4 500 4 500	000 6	10 000 8 000 8 000
BC152 DC44	121 382		145 000	121 382	122 800	145 000							2 000	4 500	000 6	2 000	4 500	000 6
Total: Alfred Nzo Municipalities	121 382	122 800		121 382	122 800	145 000							15 500	31 500	35 000	15 500	31 500	35 000
Total: Eastern Cape Municipalities	316 202	366 650	417 000	316 202	366 650	417 000			$\parallel$				000 59	136 500	133 000	000 29	136 500	133 000

ANNEXURE W6
ALLOCATIONS-IN-KIND TO MUNICIPALITIES (SCHEDULE 7) 1 OF 3

		Regio	nal Bulk Infr	Regional Bulk Infrastructure Grant	ınt			Water	Water Services Operating Subsidy	rating Subsic	Ŋ			Rural	Households Ir	Rural Households Infrastructure Grant	Frant	
	Nation	National Financial Year	ear	Municit	Municipal Financial Year	Year	Nations	National Financial Year	ear	Municit	Municipal Financial Year	Year	Nation	National Financial Year	Year	Munic	Municipal Financial Year	Year
Category Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R 000)	2010/11 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2010/11 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2010/11 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2010/11 (R'000)	2011/12 (R'000)	2012/13 (R'000)
FREE STATE																		
A MAN Mangaung				+			+			1	+		1					
B FS161 Letseneng B FS162 Kopanong D FS162 Abbelone	30 000	72 300	900	30 000	72 300	96												
FS164 DC16	2	2000	10 000	6000	000 51	10 000												
Total: Xhariep Municipalities	40 894	85 300	35 000	40 894	85 300	35 000												
B FS181 Masilonyana   B FS182 Tekologo   B FS183 Tewelpade   B FS184 Majhaleng   B FS184 Majhaleng   B FS184 Majhaleng   C FS185 Leiwelentsaxa District Municipality	49 000	20 300	3 000	49 000	20 300	3 000												
181	49 000	20 300	3 000	49 000	20 300	3 000												
B         FS191         Setsoto           B         FS192         Dilabbeng           B         FS193         Natetoma           B         FS194         Maluti a Profung           B         FS195         Phunnelela           B         FS196         Phunnelela           B         FS196         Phunnelela           C         TY10         Thack Modification	68 780 9 000	30 400	15 500	9 000	30 400	15 500							4 000 4 000	000 6 000 6	10 000	4 000	000 6 6	10 000
Ž	77 780	30 400	25 500	77 780	30 400	25 500							8 000	18 000	20 000	8 000	18 000	20 000
B FS201 Moghaka B FS203 Ngwathe B FS204 Masimaholo B FS205 Maribabio FS205 Maribabiostrict Manicipality													2 000			2 000		
Total: Fezile Dabi Municipalities					H								2 000			2 000		
Posts I. Duce Cleate Manifolium History	107 674	13,500	002 69	150 501	132,000	002 67		$\uparrow$					90	90	000 00	000 01	000	000 00
Total: Free State Municipannes	10 / 0 / 4	000 001	000 00	10/0/4	130 000	000 00			1	1		1	10 000	19 000	70 000	000 01	non er	70 000

ANNEXURE W6
ALLOCATIONS-IN-KIND TO MUNICIPALITIES (SCHEDULE 7) 1 OF 3

		Regio	nal Bulk Infra	Regional Bulk Infrastructure Grant	ii ii			Water	· Services Ope	Water Services Operating Subsidy	À			Rural I	Rural Households Infrastructure Grant	frastructure	Grant	
	Natio	National Financial Year	ear	Municit	Municipal Financial Year	Year	Nationa	National Financial Year	sar	Municip	Municipal Financial Year	ear	Nation	National Financial Year	Year	Munic	Municipal Financial Year	Year
Category Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R 000)	2010/11 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2010/11 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2010/11 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2010/11 (R'000)	2011/12 (R'000)	2012/13 (R'000)
GAUTENG																		
A EKU Ekurhuleni A JHB City of Johannesburg A TCH City of Tahuma	30,000	900	900	90	900	98	Ę			33								
	200.00	44 000	000 10	30,000	45,000	000 10	430			430								
B GT421 Emfuleni	30 000	57 000	70 000	30 000	57 000	70 000												
B GT422 Midvaal																		
B GT423 Lesedi																		
C DC42 Sedibeng District Municipality																		
Total: Sedibeng Municipalities	30 000	27 000	20 000	30 000	27 000	20 000												
CHACLE CO.																		
B GT482 Randfontein																		
	20 000	42 500	000 09	20 000	42 500	000 09												
C DC48 West Rand District Municipality																		
Total: West Rand Municipalities	20 000	42 500	000 09	20 000	42 500	000 09				_	_							
Total: Cantona Municipalities	000 00	148 500	101 000	000 08	148 500	101 000	030			430								

ANNEXURE W6
ALLOCATIONS-IN-KIND TO MUNICIPALITIES (SCHEDULE 7) 1 OF 3

		Region	Regional Bulk Infra	frastructure Grant	Ħ			Wate	Water Services Operating Subsidy	erating Subs	dy			Rural	Rural Households Infrastructure Grant	frastructure (	Frant	
	Nations	National Financial Year	ear	Municit	Municipal Financial Year	Year	Nation	National Financial Year	'ear	Munic	Municipal Financial Year	Year	Natio	National Financial Year	Year	Munic	Municipal Financial Year	Year
Category Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R 000)	2010/11 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2010/11 (R'000)	2011/12 (R'000)	2012/13 (R 000)	2010/11 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2010/11 (R'000)	2011/12 (R'000)	2012/13 (R'000)
KWAZULU-NATAL																		
A ETH eThekwini																		
													4 000	10 000	10 000	4 000	10 000	10 000
B KZN215 Ezinqolweni B KZN216 Hibiscus Coast C DC21 Ugu District Municipality	40 000	40 000	42 823	40 000	40 000	42 823												
Total: Ugu Municipalities	40 000	40 000	42 823	40 000	40 000	42 823							4 000	10 000	10 000	4 000	10 000	10 000
													4 000	8 000	10 000	4 000	8 000	10 000
B KZN227 Richmond C DC22 uMgungundlovu District Municipality	25 000	33 000	25 000	25 000	33 000	25 000							9			4	;	
Total: uMgungundlovu Municipalities	25 000	33 000	25 000	25 000	33 000	25 000							9 000	12 500	17 000	000 9	12 500	17 000
B KZN232 Emnambethi-Ladysmith B KZN233 Indaka B KZN234 Umskezi B KZN235 Okhahlamba													4 000	8 000	10 000	4 000	8 000 8	10 000
B KZN236 Imbabazane C DC23 Uthukela District Municipality	18 430	44 000	65 910	18 430	44 000	65 910												
Total:Uthukela Municipalities	18 430	44 000	65 910	18 430	44 000	65 910							8 000	16 000	20 000	8 000	16 000	20 000
B KZN241 Endumeni B KZN242 Nqubu B KZN244 Msinga B KZN245 Unvoit ii C DC34 Unzanyaulii Disrict Municipality	7 437	37 000	41 813	7 437	37 000	41 813							4 000	7 500	10 000 10 000 8 000	4 000	7 500	10 000 10 000 8 000
Total: Umzinyathi Municipalities	7 437	37 000	41 813	7 437	37 000	41 813							4 000	7 500	28 000	4 000	7 500	28 000
B KZN252 Newcaste B KZN253 Emadlangeni B KZN254 Damithuser C DZ35 Amanjaba Berret Municipality	22 871		25 000	22 871		25 000							4 000		10 000	4 000	10 000	10 000
Total: Amajuba Municipalities	22 871		25 000	22 871		25 000							4 000	10 000	10 000	4 000	10 000	10 000

ANNEXURE W6
ALLOCATIONS-IN-KIND TO MUNICIPALITIES (SCHEDULE 7) 1 OF 3

		Region	Regional Bulk Infrastructure Grant	structure Gra	nt			Water	Water Services Operating Subsidy	erating Subsi-	dy.			Rural	Rural Households Infrastructure Grant	frastructure	rant	
	Nation	National Financial Year	ear	Municip	Municipal Financial Year	ear	Nationa	National Financial Year	ear	Munici	Municipal Financial Year	Year	Nation	National Financial Year	Year	Munic	Municipal Financial Year	Year
Category Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R 000)	2010/11 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2010/11 (R'000)	2011/12 (R'000)	2012/13 (R 000)	2010/11 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2010/11 (R'000)	2011/12 (R'000)	2012/13 (R'000)
B KZN261 eDumbe B KZN3c2 UPhongolo B KZNA63 Abaqulusi B KZN265 Nongoma B KZN266 Ulundi													4 000	10 000	000 8	4 000	10 000	8 000
C DC26 Zululand District Municipality  Total: Zululand Municipalities	000 09	74 928 <b>74 928</b>	48 413 48 413	000 09	74 928 <b>74 928</b>	48 413 <b>48 413</b>							4 000	10 000	8 000	4 000	10 000	8 000
													4 000 4	10 000	8 000 8	4 000	10 000	8 000 8
B KAN2.5 I Be Big 3 False Bay B KAN274 Habisa B KAN275 Mubanba C DC27 Umkhanyakude District Municipality	101 000	113 878	85 000	101 000	113 878	85 000							4 000	10 000	8 000	4 000	10 000	8 000
Total: Umkhanyakude Municipalities	101 000	113 878	85 000	101 000	113 878	85 000							12 000	30 000	24 000	12 000	30 000	24 000
		100	500 52	2000	18								5 000	8 000 8	10 000	4 000 5 000 5 000	8 000	10 000
C DC28 uThungulu District Municipality	33 999	40 001	20 000	33 999	40 001	20 000		1	1	1	1		000	000	00000	0000	000	000 00
Total: uThungulu Municipalitie	33 999	40 001	20 000	33 999	40 001	20 000							14 000	26 000	30 000	14 000	26 000	30 000
B KZN291 Mandeni B KZN292 KwaDukuza B KZN293 Ndwedwe B KZN294 Maphumulo C DC29 iLembe District Municipality	17 663	37 576	37 000	17 663	37.576	37 000							4 500	8 000	10 000	4 500	8 000	10 000
Total: iLembe Municipalities	17 663	37 576	37 000	17 663	37 576	37 000							4 500	8 000	10 000	4 500	8 000	10 000
B KZN431 Ingwe B KZN432 Kwa Sani B KZN433 Greater Kokstad B KZN434 Ubulicheczwe B KZN435 Umzinkhulu C DC43 Sisonke District Municipality	00091	20 000	35 441	16 000	20 000	35 441							4 000	7 000	9 000	4 000	7 000	9 000 01
7	16 000	20 000	35 441	16 000	20 000	35 441							8 000	17 000	19 000	8 000	17 000	19 000
Total: KwaZulu-Natal Municipalities	342 400	440 383	456 400	342 400	440 383	456 400							68 500	147 000	176 000	68 500	147 000	176 000

ANNEXURE W6
ALLOCATIONS-IN-KIND TO MUNICIPALITIES (SCHEDULE 7) 1 OF 3

		Region	al Bulk Infra	Regional Bulk Infrastructure Grant	#			Water	Services Ope	Water Services Operating Subsidy				Rural House	Rural Households Infrastructure Grant	tructure Gr	ant	
	Nationa	National Financial Year	3ar	Municip	Financial Y	ear	Nationa	National Financial Year	ar	Municipal	ial Y	H	National Fit	National Financial Year		Municips	nancial Y	ear
Category Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2010/11 (R'000)	2011/12 2 (R'000) (	2012/13 (R'000)	2010/11 20 (R'000) (R	2011/12 2012/13 (R'000) (R'000)		2010/11 2011/12 (R'000) (R'000)		2012/13 20 (R'000) (A	2010/11 (R'000)	2011/12 (R'000)	2012/13 (R'000)
LIMPOPO																		
B LIM331 Greater Givani													4 000	2 000	000	4 000	2 000	8 000
													4 000	7 000	8 000	4 000	7 000	8 000
													4 000	7 000	8 000	4 000	7 000	8 000
	i i	0	000	000	000	000	0			0								
C DC55 Mopani District Municipality	000 //	68 /89	20 000	000 //	68/89	20 000	10 599			10 599				000	0000	000	000	000
Total: Mopani Municipalities	77 000	68 789	20 000	77 000	68 789	20 000	10 599		+	10 599			12 000	21 000	24 000	12 000	21 000	24 000
D I PAZAI Menina																		
													000	0000	0000	000	0000	000 8
													000	000 8	0000	900	000 8	0000
	42 000	45 200	15 000	42 000	45 200	15 000	17 767			197.71			800	0000	9000	900 +	0000	9 000
12	42 000	45 200	15 000	42 000	45 200	15 000	17 767			17 767			8 000	16 000	16 000	8 000	16 000	16 000
B LIM351 Blouberg													4 000	7 000	8 000	4 000	2 000	8 000
													4 000	7 000	8 000	4 000	7 000	8 000
													4 000	7 000	8 000	4 000	7 000	8 000
C DC35 Capricom District Municipality	39 000			39 000			26 007			26 007								
Total: Capricorn Municipalities	39 000			39 000			26 007			26 007		+	12 000	21 000	24 000	12 000	21 000	24 000
B I IM361 Thahazimbi																		
													4 000	7 000	8 000	4 000	2 000	8 000
							8			68								
	40 000	76 700	80 000	40 000	76 700	80 000	12 078			12 078			4 000	7 000	8 000	4 000	7 000	8 000
C DC36 Waterberg District Municipality																		
Total: Waterberg Municipalities	40 000	26 700	80 000	40 000	26 700	80 000	12 167			12 167			8 000	14 000	16 000	8 000	14 000	16 000
B LIM471 Enhraim Moosle																		
													4 000	7 000	8 000	4 000	7 000	8 000
				_	_													
B LIM474 Fetakgomo																		
B LIM475 Greater Tubatse													4 000	0006	8 000	4 000	000 6	8 000
C DC47 Greater Sekhukhune District Municipality	117 000	187 000	225 000	117 000	187 000	225 000	26 922			26 922								
Total: Greater Sekhukhune Municipalities	117 000	187 000	225 000	117 000	187 000	225 000	26 922			26 922			8 000	16 000	16 000	8 000	16 000	16 000
Total: Limnono Municipalities	315,000	089 222	370 000	315 000	089 222	370 000	03 462	1	$\dagger$	03 462	1	+	48 000	000 88	000 90	48 000	000 88	000 96
TOTAL EMILIPORY SAMINATION OF	OTO OTO	211 000	200 000	Topo or	211 00	200 000	TOL CY			TOT OF	-				20 000	non of	00 00	7000

ANNEXURE W6
ALLOCATIONS-IN-KIND TO MUNICIPALITIES (SCHEDULE 7) 1 OF 3

		Regic	Regional Bulk Infrastructure Grant	astructure Gra	mt			Wate	Water Services Operating Subsidy	erating Subsic	dy			Rural	Rural Households Infrastructure Grant	nfrastructure	Grant	
	Natio	National Financial Year	Year	Municia	Municipal Financial Year	Year	Nation	National Financial Year	ear	Munici	Municipal Financial Year	Year	Nation	National Financial Year	Year	Munic	Municipal Financial Year	Year
Category Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2010/11 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2010/11 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2010/11 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2010/11 (R'000)	2011/12 (R'000)	2012/13 (R'000)
MPUMALANGA																		
B MP301 Albert Luthuli B MP302 Msukaligwa	4 045 2 045		12 000	4 045 2 045		12 000												
MP304 MP305 MP305																		
	2 042			2 042														
	8 132		12 000	8 132		12 000												
B MP311 Victor Khanye	22 000	25 000	30 000	22 000	25 000	30 000												
B MP312 Emalahleni B MP313 Sleve Tshwere																		
							299			299								
B MP316 Dr.JS Moroka C DC31 Nkangala District Municipality	1 000	15 000	20 000	1 000	15 000	20 000												
Total: Nkangala Municipalities	23 000	40 000	20 000	23 000	40 000	20 000	299			299								
B MP322 Mbombela B MP323 Umjindi	9 500	22 000	30 000	9 500	22 000	30 000												
													4 000	000 6	8 000	4 000	0006	8 000
	105 900	41 000		105 900	41 000		160			160			4 000	15 000		4 000	15 000	8 000
C DC32 Ehlanzeni District Municipality	5 500		26 000	5 500		26 000												
Total: Ehlanzeni Municipalities	120 900	63 000	26 000	120 900	63 000	26 000	160			160			8 000	24 000	16 000	8 000	24 000	16 000
Total: Munmalanga Municinalities	152 032	103 000	118 000	152 032	103 000	118 000	827			827			8 000	24 000	16 000	8 000	24 000	16 000

ANNEXURE W6
ALLOCATIONS-IN-KIND TO MUNICIPALITIES (SCHEDULE 7) 1 OF 3

		Dogion	Designed Dull: Infusetunetune Cunnt	, Commont	1	-		Water	Water Couries Orometing Subside	omoting Culoi	4			Dunel II.	Dunel Householde Infracturations Curret	, occuporation,	, mont	
	Nation	National Financial Year	ar bur mit	Municir	Municipal Financial Year	ear	Nations	National Financial Year	ear ear	Munic	Municipal Financial Year	Year	National	National Financial Year	ear	Munici	Municipal Financial Year	Year
Category Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2010/11 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2010/11 (R'000)	2011/12 (R'000)	2012/13 (R 000)	2010/11 2 (R'000) (	2011/12 (R'000)	2012/13 (R'000)	2010/11 (R'000)	2011/12 (R'000)	2012/13 (R'000)
NORTHERN CAPE																		
B NC061 Richtersveld	000 >1	18 000	000 6	900	18 000	9 000												
NC062	000 01	900 14	000 001	10 000 10 000	000 14	000 501												
NC065			8 775			8 775												
B NC066 Karoo Hoogland B NC067 Khái-Ma																		
C DC6 Namakwa District Municipality																		
Total: Namakwa Municipalities	16 000	29 000	120 775	16 000	29 000	120 775												
B NC071 Ubuntu																		
	32 480	34 000	20 000	32 480	34 000	20 000												
	17 000	18 000	2 000	17 000	18 000	2 000												
	900	000 00	000	000	000 00	900												
D NCO77 Simpleme	17,000	70,000	2 000	12 000	70,000	2 000												
D NOOR Simmon																		
			2 000			2 000												
K	61 480	72 000	35 000	61 480	72 000	35 000												
n Menol Me																		
	3 505			3 595														
NC083				1														
	33 500			33 500														
B NC086 Kgatelopele																		
Total: Siyanda Municipalities	37 095			37 095														
NC091																		
NC092			14 000			14 000												
B NC094 Photware																		
			35 000			35 000												
8			49 000		H	49 000												
NC451	35 000	68 500		35 000	008 89								4 000	000 6	10 000	4 000	000 6	10 000
NC452																		
B NC453 Gamagara C DC45 John Taolo Gaetsewe District Municipality							5 2 1 6			5 216								
Total: John Taolo Gaetsewe Municipalities	35 000	08 200		35 000	008 89		5 216			5 216			4 000	000 6	10 000	4 000	000 6	10 000
Total: Northern Cape Municipalities	149 575	199 500	204 775	149 575	199 500	204 775	5 216			5 216			4 000	0006	10 000	4 000	0006	10 000
			-	-	-	-			-						-			

ANNEXURE W6
ALLOCATIONS-IN-KIND TO MUNICIPALITIES (SCHEDULE 7) 1 OF 3

		Regi	Regional Bulk Infrastructure Grant	astructure Gr	ını			Water	Water Services Onerating Subsidy	rating Subsig	J.			Rural	Households In	Rural Households Infrastructure Grant	Cant	
	Natio	National Financial Year	Year	Munici	Municipal Financial Year	Year	Nation	National Financial Year	ear	Municin	Municipal Financial Year	Year	Nation	National Financial Year	Year	Municit	Municipal Financial Year	ear
Category Municipality	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	2010/11		2012/13	2010/11	2011/12	2012/13	2010/11	2011/12	2012/13	2010/11	2011/12	2012/13
- 1	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R 000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R 000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)
NORTH WEST																		
B NW371 Moretele													4 000	000 6	8 000	4 000	000 6	8 000
	49 000	52 900	000 09	49 000	52 900	000 09							4 000	000 6	8 250	4 000	0006	8 250
		000 01			900	000							4 000	11 000	10 000	4 000	11 000	10 000
C DC5/ Bojanala Platinum District Municipality		18 000			18 000	72 000	1	+	1	1	1	1						
Total: Bojanala Platinum Municipalities	49 000	70 900	82 000	49 000	70 900	82 000							12 000	29 000	26 250	12 000	29 000	26 250
B NW381 Ration		14 000	15 000		14 000	15 000							4 000	7 000	10 000	4 000	2 000	10.000
COCIMIN					000 +1	000 21							6	000	200	-	000	200.01
B NW382 Iswamg																		
B NW385 Ramotshere Moiloa													4 000	7 000	10 000	4 000	7 000	10 000
C DC38 Ngaka Modiri Molema District Municipality			10 000			10 000												
Total: Ngaka Modiri Molema Municipalities		14 000	25 000		14 000	25 000							8 000	14 000	20 000	8 000	14 000	20 000
													4 000	7 000	10 000	4 000	2 000	10 000
													4 000	7 000	10 000	4 000	7 000	10 000
C DC39 Dr Ruth Segomotsi Mompati District Municipality	65 000	80 000		65 000	80 000	80 000												
Total: Dr Ruth Segomotsi Mompati Municipalities	000 59	80 000	80 000	000 59	80 000	80 000							8 000	14 000	20 000	8 000	14 000	20 000
		000			000	000												
N W401		10 000	000 OT		10 000	10 000												
B NW402 Tlokwe																		
B NW403 City of Matlosana																		
C DC40 Dr Kenneth Kaunda District Municipality																		
Total: Dr Kenneth Kaunda Municipalities		10 000	10 000		10 000	10 000												
Total: North West Municipalities	114 000	174 900	200 000	114 000	174 900	200 000							28 000	27 000	057 99	000 86	27 000	050 99

ANNEXURE W6
ALLOCATIONS-IN-KIND TO MUNICIPALITIES (SCHEDULE 7) 1 OF 3

		Regi	Regional Bulk Infrastructure Grant	structure Gra	nt			Water Services Operating Subsidy	Operating Su	bsidy			Rura	Rural Households Infrastructure Grant	Infrastructure	Grant	
	Natio	National Financial Year	Year	Munici	Municipal Financial Year	Year	National Financial Year	ancial Year		Municipal Financial Year	ıl Year	Natio	National Financial Year	Year	Munic	Municipal Financial Year	Year
Category Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R 000)	2010/11 2011/12 (R'000) (R'000)	/12 2012/13 00) (R'000)	2010/11 (R'000)	2011/12 (R'000)	2012/13 (R 000)	2010/11 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2010/11 (R'000)	2011/12 (R'000)	2012/13 (R'000)
WESTERN CAPE																	
A CPT City of Cape Town																	
B WC011 Matzikama																	
	20 000			20 000													
B WC013 Bergrivier																	
WC014																	
C DC1 West Coast District Municipality	5 257	12 000	53 000	5 257	12 000	53 000											
Total: West Coast Municipalities	25 257	12 000	23 000	25 257	12 000	53 000											
B WC022 Witzenberg	17 000		900	17 000		000											
WC023	12,000		19 000	12 000		10000											
WC025			40 000			40 000											
B WC026 Langeberg	3,000			3 000													
ΙË	32 000		28 000	32 000		28 000											
vous cape whenever parties	200		200	000		000											
WC031																	
B WC032 Overstrand																	
		3 000	12 000		3 000	12 000											
C DC3 Overberg District Municipality	1 500			1 500													
Total: Overberg Municipalities	1 500	3 000	12 000	1 500	3 000	12 000											
Landa Wood of		11	000		11 300	9											
WC042		14 895	000 00		11 805	00000											
WC043			2		200	2											
	8 500			8 500													
B WC047 Bitou																	
D WC046 Knysna C DC4 Eden District Municipality																	
Total: Eden Municipalities	8 500	26 202	25 599	8 200	26 292	25 599											
B WC051 Laingsburg																	
WC052																	
m		15 000	7 000		15 000	7 000											
C DCS Central Karoo District Municipality		000			000												
Total: Central Karoo Municipalities		15 000	7 000		15 000	7 000											
Total: Western Cape Municipalities	67 257	26 2 3 2	155 599	67 257	26 292	155 599											
Unallocated:																	
					- 1												
National Total	1 704 140	2 003 217	2 176 274	1 704 140	2 003 217	2 176 274	99 935		99 935	w		231 500	479 500	517 250	231 500	479 500	517 250

ALLOCATIONS-IN-KIND TO MUNICIPALITIES (SCHEDULE 7) 2 OF 3

		Integr	ated National I	Integrated National Electrification Programme (Eskom) Grant	Programme (	Eskom) Gra	ıt	Elec	tricity Dema	and Side Ma	Electricity Demand Side Management (Eskom) Grant	Eskom) Gra	nt
		Nation	National Financial Year	ear	Munici	Municipal Financial Year	Year	Nation	National Financial Year	Year	Munici	Municipal Financial Year	l Year
Category	Municipality	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	20/1/12	2012/13	2013/14
0		(8,000)	(R000)	(R.000)	(8.000)	(K'000)	(K'000)	(K'000)	(R.000)	(R'000)	(R'000)	(R000)	(R.000)
EASTERN CAPE													
	;	,	,										
A BUF A NMA	Buffalo City Nelson Mandela Bay	9 9 9 4	0 400	/ 466	9 964	0 400	/ 400						
	Camdeboo	62		28	62		28						
	Blue Crane Route												
	Ikwezi	79		36	79		36						
	Makana	4 236	1 693	2 695	4 236	1 693	2 695						
	Ndlambe	5 045	2 049	3 225	5 045	2 049	3 225						
	Sundays River Valley	1 901	720	1 191	1 901	720	1 191						
B EC107	Baviaans	29		30	29		30						
B EC108	Kouga	52		24	52		24						
B EC109	Kou-kamma	2 970	1 255	1 921	2 970	1 255	1 921						
C DC10	Cacadu District Municipality												
Total: Cacadu Municipalities	palities	14 412	5717	9 149	14 412	5 717	9 149						
B EC121	Mbhashe	8 210	3 421	5 287	8 210	3 421	5 287						
B EC122	Mnguma	63 549	32 393	43 610	63 549	32 393	43 610						
B EC123	Great Kei	7 031	2 921	4 524	7 031	2 921	4 524						
B EC124	Amahlathi	868 9	8 611	7 050	868 9	8 611	7 050						
B EC126	Ngqushwa	4 671	1 940	3 005	4 671	1 940	3 005						
B EC127	Nkonkobe	4 4 1 2	2 059	2 941	4 412	2 059	2 941						
B EC128	Nxuba												
C DC12	Amathole District Municipality												
Total: Amathole Municipalities	cipalities	94 772	51 344	66 416	94 772	51 344	66 416						
	Inxuba Yethemba	343		156	343		156						
	Tsolwana	3 685	1 544	2 377	3 685	1 544	2 377						
	Inkwanca	234		107	234		107						
	Lukhanji	273		124	273		124						
	Intsika Yethu	21 301	18 229	17 968	21 301	18 229	17 968						
	Emalahleni	8 128	7 047	868 9	8 128	7 047	868 9						
	Engcobo	4 638	2 978	3 403	4 638	2 978	3 403						
B EC138	Sakhisizwe	5 382	2 2 5 9	3 473	5 382	2 2 2 5 9	3 473						
C DC13	Chris Hani District Municipality												
Total: Chris Hani Municipalities	nicipalities	43 984	32 057	34 506	43 984	32 057	34 506						
	Elundini	19 121	10 581	13 468	19 121	10 581	13 468						
	Sengu	7 316	3 044	4 709	7 316	3 044	4 709						
	Maletswai	74		33	74		33						
B EC144	Gariep	1 039		472	1 039		472						
C DC14 Ukhahlamba	Ukhahlamba District Municipality	200	20, 67	00,04	25.1	20, 67	60, 64						
TOTAL: OKHAIHAIHDA IME	unicipanues	71.349	C70 CI	000 01	71.349	C70 CI	000 01						

AND ALLOCATIONS-IN-KIND TO MUNICIPALITIES (SCHEDULE 7) 2 OF 3

		Integra	Integrated National Electrification Programme (Eskom) Grant	Jectrification	Programme (	Eskom) Grai	=	Flec	глсту Бет	and Side M	Electricity Demand Side Management (Eskom) Grant	ESKOM) Gra	1
		Nation	National Financial Year	ear	Munici	Municipal Financial Year	Year	Nation	National Financial Year	Year	Munici	Municipal Financial Year	ıl Year
2	Manipipalita	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14
Category	Municipanty	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)
B EC	EC153 Ngquza Hill	28 775	29 338	26 338	28 775	29 338	26 338						
B EC	EC154 Port St Johns	6 657	3 971	4 831	6 657	3 971	4 831						
B EC	EC155 Nyandeni	33 541	16 219	22 613	33 541	16 219	22 613						
B EC	EC156 Mhlontlo	191.6	9 490	8 753	1916	9 490	8 753						
B EC	EC157 King Sabata Dalindyebo	14 139	11 387	11 602	14 139	11 387	11 602						
C DC	DC15 OR Tambo District Municipality												
Total: O R.Tamb	Fotal: O R.Tambo Municipalities	92 878	70 404	74 138	92 878	70 404	74 138						
B EC.	EC441 Matatiele	48 961	42 107	41 218	48 961	42 107	41 218						
B EC.	EC442 Umzimvubu	177 071	154 438	147 992	177 071	154 438	147 992						
B EC.	EC443 Mbizana	53 838	138 031	82 300	53 838	138 031	82 300						
B EC	EC152 Ntabankulu	1 326	14 316	6 507	1 326	14 316	6 507						
C DC	DC44 Alfred Nzo District Municipality												
Total: Alfred Nzo Municipalities	Municipalities	281 195	348 893	278 018	281 195	348 893	278 018						
Total: Eastern Ca	Total: Eastern Cape Municipalities	564 755	528 500	488 375	564 755	528 500	488 375						

ANNEXURE W6
ALLOCATIONS-IN-KIND TO MUNICIPALITIES (SCHEDULE 7) 2 OF 3

	Int	Integrated National Electrification Programme (Eskom) Grant	l Electrification	Programme	Eskom) Gra	ıt	Elect	ricity Dema	nd Side Ma	Electricity Demand Side Management (Eskom) Grant	Eskom) Gra	ıı
	Na	National Financial Year	Year	Munici	Municipal Financial Year	Year	Nationa	National Financial Year	Year	Municip	Municipal Financial Year	l Year
Category Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
FREE STATE												
A MAN Mangaung	9	903 821	862	903	821	862						
B FS161 Letsemeng		41	21	41		21						
FS162		82	94	187		94						
B FS163 Mohokare B FS164 Naledi		41	21	41		21						
DC16												
Total: Xhariep Municipalities	2	569	135	569		135						
B FS181 Masilonyana		2 245	1 123		2 245	1 123						
FS182	1	123		123		62						
FS183	1 831			1831	674	1 252						
FS184	11 764	9 2 1 7	10	11 764	9 217	10 491						
B FS185 Nala		246	123	246		123						
[	13 965	65 12 136	13 050	13 965	12 136	13 050						
FS191	2 097	97 2 049	2 073	2 097	2 049	2 073						
B FS192 Ulniabeng				/ 401	5 824	2 628						
FS194		83	41	82		41						
FS195												
C DC19 Thabo Mofutsanyana District Municipality												
Total: Thabo Mofutsanyana Municipalities	9 640	40 5 903	7 7 7 7 2	9 640	5 903	7 772						
B FS201 Moqhaka		82	41	82		41						
FS203			i			i						
FS204		41	21	41		21						
B F3203 Matube C DC20 Fezile Dabi District Municipality												
ΙBΙ	1	123	62	123		62						
Total: Free State Municipalities	24 900	00 18 859	21 880	24 900	18 859	21 880						

ANNEXURE W6
ALLOCATIONS-IN-KIND TO MUNICIPALITIES (SCHEDULE 7) 2 OF 3

			Integra	ated National l	Integrated National Electrification Programme (Eskom) Grant	Programme ()	Eskom) Gran	ıt	Elect	ricity Dema	and Side Ma	Electricity Demand Side Management (Eskom) Grant	Eskom) Gra	ınt
			Nation	National Financial Year	ear	Municip	Municipal Financial Year	Year	Nationa	National Financial Year	Year	Municip	Municipal Financial Year	d Year
3,5		10101	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14
Category		Municipanty	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)
GAUTENG														
A	EKU Ekurl	Ekurhuleni	92 070	101 117	96 593	92 070	101 117	96 593						
A		City of Johannesburg	31 996	31 996	31 996	31 996	31 996	31 996						
A T	TSH City of	City of Tshwane	5 701	9 145	7 423	5 701	9 145	7 423						
	GT421 Emfuleni	ıleni	5 137	5 137	5 137	5 137	5 137	5 137						
B G	TY22 Midv	/aal												
B		di												
C D	DC42 Sedib	Sedibeng District Municipality							54 400			54 400		
Total: Sedibeng Municipalities	g Municipaliti	ies	5 137	5 137	5 137	5 137	5 137	5 137	54 400			54 400		
<u>r</u>	GT481 Moga	Monale City	896 0	8966	9926	9768	976	9926						
. E		Randfontein	85	82	82	82	82	82						
B	GT483 West	onaria												
<u>B</u>		Merafong City												
C D		West Rand District Municipality												
<b>Total: West Rand Municipalities</b>	nd Municipal	lities	9 350	9 350	9 350	9 350	9 350	9 350						
Total: Gauteng Municipalities	Municipalitie	es	144 255	156 745	150 500	144 255	156 745	150 500	54 400			54 400		

ANNEXURE W6
ALLOCATIONS-IN-KIND TO MUNICIPALITIES (SCHEDULE 7) 2 OF 3

	Integra	ated National 1	Integrated National Electrification Programme (Eskom) Grant	Programme (	Eskom) Grar	ıt	Elect	ricity Dema	ind Side Ma	Electricity Demand Side Management (Eskom) Grant	Skom) Gra	nt
	Nation	National Financial Year	ear	Munici	Municipal Financial Year	Year	Nations	National Financial Year	Year	Municip	Municipal Financial Year	l Year
Category Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
KWAZULU-NATAL												
A ETH eThekwini	10 885	11 736	15 317	10 885	11 736	15 317						
B KZN211 Vulamehlo												
KZN212	13 209	13 209	19 570	13 209	13 209	19 570						
B KZN213 Umzumbe B KZN214 uMuziwahantu	1 340	1 914	2.816	1 340	1 914	2.816						
KZN215	4 777		)	4 777								
B KZN216 Hibiscus Coast C DC21 Ugu District Municipality	5 852	4 269	6 130	5 852	4 269	6 130						
Total: Ugu Municipalities	25 717	19 393	28 515	25 717	19 393	28 515						
B KZN221 uMshwathi		1 026	160		1 026	160						
KZN222		1 106	819		1 106	819						
KZN223												
B KZN224 Impendle B KZN235 Mennduzi	103	2276	3 372	103	2 276	3 372						
KZN226	25 573	25 404	38 186	25 573	25 404	38 186						
KZN227	3 122	3 122	4 625	3 122	3 122	4 625						
C DC22 uMgungundlovu District Municipality												
Total: uMgungundlovu Municipalities	31 210	32 934	47 762	31 210	32 934	47 762						
B KZN232 Emnambethi-Ladysmith		33 262	24 639		33 262	24 639						
KZN233												
KZN234		413	306		413	306						
B KZN235 Okhahlamba B KZN236 Imbabazane	517			517								
DC23												
Total: Uthukela Municipalities	42 432	33 675	24 944	42 432	33 675	24 944						
B KZN241 Endumeni												
B KZN242 Nguthu	4 953	9 157	10 174	4 953	9 157	10 174						
KZN244	19 147	14 813	25 156	19 147	14 813	25 156						
B KZN245 Umvoti C DC24 Ilmzinvarhi District Municinality	569	269	843	569	995	843						
٤	24.670	24 530	36 173	04 670	24 530	36 173						
commission of the commission o												
	9 9	9 200	9 630	9 200	9 200	9 630						
KZN253												
Total: Amajuba Municipalities	005 9	9 200	9 630	9 200	9 200	9 630						

ANNEXURE W6
ALLOCATIONS-IN-KIND TO MUNICIPALITIES (SCHEDULE 7) 2 OF 3

	Integ	Integrated National Electrification Programme (Fskom) Grant	Electrification	Programme (	Eskom) Grai	1	Elect	Electricity Demand Side Management (Eskom) Grant	d Side Ma	nagement (E	Skom) Gra	ı
	Natio	National Financial Year	ear	Munici	Municipal Financial Year	Year	Nationa	National Financial Year	ear	Municip	Municipal Financial Year	l Year
Category Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
1001 NEAR												
KZNZ61	33			135								
KZN262	16			16		,						
KZN263	428		6 364	428		6 364						
	23 886	8 053	25 585	23 886	8 053	25 585						
B KZN266 Ulundi	5 794	1 864	1 381	5 794	1 864	1 381						
C DC26 Zululand District Municipality		4 203	5 672		4 203	5 672						
Total: Zululand Municipalities	30 294	14 119	39 001	30 294	14 119	39 001						
B KZN271 Umhlabuyalingana		542	401		542	401						
B KZN272 Jozini	262		16 467	262		16 467						
B KZN273 The Big 5 False Bay			10 184			10 184						
B KZN274 Hlabisa			16 201			16 201						
B KZN275 Mtubatuba	23 037	18 791	27 838	23 037	18 791	27 838						
C DC27 Umkhanyakude District Municipality		4 203	5 672		4 203	5 672						
Total: Umkhanyakude Municipalities	23 299	23 535	76 763	23 299	23 535	76 763						
B KZN281 Mfolozi	672		966	672	672	966						
B KZN282 uMhlathuze	10 616	48 802	41 977	10 616	48 802	41 977						
KZN283												
B KZN284 Umlalazi	49 773		36 150	49 773		36 150						
,	298			298								
C DC28 uThungulu District Municipality		4 203	5 672		4 203	5 672						
Total: uThungulu Municipalities	61 359	53 677	84 794	61 359	53 677	84 794						
		105	0.50		105	020						
NZN291		001	600		001	900						
KZN292	1 411	2 822	3 136	1411	2 822	3 136						
KZN293	55 332	89 640	107 196	55 332	89 640	107 196						
_	4 234	4 690	6 401	4 234	4 690	6 401						
C DC29 iLembe District Municipality		4 203	5 672		4 203	5 672						
Total: iLembe Municipalities	226 09	101 840	122 763	226 09	101 840	122 763						
LCVINZA	00			100								
NZIN431	970	700	190	100	.,,	130.01						
KZN432	/ 364	/ 761	/5/.01	/ 364	/ 761	10.72/						
KZN433	329	57	45	329	57	42						
KZN434	72 403	79 927	112 837	72 403	79 927	112 837						
B KZN435 Umzimkhulu	24 725	63 012	161 791	24 725	63 012	161 191						
C DC43 Sisonke District Municipality	8 910	8 910	13 200	8 910	8 910	13 200						
Total: Sisonke Municipalities	114 624	159 167	204 633	114 624	159 167	204 633						
Total: KwaZulu-Natal Municipalities	431 967	481 114	690 297	431 967	481 114	690 297						
	-				-		-					

ALLOCATIONS-IN-KIND TO MUNICIPALITIES (SCHEDULE 7) 2 OF 3

	Integ	rated National	Integrated National Electrification Programme (Eskom) Grant	Programme (	Eskom) Grai	ıt	Elec	tricity Dema	and Side Ma	Electricity Demand Side Management (Eskom) Grant	Eskom) Gra	nt
	Natio	National Financial Year	'ear	Munici	Municipal Financial Year	Year	Nations	National Financial Year	Year	Municiț	Municipal Financial Year	l Year
Category Municipality	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	20/11/12	2012/13	2013/14
	(R000)	(R000)	(R'000)	(K000)	(R.000)	(R'000)	(R'000)	(R.000)	(R'000)	(R.000)	(R000)	(R'000)
LIMPOPO												
LIM331	5 204	3 529	4 367	5 204	3 529	4 367						
LIM332	10 043	41 084	25 563	10 043	41 084	25 563						
B LIM333 Greater Tzaneen	9 404	6 135	7 770	9 404	6 135	7 770						
B LIM334 Ba-Phalaborwa	3 121	2 223	2 672	3 121	2 223	2 672						
B LIM335 Maruleng	3 858	2 494	3 176	3 858	2 494	3 176						
C DC33 Mopani District Municipality												
Total: Mopani Municipalities	31 629	55 466	43 548	31 629	55 466	43 548						
R IM341 Musina												
THE CANAL	i i	ī			i		000			000		
LIM342	17 021	7 104	12 063	17 021	7 104	12 063	54 400			24 400		
LIM343	14 157	42 163	28 160	14 157	42 163	28 160						
LIM344	6 227	58 766	32 496	6 227	58 766	32 496						
C DC34 Vhembe District Municipality												
Total: Vhembe Municipalities	37 405	108 033	72 719	37 405	108 033	72 719	54 400			54 400		
LIM351	9 168	6 924	8 046	9 168	6 924	8 046						
LIM352		5 154	2 577		5 154	2 577						
B LIM353 Molemole												
B LIM354 Polokwane	27 953	19 896	23 924	27 953	19 896	23 924						
B LIM355 Lepelle-Nkumpi	1 270	10 810	6 040	1 270	10 810	6 040						
C DC35 Capricom District Municipality	684	2 736	1 710	684	2 736	1 710						
Total: Capricorn Municipalities	39 075	45 519	42 297	39 075	45 519	42 297						
B LIM361 Thabazimbi												
B LIM362 Lephalale	4 195	6 047	5 121	4 195	6 047	5 121						
LIM364												
LIM365												
B LIM366 Bela Bela	171		98	171		98						
B LIM367 Mogalakwena	13 556	9 378	11 467	13 556	9 378	11 467						
C DC36 Waterberg District Municipality	513	912	713	513	912	713						
Total: Waterberg Municipalities	18 436	16 337	17 386	18 436	16 337	17 386						
R 1 IM471 Enhraim Mogale	6 053	1 589	3 821	6.053	1 589	3 821						
CLVM1 1	V 55A	2518	3 5 3 6	7 554	2518	3 536						
LIM4/2	17 030	12 615	728 71	17 030	12 615	14 827						
14. MILL	0.03	C10 71	14027	17 039	C10 71	14041						
LIM4/4	9 359	4 209	6 784	9359	4 209	6 784						
_	21 019	33 555	787.78	21 019	33 555	787.78						
C DC47 Greater Sekhukhune District Municipality												
Total: Greater Sekhukhune Municipalities	58 024	54 486	56 255	58 024	54 486	56 255						
Total: Limpopo Municipalities	184 569	279 841	232 205	184 569	279 841	232 205	54 400			54 400		

ANNEXURE W6
ALLOCATIONS-IN-KIND TO MUNICIPALITIES (SCHEDULE 7) 2 OF 3

		Integr	rated National	Integrated National Electrification Programme (Eskom) Grant	Programme (	Eskom) Grai	ıt	Elect	tricity Dema	ınd Side Ma	Electricity Demand Side Management (Eskom) Grant	Eskom) Gra	ınt
		Natio	National Financial Year	/ear	Munici	Municipal Financial Year	Year	Nations	National Financial Year	Year	Munici	Municipal Financial Year	ıl Year
Category	Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
MPUMALANGA													
	•	19 618	14 490	17 054	19 618	14 490	17 054						
B MP302 B MP303	Msukaligwa Mkhondo	9 713	4 988	7 351	9 713	4 988	7 351						
B MP304													
		934	1 231	1 082	934	1 231	1 082						
B MP307	Govan Mbeki Gart Sihanda District Municipality												
Total: Gert Sibande Municinalities	Municipalities	30.05	20 710	25 487	30.05	20 710	25 487						
	•												
B MP311	Victor Khanye	205		103	205		103						
B MP312	Emalahleni												
B MP313	Steve Tshwete	4 161	3 010	3 585	4 161	3 010	3 585						
B MP314	Emakhazeni												
B MP315	Thembisile	5 586	6 384	5 985	5 586	6 384	5 985						
B MP316	Dr JS Moroka	1 789	1 269	1 529	1 789	1 269	1 529						
C DC31	Nkangala District Municipality	1 508	1 550	1 529	1 508	1 550	1 529						
Total: Nkangala Municipalities	nicipalities	13 248	12 213	12 730	13 248	12 213	12 730						
B MP321	Thaba Chweu												
B MP322	Mbombela	8 789	11 026	806 6	8 789	11 026	806 6						
B MP323	Umjindi												
B MP324	Nkomazi	42 106	22 230	32 168	42 106	22 230	32 168						
B MP325	Bushbuckridge	17 076	40 064	28 570	17 076	40 064	28 570						
C DC32	Ehlanzeni District Municipality												
Total: Ehlanzeni Municipalities	nicipalities	67 971	73 320	70 646	67 971	73 320	70 646						
Total: Mpumalanga Municipalities	Municipalities	111 484	106 243	108 864	111 484	106 243	108 864						

ALLOCATIONS-IN-KIND TO MUNICIPALITIES (SCHEDULE 7) 2 OF 3

		Integr	ated National	Integrated National Electrification Programme (Eskom) Grant	Programme	Eskom) Gra	nt	Elec	tricity Demai	nd Side Ma	Electricity Demand Side Management (Eskom) Grant	skom) Gra	nt
		Natio	National Financial Year	/ear	Munici	Municipal Financial Year	Year	Nation	National Financial Year	Year	Municipa	Municipal Financial Year	Year
Category	Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
NORTHERN CAPE	я												
B NC061	51 Richtersveld												
B NC060	ob Karoo Hoogiand	1 204	18 603	0000	1 204	18 603	0 007						
		†67 I	660 01	4666	1 294	10 073	7 774						
Total: Namakwa Municipalities	lunicipalities	1 294	18 693	9 994	1 294	18 693	9 994						
100N	71 Thursts	33	33	33	33	33	33						
B NC072		1310	2152	1 731	1 310	21.5	1 731						
		935	935	935	935	2 132	935						
		,	,	)			,						
B NC076		82		41	82		41						
	77 Siyathemba	75	496	285	75	496	285						
B NC078		1 195	1113	1 154	1 195	1 113	1 154						
C DC7	Pixley Ka Seme District Municipality												
Total: Pixley Ka Seme Municipalities	me Municipalities	3 630	4 730	4 180	3 630	4 730	4 180						
B NC081	51 Ivner 22 IKai IGarik	115	114		115	114	114						
		787	714	750	787	714	750						
		115	114		115	114	114						
B NC085	35 Tsantsabane	2 140	2 058	2	2 140	2 058	2 099						
Į													
C DC8 Siyand	Siyanda District Municipality	3.156	3,000	3 070	3.156	3,000	2.070						
10tal: Siyanua Mili	шеграниея	OCT C	0000 C	3010	OCT C	2000	0/00						
B NC091	11 Sol Plaatje							5 000			2 000		
		200		350	700		350						
		11 693	11 693	11 693	11 693	11 693	11 693						
B NC094		20 641	20 558	20 599	20 641	20 558	20 599						
SOC I	Frances Baard District Municipality	10000	01000	07,00	, 0000	0.00	017.00	000			000		
Total: Frances Baard Municipalities	rd Municipalities	33 034	757.75	32 643	55 034	757 75	32 643	9000 c			000 c		
B NC451	11 Moshaweng	7 403	7 403	7 403	7 403	7 403	7 403						
		8 053	8 053	8 053	8 053	8 053	8 053						
C DC45	5 John Taolo Gaetsewe District Municipality	1 524	5 645	3 585	1 524	5 645	3 585						
Total: John Taolo (	Total: John Taolo Gaetsewe Municipalities	16 981	21 102	19 041	16 981	21 102	19 041						
Total: Northern Cape Municipalities	ne Municipalities	58 094	777 67	68 935	58 094	777 6L	68 935	5 000			2 000		
						:							

AND ALLOCATIONS-IN-KIND TO MUNICIPALITIES (SCHEDULE 7) 2 OF 3

	Integ	rated National	Integrated National Electrification Programme (Eskom) Grant	Programme (	Eskom) Grai	ı	Elec	tricity Dema	and Side Ma	Electricity Demand Side Management (Eskom) Grant	Eskom) Gra	ııt
	Natio	National Financial Year	/ear	Municil	Municipal Financial Year	Year	Nation	National Financial Year	Year	Munici	Municipal Financial Year	l Year
Category Municipality	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14 (R'000)	2011/12	2012/13 (R'000)	2013/14	2011/12	2012/13	2013/14 (R'000)
	(000 W)	(222 11)	(000 W)	(000 W)	(000 11)	(000 11)	(moon)	(000 11)	(11000)	(000 11)	(000 W)	(000 11)
NORTH WEST												
B NW371 Moretele	6 752	4 193	5 473	6 752	4 193	5 473						
B NW372 Madibeng	12 410	9 0 2 6	10 718	12 410	9 0 5 0	10 718						
B NW373 Rustenburg	5 661	4 697	5 179	5 661	4 697	5 179						
B NW374 Kgetlengrivier												
NW375	13 421	10 041	11 731	13 421	10 041	11 731						
C DC37 Bojanala Platinum District Municipality												
Total: Bojanala Platinum Municipalities	38 245	27 957	33 101	38 245	27 957	33 101						
B NW381 Ration	4 802	6199	5 501	4 802	6 1 9 9	5 501						
NW382	3 242	2 594	2 918	3 242	2 594	2 918						
NW383	29 151	38 178	33 664	29 151	38 178	33 664						
B NW384 Ditsobotla	960 6	11 654	10 375	960 6	11 654	10 375						
B NW385 Ramotshere Moiloa	5 865	4 692	5 279	5 865	4 692	5 279						
C DC38 Ngaka Modiri Molema District Municipality												
Total: Ngaka Modiri Molema Municipalities	52 156	63 317	57 736	52 156	63 317	57 736						
B NW392 Naledi	2 495	1 996	2 245	2 495	1 996	2 245	5 000			5 000		
B NW393 Mamusa	41	33	37	41	33	37						
B NW394 Greater Taung	5 902	9 189	7 546	5 902	9 189	7 546						
B NW396 Lekwa-Teemane	4 211	3 369	3 790	4 211	3 369	3 790						
	8 837	7 070	7 954	8 837	7 070	7 954						
C DC39 Dr Ruth Segomotsi Mompati District Municipality												
Total: Dr Ruth Segomotsi Mompati Municipalities	21 487	21 656	21 572	21 487	21 656	21 572	5 000			2 000		
B NW401 Ventersdorp												
B NW402 Tlokwe												
B NW403 City of Matlosana	5 166	4 133	4 650	5 166	4 133	4 650						
_	10 467	8 373	9 420	10 467	8 373	9 420						
C DC40 Dr Kenneth Kaunda District Municipality												
Total: Dr Kenneth Kaunda Municipalities	15 633	12 507	14 070	15 633	12 507	14 070						
Total: North West Municipalities	127 521	125 436	126 479	127 521	125 436	126 479	2 000			2 000		

ALLOCATIONS-IN-KIND TO MUNICIPALITIES (SCHEDULE 7) 2 OF 3

		Integr	ated National	Integrated National Electrification Programme (Eskom) Grant	Programme (	Eskom) Grai	ıt	Elect	tricity Demai	nd Side Ma	Electricity Demand Side Management (Eskom) Grant	Skom) Gra	nt
		Nation	National Financial Year	ear	Munici	Municipal Financial Year	Year	Nationa	National Financial Year	Year	Municip	Municipal Financial Year	l Year
Category	Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
WESTERN CAPE													
A CPT	City of Cape Town	68 892	84 091	76 623	68 892	84 091	76 623						
B WC011	11 Matzikama	115	91	103	115	91	103						
B WC012		853	822	838	853	822	838						
		115	121	118	115	121	118						
> >		3 644	2 915	3 279	3 644	2 915	3 279						
Total: West Coast Municipalities	West Coast District Municipality Municipalities	4 7 2 6	3 950	4 338	4 726	3 950	4 338						
B WC022	22 Witzenhero	115	202	158	115	200	158						
B WC023		115	209	162	115	209	162						
		1 916	2 313	2 115	1916	2 313	2 115						
B WC026		115	285	200	115	285	200						
Total. Cana Winele	Cape winelands District Municipanty	03.0	3.010	3636	0300	3.010	3636						
Total: Cape Winelands Municipanties	nds Municipanues	7 700	3 010	2 0 2	7 700	2 010	CC0 7						
		3 489	2 789	3 139	3 489	2 789	3 139						
			o o			9							
B wC034	34 Swellendam Overhere District Municipality	cII	986	106	cII	86	106						
otal: Overber	unicipalities	3 604	2 887	3 246	3 604	2 887	3 246						
B WC041	11 Kannaland												
		358	286	322	358	286	322						
		5 801	4 640	5 220	5 801	4 640	5 220						
		4 048	3 331	3 689	4 048	3 331	3 689						
B WC047	1/ Bitou 18 Knyena												
Total: Eden Municipalities	palities	10 206	8 257	9 231	10 206	8 257	9 231						
B WC051	51 Laingsburg	115	200	157	115	200	157						
B WC053	53 Beaufort West Control Konco Dietrict Municipality	463	3 148	1 806	463	3 148	1 806						
otal: Central	oo Municipalities	578	3 348	1 963	578	3 348	1 963						
	•												
Total: Western Cape Municipalities	be Municipalities	90 265	105 543	98 036	90 265	105 543	98 036						
Unallocated:													
National Total		1 737 812	1 882 057	1 985 570	1 737 812	1 882 057	1 985 570	118 800			118 800		

ALLOCATIONS-IN-KIND TO MUNICIPALITIES (SCHEDULE 7) 3 OF 3

			Neighbourl	hood Develop	Neighbourhood Development Partnership Grant	np Grant				SUB-TOTAL: INDIRECT	: INDIRECT		
		Nation	National Financial Year	Year	Munici	Municipal Financial Year	Year	Natio	National Financial Year	ear	Munici	Municipal Financial Year	ar
Category	ry Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
EASTERN CAPE	CAPE												
A BUF		4 000	4 000	3 000	4 000	4 000	3 000	13 964	10 460	10 466	13 964	10 460	10 466
A NMA	1A Nelson Mandela Bay	2 300	1 800			1 800	1 300	2 300	1 800		2 300	1 800	1 300
	_							62		28	62		28
B EC102								1		00	ç		0
	103 IKWezi	005	200	001	200	00%	901	6/	2 193	30	736	2 103	30
B EC105		995	000			000	801	5 045	2 049		5 045	2 049	3 225
								17 901	9 020		17 901	9 020	12 191
								19		30	29		30
B EC108	108 Kouga							52	1 255	24	22	1 255	1 02 1
								7 000	17 000		7 000	17 000	31 000
Total: Cac	Total: Cacadu Municipalities	200	200	100	200	200	100	37 912	31 517	51 249	37 912	31 517	51 249
B EC121	121 Mbhashe							12 710	16 421	15 287	12 710	16 421	15 287
								68 049	45 393		68 049	45 393	53 610
								11 531	11 921		11 531	11 921	14 524
								11 398	17 611		11 398	17 611	17 050
B EC126	126 Ngqusnwa 127 Nkonkobe							46/1	2 059	3 003	4 6/1	2 059	3 003 2 941
								7	600 4	1	7	001	1 7
C DC12	12 Amathole District Municipality	3 000	3 000	3 000	3 000	3 000	3 000	58 700	36 200	13 000	58 700	36 200	13 000
Total: Am.	Total: Amathole Municipalities	3 000	3 000	3 000	3 000	3 000	3 000	171 472	131 544	119 416	171 472	131 544	119 416
B EC131	131 Inxuba Yethemba							343		156	343		156
								3 685	1 544	2	3 685	1 544	2 377
B EC133	,							234			234		107
		1 000	1 000	1 000	1 000	1 000	1 000	5 773	8 000		5 773	8 000	5 124
B EC135	135 Intsika Yethu	1 000	OUS	002		008	005	21 301	18 229	17 968	21 301	18 229	17 968
		-	2			8	3	9 138	11 978		9 138	11 978	13 403
B EC138								5 382	2 259		5 382	2 2 2 5 9	3 473
C DC13	13 Chris Hani District Municipality							112 000	179 350		112 000	179 350	210 000
Total: Chr	Total: Chris Hani Municipalities	2 000	1 800	1 700	2 000	1800	1 700	166 984	229 207	260 206	166 984	229 207	260 206
B EC141								23 621	19 581	23 468	23 621	19 581	23 468
		400	236		400	236		12 2 16	12 280	14 709	12 216	12 280	14 709
B EC143	143 Maletswai							74		33	74		33
								601		7	600 1		†
Total: Ukb	Total: Ukhahlamba Municipalities	400	236		400	236		36 949	31 861	38 683	36 949	31 861	38 683

ANNEXURE W6 ALLOCATIONS-IN-KIND TO MUNICIPALITIES (SCHEDULE 7) 3 OF 3

		Neiohhour	hood Develon	Neighbourhood Develonment Partnershin Grant	hin Grant				SIR-TOTAL INDIRECT	· INDIRECT		
	Notional	onel Financial Veer	Voor	Munic	Municipal Financial Vear	Voor	Notio	National Financial Voor	- Coor	Munic	Municipal Financial Voor	
	C1/11/0C	21/01/00	2012/14	C1/110C	21/01/00	2012/14	C1/110C	21/01/00	100000	C1/110C	6276106	2012/14
Cotonom Municipality	71/11/2	2017/13	+1/0107	2011/11/2	20177107	41/6107	2011/12	2017/102	41/0107	2011/12	C1/7107	41/6107
Category Municipanty	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)
B EC153 Ngquza Hill							28 775	29 338	26 338	28 775	29 338	26 338
B EC154 Port St Johns	1 000	1 000	400	1 000	1 000	400	7 657	4 971	5 231	7 657	4 971	5 231
B EC155 Nyandeni							38 041	25 219	30 613	38 041	25 219	30 613
B EC156 Mhlontlo							14 267	18 490	16 753	14 267	18 490	16 753
B EC157 King Sabata Dalindyebo							18 639	20 387	19 602	18 639	20 387	19 602
C DC15 OR Tambo District Municipality							4 120	9 000	10 000	4 120	0009	10 000
Total: O.R.Tambo Municipalities	1 000	1 000	400	1 000	1 000	400	111 498	104 404	108 538	111 498	104 404	108 538
B EC441 Matatiele							53 461	51 107	51 218	53 461	51 107	51 218
B EC442 Umzimvubu							181 571	163 438	155 992	181 571	163 438	155 992
B EC443 Mbizana							58 338	147 031	90 300	58 338	147 031	90 300
B EC152 Ntabankulu							3 326	18 816	15 507	3 326	18 816	15 507
C DC44 Alfred Nzo District Municipality							121 382	122 800	145 000	121 382	122 800	145 000
Total: Alfred Nzo Municipalities							418 077	503 193	458 018	418 077	503 193	458 018
Total: Eastern Cape Municipalities	13 200	12 336	9 500	13 200	12 336	9 500	959 157	1 043 986	1 047 875	959 157	1 043 986	1 047 875

ALLOCATIONS-IN-KIND TO MUNICIPALITIES (SCHEDULE 7) 3 OF 3

			Neighbour	nood Developi	Neighbourhood Development Partnership Grant	nip Grant				SUB-TOTAL: INDIRECT	: INDIRECT		
		National	nal Financial Year	Year	Munici	Municipal Financial Year	Year	Natio	National Financial Year	ear	Muni	Municipal Financial Year	ear
Category	Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
FREE STATE													
A MAN	Mangaung							903	821	862	903	821	862
	Letsemeng							41		21	41		21
FS162 FS163	Kopanong Mohokare							30 187	72 300	94	30 187	72 300	94
FS164 DC16	Naledi Xhariep District Municipality									10 000			10 000
Total: Xhariep Municipalities	<b>Junicipalities</b>							41 163	85 300	35 135	41 163	85 300	35 135
B FS181 N	Masilonyana								2 245	1 123		2 2 4 5	1 123
FS182	Tokologo							49 123	20 300	3 062		20300	3 062
B FS183	Tswelopele Matihaheng	2 300	1 400		2 300	1 400		1 831	674	1 252	1 831	10 674	1 252
FS185	Nala	1	•		1			246		123			123
Total: Lejwelepu	C DC18 Lejweleputswa District Municipality Total: Lejweleputswa Municipalities	2 300	1 400		2 300	1 400		65 265	33 836	16 050	65 265	33 836	16 050
10100								2000 6	010.0	0.00		0700	
FS191	Setsoto Dihlabeng	2 000	500	200	2 000	200	200	2 097	4 354	6 158	2 09/ 9 461	4 354	6 158
FS193	Nketoana												
FS194	Maluti a Phofung							72 862	39 400	10 041	72 862	39 400	10 041
B FS195 I	Phumelela Mantsona							13 000	000 6	25 500	13 000	000 6	10 000
DC19	Thabo Mofutsanyana District Municipality												
Total: Thabo Mo	Total: Thabo Mofutsanyana Municipalities	2 000	200	200	2 000	200	200	97 420	54 803	53 772	97 420	54 803	53 772
B FS201 1	Moqhaka							82		41	82		41
FS203	Ngwathe							2 000			2 000		
FS204	Metsimaholo							41		21	41		21
B FS205 N	Mafube												
otal: Fezile Dal	Municipalities							2 123		62	2 123		62
Total: Free State Municipalities	Municipalities	4 300	1 900	200	4 300	1900	200	206 874	174 759	105 880	206 874	174759	105 880

ALLOCATIONS-IN-KIND TO MUNICIPALITIES (SCHEDULE 7) 3 OF 3

			Neighbourh	ood Developn	Neighbourhood Development Partnership Grant	up Grant				SUB-TOTAL: INDIRECT	: INDIRECT		
		National	nal Financial Year	ear	Municil	Municipal Financial Year	Year	Natio	National Financial Year	ear	Munic	Municipal Financial Year	ar
240	M	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14
Category	Category Municipanty	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)
GAUTENG													
A EKU	Ekurhuleni	2 746	3 000	2 000	2 746	3 000	2 000	94 816	104 117	98 593	94 816	104 117	98 593
A JHB	City of Johannesburg	5 388	2 000	5 000	5 388	2 000	2 000	37 384	36 996	36 996	37 384	36 9 96	36 996
A TSH	City of Tshwane	7 500	7 900	5 750	7 500	7 900	5 750	43 631	66 045	74 173	43 631	66 045	74 173
B GT421	Emfuleni							35 137	62 137	75 137	35 137	62.137	75 137
B GT422	Midvaal											1	)
B GT423	Lesedi												
C DC42	Sedibeng District Municipality	2 800	3 500	1 763	2 800	3 500	1 763	57 200	3 500	1 763	57 200	3 500	1 763
Total: Sediben	Total: Sedibeng Municipalities	2 800	3 500	1 763	2 800	3 500	1 763	92 337	65 637	006 92	92 337	65 637	26 900
B GT481	Mogale City	3 500	4 000	1 100	3 200	4 000	1 100	12 768	13 268	10 368	12 768	13 268	10 368
B GT482	Randfontein							82	82	82	82	82	83
B GT483	Westonaria							20 000	42 500	000 09	20 000	42 500	000 09
B GT484													
C DC48	West Rand District Municipality	1 000	4 000	2 564	1 000	4 000	2 564	1 000	4 000	2 564	1 000	4 000	2 5 6 4
Total: West R.	Total: West Rand Municipalities	4 500	8 000	3 664	4 500	8 000	3 664	33 850	29 850	73 014	33 850	29 850	73 014
Total: Gauten;	Total: Gauteng Municipalities	22 934	27 400	18 177	22 934	27 400	18 177	302 019	332 645	359 677	302 019	332 645	359 677
											-		

ALLOCATIONS-IN-KIND TO MUNICIPALITIES (SCHEDULE 7) 3 OF 3

		Neighbour	hood Develop	Neighbourhood Development Partnership Grant	hip Grant				SUB-TOTAL: INDIRECT	: INDIRECT		
	Natio	National Financial Year	Year	Munici	Municipal Financial Year	Year	Natio	National Financial Year	ear	Munic	Municipal Financial Year	ar
Category Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
KWAZULU-NATAL												
A ETH eThekwini	2 800	3 500	2 000	2 800	3 500	2 000	13 685	15 236	17 317	13 685	15 236	17 317
B KZN211 Vulamehlo												
KZN212							13 209	13 209	19 570	13 209	13 209	19 570
B KZN213 Umzumbe							4 539	10 000	10 000	4 539	10 000	10 000
KZN214							1 340	1 914	2 816	1 340	1914	2 816
KZN215							4 7 7 7			4 777		
B KZN216 Hibiscus Coast C DC21 Ugu District Municipality							5 852 40 000	4 269	6 130	5 852	4 269	6 130
15							69 717	69 393	81 338	69 717	69 393	81 338
B KZN221 uMshwathi							4 000	970 6	10 760	4 000	9006	10 760
KZN222								1 106	819		1 106	819
KZN223												
KZN224							103	i i	0.00	103	t	0
KZN225							4413	9//.9	10 3/2	4 413	9//9	10 3/2
B KZN227 Richmond							3 1 2 3	25 404	38 I86 4 625	2,5 5,5	25 404	38 180
DC22							25 000	33 000	25 000	25 000	33 000	25 000
Total: uMgungundlovu Municipalities							62 210	78 434	89 762	62 210	78 434	89 762
	456	400		456	709		4 4 5 6	41 971	34 639	4 456	41 971	34 639
B KZN233 Indaka B KZN234 Umtshezi	1 000	500	446	1 000	500	446	1 000	913	752	1 000	913	752
KZN235							4 517	8 000	10 000	4 517	8 000	10 000
B KZN236 Imbabazane C DC23 Uthukela District Municipality							41 914	44 000	65 910	41 914	44 000	65 910
Total:Uthukela Municipalities	1 456	1 209	446	1 456	1 209	446	70 318	94 884	111 300	70 318	94 884	111 300
B KZN241 Endumeni							7.053	751.0	251 OC	7 053	0.157	451 OC
VZN242							1 - 0	14 913	25 156	th - 01	1010	751 26
							19 14/	8 069	8 843	4 569	8 069	35 150 8 843
DC24							7 437	37 000	41 813	7 437	37 000	41 813
Total: Unzinyathi Municipalities							36 107	69 039	105 986	36 107	68 039	105 986
B KZN252 Newcastle	200	200	1 160	200	200	1 160	9 6 7 0 0	7 000	10 790	9 2 200	7 000	10 790
B KZN253 Emadlangeni							000	10.000	000 01	000	10.000	10.000
C DC25 Amajuba District Municipality							22 871		25 000	22 871		25 000
Total: Amajuba Municipalities	200	500	1 160	200	200	1 160	33 571	17 000	45 790	33 571	17 000	45 790

ALLOCATIONS-IN-KIND TO MUNICIPALITIES (SCHEDULE 7) 3 OF 3

		Neighbour	hood Developr	Neighbourhood Development Partnership Grant	hip Grant				SUB-TOTAL: INDIRECT	: INDIRECT		
	Natio	National Financial Year	Year	Munici	Municipal Financial Year	Year	Natio	National Financial Year	(ear	Munic	Municipal Financial Year	ear
Category Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
B KZN261 eDumbe							135			135		
KZN263							428		6 364			6 364
B KZN265 Nongoma							27 886	18 053		27 886	18 053	33 585
NZNZ66 DC26							5 / 94	79 130		60 000	79 130	54 085
모							94 294	99 047		94 294	99 047	95 414
B KZN271 Umhlabuyalingana							4 000	10 542	8 401	4 000	10 542	8 401
KZN272							4 262	10 000		4 262	10 000	24 467
B KZN273 The Big 5 False Bay							000	000 01	10 184	900	000	10 184
KZN275							23 037	10 000	27 838		10 000	24 201
DC27							101 000	118 081	90 672	_	118 081	90 672
Total: Umkhanyakude Municipalities							136 299	167 413	185 763	136 299	167 413	185 763
B KZN281 Mfolozi							672	672		672	672	966
KZN282							14 616	58 802	51 977	14 616	58 802	51 977
B KZN283 Ntambanana							5.4.773	000 8	150	57773	000 8	150
KZN284							0,7	000 6			000 6	00104
KZN286							5 298	8 000		5 298	8 000	10 000
C DC28 uThungulu District Municipality							33 999	44 203	55 672	33 999	44 203	55 672
Total: uThungulu Municipalities							109 358	119 677	164 794	109 358	119 677	164 794
B KZN291 Mandeni	2 800	786		2 800	786		2 800	1 271	359	2 800	1 2 7 1	359
B KZN292 KwaDukuza	5 000	3 350		5 000	3 350		6411	6 172			6 172	3 136
	2 500	2 500	1 500	2 500	2 500	1 500	62 332	100 140	1		100 140	118 696
_							4 234	4 690	6 401	4 234	4 690	6 401
Total: iLembe Municipalities	10 300	6 636	1 500	10 300	96 9 9	1500	93 440	154 052	171 263	93 440	154 052	171 263
B KZN431 Inewe							4 894	7 000	000 6	4 894	7 000	000 6
KZN432							7364	7 261	10 757	7 364	7 261	10 757
KZN433							329	57	42	329	57	42
KZN434							72 403	79 927	112 837	72 403	79 927	112 837
B KZN435 Umzimkhulu	1 100	100	400	1 100	100	400	29 825	73 112	78 197	29 825	73 112	78 197
Total: Sisonke Municipality	1 100	100	400	1100	100	400	139 724	196 267	259 475	139 724	196 267	259 475
Total: KwaZulu-Natal Municipalities	15 856	11 945	2 506	15 856	11 945	5 506	858 724	1 080 442	1 328 203	858 723	1 080 442	1 328 203

ALLOCATIONS-IN-KIND TO MUNICIPALITIES (SCHEDULE 7) 3 OF 3

			Neighbourl	nood Develonr	Neighbourhood Develonment Partnershin Grant	in Grant				SUB-TOTAL: INDIRECT	INDIRECT		
		Natio	National Financial Year	Year	Municip	Municipal Financial Year	/ear	Natio	National Financial Year	ear	Munici	Municipal Financial Year	ar
Category	Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
LIMPOPO													
B LIM331	Greater Giyani							9 204	10 529	12 367	9 204	10 529	12 367
B LIM332	Greater Letaba							14 043	48 084	33 563	14 043	48 084	33 563
B LIM333								13 404	13 135	15 770	13 404	13 135	15 770
B LIM334	Ba-Phalaborwa	1 000	1 000	1 000	1 000	1 000	1 000	4 121	3 223	3 672	4 121	3 223	3 672
_								3 858	2 494	3 176	3 858	2 494	3 176
C DC33	Mopani District Municipality							87 599	68 789	20 000	87 599	68 7 89	50 000
Total: Mopani	Total: Mopani Municipalities	1 000	1 000	1 000	1 000	1 000	1 000	132 228	146 255	118 548	132 228	146 255	118 548
D 11M341	Mercina												
B LIM342	Mutale Mutale							71 421	7 104	12 063	71 421	7 104	12 063
		5 610	1 900	1 500	5 610	1 900	1 500	23.767	52 063	37 660	23 767	52 063	37 660
								10 227	992 99	40 496	10 227	992 99	40 496
C DC34	Vhembe District Municipality							29 767	45 200	15 000	59 767	45 200	15 000
Total: Vhembe	Total: Vhembe Municipalities	5 610	1 900	1 500	5 610	1900	1 500	165 182	171 133	105 219	165 182	171 133	105 219
B LIM351	Blouberg							13 168	13 924	16 046	13 168	13 924	16 046
								4 000	12 154	10 577	4 000	12 154	10 577
B LIM353	Molemole Polokwane	3 000	1 900	1 278	3 000	1 900	1 278	30 953	21 796	25 202	30 053	21 796	25 202
B LIM355		000		0/1	9	000	9/7	5 270	17 810		5 270	17810	14 040
								65 691	2 736		65 691	2 736	1710
Total: Caprico	Total: Capricorn Municipalities	3 000	1 900	1 278	3 000	1 900	1 278	119 082	68 419	67 575	119 082	68 419	67 575
B LIM361								0	c ·		i c	i d	
B LIM362								8 195	13 047	13 121	8 195	13 047	13 121
	Modimolle												
								260		98	260		98
	Mogalakwena	1 500			1 500			71 134	93 078	99 467	71 134	93 078	99 467
DC36	Waterberg District Municipality	9			9			513	912	713	513	216	713
10tal: waterb	total: waterberg Municipalities	1 200			00C T			SU 103	10/03/	086 611	80 103	10 / 03 /	113 380
B LIM471	Ephraim Mogale							6 053	1 589		6 053	1 589	3 821
								8 554	9 518		8 554	9 5 1 8	11 536
								17 039	12 615	14 827	17 039	12 615	14 827
B LIM474		6		1	6	i	1	9359	4 209	6 784	9 359	4 209	6 784
B LIM4/5	Greater Tubatse Greater Sabhubhuna Dietrict Municipality	2 200	06/	000	2 200	08/	066	27.219	43 305	35 837	27.219	43 305	35 837
Total: Greater	Total: Greater Sekhukhune Municipalities	2 200	750	550	2 200	750	550	212 146	258 236	297 805	212 146	258 236	297 805
Total: Limpop	Total: Limpopo Municipalities	13 310	5 550	4 328	13 310	5 5 5 0	4 328	708 741	751 080	702 533	708 741	751 080	702 533

ALLOCATIONS-IN-KIND TO MUNICIPALITIES (SCHEDULE 7) 3 OF 3

			Neighbour	hood Develop	Neighbourhood Development Partnership Grant	hip Grant				SUB-TOTAL: INDIRECT	INDIRECT		
		Natio	National Financial Year	Year	Munici	Municipal Financial Year	Year	Natio	National Financial Year	ear	Munic	Municipal Financial Year	ar
Category Municipality	pality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
MPUMALANGA													
B MP301 Albert Luthuli	Luthuli							23 663	14 490	29 054	23 663	14 490	29 054
MP302	ligwa							2 045			2 045		
B MP303 Mkhondo	Mkhondo							9 713	4 988	7 351	9 713	4 988	7 351
MP305	Na Jeine							934	1 231	1 082	934	1 231	1 082
B MP306 Dipaleseng B MP307 Govan Mbeki	seng Mbeki							2 0 4 2			2 042		
DC30	Gert Sibande District Municipality												
Total: Gert Sibande Municipalities	nnicipalities							38 397	20 710	37 487	38 397	20 710	37 487
R MP311 Victor Khanve	Khanve							20226	050 56	30.103	20 202	05000	30.103
MP312	ileni							1	3		81		
	Steve Tshwete	1 100			1 100			5 261	3 010	3 585	5 261	3 010	3 585
MP314	azeni								700	1	0	000	000
MP315	ISIDE							0 233	0 384	2 985	0 233	0.384	2 985
B MP316 Dr JS Moroka C DC31 Nkangala Dist	Dr JS Moroka Nkangala District Municipality							1 /89	16 550	21 529	1 /89 2 508	16550	21 529
Total: Nkangala Municipalities	ipalities	1 100			1 100			38 015	52 213	62 730	38 015	52 213	62 730
B MP321 Thaba Chweu	Chweu												
	oela	2 500	378		2 500	378		20 789	33 404	39 908	20 789	33 404	39 908
B MP323 Umjindi								16 106	21 720	40.160	46.106	21 230	40.169
MP324	Inkollidzi Bushkudzeidae	0000	2 500	000	0000	2 500	000	129 136	98 564	37 570	129 136	98 564	37 570
DC32	Enlanzeni District Municipality	1	1		2	1		5 500		26 000	5 500		26 000
Total: Ehlanzeni Municipalities	ipalities	4 500	2 878	1 000	4 500	2 878	1 000	201 531	163 198	143 646	201 531	163 198	143 646
Total: Mpumalanga Municipalities	unicipalities	2 600	2 878	1 000	2 600	2 878	1 000	277 943	236 121	243 864	277 943	236 121	243 864

ALLOCATIONS-IN-KIND TO MUNICIPALITIES (SCHEDULE 7) 3 OF 3

			Neighbour	hood Develop	Neighbourhood Develonment Partnershin Grant	in Grant				SUB-TOTAL: INDIRECT	INDIRECT		
		National		Year	Municip	Municipal Financial Year	Year	Natio	National Financial Year	ear	Munici	Municipal Financial Year	ır
Category	Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
NORTHERN CAPE	CAPE												
B NC061	Richtersveld Nama Khai							00091	18 000	9 000	000 91	18 000	9 000
								2000	1+	000 501	000 01	7	000 501
										8 775			8 775
4 4								1 294	18 693	9 994	1 294	18 693	9 994
C DC6 Total: Namakv	C DC6 Namakwa District Municipality Total: Namakwa Municipalities							17 294	77 693	130 769	17 294	77 693	130 769
B NC071	Hhuntu							33	33	33	33	33	33
B NC072								33 790	36 152	21 731	33 790	36 152	21 731
								17 935	18 935	5 935	17 935	18 935	5 935
B NC074	Kareeberg												
B NC076								12 082	20 000	5 041	12 082	20 000	5 041
								75	496	285	75	496	285
B NC078	Siyancuma Biylari Po Sama Dietriot Municipality							1 195	1 113	1 154	1 195	1 113	1 154
Total: Pixley K	Total: Pixley Ka Seme Municipalities							65 110	76 730	39 180	65 110	76730	39 180
R NC081	Mier												
								3 710	114	114	3 710	114	114
B NC083								787	714	750	787	714	750
								115	114	114	115	114	114
4 4								35 640	7 058	2 099	35 040	7 028	7 099
C DC8 Total: Siyanda	C DC8 Siyanda District Municipality  Total: Siyanda Municipalities							40 251	3 000	3 078	40 251	3 000	3 0 78
10007	Cal Disast:	000 0			000			1			1		
B NC092		7 000			7 000			700		14 350	700		14 350
								11 693	11 693	11 693	11 693	11 693	11 693
B NC094								20 641	20 558	20 599	20 641	20 558	20 599
Total: Frances	Total: Frances Baard Municipality  Total: Frances Baard Municipalities	2 000			2 000			40 034	32 252	81 643	40 034	32 252	81 643
B NC451	Moshaweng							46 403	84 903	17 403	46 403	84 903	17 403
		1 000	1 000	800	1 000	1 000	800	9 053	9 053	8 853	9 053	9 053	8 853
B NC453 C DC45	Gamagara John Taolo Gaetsewe District Municipality							5 216 1 524	5 645	3 585	5 216 1 524	5 645	3 585
Total: John Ta	Total: John Taolo Gaetsewe Municipalities	1 000	1 000	800	1 000	1 000	800	62 197	99 602	29 841	62 197	60 602	29 841
Total: Norther	Total: Northern Cape Municipalities	3 000	1 000	800	3 000	1 000	800	224 885	289 277	284 510	224 885	289 277	284 510

ALLOCATIONS-IN-KIND TO MUNICIPALITIES (SCHEDULE 7) 3 OF 3

				Neighbourl	lood Develop	Neighbourhood Development Partnership Grant	hip Grant				SUB-TOTAL: INDIRECT	INDIRECT		
			Natio	National Financial Year	Year	Munici	Municipal Financial Year	Year	Natic	National Financial Year	ear	Muni	Municipal Financial Year	ar
Cate	Category Municipality	ality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
NORTI	NORTH WEST													
	NW371 Moretele								10 752	13 193	13 473	10 752	13 193	13 473
N B	NW372 Madibeng	50							65 410	70 926	896 82	65 410	70 926	78 968
		SI	2 000	3 000	2 800	2 000	3 000	2 800	7 661	7 69 7	7 979	7 661	7 69 T	7 979
		rivier												
N B		otane							17 421	21 041	21 731	17 421	21 041	21 731
C	DC37 Bojanala I	Bojanala Platinum District Municipality								18 000	25 000		18 000	25 000
Total: I	Fotal: Bojanala Platinum Municipalities	Municipalities	2 000	3 000	2 800	2 000	3 000	2 800	101 245	130 857	147 151	101 245	130 857	147 151
Z g	NW381 Ratlou								8 802	27 199	30 501	8 802	27 199	30 501
В									3 242	2 594	2 918	3 242	2 594	2 918
	NW383 Mafikeng	500							29 151	38 178	33 664	29 151	38 178	33 664
N B	NW384 Ditsobotla	a .							960 6	11 654	10 375	960 6	11 654	10 375
Z B	NW385 Ramotshe	Ramotshere Moiloa	2 500	1 246		2 500	1 246		12 365	12 938	15 279	12 365	12 938	15 279
Ü	DC38 Ngaka Mc	Ngaka Modiri Molema District Municipality									10 000			10 000
Total: 1	gaka Modiri Mok	Total: Ngaka Modiri Molema Municipalities	2 500	1 246		2 500	1246		62 656	92 563	102 736	62 656	92 563	102 736
Z B	NW392 Naledi								7 495	1 996	2 245	7 495	1 996	2 245
N B	NW393 Mamusa								4 041	7 033	10 037	4 041	7 033	10 037
N B	NW394 Greater Taung	aung	2 500	906		2 500	906		12 402	17 095	17 546	12 402	17 095	17 546
B	NW396 Lekwa-Teemane	eemane							4 211	3 369	3 790	4 211	3 3 6 9	3 790
Z B	NW397 NW397								8 837	7 070	7 954	8 837	7 070	7 954
ر ن	DC39 Dr Ruth S	Dr Ruth Segomotsi Mompati District Municipality							92 000	80 000	80 000	92 000	80 000	80 000
Total: I	r Ruth Segomotsi	Total: Dr Ruth Segomotsi Mompati Municipalities	2 500	906		2 500	906		101 987	116 562	121 572	101 987	116 562	121 572
Z M	NW401 Ventersdorp	dio								10 000	10 000		10 000	10 000
N B	NW402 Tlokwe		2 000		3 000	2 000	2 200	3 000	2 000	2 200	3 000	2 000	2 2 0 0	3 000
В	NW403 City of Matlosana	Iatlosana	3 000	2 000	2 795	3 000	2 000	2 795	8 166	6 133	7 445	8 166	6 133	7 445
Z B		Hills							10 467	8 373	9 420	10 467	8 373	9 420
C	DC40 Dr Kennel	Dr Kenneth Kaunda District Municipality												
Total: I	Fotal: Dr Kenneth Kaunda Municipalities	da Municipalities	5 000	4 200	5 795	2 000	4 200	5 795	20 633	26 707	29 865	20 633	26 707	29 865
Total:	Total: North West Municipalities	ipalities	12 000	9 352	8 595	12 000	9 352	8 595	286 521	366 688	401 324	286 521	366 688	401 324

ALLOCATIONS-IN-KIND TO MUNICIPALITIES (SCHEDULE 7) 3 OF 3

			Neighbourk	ood Develon	Neighbourhood Development Partnership Grant	in Grant				SIB-TOTAL: INDIRECT	INDIRECT		
		Nation	National Financial Year	/ear	Munici	Municipal Financial Year	/ear	Natio	National Financial Year	ear	Munic	Municipal Financial Year	ar
Category	Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
WESTERN CAPE	PE												
A CPT	City of Cape Town	4 500	3 032	3 000	4 500	3 032	3 000	73 392	87 123	79 623	73 392	87 123	79 623
B WC011	Matzikama	100	148		100	148		215	239	103	215	239	103
	Cederberg							20 853	822	838	20 853	822	838
	Bergrivier							115	121	118	115	121	118
B WC014 B WC015	Saldanha Bay Swartland							3 644	2 915		3 644	2 9 1 5	3 279
C DC1	West Coast District Municipality							5 2 5 7	12 000	53 000	5 257	12 000	53 000
Total: West Co.	Total: West Coast Municipalities	100	148		100	148		30 083	16 098	57 338	30 083	16 098	57 338
B WC022	Witzenberg							17 115	202	158	17 115	202	158
B WC023	Drakenstein Steallankook							12 115	209	18 162	12 115	209	18 162
B WC025	Stellenbosch Breede Valley							1916	2 313	42 115	1 916	2 3 1 3	42 115
B WC026	Langeberg Cons Windowde Dietrict Municipality	200	392		700	392		815	<i>LL</i> 9	200	815	219	200
Total Cane Wi	Total: Cane Winelands Municipalities	002	302		002	302		34 960	3 402	589 09	3,000	3 400	569 09
B WC031	Thewaterskloof	200	766		8	766		3 480	0 789	3 139	3 489	2789	3 139
	Overstrand	250	150	85	250	150	85	250	150	85	250	150	85
B WC033	Cape Agulhas Swellendam							711	3.098	12 106	711	3 008	12 106
	Overberg District Municipality							1 500		201	1 500		001 71
Total: Overberg	Total: Overberg Municipalities	250	150	85	250	150	85	5 3 5 4	6 037	15 331	5 354	6 0 37	15 331
	Kannaland								11 700	5 599		11 700	5 599
B WC042	Hessequa							358	15 181	20 322	358	15 181	20 322
B WC044	Mossel Bay George							14 301	4 640		14 301	4 640	5 220
	Oudtshoorn	800	200	200	800	200	200	4 848	3 831		4 848	3 831	3 889
	Bitou	1 400	1 300	1 309	1 400	1 300	1 309	1 400	1 300	1 309	1 400	1 300	1 309
B wC048 C DC4	Knysna Eden District Municipality	1 /00	2 000	2 000	1700	2 000	7 000	1 700	2 000		1 700	2 000	2 000
otal: F	unicipalities	3 900	3 800	3 509	3 900	3800	3 509	22 606	38 652	38 339	22 606	38 652	38 339
	Laingsburg							115	200	157	115	200	157
	Prince Albert Beaufort West	350	1117		350	117		813	18 265	8 806	813	18 265	8 806
C DC5	Central Karoo District Municipality												
Total: Central l	Total: Central Karoo Municipalities	350	117		350	117		928	18 465	8 963	928	18 465	8 963
Total: Western	Total: Western Cape Municipalities	008 6	7 639	6 594	0086	7 639	6 594	167 322	169 777	260 229	167 322	169 777	260 229
Unallocated:													
National Total		100 000	80 000	55 000	100 000	80 000	55 000	3 992 187	4 444 774	4 734 094	3 992 187	4 444 774	4 7 3 4 0 9 4

#### ANNEXURE W7

# INCENTIVES TO MUNICIPALITIES TO MEET TARGETS WITH REGARDS TO PRIORITY GOVERNMENT PROGRAMMES (SCHEDULE 8)

(National and Municipal Financial Years)

ANNEXURE W7
INCENTIVES TO MUNICIPALITIES TO MEET TARGETS WITH REGARDS TO PRIORITY GOVERNMENT PROGRAMMES (SCHEDULE 8)

			Ex	panded Public W	orks Programme	Expanded Public Works Programme Incentive Grant for Municipalities	for Municipalit	ies		L-BUS	OTAL: INCENT	SUB-TOTAL: INCENTIVE ALLOCATIONS	IONS	
			Nati	National Financial Ye	Year	Munic	Municipal Financial Year	(ear	Na	National Financial Year	/ear	Munic	Municipal Financial Year	/ear
Catı	Category	Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
EASTERN CAPE	CAPE													
4	BUF NMA	Buffalo City Nelson Mandela Bay	1 512 8 923			2 268			1 512 8 923			2 2 68 13 3 8 4		
Д	EC101	Camdeboo	996			1 449			996			1 449		
В	EC102		357			536			357			536		
а м	EC103		357			536			357			536		
В	EC105		357			536			357			536		
В	EC106													
В	EC107		996			1 449			996			1 449		
м,	EC108		1 251			1 876			1251			1 876		
a C	EC109	Kou-kamma Cacadu District Municipality	357			536			357			536		
Total: Cacadu Municipalities	ıdu Munici	ipalities	4 968			7 454			4 968			7 454		
		1												
В	EC121													
д	EC122		357			536			357			536		
<b>A</b> 10	EC123		007			210			002			710		
<b>മ</b> മ	EC124	Amaniami	600			914			600			914		
αм	EC120		357			536			357			536		
В	EC128													
C	DC12	Amathole District Municipality	7 803			11 704			7 803			11 704		
Total: Amathole Municipalities	thole Mun	nicipalities	9 126			13 690			9 126			13 690		
ш	EC131	Inxuba Yethemba	357			536			357			536		
В	EC132		357			536			357			536		
В	EC133		357			536			357			536		
В	EC134	Lukhanji	996			1 449			996			1 449		
В	EC135		996			1 449			996			1 449		
В	EC136	Emalahleni	1 625			2 438			1 625			2 438		
В	EC137	Engcobo	609			914			609			914		
В	EC138		357			536			357			236		
C	DC13	Chris Hani District Municipality	16 400			24 600			16 400			24 600		
Total: Chr.	e Hani Min	Ortal: Chris Hani Municipalities	21 994			32 994			21 994			32,004		

ANNEXURE W7
INCENTIVES TO MUNICIPALITIES TO MEET TARGETS WITH REGARDS TO PRIORITY GOVERNMENT PROGRAMMES (SCHEDULE 8)

	Ė	Fynandad Public V	Vorke Programm	blic Works Programme Incentive Creat for Municipalities	for Municipalit	soi		CTIR.7	POTAI - INCEN	STIB-TOTAL: INCENTIVE ALLOCATIONS	SNOT	
		banaca r annuc.	TOTAL TOPICALITY	in meeting of an	ior intermediate	Tropies of the second		100	TOTAL INCEN	I VE ALLOCAL	in in it	
		National Financial Year		Muni	Municipal Financial Year	/ear	Nati	National Financial Year	Year	Muni	Municipal Financial Year	ear
Cotocour	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14
Category Municipality	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)
B EC141 Elundini	357			536			357			236		
B EC142 Sengu	996			1 449			996			1 449		
B EC143 Maletswai	357			536			357			536		
B EC144 Gariep	357			536			357			536		
C DC14 Ukhahlamba District Municipality	3 476			5 2 1 4			3 476			5 2 1 4		
Total: Ukhahlamba Municipalities	5 513			8 271			5 513			8 271		
				, c			, c			C		
	/cc			956			/cc			occ		
B EC154 Port St Johns	996			1 449			996			1 449		
B EC155 Nyandeni				536						536		
B EC156 Millontlo	996			1 449			996			1 449		
B EC157 King Sabata Dalindyebo	609			914			609			914		
C DC15 OR Tambo District Municipality	7 792			11 688			7 792			11 688		
Total: O.R.Tambo Municipalities	10 690			16 572			10 690			16 572		
				1000						1000		
	1 934			7 201			1 934			1067		
B EC442 Umzimvubu	996			1 449			996			1 449		
B EC443 Mbizana												
B EC152 Ntabankulu	996			1 449			996			1 449		
C DC44 Alfred Nzo District Municipality	9 530			14 295			9 530			14 295		
Total: Alfred Nzo Municipalities	13 396			20 094			13 396			20 094		
Total: Eastern Cape Municipalities	76 122			114 727			76 122			114 727		

ANNEXURE W7 INCENTIVES TO MUNICIPALITIES TO MEET TARGETS WITH REGARDS TO PRIORITY GOVERNMENT PROGRAMMES (SCHEDULE 8)

	Ex	panded Public W	orks Programm	Expanded Public Works Programme Incentive Grant for Municipalities	for Municipalit	ies		SUB-C	TOTAL: INCEN	SUB-TOTAL: INCENTIVE ALLOCATIONS	SNOL	
	Nati	National Financial Year	ear	Munic	Municipal Financial Year		Nat	National Financial Year	Year	Munic	Municipal Financial Year	ear
Category Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
FREE STATE												
A MAN Mangaung	5 713			8 570			5 713			8 570		
B FS161 Letsemeng	357			536			357			536		
FS162	357			536			357			536		
FS163	357			536			357			536		
B FS164 Naledi C DC16 Xhariep District Municipality	357			1 449			966			1 449		
ĿĦ	2 394			3 593			2 394			3 593		
B FS181 Masilonvana	357			536			357			536		
FS182	357			536			357			536		
FS183	357			536			357			536		
FS184	3 335			5 002			3 335			5 002		
	357			536			357			989		
C DC18 Lejweleputswa District Municipality	996			1 449			996			1 449		
Total: Lejweleputswa Municipalities	5 729			8 595			5 729			8 595		
B FS191 Setsoto	7 095			10 642			7 095			10 642		
FS192	367			550			367			550		
FS193	357			536			357			536		
	902 9			10 059			902 9			10 059		
FS195												
B FS196 Mantsopa	357			536			357			536		
a	16 229			24 343			16 229			24 343		
FS201	1 726			2 589			1 726			2 589		
FS203	6/6			1 462			6/6			1 462		
B FS204 Metsimaholo	357			536			357			536		
DC30	996			1 449			990			1 440		
otal: Fezile Dabi Muni	4 381			6 572			4 381			6 572		
Total: Free State Municipalities	34 446			51 673			34 446			51 673		

ANNEXURE W7 INCENTIVES TO MUNICIPALITIES TO MEET TARGETS WITH REGARDS TO PRIORITY GOVERNMENT PROGRAMMES (SCHEDULE 8)

	Ex	panded Public V	Vorks Programn	Expanded Public Works Programme Incentive Grant for Municipalities	t for Municipali	ies		SUB-1	FOTAL: INCEN	SUB-TOTAL: INCENTIVE ALLOCATIONS	SNOI		
	Nati	National Financial Year	(ear	Mun	Municipal Financial Year	/ear	Nati	National Financial Year	Year	Muni	Municipal Financial Year	/ear	
	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	21/1102	2012/13	2013/14	
Category Municipality	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	
GAUTENG													
A EKU Ekurhuleni	6 222			9 333			6 222			9 333			
A JHB City of Johannesburg	110 161			286 581			110 161			286 581			
	7 682			11 520			7 682			11 520			
	7 693			11 540			7 693			11 540			
B GT422 Midvaal	357			536			357			536			
B GT423 Lesedi													
C DC42 Sedibeng District Municipality	357			536			357			536			
Total: Sedibeng Municipalities	8 407			12 612			8 407			12 612			
B GT481 Mogale City	2 709			4 064			2 709			4 064			
B GT482 Randfontein	2 009			3 013			2 009			3 013			
B GT483 Westonaria	4 682			7 023			4 682			7 023			
B GT484 Merafong City	3 952			5 928			3 952			5 928			
C DC48 West Rand District Municipality	686			1 484			686			1 484			
Total: West Rand Municipalities	14 341			21 212			14 341			21 212			
Patel. Content Maniellalities	233 244			933 116			233 244			922 112			

ANNEXURE W7 INCENTIVES TO MUNICIPALITIES TO MEET TARGETS WITH REGARDS TO PRIORITY GOVERNMENT PROGRAMMES (SCHEDULE 8)

	Ex	spanded Public W	orks Programme	Expanded Public Works Programme Incentive Grant for Municipalities	for Municipaliti	es		L-SUB-1	OTAL: INCEN	SUB-TOTAL: INCENTIVE ALLOCATIONS	SNOI	
	Nat	National Financial Year		Munic	Municipal Financial Year	ear		National Financial Year	/ear	Muni	Municipal Financial Year	ear
Category Municipality	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14
	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)
KWAZULU-NATAL												
A ETH eThekwini	72 084			108 126			72 084			108 126		
B KZN211 Vulamehlo												
KZN212												
KZN214												
KZNZIS				-			Š			-		
	1 802			1 449 2 704			966			2 704		
otal: Usu Municipalit	2 768			4 153			2 768			4 153		
romi cga rramchanaca	8			201			8			201		
B KZN221 uMshwathi												
KZN223												
KZN224												
KZN225	1 512			2 2 68			1 512			2 2 6 8		
KZN227												
C DC22 uMgungundlovu District Municipality												
Total: uMgungundlovu Municipalities	1 512			2 268			1 512			2 268		
COCINE				Č			t			i i		
	35/			930			/cc			230		
KZN233												
B KZN234 Umtshezi												
CCZNIZN												
KZN236										,		
C DC23 Uthukela District Municipality	11 153			16 729			11 153			16 729		
Total:Uthukela Municipalities	11 510			17 265			11 510			17 265		
B KZN241 Endumeni												
CYCNZX												
KZNZ4Z												
DC24	3 537			5 306			3 537			5 306		
Total: Umzinyathi Municipalities	3 537			2 306			3 537			2 306		
KZN252	996			1 449			996			1 449		
B VZNZ5 Emadiangen												
	996			1 449			996			1 449		
otal: Amainha Munici	1 932			2.898			1 932			2.898		

ANNEXURE W7 INCENTIVES TO MUNICIPALITIES TO MEET TARGETS WITH REGARDS TO PRIORITY GOVERNMENT PROGRAMMES (SCHEDULE 8)

	Ex	panded Public W	Vorks Programm	Expanded Public Works Programme Incentive Grant for Municipalities	for Municipalit	ies		SUB-7	FOTAL: INCENT	SUB-TOTAL: INCENTIVE ALLOCATIONS	SNOL	
	Nati	National Financial Year	ear	Munic	Municipal Financial Year	/ear	Nat	National Financial Year	/ear	Munic	Municipal Financial Year	Year
Category Municipality	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14
	(nnn v)	(N 000)	(N 000)	(non)	(N 000)	(NOO V)	(nnn <b>v</b> )	(N 000)	(NOON W)	(non)	(000 V)	(NOO)
B KZN261 eDumbe	357			536			357			536		
	357			536			357			536		
KZNZX	357			536			357			536		
KZN265	357			536			357			536		
DC26	1 2 1 7			1 826			1 217			1 826		
Total: Zululand Municipalities	2 645			3 970			2 645			3 970		
KZN272												
B KZN273 The Big 5 False Bay												
B KZN274 Hlabisa												
B KZN275 Mtubatuba												
C DC27 Umkhanyakude District Municipality	17 369			26 053			17 369			26 053		
Total: Umkhanyakude Municipalities	17 369			26 053			17 369			26 053		
B KZN281 Mfolozi												
B KZN283 Ntambanana												
B KZN284 Umlalazi												
B KZN285 Mthonjaneni												
B KZN286 Nkandla												
C DC28 uThungulu District Municipality	4 868			7 301			4 868			7 301		
Total: uThungulu Municipalities	4 8 6 8			7 301			4 868			7 301		
B KZN291 Mandeni												
B KZN292 KwaDukuza												
B KZN293 Ndwedwe												
B KZN294 Maphumulo												
C DC29 iLembe District Municipality	2 0 0 2			10 901			7 067			10 901		
Total: iLembe Municipalities	290 2			109 01			2 0 0 2			109 01		
R KZN431 Inowe												
VZN433	257			526			257			526		
NZN455	/66			930			/cc			930		
KZN434				1			1			6		
_	357			536			357			536		
C DC43 SISONKE DISTRICT MUNICIPALITY	19/5			7 963			1975			7 903		
Total: Sisonke Municipalities	2 689			4 035			2 689			4 035		
Total: KwaZulu-Natal Municipalities	127 981			101 076			127 981			101 076		
LOCAL INTRACTOR COMME STREET	1000		•	D			A C C C S			4/4/10		

ANNEXURE W7 INCENTIVES TO MUNICIPALITIES TO MEET TARGETS WITH REGARDS TO PRIORITY GOVERNMENT PROGRAMMES (SCHEDULE 8)

	Expanded P	ablic Works Programn	Expanded Public Works Programme Incentive Grant for Municipalities	Municipalities		20	SUB-TOTAL: INCENTIVE ALLOCATIONS	NTIVE ALLOCAT	IONS	
	National Financial Year	ncial Year	Municipa	Municipal Financial Year		National Financial Year	cial Year	Muni	Municipal Financial Year	
Category Municipality	2011/12 2012/13 (R'000) (R'000)	13 2013/14 (R'000)	2011/12 (R'000)	2012/13 2013/14 (R'000) (R'000)	14 2011/12 9) (R'000)	12 2012/13 9) (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 2013/14 (R'000) (R'000)	2013/14 (R'000)
LIMPOPO										
B LIM331 (Treater Givani	357		536			357		536		
	2 643		3 965			2 643		3 965		
LIM333	911		1 367			911		1 367		
LIM334	357		536			357		536		
LIM335	996		1 449			996		1 449		
C DC33 Mopani District Municipality	7 746		11 620			7 746		11 620		
Total: Mopani Municipalities	12 980		19 473			12 980		19 473		
LIM341	ţ									
LIM342	1 621		2 432			1 621		2 432		
LIMS43	2 291		3 43/			1677		3 43/		
B LIM344 Makhado	609		914			609		914		
C DC54 VIEIDE DISTICUMUNICIPANTY	1100		71060		 	1100		1000		
Total: Vnembe Municipanties	705 01		nne ct		<u>]</u> T	10 532		noe cr		
B LIM351 Blouberg	483		725			483		725		
LIM352	357		536			357		536		
LIM353										
	10 244		15 366		_	10 244		15 366		
LIM355	996		1 449			996		1 449		
DC35	4 269		6 404			4 269		6 404		
Total: Capricorn Municipalities	16 319		24 480			16 319		24 480		
	357		536			357		536		
	2 554		3 831			2 554		3 831		
LIM365	357		536			357		536		
	357		236			357		236		
LIM367	1 693		2 540			1 693		2 540		
C DC36 Waterberg District Municipality	996		1 449			996		1 449		
Total: Waterberg Municipalities	6 284		9 428			6 284		9 428		
B LIM471 Enhraim Mogale	357		536			357		536		
CLAMI I	2 661		3 991			2 661		3 001		
I IM473	1 492		2 238			1 492		2 2 38		
I IM474	357		536			357		536		
I IM475	1 284		1 976			1 284		1 926		
	12.333		18 499			12 333		18 499		
otal: Greater Sekhukl	18 484		27 726			18 484		27 726		
					<u>]</u>					
Total: Limpopo Municipalities	64 599	_	206 96		_	64 599		206 96		_

ANNEXURE W7 INCENTIVES TO MUNICIPALITIES TO MEET TARGETS WITH REGARDS TO PRIORITY GOVERNMENT PROGRAMMES (SCHEDULE 8)

		1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	d 1 %	, ,				CTID	POTAT. INCENT	SMORE ADOLLA SMERINSOM : LAROR GIR	SMOE	
	1	xpanded rubiic	WOFKS Programm	Expanded Fublic Works Frogramme incentive Grant for Municipalities	ior iviunicipaliti	8		-gne	IOIAE: INCEN	HVE ALLUCAL	CNOT	
		National Financial Year		Munic	Municipal Financial Year	ear		National Financial Year	Year	Muni	Municipal Financial Year	ear
Category Municipality	2011/12	2012/13 (R'000)	2013/14 (R'000)	2011/12	2012/13 (R'000)	2013/14 (R'000)	2011/12	2012/13 (R'000)	2013/14 (R:000)	2011/12	2012/13	2013/14 (R'000)
	(000 )	(200 V)	(2001)	(2003)	(200 11)	(2001)	(000 11)	(00011)	(00011)	(000 11)	(000 11)	(2221)
MPUMALANGA												
B MP301 Albert Luthuli	996	10		1 449			996			1 449		
MP302	357	7		536			357			536		
MP303	996	2		1 449			996			1 449		
MP304	9 477	7		14 215			9 477			14 215		
MP305	996	20		1 449			996			1 449		
MP306	996	2		1 449			996			1 449		
	1 574	+		2 361			1 574			2 361		
C DC30 Gert Sibande District Municipality	3 420	)		5 131			3 420			5 131		
Total: Gert Sibande Municipalities	18 692	2		28 039			18 692			28 039		
				i i			į			i i		
MP311	35/			036			35/			236		
MP312	996			1 449			996			1 449		
MP313	1 259	•		1 888			1 259			1 888		
MP314 I	357	7		536			357			536		
MP315	996	2		1 449			996			1 449		
	1 788	~		2 683			1 788			2 683		
C DC31 Nkangala District Municipality	3 242	2		4 863			3 242			4 863		
Total: Nkangala Municipalities	8 935	16		13 404			8 935			13 404		
B MP321 Thaba Chweu	996	, c		1 449			996			1 449		
MP322	11911			2 867			1 911			2 867		
B MP323 Umjindi	357	7		536			357			536		
B MP324 Nkomazi	1 092	2		1 638			1 092			1 638		
B MP325 Bushbuckridge	1 2 5 1	_		1 876			1 251			1 876		
C DC32 Ehlanzeni District Municipality	563	3		844			563			844		
Total: Ehlanzeni Municipalities	6 140	0		9 210			6 140			9 210		
Total: Mpumalanga Municipalities	33 767	7		50 653			33 767			50 653		

ANNEXURE W7
INCENTIVES TO MUNICIPALITIES TO MEET TARGETS WITH REGARDS TO PRIORITY GOVERNMENT PROGRAMMES (SCHEDULE 8)

	Exnai	Exnanded Public Works Programme Incentive Grant for Municipalities	Programme In	centive Grant fo	r Municipalitie	9		SIIB.	TOTAL: INCEN	SIB-TOTAL: INCENTIVE ALL OCATIONS	SNOT	
	Nation	National Financial Year	_	Municip	Municipal Financial Year			National Financial Year	Year	Munic	Municipal Financial Year	ear
Category Municipality	2011/12 (R'000)	2012/13 20 (R'000) (H	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
NORTHERN CAPE												
B NC062 Nama Khoi												
NC065												
NC066												
B NC067 Khāi-Ma	400.4			002			4 406			000		
otal: Namakwa Munic	4 405			809 9			4 405			809 9		
1001	I de			i i						i i		
B NC071 Ubuntu R NC072 Umsohomvii	357			536			357			536		
NC073	357			536			357			536		
NC074												
B NC075 Renosterberg	996			1 449			996			1 449		
NC077	996			1 449			996			1 449		
NC078												
C DC7 Pixley Ka Seme District Municipality				omo e			717.0			on o		
Total: Pixley Ka Seme Municipalities	7 040			3 9/0			7 040			39/0		
B NC081 Mior												
NC083												
NC085	357			536			357			536		
B NCU86 Kgatelopele C DC8 Siyanda District Municipality	888			13 332			8 888			13 332		
otal: Siyanda Municip	9 245			13 868			9 245			13 868		
1000018	200 71			001.10			1000			001.00		
B NC097 Dikestleng	10 08/			24 130			10 08/			24 130		
NC093												
NC094	357			536			357			536		
C DC9 Frances Baard District Municipality	6 588			9 882			6 588			9 882		
Total: Frances Baard Municipalities	72 037			34 348			23 032			34 348		
NC451	357			536			357			536		
NC452	996			1 449			996			1 449		
B NC453 Gamagara C DC45 John Taolo Gaetsewe District Municipality	7 113			10 669			7 113			10 669		
Ιŧ	8 436			12 654			8 436			12 654		
Total: Northern Cape Municipalities	47 764			71 648			47 764			71 648		
	-	-			-							

ANNEXURE W7 INCENTIVES TO MUNICIPALITIES TO MEET TARGETS WITH REGARDS TO PRIORITY GOVERNMENT PROGRAMMES (SCHEDULE 8)

	Ex	panded Public W	Vorks Programm	Expanded Public Works Programme Incentive Grant for Municipalities	t for Municipalit	ies		L-SUB-1	FOTAL: INCEN	SUB-TOTAL: INCENTIVE ALLOCATIONS	NOL	
	_	National Financial Year	ear	Muni	Municipal Financial Year			National Financial Year	Year	Muni	Municipal Financial Year	/ear
Category Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
NORTH WEST												
B NW371 Moretele	1 759			2 638			1 759			2 638		
NW372	3 417			5 125			3 417			5 125		
NW373	8 315			12 473			8 315			12 473		
B NW374 Kgetlengrivier	357			536			357			536		
	1 242			1 863			1 242			1 863		
C DC37 Bojanala Platinum District Municipality	1 301			1 952			1 301			1 952		
Total: Bojanala Platinum Municipalities	16 301			24 587			16 391			24 587		
B NW381 Botlon	009			410			009			017		
	990			1440			990			1 440		
NW 362	996			1 449			996			1 449		
	996			1 449			996			1 449		
NW385	609			914			609			914		
DC38	1 955			2 932			1 955			2 9 3 2		
Total: Ngaka Modiri Molema Municipalities	6 071			6 107			6 071			6 107		
				9			1			9		
NW392	357			530			357			530		
NW393	357			936			357			956		
NW394	966			1 449			966			1 449		
NW 396	357			230			357			530		
B NW39/ NW39/ NW39/ C DC39 Dr Ruth Segomotsi Mompati District Municipality	4 765			7 147			4 765			7147		
18	7 159			10 740			7 159			10 740		
1 AT TOTALINA	Š			-						,		
104WN	900			644			906			1 449		
NW402	996			1 449			996			1 449		
NW403	7 334			11 002			7 334			11 002		
	996			1 449			996			1 449		
C DC40 Dr Kenneth Kaunda District Municipality	357			536			357			536		
Total: Dr Kenneth Kaunda Municipalities	10 589			15 885			10 589			15 885		
Total: North West Municipalities	40 210			60 319			40 210			60 319		
			•			1					•	•

ANNEXURE W7
INCENTIVES TO MUNICIPALITIES TO MEET TARGETS WITH REGARDS TO PRIORITY GOVERNMENT PROGRAMMES (SCHEDULE 8)

	Exp	anded Public W	orks Programm	Expanded Public Works Programme Incentive Grant for Municipalities	t for Municipalit	ies		T-8UB-T	OTAL: INCENT	SUB-TOTAL: INCENTIVE ALLOCATIONS	NOL	
	~	National Financial Year	ear	Muni	Municipal Financial Year		Natio	National Financial Year	ear	Muni	Municipal Financial Year	ear
Category Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
WESTERN CAPE												
A CPT City of Cape Town	16 989			25 484			16 989			25 484		
WC011	357			536			357			536		
B WC012 Cederberg	357			536			357			236		
WC013	357			536			357			536		
B WC014 Saldanha Bay												
wcols DC1	357			536			357			536		
	1 428			2 144			1 428			2 144		
COOCIA												
B WC023 Drakenstein	357			536			357			536		
WC024												
WC025	357			536			357			536		
B WC026 Langeberg C DC2 Cane Winelands District Municipality												
spu	714			1 072			714			1 072		
2 - 11 - WYOOYN W				-			-			-		
B WC031 Illeewatershoot	357			1 / 01			357			1 /01		
WC033	357			536			357			536		
B WC034 Swellendam	357			536			357			536		
C DC3 Overberg District Municipality	357			536			357			536		
Total: Overberg Municipalities	2 562			3 845			2 562			3 845		
WC041	357			536			357			536		
WC042	357			536			357			536		
WC043	609			914			609			914		
WC044	1585			2 378			1 585			2 378		
B WC045 Oudishoom	1 359			2 038			1 359			2 038		
	, c			000			Ì			000		
C DC4 Eden District Municipality												
Total: Eden Municipalities	4 624			866 9			4 624			866 9		
B WC051 Laingsburg												
WC053	357			0			357					
C DC5 Central Karoo District Municipality	35/			3 26/			700			3 26/		
Total: Central Karoo Municipalities	714			3 267			714			3 267		
Total: Western Cape Municipalities	27 031			42 750			27 031			42 750		
Unallocated:												
National Total	679 583			1 022 211			679 583			1 022 211		

### ANNEXURE W8

EQUITABLE SHARE AND TOTAL ALLOCATIONS TO MUNICPALITIES

## (National and Municipal Financial Years)

ANNEXURE W8
EQUITABLE SHARE AND TOTAL ALLOCATIONS TO MUNICIPALITIES

					-								
				EQUITABLE SHARE	SHARE				TOTAL A	TOTAL ALLOCATIONS TO MUNICIPALITIES	TO MUNICIPA	ALITHES	
		Nation	Œ	ear	Muni	Municipal Financial Year	Year	Natio	National Financial Year	ear	Munic	Municipal Financial Year	Year
Category	Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
EASTERN CAPE	CAPE												
A BUF	Buffalo City	583 628	644 542	698 989	583 628	644 542	698 989	1 230 680	1 367 267	1 601 173	1 231 436	1 367 267	1 601 173
	- 1	656 653	730 416	778 401	656 653	730 416	778 401	2 066 752	1 801 566	2 185 025	2 071 213	1 801 566	2 185 025
B EC101	Camdeboo	33 092	36 581	38 967	33 092	36 581	38 967	50 883	59 154	58 485	51 366	59 154	58 485
B EC102	2 Blue Crane Route	34 998	38 693	41 218	34 998	38 693	41 218	54 396	63 386	65 550	54 575	63 386	65 550
	3 Ikwezi	13 753	15 192	16 183	13 753	15 192	16 183	24 875	27 760	29 401	25 054	27 760	29 401
		59 143	65 410	<i>LL</i> 69	59 143	65 410	229 69	105 787	117 462	119 078	105 966	117 462	119 078
		49 522	54 813	58 398	49 522	54 813	58 398	82 163	96 96	101 268	82 342	96 96	101 268
		32 105	35 517	37 845	32 105	35 517	37 845	71 800	70 551	77 405	71 800	70 551	77 405
		15 153	16 751	17 848	15 153	16 751	17 848	28 922	31 503	33 628	29 405	31 503	33 628
		41 038	45 585	48 593	41 038	45 585	48 593	69 583	75 852	80 472	70 208	75 852	80 472
B EC109		25 910	28 642	30 511	25 910	28 642	30 511	45 921	50 001	53 565	46 100	50 001	53 565
C DC10	Cacadu District Municipality	67 220	999 69	72 317	67 220	999 69	72 317	76 260	89 166	105 817	76 260	89 166	105 817
Total: Caca	Total: Cacadu Municipalities	371 935	406 850	431 557	371 935	406 850	431 557	610 590	681 791	724 670	613 076	681 791	724 670
		000	000	20.	000	0.00		000		000	000		000
		105 258	116 503	124 1/3	105 238	116 503	1.24 1/3	1/3 /28	193 117	233 94 /	1/3 /28	195117	255 94 /
		136 070	150 518	160 377	136 070	150 518	160 377	261 079	267 152	288 294	261 258	267 152	288 294
		28 209	31 184	33 221	28 209	31 184	33 22 1	55 922	59 864	65 399	55 922	59 864	65 399
		79 616	87 995	93 737	79 616	87 995	93 737	116 797	135 730	142 541	117 102	135 730	142 541
		55 140	960 965	64 949	55 140	960 962	64 949	79 308	86 188	92 491	79 308	86 188	92 491
		76 099	84 004	89 451	76 099	84 004	89 451	104 801	114 740	122 620	104 980	114 740	122 620
B EC128		18 627	20 591	21 935	18 627	20 591	21 935	30 536	34 647	36 738	30 536	34 647	36 738
C DC12	Amathole District Municipality	530 281	583 693	627 182	530 281	583 693	627 182	965 168	1 066 587	1 111 638	690 696	1 066 587	1 111 638
Total: Ama	Total: Amathole Municipalities	1 029 279	1 135 453	1 215 023	1 029 279	1 135 453	1 215 023	1 787 339	1 958 025	2 093 668	1 791 903	1 958 025	2 093 668
B EC131	I Inxuba Yethemba	35 770	39 535	42 109	35 770	39 535	42 109	50 664	65 093	74 603	50 843	65 093	74 603
B EC132	2 Tsolwana	21 878	24 165	25 740	21 878	24 165	25 740	39 591	39 719	47 821	39 770	39 719	47 821
B EC133	3 Inkwanca	15 748	17 399	18 534	15 748	17 399	18 534	27 365	29 714	31 556	27 544	29 714	31 556
B EC134	1 Lukhanji	96 062	106 129	113 038	96 062	106 129	113 038	132 572	148 626	163 480	133 055	148 626	163 480
B EC135	5 Intsika Yethu	78 404	86 613	92 256	78 404	86 613	92 256	128 897	138 677	145 843	129 380	138 677	145 843
B EC136	5 Emalahleni	619 19	68 112	72 564	619 19	68 112	72 564	98 360	108 624	122 163	99 173	108 624	122 163
		965 99	73 753	78 619	962 99	73 753	78 619	154 203	174 320	182 107	154 508	174 320	182 107
B EC138	S Sakhisizwe	34 165	37 799	40 278	34 165	37 799	40 278	59 624	58 688	69 329	59 803	58 688	69 329
C DC13	Chris Hani District Municipality	325 908	360 052	384 758	325 908	360 052	384 758	750 151	926 868	190 696	758 351	926 868	190 696
Total: Chris	Total: Chris Hani Municipalities	736 151	813 558	867 896	736 151	813 558	867 896	1 441 428	1 662 437	1 805 962	1 452 428	1 662 437	1805962

ANNEXURE W8
EQUITABLE SHARE AND TOTAL ALLOCATIONS TO MUNICIPALITIES

			EQUITABI	EQUITABLE SHARE				TOTAL A	LLOCATIONS	TOTAL ALLOCATIONS TO MUNICIPALITIES	ALITIES	
	Nati	National Financial Year	ear	Muni	Municipal Financial Year	Year	Natio	National Financial Year	(ear	Munic	Municipal Financial Year	Year
Cotocom: Municipality	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14
Category intuincipanty	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)
B EC141 Elundini	64 855	71 762	76 475	64 855	71 762	76 475	114 589	120 959	146 163	114 768	120 959	146 163
B EC142 Sengu	79 190	87 568	93 295	79 190	87 568	93 295	120 614	130 239	140 039	121 097	130 239	140 039
B EC143 Maletswai	20 224	22 351	23 809	20 224	22 351	23 809	38 558	41 987	44 423	38 737	41 987	44 423
B EC144 Gariep	22 167	24 507	26 107	22 167	24 507	26 107	36 732	41 879	49 715	36 911	41 879	49 715
C DC14 Ukhahlamba District Municipality	149 031	164 710	175 859	149 031	164 710	175 859	300 192	343 578	364 793	301 930	343 578	364 793
Total: Ukhahlamba Municipalities	335 467	370 898	395 545	335 467	370 898	395 545	610 685	678 642	745 132	613 443	678 642	745 132
B EC153 Ngquza Hill	97 182	107 542	114 611	97 182	107 542	114 611	169 097	197 533	203 761	169 276	197 533	203 761
B EC154 Port St Johns	59 537	65 862	70 188	59 537	65 862	70 188	91 204	97 759	103 749	91 687	97 759	103 749
B EC155 Nyandeni	111 925	123 822	131 944	111 925	123 822	131 944	194 977	211 187	246 944	195 513	211 187	246 944
B EC156 Mhlontlo	87 213	96 486	102 818	87 213	96 486	102 818	134 025	181 732	243 272	134 508	181 732	243 272
B EC157 King Sabata Dalindyebo	153 086	169 585	180 725	153 086	169 585	180 725	230 853	258 542	270 043	231 158	258 542	270 043
C DC15 OR Tambo District Municipality	416 223	450 392	480 874	416 223	450 392	480 874	1 010 205	1 162 021	1 222 009	1 014 101	1 162 021	1 222 009
Total: O.R. Tambo Municipalities	925 167	1 013 688	1 081 159	925 167	1 013 688	1 081 159	1 830 360	2 108 773	2 289 780	1 836 242	2 108 773	2 289 780
B FC441 Meratiolo	92 449	102 304	109 073	92 449	102 304	109073	178 876	190 719	214 525	179.843	190 719	214 525
B FC442 Ilmzimunbu	00 003		100 667	02 003	102 201	109 667	308 027	306 528	220 022	300 4 10	306 528	322 077
B EC443 Mhizana	098.860		116 530	098 860	109 353	116 530	208 883	314 486	266 950	208 883	314 486	266 950
B EC152 Ntabankulu	54 929		64 740	54 929	60 753	64 740	96 175	125 779	117 822	96 658	125 779	117 822
C DC44 Alfred Nzo District Municipality	265 535	(4	315 328	265 535	294 916	315 328	714 381	805 543	864 832	719 146	805 543	864 832
Total: Alfred Nzo Municipalities	604 767	670 232	715 289	604 767	670 232	715 289	1 507 242	1 743 054	1 787 105	1 513 940	1 743 054	1 787 105
Total: Eastern Cape Municipalities	5 243 046	5 785 638	6 171 740	5 243 046	5 785 638	6 171 740	11 085 076	12 001 556	13 232 516	11 123 681	12 001 556	13 232 516

ANNEXURE W8
EQUITABLE SHARE AND TOTAL ALLOCATIONS TO MUNICIPALITIES

			EQUITABLE SHARE	E SHARE				TOTAL A	TOTAL ALLOCATIONS TO MUNICIPALITIES	TO MUNICIPA	ALITIES	
	Natio	na		Muni	Municipal Financial Year	Year	Natio	National Financial Year	Vear	Munic	Municipal Financial Year	/ear
Category Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
FREE STATE												
A MAN Mangaung	546 417	603 089	642 590	546 417	603 086	642 590	1 006 478	1 118 836	1 204 836	1 009 335	1 118 836	1 204 836
B FS161 Letsemeng	45 212	49 982	53 241	45 212	49 982	53 241	66 654	74 423	78 970	66 833	74 423	78 970
FS162	79 334	87 704	93 418	79 334	87 704	93 418	132 308	186 380	121 698	132 487	186 380	121 698
FS163	45 632		53 741	45 632	50 451	53 741	75 373	85 135	101 035	75 552	85 135	101 035
	33 288		39 204	33 288	36 803	39 204	50 643	56 306	69 703	51 126	56306	69 703
C DC16 Xhariep District Municipality	20 629		23 067	20 629	22 002	23 067	23 026	24 052	25 167	23 205	24 052	25 167
Total: Xhariep Municipalities	224 095	246 941	262 671	224 095	246 941	262 671	348 005	426 296	396 573	349 204	426 296	396 573
B FS181 Masilonyana	72 352	086 62	85 192	72 352	086 62	85 192	105 072	121 394	127 560	105 251	121 394	127 560
B FS182 Tokologo	38 552	42 614	45 393	38 552	42 614	45 393	108 127	86 923	73 707	108 306	86 923	73 707
	55 333	61 187	65 182	55 333	61 187	65 182	84 826	94 616	100 873	85 005	94 616	100 873
B FS184 Matjhabeng	390 659	7	460 936	390 659	432 635	460 936	582 481	652 551	692 803	584 148	652 551	692 803
B FS185 Nala	120 920	133 667	142 372	120 920	133 667	142 372	169 456	191 464	203 394	169 635	191 464	203 394
C DC18 Lejweleputswa District Municipality	93 735	97 203	101 043	93 735	97 203	101 043	96 741	99 453	103 143	97 224	99 453	103 143
Total: Lejweleputswa Municipalities	771 551	847 286	900 117	771 551	847 286	900 117	1 146 702	1 246 400	1 301 480	1 149 568	1 246 400	1 301 480
B FS191 Setsoto	147 875	163 504	174 167	147 875	163 504	174 167	222 270	244 410	259 358	225 817	244 410	259 358
B FS192 Dihlabeng	114 851	127 105	135 413	114 851	127 105	135 413	173 566	190 613	205 627	173 749	190 613	205 627
B FS193 Nketoana	195 69	76 929	81 951	29 567	76 929	81 951	100 208	113 327	120 274	100 387	113 327	120 274
FS194	305 453	338 239	360 402	305 453	338 239	360 402	589 859	608 267	619 580	593 212	608 267	619 580
B FS195 Phumelela	49 899		58 766	49 899	55 167	58 766	89 576	93 152	114 770	89 576	93 152	114 770
	59 517		20 096	59 517	37 002	70 096	85 697	66 992	111 658	85 876	66 992	111 658
C DC19 Thabo Mofutsanyana District Municipality	72 399	76 038	79 952	72 399	76 038	79 952	75 786	78 288	82 052	76 459	78 288	82 052
Total: Thabo Mofutsanyana Municipalities	819 560	873 983	960 747	819 560	813 983	960 747	1 332 961	1 395 050	1 513 319	1 341 075	1 395 050	1 513 319
B FS201 Moqhaka	145 181	160 452	170 892	145 181	160 452	170 892	191 250	211 759	226 876	192 113	211 759	226 876
B FS203 Ngwathe	137 311	151 789	161 675	137 311	151 789	161 675	189 291	209 648	224 366	189 778	209 648	224 366
B FS204 Metsimaholo	88 125	97 702	104 114	88 125	97 702	104 114	132 389	152 230	169 311	132 568	152 230	169 311
B FS205 Mafube	67 075	74 135	78 962	67 075	74 135	78 962	103 075	107 336	116 638	103 254	107 336	116 638
C DC20 Fezile Dabi District Municipality	127 132		135 790	127 132	131 308	135 790	130 138	133 558	138 040	130 621	133 558	138 040
Total: Fezile Dabi Municipalities	564 823	615 385	651 434	564 823	615 385	651 434	746 142	814 531	875 230	748 333	814 531	875 230
Total: Free State Municipalities	2 926 447	3 186 684	3 417 559	2 926 447	3 186 684	3 417 559	4 580 288	5 001 113	5 291 438	4 597 515	5 001 113	5 291 438

ANNEXURE W8
EQUITABLE SHARE AND TOTAL ALLOCATIONS TO MUNICIPALITIES

			EQUITABLE SHARE <sup>1</sup>	E SHARE <sup>1</sup>				TOTAL A	TOTAL ALLOCATIONS TO MUNICIPALITIES	TO MUNICIPA	LITHES	
	Nati	National Financial Year	ear	Muni	Municipal Financial Year	Year	Natio	National Financial Year	(ear	Munic	Municipal Financial Year	/ear
	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14
Category Municipanty	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)
GAUTENG												
A EKU Ekurhuleni	1 644 128	1 828 391	1 949 038	1 644 128	1 828 391	1 949 038	3 007 693	3 296 398	3 554 003	3 010 804	3 296 398	3 554 003
A JHB City of Johannesburg	1 897 561	2 134 780	2 276 247	1 897 561	2 134 780	2 276 247	4 973 159	5 029 286	5 288 052	5 068 729	5 029 286	5 288 052
A TSH City of Tshwane	923 020	1 031 527	1 100 611	923 020	1 031 527	1 100 611	2 184 265	3 066 428	3 330 476	2 188 103	3 066 428	3 330 476
B GT421 Emfuleni	539 842	597 880	986 989	539 842	597 880	636 986	733 962	816 668	877 266	737 809	816 668	877 266
B GT422 Midvaal	44 379	49 238	52 480	44 379	49 238	52 480	72 230	79 265	84 034	72 409	79 265	84 034
B GT423 Lesedi	52 626	58 222	62 021	52 626	58 222	62 021	80 815	88 607	93 954	80 815	88 607	93 954
C DC42 Sedibeng District Municipality	220 439	227 627	233 903	220 439	227 627	233 903	294 046	258 377	262 916	294 225	258 377	262 916
Total: Sedibeng Municipalities	857 285	932 967	985 390	857 285	932 967	985 390	1 181 053	1 242 916	1 318 170	1 185 258	1 242 916	1 318 170
B GT481 Mooale City	189 605	210 233	224 025	189 605	210 233	224 025	308 090	354 188	370 493	309 445	354 188	370 493
B GT482 Randfontein	81 638	90 469	96 398	81 638	90 469	96 398	119 825	129 623	139 777	120 829	129 623	139 777
B GT483 Westonaria	87 796	97 063	103 392	87 796	97 063	103 392	168 712	207 695	235 148	171 053	207 695	235 148
B GT484 Merafong City	167 868	185 935	198 124	167 868	185 935	198 124	243 407	262 771	279 049	245 383	262 771	279 049
C DC48 West Rand District Municipality	163 221	168 881	173 109	163 221	168 881	173 109	172 260	185 131	187 923	172 755	185 131	187 923
Total: West Rand Municipalities	690 128	752 581	795 048	690 128	752 581	795 048	1 012 294	1 139 409	1 212 391	1 019 465	1 139 409	1 212 391
Total: Gauteng Municipalities	6 012 123	6 680 246	7 106 335	6 012 123	6 680 246	7 106 335	12 358 464	13 774 437	14 703 092	12 472 359	13 774 437	14 703 092

ANNEXURE W8
EQUITABLE SHARE AND TOTAL ALLOCATIONS TO MUNICIPALITIES

			FOUTTABLE SHAPE	F CHAPE <sup>1</sup>				TOTAL A	LOCATIONS	TOTAL ALLOCATIONS TO MINICIPALITIES	LITTES	
	Nationa	nal Financial Y	ar	Munic	Municipal Financial Year	Year	Natio	National Financial Year	ear	Munici	Municipal Financial Year	Year
Category Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
KWAZULU-NATAL												
A ETH eThekwini	1 595 941	1 773 889	1 891 834	1 595 941	1 773 889	1 891 834	3 120 134	3 315 330	3 513 328	3 156 176	3 315 330	3 513 328
B KZN211 Vulamehlo	31 146	34 452	36 712	31 146	34 452	36712	48 962	53 199	56 713	48 962	53 199	56 713
	24 104	26 677	28 423	24 104	26 677	28 423	54 881	73 027	71 890	54 881	73 027	71 890
, ,	76 198	84 283	808 68	76 198	84 283	808 68	107 682	126 561	134 085	107 682	126 561	134 085
	38 288	42 368	45 149	38 288	42 368	45 149	966 95	64 976	70 021	966 95	64 976	70 021
	23 010	25 457	27 127	23 010	25 457	27 127	41 203	41 284	44 048	41 203	41 284	44 048
B KZN216 Hibiscus Coast C DC21 Hon District Municipality	76 844	85 416	91 051	76 844	85 416	91 051	113 801	162 844	150 387	114 284	162 844	150 387
	505 328	558 995	596 755	505 328	558 995	596755	944 049	1 116 920	1 164 214	945 434	1 116 920	1 164 214
B KZN221 nMshwathi	49 180	54 362	57 915	49 180	54 362	57 915	70 641	84 135	987 06	70 641	84 135	987 06
B KZN222 uMngeni	30 559	33 938	36 175	30 559	33 938	36 175	46 146	53 571	56 764	46 146	53 571	56 764
	19 485	21 553	22 962	19 485	21 553	22 962	31 444	35 608	38 014	31 444	35 608	38 014
, .	21 087	23 335	24 868	21 087	23 335	24 868	37 787	39 421	40 257	37 787	39 421	40 257
	304 835	339 004	361 409	304 835	339 004	361 409	512 447	592 838	581 646	513 203	592 838	581 646
	24 863	27 485	29 282	24 863	27 485	29 282	69 983	099 29	83 276	69 983	099 29	83 276
B KZN227 Richmond	25 849	28 573	30 440	25 849	28 573	30 440	48 761	50 409	55 032	48 761	50 409	55 032
Total: uMgungundlovu Municipalities	761 877	314 348 842 598	901 755	761 877	842 598	901 755	1 217 805	1 379 408	1 423 726	1 218 561	1 379 408	1 423 726
B KZN232 Emnambethi-Ladysmith	93 368	103 324	110 079	93 368	103 324	110 079	156 778	199 425	192 953	156 956	199 426	192 953
B KZN234 Umtshezi	25 843	28 621	30 510	25 843	28 621 28 630	30 510	/9 U63 48 093	81 512 47 621	86 621	/9 063 48 093	81 512	86 621
	55 031	60 875	64 865	55 031	60 875	64 865	88 421	94 462	102 083	88 421	94 462	102 083
B KZN236 Imbabazane	57 268	63 321	67 462	57 268	63 321	67 462	124 903	86 633	92 280	124 903	86 633	92 280
C DC23 Uthukela District Municipality	229 471	253 506	270 913	229 471	253 506	270 913	413 522	484 823	536 309	419 098	484 823	536 309
Total: Uthukela Municipalities	513 982	568 277	606 286	513 982	568 277	606 286	910 779	994 277	1 059 299	916 534	994 278	1 059 299
B KZN241 Endumeni	27 416	30 369	32 368	27 416	30 369	32 368	44 906	45 132	48 167	44 906	45 132	48 167
	62 785	69 485	74 053	62 785	69 485	74 053	91 048	136 499	123 841	91 048	136 499	123 841
	61 218	67 731	72 182	61 218	67 731	72 182	111 587	121 572	138 185	111 587	121 572	138 185
B KZN245 Umvoti	36 785	40 689	43 354	36 785	40 689	43 354	64 326	68 970	73 745	64 326	68 970	73 745
Total: Umzinvathi Municinalities	348 655	385 683	411 503	348 655	385 683	411 503	648 962	787 520	829 152	650 731	787 520	829 152
R KZNOSO Nauvosetla	244 400	270 655	288 355	244 400	270 655	388 355	361 441	390.204	431 316	361 024	390.204	431 316
B KZN253 Emadlangeni	12.256	13 549	14 436	12.256	13 549	14 436	23 170	26 334	28 099	23 170	26 334	28 099
	42 091	46 545	49 591	42 091	46 545	49 591	71 468	77 242	81 865	71 468	77 242	81 865
C DC25 Amajuba District Municipality	88 571	97 343	104 815	88 571	97 343	104 815	157 347	151 732	189 049	157 830	151 732	189 049
Total: Amajuba Municipalities	387 318	428 091	457 198	387 318	428 091	457 198	613 425	645 512	730 328	614 391	645 512	730 328

ANNEXURE W8
EQUITABLE SHARE AND TOTAL ALLOCATIONS TO MUNICIPALITIES

			EOUTTABLE SHARE	E SHARE1				TOTALA	TOTAL ALLOCATIONS TO MUNICIPALITIES	TO MUNICIPA	ALITIES	
	Nationa	onal Financial Y	ear	Muni	Municipal Financial Year	Year	Natio	National Financial Year	(ear	Munic	Municipal Financial Year	Year
Category Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
B KZN261 eDumbe	32 113	35 522	37 849	32 113	35 522	37 849	47 591	53 521	96 850	47 770	53 521	99 99
B KZN262 UPhongolo	52 612	58 176	61 981	52 612	58 176	61 981	80 263	96 095	95 173	80 442	96 095	95 173
	69 224	76 515	81 515	69 224	76 515	81 515	107 207	124 518	125 435	107 386	124 518	125 435
B KZN265 Nongoma	62 891	69 561	74 122	62 891	69 561	74 122	124 243	165 286	147 126	124 422	165 286	147 126
	68 274	75 482	80 420	68 274	75 482	80 420	106 529	111 609	115 175	106 529	111 609	115 175
C DC26 Zululand District Municipality	234 326	258 934	276 726	234 326	258 934	276 726	526 580	618 227	631 249	527 189	618 227	631 249
Total: Zululand Municipalities	519 441	574 191	612 613	519 441	574 191	612 613	992 414	1 169 256	1 181 007	993 739	1 169 256	1 181 007
B KZN271 Umhlabuvalineana	49 682	54 977	58 595	49 682	54 977	58 595	83 829	107 024	109 183	83 829	107 024	109 183
B KZN272 Jozini	64 897	71 806	76 525	64 897	71 806	76 525	103 572	123 497	149 649	103 572	123 497	149 649
	12 396	13 701	14 599	12 396	13 701	14 599	29 870	57 229	39 278	29 870	57 229	39 278
B KZN274 Hlabisa	38 543	24 972	26 589	38 543	24 972	26 589	99 99	79 215	81 039	99 99	79 215	81 039
B KZN275 Mtubatuba	42 041	46 763	49 926	42 041	46 763	49 926	84 313	88 761	102 471	84 313	88 761	102 471
C DC27 Umkhanyakude District Municipality	159 548	176 416	188 500	159 548	176 416	188 500	468 587	525 831	528 097	477 271	525 831	528 097
Total: Umkhanyakude Municipalities	367 107	388 636	414 735	367 107	388 636	414 735	836 776	981 557	1 009 718	845 460	981 557	1 009 718
B KZN281 Mfolozi	37 931	41 898	44 633	37 931	41 898	44 633	55 318	62 470	948	55 318	62 470	98899
B KZN282 uMhlathuze	161 654	179 231	190 994	161 654	179 231	190 994	249 914	327 153	337 216	249 914	327 153	337 216
B KZN283 Ntambanana	17 675	19 514	20 786	17 675	19 514	20 786	46 108	34 207	62 911	46 108	34 207	62 911
	69 515	76 827	81 845	69 515	76 827	81 845	149 473	114 222	161 230	149 473	114 222	161 230
	22 096	24 442	26 043	22 096	24 442	26 043	45 097	39 825	42 496	45 097	39 825	42 496
	44 648	49 392	52 635	44 648	49 392	52 635	82 122	101 440	98 230	82 122	101 440	98 230
C DC28 uThungulu District Municipality	299 178	328 816	354 199	299 178	328 816	354 199	502 706	572 729	625 432	505 139	572 729	625 432
Total: uThungulu Municipalities	652 698	720 120	771 135	652 698	720 120	771 135	1 130 737	1 252 046	1 394 360	1 133 170	1 252 046	1 394 360
B KZN291 Mandeni	57 058	63 057	67 171	57 058	63 057	67 171	95 250	105 547	112 264	95 250	105 547	112 264
B KZN292 KwaDukuza	65 237	72 636	77 471	65 237	72 636	77 471	119 864	124 339	120 376	119 864	124 339	120 376
B KZN293 Ndwedwe	50 875	56 207	59 874	50 875	56 207	59 874	145 640	201 200	228 073	145 640	201 200	228 073
	43 137	47 728	50 862	43 137	47 728	50 862	76 492	90 244	91 517	76 492	90 244	91 517
C DC29 iLembe District Municipality	206 729	228 084	244 242	206 729	228 084	244 242	382 987	453 604	482 478	386 521	453 604	482 478
Total: iLembe Municipalities	423 035	467 712	499 620	423 035	467 712	499 620	820 232	974 934	1 034 708	823 766	974 934	1 034 708
B KZN431 Ingwe	44 868	49 640	52 898	44 868	49 640	52 898	74 571	980 66	90 752	74 571	980 66	90 752
B KZN432 Kwa Sani	10 517	11 617	12 375	10 517	11 617	12 375	28 057	30 767	35 849	28 057	30 767	35 849
B KZN433 Greater Kokstad	42 946	47 560	50 680	42 946	47 560	50 680	61 600	69 282	73 752	61 779	69 282	73 752
B KZN434 Ubuhlebezwe	44 405	49 112	52 331	44 405	49 112	52 331	136 257	152 506	190 099	136 257	152 506	190 099
B KZN435 Umzimkhulu	75 003	82 979	88 423	75 003	82 979	88 423	162 112	242 648	250 131	162 291	242 648	250 131
C DC43 Sisonke District Municipality	182 881	202 113	215 785	182 881	202 113	215 785	350 425	401 157	443 686	351 413	401 157	443 686
Total: Sisonke Municipalities	400 620	443 021	472 492	400 620	443 021	472 492	813 021	995 447	1 084 268	814 367	995 447	1 084 268
Total: KwaZulu-Natal Municipalities	6 476 001	7 151 211	7 635 926	6 476 001	7 151 211	7 635 926	12 048 335	13 612 207	14 424 109	12 112 329	13 612 207	14 424 109

ANNEXURE W8
EQUITABLE SHARE AND TOTAL ALLOCATIONS TO MUNICIPALITIES

			EQUITABLE SHARE	E SHARE				TOTAL	LLOCATIONS	TOTAL ALLOCATIONS TO MUNICIPALITIES	ALITIES	
	Nati	National Financial Y	ear	Muni	Municipal Financial Year	Year	Nati	National Financial Year	Year	Muni	Municipal Financial Year	/ear
Category Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
LIMPOPO												
B LIM331 Greater Givani	116 853	129 173	137 610	116 853	129 173	137 610	162 404	194 418	202 795	162 583	194 418	202 795
B LIM332 Greater Letaba	117 473	129 869	138 353	117 473	129 869	138 353	169 196			170 518	220 374	216 644
B LIM333 Greater Tzaneen	176 879	195 784	208 601	176 879	195 784	208 601	258 946			259 402	283 018	316 692
	53 751	59 394	63 265	53 751	59 394	63 265	88 398		_	88 577	102 744	109 809
	47 176		55 552	47 176	52 144	55 552	79 165		93 036	79 648	87 183	93 036
C DC33 Mopani District Municipality	419 718	463 587	495 550	419 718	463 587	495 550	851 688		942 572	855 562	917 065	942 572
Total: Mopani Municipalities	931 851	1 029 952	1 098 930	931 851	1 029 952	1 098 930	1 609 797	1 804 802	1 881 548	1 616 290	1 804 802	1 881 548
B LIM341 Musina	27 908	30.858	32 874	27 908	30.858	32 874	41 987	47.546		41 987	47 546	50 717
B LIM342 Mutale	37 628		44 316	37 628		44 316	129 705		76 731	130 516	68 014	76 731
	235 608	( )	277 899	235 608	( )	277 899	390 002	4	466 588	391 148	468 702	466 588
LIM344	212 830		250 972	212 830		250 972	284 768			285 073	382 176	380 141
C DC34 Vhembe District Municipality	431 171	476 629	508 893	431 171	476 629	508 893	949 571	1 018 108	998 015	952 577	1 018 108	998 015
Total: Vhembe Municipalities	945 145	1 045 443	1 114 955	945 145	1 045 443	1 114 955	1 796 033	1 984 547	1 972 193	1 801 301	1 984 547	1 972 193
B LIM351 Blouberg	79 413		93 693	79 413	87 909	93 693	123 831	138 610			138 610	149 819
B LIM352 Aganang	0/1/99	73 814	78 641	022 99	73 814	78 641	94 906	114 701	119 505	95 085	114 701	119 505
B LIM353 Molemole	66 171	73 146	77 922	66 171	73 146	77 922	87 926			87 926	99 418	105 612
	350 705	388 419	413 878	350 705	388 419	413 878	692 237	691 794	750 653	697 359	691 794	750 653
	109 337	120 857	128 749	109 337	120 857	128 749	149 331	175 884	184 527	149 814	175 884	184 527
C DC35 Capricorn District Municipality	365 229	402 266	431 885	365 229	402 266	431 885	655 450	644 136	670 807	657 585	644 136	670 807
Total: Capricorn Municipalities	1 037 625	1 146 411	1 224 768	1 037 625	1 146 411	1 224 768	1 803 681	1 864 542	1 980 923	1 811 842	1 864 542	1 980 923
B LIM361 Thabazimbi	53 095		62 602	53 095		62 602	92 092	106 307	114 578	92 271	106 307	114 578
B LIM362 Lephalale	73 300		84 829	73 300		84 829	125 823			127 100	138 675	146 411
B LIM364 Mookgopong	22 822		26 918	22 822	25 260	26 918	40 399			40 399	49 019	46 682
	50 075		28 990	50 075		58 990	86 629		105 205	808 98	99 207	105 205
B LIM366 Bela Bela	39 790		46 885	39 790		46 885	28 096		69 277	58 275	65 181	69 277
B LIM367 Mogalakwena	225 142	249 257	265 591	225 142	249 257	265 591	458 376	500 382	522 698	459 223	500 382	522 698
Total: Waterberg Municipalities	552 105		639 917	552 105	603 417	639 917	952 813	1(	1 101 913	955 957	1 053 018	1 101 913
B LIM471 Ephraim Mogale	60 529	968 892	71 260	60 529	66 895	71 260	86 850		100 406	87 029	92 513	100 406
B LIM472 Elias Motsoaledi	114 134	126	134 349	114 134	126 125	134 349	156 426	173 249		157 756	173 249	185 533
B LIM473 Makhuduthamaga	126 339	139 682	148 812	126 339	139 682	148 812	188 818			189 564	196 130	219 856
B LIM474 Fetakgomo	40 562		47 774	40 562	44 840	47 774	576 19		76722	68 154	70 083	76 722
B LIM475 Greater Tubatse	114 137	126 220	134 480	114 137	126 220	134 480	200 791	225 976	238 297	201 433	225 976	238 297
C DC47 Greater Sekhukhune District Municipality	330 877	365 664	390 926	330 877	365 664	390 926	873 923	1 013 871	1 101 589	880 088	1 013 871	1 101 589
Total: Greater Sekhukhune Municipalities	786 577	869 427	927 602	786 577	869 427	927 602	1 574 783	1 771 822	1 922 402	1 584 025	1 771 822	1 922 402
Total: Limpopo Municipalities	4 253 303	4 694 650	5 006 171	4 253 303	4 694 650	5 006 171	7 737 108	8 478 730	8 858 979	7 769 416	8 478 730	8 858 979
	_	l										

ANNEXURE W8
EQUITABLE SHARE AND TOTAL ALLOCATIONS TO MUNICIPALITIES

			EQUITABLE SHARE	E SHARE <sup>1</sup>				TOTAL A	FOTAL ALLOCATIONS TO MUNICIPALITIES	TO MUNICIPA	ALITIES	
	Natio	National Financial Yea	ear	Muni	Municipal Financial Year	Year	Natio	National Financial Year	ear	Munic	Municipal Financial Year	ear
Category Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
MPUMALANGA												
B MP301 Albert Luthuli	141 281	156 282	166 510	141 281	156 282	166 510	238 583	257 202	284 589	239 066	257 202	284 589
MP302	93 142	103 000	109 722	93 142	103 000	109 722	138 105	150 871	156 870	138 284	150 871	156870
B MP303 Mkhondo	88 732	98 160	104 590	88 732	98 160	104 590	150 155	164 668	176 766	150 638	164 668	176766
B MP304 Pixley Ka Seme	74 975	82 907	88 318	74 975	82 907	88 318	113 068	117 520	124 759	117 806	117 520	124 759
B MP305 Lekwa	69 69	77 364	82 412	69 69	77 364	82 412	111 165	124 267	131 866	111 648	124 267	131 866
B MP306 Dipaleseng	39 319	43 465	46 299	39 319	43 465	46 299	63 782	67 507	71 587	64 265	67 507	71 587
B MP307 Govan Mbeki	171 429	190 045	202 507	171 429	190 045	202 507	255 920	294 456	312 034	256 707	294 456	312 034
C DC30 Gert Sibande District Municipality	246 282	254 200	261 625	246 282	254 200	261 625	251 952	256 450	263 875	253 663	256 450	263 875
Total: Gert Sibande Municipalities	925 119	1 005 423	1 061 984	925 119	1 005 423	1 061 984	1 322 729	1 432 940	1 522 344	1 332 076	1 432 940	1 522 344
B MP311 Victor Khanye	45 078	49 867	53 124	45 078	49 867	53 124	90 611	102 519	111 951	90 790	102 519	111 951
B MP312 Emalahleni	163 854	181 726	193 663	163 854	181 726	193 663	263 611	285 315	302 527	264 094	285 315	302 527
B MP313 Steve Tshwete	77 312	85 857	91 509	77 312	85 857	91 509	133 269	149 007	167 155	133 898	149 007	167 155
B MP314 Emakhazeni	31 562	34 885	37 159	31 562	34 885	37 159	47 724	53 151	56353	47 903	53 151	56353
B MP315 Thembisile	196 665	217 431	231 624	196 665	217 431	231 624	303 732	341 199	359 363	304 215	341 199	359 363
B MP316 Dr JS Moroka	205 518	227 206	242 036	205 518	227 206	242 036	331 244	359 025	378 064	332 139	359 025	378 064
C DC31 Nkangala District Municipality	291 974	301 317	309 985	291 974	301 317	309 985	299 974	320 118	334 014	301 595	320 118	334 014
Total: Nkangala Municipalities	1 011 962	1 098 290	1 159 101	1 011 962	1 098 290	1 159 101	1 470 164	1 610 333	1 709 427	1 474 633	1 610 333	1 709 427
D MD221 Thoke Chanca	909 99	72 750	70 562	909 99	72 750	70 563	00 101	10.9 503	115 200	290 00	108 503	115 200
B MD322 Mhombela	787 081	212 447	332 931	282 081	21.2 747	337 931	575 887	572 655	613 100	516 813	572 655	613 109
	190 202	77 758	166 266	13 164	77 758	160 260	80.757	000 50	201 610	90.031	000 210	201 010
	23.4.566	250 743	987 970	734 566	250 743	987 977	424 380	443 193	478 296	424 926	443 193	478 296
	398 491	440 908	469 771	398 491	440 908	469 771	799 801	859 887	857 264	800 426	859 887	857 264
	170414	178 778	185 817	170 414	178 778	185 817	178 727	181 278	214 317	179 008	181 278	214 317
Total: Ehlanzeni Municipalities	1 195 411	1 313 385	1 394 754	1 195 411	1 313 385	1 394 754	2 128 031	2 251 319	2 368 712	2 131 101	2 251 319	2 368 712
Total: Monmalanga Municipalities	3 132 492	3 417 098	3 615 839	3 132 492	3 417 098	3 615 839	4 920 925	5 294 592	5 600 484	4 937 811	5 294 592	5 600 484

ANNEXURE W8
EQUITABLE SHARE AND TOTAL ALLOCATIONS TO MUNICIPALITIES

			EQUITABLE SHARE	E SHARE				TOTAL A	LLOCATIONS	TOTAL ALLOCATIONS TO MUNICIPALITIES	ALITIES	
	Natio	National Financial Y	ear	Munic	Municipal Financial	Year	Nati	National Financial Year	Year	Munic	Municipal Financial Year	ear
Category Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
NORTHERN CAPE												
NC061	10415		12 257	10 415	11 507	12 257			33 031	19 768	40 456	33 031
NC062	28 625	31 609	33 661	28 625	31 609	33 661	70 284		164 324	70 284	91 035	164 324
	10 907		12 824	10 907	12 041	12 824			27 832	22 781	26 054	27 832
NC065	17 683		20 817	17 683	19 543	20 817			45 717	31 941	34 615	45 717
NC066	11 544	12 740	13 568	11 544	12 740	13 568			28 254	25 466	26 448	28 254
NC067	10 921	12 053	12 837	10 921	12 053	12 837			36 820	23 095	43 795	36 820
C DC6 Namakwa District Municipality	29 792	31 219	32 445	29 792	31 219	32 445			34 695	38 440	33 469	34 695
Total: Namakwa Municipalities	288 611	130 712	138 409	119 887	130 712	138 409	229 573	295 872	370 672	231 776	295 872	370 672
B NC071 Thursts	15 660	17 3 13	18 441	15 660	17 3 13	18 441	287.70	31 183	33 705	77 967	31 183	33 705
	200 51		30 714	26,005	78 926	30 71 /	125 060	91 163	70 466	135 060	91 163	70 466
NC072	20 033	22 647	35 837	30 439	22 647	36 714	153 060	71.056	62 02	153 000	71.056	62 026
NC074	10.456		12 310	10.466	11 565	12 310		73.861	25 020	21,008	73.861	25 003
NC075	10 460		15 319	12 012	11 363	12.319	24.280	25 601	23 093	24 772	26 401	25 093
NCO75	12,912		001 51	12 912	14 25/	15 185		20 05	28 313	74 / 17	26 491	28 21 20
NC0/6	12 849		15 120	12 849	14 194	15 120	40.907	50 094	37 160	40.907	50 094	37 160
	17829		21 003	17.829	19 / 15	21 003	31 336	34 969	37 082	31 839	34 969	37 082
20	31 /26	35 106	37 408	31 /26	35 106	37.408	55 036	180.65	04 63 /	25 036	180 65	04 63 /
C DC7 Pixley Ka Seme District Municipality	24 727	26 159	27 319	24 727	26 159	27 319	27 177	28 609	34 569	27 177	28 609	34 569
Total: Pixley Ka Seme Municipalities	182 713	200 786	213 341	182 713	200 786	213 341	427 158	406 936	392 643	428 482	406 936	392 643
D NCOOL Miss	000	0140	01101	000	0130	01	25000	00000	000	2000	22.220	0000
	0.020	`	10 140	02020	15 050	10 140	1000	622.62	72 000	47.07	622.62	72 000
NC082	41 330		46 933	41 330	51 070	46 933	70 588	187.80	83 281	20 588	187.60	83.281
NC084	13 955		16 422	13 955	15 416	16 422	27 744	31 732	33.853	27 744	31 732	33.853
NC085	22,454		26 473	22.454	24 844	26 473	72 579	43 656	46 471	72.758	43 656	46 471
NC086	12.918		15 194	12.918	14 266	15 194	22.935	25 961	27.756	22 935	25 961	27.756
DC8	39 818		43 304	39 818	41 647	43 304	50 746	43 897	45 554	55 190	43 897	45 554
otal: Siyanda	185 443	202	214 900	185 443	202 712	214 900	330 046	316 545	335 643	334 669	316 545	335 643
NC091	132 176		156 193	132 176	146 577	156 193	230 869	229 034	237 034	238 912	229 034	237 034
NC092	40 046	44 285	47 179	40 046	44 285	47 179	64 104	72 262	91 269	64 104	72 262	91 269
NC093	25 546		30 080	25 546	28 238	30 080	50 327	55 360	58 27 5	50 327	55 360	58 275
+	56 719		166 797	56 719	62 706	164 99	105 283	112 155	118 091	105 462	112 155	118 091
C DC9 Frances Baard District Municipality	79 281		88 994	79 281	86 061	88 994	88 319	88 511	126 244	91 613	88 511	126 244
Total: Frances Baard Municipalities	333 769	367 868	389 243	333 769	367 868	389 243	538 903	557 323	630 911	550 419	557 323	630 911
B NC451 Moshaweng	63 808	70 638	75 287	63 808	70 638	75 287		207 848	148 098	157 990	207 848	148 098
NC452	58 219		68 626	58 219	64 408	68 626		141 053	148 462	122 916	141 053	148 462
NC453	18 283	20 213	21 532	18 283	20 213	21 532		51 400	42 460	43 521	51 400	42 460
C DC45 John Taolo Gaetsewe District Municipality	50 939		56 721	50 939	54 196	56 721		62 091	62 556	65 172	62 091	62 556
Total: John Taolo Gaetsewe Municipalities	191 248	209 455	222 165	191 248	209 455	222 165	385 382	462 393	401 575	389 600	462 393	401 575
Total: Northern Cane Municipalities	1 013 059	1 111 533	1 178 060	1 013 059	1 111 533	1 178 060	1 911 062	2 039 069	2 131 445	1 934 946	2 039 069	2 131 445
LOGAL MANNER IN Super standard purchase											1	

ANNEXURE W8
EQUITABLE SHARE AND TOTAL ALLOCATIONS TO MUNICIPALITIES

			EOUTTABLE SHARE	E SHARE <sup>1</sup>				TOTAL A	FOTAL ALLOCATIONS TO MUNICIPALITIES	TO MUNICIPA	ALITIES	
	Natic	National Financial Y	l ä	Muni	Municipal Financial Year	Year	Natio	National Financial Year	Year	Munic	Municipal Financial Year	ear
Category Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
NORTH WEST												ì
B NW371 Moretele	138 282	152 916	162 918	138 282	152 916	162 918	241 961	272 827	288 741	242 840	272 827	288 741
B NW372 Madibeng	247 326	273 870	291 812	247 326	273 870	291 812	495 496	546 631	598 479	497 204	546 631	598 479
B NW373 Rustenburg	231 669	256 655	273 476	231 669	256 655	273 476	622 638	721 843	872 512	626 796	721 843	872 512
B NW374 Kgetlengrivier	38 032	42 057	44 806	38 032	42 057	44 806	57 029	64 740	68 499	57 208	64 740	68 499
B NW375 Moses Kotane	203 756	225 305	240 017	203 756	225 305	240 017	329 267	372 937	397 563	329 888	372 937	397 563
C DC37 Bojanala Platinum District Municipality	232 094	239 987	247 826	232 094	239 987	247 826	235 435	260 237	275 076	236 086	260 237	275 076
Total: Bojanala Platinum Municipalities	1 091 159	1 190 789	1 260 855	1 091 159	1 190 789	1 260 855	1 981 826	2 239 215	2 500 871	1 990 022	2 239 215	2 500 871
B NW381 Ration	59 576	65 873	70 182	59 576	65 873	70 182	89 351	117 346	126 266	89 656	117 346	126 266
	56 186	62 108	66 164	56 186	62 108	66 164	95 095	92 497	98 379	95 578	92 497	98 379
B NW383 Mafikeng	109 725	121 500	129 460	109 725	121 500	129 460	178 052	215 198	211 010	178 535	215 198	211 010
B NW384 Ditsobotla	969 69	666 92	82 017	969 69	666 92	82 017	107 601	122 530	127 467	108 084	122 530	127 467
B NW385 Ramotshere Moiloa	70 458	77 882	82 968	70 458	77 882	82 968	118 520	135 586	135 209	118 825	135 586	135 209
C DC38 Ngaka Modiri Molema District Municipality	357 615	394 150	422 535	357 615	394 150	422 535	548 501	601 800	666 473	549 478	601 800	666 473
Total: Ngaka Modiri Molema Municipalities	723 256	798 513	853 327	723 256	798 513	853 327	1 137 121	1 284 955	1 364 803	1 140 157	1 284 955	1364803
B NW392 Naledi	29 119	32 204	34 305	29 119	32 204	34 305	55 589	53 101	58 354	55 768	53 101	58 354
B NW393 Mamusa	28 210	31 172	33 205	28 210	31 172	33 205	48 447	57 283	63 293	48 626	57 283	63 293
B NW394 Greater Taung	88 632	97 982	104 392	88 632	97 982	104 392	159 016	170 176	173 890	159 499	170 176	173 890
B NW396 Lekwa-Teemane	24 989	27 613	29 413	24 989	27 613	29 413	45 026	49 306	52 459	45 205	49 306	52 459
B NW397 NW397	58 504	64 639	68 853	58 504	64 639	68 853	94 311	104 635	111 379	94 490	104 635	111 379
C DC39 Dr Ruth Segomotsi Mompati District Municipality	188 347	207 978	179 335	188 347	207 978	179 335	360 416	403 366	395 937	362 798	403 366	395 937
Total: Dr Ruth Segomotsi Mompati Municipalities	417 801	461 588	449 503	417 801	461 588	449 503	762 806	837 867	855 312	766 387	837 867	855 312
B NW401 Ventersdom	40 951	45 287	48 247	40 951	45 287	48 247	688 99	81 961	86 332	67 372	81 961	86 332
B NW402 Tlokwe	76 801	85 191	90 788	76 801	85 191	90 788	119 997	150 264	155 047	120 480	150 264	155 047
B NW403 City of Matlosana	303 560	336 019	357 961	303 560	336 019	357 961	435 429	476 783	516866	439 097	476 783	516 866
B NW404 Maquassi Hills	69 259	76 618	81 631	69 259	76 618	81 631	111 928	121 842	129 902	112 411	121 842	129 902
C DC40 Dr Kenneth Kaunda District Municipality	153 622	158 939	155 901	153 622	158 939	155 901	156 019	161 189	158 151	156 198	161 189	158 151
Total: Dr Kenneth Kaunda Municipalities	644 194	702 054	734 528	644 194	702 054	734 528	890 263	992 040	1 046 298	895 559	992 040	1 046 298
Total: North West Municipalities	2 876 410	3 152 944	3 298 214	2 876 410	3 152 944	3 298 214	4 772 016	5 354 077	5 767 284	4 792 125	5 354 077	5 767 284

ANNEXURE W8
EQUITABLE SHARE AND TOTAL ALLOCATIONS TO MUNICIPALITIES

			EOUTTABLE SHARE	E SHARE <sup>1</sup>				TOTAL A	TOTAL ALLOCATIONS TO MUNICIPALITIES	TO MUNICIPA	ALITIES	
	Nati	National Financial Year	ear	Muni	Municipal Financial Year	Year	Natio	National Financial Year	Tear	Munic	Municipal Financial Year	Year
Category Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
WESTERN CAPE												
A CPT City of Cape Town	970 473	1 090 987	1 163 859	970 473	1 090 987	1 163 859	3 788 633	3 836 201	4 012 217	3 797 128	3 836 201	4 012 217
B WC011 Matzikama	32 066		37 821	32 066	35 495	37 821	55 591	59 412	61 726	55 770	59 412	61 726
B WC012 Cederberg	23 008		27 106	23 008	25 447	27 106	62 065	43 900	50 531	62 244	43 900	50 531
WC013	19 898		23 445	19 898	22 010	23 445	32 137	36 008	38 190	32 316	36 008	38 190
	29 296		34 724	29 296	32 575	34 724	51 500	53 862	58 308	51 500	53 862	58 308
B WC015 Swartland	22 887	25 392	27 058	22 887	25 392	27 058	40 544	44 915	47 847	40 544	44 915	47 847
otal: West C	195 808	211 911	222 306	195 808	211 911	222 306	318 445	323 139	388 003	319 161	323 139	388 003
D W/COOD Witnesshows	17501	14 0 5 5	107.77	10 501	14 055	107.77	330 00	127 03	CF3 CF	330 00	127 03	CT3 CT
	40.361	4 633	74 007	40.301	69 437	74 007	709 201	112 673	136415	108 086	112 673	136415
WC024	36 977		74 007	36 977	41 252	43 992	59 972	68 781	73 023	59 972	68 781	73 023
WC025	56 804		67 160	56 804	63 024	67 160	660 88	98 371	144 112	88 278	98 371	144 112
WC026	45 172	49 943	53 197	45 172	49 943	53 197	68 176	74 780	690 77	68 176	74 780	690 77
C DC2 Cape Winelands District Municipality	200 739	207 135	210 408	200 739	207 135	210 408	205 779	209 385	212 658	205 779	209 385	212 658
Total: Cape Winelands Municipalities	442 730	475 646	496 545	442 730	475 646	496 545	608 288	633 461	716 850	608 646	633 461	716 850
B WC031 Theewaterskloof	46 935	51 892	55 272	46 935	51 892	55 272	76 585	83 416	88 589	77 152	83 416	88 589
B WC032 Overstrand	31 156	34 666	36 957	31 156	34 666	36 957	53 974	53 853	60 752	54 153	53 853	60 752
	14 805		17 477	14 805	16 402	17 477	25 681	28 761	30 502	25 860	28 761	30 502
WC034	16 648		19 629	16 648	18 425	19 629	30 331	34 781	45 650	30 510	34 781	45 650
C DC3 Overberg District Municipality	41 692	43 229	44 363	41 692	43 229	44 363	45 589	45 479	46 613	45 768	45 479	46 613
Total: Overberg Municipalities	151 236	164 615	173 698	151 236	164 615	173 698	232 159	246 290	272 106	233 442	246 290	272 106
R WCM1 Kannaland	16.840	18 619	19.833	16 840	18 619	19.833	33 661	44 613	39 874	33.840	44 613	39.874
	23 227	25 750	27 438	23 227	25 750	27 438	44 234	56 810	68 744	44 413	56810	68 744
WC043	36 942		43 589	36 942	40 915	43 589	59 735	61 161	71 381	60 040	61 161	71 381
WC044	72 201		85 706	72 201	80 365	85 706	135 784	132 428	141 651	136 577	132 428	141 651
WC045	37 618		44 281	37 618	41 577	44 281	72 040	75 828	74 631	72 719	75 828	74 631
WC047	20 415	22 696	24 195	20 415	22 696	24 195	56 294	56 735	58 151	56 473	56 735	58 151
B WC048 Knysna	26 68/	126 954	31 695	26 68/	29 /26	31 693	26 660	6/012	131 951	26 660	6/012	131 951
otal: Eden N	356 842		406 437	356 842	386 601	406 437	587 360	623 790	669 250	589 674	623 790	669 250
B WC051 Laingsburg	8 239	6 097	069 6	8 239	6 097	069 6	17 399	118 61	20 913	17 399	118 61	20 913
B WC052 Prince Albert	9 200	10 149	10 809	9 200	10 149	10 809	18 295	20 778	22 009	18 295	20 778	22 009
WC053	28 956		34 194	28 956	32 081	34 194	72 639	89 584	82 137	72 282	89 584	82 137
C DC5 Central Karoo District Municipality	11 535		12 906	11 535	12 304	12 906	13 932	14 554	15 156	16 842	14 554	15 156
Total: Central Karoo Municipalities	57 931	63 632	67 599	57 931	63 632	62 299	122 266	144 726	140 214	124 819	144 726	140 214
Total: Western Cape Municipalities	2 175 019	2 393 393	2 530 444	2 175 019	2 393 393	2 530 444	5 657 152	5 807 607	6 198 641	5 672 871	5 807 607	6 198 641
Unallocated:							520 000	405 000	450 000	470 000	330 000	350 000
National Total	34 107 901	37 573 306	30 060 288	34 107 901	902 225 28	39 090 08	82 500 428	71 768 388	880 129 91	65 883 055	71 603 388	76 557 988

National Total

Includes equitable share formula allocations, RSC levies replacement and special contribution towards Councillor remuneration, but excludes the sharing of the general fuel levy with metropolitan municipalities.

See Appendix WI.

## APPENDIX W1: APPENDIX W1: APPENDIX TO SCHEDULE 3: EQUITABLE SHARE ALLOCATIONS TO MUNICIPALITIES

# (EQUITABLE SHARE FORMULA ALLOCATIONS + RSC LEVIES REPLACEMENT + SPECIAL CONTRIBUTION TOWARDS

COUNCILLOR REMUNERATION)

(National and Municipal Financial Year)

APPENDIX W1:
APPENDIX TO SCHEDULE 3: EQUITABLE SHARE ALLOCATIONS TO MUNICIPALITIES
EQUITABLE SHARE FORMULA ALLOCATIONS + RSC LEVIES REPLACEMENT + SPECIAL CONTRIBUTION TOWARDS COUNCILLOR REMUNERATION

			Equitable Share Formula	Formula				J50	DSC Loxios Donlacoment	mont			Stacial Contribution towards Councillor Bannaration	bution towards	Comeiller Be	noiterounu	
	Natio	lal Y	ear	Munici	cial Y		National 1	<u>.</u>		ŀ₽	icial Year	Nati	National Financial Year	Year	Municip	al Financial N	ear
Category Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12   20 (R'000)   (R	2012/13 20. (R'000) (R'	2013/14 2011/12 (R'000) (R'000)	2011/12 2012/13 (R'000) (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	712 2012/13 201. 30) (R'000) (R'C	2013/14 (R'000)
EASTERN CAPE																	
A BUF Buffalo City A NMA Nelson Mandela Bay	583 628 656 653	644 542 730 416	686 869 778 401	583 628 656 653	644 542 730 416	686 869 778 401											
B EC101 Camdeboo	32 107	35 541	37 866	32 107	35 541	37 866						986	1 041	1011	986	1 041	1011
	13 059	14 459	15 408	13 059	14 459	15 408						694	732	775	694	732	775
	57 458	63 631	67 795	57 458	63 631	67 795						1 684	1 779	1 882	1 684	1 779	1 882
B EC105 Ndlambe B EC106 Sundays River Valley	48 319	53 542 34 327	36 587	30 979	53 542 34 327	57 054 36 587						1 203	1 270	1 344	1 203	1 270	1 259
	14 459	16 019	17 073	14 459	16 019	17 073						694	732	775	694	732	775
B EC108 Kouga B EC109 Kou-kamma	39 294 25 136	43 743 27 824	29 646 29 646	39 294 25 136	43 743 27 824	46 644 29 646						1 744	1842	1 949	1 744	1 842 818	1 949
C DC10 Cacadu District Municipality	12 786	13 554	14 468	12 786	13 554	14 468	52 629	54 207	55 833	52 629 54 207	07 55 833		1 906	2016	1 805	1 906	2016
Total: Cacadu Municipalities	307 821	340 516	362 893	307 821	340 516	362 893	52 629	54 207				11 485	12 128	12 831	11 485	12 128	12 831
	101 569		120 073	101 569	112 628	120 073						3 669	3 875	4 100	3 669	3 875	4 100
EC122	132 340		156 210	132 340	146 580	156 210						3 730	3 938	4 167	3 730	3 938	4 167
B EC123 Great Kei	27 293	30 217	32 198	27 293	30 217	32 198						915	967	1 023	915	967	1 023
	53 636		63 268	53 636	59 377	63 268						1 504	1588	1 680	1 504	1 588	1 680
	73 632		969 98	73 632	81 400	969 98						2 466	2 604	2 755	2 466	2 604	2 755
B EC128 Nxuba C DC12 Amathole District Municipality	321 457	356 075	379 078	321 457	356 075	379 078	208 824		248 104 2	208 824 227 6	248 104		666	679	200	ckc	679
jici	805 201		949 877	805 201	891 726	949 877	Ш	227 618		208 824 227 618		15 254	16 108	17 042	15 254	16 108	17 042
B EC131 Inxuba Yethemba	34 687	38 391	40 900	34 687	38 391	40 900						1 083	1 143	1 210	1 083	1 143	1 210
	20 887	23 119	24 633	20 887	23 119	24 633						991	1 046	1 107	166	1 046	1 107
	15 054	16 667	17 759	15 054	16 667	17 759						694		2775	694	732	775
B EC134 Lukhanji B EC135 Intsika Yethu	92 814 75 447	102 699 83 491	109 409 88 953	75 447	102 699 83 491	109 409 88 953						3 248 2 957		3 304	3 248 2 957	3 123	3 304
	59 226	65 584	688 69	59 226	65 584	688 69						2 394		2 674	2 394	2 528	2 674
B EC137 Engcobo	64 190	71 212	75 931	64 190	71 212	75 931						2 406	2 541	2 688	2 406	2 541	2 688
	281 451	311 594	331 939	281 451	311 594	331 939	44 457	48 458		44 457 48 458	58 52 819		1 080	641	1 023	1 000	C+I I
ΙÈΙ	868 929	749 476	798 547	868 929	749 476	798 547	44 457	48 458	52 819			14 795	15 624	16 530	14 795	15 624	16 530
B EC141 Elundini	62 461	69 234	73 801	62.461	69 234	73 801						2,394		2 674	2,394	2.528	2.674
EC142	76965	85 218	608 06	76 965	85 218	608 06						2 226		2 487	2 226	2 350	2 487
	19 379	21 459	22 865	19 379	21 459	22 865						845	892	944	845	892	944
B EC144 Garlep C DC14 Ukhahlamba District Municipality	21 463	23 /63	157 462	21 463	23 763	25 520	14 127	15 398	16 784	14 127 15 3	16 784	4 4	1 525	1 613	4 4 4	1 525	1613
	313 728	347 462	370 257	313 728	347 462	370 257	14 127	15 398		14 127 15 398			8 038	8 505	7 612	8 038	8 505
B EC153 Ngquza Hill	93 453	103 604	110 444	93 453	103 604	110 444						3 730	3 938	4 167	3 730	3 938	4 167
EC154	56 791	62 962	67 120	56 791	62 962	67 120						2 746		3 068	2 746	2 900	3 068
B EC155 Nyandeni B EC156 Milontlo	108 256	119 947	127 844	108 256	119 947	127 844						3 669	3 875	3 495	3 669	3 875	3 495
EC157	153 086	169 585	180 725	153 086	169 585	180 725											
C DC15 O.K. Lambo District Municipality Total: O R Tamba Municipalities	362 884	944 802	1 006 522	302 884 861 555	944 802	1 006 522	50 330	54 870 54 870	59 808 59 808	50 339 54 870	20 59 808	13 273	14 016	14.829	13 273	14 016	14.820
Total: O.K. ranno Municipanues	CCC 100	700 ++4	770 000 1	CCC 100	700 ++4	776 000 1	ACC AC	0/0+0						14 073	CITCI	010 +1	14 027
EC441	89 321	99 001	105 529	89 321	99 001	105 529						3 128	3 303	3 495	3 128	3 303	3 495
B EC442 Umzimvubu	89 745	99 4/6	106 038	89 745	994/6	106 038						3 248		3 629	3 248	3 430	3 629
EC152	52 394	58 077	61 908	52 394	58 077	61 908						2 535	2 676	2 832	2 535	2 676	2 832
C DC44 Alfred Nzo District Municipality	238 914	265 981	283 870	238 914	265 981	283 870	24 215	26 394	28 769	24 215 26 394	94 28 769			2 688	2 406	2 541	2 688
Total: Affred Nzo Municipalities	564 940	627 352	220 699	564 940	627 352	220 699	24 215	26 394				15 612	16 486	17 442	15 612	16 486	17 442
Total: Eastern Cape Municipalities	4 770 425	5 276 293	5 622 444	4 770 425	5 276 293	5 622 444	394 591	426 945	462 118 3	394 591 426 945	45 462 118	78 030	82 400	87 179	78 030	82 400	87 179
															•		

APPENDIX W1:
APPENDIX TO SCHEDULE 3: EQUITABLE SHARE ALLOCATIONS TO MUNICIPALITIES
EQUITABLE SHARE FORMULA ALLOCATIONS + RSC LEVIES REPLACEMENT + SPECIAL CONTRIBUTION TOWARDS COUNCILLOR REMUNERATION

		-	Fanitable Share Formula	Formula				RSC	RSC Levies Renlacement	ement			Special Contribution towards Councillor Remuneration	ibution toward	S Corneillor R	emmeration	
	Natio	National Financial Year	ar	Munici	Municipal Financial Year	ear	National	National Financial Year		Municipal Financial Year	icial Year	Nati	National Financial Year	Year	Munici	Municipal Financial Year	/ear
Category Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 2 (R'000) (4	2012/13 20, (R'000) (R'	2013/14 2011/12 (R'000) (R'000)	712 2012/13 90) (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
FREE STATE																	
A MAN Mangaung	546 417	603 089	642 590	546 417	603 089	642 590											
B FS161 Letsemeng	44 437	49 164	52 375	44 437	49 164	52 375						774	818	865	774	818	865
	78 432	86 751	92 410	78 432	86 751	92 410						902	953	_	902	953	1 008
	44 858	49 633	52 876	44 858	49 633	52 876						774	818		774	818	865
	32 725	36 208	38 574	32 725	36 208	38 574	0	-		000			595	-	563	595	629
C DC16 Xhariep District Municipality Total: Xharian Municipalities	200 175	21/ 6	246 611	200 175	21/6	246 611	10 779	11 100	11 434	001 11 677 01	00 11 434	4 141	4 373	4621	4 141	4 373	462.1
Total Analiep intuite panties	C11 C07	107	710 047	C11 C07	004 107	110.047	(11) OT	007 17	101				CICT	070+	1111	010	070+
	71 149	78 710	83 847	71 149	78 710	83 847						1 203	1 270	1	1 203	1 270	1 344
	37 759	41 777	44 507	37 759	41 777	44 507						793	837		793	837	988
	54 431	60 234	64 174	54 431	60 234	64 174						902	953	1 008	905	953	1 008
	390 659	432 635	460 936	390 659	432 635	460 936											
B FS185 Nala	119 476	132 142	140 759	119 476	132 142	140 759	000	72	100 30	000 22	100.35	1 444	1 525	1 613	1 444	1 525	1 613
12	706 361	177 57	24 047	706 361	122 52	24 041	71 676	73 062				1 347	1 595	1951	1 342	1 595	1 951
Total: Lejwelepuiswa iviuncipanues	100 000	(T) 90/	619 019	196 560	/100 / 173	200 610	070 1/	79% 61					4 2002		7 + 0 +	4 200	100+
B FS191 Setsoto	145 769	161 280	171 815	145 769	161 280	171 815						2 105	2 223	2352	2 105	2 223	2 352
	112 444	124 564	132 725	112 444	124 564	132 725						2 406	2 541	2 688	2 406	2 541	2 688
	68 484	75 786	80 741	68 484	75 786	80 741						1 083	1 143	1 210	1 083	1 143	1 210
	305 453	338 239	360 402	305 453	338 239	360 402						1					
	48 996	54 214	57.758	48 996	54 214	57 758						902	953	1 008	902	953	1 008
B FS196 Mantsopa	58 494	35 923	68 953	58 494	35 923	68 953	40 130	102.01	00112	40 621	100		1 080		1 023	1 080	1 143
Ιź	763 862	816 422	901 237	763 862	816 422	901 237	48 179	49 621				7 519	7 940	8 401	7 519	7 940	8 401
TEACOL METERS		700	000			203						000 6	,,,,	000	000	,	0,00
	270101	0.72.01.	100.001	27.04.01	140.011	107 024						2 25.0	0 170	2000	2 2 2 2	0.170	3,000
B FS203 Ngwame	134 965	97 702	104 114	88 125	97 702	104 114						7 340	7/47	7 071	2 340	7/47	7 971
	66 053	73 055	77 820	66 053	73 055	77 820						1 023	1 080	1 143	1 023	1 080	1 143
C DC20 Fezile Dabi District Municipality	12 105	12 831	13 758	12 105	12 831	13 758	115 027	118 477	122 031	115 027 118 477	122 031						
Total: Fezile Dabi Municipalities	443 420	490 175	522 279	443 420	490 175	522 279	115 027	118 477	122 031	115 027 118 477	77 122 031	6 376	6 733	7 124	926 9	6 733	7 124
Total: Free State Municipalities	2 658 256	2 909 872	3 131 781	2 658 256	2 909 872	3 131 781	245 813	253 180 2	2 2 2 2	245 813 253 180	80 260 776	22 378	23 632	25 002	22 378	23 632	25 002

APPENDIX W1:
APPENDIX TO SCHEDULE 3: EQUITABLE SHARE ALLOCATIONS TO MUNICIPALITIES
EQUITABLE SHARE FORMULA ALLOCATIONS + RSC LEVIES REPLACEMENT + SPECIAL CONTRIBUTION TOWARDS COUNCILLOR REMUNERATION

			<u>ର</u>	Equitable Share Formula	Formula				2	RSC Levies Replacement	placement			Spec	ial Contributi	Special Contribution towards Councillor Remuneration	ouncillor Rer	nuneration	
		Nationa	National Financial Year	ı.	Munici	Municipal Financial Year	'ear	Nation	National Financial Year	ear	Municipal	Municipal Financial Year		National I	National Financial Year	_	Municipa	Municipal Financial Year	ear
	N	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14 2	2011/12 20	2012/13 201	//4	2011/12 20	2012/13 20	2013/14	2011/12	2012/13	2013/14
Category Municipanty	9	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000) (	(R'000) (	(R'000) (A	(R'000) (R'0	(R'000) (R'	(R'000) (R	(R'000) (A	(R'000) (	(R'000)	(R'000)	(R'000)
Olwardi. F																			
GAULENG																			
A EKU Ekurhuleni		1 644 128	1 828 391	1 949 038	1 644 128	1 828 391	1 949 038												
A JHB City of Johannesburg	burg	1 897 561	2 134 780	2 276 247	1 897 561	2 134 780	2 276 247												
A TSH City of Tshwane		923 020	1 031 527	1 100 611	923 020	1 031 527	1 100 611												
B GT421 Emfuleni		539 842	597 880	636 986	539 842	597 880	636 986												
B GT422 Midvaal		42 754	47 523	20 666	42 754	47 523	999 09							1 624	1 715	1 815	1 624	1715	1815
B GT423 Lesedi		51 062	56 570	60 274	51 062	56 570	60 274							1 564	1 652	1 747	1 564	1 652	1 747
C DC42 Sedibeng District Municipality	Municipality	19 172	20 322	20 379	19 172	20 322	20 379	201 267	207 305	213 524	201 267	207 305 2	213 524						
Total: Sedibeng Municipalities		652 830	722 295	768 304	652 830	722 295	768 304	201 267	207 305	213 524	201 267	207 305 2	213 524	3 188	3 367	3 562	3 188	3 367	3 562
B GT481 Mogale City		189 605	210 233	224 025	189 605	210 233	224 025												
B GT482 Randfontein		78 991	87 674	93 440	78 991	87 674	93 440							2 647	2 795	2 957	2 647	2 795	2 957
B GT483 Westonaria		85 931	95 094	101 309	85 931	95 094	101 309							1 865	1 969	2 083	1 865	1 969	2 083
B GT484 Merafong City		167 868	185 935	198 124	167 868	185 935	198 124												
C DC48 West Rand District Municipality	ct Municipality	25 474	27 002	26 974	25 474	27 002	26 974	137 747	141 879	146 135	137 747	141 879 1	146 135						
Total: West Rand Municipalities		547 869	605 938	643 872	547 869	605 938	643 872	137 747	141 879	146 135	137 747	141 879 1	146 135	4 512	4 764	5 041	4 512	4 764	5 041
Total: Gauteng Municipalities		5 665 409	6 322 931	6 738 073	5 665 409	6 322 931	6 738 073	339 014	349 184	359 660	339 014	349 184 3	359 660	7 700	8 131	8 603	7 700	8 131	8 603

APPENDIX W1:
APPENDIX TO SCHEDULE 3: EQUITABLE SHARE ALLOCATIONS TO MUNICIPALITIES
EQUITABLE SHARE FORMULA ALLOCATIONS + RSC LEVIES REPLACEMENT + SPECIAL CONTRIBUTION TOWARDS COUNCILLOR REMUNERATION

			E	Equitable Share Formula	Formula				RSC	RSC Levies Replacement	ement			Special Contribution towards Councillor Remuneration	bution toward	s Councillor Re	muneration	
		Natio	National Financial Year	ar	Munici	Municipal Financial Year	ear	National	National Financial Year		Municipal Financial Year	ncial Year	Nati	National Financial Year	Year	Municit	Municipal Financial Year	ear
Cat	Category Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 20 (R'000) (1	2012/13 20 (R'000) (R	2013/14 2011/12 (R'000) (R'000)	2011/12 2012/13 (R'000) (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
KWAZUL	KWAZULU-NATAL																	
٧	ETH eThekwini	1 595 941	1 773 889	1 891 834	1 595 941	1 773 889	1 891 834											
<u>m</u>	KZN211 Vulamehlo	29 943	33 182	35 368	29 943	33 182	35 368						1 203	1 270	1 344	1 203	1 270	1 344
<u> </u>		22 961	25 470	27 146	22 961	25 470	27 146						1 143	1 207	1 277	1 143	1 207	1 277
В	KZN213 Umzumbe	73 523	81 458	86 819	73 523	81 458	86 819						2 675	2 825	2 989	2 675	2 825	2 989
В		37 145	41 161	43 872	37 145	41 161	43 872						1 143	1 207	1 277	1 143	1 207	1 277
В		22 236	24 639	26 261	22 236	24 639	26 261						774	818	865	774	818	865
<u>м</u> С	KZN216 Hibiscus Coast DC21 Hon District Municipality	76 844	214 204	91 051	76 844	214 204	91 051	42, 329	46 138	50 290	42 329 46 138	38 50 290						
Total: Ugu	ΙĒ	456 061	505 530	538 712	456 061	505 530	538 712	42 329	46 138				0 6 939	7 327	7 752	6 6 9 3 9	7 327	7752
м		47 349	52 429	55 870	47 349	52 429	55 870						1 830	1 933	2 045	1 830	1 933	2 045
В		29 176	32 477	34 629	29 176	32 477	34 629						1 384	1 461	1 546	1 384	1 461	1 546
<u>m</u> :		18 992	21 032	22 412	18 992	21 032	22 412						493	520	551	493	520	551
<u> </u>	KZN224 Impendie	20 393	22 602	24 093	20 393	22 602	24 093						694	732	775	694	732	775
цш		33 877	26 444	28 181	23 877	26 444	28 181						980		1011	980	1001	101
<u>q</u> <u>m</u>		24 863	27 532	29 339	24.863	27 532	29 339						986	1 1	1011	986	1 1 2	101
<u>. U</u>		137 524	152 488	162 276	137 524	152 488	162 276	148 495	161 860	176 427	148 495 161 8	60 176 427						
Total: uMg	otal: uMgungundlovu Municipalities	607 010	674 009	718 209	010 209	674 009	718 209	148 495		176 427	148 495 161 860		6 372	6 7 2 9	7 119	6 372	6 7 2 9	7 119
<u>m</u>	KZN232 Emnambethi-Ladvsmith	90.180	99 958	106517	90 180	99 958	106 517						3 188	3 367	3.562	3 188	3 367	3.562
ш		51 594	57 134	60 885	51 594	57 134	60 885						1 408	1 487	1 573	1 408	1 487	1 573
В		24 820	27 550	29 367	24 820	27 550	29 367						1 023	1 080	1 143	1 023	1 080	1 143
В		53 346	960 65	62 983	53 346	29 096	62 983						1 684	1 779	1 882	1 684	1 779	1 882
В		55 507	61 462	65 495	55 507	61 462	65 495						1 760	1 859	1 966	1 760	1 859	1 966
ບ	DC23 Uthukela District Municipality	197 703	218 879	233 170	197 703	218 879	233 170	31 768	34 627									
Total: Uthr	Total:Uthukela Municipalities	473 151	524 079	558 417	473 151	524 079	558 417	31 768	34 627	37 743	31 768 34 627	27 37 743	3 9 063	9 571	10 126	9 063	9 571	10 126
<u>m</u>	KZN241 Endumeni	25 370	28 209	30.083	25 370	28 209	30 083						2 045		2 285	2 045	2 160	2 285
В		60 559	67 134	71 566	60 559	67 134	71 566						2 226	2 350	2 487	2 226	2 350	2 487
В		58 613	64 981	69 272	58 613	64 981	69 272						2 605		2 910	2 605	2 751	2910
М	10	35 462	39 291	41 876	35 462	39 291	41 876								1 479	1 323	1 398	1 479
၁	DC24 Umzinyathi District Municipality	141410	156 656	166 924	141 410	156 656	166 924	19 041	20.754				2					
Total: Um	Fotal: Umzinyathi Municipalities	321 415	356 271	379 721	321 415	356 271	379 721	19 041	20 754	22 622	19 041 20 754	54 22 622	8 199	8 658	9 161	8 199	8 658	9 161
В	KZN252 Newcastle	244 400	270 655	288 355	244 400	270 655	288 355											
В	KZN253 Emadlangeni	11 562	12 816	13 661	11 562	12 816	13 661						694		775	694	732	775
В	-+	40 612	44 983	47 939	40 612	44 983	47 939						1 478	1 561	1 652	1 478	1 561	1 652
ا ا	DC25 Amajuba District Municipality	44 873	49 713	52 899	44 873	49.713	52 899	43 698	47 630						-			
Iotal: Am	Lotat: Amajuba Mumcipanties	341 448	3/8 16/	407 824	341 448	3/8 16/	407 824	43 698	4/ 630	71616	43 698 47 630	30 31 917	7 7 7 7	7 294	7747	7/17	7 734	7747

APPENDIX TO SCHEDULE 3: EQUITABLE SHARE ALLOCATIONS TO MUNICIPALITIES
EQUITABLE SHARE FORMULA ALLOCATIONS + RSC LEVIES REPLACEMENT + SPECIAL CONTRIBUTION TOWARDS COUNCILLOR REMUNERATION

				Equitable Share Formula	2 Formula				RSC	RSC Levies Replacement	ement			Special Contribution towards Councillor Remuneration	bution toward	s Councillor Re	muneration	
		Natio	National Financial Year	ear	Munic	Municipal Financial Year	?ear	National	National Financial Year		Municipal Financial Year	ncial Year	Nati	National Financial Year	Year	Municit	Municipal Financial Year	ear
		2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	2011/12 20	2012/13 20	2013/14 2011/12	7/12 2012/13	3 2013/14	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14
Category Municipality		(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000) (I	(R'000) (R	(R'000) (R'000)		$\dashv$	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)
					;									1				
KZN261		31 211	34 570	36 841	31 211	34 570	36 841						905	953	1 008	905	953	1 008
KZN262		20 987	56 461	60 167	20 987	56 461	60 167						1 624	1715	1815	1 624	1715	1815
B KZN263 Abaqulusi		22 99	73 720	78 558	22 99	73 720	78 558						2 647	2 795	2 957	2 647	2 795	2 957
B KZN265 Nongoma		90 365	66 893	71 300	60 365	66 893	71 300						2 526	2 668	2 823	2 526	2 668	2 823
B KZN266 Ulundi		65 447	72 497	77 261	65 447	72 497	77 261						2 827	2 986	3 159	2 827	2 986	3 159
C DC26 Zululand Distri	Zululand District Municipality	202 202	223 919	238 559	202 202	223 919	238 559	32 124	35 015	38 166	32 124 35 015	15 38 166	16					
Total: Zululand Municipalities		476 790	528 059	562 686	476 790	528 059	562 686	32 124	35 015		Ш		10 527	11 117	11 761	10 527	11 117	11 761
	gana	47 288	52 449	55 921	47 288	52 449	55 921						2 394	2 528	2 674	2 394	2 528	2 674
B KZN272 Jozini		62 081	68 832	73 379	62 081	68 832	73 379						2 816	2 974	3 146	2 816	2 974	3 146
B KZN273 The Big 5 False Bay	e Bay	11 702	12 969	13 824	11 702	12 969	13 824						694	732	775	694	732	775
B KZN274 Hlabisa		37 417	23 783	25 331	37 417	23 783	25 331						1 126	1 190	1 259	1 126	1 190	1 259
B KZN275 Mtubatuba		39 365	43 938	46 937	39 365	43 938	46 937						2 675	2 825	2 989	2 675	2 825	2 989
C DC27 Umkhanyakude	Umkhanyakude District Municipality	140 137	155 258	165 438	140 137	155 258	165 438	19 411	21 158		19 411 21 1	58 23 062						
Total: Umkhanyakude Municipalities		337 991	357 229	380 830	337 991	357 229	380 830	19 411	21 158		19 411 21 158	58 23 062	9 705	10 249	10 843	9 705	10 249	10 843
isologi Medosi		35 910	30.669	47.773	25 910	30 669	47 273						2112	2 230	2360	2 112	2 230	2360
INZIAZ61		20 000	32,000	01774	20000	22 000	0.777						7117	007 7	2 300	7117	0677	7 200
KZN282		161 654	179 231	190 994	161 654	179 231	190 994											
KZN283		16 548	18 325	19 528	16 548	18 325	19 528						1 126	1 190	1 259	1 126	1 190	1 259
KZN284		982	73 524	78 350	66 387	73 524	78 350						3 128	3 303	3 495	3 128	3 303	3 495
B KZN285 Mthonjaneni		21 434	23 743	25 304	21 434	23 743	25 304						662	669	739	999	669	739
B KZN286 Nkandla		42 747	47 385	50 511	42 747	47 385	50 511						1 901	2 007	2 124	1 901	2 007	2 124
C DC28 uThungulu Dist	uThungulu District Municipality	146 964	162 903	173 353	146 964	162 903	173 353	152 214	165 913		152 214 165 913	13 180 845						
Total: uThungulu Municipalities		491 554	544 778	580 313	491 554	544 778	580 313	152 214	165 913		152 214 165 913	13 180 845	8 929	9 429	9266	8 929	9 429	9266
R KZN291 Mandeni		55 012	868 09	44 886	55.012	808 09	988						2 045	091 ¢	2385	2 0.45	091.0	2385
CPCNZX		750 23	72 636	77 471	65 237	72 636	77 471						1			1		
KZN293		48 649	53 857	57 387	48 649	53 857	57 387						2 226	2 350	2 487	2 226	2.350	2 487
KZN294		41.588	46 092	49 131	41.588	46 092	49 131						1 549	1 636	1 730	1 549	1 636	1 730
DC29	t Municipality	158 500	175 514	186 941	158 500	175 514	186 941	48 229	52 570	57 301	48 229 52 5	70 57 301				!		
Total: iLembe Municipalities		368 986	408 996	435 816	368 986	408 996	435 816	48 229	52 570		48 229 52 570	70 57 301	5 820	6 146	6 502	5 820	6 146	6 502
R KZN431 Inowe		43 319	48 004	51.167	43 319	48 004	21 167						1 549	1 636	1 730	1 549	1 636	1 730
VZNA32		0.673	10 00	11,600	0.873	50001	11,600						709	733	277	694	732	277
		11 094	10 663	40.604	11 004	10 000	40 604						160	307	2001	160	201	2001
NZIN433	Dis	41 984	40 243	49 604	41 984	40 243	49 604						796	910 1	0.01	796	1010	0.01
KZN434		42 716	47 328	50 443	42 716	47 328	50 443						1 690	1 784	1 888	1 690	1 784	1 888
		72 596	80 438	85 735	72 596	80 438	85 735						•	2 541	2 688	2 406	2 541	2 688
C DC43 Sisonke District Municipality	et Municipality	164 070	181 662	193 547	164 070	181 662	193 547	17 247	18 799	20 491	17 247 18 799	99 20 491	1 564	1 652	1 747	1 564	1 652	1 747
Total: Sisonke Municipalities		374 508	414 861	442 097	374 508	414 861	442 097	17 247	18 799	20 491	17 247 18 799	99 20 491	598 8	9 361	9 904	8 865	9 361	9 904
Total: KwaZulu-Natal Municinalities		5 844 854	6 465 867	6 891 489	5 844 854	6 465 867	6891 489	554 556	604 464	998 869	554 556 604 464	658 866	165 92	80 880	85 572	16 201	80 880	85 572
			1.22	10.	-		700							-	1	1 1 1 1	1,22,00	

APPENDIX W1:
APPENDIX TO SCHEDULE 3: EQUITABLE SHARE ALLOCATIONS TO MUNICIPALITIES
EQUITABLE SHARE FORMULA ALLOCATIONS + RSC LEVIES REPLACEMENT + SPECIAL CONTRIBUTION TOWARDS COUNCILLOR REMUNERATION

			Equitable Share Formula	e Formula				RSC	RSC Levies Replacement	ment			Special Contribution towards Councillor Remuneration	bution toward	s Councillor R	muneration	
	Nati	National Financial Year	/ear	Muni	Municipal Financial Year	Year	Nationa	National Financial Year		Municipal Financial Year	cial Year	Natio	National Financial Year	Year	Munici	Municipal Financial Year	ear
Category Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 201 (R'000) (R'	2013/14 2011/12 (R'000) (R'000)	12 2012/13 9) (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
LIMPOPO																	
B ITM331 Greater Givani	113 244	125 361		113 244	125 361	133 577						3,609	3 811	4.032	3,609	3.811	4.032
LIM332	114 044	126 248	134 522	_	126 248							3 429	3 621	3 831	3 429	3 621	3 831
LIM333	176 879	195 784		_	195 784												
LIM334	51 586	57 108			57 108							2 166	2 287	2 419	2 166	2 287	2 419
	45 276	50 137	53 428		50 137								2 007	2 124	1 901	2 007	2 124
C DC33 Mopani District Municipality	356 295	394 456	420 198		394 456	420 198	63 423	69 131									
Total: Mopani Municipalities	857 324	949 095	1 011 171	857 324	949 095	1 011 171	63 423	69 131	75 353 6.	63 423 69 131	15 353	11 105	11 726	12 406	11 105	11 726	12 406
n interior	270 20		21 020	2000		00010						40	000	7	970	000	7
LECTION 241	27,003			27 003	29 900							040	760	1 :	C+0 ,	760	1 :
LIM342	35 797			35 797								1 830	1 933	2 045	1 830	1 933	2 045
LIM343	235 608			235 608	260 810												
_	212 830	235 552		212 830													
C DC34 Vhembe District Municipality	389 957			389 957	431 706		41 214	44 923		41 214 44 923							
Total: Vhembe Municipalities	901 256	997 695	1 063 000	901 256	997 695	1 063 000	41 214	44 923	48 966 4	41 214 44 9	3 48 966	2 675	2 825	2 989	2 675	2 825	2 989
B LIM351 Bloubere	76 947	85 305	90 937	76 947	85 305	90 937						2 466	2 604	2755	2 466	2 604	2 755
	64 165	71 063	75 731	64 165	71 063							2 605	2.751	2 910	2 605	2.751	2 910
L.IM353	64 270	71.138	75 798		71.138							1 901	2.007	2 124	1 901	2 007	2 124
1 TM354	350.705	388 419	413.878	350.705	388 419	413.878							1	1		1	1
1 TM355	105 908	117 236	127 919		117 236	124 919						3 730	3 621	3 831	3 420	3 631	3 831
DC35	735 675	261 052	C96 +21	235 675	261.052	616 +21	129 554	141214	153 923	129 554 141 214	153 923		2 021	1000	2459	3 021	1000
ŀĕ	079 768	994 214	1 059 224	897 670	994 214	1 059 224	129 554					10 401	10 983	11 620	10 401	10 983	11 620
											L						
B LIM361 Thabazimbi	51 712		61 057		57 294							1 384	1 461	1 546	1 384	1 461	1 546
B LIM362 Lephalale	71 856		83 216		78 145							1 444	1 525	1 613	1 444	1 525	1 613
B LIM364 Mookgopong	22 118	24 517	26 131		24 517							704	743	787	704	743	787
B LIM365 Modimolle	48 992		57 780		54 233	57 780						1 083	1 143	1 210	1 083	1 143	1 210
B LIM366 Bela Bela	38 768		45 743	38 768	42 932	45 743						1 023	1 080	1 143	1 023	1 080	1 143
B LIM367 Mogalakwena	225 142	249 257	265 591	61	249 257	265 591											
C DC36 Waterberg District Municipality	18 994	20 134	21 019	18 994	20 134	21 019	988 89		73 082 68	68 886 70 953							
Total: Waterberg Municipalities	477 582	526 511	560 538	477 582	526 511	560 538	988 89	70 953		68 886 70 9.	3 73 082	5 637	5 952	6 298	5 637	5 952	6 298
B LIM471 Ephraim Mogale	58 604	64 863			64 863	69 110						1 925	2 033	2 151	1 925	2 033	2 151
I.IM472	110 524	122 314	130 316	110 524	122 314							3 609	3 811	4 032	3 609	3.811	4 032
1.IM473	122 669	135 808			135.808							3 669	3 875	4 100	3 669	3 875	4 100
	38 802	42.981		38 802								1 760	1859	1 966	1 760	1 859	1 966
1 TA 472	10.402	100 251				130 213						2 730	0000	4 162	2 130	0000	1 163
DC47	110 40/ iv 281 448	311 786	332 199			332 199	49 429	53.878		49 429 53 878	58 727		00000	4 10/	06/ 6	2 930	10/
5		800 033			800 033	957 459	40.420		26 777			14 603	15 516	16.416	14 603	15 516	16.416
Total Overes Overmente Parine		2000					i i										
Total: Limnono Municinalities	3 856 286	4 267 547	4 546 391	3856 286	4 267 547	4 546 391	352.506	380 099	410 051	352 506 380 099	410 051	44 511	47 003	49 730	44.511	47 003	49 730

APPENDIX W1:
APPENDIX TO SCHEDULE 3: EQUITABLE SHARE ALLOCATIONS TO MUNICIPALITIES
EQUITABLE SHARE FORMULA ALLOCATIONS + RSC LEVIES REPLACEMENT + SPECIAL CONTRIBUTION TOWARDS COUNCILLOR REMUNERATION

			Equitable Share Formula	Formula				RSC	RSC Levies Replacement	ement			Special Contri	Special Contribution towards Councillor Remuneration	s Councillor Re	muneration	
	Natio	National Financial Year	ear	Munic	Municipal Financial Year	/ear	National	National Financial Year		Municipal Financial Year	ancial Year	Nat	National Financial Year	Year	Municip	Municipal Financial Year	ear
Category Municipality	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	_	_	,		_	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14
	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000) (	(R'000) (R'	(R'000) (R'000)	00) (R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)
MPUMALANGA																	
	138 333	153 170	163 217	138 333	153 170	163 217						2 948	3 113	3 293	2 948	3 113	3 293
B MP302 Msukaligwa	90 826	100 586	107 169	90 826	100 586	107 169						2 286	2 414	2 554	2 286	2 4 1 4	2 554
B MP303 Mkhondo	86 446	95 747	102 036	86 446	95 747	102 036						2 286	2 414	2 554	2 286	2 414	2 554
B MP304 Pixley Ka Seme	73 712	81 573	86 907	73 712	81 573	86 907						1 263	1 334	1 411	1 263	1 334	1411
B MP305 Lekwa	68 155	75 458	80 395	68 155	75 458	80 395						1 805	1 906	2 0 1 6	1 805	1 906	2 0 1 6
B MP306 Dipaleseng	38 474	42 573	45 355	38 474	42 573	45 355						845	892	944	845	892	944
B MP307 Govan Mbeki	171 429	190 045	202 507	171 429	190 045	202 507											
C DC30 Gert Sibande District Municipality	17 634	18 693	19 053	17 634	18 693	19 053	228 648				235 507 242 572	.2					
Total: Gert Sibande Municipalities	682 030	757 844	806 639	685 039	757 844	806 639	228 648	235 507 2	242 572 2	228 648 235	235 507 242 572	2 11 432	12 072	12 772	11 432	12 072	12 772
B MP311 Victor Khanve	44 055	48 787	51 982	44 055	48 787	51 982						1 023	1 080	1 143	1 023	1 080	1 143
	163 854	181 726	193 663	163 854	181 726	193 663											
B MP313 Steve Tshwete	77 312	85 857	91 509	77 312	85 857	91 509											
B MP314 Emakhazeni	30 506	33 770	35 979	30 506	33 770	35 979						1 056	1 115	1 180	1 056	11115	1 180
B MP315 Thembisile	192 816	213 366	227 323	192 816	213 366	227 323						3 850	4 065	4 301	3 850	4 065	4 301
B MP316 Dr JS Moroka	201 153	222 597	237 159	201 153	222 597	237 159						4 365	4 609	4 877	4 365	4 609	4 877
C DC31 Nkangala District Municipality	19 468	20 636	20 883	19 468	20 636	20 883	272 506		289 101 2		280 681 289 101	11					
Total: Nkangala Municipalities	729 162	806 739	858 499	729 162	806 739	858 499	272 506	280 681 2	289 101 2	272 506 280 681	681 289 101	10 294	10 870	11 500	10 294	10 870	11 500
B MP321 Thaba Chweu	65 071	72 035	76748	65 071	72 035	76 748						1 624	1 715	1 815	1 624	1 715	1 815
B MP322 Mbombela	282 081	312 447	332 931	282 081	312 447	332 931											
	42 081	46 614	49 674	42 081	46 614	49 674						1 083	1 143	1 210	1 083	1 143	1 210
B MP324 Nkomazi	234 566	259 743	276 789	234 566	259 743	276 789											
	394 039	436 207	464 797	394 039	436 207	464 797						4 451	4 701	4 973	4 451	4 701	4 973
C DC32 Ehlanzeni District Municipality	35 267	39 577	42 440	35 267	39 577	42 440	135 147					7					
Total: Ehlanzeni Municipalities	1 053 106	1 166 624	1 243 380	1 053 106	1 166 624	1 243 380	135 147	139 201	143 377 1.	135 147 139	139 201 143 377	7 7 158	7 559	2 998	7 158	7 559	7 998
Total: Mpumalanga Municipalities	2 467 307	2 731 207	2 908 517	2 467 307	2 731 207	2 908 517	636 301	9 682 389	675 051 6	636 301 655	655 389 675 051	1 28 884	30 501	32 271	28 884	30 501	32 271
																	1

APPENDIX W1:
APPENDIX TO SCHEDULE 3: EQUITABLE SHARE ALLOCATIONS TO MUNICIPALITIES
EQUITABLE SHARE FORMULA ALLOCATIONS + RSC LEVIES REPLACEMENT + SPECIAL CONTRIBUTION TOWARDS COUNCILLOR REMUNERATION

			Equitable Share Formul	re Formula				R	RSC Levies Replacement	placement			Š	ecial Contrib	ution towards	Special Contribution towards Councillor Remuneration	muneration	
	Nati	National Financial Year	/ear	Muni	Municipal Financial Year	Year	Nations	National Financial Year	ear	Municipa	Municipal Financial Year	ear	Nation	National Financial Year	ear	Municit	Municipal Financial Year	ear
	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	2011/12	2012/13	3/14	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14
Category Municipality	(R000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	$\dashv$		-	$\dashv$	$\dashv$	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)
NORTHERN CAPE																		
	1			4									1		•	1	1	•
	9 852			. د	10 913	11 628							505	666	679	505	666	679
NC062	2/ 428		32 323		30 345								1.197	1 264	1337	197	1 264	1337
NC064	10213			2 !	11 308								694	/32	5//	694	/32	5//
	17 049				188/4								634	699	108	634	699	80/
NC066	10 850	12 007	12 793	2	12 007								694	732	775	694	732	775
_	10 227			2	11 321								694	732	775	694	732	775
C DC6 Namakwa District Municipality	5 396		6 513	5	6 068	6513	23 494	24 198	24 924	23 494	24 198	24 924	902	953	1 008	902	953	1 008
Total: Namakwa Municipalities	91 016	100 836	107 478	91016	100 836	107 478	23 494	24 198	24 924	23 494	24 198	24 924	5 377	5 678	2009	5 377	2 678	6 007
NC071	15 106				16 718								263	265	629	263	265	629
B NC072 Umsobomvu	25 391			25	28 092								704	743	787	704	743	787
B NC073 Emthanjeni	29 454		34 731	29 454	32 601	34 731							986	1 041	1 101	986	1 041	1 101
B NC074 Kareeberg	9 973	11 044		6	11 044	11 768							493	520	551	493	520	551
B NC075 Renosterberg	12 219			12	13 525	14 410							694	732	775	694	732	775
B NC076 Thembelihle	12 155			12	13 462	14 345							694	732	775	694	732	775
	17 266			17	19 120								563	595	629	563	595	629
NC078	30 952			30	34 289	36 543							774	818	865	774	818	865
DC7			9 251	7	8 653		15 697	16 167	16 652	15 697	16 167	16 652	1 267	1 338	1416	1 267	1338	1416
e N	1	1	189 161	160	177 504	189 161	15 697	16 167	16 652	15 697	16 167	16 652	6 738	7 115	7 528	6 738	7 115	7 528
B NC081 Mier	7 926		9 365	7 926	8 786	9 365							694	732	775	694	732	775
NC082	40 533			40	44 871								1 023	1 080	1 143	1 023	1 080	1 143
NC083	44 497			4	49 355								1 624	1 715	1815	1 624	1 715	1815
NC084	13 262		15 647	13	14 684								694	732	775	694	732	775
NC085	21 680		25 608	21	24 026								774	818	865	774	818	865
	12 355		14 565	12	13 671	14 565							263	262	629	263	262	629
C DC8 Siyanda District Municipality	7 892		9 363		8 731	9 3 6 3	30 663	31 582	32 529	30 663	31 582	32 529	1 263	1 334	1411	1 263	1 334	1411
Total: Siyanda Municipalities	148 145	164 124	174 958	148 145	164 124	174 958	30 663	31 582	32 529	30 663	31 582	32 529	6 635	2 006	7 413	6 635	2 000	7 413
B NC091 Sol Plantie	132 176	146 577	156 193		146 577	156 193												
NC092	39 131			39	43 318								915	196	1 023	915	196	1 023
NC093	24 912			24	27 569								634	699	708	634	699	708
B NC094 Phokwane	55 636				61 563								1 083	1 143	1 210	1 083	1 143	1 210
DC9				10	10 613		67 645	73 733	75 945	67 645	73 733	75 945	1 624	1 715	1815	1 624	1 715	1815
Total: Frances Baard Municipalities			63	261	289 641		67 645	73 733	75 945	67 645	73 733	75 945	4 256	4 494	4755	4 256	4 4 4 9 4	4755
					101 02									ì				
	99/ 19				68 481								2 042	7 136	7.781	7 047	2 150	7.781
NC452	56715			56	62 820								1 504	1 588	1 680	1 504	1 588	1 680
		19 470		17	19 470	20 745							704	743	787	704	743	787
C DC45 John Taolo Gaetsewe District Municipality				19	22 166		29 864	30 759	31 682	29 864	30 759	31 682	1 203	1 270	1 344	1 203	1 270	1344
Total: John Taolo Gaetsewe Municipalities	155 931	172 938	184 392		172 938	184 392	29 864	30 759	31 682	29 864	30 759	31 682	5 453	5 758	6 092	5 453	5 758	6 092
Total: Northern Cane Municipalities	817 238	905 042	964 533	817 238	905 042	964 533	167 363	176 439	181 732	167 363	176 439	181 732	28 458	30 052	31 795	28 458	30 052	31 795

APPENDIX W1:
APPENDIX TO SCHEDULE 3: EQUITABLE SHARE ALLOCATIONS TO MUNICIPALITIES
EQUITABLE SHARE FORMULA ALLOCATIONS + RSC LEVIES REPLACEMENT + SPECIAL CONTRIBUTION TOWARDS COUNCILLOR REMUNERATION

			Equitable Share Formula	Formula				RSC	RSC Levies Replacement	ment			Special Contri	Special Contribution towards Councillor Remuneration	S Councillor Re	muneration	
	Natio	National Financial Year	ear	Munici	Municipal Financial Year	ear	National F	National Financial Year		Municipal Financial Year	incial Year	Nat	National Financial Year	Year	Municip	Municipal Financial Year	ear
	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14	2011/12 201	2012/13 201	2013/14 2011/12	712 2012/13	3 2013/14	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14
Category Municipality	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)				(R'000) (R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)		(R'000)
NORTH WEST																	
B NW371 Moretele	134 410	148 827	158 592	134 410	148 827	158 592						3 872	4 089	4 326	3 872	4 089	4 326
B NW372 Madibeng	247 326	273 870	291 812	247 326	273 870	291 812											
B NW373 Rustenburg	231 669	256 655	273 476	231 669	256 655	273 476											
B NW374 Kgetlengrivier	37 187	41 165	43 862	37 187	41 165	43 862						845	892	944	845	892	944
B NW375 Moses Kotane	200 027	221 367	235 850	200 027	221 367	235 850						3 730	3 938	4 167	3 730	3 938	4 167
C DC37 Bojanala Platinum District Municipality	30 973	32 832	34 457	30 973	32 832	34 457	201 121 2	207 155 2	213 370 20	201 121 207	207 155 213 370	0.					
Total: Bojanala Platinum Municipalities	881 592	974 714	1 038 049	881 592	974 714	1 038 049		207 155 2	213 370 20	201 121 207	207 155 213 370	0 8 447	8 920	9 437	8 447	8 920	9 437
NW381	27 605	167 59	61.619		63.791	67.679						1 971		2 202	1 971	2 082	2 202
NW382	54 381	60 203	64 148	54 381	60 203	64 148						1 805	1 906	2 0 1 6	1 805	1 906	2 0 1 6
B NW383 Mafikeng	109 725	121 500	129 460	109 725	121 500	129 460											
B NW384 Ditsobotla	67 169	74 331	79 195	67 169	74 331	79 195						2 526		2 823	2 526	2 668	2 823
B NW385 Ramotshere Moiloa	68 112	75 405	80 347	68 112	75 405	80 347						2 346		2 621	2 346	2 477	2 621
C DC38 Ngaka Modiri Molema District Municipality	254 586	281 848	300 126	254 586	281 848	300 126	103 029	112 302 1	122 409 10	103 029 112	112 302 122 409	6					
Total: Ngaka Modiri Molema Municipalities	611 578	810 119	721 256	611 578	810 119	721 256	103 029	112 302	122 409 10	103 029 112	112 302 122 409	9 8 648	9 133	9 662	8 648	9 133	9 662
R NW307 Noladi	38.036	31 061	33.006	28.036	31.061	33 006						1.083	1 1/3	1 210	1 083	1 1 1 3 3	1 210
	27 154	30.057	32 025		30.057	32 025						1 056		081	1 056	1115	1 180
NW394	85 042	94 190	100380	85 042	94 190	100 380						3 591		4 012	3.591	3 792	4 012
	24 003	26 572	28 312	24 003	26 572	28 312						986		1 101	986	1 041	1 101
B NW397 NW397	56392	62 409	66 493	56 392	62 409	66 493						2 112	2 230	2 360	2 112	2 230	2 360
C DC39 Dr Ruth Segomotsi Mompati District Municipality	163 012	180 422	149 358	163 012	180 422	149 358	23 591	25 714	28 028 2	23 591 25	25 714 28 028			1 949	1 744	1 842	1 949
Total: Dr Ruth Segomotsi Mompati Municipalities	383 639	424 710	409 663	383 639	424 710	409 663	23 591	25 714	28 028 2	23 591 25	25 714 28 028	8 10 572	11 164	11 811	10 572	11 164	11811
B NW401 Ventersdorn	40 106	44 395	47 303	40 106	44 395	47 303						845	897	944	845	68	944
NW402	76 801	85 191	90 788		85 191	90 788									!		
NW403	303 560	336 019	357 961	303 560	336 019	357 961											
NW404	966 19	75 284	80 220	966 29	75 284	80 220						1 263	1 334	1 411	1 263	1 334	1 411
C DC40 Dr Kenneth Kaunda District Municipality	23 596	25 012	17 956		25 012	17 956	130 026	133 927	137 945 13	130 026 133	133 927 137 945						
Total: Dr Kenneth Kaunda Municipalities	512 060	565 901	594 228	512 060	565 901	594 228	130 026	133 927 1	137 945 13	130 026 133	133 927 137 945	5 2 108	2 226	2 355	2 108	2 226	2 355
Total: North West Municipalities	0 388 860	2,642,404	2 763 196	2 388 869	2 642 404	2.763.196	457 767	479 008	501 752 45	257 767	501 752	277 90	31 442	33 266	277 90	31 442	33,756

APPENDIX W1:
APPENDIX TO SCHEDULE 3: EQUITABLE SHARE ALLOCATIONS TO MUNICIPALITIES
EQUITABLE SHARE FORMULA ALLOCATIONS + RSC LEVIES REPLACEMENT + SPECIAL CONTRIBUTION TOWARDS COUNCILLOR REMUNERATION

			Equitable Share Formula	Formula				RSC	RSC Levies Replacement	ment			Special Contribution towards Councillor Remuneration	bution toward	s Councillor R	emuneration	
	Natio	National Financial Year	ar	Munic	Municipal Financial Year	ear	National	ial		멾	ncial Year	Natic	National Financial Year	Year	Munici	Municipal Financial Year	/ear
Category Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 2 (R'000) (	2012/13 20. (R'000) (R'	2013/14 2011/12 (R'000) (R'000)	712 2012/13 0) (R'000)	R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
WESTERN CAPE																	
A CPT City of Cape Town	970 473	1 090 987	1 163 859	970 473	1 090 987	1 163 859											
B WC011 Matzikama	31 164	34 542	36 813	31 164	34 542	36 813						902	953	1 008	902	953	1 008
B WC012 Cederberg	22 233	24 629	26 241	22 233	24 629	26 241						774	818	865	774	818	865
WC013	18 983	21 044	22 422	18 983	21 044	22 422						915	296	1 023	915	296	1 023
WC014	27 792	30 987	33 044	27 792	30 987	33 044						1 504	1 588	1 680	1 504	1 588	1 680
9	21 503	23 930	25 512	21 503	23 930	25 512						1 384	1 461	1 546	1 384	1 461	1 546
C DCI West Coast District Municipality	9 356	9 9 18	9 245	9 356	9 9 18	9 245	59 296	61 074	62 906 5	59 296 61 074	74 62 906	027	701.3	(11)	02.70	701.3	(11)
Lotal: West Coast Municipalities	7C0 ICI	160 641	155 2/0	700 101	160 641	0/7 CCI	29.790						90/6	7710	0419	90/0	7710
B WC022 Witzenberg	39 177	43 394	46 235	39 177	43 394	46 235						1 384	1 461	1 546	1 384	1 461	1 546
	62 476	69 437	74 007	62 476	69 437	74 007											
WC024	36 977	41 252	43 992	36 977	41 252	43 992											
	56 804	63 024	67 160	56 804	63 024	67 160											
2	43 789	48 482	51 651	43 789	48 482	51 651	000					1 384	1 461	1 546	1 384	1 461	1 546
C DC2 Cape Winelands District Municipality	12 461	13 209	10 664	12 461	13 209	10 664	188 278										
Total: Cape Winelands Municipalities	251 685	278 798	293 710	251 685	278 798	293 710	188 278	193 926	199 744 18	188 278 193 926	26 199 744	2 767	2 922	3 092	2 767	2 922	3 092
B WC031 Theewaterskloof	45 431	50 304	53 592	45 431	50 304	53 592						1 504	1 588	1 680	1 504	1 588	1 680
	29 652	33 078	35 277	29 652	33 078	35 277						1 504	1 588	1 680	1 504	1 588	1 680
WC033	14 172	15 733	16 769	14 172	15 733	16 769						634	699	708	634	699	708
4	16 014	17 756	18 921	16 014	17 756	18 921						634	699	708	634	699	708
C DC3 Overberg District Municipality	8 495	500 6	5006	8 495	500 6	5/06	31 933						1 534	1411	1 203	1 534	1411
Total: Overberg Municipalities	113 765	125 877	133 633	113 765	125 877	133 633	31 933	32 890	33 877 3	31 933 32 890	90 33 877	5 538	5 848	9819	5 538	5 848	6 188
B WC041 Kannaland	16 347	18 099	19 282	16347	18 099	19 282						493	520	551	493	520	551
B WC042 Hessequa	22 171	24 634	26 258	22 171	24 634	26 258						1 056	1 115	1 180	1 056	1115	1 180
WC043	35 318	39 199	41 774	35 318	39 199	41 774						1 624	1 715	1815	1 624	1715	1815
B WC044 George	72 201	80 365	85 706	72 201	80 365	85 706						100	001	900	100	000	90
	19 500	21 729	73 172	10 500	21 729	23 172						1 304	1 388	1 080	1 304	1 388	1 003
WC048	25 544	28 519	30 418	25 544	28 519	30 418						1 143	1 207	1 277	1 143	1 207	1 277
	11 834	12 544	11 858	11 834	12 544	11 858	111 078	114 410	117 842	111 078 114 410	10 117 842						
Total: Eden Municipalities	239 029	265 079	281 070	239 029	265 079	281 070	111 078	114410		111 078 114 410		6 735	7 112	7 525	6 735	7 112	7 525
B WC051 Laingsburg	7 546	8 365	8 915	7 546	8 365	8 915						694	732	775	694	732	775
	8 506	9 417	10 034	8 506	9 417	10 034						694	732	775	694	732	775
WC053	28 041	31 115	33 172	28 041	311115	33 172							296	1 023	915	196	1 023
C DC5 Central Karoo District Municipality	4 979	5 528	5 899	4 979	5 528	5 899	5 641	5 810	5 984	5 641 5 8	5 810 5 984		196	1 023	915	196	1 023
Total: Central Karoo Municipalities	49 072	54 424	58 019	49 072	54 424	58 019	5 641	5 810	5 984	5 641 5 8	5 810 5 984	3 218	3 398	3 295	3 218	3 398	3 595
Total: Western Cape Municipalities	1 755 056	1 960 216	2 083 570	1 755 056	1 960 216	2 083 570	396 226	408 110	420 353 39	396 226 408 110	10 420 353	23 738	25 067	26 521	23 738	25 067	26 521
Unallocated:																	
National Total	009 266 02	33 481 379	35 649 994	30 223 699	33 481 379	35 649 994	3 544 137	3 732 908	1 930 357	3 544 137 3 732 908	3 9 3 9 3 5 7	340 065	359 100	279 937	340 065	359 109	379 937
L'ALL'OLIGI L'OLGI		1212 101 20	20 040	No man Oct.									1004 000		200 010	1 704 700	

## APPENDIX W2: APPENDIX W2: APPENDIX TO SCHEDULE 3: EQUITABLE SHARE ALLOCATIONS TO MUNICIPALITIES

# (BREAKDOWN OF EQUITABLE SHARE ALLOCATIONS PER LOCAL MUNICIPALITY PER SERVICE FOR DISTRICT

MUNICIPALITIES AUTHORISED FOR SERVICES)

(National and Municipal Financial Year)

APPENDIX W2:
APPENDIX TO SCHEDULE 3: EQUITABLE SHARE ALLOCATIONS TO MUNICIPALITIES
BREAKDOWN OF EQUITABLE SHARE ALLOCATIONS PER LOCAL MUNICIPALITY PER SERVICE FOR DISTRICT MUNICIPALITIES AUTHORISED FOR SERVICES

		BF	REAKDOWN OF	EQUITABLE	SHARE FOR I	BREAKDOWN OF EQUITABLE SHARE FOR DISTRICT MUNICIPALITIES AUTHORISED FOR SERVICES	PALITIES AUT	THORISED F	OR SERVICES	
						National/M	National/Municipal Financial Year	al Year		
Number	Municipality	Water	Sanitation	Refuse	Water	Sanitation	Refuse	Water	Sanitation	Refuse
		20	2011/12 (R'000)		=	2012/13 (R'000)		-	2013/14 (R'000)	
EASTERN CAPE										
BITE	Buffelo City									
A NMA	Danian Cary Nelson Mandela Bay									
B FC101	Camdahoo									
	Div Canada									
	Dide Ciane Noute									
B EC104	Makana									
B EC106										
B EC108	Kouga									
B EC109	Kou-kamma									
C DC10	Cacadu District Municipality									
Total: Cacadu Municipalities	alities									
	,									
B EC121		31 854	32 285		35 285	35 761		37 564	38 072	
		43 948	38 9 / 1		48 681	43 168		51 826	156 54	
	Great Kei	8 832	7 059		9 783	7 819		10415	8 324	
		24 777	19 854		27 445	21 992		29 218	23 412	
		17 312	12 466		19 176	13 808		20 415	14 700	
		23 851	18 345		26 419	20 321		28 126	21 634	
B EC128		5 691	5 495		6 304	980 9		6 711	6 4 1 9	
C DC12	Amathole District Municipality									
Total: Amathole Municipalities	cipalities	156 264	134 474		173 092	148 955		184 274	158 578	
200	T		770			0000		00	2	
	mynoa rememba	12 6/1	12 346		14 249	13 006		13 180	14 360	
D EC132	Isolwana	0.132	500 \$		00/00	5 564		7 232	5 029	
	Inkwalica Lukhanii	35 828	195 66		30,665	32 777		42 255	34 864	
	Intsika Yethu	24 947	20 835		27 618	23 066		29 422	24 572	
	_	19 870	16 456		21 998	18 219		23 434	19 408	
B EC137	Engcobo	20 671	19 838		22 885	21 962		24 379	23 396	
B EC138		11 267	9 854		12 474	10 910		13 288	11 622	
C DC13	Chris Hani District Municipality									
Total: Chris Hani Municipalities	icipalities	136 755	118 688		151 401	131 400		161 287	139 979	
B EC141	Elundini	22.748	21 913		25 190	24 265		26 838	25.854	
B FC147	Senon	26 120	20 02		28 924	22 22		30.817	23 692	
		6 932	5 735		7 676	6 351		8 178	6 767	
		7 783	7 710		8 619	8 537		9 183	960 6	
C DC14	Ukhahlamba District Municipality									
Total: Ukhahlamba Municipalities	nnicipalities	63 582	55 439		70 408	61 390		75 017	65 409	

APPENDIX W2:
APPENDIX TO SCHEDULE 3: EQUITABLE SHARE ALLOCATIONS TO MUNICIPALITIES
BREAKDOWN OF EQUITABLE SHARE ALLOCATIONS PER LOCAL MUNICIPALITY PER SERVICE FOR DISTRICT MUNICIPALITIES AUTHORISED FOR SERVICES

		BREAKDOW	N OF EQUITABL	E SHARE FOR	BREAKDOWN OF EQUITABLE SHARE FOR DISTRICT MUNICIPALITIES AUTHORISED FOR SERVICES	TPALITIES A	UTHORISED F	OR SERVICES	
					National/M	National/Municipal Financial Year	icial Year		
Number Municipality	Water	er Sanitation	Refuse	Water	Sanitation	Refuse	Water	Sanitation	Refuse
		2011/12 (R'000)	(6		2012/13 (R'000)		· •	2013/14 (R'000)	
B EC153 Ngquza Hill	33	32 654 30 5	30 507	35 299	32 978		37 579	35 108	
	18		17 980	19 733	19 437		21 008	20 692	
		33 680 32 6	32 668	36 408	35 315		38 759	37 595	
B EC156 Millontlo		28 305 25 6	25 635	30 598	27 711		32 574	29 501	
		54 732 53 7	53 752	59 165	58 106		62 986	61 859	
	rict Municipality								
Total: O.R.Tambo Municipalities	191	167 626 160 542	542	181 204	173 547		192 907	184 755	
B EC441 Matatiele	35	35 041 29 7	29 710	39 011	33 076		41 635	35 301	
B EC442 Umzimvubu	30	30 758 28 1	28 129	34 242	31 316		36 545	33 422	
B EC443 Mbizana	25	29 731 28 5	28 547	32 140	30 860		34 215	32 853	
B EC152 Ntabankulu	16	16 910 18 (	18 049	18 280	119 511		19 460	20 77 1	
C DC44 Alfred Nzo District Municipality	ict Municipality								
Total: Alfred Nzo Municipalities	112	112 440 104 436	436	123 673	114 763		131 856	122 347	
Total: Eastern Cape Municipalities	969	636 667 573 579	625	622 669	630 055		745 340	671 068	

APPENDIX W2:
APPENDIX TO SCHEDULE 3: EQUITABLE SHARE ALLOCATIONS TO MUNICIPALITIES
BREAKDOWN OF EQUITABLE SHARE ALLOCATIONS PER LOCAL MUNICIPALITY PER SERVICE FOR DISTRICT MUNICIPALITIES AUTHORISED FOR SERVICES

			BREAKDOWN	)F EQUITABLE	SHARE FOR	BREAKDOWN OF EQUITABLE SHARE FOR DISTRICT MUNICIPALITIES AUTHORISED FOR SERVICES	PALITIES	UTHORISED	FOR SERVICES	
				,		National/M	National/Municipal Financial Year	cial Year		
Number	Municipality	Water	Sanitation	Refuse	Water	Sanitation	Refuse	Water	Sanitation	Refuse
			2011/12 (R'000)			2012/13 (R'000)			2013/14 (R'000)	
FREE STATE										
A MAN	Mangaung									
B FS161 B FS162 B FS163 B FS164 C DC16	Letsemeng Kopanong Mohokare Naledi Xharriep District Municipality									
Total: Xhariep Municipalities	palities									
B FS181 B FS182 B FS183 B FS184 B FS184 C DC18	Masilonyana Tokologo Tswelopele Matjhabeng Nala Nala Leiweleputswa District Municipality									
Total: Lejweleputswa Municipalities	Municipalities									
B FS191 B FS192 B FS193 B FS194 B FS195 C DC19	Setsoto Dihlabeng Nketoana Maluti a Phofung Phumelial Mantsopa Thabo Mofutsanyana District Municipality									
Total: Thabo Mofutsanyana Municipalities	ıyana Municipalities									
B FS201 B FS203 B FS204 B FS205 C DC20	Moqhaka Ngwathe Metsimaholo Mariube Fezile Dabi District Municipality									
Total: Fezile Dabi Municipalities	nicipalities									
Total: Free State Municipalities	icipalities									

APPENDIX W2:
APPENDIX TO SCHEDULE 3: EQUITABLE SHARE ALLOCATIONS TO MUNICIPALITIES

			BREAKDOWN	JE EQUITABL	E SHARE FOR	BREAKDOWN OF EQUITABLE SHARE FOR DISTRICT MUNICIPALITIES AUTHORISED FOR SERVICES	CIPALITIES /	AUTHORISED	FOR SERVICES	
						National	National/Municipal Financial Year	ncial Year		
Number	Municipality	Water	Sanitation	Refuse	Water	Sanitation	Refuse	Water	Sanitation	Refuse
			2011/12 (R'000)	-		2012/13 (R'000)			2013/14 (R'000)	
GAUTENG										
A EKU A JHB A TSH	Ekurhuleni City of Johannesburg City of Tshwane									
B GT421										
B GT422	2 Midvaal									
B GT423	3 Lesedi Sedibene District Municipality									
Total: Sedibeng Municipalities	icipalities									
B GT481 B GT482	1 Mogale City 2 Randfontein									
B GT483	3 Westonaria									
C DC48										
Total: West Rand Municipalities	micipalities									
Total: Gauteng Municipalities	cipalities									

APPENDIX W2:
APPENDIX TO SCHEDULE 3: EQUITABLE SHARE ALLOCATIONS TO MUNICIPALITIES
BREAKDOWN OF EQUITABLE SHARE ALLOCATIONS PER LOCAL MUNICIPALITY PER SERVICE FOR DISTRICT MUNICIPALITIES AUTHORISED FOR SERVICES

			BREAKDOWN OI	F EQUITABLE	SHARE FOR	BREAKDOWN OF EQUITABLE SHARE FOR DISTRICT MUNICIPALITIES AUTHORISED FOR SERVICES	ALITIES AUTHOR	(ISED FO	R SERVICES	
		•	-	•	-	National/Mur	National/Municipal Financial Year	r	-	
Number	Municipality	Water	Samitation 2011/12 (R'000)	Refuse	Water	Sanitation 2012/13 (R'000)	Refuse Water		Sanitation 2013/14 (R'000)	Refuse
KWAZULU-NATAL										
A ETH e	eThekwini									
KZN211	Vulamehlo	9 445	9 7 65		10 460	10 815		11 143	11 521	
	uMdoni	7 818	9 538		8 659	10 564		9 225	11 254	
KZN213	Umzumbe	23 103	22 880		25 587	25 340	- 2	27 258	26 995	
KZN214	uMuziwabantu	13 096	11 415		14 504	12 643		15 451	13 469	
KZN215	Ezinqolweni	06 2 90	6 701		7 299	7 421		7 7 7 6	2 906	
B KZN216 F C DC21 L	Hibiscus Coast Ugu District Municipality	26 945	24 621		29 842	27 269		31 791	29 050	
Total: Ugu Municipalities	, ,	966 98	84 921		96 351	94 052	10	102 643	100 194	
. ICCNEA	Mahoth:	17 524	130 C1		10.442	036 11		009.00	200 31	
CCNZA	uMagani	17 042	10,488		13 350	11 629		14 200	202 CI 202 CI	
KZN223	Moofana	6 883	6 2 5 9		7 632	6 940		8 122	7 385	
KZN224	Impendie	7 662	6 807		8 495	7 548		9 040	8 032	
KZN225	Msunduzi									
KZN226	Mkhambathini	8 7 5 8	6 778		9 711	7 516		10 334	7 998	
7	Richmond	7 523			8 342	7 921		8 877	8 429	
C DC22 u	uMgungundlovu District Municipality									
Total: uMgungundlovu Municipalities	micipalities	60 402	50 426		66 974	55 913	7	71 273	59 502	
B KZN232 E	Emnambethi-Ladysmith	34 001	29 108		37 643	32 226	4	40 101	34 330	
KZN233	Indaka	20 384	14 816		22 568	16 403	2	24 041	17 474	
KZN234	Umtshezi	8 730	7 913		6 665	8 760		10 296	9 332	
KZN235	Okhahlamba	18 243	15 128		20 197	16 748	2	21 516	17 842	
B KZN236 I C DC23 I	Imbabazane Uthukela District Municipality	16268	13 462		18 010	14 904		19 186	15 877	
otal:Uthukela Municipa	ties	97 627	80 427		108 084	89 041	111	115 141	94 855	
B K7N741	Brdinneni	9 073	10 314		10 01	11 425		002.01	12 174	
CVCNZX	Nouthu	27.07.	701 61		75 307	21.167		72.02	22 554	
KZN242 KZN244	Meinea	19 317	19 189		23 392	21 16/		27 802	22 334	
KZN245	Umvoti	12 385	13 672		13 721	15 146		14 620	16139	
DC24	Umzinyathi District Municipality								)	
Total: Umzinyathi Municipalities	palities	63 692	62 282		70 562	266 89	7	75 187	73 519	
KZN252	Newcastle									
KZN253	Emadlangeni	3 575	3 598		3 961	3 986	_	4 215	4 241	
B KZN254 I	Dannhauser A mainta Dietrict Municipality	12 364			13 698	12 369		14 575	13 162	
otal: Amajuba Municipa	lities	15 939	14 763		17 658	16 355		18 790	17 403	

APPENDIX W2:
APPENDIX TO SCHEDULE 3: EQUITABLE SHARE ALLOCATIONS TO MUNICIPALITIES
BREAKDOWN OF EQUITABLE SHARE ALLOCATIONS PER LOCAL MUNICIPALITY PER SERVICE FOR DISTRICT MUNICIPALITIES AUTHORISED FOR SERVICES

				BREAKDOWN O	F EQUITABLI	SHARE FOR	BREAKDOWN OF EQUITABLE SHARE FOR DISTRICT MUNICIPALITIES AUTHORISED FOR SERVICES	TPALITIES A	UTHORISED	FOR SERVICES	
			•	•		•	National/M	National/Municipal Financial Year	cial Year		
Number		Municipality	Water	Sanitation	Refuse	Water	Sanitation	Refuse	Water	Sanitation	Refuse
				2011/12 (R'000)			2012/13 (R'000)			2013/14 (R'000)	
22	KZN261	eDimbe	11 800	9 166		13.068	10 150		13 922	10.814	
n <u>m</u>	KZN262	OlomodII	18 488	14 151		20 473	15 671		21 812	16 695	
В	KZN263		24 609	23 184		27 252	25 674		29 034	27 353	
В	KZN265		18 852	18 839		20 876	20 862		22 241	22 226	
В	KZN266		21 558	20 433		23 873	22 627		25 434	24 107	
C	DC26	Zululand District Municipality									
Total: Zululand Municipalities	and Munici	palities	95 306	85 772		105 543	94 984		112 443	101 194	
<u></u>	PONTY	The bloker of he can	16 500	16 433		10 202	200 01		10.500	0000	
n m	KZNZZZ	Jozini Iozini	73 537	20.858		26.017	23 109		27 786	24 624	
<u> </u>	KZN273	The Big 5 False Bay	3 611	3 885		4 000	4 304		4 262	4 587	
п	KZN274	Habisa	6 492	6 982		7 193	7 735		7 664	8 242	
В	KZN275	Mtubatuba	12 809	12 223		14 191	13 542		15 122	14 430	
C	DC27	Umkhanyakude District Municipality									
Total: Umkh	nanyakude 1	Total: Umkhanyakude Municipalities	63 041	09 381		69 843	968 99		74 423	71 282	
Ω.	VZN781	Medori	0 648	11 676		10,694	12 042		11 380	13 777	
<u> </u>	KZN282		1				71.71		00011	7	
п	KZN283		6 311	6 643	5 452	966 9	7 364	6 043	7 444	7 836	6 431
В	KZN284		20 245	19 105		22 441	21 176		23 880	22 535	
В	KZN285		7 126	7 028		7 899	7 790		8 406	8 290	
В	KZN286		15 520	15 062		17 203	16 696		18 307	17 767	
C	DC28	uThungulu District Municipality									
Total: uThungulu Municipalities	ngulu Muni	icipalities	58 850	59 513	5 452	65 232	896 29	6 043	69 417	70 200	6 431
щ	192NZX	Mandeni	18 604	16 947		20 601	997.81		21 942	19 988	
1 2	KZN292		24 415			27 036	24 518		28.736	26 114	
п	KZN293		16737			18 534	16 856		19 740	17 954	
В	KZN294		12 627	13 796		13 983	15 277		14 893	16 272	
C	DC29	iLembe District Municipality									
Total: iLembe Municipalities	be Municipa	alities	72 383	68 107		80 153	75 418		85 372	80 328	
<u>~</u>	KZN431	Inowe	16 798	277 C1		18 599	14 145		19.816	15.070	
п	KZN432		3 334			3 691	3 527		3 932	3758	
В	KZN433		19 458			21 544	18 268		22 954	19 463	
В	KZN434	Ubuhlebezwe	13 925	13 934		15 418	15 428		16 426	16 437	
В	KZN435	Umzimkhulu	25 857	21 779		28 630	24 114		30 503	25 691	
C	DC43	Sisonke District Municipality									
Total: Sisonke Municipalities	ke Municipa	alities	79 371	68 172		87 882	75 481		93 631	80 419	
Total: KwaZ	Julu-Natal	Total: KwaZulu-Natal Municipalities	693 611	634 763	5 452	768 282	703 105	6 043	818 320	748 897	6 431

APPENDIX W2:
APPENDIX TO SCHEDULE 3: EQUITABLE SHARE ALLOCATIONS TO MUNICIPALITIES
BREAKDOWN OF EQUITABLE SHARE ALLOCATIONS PER LOCAL MUNICIPALITY PER SERVICE FOR DISTRICT MUNICIPALITIES AUTHORISED FOR SERVICES

		BREAKDOWNO	F EQUITABLE	SHARE FOR D	BREAKDOWN OF EQUITABLE SHARE FOR DISTRICT MUNICIPALITIES AUTHORISED FOR SERVICES	ITIES AUTHORI	SED FOR S	ERVICES	
				Ē	National/Munici	National/Municipal Financial Year		-	
Number Municipality	Water	Sanitation 2011/12 (R'000)	Refuse	Water 2	Sanitation Re 2012/13 (R'000)	Refuse Water		Sanitation 2013/14 (R'000)	Refuse
LIMPOPO									
B LIM331 Greater Giyani	42 878			47 470	32 026	5(	9950	34 116	
B LIM332 Greater Letaba	45 175			50 014	32 017	53	3 278	34 107	
LIM333	690 99			73 145	52 321	77	77 918	55 735	
LIM334	20 002			22 144	15 546	23	3 589	16 561	
B LIM335 Maruleng C DC33 Monori Diereies Municipality	18 517	11 536		20 500	12 772	21	1 838	13 605	
otal: Mopani Municipali	192 641	130 685		213 274	144 682	227	161 722	154 124	
	11 193			12 392	9 757	- 1	13 202	10 394	
LIM342	13 238			14 655	10 722	15	5 613	11 423	
LIM343	89 163			109 337	74 043	116	116 484	78 883	
B LIM344 Makhado	88 332	57 285		684 46	63 418	10/	104 181	67 564	
PCOd	211526	142 666		234 172	157 030	376	040 480	176 971	
Tom: Themse stameters	000110				707 101			107 007	
	32 097	20 519		35 553	22 728	37	37 856	24 201	
B LIM352 Aganang	27 635			30 611	18 661	32	2 594	19 870	
LIM353	23 544	16 336		26 080	18 095	27	692 2	19 267	
LIM354									
	36 472	27 648		40 400	30 625		43 017	32 608	
C DC35 Capricorn District Municipality						;			
Total: Capricorn Municipalities	119 749	81 349		132 644	90 109	141	141 236	95 946	
B IIM361 Thehesimbi	707 61			21 373	15 143		2773	16.135	
	33 598	25 145		37 220	27.855		39 657	629 62	
LIM364	37.870			41 953	33 493		44 699	35 686	
LIM365	13 199			14 622	11 943	15	5 579	12 724	
LIM366	38 329			42 460	33 915	44	45 240	36 135	
C DC36 Waterberg District Municipality									
Total: Waterberg Municipalities	142 290	110 444		157 628	122 349	16	167 948	130 359	
B LIM471 Ephraim Mogale									
LIM472									
LIM473									
LIM474									
LIM475									
C DC4/ Greater Sekhukhune District Municipality									
Total: Greater Sekhukhune Municipanties									
Total: Limpopo Municipalities	902 999	465 144		737 718	515 080	785	785 855	548 693	

APPENDIX W2:
APPENDIX TO SCHEDULE 3: EQUITABLE SHARE ALLOCATIONS TO MUNICIPALITIES
BREAKDOWN OF EQUITABLE SHARE ALLOCATIONS PER LOCAL MUNICIPALITY PER SERVICE FOR DISTRICT MUNICIPALITIES AUTHORISED FOR SERVICES

			BREAKDOWN OI	F EOUITABLE	SHARE FOR	BREAKDOWN OF EQUITABLE SHARE FOR DISTRICT MUNICIPALITIES AUTHORISED FOR SERVICES	IPALITIES A	UTHORISED	FOR SERVICES	
						National/N	National/Municipal Financial Year	cial Year		
Number	Municipality	Water	Sanitation	Refuse	Water	Sanitation	Refuse	Water	Sanitation	Refuse
			2011/12 (R'000)			2012/13 (R'000)			2013/14 (R'000)	
MPUMALANGA										
B MP304	Fixiey Na Seme									
Total: Gert Sibande Municipalities	unicipalities									
B MP311	Victor Khanye									
B MP312										
B MP313										
B MP314	Emakhazeni									
B MP315	Thembisile									
B MP316	Dr JS Moroka									
C DC31	Nkangala District Municipality									
Total: Nkangala Municipalities	ripalities									
B MP321	Тъвра Срмен									
C DC32										
Total: Ehlanzeni Municipalities	cipalities									
Total: Moumalanga Municipalities	unicipalities									

APPENDIX W2:
APPENDIX TO SCHEDULE 3: EQUITABLE SHARE ALLOCATIONS TO MUNICIPALITIES
BREAKDOWN OF EQUITABLE SHARE ALLOCATIONS PER LOCAL MUNICIPALITY PER SERVICE FOR DISTRICT MUNICIPALITIES AUTHORISED FOR SERVICES

			BREAKDOWN C	F EQUITABLE	SHARE FOR	BREAKDOWN OF EQUITABLE SHARE FOR DISTRICT MUNICIPALITIES AUTHORISED FOR SERVICES	CIPALITIES /	AUTHORISED	FOR SERVICES	
				,		National/	National/Municipal Financial Year	ncial Year		
Number	Municipality	Water	Sanitation 2011/12 (R'000)	Refuse	Water	Sanitation 2012/13 (R'000)	Refuse	Water	Sanitation 2013/14 (R'000)	Refuse
NORTHERN CAPE										
B NC061	Richtersveld									
	Nama Khoi									
NC064	Kamiesberg									
NC065	Hantam									
B NC066	Karoo Hoogiand									
NC06/ DC6	Knat-Ma Namakwa District Municipality									
Total: Namakwa Municipalities	alities									
B NC071	Միստի									
NC072	Umsobomvu									
	Emthanjeni									
NC074	Kareeberg									
	Renosterberg									
NC076	Thembelinle									
NC078	Siyanciima									
DC7	Pixlev Ka Seme District Municipality									
Total: Pixley Ka Seme Municipalities	nicipalities									
B NC081	Mior									
NC082	Kaj   Garib									
NC083	//Khara Hais									
NC084	!Kheis									
NC085	Tsantsabane									
DC8	Kgatelopele Sivanda District Municipality									
pal	ities									
100014	Cal Blackia									
	Sol Flaatje Dikeatlone									
NC093	Magareng									
<b>-</b>	Phokwane									
C DC3	Frances Baard District Municipality									
Total: Frances Baard Municipalities	nicipalities									
B NC451	Moshaweng			10 703			11 939			12 762
NC452	Ga-Segonyana									
B NC453	Gamagara									
Į ĝ	John I aolo Gaetsewe District Municipality we Municipalities			10 703			11 939			12.762
Total: Northern Cape Municipalities	micipalities			10 703			11 939			12 762

APPENDIX W2:
APPENDIX TO SCHEDULE 3: EQUITABLE SHARE ALLOCATIONS TO MUNICIPALITIES
BREAKDOWN OF EQUITABLE SHARE ALLOCATIONS PER LOCAL MUNICIPALITY PER SERVICE FOR DISTRICT MUNICIPALITIES AUTHORISED FOR SERVICES

			BREAKDOWN OF EQUITA	BLE SHARE FOR	BREAKDOWN OF EQUITABLE SHARE FOR DISTRICT MUNICIPALITIES AUTHORISED FOR SERVICES	S AUTHORISED I	FOR SERVICES	
	_				National/Municipal Financial Year	nancial Year		
Number	Municipality	Water	Sanitation Refuse	Water	Sanitation Refuse	Water	Sanitation	Refuse
			2011/12 (R'000)		2012/13 (R'000)		2013/14 (R'000)	
NORTH WEST								
B NW371	Moretele							
B NW3/4	Kgetlengrivier Meese Return							
Total: Bojanala Platinum Municipalities	ım Municipalities							
1								
B NW381	Ration	17 383	12 684	19 244	14 043	20 492	14 953	
B NW382	Tswaing	21 398	16 257	23 690		25 226	19 165	
B NW383		39 741	30 800	43 996		46 850	36 309	
B NW384		25 884	22 015	28 656		30 514	25 953	
B NW385	Ramotshere Moiloa	26 638	17 370	29 491		31 403	20 477	
C DC38	Ngaka Modiri Molema District Municipality							
Total: Ngaka Modiri Molema Municipalities	Iolema Municipalities	131 045	99 125	145 077	109 739	154 485	116 856	
						!		
		11 872	9 858	13 140		13 962	11 593	
		9 533	8 348	10 551		11 211		
B NW394	Greater Taung	32 832	22 503	36 338		38 612	26 465	
		8 740	8 823	9 673	9 765	10 279	10 376	
-		19 997	13 996	22 133		23 518	16 460	
C DC39	Dr Ruth Segomotsi Mompati District Municipality							
Total: Dr Ruth Segomo	Total: Dr Ruth Segomotsi Mompati Municipalities	82 973	63 527	91 835	70 311	97 582	74 712	
G. 107.WIM								
B NW/407	Veniciscoip							
Total: Dr Kenneth Kaunda Municipalities	nda Municipalities							
Total. Month Wort Municipalities		014 010	163 653	226 017	150 051	250 050	101 500	
10tal: Morth West Man.	nerpannes	OTA 5.77	7C0 70T	71 C OC7		100 707	121 200	

APPENDIX W2:
APPENDIX TO SCHEDULE 3: EQUITABLE SHARE ALLOCATIONS TO MUNICIPALITIES
BREAKDOWN OF EQUITABLE SHARE ALLOCATIONS PER LOCAL MUNICIPALITY PER SERVICE FOR DISTRICT MUNICIPALITIES AUTHORISED FOR SERVICES

Number         Municipality           WESTERN CAPE         A           CPT         City of Cape Town           B         WC011           Marzikama         B           B         WC012           Cederberg         B           B         WC013           B         WC014           B         WC015           B         WC015           B         WC015           C         DC1           Total:         West Coast Municipalities	Water	Sanitation	Refuse			National/Municipal Financial Year	ncial Year		
WESTERN CAPE	1	Salitration		Woton	Conitotion	Dofines		Conitotion	Contract
WESTERN CAPE           A         CPT         City of Cape Town           B         WC011         Marzikama           B         WC012         Cederberg           B         WC013         Begriver           B         WC014         Saldanha Bay           B         WC015         Swartland           C         DC1         West Coast District Municipality           Total: West Coast Municipalities	,	2011/12 (K.000)		water	2012/13 (R'000)	weinse	Water	2013/14 (R'000)	метизс
A CPT City of Cape Town  B WC011 Mazikama B WC012 Cederberg B WC013 Bergrivier B WC014 Saldanha Bay WC015 Swartland C DC1 West Coast District Municipality  Total: West Coast Municipalities									
WC011 Mazikama   WC011 Mazikama   WC012 Cederberg   WC013 Begrivier   WC014 Saldama Bay   WC014 Saldama Bay   WC015 Swartland   C   DC1 West Coast District Municipality   Total: West Coast Municipalities									
B   WC011 Matzikama									
B   WC012 Cederberg									
B WC013 Bergriver B WC014 Saldanha Bay B WC015 Swartland C DC1 West Coast District Municipality Total: West Coast Municipalities									
B WCO15 Swarfand C DC1 West Coast District Municipality  Total: West Coast Municipalities									
C DCI West Coast District Municipality Total: West Coast Municipalities									
Total: West Coast Municipalities									
_									
B WC022 Witzenberg									
WC024									
WC025									
B WC026 Langeberg C DC2 Cane Winelands District Municipality									
ofal: Cane Winelands M									
Commission of the commission o									
WC031									
WC032									
WC033									
B WC034 Swellendam C DC3 Overhere District Municipality									
Total: Overberg Municipalities									
4									
WC042									
WC043									
B WC045 Oudishoom									
WC047									
WC048									
C DC4 Eden District Municipality									
Total: Eden Municipalities									
B WC051 Lainoshuro									
WC052									
DC5									
Total: Central Karoo Municipalities									
Total: Western Cape Municipalities									
Unallocated:									
N-410-4-1	2 210 502	1 037 137	17.155	2 442 600	000 0000	17.000	2 (01 502	2160226	10 103

### APPENDIX W3: APPENDIX W3: APPENDIX TO SCHEDULE 6: MUNICIPAL INFRASTRUCTURE GRANT

### (BREAKDOWN OF MIG ALLOCATIONS PER LOCAL MUNICIPALITY FOR DISTRICT MUNICIPALITIES AUTHORISED FOR **SERVICES**)

(National and Municipal Financial Years)

### APPENDIX W3: APPENDIX TO SCHEDULE 6: MUNICIPAL INFRASTRUCTURE GRANT

### BREAKDOWN OF MIG ALLOCATIONS PER LOCAL MUNICIPALITY FOR DISTRICT MUNICIPALITIES AUTHORISED FOR SERVICES

		Breakdown	of MIG alloca	ations for distr	ict municipali	ties authorised	l for services
			onal Financial			cipal Financia	
G-1	3.6	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14
Category	Municipality	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)
FREE STATE							
A MAN	Mangaung						
B FS161	Letsemeng						
B FS162	E						
B FS163							
B FS164							
C DC16							
Total: Xhariep Mun							
_							
B FS181	Masilonyana						
B FS182	Tokologo						
B FS183	Tswelopele						
B FS184	Matjhabeng						
B FS185	Nala						
C DC18	Lejweleputswa District Municipality						
Total: Lejweleputsw							
B FS191	Setsoto						
B FS192	Dihlabeng						
B FS193							
B FS194	E						
B FS195							
B FS196							
C DC19	Thabo Mofutsanyana District Municipality						
Total: Thabo Mofuts	sanyana Municipalities						
B FS201	Moqhaka						
B FS203	•						
B FS204							
B FS205							
C DC20							
Total: Fezile Dabi M							
Total. Pezile Dabi W	инстранись						
Total: Free State Mu	micipalities						

		Breakdown	of MIG alloca	ations for distr	ict municipali	ties authorised	l for services
		Natio	onal Financial	Year	Muni	cipal Financia	l Year
Category	Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
GAUTENG						, ,	
A EKU	Ekurhuleni						
A JHB	City of Johannesburg						
A TSH	City of Tshwane						
B GT421							
B GT422							
B GT423	3 Lesedi						
C DC42	Sedibeng District Municipality						
Total: Sedibeng Mu	nicipalities						
B GT481	Mogale City						
B GT482	. 8						
B GT483							
B GT484							
C DC48	ē ,						
Total: West Rand M							
	•						
Total: Gauteng Mun	icipalities						

	Breakdown	of MIG allocat	tions for distri	ct municipalit	ies authorised	for services
	Natio	nal Financial `	Year	Munio	ipal Financial	Year
Category Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
KWAZULU-NATAL						
A ETH eThekwini						
B KZN211 Vulamehlo	31 080	37 790	39 869	31 080	37 790	39 869
B KZN212 uMdoni	14 775	17 965	18 953	14 775	17 965	18 953
B KZN213 Umzumbe	78 115	94 981	100 205	78 115	94 981	100 205
B KZN214 uMuziwabantu	33 657	40 924	43 175	33 657	40 924	43 175
B KZN215 Ezinqolweni	20 316	24 703	26 061	20 316	24 703	26 061
B KZN216 Hibiscus Coast	61 102	74 295	78 381	61 102	74 295	78 381
C DC21 Ugu District Municipality						
Total: Ugu Municipalities	239 046	290 658	306 644	239 046	290 658	306 644
B KZN221 uMshwathi	28 747	34 953	36 876	28 747	34 953	36 876
B KZN222 uMngeni	8 073	9 816	10 356	8 073	9 816	10 356
B KZN223 Mpofana	7 421	9 023	9 519	7 421	9 023	9 519
B KZN224 Impendle	7 105	8 640	9 115	7 105	8 640	9 115
B KZN225 Msunduzi	15.005	10.425	20.505	15.005	10.425	20.505
B KZN226 Mkhambathini	15 986	19 437	20 506	15 986	19 437	20 506
B KZN227 Richmond	18 519	22 517	23 756	18 519	22 517	23 756
C DC22 uMgungundlovu District Municipality  Total: uMgungundlovu Municipalities	85 850	104 386	110 127	85 850	104 386	110 127
Total: uvigungundiovu viumcipanties	05 050	104 360	110 127	05 050	104 360	110 127
B KZN232 Emnambethi-Ladysmith	39 826	48 424	51 088	39 826	48 424	51 088
B KZN233 Indaka	24 059	29 254	30 863	24 059	29 254	30 863
B KZN234 Umtshezi	9 966	12 117	12 784	9 966	12 117	12 784
B KZN235 Okhahlamba	40 050	48 697	51 376	40 050	48 697	51 376
B KZN236 Imbabazane	36 840	44 794	47 257	36 840	44 794	47 257
C DC23 Uthukela District Municipality						
Total: Uthukela Municipalities	150 740	183 286	193 367	150 740	183 286	193 367
· · · · · · · · · · · · · · · · · · ·						
B KZN241 Endumeni	4 926	5 989	6 319	4 926	5 989	6 319
B KZN242 Nquthu	53 018	64 465	68 011	53 018	64 465	68 011
B KZN244 Msinga	73 386	89 231	94 138	73 386	89 231	94 138
B KZN245 Umvoti	30 612	37 222	39 269	30 612	37 222	39 269
C DC24 Umzinyathi District Municipality						
Total: Umzinyathi Municipalities	161 942	196 907	207 737	161 942	196 907	207 737
D VITTO TO VI						
B KZN252 Newcastle		10.550		0.50	10.5-0	
B KZN253 Emadlangeni	8 684	10 558	11 139	8 684	10 558	11 139
B KZN254 Dannhauser	32 528	39 550	41 725	32 528	39 550	41 725
C DC25 Amajuba District Municipality	44.044	F0 100	52.045	41.211	E0 100	F0 0 < =
Total: Amajuba Municipalities	41 211	50 109	52 865	41 211	50 109	52 865

### BREAKDOWN OF MIG ALLOCATIONS PER LOCAL MUNICIPALITY FOR DISTRICT MUNICIPALITIES AUTHORISED FOR SERVICES

Breakdown of MIG allocations for district municipalities authorised for services **Municipal Financial Year** National Financial Year 2013/14 2011/12 2011/12 2013/14 2012/13 2012/13 Category Municipality (R'000)(R'000)(R'000)(R'000)(R'000)(R'000)22 303 27 119 22 303 27 119 KZN261 eDumbe 28 610 28 610 37 485 45 578 48 085 В KZN262 UPhongolo 48 085 37 485 45 578 В KZN263 Abaqulusi 45 393 55 194 58 229 45 393 55 194 58 229 В KZN265 65 563 79 719 84 104 65 563 79 719 84 104 Nongoma lв KZN266 Ulundi 56 355 68 523 72 291 56 355 68 523 72 291 Zululand District Municipality DC26 **Total: Zululand Municipalities** 227 100 291 319 227 100 291 319 276 132 276 132 В KZN271 Umhlabuyalingana 53 292 64 798 68 362 53 292 64 798 68 362 В KZN272 Jozini 63 513 77 225 81 473 63 513 77 225 81 473 В KZN273 The Big 5 False Bay 10 523 12 795 13 498 10 523 12 795 13 498 В KZN274 Hlabisa 21 673 26 353 27 802 21 673 26 353 27 802 В KZN275 Mtubatuba 37 941 46 133 48 670 37 941 46 133 48 670 DC27 Umkhanyakude District Municipality Total: Umkhanyakude Municipalities 186 942 227 304 239 806 186 942 227 304 239 806 29 905 KZN281 Mfolozi 29 905 36 362 38 362 36 362 38 362 lв KZN282 uMhlathuze lв KZN283 Ntambanana 20.814 25 308 26 700 20.814 25 308 26 700 В KZN284 Umlalazi 58 648 71 311 75 233 58 648 71 311 75 233 В KZN285 Mthonjaneni 13 091 15 917 16 793 13 091 15 917 16 793 В KZN286 Nkandla 38 475 46 782 49 355 38 475 46 782 49 355 DC28 uThungulu District Municipality Total: uThungulu Municipalities 160 933 195 680 206 442 160 933 195 680 206 442 В KZN291 Mandeni 30 696 37 324 30 696 37 324 39 377 33 336 40 533 42 762 33 336 40 533 42 762 lв KZN292 KwaDukuza 52 056 54 919 54 919 lв KZN293 Ndwedwe 42 813 42 813 52 056 KZN294 Maphumulo 52 537 52 537 lв 40 956 49 798 40 956 49 798 DC29 iLembe District Municipality 147 801 179 711 189 596 147 801 179 711 189 596 Total: iLembe Municipalities В KZN431 Ingwe 31 466 38 260 40 365 31 466 38 260 40 365 В KZN432 Kwa Sani 3 701 4 501 4 748 3 701 4 501 4 748 В KZN433 Greater Kokstad 8 744 10 632 11 217 8 744 10 632 11 217 34 338 41 751 44 048 34 338 44 048 В KZN434 Ubuhlebezwe 41 751 В KZN435 Umzimkhulu 58 360 70 960 74 863 58 360 70 960 74 863 DC43 Sisonke District Municipality Total: Sisonke Municipalities 136 610 166 105 175 240 136 610 166 105 175 240 Total: KwaZulu-Natal Municipalities 1 538 176 1 870 278 1 973 143 1 538 176 1 870 278 1 973 143

### BREAKDOWN OF MIG ALLOCATIONS PER LOCAL MUNICIPALITY FOR DISTRICT MUNICIPALITIES AUTHORISED FOR SERVICES

Breakdown of MIG allocations for district municipalities authorised for services National Financial Year **Municipal Financial Year** 2011/12 2011/12 2013/14 2013/14 2012/13 2012/13 Category Municipality (R'000)(R'000)(R'000)(R'000)(R'000)(R'000)LIMPOPO LIM331 Greater Giyani 59 816 72 731 76 731 59 816 72 731 76 731 LIM332 Greater Letaba 58 183 70 745 74 636 58 183 70 745 74 636 В LIM333 Greater Tzaneen 102 351 124 449 131 293 102 351 124 449 131 293 lв 22 340 lв LIM334 Ba-Phalaborwa 22 340 27 163 28 657 27 163 28 657 LIM335 Maruleng 20.529 24 962 20 529 24 962 26 334 lΒ 26 334 Mopani District Municipality DC33 337 652 263 219 337 652 263 219 320 050 Total: Mopani Municipalities 320 050 В LIM341 Musina 6 487 7 888 8 321 6 487 7 888 8 321 LIM342 Mutale 24 148 29 361 30 976 24 148 29 361 30 976 В LIM343 Thulamela 145 826 177 311 187 063 145 826 177 311 187 063 119 816 119 816 В LIM344 Makhado 145 685 153 698 145 685 153 698 DC34 Vhembe District Municipality Total: Vhembe Municipalities 296 276 360 245 380 058 296 276 360 245 380 058 42 056 53 948 lв LIM351 Blouberg 42 056 51 136 53 948 51 136 lв LIM352 Aganang 38 440 46 740 49 310 38 440 46 740 49 310 В LIM353 Molemole 30 383 36 943 38 975 30 383 36 943 38 975 lв LIM354 Polokwane В LIM355 Lepelle-Nkumpi 63 035 76 644 80 860 63 035 76 644 80 860 DC35 Capricorn District Municipality Total: Capricorn Municipalities 173 914 211 462 223 093 173 914 211 462 223 093 В LIM361 Thabazimbi LIM362 Lephalale lв LIM364 Mookgopong lв LIM365 Modimolle lв lв LIM366 Bela Bela lв LIM367 Mogalakwena DC36 Waterberg District Municipality Total: Waterberg Municipalities Ephraim Mogale LIM471 50 598 85 256 89 945 50 598 85 256 89 945 113 430 191 125 201 637 113 430 191 125 201 637 В LIM472 Elias Motsoaledi 150 210 LIM473 Makhuduthamaga 253 098 267 018 150 210 253 098 267 018 lв LIM474 Fetakgomo 51 361 86 541 91 301 51 361 86 541 91 301 В LIM475 Greater Tubatse 142 638 240 339 253 558 142 638 240 339 253 558 lв Greater Sekhukhune District Municipality DC47 508 236 903 459 508 236 903 459 856 359 856 359 Total: Greater Sekhukhune Municipalities **Total: Limpopo Municipalities** 1 241 645 1 748 116 1 844 262 1 241 645 1 748 116 1 844 262

		Breakdown	of MIG alloca	tions for distr	rict municipali	ties authorised	l for services
		Natio	nal Financial	Year	Muni	cipal Financia	l Year
G .	36	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14
Category	Municipality	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)
MPUMALANGA							
B MP30	l Albert Luthuli						
B MP30	2 Msukaligwa						
B MP30	3 Mkhondo						
B MP30	4 Pixley Ka Seme						
B MP30	5 Lekwa						
B MP30	6 Dipaleseng						
B MP30	7 Govan Mbeki						
C DC3							
Total: Gert Sibando	Municipalities						
B MP31							
B MP31							
B MP31							
B MP31							
B MP31							
B MP31							
C DC3	<u> </u>						
Total: Nkangala M	ınicipalities						
B MP32	1 Thaba Chweu						
B MP32							
B MP32							
B MP32	3						
B MP32							
C DC3	č						
Total: Ehlanzeni M							
Total: Mpumalanga	Municipalities						

		Breakdown	of MIG alloca	ntions for distr	ict municipali	ties authorised	l for services
		Natio	onal Financial	Year	Muni	cipal Financia	l Year
Category	Municipality	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14
		(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)
NORTHERN CAPE							
B NC061	Richtersveld						
B NC062	Nama Khoi						
B NC064	Kamiesberg						
B NC065							
B NC066	2						
B NC067	Khâi-Ma						
C DC6	Namakwa District Municipality						
Total: Namakwa Mu	nicipalities						
B NC071	Ubuntu						
B NC072							
B NC073							
B NC074							
B NC075							
B NC076	E						
B NC077							
B NC078	•						
C DC7	Pixley Ka Seme District Municipality						
Total: Pixley Ka Sem	e Municipalities						
B NC081	Mier						
B NC082	!Kai !Garib						
B NC083	//Khara Hais						
B NC084	!Kheis						
B NC085	Tsantsabane						
B NC086	Kgatelopele						
C DC8	Siyanda District Municipality						
Total: Siyanda Muni	cipalities						
B NC091	Sol Plaatje						
B NC092	3						
B NC093							
B NC094	C C						
C DC9	Frances Baard District Municipality						
Total: Frances Baard							
- Juni Frances Dadio							
B NC451	Moshaweng						
B NC452							
B NC453							
C DC45	John Taolo Gaetsewe District Municipality						
Total: John Taolo Ga	etsewe Municipalities						
Total: Northern Cap	Municipalities						
u							

	Breakdown	of MIG alloca	tions for distri	ict municipalit	ties authorised	for services
	Natio	nal Financial		Munio	cipal Financia	Year
Category Municipality	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14
omiga-y mana-pan-y	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)
NORTH WEST						
B NW371 Moretele						
B NW372 Madibeng						
B NW373 Rustenburg						
B NW374 Kgetlengrivier						
B NW375 Moses Kotane						
C DC37 Bojanala Platinum District Municipality						
Total: Bojanala Platinum Municipalities						
D NW201 Deflere	22.202	40.272	12.502	22 202	40.272	42 592
B NW381 Ratlou B NW382 Tswaing	33 203 20 928	40 372 25 447	42 592 26 846	33 203 20 928	40 372 25 447	42 392 26 846
B NW382 Tswaing B NW383 Mafikeng	61 239	74 460	78 556	61 239	74 460	26 846 78 556
B NW384 Ditsobotla	25 616	31 146	32 859	25 616	31 146	76 336 32 859
B NW384 Ditsobotia B NW385 Ramotshere Moiloa	26 478	32 195	32 839	26 478	32 195	32 839
C DC38 Ngaka Modiri Molema District Municipality	20 478	32 193	33 903	20 4 / 8	32 193	33 903
Total: Ngaka Modiri Molema Municipalities	167 463	203 619	214 818	167 463	203 619	214 818
Total: Agana Mount Motema Municipalities	107 403	203 017	214 010	107 403	203 017	214 010
B NW392 Naledi	4 214	5 124	5 406	4 214	5 124	5 406
B NW393 Mamusa	8 517	10 355	10 925	8 517	10 355	10 925
B NW394 Greater Taung	45 030	54 752	57 764	45 030	54 752	57 764
B NW396 Lekwa-Teemane	4 592	5 583	5 890	4 592	5 583	5 890
B NW397 NW397	29 232	35 544	37 498	29 232	35 544	37 498
C DC39 Dr Ruth Segomotsi Mompati District Municipality						
Total: Dr Ruth Segomotsi Mompati Municipalities	91 585	111 358	117 483	91 585	111 358	117 483
B NW401 Ventersdorp						
B NW402 Tlokwe						
B NW403 City of Matlosana						
B NW404 Maquassi Hills						
C DC40 Dr Kenneth Kaunda District Municipality						
Total: Dr Kenneth Kaunda Municipalities						
Total: North West Municipalities	259 048	314 978	332 301	259 048	314 978	332 301

		Breakdown	of MIG alloca	tions for distr	ict municipalit	ies authorised	for services
			nal Financial			ripal Financial	
Category	Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
WESTERN CAPE		(11 000)	(11 000)	(11 000)	(11 000)	(N ooo)	(11 000)
A CPT	City of Cape Town						
B WC011	Matzikama						
B WC011							
B WC013	Bergrivier						
B WC014	Saldanha Bay						
B WC014	Swartland						
C DC1	West Coast District Municipality						
Total: West Coast Mu							
Total. West Coast Mu	пераписа						
B WC022	Witzenberg						
B WC023	Drakenstein						
B WC024	Stellenbosch						
B WC025	Breede Valley						
B WC026	·						
C DC2	Cape Winelands District Municipality						
Total: Cape Winelands							
	· · · · · · · · · · · · · · · · · · ·						
B WC031	Theewaterskloof						
B WC032	Overstrand						
B WC033	Cape Agulhas						
B WC034	Swellendam						
C DC3	Overberg District Municipality						
Total: Overberg Muni	cipalities						
B WC041	Kannaland						
B WC042	Hessequa						
B WC043	Mossel Bay						
B WC044	George						
B WC045	Oudtshoorn						
B WC047	Bitou						
B WC048	Knysna						
C DC4	Eden District Municipality						
Total: Eden Municipal	ities						
D WC051	Lainachana						
B WC051 B WC052	Laingsburg Prince Albert						
B WC052 B WC053	Beaufort West						
C DC5	Central Karoo District Municipality						
Total: Central Karoo							
Totali Celli al Isal 00	viumeipanues					+	
Total: Western Cape N	<b>Junicipalities</b>						
Unallocated:							
National Total		4 659 673	5 904 118	6 228 845	4 659 673	5 904 118	6 228 845
/		1					

# APPENDIX TO SCHEDULE 7: REGIONAL BULK INFRASTRUCTURE GRANT

BREAKDOWN OF REGIONAL BULK INFRASTRUCTURE GRANT ALLOCATIONS PER LOCAL MUNICIPALITY PER PROJECT

(National and Municipal Financial Years)

APPENDIX TO SCHEDULE 7: REGIONAL BULK INFRASTRUCTURE GRANT

BREAKDOW	BREAKDOWN OF REGIONAL BULK INFRASTRUCTURE GRANT ALLOCATIONS PER LOCAL	ALLOCATION	S PER LOCAL MUNICIPALITY PER PROJECT		Natio	National Financial Year	ear	Munic	Municipal Financial Year	ear
Project Code	Project Name	Category	Water Service Authority	Benefiting Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
	EASTERN CAPE									
ECR011	Ndlambe Dam/ Albany Coast BWS (Grahanstown& Port	C DCI0	Cacadu Distrct Municipality	Ndlame Local Municipality	7 000	17 000	31 000	7 000	17 000	31 000
ECR024	Sundays River- Paterson Bulk Water Supply	B EC106	Sundays River Valley Local Municipality	Sundays River Valley Local Municipality	16 000	8 300	11 000	16 000	8 300	11 000
		Total: Cacadu	Total: Cacadu Municipalities		23 000	25 300	42 000	23 000	25 300	42 000
ECR010	Mncwansa Bulk Water Supply	C DC12	Amatole District Municipality	Mbashe Local Municipality	24 000	12 000		24 000	12 000	
ECR015	Xhora East Water Supply	C DC12	Amatole District Municipality	Mbashe Local Municipality	11 700			11 700		
ECR007	Idutywa Water Supply	C DC12	Amatole District Municipality	Mbashe Local Municipality		5 000	2 000		2 000	5 000
ECR018	Sundwana Water Supply	C DC12	Amatole District Municipality	Mbashe Local Municipality		2 000	2 000		2 000	5 000
ECR006	Ibika Water supply	C DC12	Amatole District Municipality	Mnquma Local Municipality	20 000	11 200		20 000	11 200	
		Total: Amatole	Total: Amatole District Municipalities		25 700	33 200	10 000	55 700	33 200	10 000
FCR033	Cluster 4 CHDM Bulk Water Sumuly	DC13	Chris Hani District Municinality	Enecoho Local Municinality	8 500	23.250	40 000	8 500	23.250	40 000
FCR028	Cluster 6 CHDM Bulk Water Sumuly	C DC13	Chris Hani District Municipality	Enecoho Local Municipality	20 000	57 000	20 000	20 000	57 000	20 000
ECR029	Cluster 9 CHDM Bulk Water Supply	C DC13	Chris Hani District Municipality	Intsika yethu Local Municipality	21 500	35 500	45 000	21 500	35 500	45 000
ECR002	Xonxa Bulk Water Supply	C DC13	Chris Hani District Municipality	Engcobo Local Municipality	38 000	45 100	55 000	38 000	45 100	55 000
ECR005	Hofmeyer Water Supply	C DC13	Chris Hani District Municipality	Engcobo Local Municipality	12 000	9 250		12 000	9 250	
ECR005a	Middelburg Bulk Water Supply	C DC13	Chris Hani District Municipality	Engcobo Local Municipality	12 000	9 250		12 000	9 250	
		Total: Chris Ha	Total: Chris Hani District Municipalities		112 000	179 350	210 000	112 000	179 350	210 000
ECR012	OR Tambo Mthatha Feasibility Study	C DCI5	OR Tambo District Municipality	King Sabatha Dalindyebo LM	2 000			2 000		
ECR016	OR Tambo regional extensions (Coffee Bay)	C DC15	OR Tambo District Municipality	King Sabatha Dalindyebo LM	2 120	000 9	10 000	2 120	0009	10 000
		Total: O.R. Ta	Total: O.R. Tambo District Municipalities		4 120	000 9	10 000	4 120	000 9	10 000
ECR008	Mbizana Regional Bulk Water Supply		Alfred Nzo District Municipality	Mbizana Local Municipality	92 780	98 800	95 000	92 780	98 800	95 000
ECK001	Alfred Nzo (Matatiele & Mount Ayliff) BWS scheme	C DC4	C DC44 Alfred Nzo District Municipality	Mzimvubu & Matatiele LMs	28 602	24 000	20 000	28 602	24 000	20 000
		1 otal: Alfred N	Zo District Municipanties		786 171	177 900	145 000	171 397	177 900	145 000
		Total: Eastern	Total: Fastern Cape Municipalities		316 202	366.650	417 000	316 202	366 650	417 000
		* Common annual **	Calbo transmission						- 20 000	

	FREE STATE									
FSR001/7	Letsemeng - Petrusburg, Koffiefontein	B FS161	Letsemeng Local Municipality	Letsemeng Local Municipality						
FSR002	Jagersfontein /Fauresmith Bulk Water Supply	B FS162	Kopanong Local Municipality	Kopanong Local Municipality	30 000	72 300		30 000	72 300	
FSR005	Rouxville /Smithfield /Zastron Bulk Water Supply	B FS163	Mohokare Local Municipality	Mohokare Local Municipality	10 894	13 000	25 000	10 894	13 000	25 000
FSR001/5	Naledi Bulk Water Supply	B FS164	Naledi Local Municipality	Naledi Local Municipality			10 000			10 000
		C DC16	Xhariep District Municipality		40 894	85 300	35 000	40 894	85 300	35 000
FSR008	Tokologo Regional Water Supply	B FS182	B FS182 Tokologo Local Municipality	Tokologo, Saul Platije LMs	49 000	20 300	3 000	49 000	20 300	3 000
		C DC18	Lejweleputswa District Municipality		49 000	20 300	3 000	49 000	20 300	3 000
FSR007	Sterkfontein Dam Bulk Water Sumly	B FS194	Maluti-a-Phofung Local Municipality	Maluti-a-Phofung Local Municipality	08 780	30 400		08 7 80	30 400	
FSR001/9	Phumelela Bulk Water Supply	B FS195		Phumelela Local Municipality	000 6		15 500	000 6		15 500
FSR001/6	Mantsopa Bulk Water Supply	B FS196	Mantsopa Local Municipality	Mantsopa Local Municipality			10 000			10 000
		C DC19	C DC19 Thabo Mofutsanyana District Municipality		177 780	30 400	25 500	77 780	30 400	25 500
		Total: Free St	Total: Free State Municipalities		167 674	136 000	63 500	167 674	136 000	63 500

440 383

342 400

456 400

440 383

Total: KwaZulu-Natal Municipalities

### APPENDIX TO SCHEDULE 7: REGIONAL BULK INFRASTRUCTURE GRANT

BREAKDOV	BREAKDOWN OF REGIONAL BULK INFRASTRUCTURE GRANT ALLOCATIONS PER LOCAL MUNICIPALITY PER PROJECT	T ALLOCATION	S PER LOCAL MUNICIPALITY PER PROJECT		Nation	National Financial Year	ear	Munic	Municipal Financial Year	ear
Project Code	e Project Name	Category	Water Service Authority	Benefiting Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
	GAUTENG									
GPR003	Upgrade of Western Highveld Water Treatment Plant	B GT462	City of Tshwane	City of Tshwane	30 000	49 000	61 000	30 000	49 000	61 000
					000 00	200 6	000 00	000 00	000 64	000 10
GPR001	Sedibeng Bulk Regional Sewerage Scheme	B GT421 C DC42	Emfuleni Local Municipality Sedibeng District Municipality	Emfuleni Local Municipality	30 000	57 000 57 000	70 000	30 000	57 000 57 000	70 000
CODBOOL	Wastonaria Baritanal Bulb Sanitation		Watrangia I ocal Municipality	Waeronaria I coal Municipality	00000	005 CF	00009	00000	005 CF	00009
7000	Westerlain Negional Dan Samanon		West Rand District Municipality	resonana zoca manerpanej	20 000	42 500	000 09	20 000	42 500	000 09
		Total: Gauteng	Total: Gauteng Municipalities		80 000	148 500	191 000	80 000	148 500	191 000
	KWAZULU-NATAL									
KNR013	Mhlabatshane Bulk Water Supply	C DC21	Ugu District Municipality	uMzumbe Local Municipality	40 000	40 000	42 823	40 000	40 000	42 823
		Total: Ugu Municipalities	nicipalities	•	40 000	40 000	42 823	40 000	40 000	42 823
KNR006	Greater Eston Water Scheme	C DC 22	uMgungundlovu District Municipality	Mkhambathini/ Richmond LM's	25 000	33 000	25 000	25 000	33 000	25 000
		Total: uMgung	Total: uMgungundlovu Municipalities		25 000	33 000	25 000	25 000	33 000	25 000
KNR010 KNR016	Driefontein Complex Bulk Water Supply Driefontein Phase 2	C DC23 C DC23	Uthukela District Municipality Uthukela District Municipality	Emnambithi/Ladysmith Local municipality Emnambithi/Ladysmith Local municipality	18 430	44 000	46 910	18 430	44 000	46 910 19 000
		Total: Uthukel	Total: Uthukela Municipalities		18 430	44 000	65 910	18 430	44 000	65 910
KNR008	Greytown Regional Bulk Scheme	C DC24	Umzinyathi District Municipality	Umvoti Local Municipality	7 437	37 000	41 813	7 437	37 000	41 813
		Total: Umzinya	Total: Umzinyathi Municipalities		7 437	37 000	41 813	7 437	37 000	41 813
KNR009 KNR017	Emadlangeni Bulk Regional Scheme Buffalo Flats Phase 2	C DC25 C DC25	Amajuba District Municipality Amajuba District Municipality	Newcastle Local Municipality Newcastle Local Municipality	22 871		25 000	22 871		25 000
		Total: Amajub	Total: Amajuba Municipalities		22 871		25 000	22 871		25 000
KNR001 KNR002	Nongoma Bulk Water Supply Mandlakazi Bulk Water Supply	C DC26	Zululand District Municipality Zululand District Municipality	Nongoma Local Municipality uPhongola & Nongoma LMs	25 000	28 000	48 413	25 000	28 000	48 413
		Total: Zululan	Total: Zululand Municipalities	3	000 09	74 928	48 413	000 09	74 928	48 413
KNR015 KNR003	Pongolapoort Bulk water Scheme Habita Bulk Water Sumely	C DC27	Umkhanyakude District Municipality IImkhannsakude District Municipality	Jozini Local Municipality Hishica Local Municipality	26 000	37 000	38 000	26 000	37 000	38 000
KNR014	Dukuduku Resettlement		Umkhanyakude District Municipality	Big 5 False Bay Local Municipality	35 000	41 000	47 000	35 000	41 000	47 000
		Total: Umkhar	Total: Umkhanyakude Municipalities		101 000	113 878	85 000	101 000	113 878	85 000
KNR005	Greater Mthonjaneni Bulk Water Supply Phase 2	C DC28	UThungulu District Municipality	Mtonjaneni Local Municipality	33 999	40 001	20 000	33 999	40 001	50 000
		Total: u Tuung	Total: utnungun Muncipanues		666.00	T00 0#	000 00	666 66	100 04	000 00
KNR011	Ngcebo Regional Bulk (Lower Tugela)	C DC29	iLembe District Municipality	KwaDukuza Local Municipality	17 663	37 576	37 000	17 663	37 576	37 000
		Total: il.embe	Lotal: iLembe Municipalities		17 663	37.576	37 000	17 663	37 576	37 000
KNR007	Greater Bulwer Donnybrook Water Scheme		Sisonke District Municipality	Ingwe Local Municipality	16 000	20 000	20 441	16 000	20 000	20 441
KNR018	Sisonke Bulk Water Supply	C DC43 Sisonke Distri Total: Sisonke Municipalities	Sisonke District Municipality Municipalities	Sisonke District Municipality	16 000	20 000	35 441	16 000	20 000	35 441

APPENDIX TO SCHEDULE 7: REGIONAL BULK INFRASTRUCTURE GRANT

BREAKDOW	BREAKDOWN OF REGIONAL BULK INFRASTRUCTURE GRANT ALLOCATIONS FER LOCAL	T ALLOCATIONS	S PER LOCAL MUNICIPALITY PER PROJECT		Natio	National Financial Year	ear	Munic	Municipal Financial Year	/ear
Project Code	Project Name	Category	Water Service Authority	Benefiting Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
	LIMPOPO									
LPR011	Nebo Bulk Water Supply	C DC47	Greater Sekhukhune District Municipality	Greater Tubatse/ Makhudutmahaga LMs	000 09	77 000	75 000		77 000	75 000
LPRNEW1 LPR012	Moutse Bulk Water Supply   Mooihoek/Tubatse Bulk Water Supply	C DC47	Greater Sekhukhune District Municipality Greater Sekhukhune District Municipality	Elias Motsoaledi/ Greater Marble Hall Greater Tubatse Local Municipality	9 000	40 000	70 000	9 000 4	40 000	70 000
LPR009	Lebalelo Central & North RWS	C DC47	Greater Sekhukhune District Municipality	Greater Tubatse Local Municipality			10 000			10 000
		Total: Greater	Total: Greater Sekhukhune District Municipalities		117 000	187 000	225 000	117 000	187 000	225 000
LPR018	Giyani Bulk Water Supply Drought relief	C DC33	Mopani District Municipality	Greater Giyani Local Municipality	37 000	17 000			17 000	
LPR017	Mametja Sekororo	C DC33	Mopani District Municipality	Maruleng/ Richmond/ Oaks/ Willows	40 000	51 789	50 000	40 000	51 789	50 000
		Total: Mopani	Total: Mopani District Municipalities		77 000	68 289	20 000	27 000	68 789	20 000
LPR016	Sinthumule Kutama Bulk Water Supply	C DC34	Vhembe District Municipality Whombe District Municipality	Makhodo/ Kutama/ Sinthumule	42 000	45 200	900	42 000	45 200	900
7007	TABLETO VAILEY DAYS	Total: Vhembe District Munici	District Municipalities	Mariago Local Muncipanty	42 000	45 200	15 000	42 000	45 200	15 000
LPR014	Olifantspoort Water Treatment Works	C DC35	Capricorn District Municipality	Lepelle-Nkumpi/ Polokwane	39 000			39 000		
		Total: Capricon	Total: Capricorn District Municipality		39 000			39 000		
LPR015	Mogalakwena Bulk Water Supply	B NP367	Mogalakwena Local Municipality	Mogalakwena Local Municipality	40 000	76 700	80 000	40 000	76 700	80 000
		C DC36	Waterberg District Municipality		40 000	002 92	80 000	40 000	002 92	80 000
		Total: Limpopo Municipalities	Municipalities		315 000	377 689	370 000	315 000	377 689	370 000

DC30	Albert Luthuli Local Municipality	Albert Luthuli Local Municipality	2 000		12 000	2 000		12 000
	Albert Luthuli Local Municipality	Albert Luthuli Local Municipality	2 045			2 045		
DC30 N	Msukaligwa Local Municipality	Msukaligwa Local Municipality	2 045			2 045		
2 DC30 D	Dipaleseng Municipality	Dipaleseng Municipality	2 042			2 042		
DC30 G	Gert Sibande District Municipality		8 132		12 000	8 132		12 000
	Victor Kanye Local Municipality	Delmas Local Municipality	22 000	25 000	30 000	22 000	25 000	30 000
DC31 NI	Nkangala District Municipality	Dr JS Moroka/ Thembisile	1 000	15 000	20 000	1 000	15 000	20 000
C DC31 NF	Nkangala District Municipality		23 000	40 000	20 000	23 000	40 000	50 000
DC32 EI	Ehlanzeni District Municipality	Ehlanzeni District Municipality	5 500		26 000	5 500		26 000
	Mbombela Local Municipality	Mbombela Local Municipality	3 000	000 9	10 000	3 000	0009	10 000
3 MP322	Mbombela Local Municipality	Mbombela Local Municipality	3 500	10 000	10 000	3 500	10 000	10 000
	Mbombela Local Municipality	Mbombela Local Municipality	3 000	000 9	10 000	3 000	0009	10 000
3 MP325 F	Bushbuckridge Local Municipality	Bushbuckridge Local Municipality	105 900	41 000		105 900	41 000	
C DC32	Ehlanzeni District Municpality		120 900	63 000	26 000	120 900	63 000	<b>26</b> 000
							_	
Total: Mpumalanga Municip	ga Municipalities		152 032	103 000	118 000	152 032	103 000	118000

APPENDIX TO SCHEDULE 7: REGIONAL BULK INFRASTRUCTURE GRANT

BREAKDOW	BREAKDOWN OF REGIONAL BULK INFRASTRUCTURE GRANT ALLOCATIONS PER LOCAL MUNICIPALITY PER PROJECT	ALLOCATION	S PER LOCAL MUNICIPALITY PER PROJECT		Nati	National Financial Year	ear	Munic	Municipal Financial Year	ear
Project Code	Project Name	Category	Water Service Authority	Benefiting Municipality	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
	NORTHERN CAPE									
NCR005	Heuningvlei / Moshaweng Bulk Water Supply	B NC451	Moshaweng Local Municipality	Moshaweng Local Municipality	35 000	68 500		35 000	68 500	
		Total: John Taolo Gaetsewe	oto Gaetsewe Municipalities		000 cc	000 00		000 66	008 200	
NCRNEW3	Bulk water supply to Porth Nolloth	B NC061	Richtersveld Local Municipality	Richtersveld Local Municipality		18 000	000 6		18 000	000 6
NCRNEW	Refurbishment of Namakwa WTW	B NC062	Nama Khoi Local Municipality	Nama Khoi Local Municipality	16 000	41 000	103 000	16 000	41 000	103 000
NCR004	Kammiesberg / Namakwa pipeline ext	C DC6	Namakwa District Municipality	Kamiesberg Local Municipality						
NCNEW4	Hantam (Brandvlei) Desalination of groundwater	B NC065	Hantam Local Municipality	Hantam Local Municipality			8 775			8 775
		Total: Namakw	Total: Namakwa Municipalities		16 000	29 000	120 775	16 000	29 000	120 775
NCR010	Oranje river - Colesberg - Noupoort Bulk WS		Umsobomvu Local Municipality	Umsobomvu Local Municipality	32 480	34 000	20 000	32 480	34 000	20 000
NCR006/2	Van der Kloof/ Petrusville Pipeline	B NC073	Emthanjeni Local Municipality	Emthanjeni Local Municipality	17 000	18 000	2 000	17 000	18 000	5 000
NCR006/3	Thembelilhe Bulk Water Supply	B NC076	Thembelihle Local Municipality	Thembelihle Local Municipality	12 000	20 000	2 000	12 000	20 000	5 000
NCR003	Pixley Ka Seme BWS FS	C DC7	Pixley Ka Seme District Municipality	Varoius			5 000			5 000
		Total: Pixley ka	Total: Pixley ka Seme Municipalities		61 480	72 000	35 000	61 480	72 000	35 000
NCR007	Kenhari Bulk Water Sumily	B NC082	Kai Garih Local Municinality	Kai Garih I ocal Municinality	3 505			3 595		
NCR008/1	Tsantsabane Bulk Scheme	NC085	Tsantsabane Local Municipality	Tsantsabane Local Municipality	33 500			33 500		
		Total: Siyanda Municipalities	Municipalities		37 095			37 095		
NCRNEW1	Windsorton to Holpan Bulk Water Supply	B NC092	Dikgatlong	Kai Garib Local Municipality			14 000			14 000
NCR009	Vaal Gamagara pipeline investigation	C DC3	Frances Baard District Municipality	Frances Baard District Municipality			35 000			35 000
		Total: Frances	Total: Frances Baard Municipalities				49 000			49 000
		Total: Northern Cape Muni	n Cape Municipalities		149 575	199 500	204 775	149 575	199 500	204 775

	NORTH WEST									
NWR005 NWRNEW4	NWR005 Madibeng Bulk Water Supply NWRNEW4 Bakwena Regional Bulk Water Supply	B NW372 C DC37	Madibeng Local Municipality Bojanala Platinum District Municipality	Madibeng Local Municipality Bojanala Platinum District Municipality	49 000	52 900 18 000	60 000 25 000	49 000	52 900	60 000 25 000
		C DC37	Bojanala Platinum District Municipality		49 000	20 900	85 000	49 000	20 900	82 000
NWR002 NWR006	Ratlou Local Municipality Bulk Water Supply Feasbility Studies	B NW381 C DC37	Ratlou Local Municipality Ngaka Modiri Molema District Municipality	Ratlou Local Municipality Ngaka Modiri Molema District Municipality		14 000	15 000		14 000	15 000 10 000
		C DC37	Ngaka Modiri Molema District Municipality			14 000	25 000		14 000	25 000
NWR009 NWRNew1	Taung/ Naledi Bulk Water Supply Greater Mamusa Bulk Water Supply	C DC39	C DC39 Dr Ruth Mompati District Municipality C DC39 Dr Ruth Mompati District Municipality	Greater Taung Local Municipality Mamusa Local Municipality	65 000	72 000	70 000	02 000	72 000	70 000
		Total: Dr R	Total: Dr Ruth Segomotsi Mompati Municipalities		000 59	80 000	80 000	000 59	80 000	80 000
NWR010	Ventersdorp Bulk Water Supply	B NW401		Ventersdorp Local Municipality		10 000	10 000		10 000	10 000
		C DC40	Dr Kenneth Kaunda Distrct Municipality			10 000	10 000		10 000	10 000
		Total: Nort	Total: North West Municipalities		114 000	174 900	200 000	114 000	174 900	200 000

APPENDIX TO SCHEDULE 7: REGIONAL BULK INFRASTRUCTURE GRANT

### 18 000 40 40 000 12 000 5 599 20 000 2 176 274 53 000 53 000 155 599 28 000 25 599 2013/14 (R'000) Municipal Financial Year 12 000 8 000 3 700 14 895 15 000 15 000 56 595 12 000 26 295 2 003 217 3 000 3 000 2012/13 (R'000) 20 000 3 957 1 300 17 000 12 000 8 500 3 000 200 67 257 1 704 140 25 257 2011/12 (R'000) 53 000 18 000 40 40 000 5 599 20 000 155 599 2 176 274 58 000 12 000 12 000 53 000 25 599 7 000 2013/14 (R'000) National Financial Year 8 000 3 700 14 895 15 000 15 000 12 000 26 295 56 595 12 000 3 000 3 000 2 003 217 2012/13 (R'000) 1 704 140 3 957 1 300 25 257 67 257 17 000 12 000 3 000 32 000 500 500 8 500 2011/12 (R'000) Cape Winelands District Municipality Cederberg, and Matzikamma LM's West Coast District Municipality Drakenstein Local Municipality Breede Valley Local Municipality Beaufort West Local Municipality West Coast District Municipality Kannaland Local Municipality Kannaland Local Municipality Hessequa Local Municipality George Local Municipality Witzenberg Local Municipality Drakenstein Local Municipality Swellendam Local Municipality Overberg District Municipality Benefiting Municipality BREAKDOWN OF REGIONAL BULK INFRASTRUCTURE GRANT ALLOCATIONS PER LOCAL MUNICIPALITY PER PROJECT C DC2 Cape Winelands District Municipality Total: Cape Winelands Municipalities B WC053 Beaufort West Local Municipality Total: Central Karoo Municipalities Breede Valley Local Municipality Cederberg Local Municipalty West Coast District Municipality C DC1 West Coast District Municipality Total: West Coast Municipalities Witzenberg Local Municipality Drakenstein Local Municipality Swellendam Local Municipality Drakenstein Local Municipality C DC3 Overberg District Municipality Total: Overberg Municipalities | 15 WC042 | Hessequa Local Municipality | B WC044 | George Local Municipality | Total: Eden Municipalities Kannaland Local Municipality Kannaland Local Municipality Water Service Authority Total: Western Cape Municipalities Category WC023 WC023 WC025 WC012 WC034 WC041 WC041 DC DC Clanwilliam /Lambertsbaai Regional Water Supply West Coast Bulk Water Supply Tulbagh Bulk Water Supply Drakenstein Waste Water Treatment Plant Hessequa Water Supply George Bulk Water Supply Augmentation Project Name Worcester Water Supply Cape Winelands WS & WW Beaufort West Water Supply aarl South efluent pipeline Swellendam Water Supply Overberg DM WWTW Cannaland Water Supply West Coast-Cedeberg WESTERN CAPE adismith WWTW National Total Project Code WCRNew10 WCRNew11 WCR003 WCR006 WCRNew2 WCRNew7 WCRNew5 WCR002 WCRNew3 WCRNew8 WCRNew6 WCRNew4 WCR001 WCR004 WCR005

## APPENDIX TO SCHEDULE 5: BREAKDOWN OF EPWP SOCIAL SECTOR ALLOCATIONS PER PROVINCIAL DEPARTMENT FOR THE 2011/12 FINANCIAL YEAR

APPENDIX W5:

APPENDIX TO SCHEDULE 5: BREAKDOWN OF EPWP SOCIAL SECTOR ALLOCATIONS PER PROVINCIAL DEPARTMENT FOR THE 2011/12 FINANCIAL YEAR

	Social Sector Expanded	l Public Works l Province	-	tive Grant for
Province/Provincial Department	Number of FTEs		Financial Year	
	to be created from allocations in 2011/12	2011/12 (R'000)	2012/13 (R'000)	2013/14 ('R000)
EASTERN CAPE				
Social Development	294	5 070		
Total: Eastern Cape	294	5 070	7 120	9 139
FREE STATE				
Health	904	15 586		
Total: Free State	904	15 586	18 703	22 444
GAUTENG				
Social Development	874	15 071		
Health	1 147	19 777		
Total: Gauteng	2 020	34 848	41 818	50 181
KWAZULU-NATAL				
Social Development	261	4 495		
Health	1 494	25 774		
Total: KwaZulu-Natal	1 755	30 269	32 485	34 354
LIMPOPO				
Health and Social Development (Health Branch)	1 446	24 950		
Health and Social Development (Social Development Branch)	196	3 382		
Total: Limpopo	1 642	28 332	33 998	40 798
MPUMALANGA				
Health and Social Development (Health Branch)	192	3 310		
Education	574	9 901		
Safety and Liaison	11	196		
Total: Mpumalanga	777	13 407	16 088	19 306
NORTHERN CAPE				
Social Development	328	5 658		
Health	120	2 073		
Education	357	6 159		
Total: Northern Cape	805	13 890	16 669	20 002
NORTH WEST				
Education	805	13 893		
Health and Social Development (Social Development Branch)	635	10 955		
Health and Social Development (Health Branch)	1 007	17 374		
Total: North West	2 448	42 222	54 229	65 901
WESTERN CAPE		$\neg$	Т	
Education	468	8 070		
Health	502	8 664		
Total: Western Cape	970	16 734	20 394	23 903
Grand Total	11 615	200 358	241 504	286 028

# APPENDIX W6: APPENDIX W6: APPENDIX TO SCHEDULE 8: EXPANDED PUBLIC WORKS PROGRAMME

## INCENTIVES TO PROVINCES AND MUNICIPALITIES TO MEET TARGETS WITH REGARDS TO PRIORITY GOVERNMENT **PROGRAMMES**

(National and Municipal Financial Years)

 ${\small \textbf{APPENDIX W6:}} \\ \textbf{APPENDIX TO SCHEDULE 8: INCENTIVES TO PROVINCES TO MEET TARGETS WITH REGARDS TO PRIORITY GOVERNMENT PROGRAMMES} \\$ 

		Expanded		Programme Inc	entive Grant fo	r Provinces	
Province/Provincial Department	Infrastruc	ture Sector		t and Culture ctor		Financial Year	
	Minimum Threshold	FTE Performance	Minimum Threshold	FTE Performance	2011/12 (R'000)	2012/13 (R'000)	2013/14 (R'000)
	Tiresnoid	Target	Tiresnoid	Target	(1 000)	(N 000)	(N 000)
EASTERN CAPE Housing	11 899	11 948			711		
Roads and Transport	10 856	10 906			7 788		
Health Education	2 511 3 816	2 560 3 865		648	16 579 711		
Economic Development and Environmental Affairs	3 010	49		118	1 760		
Agriculture and Rural Development Sport, Recreation, Arts and Culture	412	461 117	1 187	49 1 236	535 1 277		
Social Development		49	1 107	1 230	535		
Public Works Total: Eastern Cape	29 494	49 30 006	1 187	2 051	535 30 431		
FREE STATE	29 494	30 000	1 10/	2 031	30 431		
Cooperative Governance and Traditional Affairs; Human Settlements	9 109	9 208		49	1 246		
Public Works and Rural Development Police, Roads and Transport	4 208	172 4 307		49	2 320 711		
Education	1 721	1 819		49	1 246		
Health Social Development	651	701 49		49	356 1 339		
Economic Development, Tourism and Environmental Affairs		49		49	711		
Sport, Arts, Culture and Recreation		49 49	617	49 666	536 536		
Agriculture Police, Roads and Transport		49	017	000	535		
Free State Provincial Treasury		49		40	535		
Police, Roads and Transport  Total: Free State	15 690	16 503	617	49 <b>962</b>	535 10 606		
GAUTENG							
Housing Education	28 531 3 154	28 630 3 203		246	711 2 669		
Health	804	853			535		
Infrastructure Development Roads and Transport	4 166	269 4 264			2 389 14 006		
Agriculture and Rural Development	4 100	49	317	2 108	19 469		
Social Development Community Safety		211 49			1 753 536		
Sport, Arts, Culture and Recreation		49		49	535		
Roads and Transport		49		49 49	1 071		
Local Government Total: Gauteng	36 655	37 578	317	2 502	536 44 210		
KWAZULU-NATAL	20.707	20.005			711		
Human Settlements Public Works	20 787	20 885 214			711 2 690		
Education	8 513	8 612		125	1 532		
Health Transport	2 883 6 803	2 981 13 246			4 493 113 287		
Arts and Culture		49		49	711		
Agriculture, Environmental Affairs and Rural Development Provincial Treasury		49 49	1 118	1 560	4 806 535		
Office of the Premier		7		49	536		
Social Development Total: KwaZulu-Natal	38 986	46 086	1 118	49 1 833	535 129 836		
LIMPOPO	30 700	40 000	1110	1 033	127 030		
Local Government and Housing Education	9 340 5 795	9 439 5 845		49 49	1 246 1 071		
Health and Social Development	1 481	1 530		49	1 071		
Public Works Roads and Transport	189 8 203	339 8 301		49	1 698 1 246		
Office of the Premier	8 203	49		49	1 071		
Economic Development, Environment and Tourism		49	1.025	1 084	1 071		
Agriculture Sport, Arts and Culture			1 035	1 084	535		
Total: Limpopo	25 008	25 552	1 035	1 429	9 545		
MPUMALANGA Public Works		99		49	1 070		
Roads and Transport	3 966	4 064		49	6 808		
Education Cooperative Governance and Traditional Affairs	2 363	2 412 49			536 536		
Agriculture, Rural Development and Land Administration			666	715	536		
Total: Mpumalanga NORTHERN CAPE	6 328	6 624	666	814	9 486		
Cooperative Governance, Human Settlements and Traditional Affairs	2 067	2 166			711		
Education Health	652 339	436 247			535 535		
Transport, Safety and Liaison	339	49			175		
Roads and Public Works	3 552	3 602		49	1 246		
Sport, Arts and Culture Agriculture, Land Reform and Rural Development		49	464	49 513	536 536		
Environment and Nature Conservation			-	49	536		
Total: Northern Cape NORTH WEST	6 611	6 548	464	661	4 810		
Human Settlements	10 416	10 465			536		
Public Works, Roads and Transport Education	4 223 2 653	4 272 2 702		107	1 163 536		
Sport, Arts and Culture	2 0.53	49		275	2 453		
Social Development Agriculture, Conservation, Environment and Rural Development		49	900	950	535 535		
Total: North West	17 292	17 538	900	1 332	5 758		
WESTERN CAPE							
Housing Transport and Public Works	14 137 2 913	14 235 3 958			711 18 718		
Environmental Affairs and Development Planning	27.3	]		192	1 552		
Agriculture Provincial Treasury			541	590 49	535 535		
Agriculture				49 49	536		
Total: Western Cape	17 049	18 193	541	881	22 587	207.000	401.500
Unallocated Grand Total	193 113	204 628	6 845	12 464	267 269	325 220 325 220	401 588 401 588

			Expand	led Public Wor	ks Programm	e Incentive G	rant for Munic	cipalities	
Cotogowy	Municipality	Eligibility	FTE		nal Financial			icipal Financi	al Year
Category	Municipality	Threshold	Performance	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14
			Target	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)	(R'000)
ASTERN CAP	Е								
BUF	Buffalo City	1 534	1 583	1 512			2 268		
BUF NMA	Nelson Mandela Bay	1 449	1 498	8 923			13 384		
EC101 EC102	Camdeboo Blue Crane Route		49 49	966 357			1 449 536		
EC102 EC103	Ikwezi		49	357			536		
EC104	Makana		49	357			536		
EC105	Ndlambe		49	357			536		
EC106	Sundays River Valley								
EC107	Baviaans		49	966			1 449		
EC108	Kouga		138	1 251			1 876		
EC109	Kou-Kamma		49	357			536		
DC10	Cacadu District Municipality		404	1000			2 151		
otal: Cacadu M	lunicipalities		481	4 968			7 454		
EC121	Mbhashe								
EC122	Mnquma		49	357			536		
EC123	Great Kei								
EC124	Amahlathi		49	609			914		
EC126	Ngqushwa								
EC127 EC128	Nkonkobe		49	357			536		
	Nxuba		340	7 902			11.704		
otal: Amathole	Amathole District Municipality  Municipalities		487	7 803 9 126			11 704 13 690		
otai. Amathole	iviumcipanues		467	9 120			13 090		
EC131	Inxuba Yethemba		49	357			536		
EC132	Tsolwana		49	357			536		
EC133	Inkwanca		49	357			536		
EC134	Lukanji		49	966			1 449		
EC135	Intsika Yethu		49	966			1 449		
EC136	Emalahleni		190	1 625			2 438		
EC137	Engcobo		49	609			914		
EC138	Sakhisizwe		49	357			536		
DC13	Chris Hani District Municipality		709	16 400			24 600		
otal: Chris Han	i Municipalities		1 242	21 994			32 994		
EC141	Elundini		49	357			536		
EC142	Senqu		49	966			1 449		
EC143	Maletswai		49	357			536		
EC144	Gariep		49	357			536		
DC14	Joe Gqabi District Municipality		213	3 476			5 214		
otal: Joe Gqabi	Municipalities		409	5 513			8 271		
EC153	Ngquza Hill		49	357			536		
EC154	Port St Johns		49	966			1 449		
EC155	Nyandeni		49	2 50			536		
EC156	Mhlontlo		49	966			1 449		
EC157	King Sabata Dalindyebo		49	609			914		
DC15	O R Tambo District Municipality		843	7 792			11 688		
otal: O.R.Tamb	oo Municipalities		1 088	10 690			16 572		
EC441	Matatiele		232	1 934			2 901		
EC441	Umzimvubu		49	966			1 449		
EC443	Mbizana		"	230			/		
EC152	Ntabankulu		49	966			1 449		
DC44	Alfred Nzo District Municipality		543	9 530			14 295		
otal: Alfred Nz	o Municipalities		873	13 396			20 094		
	27 11 114		7.00	74.11					
otai: Eastern C	ape Municipalities	2 983	7 661	76 122			114 727		

				Expan	ded Public Wor	ks Programm	e Incentive G	rant for Muni	cipalities	
		34	Eligibility	FTE	Natio	nal Financial	Year	Mur	icipal Financ	ial Year
Cat	tegory	Municipality	Threshold	Performance	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14
FREE S	STATE									
A	MAN	Mangaung	1 347	1 446	5 713			8 570		
В	FS161	Letsemeng		49	357			536		
В	FS162	Kopanong		49	357			536		
В	FS163	Mohokare		49	357			536		
В	FS171	Naledi		49	966			1 449		
C	DC16	Xhariep District Municipality		49	357			536		
Total: 2	Khariep M	Iunicipalities		245	2 394			3 593		
В	FS181	Masilonyana		49	357			536		
В	FS182	Tokologo		49	357			536		
В	FS183	Tswelopele		49	357			536		
В	FS184	Matjhabeng		49	3 335			5 002		
В	FS185	Nala		49	357			536		
C	DC18	Lejweleputswa District Municipality		49	966			1 449		
Total: I	Lejwelepu	tswa Municipalities		294	5 729			8 595		
В	FS191	Setsoto		49	7 095			10 642		
В	FS192	Dihlabeng		100	367			550		
В	FS193	Nketoana		49	357			536		
В	FS194	Maluti a Phofung		295	6 706			10 059		
В	FS195	Phumelela								
В	FS196	Mantsopa		49	357			536		
C	DC19	Thabo Mofutsanyana District Municipality		151	1 347			2 020		
Total:	Thabo Mo	futsanyana Municipalities		693	16 229			24 343		
В	FS201	Moqhaka		49	1 726			2 589		
В	FS203	Ngwathe		184	975			1 462		
В	FS204	Metsimaholo		49	357			536		
В	FS205	Mafube		49	357			536		
C	DC20	Fezile Dabi District Municipality		49	966			1 449		
Total: I	rezile Dab	i Municipalities	+	380	4 381			6 572		
Total: I	Free State	Municipalities	1 347	3 058	34 446			51 673		

				Expan	ded Public Wor	ks Programm	e Incentive G	rant for Munic	ipalities	
	-4	Municipality	Eligibility	FTE	Nation	al Financial	Year	Mun	icipal Financi	al Year
	ategory	wuntcipanty	Threshold	Performance	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14
GAU'	ГENG									
A	EKU	Ekurhuleni	3 980	4 729	6 222			9 333		
A	JHB	City of Johannesburg	4 417	15 298	191 011			286 581		
A	TSH	City of Tshwane	3 049	3 196	7 682			11 520		
B B B	GT421 GT422 GT423	Emfuleni Midvaal Lesedi		148 49	7 693 357			11 540 536		
C	DC42	Sedibeng District Municipality		49	357			536		
Total	Sedibeng 1	Municipalities		246	8 407			12 612		
B B B C Total	GT481 GT482 GT483 GT484 DC48	Mogale City Randfontein Westonaria Merafong City West Rand District Municipality d Municipalities		294 49 49 268 144 804	2 709 2 009 4 682 3 952 989 14 341			4 064 3 013 7 023 5 928 1 484 21 512		
Total	Gauteng N	Municipalities	11 446	24 273	227 663			341 558		

				Expan	ded Public Wor			rant for Munic	ripalities	
Cat	tegory	Municipality	Eligibility	FTE		nal Financial			icipal Financi	
	-	• •	Threshold	Performance	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14
WAZ	ULU-NAT	TAL								
	ETH	Ethekwini	4 730	8 425	72 084			108 126		
	LIII	Ethekwiii	4750	0 423	72 004			100 120		
		Vulamehlo								
3	KZN212	Umdoni Umzumbe								
3		UMuziwabantu								
3		Ezingoleni								
3		Hibiscus Coast		49	966			1 449		
				49	1 802			2 704		
otai: t	Jgu Munic	npanties		98	2 768			4 153		
3	KZN221	uMshwathi								
3	KZN222									
	KZN223									
	KZN224	Impendle Msunduzi	835	884	1 512			2 268		
		Mkhambathini	633	004	1 312			2 200		
		Richmond								
!		Umgungundlovu District Municipality								
otal: U	Jmgungun	dlovu Municipalities	835	884	1 512			2 268		
;	KZN232	Emnambithi/Ladysmith		49	357			536		
3	KZN232 KZN233			"	551			550		
3		Umtshezi								
3		Okhahlamba								
3		Imbabazane		402	11.152			16720		
		Uthukela District Municipality  Iunicipalities	+	482 531	11 153 11 510		-	16 729 17 265		
Judic	muncia M	numcipalities		331	11 510			1 / 203		
3	KZN241	Endumeni								
3	KZN242									
3	KZN244 KZN245									
		Umzinyathi District Municipality		343	3 537			5 306		
		i Municipalities		343	3 537			5 306		
3		Newcastle		49	966			1 449		
3		Emadlangeni Dannhauser								
		Amajuba District Municipality		49	966			1 449		
otal: A		Municipalities		98	1 932			2 898		
3 3	KZN261	eDumbe UPhongolo		49 49	357 357			536 536		
3		Abaqulusi		49	357			536		
3		Nongoma		49	357			536		
3	KZN266									
l latale 5		Zululand District Municipality  Municipalities		133 329	1 217 2 645			1 826 3 970		
otai: A	Zuiuiana iv	rumcipanties		329	2 043			3 970		
3	KZN271	Umhlabuyalingana								
3	KZN272	Jozini								
3		The Big 5 False Bay								
8	KZN274 KZN275	Hlabisa Mtubatuba								
		Umkhanyakude District Municipality		117	17 369			26 053		
		kude Municipalities		117	17 369			26 053		
3 3	KZN281									
5 5		uMhlathuze Ntambanana								
3	KZN284									
3	KZN285	Mthonjaneni								
3	KZN286									
otal: I		Uthungulu District Municipality  Municipalities		163 163	4 868 4 868		-	7 301 7 301		
otali (	Janunguiù	viuncipanues		103	4 808			/ 301		
3	KZN291	Mandeni								
;		KwaDukuza								
		Ndwedwe								
3		Maphumulo iLembe District Municipality		176	7 067			10 601		
		inicipalities		176	7 067			10 601		
				1.0						
3	KZN431									
3		Kwa Sani								
3		Greater Kokstad Ubuhlebezwe		49	357			536		
3 3		Ubuhlebezwe Umzimkhulu		49	357			536		
		Sisonke District Municipality		238	1 975			2 963		
2		unicipalities		336	2 689			4 035		
	oisonke Mit									
	sisonke Mi	•								

			Expan	ded Public Wor	ks Programm	e Incentive G	rant for Munic	ipalities	
	34	Eligibility	FTE		nal Financial			icipal Financi	al Year
Category	Municipality	Threshold	Performance	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14
LIMPOPO									
LIMPOPO									
B LIM331	Greater Giyani		49	357			536		
B LIM332	Greater Letaba		186	2 643			3 965		
B LIM333	Greater Tzaneen		49	911			1 367		
B LIM334	Ba-Phalaborwa		49	357			536		
	Maruleng		49	966			1 449		
C DC33 Total: Mopani M	Mopani District Municipality		242 <b>624</b>	7 746 12 980			11 620 19 473		
Total: Mopani M	unicipanties		024	12 980			194/3		
B LIM341	Musina								
B LIM342	Mutale		49	1 621			2 432		
B LIM343	Thulamela		49	2 291			3 437		
B LIM344	Makhado		49	609			914		
C DC34	Vhembe District Municipality		576	6 011			9 017		
Total: Vhembe M	Iunicipalities		723	10 532			15 800		
	Blouberg		116	483			725		
	Aganang		49	357			536		
	Molemole		701	10.244			15.266		
	Polokwane		781	10 244 966			15 366 1 449		
C DC35	Lepele-Nkumpi Capricorn District Municipality		504	4 269			6 404		
Total: Capricorn			1 450	16 319			24 480		
Total. Capricorn	Wunicipanities		1 430	10 317			24 400		
B LIM361	Thabazimbi		49	357			536		
	Lephalale		153	2 554			3 831		
	Mookgopong								
B LIM365	Modimolle		49	357			536		
B LIM366	Bela-Bela		49	357			536		
B LIM367	Mogalakwena		173	1 693			2 540		
C DC36	Waterberg District Municipality		49	966			1 449		
Total: Waterberg	Municipalities		522	6 284			9 428		
B LIM471	Ephraim Mogale		49	357			536		
	Elias Motsoaledi		49	2 661			3 991		
B LIM473	Makhuduthamaga		171	1 492			2 238		
	Fetakgomo		49	357			536		
	Greater Tubatse		142	1 284			1 926		
C DC47	Greater Tubasse Greater Sekhukhune District Municipality		366	12 333			18 499		
	ekhukhune Municipalities		826	18 484			27 726		
			020	101			<b>. 2</b> 0		
T. ( ) T	(F. 1.1.197		4	C4 =00			06.00=		
Total: Limpopo I	viunicipalities		4 145	64 599			96 907		

				Expan	ded Public Wor	ks Programm	e Incentive G	rant for Muni	cipalities	
		Mondalmalita	Eligibility	FTE	Natio	nal Financial	Year	Mun	icipal Financi	al Year
	ategory	Municipality	Threshold	Performance	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14
MPUN	MALANGA									
В	MP301	Albert Luthuli		49	966			1 449		
В	MP302	Msukaligwa		49	357			536		
В	MP303	Mkhondo		49	966			1 449		
В	MP304	Pixley Ka Seme		49	9 477			14 215		
В	MP305	Lekwa		49	966			1 449		
В	MP306	Dipaleseng		49	966			1 449		
В	MP307	Govan Mbeki		49	1 574			2 361		
C	DC30	Gert Sibande District Municipality		104	3 420			5 131		
Total:	Gert Sibar	de Municipalities		447	18 692			28 039		
В	MP311	Victor Khanye		49	357			536		
В	MP312	Emalahleni		49	966			1 449		
В	MP313	Steve Tshwete		139	1 259			1 888		
В	MP314	Emakhazeni		49	357			536		
В	MP315	Thembisile		49	966			1 449		
В	MP316	Dr JS Moroka		106	1 788			2 683		
C	DC31	Nkangala District Municipality		122	3 242			4 863		
Total:	Nkangala	Municipalities		563	8 935			13 404		
В	MP321	Thaba Chweu		49	966			1 449		
B	MP321	Mbombela		109	1 911			2 867		
B	MP323	Umjindi		49	357			536		
В	MP324	Nkomazi		116	1 092			1 638		
В	MP325	Bushbuckridge		138	1 251			1 876		
C	DC32	Ehlanzeni District Municipality		127	563			844		
Total:		Municipalities		588	6 140			9 210		
zoui.	Zammedii			366	0 140			7 210		
Total:	Mnumalar	nga Municipalities	1	1 598	33 767			50 653		

			Expan	ded Public Wor	ks Programme	Incentive G	rant for Munic	rinalities	
		Eligibility	FTE		nal Financial Y			icipal Financi	al Year
Category	Municipality	Threshold	Performance	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14
NORTHERN	CAPE								
B NC0	61 Richtersveld								
B NC0									
B NC0									
B NC0									
B NC0									
B NC0									
C DC			49	4 405			6 608		
e	kwa Municipalities		49	4 405			6 608		
Total. Namar	xwa Municipanties		42	4 403			0 000		
B NC0	71 Ubuntu		49	357			536		
B NC0			42	337			330		
B NC0			49	357			536		
B NC0	3		42	337			330		
B NC0			49	966			1 449		
B NC0			1	700			1 447		
B NC0			49	966			1 449		
B NC0			42	900			1 449		
C DC									
	Ka Seme Municipalities		196	2 646			3 970		
Total. I ixiey	Ka Seme Municipanues		170	2 040			3770		
B NC0	81 Mier								
B NC0									
B NC0									
B NC0									
B NC0			49	357			536		
B NC0			1	331			550		
C DC			49	8 888			13 332		
	la Municipalities		98	9 245			13 868		
Total: Blyand	a Municipanties		70	, 243			13 000		
B NC0	91 Sol Plaatje		992	16 087			24 130		
B NC0			"	10 007			27 130		
B NC0	6 6								
B NC0			49	357			536		
C DC			49	6 588			9 882		
	es Baard Municipalities		1 090	23 032			34 548		
			2 070				2.510		
B NC4	51 Moshaweng		49	357			536		
B NC4			49	966			1 449		
B NC4				,,,,			• • • • •		
C DC4			49	7 113			10 669		
	Taolo Gaetsewe Municipalities		147	8 436			12 654		
	<b>-</b>								
Total: North	ern Cape Municipalities	+	1 580	47 764			71 648		
rotar. North	ли Сарс плинстраниез		1 500	47 704			/1040		

			Expan	ded Public Wor	ks Programm	e Incentive G	rant for Munic	cipalities	-
G-4	3.6 (2.2 (	Eligibility	FTE		nal Financial			icipal Financi	al Year
Category	Municipality	Threshold	Performance	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14
NORTH WEST									
B NW371	Moretele		208	1 759			2 638		
B NW372	Madibeng		146	3 417			5 125		
B NW373	Rustenburg		507	8 315			12 473		
3 NW374	Kgetlengrivier		49	357			536		
3 NW375	Moses Kotane		137	1 242			1 863		
C DC37	Bojanala Platinum District Municipality		145	1 301			1 952		
Total: Bojanala	Platinum Municipalities		1 192	16 391			24 587		
3 NW381	Ratlou		49	609			914		
NW382	Tswaing		49	966			1 449		
NW383	Mafikeng		49	966			1 449		
NW384	Ditsobotla		49	966			1 449		
NW385	Ramotshere Moiloa		49	609			914		ĺ
DC38	Ngaka Modiri Molema District Municipality		49	1 955			2 932		
	odiri Molema Municipalities		294	6 071			9 107		
NW392	Naledi		49	357			536		
NW393	Mamusa		49	357			536		
NW394	Greater Taung		49	966			1 449		
NW396	Lekwa-Teemane		49	357			536		
NW397	NW397		49	357			536		
DC39	Dr Ruth Segomotsi Mompati District Municipality		372	4 765			7 147		
otal: Dr Ruth S	egomotsi Mompati Municipalities		617	7 159			10 740		
NW401	V		40	966			1 449		
NW401 NW402	Ventersdorp Tlokwe		49 49	966			1 449		
NW402	City of Matlosana		198	7 334			11 002		
NW403	Maquassi Hills		49	966			1 449		İ
DC40	Dr Kenneth Kaunda District Municipality		49	357			536		ĺ
	th Kaunda Municipalities		394	10 589			15 885		
otal. Di Kelille	ui Kaunua Municipanues		394	10 309			15 665		
		1							
otal: North We	st Municipalities		2 497	40 210	l	1	60 319		ĺ

				Expand	led Public Wor	ks Programm	e Incentive G	rant for Munic	rinalities	
			Eligibility	FTE		nal Financial			icipal Financi	al Year
Categ	gory	Municipality	Threshold	Performance	2011/12	2012/13	2013/14	2011/12	2012/13	2013/14
WESTER	N CAP	E								
A	CPT	City of Cape Town	3 046	3 095	16 989			25 484		
B W	VC011	Matzikama		49	357			536		
	VC012	Cederberg		49	357			536		
	VC013	Bergrivier		49	357			536		
	VC014	Saldanha Bay								
	VC015	Swartland		40	255			50.5		
	DC1	West Coast District Municipality at Municipalities		49 <b>196</b>	357 1 428			536 2 144		
Total: we	est Coas	t Municipanties		196	1 428			2 144		
B W	VC022	Witzenberg								
	VC022	Drakenstein		49	357			536		
	VC024	Stellenbosch		"	357			550		
	VC025	Breede Valley		49	357			536		
B W	VC026	Langeberg								
	DC2	Cape Winelands District Municipality								
Total: Caj	pe Wine	elands Municipalities		98	714			1 072		
	VC031	Theewaterskloof		49	1 134			1 701		
	VC032	Overstrand		176	357			536		
	VC033	Cape Agulhas		49	357			536		
	VC034	Swellendam		49	357			536		
	DC3	Overberg District Municipality		49 372	357 2 562			536 3 845		
Total: Ove	erberg l	Municipalities		3/2	2 562			3 845		
B W	VC041	Kannaland		49	357			536		
	VC042	Hessequa		49	357			536		
	VC043	Mossel Bay		49	609			914		
B W	VC044	George		49	1 585			2 378		
B W	VC045	Oudtshoorn		153	1 359			2 038		
B W	VC047	Bitou		49	357			536		
	VC048	Knysna								
	DC4	Eden District Municipality								
Total: Ede	en Mun	icipalities		398	4 624			6 938		
		Laingsburg								
	VC052			40	257					
	VC053 DC5	Beaufort West Central Karoo District Municipality		49 49	357 357			3 267		
		aroo Municipalities		98	714	1		3 267		
Iotal. Cel	iiii ai Ki	по плинирание		76	/14			3 207		
Total: We	estern C	ape Municipalities	3 046	4 257	27 031			42 750		
Unallocate	ed:									
NT. 41	T. 4. 1		2420	60.550	(FO FO2			1.022.255		
National T	ı otal		24 387	60 569	679 583			1 022 211		

APPENDIX TO SCHEDULE 4: FORMAT OF INFRASTRUCTURE PROJECTS LIST FOR PROVINCES THAT RECEIVE THE EDUCATION INFRASTRUCTURE GRANT, THE HEALTH INFRASTRUCTURE GRANT AND THE PROVINCIAL ROADS MAINTENANCE GRANT AS REQUIRED BY SECTION 9(2)(d) OF THE BILL

APPENDIX TO SCHEDULE 4: FORMAT OF INFRASTRUCTURE PROJECTS LIST FOR PROVINCES THAT RECEIVE THE EDUCATION INFRASTRUCTURE GRANT, THE HEALTH INFRASTRUCTURE GRANT AND THE PROVINCIAL ROADS MAINTENANCE GRANT AS REQUIRED BY SECTION 9(2)(d) OF THE BILL

No. Project Municipality/ name Region	Project name	Municipality/ Region	Type of infrastructure	ucture	Project duration		Budget programme	EPWP budget for current	Total project cost		Total available	Forward estimates	stimates
			School (Primary/secondary/specialised; admin block; water; electricity; sanitation/tollet; fencing, etc.)	Units (i.e. number of classrooms or facilities or square meters)	Date: Start	Date: Finish	name	financial year		previous years	2011/12	2012/13	2013/14
R thousands													
New and replacement assets													
: =													
Total: New infrastructure assets													
2. Upgrades and additions													
:: <b>c</b>													
Total: Upgrades and additions													
Rehabilitation, renovations and refurbishments        n	shments												
Total: Rehabilitation, renovations and refurbishments	ırbishments												
Maintenance and repairs   n													
Total: Maintenance and repairs				-									
5. Infrastructure transfers - current													
Total: Infrastructure transfers - current													
6. Infrastructure transfers - capital 1													
Total: Infrastructure transfers - capital													
Grand Total: Education Infrastructure Grant	±.												

APPENDIX TO SCHEDULE 4: FORMAT OF INFRASTRUCTURE PROJECTS LIST FOR PROVINCES THAT RECEIVE THE EDUCATION INFRASTRUCTURE GRANT, THE HEALTH INFRASTRUCTURE GRANT AND THE PROVINCIAL ROADS MAINTENANCE GRANT AS REQUIRED BY SECTION 9(2)(d) OF THE BILL

Health Infrastructure Grant: List of Infrastructure Projects  No. Project Municipality	of Infrastruc Project	ture Projects Municipality/	Type of infrastructure	cture	Project duration	ntion	Budget	EPWP budget	Total project	Expenditure to	Total available	Forward estimates	timates
	name	Region					programme	for current	cost	date from			
			Hospital (Regional/district/central; clinic; community health centre; pharmaceutical depots; mortuary, etc.)	Units (i.e. number of beds or facilities)	Date: Start	Date: Finish	e uga	inancial year		previous years	2011/12	2012/13	2013/14
R thousands													
1. New and replacement assets													
- ; c													
Total: New infrastructure assets													
2. Upgrades and additions													
<b>-</b> :													
u													
Total: Upgrades and additions													
3. Rehabilitation, renovations and refurbishments	bishments												
- ¦ s													
Total: Rehabilitation, renovations and refurbishments	efurbishments												
4. Maintenance and repairs													
:: =													
Total: Maintenance and repairs													
5. Infrastructure transfers - current													
· ; c													
Total: Infrastructure transfers - current													
6. Infrastructure transfers - capital													
- ::													
u												1	
Total: Infrastructure transfers - capital													
Grand Total: Health Infrastructure Grant													

# APPENDIX TO SCHEDULE 4: FORMAT OF INFRASTRUCTURE PROJECTS LIST FOR PROVINCES THAT RECEIVE THE EDUCATION INFRASTRUCTURE GRANT, THE HEALTH INFRASTRUCTURE GRANT AND THE PROVINCIAL ROADS MAINTENANCE GRANT AS REQUIRED BY SECTION 9(2)(d) OF THE BILL

Provincial Roads Maintenance Grant: List of Infrastructure Projects	Grant: List of	Infrastructure Pi											
No.	Project name	Municipality/ Region	Type of infrastructure	ture	Project duration	uration	Budget programme	EPWP budget for current	Total project cost		Total available	Forward estimates	stimates
			Surfaced; gravel (include earth and access roads); public transport; bridges; damage structures, etc.	Units (i.e. number of kilometres/square metres/facilities)	Date: Start	Date: Finish	name	financial year		previous years -	2011/12	2012/13	2013/14
R thousands													
1. New and replacement assets													
- : :													
Total: New infrastructure assets													
2. Upgrades and additions													
-													
: =													
Total: Upgrades and additions													
3. Rehabilitation, renovations and refurbishments	urbishments												
<b>⊢</b> ; c													
Total: Rehabilitation, renovations and refurbishments	refurbishments												
4. Maintenance and repairs													
<b>⊢</b> ; c													
Total: Maintenance and repairs													
5. Infrastructure transfers - current													
- ; =													
Total: Infrastructure transfers - current	ı,												
6. Infrastructure transfers - capital													
<b>-</b> ∶													
u													
Total: Infrastructure transfers - capital	_												
Grand Total: Provincial Roads Maintenance Grant	nance Grant												